

University of Huddersfield Repository

Masoud, Najeb M. H.

Libya's economic reform programme and the case for a stock market

Original Citation

Masoud, Najeb M. H. (2009) Libya's economic reform programme and the case for a stock market. Doctoral thesis, University of Huddersfield.

This version is available at https://eprints.hud.ac.uk/id/eprint/9062/

The University Repository is a digital collection of the research output of the University, available on Open Access. Copyright and Moral Rights for the items on this site are retained by the individual author and/or other copyright owners. Users may access full items free of charge; copies of full text items generally can be reproduced, displayed or performed and given to third parties in any format or medium for personal research or study, educational or not-for-profit purposes without prior permission or charge, provided:

- The authors, title and full bibliographic details is credited in any copy;
- A hyperlink and/or URL is included for the original metadata page; and
- The content is not changed in any way.

For more information, including our policy and submission procedure, please contact the Repository Team at: E.mailbox@hud.ac.uk.

http://eprints.hud.ac.uk/

LIBYA'S ECONOMIC REFORM PROGRAMME

AND THE CASE FOR A STOCK MARKET

Najeb M. H. Masoud

A thesis submitted to the University of Huddersfield in partial fulfilment of the requirements for the degree of Doctor of Philosophy

The University of Huddersfield Business School

June 2009

ABSTRACT

Libya is still in the early stages of its financial liberalisation and reform following eleven years of political chaos and nearly three decades of central planning control. However, it is advancing as a result of the removal of UN and US sanctions during the last few years, and there are signs of rapid development. Despite these advancements, no study has been found which explores the readiness of the Libyan financial market for the establishment of a stock market. This thesis was undertaken to develop a conceptual framework for a research model with a specific focus on the Libyan economic reform programme and the development of the Libyan stock market between 1999 and 2008.

The empirical study investigates the determinants of economic reform and stock market performance within the Libyan economy utilising data from three different sources and a multi-method approach. Self-administered questionnaires were distributed to the entire target population of the Libyan financial market, banking sector and a number of companies. A total of 330 questionnaires were distributed and of these, 203 were returned completed and usable, a response a rate of 61.5 *per cent*. Fourteen semi-structured interviews were held with managers in a subset of companies, selected via a stratified sample of respondents to the self-administered questionnaires. The third method of data collection used financial market data over the period 1995-2006 from 42 emerging market countries. This data was analysed to examine whether best practice from emerging stock markets is transferable to the Libyan context. As a result, this study provides some knowledge that might usefully be generalised to other developing countries, particularly to those with a similar economic structure.

The primary contribution of this study lies in the fact that it is the first attempt to study the impact of stock market development on the economic growth process of a specific-country experience and evaluates the success of the economic reform programme and Libya's readiness to complete its transition to a market-based economy. The key findings are; first, the economic reform programme variables have an impact upon various features of the stock market performance variables within a linear regression model; second, stock market development has a significant effect on economic growth, and this effect remains strong even after controlling for banking sector and other control variables using a growth model; third, although the evidence largely supports the view that there is a stable, long-term equilibrium relationship between the evolution of the stock market and the evolution of the economy, it provides no support for the view that the stock market is a leading sector in the process of Libya's economic development. The evidence supports the view that the relation between stock market development and economic growth in emerging economies is bi-directional. The findings describe that the stock market and the banking sector in Libya in particular and emerging economy in general are complementary rather than substitutes in providing financial services to the economy.

This study seeks to make an original contribution to knowledge on the academic and practical levels as one of the first attempts at empirically investigating the impact of an economic reform programme on stock market performance in an emerging economy. The research represents an applied study of a type that has not appeared elsewhere, and the framework offered may therefore not only be appropriate to Libya as a case study, but also to other countries in similar circumstances. The research provides an important introduction to this area and has attempted to explore its significance for both the economy and business. This research adds to the existing body of literature regarding development and application of a series of models of economic reform programmes, stock market performance and economic growth in a developing country. Additionally, brief recommendations are offered regarding potential useful directions for future research arising from the conclusions of this research. These develop into a strategic framework for the improvement of an economic reform programme and stock market performance.

DECLARATION

I, Najeb Masoud, declare that the thesis entitled:

LIBYA'S ECONOMIC REFORM PROGRAMME AND THE CASE FOR A STOCK MARKET

To the best of my knowledge and belief, contains no material previously published or written by another person except where due reference is made in the thesis itself and the work presented in it is my own. I confirm that:

- This work was done wholly or mainly while in candidature for a research degree at Huddersfield University,
- This work has not previously been submitted for a degree or any other qualification at this University or any other institution of learning,
- Where I have quoted from the work of others, the source is always given; with the exception of such quotations, this thesis is entirely my own work,
- I have acknowledged all main sources of help,
- None of this work has been published before submission.

Signed Najeb Masoud Dated 24 June 2009

DEDICATION

In the name of Allah the All-Merciful, the All-Compassionate



(إن أريدُ إلَّا ٱلإصلاح مَا ٱستَطعْت أَومَا تَوفِيقِي إلَّا بِٱللَّهِ عَليهِ تَوكَلت وَإليهِ أنيب) (هود: 88)

(I only desire reform to the best of my power. And my guidance cannot come except from Allah, in Him I trust and unto Him I turn repent) (The Holy Quran, Sura/Hud, Verse 88)

This thesis is dedicated to my own dear family; to my precious Mother *Saide*, and to my Father *Muhammed*, for their constant support, interest and encouragement. And special dedication must be made to my lovely wife *Asma*, and our son *Muhammed* and beautiful daughter *Yaqeen*, to my dear sisters, brothers and their families, for their continuous prayers and encouragement. I am also indebted to my brother *Abdellah* for his support throughout the course of this study.

ACKNOWLEDGEMENTS

In the name of Allah the All-Merciful, the All-Compassionate

"Who does not thank people does not thank Allah" (*Prophet Mohammed peace be upon him*)

First, and most significantly, my deepest gratitude and appreciation go to almighty ALLAH (God), the merciful, the compassionate and my saviour and source of hope. It is he who guided me to every success that I have experienced throughout my life. I would also like to present my thanks to a number of individuals and organisations for their invaluable contribution to this thesis. I would like to express my sincere gratitude to my director of studies, Dr. John Cook and to John Day at the University of Huddersfield (Business School) for guidance, encouragement, invaluable comments, advice and criticism, which have assisted immensely in the shaping and completion of my PhD studies; the best I can offer is my heartfelt thanks. My sincere appreciation goes, too, to my former third supervisor, Dr. Vincent Gerard Fitzsimons for his early guidance and valuable advice. Also, special thanks to my dissertation examiners, Prof. Chris Cowton Dean of Business School at the University of Huddersfield and Dr. Roman Matousek, Department of Economics, Finance and International Business at the London Metropolitan University for their valuable suggestions and comments. I would like to express my sincere appreciation to Prof. Glenn Hardaker, Director of Research in the Business School, who contributed time and effort besides providing suggestions regarding the research model and technical analysis. My appreciation is also extended to all members of the library staff who contributed to this research. It would be an unpardonable omission, meanwhile, not to thank the ladies of the Business School's administration at the University of Huddersfield.

I wish to acknowledge my appreciation of all those who contributed to this research through their time and effort, in particular experts: Prof. Ross Levine in the Department of Economics and Finance at Brown University, United States, for his early notes in writing on the proposal for this research and the recommendation of several relevant articles from his own website; Prof. Mohammed Omran, Arab Academy for Science and Technology in Alexandria, Egypt and member of staff at the Arab Monetary Fund in United Arab Emirates, for his many comments and suggestions regarding early notes in writing concerning the proposal and research questionnaire. I wish to express deepest thanks to all members of the academic staff in the Department of Economics at the Higher Academic Studies, Tripoli, Libya. Very special thanks go to Dr. Frhat Shrnan (my past MSc supervisor) for his considerable help, generous support, criticism, encouragement and friendship which I have received while preparing the thesis. I would like to acknowledge Dr. Noury Baruon for his encouragement and useful feedback on drafts of the research questionnaire. Great thanks are due to Prof. Amar Allmquery for his support and encouragement. My appreciation is also extended to Mohamed Sugheiar Head of Business School at Elmergib University, Libya, for his encouragement and friendship. My special gratitude goes to Mr. Mohamed Sugheiar of the oil company in Tripoli for his generous continuous support from the initial period when I arrived in the UK and then throughout my PhD; his unending support, phone calls and encouragement are greatly valued.

To the many respondents from the Libyan financial market and companies who generously contributed their time and expertise and made this research possible, for their co-operation, insights and assistance: thank you. I hope that Libyan respondents regard their efforts as worthwhile. My thanks are also due to my friends, in Libya and the UK, for their support, understanding and encouragement throughout the study period. Therefore, I wish to thank all the people who created and added to my knowledge.

LIST OF CONTENTS

LIBYA'S ECONOMIC REFORM PROGRAMME AND THE CASE FOR A STOCK MARKET	1
ABSTRACT	2
DECLARATION	
DEDICATION	4
ACKNOWLEDGEMENTS	5
LIST OF CONTENTS	6
LIST OF TABLES	
LIST OF FIGURES	
LIST OF APPENDICES	17
LIST OF ACRONYMS AND ABBREVIATIONS	

Chapter One Introduction

1.1 1.2 1.3 1.3.1 1.3.2 1.3.3 1.4 1.5 1.6 1.7

Chapter Two

An Overview of Libya's Economy Structural Changes and Development Patterns

2.1	Introdu	ction			
2.2	Historic	Backgrou	nd in Libya		
2.2.1	Geogra	phical and	Population		
2.2.2	Politica	l and Econ	omic		
2.3	Backgro	ound to the	Libyan Ecor	omic Reform Programme Period	41
2.3.1	Lessons	and Chall	enge		
		2.3.1.1	Acceleration	of the Transition to a Market Economy	
			2.3.1.1.1	Deregulation Programme	43
			2.3.1.1.2	Privatisation Programme	45
			2.3.1.1.3	Liberalisation Programme	
	2.3.2	Macro-ec	onomic Refor	m Performance	
	2.3.3	The Resu	lts of Econom	ic Reform Programme	
	2.3.4	Foreign D	Direct Investm	ent in the Libyan Economy	49
2.4	Financia	al System.			
	2.4.1	Developn	nent of the Ba	nking System	

	2.4.2	2 Monetary Policy		
	2.4.3	Initiative	es for a Libyan Stock Market	53
		2.4.3.1	Umbrella: Libya's Stock Market under the Economic Reform Programme	
		2.4.3.2	Establishment of an Organised Stock Market in Libya	54
		2.4.3.3	The Stock Market Structure	
		2.4.3.4	The Main Obstacles to a Sound and Efficient Stock Market in Libya	
2.5	Summ	ary and Co	onclusion	

Chapter Three

The Economic Reform Programme

3.1	Introdu	uction	60				
3.2	Defini	Definitions of Economic Reform					
3.3	The In	portance, Aims and Objectives of Economic Reform Programmes	63				
3.4	Assess	ing the Economic Reform Programme Variables and Macro-economic Reform Variables	64				
3.5	The Pr	rocess of Economic Reform	65				
3.6	The Ec	conomic Reform Programme Experiments	66				
	3.6.1	Central and East European Countries Experiment	66				
	3.6.2	China Experiment	68				
	3.6.3	India Experiment	69				
	3.6.4	Latin American Experiment	70				
	3.6.5	Middle Eastern and North African Countries Experiment	72				
3.7	Compa	arison between Experiment Countries	73				
3.8	Summ	ary and Conclusion	75				

Chapter Four

Economic Growth Theory

4.1	Introduction	.77
4.2	Neoclassical Growth Theory	.78
4.3	New Endogenous Growth Theory and the Macro-determinants of Growth	.81
4.4	How Stock Markets Affect Economic Growth	. 87
4.5	Summary and Conclusion	.92

Chapter Five Emerging Stock Markets

5.1	Introdu	ction	94	
5.2	Definiti	on of a Stock Market	96	
5.3	The Fu	nction of Stock Markets	97	
5.4	The Imp	portance of Stock Markets		
5.5	Emergi	ng Stock Markets in Developing Countries		
5.6	.6 Measuring Stock Market Variables			
	5.6.1	Market Size	103	
	5.6.2	Market Liquidity	103	
	5.6.3	Market Activity		
	5.6.4	Market Efficiency	105	

	5.6.5	Market Volatility	106
5.7	The M	easurement of Financial Maturity	
5.8	Financi	ial Crisis in Developed and Emerging Markets	
5.9	The Re	elationship between Economic Reform Programmes and Stock Market Performance	113
	5.9.1	Relationship between Deregulation and Stock Market Performance	113
	5.9.2	Relationship between Privatisation and Stock Market Performance	114
5.10	The Re	elationship between Macro-economic Reform and Stock Market Performance	115
	5.10.1	Relationship between Interest Rates and Stock Market Performance	116
	5.10.2	Relationship between Inflation Rates and Stock Market Performance	117
	5.10.3	Relationship between Exchange Rates and Stock Market Performance	118
	5.10.4	Relationship between Budget Deficits and Stock Market Performance	119
	5.10.5	Relationship between Per-capita Income and Stock Market Performance	120
	5.10.6	Relationship between GDP Growth Rate and Stock Market Performance	120
5.11	Summa	ary and Conclusion	121

Chapter Six Research Methodology

6.1	Introdu	roduction				
6.2	Resear					
6.3	Case S	tudy and J	ustifications fo	r its Use		
6.4	Resear	ch Philoso	phy			
6.5	Justific	cation for t	he Research Pl	nilosophy		
6.6	Resear	ch Framev	vork			
	6.6.1	Building	the Research I	Framework		
	6.6.2	The Prin	hary Research	Model		
	6.6.3	Backgro	und Statistical	Technical Growth Model		
	6.6.4	Backgro	und Quantitativ	ve and Qualitative Data Analysis		
6.7	Resear	ch Design	- Survey Strate	egy		
6.8	Data C	ollection N	Aethods			
	6.8.1	Secondar	ry Data	y Data		
	6.8.2	Primary	Data			
		6.8.2.1	Self-Adminis	stered Questionnaire		
			6.8.2.1.1	Population and Samples		
			6.8.2.1.2	Designing and Developing the Questionnaire		
			6.8.2.1.2.1	Planning what to Measure		
			6.8.2.1.2.2	Formatting the Questions		
			6.8.2.1.2.3	Question Wording and Layout Decisions		
			6.8.2.1.2.4	Translation of the Research Questionnaire		
			6.8.2.1.2.5	Pilot Testing the Questionnaire		
			6.8.2.1.2.6	Fieldwork		
			6.8.2.1.3	Response Rate		
		6.8.2.2	Semi-structur	red Interviews		
			6.8.2.2.1	Pre-testing and Piloting the Interview		

			6.8.2.2.2	Interview Sample	151
			6.8.2.2.3	Interview Process	
6.9	Validit	y and Relia	ability		
	6.9.2	Validity.	•••••		
	6.9.3	Reliabilit	ty		
6.10	Data A	nalysis Sta	atistics		
	6.10.1	Quantitat	tive Data Analy	zsis	
		6.10.1.1	Inferential an	d Differences Statistics Test	
			6.10.1.1.1	Descriptive Statistics	
			6.10.1.1.2	Kurtosis and Skewness	
		6.10.1.2	The Statistica	l Techniques used to Test the Research Hypotheses	
			6.10.1.2.1	Linear Multiple Regression Models	
			6.10.1.2.1.1	Multiple Regression Analysis Model	
			6.10.1.2.1.2	Correlation Statistics	
			6.10.1.2.1.3	Chi-square-based Measures	
			6.10.1.2.1.4	Statistical Significance	
			6.10.1.2.2	Econometric Growth Model	
			6.10.1.2.2.1	Two-Stage Least-Squares (2SLS)	
	6.10.2	Qualitati	ve Data Analys	is	
6.11	Summa	ary and Co	nclusion		

Chapter Seven

Presentation of the Findings

7.1	Introdu	uction				
7.2	Data S	Data Sources and Analysis				
	7.2.1	Primary	Data Sources (Self-administered Questionnaires and Semi-structured Interv	iews) 167		
	7.2.2	Data An	alysis			
7.3	Genera	al Demogra	aphic Analysis			
	7.3.1	Sample	Description Characteristics			
	7.3.2	Characte	eristics of Respondents			
	7.3.3	The Characteristics of Participants' Companies				
7.4	Practic	es of the L	Libyan Economic Characteristics			
	7.4.1	Econom	ic Reform Programme Independent Variables			
		7.4.1.1	Deregulation			
		7.4.1.2	Corporatisation	174		
		7.4.1.3	Privatisation			
		7.4.1.4	Liberalisation			
	7.4.2	Macro-e	conomic Reform Independent Variables			
		7.4.2.1	Interest Rate			
		7.4.2.2	Exchange Rate			
		7.4.2.3	Inflation Rate			
		7.4.2.4	Budget Deficit			
		7.4.2.5	Per-capita Income			
		7.4.2.6	Real GDP Growth Rate			

7.5

7.4.3	Financia	l Maturity Independent Variables	192
	7.4.3.1	Banking Assets Ratio	192
	7.4.3.2	Domestic Credit Ratio	194
	7.4.3.3	Stock Market Capitalisation Ratio	196
7.4.4	Stock M	arket Performance Dependent Variables	197
	7.4.4.1	Market Size	198
	7.4.4.2	Market Liquidity	199
	7.4.4.3	Market Activity	201
	7.4.4.4	Market Concentration	203
Summa	ary and Co	nclusion	205

Chapter Eight Test of Research Hypotheses

8.1	Introdu	uction		207			
8.2	Descri	ptive Statis	stics for Research Variables used in Testing the Hypotheses				
8.3	Structural Multiple Regression Model						
	8.3.1	The Sim	Simple Regression Linear Model				
	8.3.2	The Cor	relation Matrix				
8.4	Result	s from Mo	delling Economic Reform Programme	213			
	8.4.1	Testing of Stock M	of Hypothesis Theorising the Relationship between Economic Reform Progra arket Performance	amme and			
		8.4.1.1	Correlation Matrix Statistics	214			
		8.4.1.2	Overall Fit of the Summary Model	214			
		8.4.1.3	Model Parameters				
	8.4.2	Subsidia	ry Hypotheses				
		8.4.2.1	Deregulation				
		8.4.2.2	Corporatisation				
		8.4.2.3	Privatisation				
		8.4.2.4	Liberalisation				
	8.4.3	Discussi	on: Findings from Testing the Economic Reform Programme Hypotheses				
8.5	Result	s from Mo	delling Macro-economic Reform				
	8.5.1	Testing of Market I	of Hypothesis Theorising the Relationship between Macro-economic Reform Performance	and Stock			
		8.5.1.1	Correlation Matrix Statistics				
		8.5.1.2	Overall Fit of the Summary Model				
		8.5.1.3	Model Parameters				
	8.5.2	Subsidia	ry Hypotheses				
		8.5.2.1	Interest Rate				
		8.5.2.2	Exchange Rate				
		8.5.2.3	Inflation Rate	232			
		8.5.2.4	Budget Deficit	233			
		8.5.2.5	Per-capita Income	235			
		8.5.2.6	GDP Growth Rate	236			
	8.5.3	Discussi	on: Findings from Testing the Macro-economic Reform Hypotheses				

LIST OF CONTENTS

8.6	Result	s from Mo	delling Financial Maturity	
	8.6.1	Testing of Performation	of Hypothesis Theorising the Relationship between Financial Maturity a ance	und Stock Market
		8.6.1.1	Correlation Matrix Statistics	
		8.6.1.2	Overall Fit of the Summary Model	
		8.6.1.3	Model Parameters	
	8.6.2	Subsidia	ry Hypotheses	
		8.6.2.1	Banking Assets Ratio	
		8.6.2.2	Domestic Credit Ratio	
		8.6.2.3	Stock Market Capitalisation Ratio	
	8.6.3	Discussi	on: Findings from Testing the Financial Maturity Hypotheses	
8.7	Summ	ary and Co	onclusion	

Chapter Nine

An Empirical Investigation of Stock Market Performance in Emerging Market Countries

9.1	Introdu	ction		250
9.2	MENA	Within the	e Context of Emerging Markets	
9.3	Empiric	cal Model	and Variable Measure	
	9.3.1	Empirica	l Model	
	9.3.2	Variable	Measure	
		9.3.2.1	The Stock Market Development Variables	
		9.3.2.2	The Banking Sector Development Variables	
		9.3.2.3	Macro-economic Reform Variables	
		9.3.2.4	The Conditioning Information Set Variables	
9.4	Data an	d Summar	y Statistical Analysis	
	9.4.1	The Sam	ple	
	9.4.2	Summary	/ Statistical Analysis	
9.5	Analyti	cal Techni	que	
9.6	Empiric	cal Results	and Analysis	
9.7	Determ	inants of E	Banking and Financial Market Development Indicators	
9.8	Summa	ry and Co	nclusion	

Chapter Ten Conclusions, Contribution and Implications

10.1	Introdu	ction	292
10.2	Overvie	ew of the Current Research	292
	10.2.1	The Theoretical Part of the Study	293
	10.2.2	The Methodological and Methodical Part of the Study	296
	10.2.3	The Empirical Part of the Study	296
10.3	The Ke	y Findings	298
	10.3.1	The Relationship between Economic Reform Programme and Stock Market Performance	299
	10.3.2	The Relationship between Macro-economic Reform and Stock Market Performance	300
	10.3.3	The Relationship between Financial Maturity and Stock Market Performance	301
	10.3.4	Comparison between Emerging Countries and the Libyan Stock Market	302

LIST OF CONTENTS

10.4	Contrib	ution to Knowledge	304
	10.4.1	Theoretically	304
	10.4.2	Empirically	305
	10.4.3	Methodologically	306
10.5	The Imp	plication of the Study for Policy Reform	307
10.6	Limitati	ions of the Research	312
	10.6.1	Theoretically	312
	10.6.2	Empirically	313
	10.6.3	Methodologically	313
10.7	Outline	for Future Research	314
	10.7.1	Research Analysis	314
	10.7.2	Economic Reform Programme	315
	10.7.3	Macro-economic Reform	315
	10.7.4	Financial Maturity	315
	10.7.5	Stock Market Performance	315
	10.7.6	Comparison between Emerging Countries and the Libyan Stock Market	316
REFE	RENCES	5	319
APPE	NDICES		354

LIST OF TABLES

Table 1.1	Research Methodology and Objectives	36
Table 2.1	Transfer of Ownership of a Number of Factories and Production Units, 1987-1992	45
Table 2.2	Number and Classification of Units that will be Transformed between 2004 and 2008	46
Table 2.3	Measuring the Macro-economic Reform Programme in Libya, 1999-2004	48
Table 2.4	Development in the Size of FDI in Economic Sectors, 2000-2005	50
Table 2.5	Number of Libyan Commercial Banks' Branches and Agencies	52
Table 2.6	Changes in Money Supply and Factors Affecting, 1970-1990	53
Table 2.7	Companies Listed in the Libyan Stock Market	57
Table 3.1	Growth Rates in CEECs Following Reforms	68
Table 3.2	China's Average Real Annual GDP Growth Rates, 1960-2007	69
Table 3.3	Indian Rates of Economic Growth, 1950-2000	70
Table 3.4	The Economic Reform in Latin America	71
Table 4.1	The Macrodeterminants of Growth	87
Table 5.1	The Population and Income Per-capita Indicators of the World including Emerging Market	
	Countries, 2005	101
Table 5.2	Stock Market Performance after Liberalisation in Emerging Markets	102
Table 6.1	Difference between Positivist and Phenomenological Philosophies	126
Table 6.2	Study Population (Sample Size)	141
Table 6.3	Sample Size: Previous Studies	143
Table 6.4	Number of Respondents who did not Respond	149
Table 6.5	Sample Representative by Financial and Non-financial Sectors	150
Table 6.6	Data Quality Issues Related to Questionnaires and Semi-structured Interviews	153
Table 6.7	Values of Correlation Coefficient (r)	160
Table 7.1	Distribution of Responses by Sample Size of Population Groups	169
Table 7.2	Descriptive Statistics for Deregulation	174
Table 7.3	Descriptive Statistics for Corporatisation	176
Table 7.4	Descriptive Statistics for Privatisation	176
Table 7.5	Descriptive Statistics for Liberalisation	180
Table 7.6	Descriptive Statistics for Interest Rates	181
Table 7.7	Descriptive Statistics for Exchange Rates	183
Table 7.8	Descriptive Statistics for Inflation Rates	185
Table 7.9	Descriptive Statistics Budget Deficit	188
Table 7.10	Descriptive Statistics for Per-capita Income	190
Table 7.11	Descriptive Statistics for Real GDP Growth Rate	190
Table 7.12	Descriptive Statistics for Banking Assets Ratio of GDP	194
Table 7.13	Descriptive Statistics for Domestic Credit Ratio of GDP	194
Table 7.14	Descriptive Statistics for Stock Market Capitalisation Ratio of GDP	196

Table 7.15	Respondents' Interviews to Stock Market Development	
Table 7.16	Descriptive Statistics for Market Size	
Table 7.17	Respondents' Interviews to Market Size	
Table 7.18	Descriptive Statistics for Market Liquidity	
Table 7.19	Respondents' Interviews to Market Liquidity	
Table 7.20	Descriptive Statistics for Market Activity	
Table 7.21	Respondents' Interviews to Market Activity	
Table 7.22	Descriptive Statistics for Market Concentration	
Table 7.23	Respondents' Interviews to Market Concentration	
Table 8.1	Descriptive Statistics for Research Variables	
Table 8.2	Correlation of Descriptive Statistics for Regression Analysis	
Table 8.3	Multiple Regression Model Summary	
Table 8.4	ANOVA for Multiple Regressions	
Table 8.5	Coefficient of the Multiple Regression Model	
Table 8.6	Coefficient of the Multiple Regression Model	
Table 8.7	Coefficient of the Multiple Regression Model	
Table 8.8	Coefficient of the Multiple Regression Model	
Table 8.9	Coefficient of the Multiple Regression Model	
Table 8.10	Structural Model Results for Scope of Testing Hypotheses	
Table 8.11	Correlation of Descriptive Statistics for Regression Analysis	
Table 8.12	Multiple Regression Model Summary	
Table 8.13	ANOVA for Multiple Regressions	
Table 8.14	Coefficient of the Multiple Regression Model	
Table 8.15	Coefficient of the Multiple Regression Model	
Table 8.16	Coefficient of the Multiple Regression Model	
Table 8.17	Coefficient of the Multiple Regression Model	
Table 8.18	Coefficient of the Multiple Regression Model	
Table 8.19	Coefficient of the Multiple Regression Model	
Table 8.20	Coefficient of the Multiple Regression Model	
Table 8.21	Structural Model Results for Scope of Testing Hypotheses	
Table 8.22	Correlation of Descriptive Statistics for Regression Analysis	
Table 8.23	Multiple Regression Model Summary	
Table 8.24	ANOVA for Multiple Regressions	
Table 8.25	Coefficient of the Multiple Regression Model	
Table 8.26	Coefficient of the Multiple Regression Model	
Table 8.27	Coefficient of the Multiple Regression Model	
Table 8.28	Coefficient of the Multiple Regression Model	
Table 8.29	Structural Model Results for Scope of Testing Hypotheses	
Table 8.30	Structural Model Results for Scope of Research Testing Hypotheses	

LIST OF TABLES

Table 9.1	Stock Market Indicators: Market Size, Activity, Efficiency and Performance during 1995-200)6
		. 254
Table 9.2	Descriptive Statistics: Forty Two Emerging Markets (Annual Average 1995-2006)	. 269
Table 9.3	Correlation Matrix among Variables	. 272
Table 9.4	Regression Results: Bank-based Indicators	. 276
Table 9.5	Regression Results: Market-based Indicators	. 278
Table 9.6	Banking Sector and Growth: Per-capita Real GDP as a Function of Proxy Measures for Bank	ing
	Development Indicators	. 281
Table 9.7	Stock Market and Growth: Per-capita Real GDP as a Function of Proxy Measures for Stock	
	Market Development Indicators	. 282
Table 9.8	Initial Stock Market Development, Banks and Growth, 42 Country Sample (Dependent Varia	ıbles:
	Real Per-capita GDP Growth 1995-2006)	. 285

LIST OF FIGURES

LIST OF FIGURES

Figure 1.1	Questions Related to the Aim and Objectives	
Figure 1.2	Flowchart for Thesis Navigation	
Figure 2.1	The Libyan Geography and Location	39
Figure 2.2	Political and Historical Landmarks	
Figure 2.3	Potential Economic Reform Programme	
Figure 2.4	Percentage of Real GDP Growth and GDP Per-capita in US\$, 1980-2007	49
Figure 2.5	Indicators of the Development of Commercial Banks in Libya, 1970-2003	
Figure 3.1	Key Events and Real World Oil Prices, 1970-2006	73
Figure 4.1	Relationships between Endogenous and Exogenous Variables	78
Figure 4.2	The Living Standards for the Long-term Process	81
Figure 4.3	Theoretical Approaches to Finance and Growth	89
Figure 5.1	Classification of Stock Markets	
Figure 5.2	Flow of Funds through the Financial System	97
Figure 5.3	Distribution of World GDP and Population, 2005	101
Figure 5.4	The World Equities Crisis in Europe, American and Asia Pacific in September 2008	110
Figure 5.5	Structural Changes in GDP by Economic Activity (1980-2007)	
Figure 5.6	Conceptual Framework of Privatisation and Regulatory Reforms	115
Figure 6.1	Major Steps Influencing Research Design and Process	123
Figure 6.2	Conceptual Framework for the Impact of Stock Market Performance	131
Figure 6.3	The Primary Model for the Impact of Stock Market Performance	132
Figure 6.4	Theoretical Model from other Emerging Markets	133
Figure 6.5	Research Typology Diagram	136
Figure 6.6	Deductive and Inductive Approaches	137
Figure 6.7	Research Activities and the Pilot Test	146
Figure 6.8	Values of the Correlation Coefficient	160
Figure 6.9	Type I and Type II Errors	162
Figure 8.1	Framework of Research Data Analysis Excluding the Measurement Analysis	209
Figure 8.2	The Primary Model for the Impact of Stock Market Performance	
Figure 9.1	Theoretical Model from other Emerging Markets Error! Bookmark no	ot defined.
Figure 9.2	Forms of Variable Measures	
Figure 9.3	Scatterplot of Predicted and Observed Variables	

LIST OF APPENDICES

Appendix A	General Information about Libya	354
Appendix B	Research Methodology Diagram	372
Appendix C	Covering Letter from the Researcher Sent with the Self-administered Questionnaires	374
Appendix D	Covering Latter from Researcher Sent with the Self-administered Questionnaires (in Arabic) .	375
Appendix E	Covering Latter form Cultural Affairs the Libyan People's Bureau Sent with Self-administered	ł
	Questionnaires in (Arabic)	376
Appendix F	Free Order Form	377
Appendix G	Economic Reform Programme and Stock Market Performance Questionnaires (Original in	
	Design)	378
Appendix H	Design) Covering Letter from the University of Huddersfield with Regard to Conducting the Semi-	378
Appendix H	Design) Covering Letter from the University of Huddersfield with Regard to Conducting the Semi- structured Interviews	378 388
Appendix H Appendix I	Design) Covering Letter from the University of Huddersfield with Regard to Conducting the Semi- structured Interviews Interview Questions	378 388 389
Appendix H Appendix I Appendix J	Design) Covering Letter from the University of Huddersfield with Regard to Conducting the Semi- structured Interviews Interview Questions Respondent Interviews	378 388 389 390
Appendix H Appendix I Appendix J Appendix K	Design) Covering Letter from the University of Huddersfield with Regard to Conducting the Semi- structured Interviews Interview Questions Respondent Interviews Chapter Seven: Presentation of the Findings	378388389390392
Appendix H Appendix I Appendix J Appendix K Appendix L	Design) Covering Letter from the University of Huddersfield with Regard to Conducting the Semi- structured Interviews Interview Questions Respondent Interviews Chapter Seven: Presentation of the Findings Chapter Eight: Statistics Tests; Appropriate Use	 378 388 389 390 392 419

LIST OF ACRONYMS AND ABBREVIATIONS

1- Abbreviations	
AMF	Arab Monetary Fund
BPC	Basic People's Congresses
BGC	Board of the General Committee
CEECs	Central and East European Countries
CBL	Central Bank of Libya
EMDB	Emerging Market Database
EU	European Union
GPC	General People's Committee
GDP	Gross Domestic Product
IFC	International Financial Corporation
IMF	International Monetary Fund
LD	Libyan Dinar
LSM	Libyan Stock Market
MENA	Middle East and North Africa
PCs	People's Congresses
PhD	Doctor of Philosophy
Prof	Professor
TFP	Total Factor Productivity
2SLS	Two-Stage Least-Squares
UAE	United Arab Emirates
UHBS	University of Huddersfield Business School
UK	United Kingdom
UN	United Nations
US	United States
US\$	United States Dollar
WTO	World Trade Organisation
WWW	World Wide Web
2- Code	Variables
BAR	Bank Asset Ratio is the Total Asset of the Banking Sector Divided by GDP
CLM	Capital Market
СҮМ	Currency Market
DC/GDP	Domestic Credit Ratio in the Ratio of Bank Claims on Private Sector GDP
EDUC	Secondary School Enrollment Rate
EF	Economic Freedom
ERP	Economic Reform Programme
ERPC	Corporatisation
ERPD	Deregulation
ERPL	Liberalisation
ERPPV	Privatisation
FLI	Financial Institutions
FM	Financial Maturity
FMBA	Banking Assets Ratio
FMDC	Domestic Credit Ratio
FMMC	Market Capitalisation Ratio
INF	Inflation Rate
INVES	Value Investment as Percentage of GDP
LBAR	Natural Logarithm of Bank Asset Ratio as Total Asset of the Banking Sector
LBAR X A PCI	Interaction with Change Per-capita Income
LDC/GD	Natural Logarithm of Domestic Credit Ratio in the Ratio of Bank Claims on Private Sector GDP
LDC/GD x APCI	Interaction with Change Per-capita Income
LIPCY	Natural Logarithm of Initial Per-capita Income
LMC/GD x ΔPCI	Interaction with Change Per-capita Income
LMC/GDP	Natural Logarithm of Market Capitalisation Ratio as the Market Capitalisation Divided by GDP
LTURN/GD x APCI	Interaction with Change Per-capita Income
LTURN/GDP	Natural Logarithm of Turnover Ratio as the Total of Equity Transactions Divided by Market
LVT/GD x APCI	Interaction with Change Per-capita Income
	Natural Logarithm of Value Traded Ratio as the Investment Ratio as the Volume of Gross Capital
LVT/GDP	Formulation Divided by GDP
MC/GDP	Market Capitalisation Ratio is the Market Capitalisation Divided by the GDP
MER	Macro-economic Reform
MERBD	Budget Deficit

	Exchange Rate
MERIFR	Inflation Rate
MERIR	Interest Rate
MERPCI	Per-capita Income
OPEN	Sum of Imports and Exports Divided by GDP
PCS	Public Sector
PCYG	Growth as Real Per-capita GDP Growth
POP	Population Growth Rate
PS	Political Stability
PVS	Private Sector
SMP	Stock Market Performance
SMPMA	Market Activity
SMPMC	Market Concentration
SMPML	Market Liquidity
SMPMZ	Market Size
TURNOVER	Total of Equity Transactions Divided by Market Capitalisation
VT/CDD	Value Traded Ratio as Investment Ratio is the Volume of Gross Capital Formulation Divided by
v I/GDP	GDP
3- Mathematical Operato	rs
SPSS	Statistical Package for the Social Sciences
Σ	This called sigma means 'adds every thing up'. For instance, if you use something like $\sum x_{i}$ it
\sum	
	just means add up all of the scores you've collected
4- Greek Symbols	
u	Probability of making a type I error
β	Probability of making a type II error
B_i	Standardised regression coefficient
χ^{2}	Chi-square test statistics
ε	Usually stands for error
μ	Mean of a population of scores
5- English Symbols	
b_i	Regression coefficient (unstandardised)
df	Degrees of freedom
DW	Durbin-Watson test for residual serial correlation
DW e _i	Durbin-Watson test for residual serial correlation Error associated with the i th person
DW e _i F	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA)
DW e _i F In	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm
DW e_i F In MS	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data
DW e_i F In MS N, n, n_i	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. N usually denotes the total sample size, whereas n usually denotes the size of a particular group
DW e_i F In MS N, n, n_i P	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. N usually denotes the total sample size, whereas n usually denotes the size of a particular group Probability (the probability value, P-value or significance of a test are usually denoted by P)
DW e_i F In MS N, n, n_i P r	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. N usually denotes the total sample size, whereas n usually denotes the size of a particular group Probability (the probability value, P-value or significance of a test are usually denoted by P) Person's correlation coefficient
DW e_i F In MS N, n, n_i P r r_s	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. <i>N</i> usually denotes the total sample size, whereas <i>n</i> usually denotes the size of a particular group Probability (the probability value, P-value or significance of a test are usually denoted by <i>P</i>) Person's correlation coefficient Person's rank correlation coefficient
DW e_i F In MS N, n, n_i P r r_s P	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. <i>N</i> usually denotes the total sample size, whereas <i>n</i> usually denotes the size of a particular group Probability (the probability value, P-value or significance of a test are usually denoted by <i>P</i>) Person's correlation coefficient Multiple correlation coefficient
DW e_i F In MS N, n, n_i P r r_s R	Durbin-Watson test for residual serial correlationError associated with the i th personF-ratio (test statistics used in ANOVA)Natural logarithmMeans squared error. The average variability in the dataSample size. N usually denotes the total sample size, whereas n usually denotes the size of a particular groupProbability (the probability value, P-value or significance of a test are usually denoted by P)Person's correlation coefficientPerson's rank correlation coefficientMultiple correlation coefficientCoefficient of determination (i.e. the proportion of variance within some data explained by the
DW e_i F In MS N, n, n_i P r r_s R R^2	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. <i>N</i> usually denotes the total sample size, whereas <i>n</i> usually denotes the size of a particular group Probability (the probability value, P-value or significance of a test are usually denoted by <i>P</i>) Person's correlation coefficient Person's rank correlation coefficient Multiple correlation coefficient Coefficient of determination (i.e. the proportion of variance within some data explained by the model)
DW e_i F In MS N, n, n_i P r r_s R R^2 S	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. N usually denotes the total sample size, whereas n usually denotes the size of a particular group Probability (the probability value, P-value or significance of a test are usually denoted by P) Person's correlation coefficient Person's rank correlation coefficient Multiple correlation coefficient Coefficient of determination (i.e. the proportion of variance within some data explained by the model) Standard deviation of a sample of data
DW e_i F In MS N, n, n_i P r r_s R R^2 S S^2	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. N usually denotes the total sample size, whereas n usually denotes the size of a particular group Probability (the probability value, P-value or significance of a test are usually denoted by P) Person's correlation coefficient Person's rank correlation coefficient Multiple correlation coefficient Coefficient of determination (i.e. the proportion of variance within some data explained by the model) Standard deviation of a sample of data
DW e_i F In MS N,n,n_i P r r_s R R^2 S S^2 SS	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. N usually denotes the total sample size, whereas n usually denotes the size of a particular group Probability (the probability value, P-value or significance of a test are usually denoted by P) Person's correlation coefficient Multiple correlation coefficient Coefficient of determination (i.e. the proportion of variance within some data explained by the model) Standard deviation of a sample of data Variance of a sample of data
DW e_i F In MS N,n,n_i P r r_s R R^2 S S^2 SS SS	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. <i>N</i> usually denotes the total sample size, whereas <i>n</i> usually denotes the size of a particular group Probability (the probability value, P-value or significance of a test are usually denoted by <i>P</i>) Person's correlation coefficient Person's rank correlation coefficient Multiple correlation coefficient Coefficient of determination (i.e. the proportion of variance within some data explained by the model) Standard deviation of a sample of data Variance of a sample of data Sum of squares or sum of squared errors to give its full title Model sum of squares
DW e_i F In MS N,n,n_i P r r_s R R^2 S S^2 SS SS_M SS	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. N usually denotes the total sample size, whereas n usually denotes the size of a particular group Probability (the probability value, P-value or significance of a test are usually denoted by P) Person's correlation coefficient Multiple correlation coefficient Coefficient of determination (i.e. the proportion of variance within some data explained by the model) Standard deviation of a sample of data Sum of squares or sum of squared errors to give its full title Model sum of squares
DW e_i F In MS N, n, n_i P r r_s R R^2 S S^2 SS SS_M SS_R SS	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. <i>N</i> usually denotes the total sample size, whereas <i>n</i> usually denotes the size of a particular group Probability (the probability value, P-value or significance of a test are usually denoted by <i>P</i>) Person's correlation coefficient Person's rank correlation coefficient Multiple correlation coefficient Coefficient of determination (i.e. the proportion of variance within some data explained by the model) Standard deviation of a sample of data Variance of a sample of data Sum of squares or sum of squared errors to give its full title Model sum of squares Residual sum of squares
DW e_i F In MS N,n,n_i P r r_s R R^2 S S^2 SS SS_M SS_R SS_T	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. <i>N</i> usually denotes the total sample size, whereas <i>n</i> usually denotes the size of a particular group Probability (the probability value, P-value or significance of a test are usually denoted by <i>P</i>) Person's correlation coefficient Multiple correlation coefficient Coefficient of determination (i.e. the proportion of variance within some data explained by the model) Standard deviation of a sample of data Sum of squares or sum of squared errors to give its full title Model sum of squares Residual sum of squares
DW e_i F In MS N,n,n_i P r r_s R R^2 S S^2 SS SS_M SS_R SS_T t	Durbin-Watson test for residual serial correlationError associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA)Natural logarithmMeans squared error. The average variability in the dataSample size. <i>N</i> usually denotes the total sample size, whereas <i>n</i> usually denotes the size of a particular groupProbability (the probability value, P-value or significance of a test are usually denoted by <i>P</i>)Person's correlation coefficientPerson's rank correlation coefficientCoefficient of determination (i.e. the proportion of variance within some data explained by the model)Standard deviation of a sample of dataSum of squares or sum of squared errors to give its full titleModel sum of squaresTotal sum of squaresTest statistics for student's t-test
DW e_i F In MS N,n,n_i P r r_s R R^2 S S^2 SS SS_M SS_R SS_T t VIF	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. <i>N</i> usually denotes the total sample size, whereas <i>n</i> usually denotes the size of a particular group Probability (the probability value, P-value or significance of a test are usually denoted by <i>P</i>) Person's correlation coefficient Multiple correlation coefficient Coefficient of determination (i.e. the proportion of variance within some data explained by the model) Standard deviation of a sample of data Variance of a sample of data Sum of squares Residual sum of squares Total sum of squares Test statistics for student's t-test Variance Inflation Factors
DW e_i F In MS N,n,n_i P r r_s R R^2 S S^2 SS SS_M SS_R SS_T t VIF VIR	Durbin-Watson test for residual serial correlation Error associated with the <i>i</i> th person <i>F</i> -ratio (test statistics used in ANOVA) Natural logarithm Means squared error. The average variability in the data Sample size. <i>N</i> usually denotes the total sample size, whereas <i>n</i> usually denotes the size of a particular group Probability (the probability value, P-value or significance of a test are usually denoted by <i>P</i>) Person's correlation coefficient Person's rank correlation coefficient Multiple correlation coefficient Coefficient of determination (i.e. the proportion of variance within some data explained by the model) Standard deviation of a sample of data Variance of a sample of data Sum of squares Residual sum of squares Total sum of squares Variance Inflation Factors Voluntary Index Ratio

CHAPTER ONE

INTRODUCTION

1.1 AN OVERVIEW

The purpose of this introductory chapter is to provide an overview and general background to the research and its motivations, outlining the topic for investigation, the research aim, objectives and associated questions. It offers a statement of the problems facing Libya and the justification for the research, providing a brief indication of the methodology to be followed and supplying detail of the structure of the thesis. The aim of this thesis is to explore which economic reform programme, macro-economic reform and institutions are required for strong, well-developed and efficient stock market performance in Libya. In connection with understanding the significance of the subject, this study seeks to explore the development of the current securities market, assesses recent efforts to create new deregulation, to create new institutions and attract further foreign investment that could modernise the market, and offers suggestions for reforms that could stimulate additional market development. In the main, the thesis provides an assessment of the quality and effectiveness of the operation of the economic reform programme and the infrastructure of the securities market in Libya. The study is of importance since, as is the case in many developing countries that have introduced an ambitious economic reform programme, the creation of a stock market in Libya has occurred on the assumption that more effectiveness in the development of economic growth will follow. However, there is no empirical evidence that provides policy markets with information relating to the particular fundamentals in developing a stock market in Libya. Therefore, there is a point of view that doubts the likely positive impact of a stock market upon economic growth in developing countries with a small population such as Libya.

This study's results will be useful in reaching policy decisions to develop stock markets in order to increase economic growth in developing countries and/or emerging economies in general, and within Libya in particular. Furthermore, providing empirical evidence regarding this critical issue within specific emerging economies will add to the literature on economic reform related to the role of stock market development and its influence on economic growth and, thus, initiate an exciting topic for further research. During the twentieth century the world was fragmented into diverging economic blocks, one following the direction of capitalism and emphasising privatisation and investment in a market-based economy and the other pursuing

20

centrally-planned, state-owned economies. Notwithstanding that, capitalism emerged as the more effective method of managing economies globally. Most of the previous socialist and communist economies then began the trend towards partial, or complete economic reform. Indeed, with the beginning of the reforms in 1985 which would eventually lead to the dissolution of the Soviet Union in 1991, the newly-independent countries in Africa, the Middle East and Latin America all came to realise the importance of economic reform programmes. Accordingly, since 1994 most developing countries have implemented similar programmes, often under pressure from global organisations such as the International Monetary Fund (IMF) and the World Bank (Williamson, 1994).

During the 1990s and first few years of the 21st century, Middle East and North Africa (MENA) countries - in particular Libya, Egypt, Tunisia, Jordan and Saudi Arabia - all began similar reform programmes aimed at achieving the stabilisation of their economies. Specifically, they found themselves incapable of attracting sufficient foreign exchange to protect their currencies and began to transform their economies from socialist and state-owned to market-based entities (Creane et al., 2004). Financial markets are today classified as bankbased or market-based. This division can be further illustrated by the Anglo-Saxon marketbased models which are prevalent in capitalist economies and allow for private investment and private ownership, and the other type, which is largely exemplified in Germany. The latter, bank-based model, has been practised more by Eastern European countries, which were centrally-planned economies or, to be politically correct, communist economies (Hall and Soskice, 2001). The relationship between financial development and economic growth has become an object of extensive analysis and debate, the question arising being whether or not markets are critical in influencing economic growth. Vastly disparate arguments are proposed by academic economists in this respect, some believing that the role of finance in economic growth is not significant, or that it is of secondary importance. Lucas (1988: 3), for instance, argues that "Economists badly over-stress the role of financial factors in economic growth". Others, though, consider that financial markets play a key role in economic growth. Such disagreement regarding the nature of financial market development in economic growth is not a novel phenomenon. Bagehot (1873) and Hicks (1969), for instance, proposed that financial market development played a critical role in British industrialisation by facilitating the mobilisation of funds for "immense works" and, in this respect, Bagehot (1873: 3-4; cited by Sinha and Macri, 1999: 4) observed:

"We have entirely lost the idea that any undertaking likely to pay, and seen to be likely, can perish for want of money; yet no idea was more familiar to our ancestors, or is more common is most countries. A citizen of London in Queen Elizabeth's time [...] would have thought that it was no use inventing railways (if he could have understood what a railway meant), for you would have not been able to collect the capital with which to make them. At this moment, in colonies and all rude economies, there is no large sum of transferable money; there is no fund from which you can borrow, and out of which you make immense works".

Gurley and Shaw (1955) made a direct correlation between financial markets and real activity, recognising that financial markets can widen borrowers' financial capacity and improve the efficiency of inter-temporal trade and result in the pooling of investment. Moreover, they asserted that financial markets are critical to economic development because they can increase physical capital accumulation. Robinson (1952: 86) did not attribute a role to financial institutions in development, believing that economic growth creates a demand for financial service and that "where enterprise leads, finance follows". She found that financial development was merely a short-term economic development and that financial institutions underpinned no significant economic growth. However, other studies, such as those of Goldsmith (1969), McKinnon (1973) and Shaw (1973), recognised that financial intermediation could be positively correlated with economic growth. Recent analysis of the link between financial development and growth, gained from insights acquired as a result of using the technique of endogenous growth models, has illustrated that growth rates could be related to technology, income distribution and institutional arrangements. This provides the theoretical background that empirical studies have lacked; thus illustrating that financial intermediation affects the level of economic growth. The resulting models have provided a new impetus to empirical research into the effects of financial development (Gertler, 1988; Levine, 1997; Bossone, 2000; Tsuru, 2000). These researchers established that financial intermediation, such as in the mobilising of capital, assisting the allocation of resources, monitoring managers and facilitating risk management, could impact upon economic growth.

The aim of this chapter, then, is to provide a general background concerning the research. As such, it is divided into five main sections. Section 1.2 describes the problems addressed by the study. In Section 1.3, the motivation behind conducting this research is discussed. In section 1.4, the overarching aim and objectives of the research are presented. In Section 1.5, presented the scope of the research. In Section 1.6, the empirical study chosen is briefly justified. Finally, in Section 1.7, a brief summary of the format and structure of this research are provided.

1.2 Problems Addressed by the Study

The problem to be addressed by this study is the economic situation of one of the MENA economies, namely that of Libya, which has witnessed organisational variations following the revolution of 1969, since when the Libyan economy has been driven by the Green Book Theory¹, which provides "for a socialist state and a state-planned economy. Investment was essentially state-driven and trade and price controls, along with subsidies, were widespread. Economic performance was [strictly] constrained by stifling government interference in the economy and an [unfavourable] business climate. Economic conditions began to deteriorate in the mid-1980s with the fall in world oil prices, and worsened in the 1990s as a result of UN economic sanctions. The authorities were reluctant to initiate any economic reforms [whilst] the country was subject to sanctions", IMF (2005: 5).

There followed the challenge of devising a strategic policy that would aim to achieve the sustainable development of the Libyan economy, especially in view of the current official tendencies for the restructuring of the Libyan economy by increasing the role of private ownership in the economy and also the continuing process of the privatisation of the public sector within the country (IMF, 2005). The privatisation programme was started in 1999 to alleviate the budget constraints of state-owned enterprises and to create economic diversity after a long dependence on oil. Oil revenues comprise over 95 per cent of Libya's hard currency earnings (75 per cent of government receipts and 30 per cent of the gross domestic product) and, thus, the economy was severely damaged by the dramatic decline in oil prices during 1998, as well as by reduced oil exports and production, in part as a result of UN and US sanctions. However, the imposition of UN sanctions in 1999 brought the Libyan economy to the brink of collapse, at which point the government introduced a rigid economic reform programme with the intention of transforming its socialist state-planned economy into one based upon capital, which (it was believed) would combat the stagnating status of the Libyan economy during the previous decade. In the late 1990s macro-economic stability was the sole intent of such reforms, which aimed to train the workforce, generate new jobs, encourage investment and sell public enterprises in order to minimise public spending.

¹ For more explanation, see Appendix A: Section A 1.1 Political Determinants of Economic Reform.

In fact, Libya, like other nations, had long suffered from deficiencies in the economy due to its political stance and UN sanctions, which had resulted in a high rate of inflation, a balance of payments' deficit, and low employment and growth rates, ultimately combining to create an imbalance in the economy. Problems of irregularity in supply and demand have a negative impact on prices and the balance of payments. For that reason, fiscal and monetary reforms constituted the core of the programme in its first stage, whereas structural reforms constituted the core aim of the remainder of the programme in the medium term. Therefore, in 1999, an economic reform programme was urgently needed to ensure improved economic performance (Alfaitori, 2003), which has now been achieved to steer the post-sanctions Libyan economy away from the edge of collapse. Nonetheless, there remains much for this reform programme to do before it can yield fruit due to numerous obstacles such as the sharp decrease in oil prices, shortage of trained labour and structural weaknesses in the financial system. However, similar to many other emerging countries, the economic reform programme was set with clear overall objectives, these being to accelerate the rate in the growth of the economy and to increase standards of living for the Libyan population. To achieve those objectives, a high level of investment is required which, it can be argued, demands a strong and healthy stock market capable of attracting more of both *domestic and foreign investments*.

1.3 MOTIVATION FOR THE RESEARCH

The motivation for conducting this research, hence, can be divided into three main subsections. Subsection 1.3.1 displays the challenge and opportunity. Subsection 1.3.2 presents the need and viability. Subsection 1.3.3 addresses the justifications for the research.

1.3.1 Challenge and Opportunity

The Libyan economy suffers from numerous negative aspects resulting from the political situation and financial crises. As mentioned earlier, different sanctions were imposed on Libya by the US and the UN. The US banned imports of Libyan crude oil in 1981 and extended it later to include direct trade, commercial contracts and travel activities. The UN embargo began in 1992 after the accusation of two Libyan citizens of involvement in the crash of an American aeroplane in Scotland in 1988. The UN embargo was eased in 1999 and completely ended in 2003 after the country accepted responsibility for the crash, while the US embargo was ended in 2004. The Libyan economy has suffered from these sanctions, especially those imposed by the UN, which have additionally influenced the health, education, oil sector and

CHAPTER 1: INTRODUCTION

other aspects of Libyan people's lives. In June 2003 President Al-Qaddafi declared that the country's public sector had failed and called for oil sector privatisation, pledging to take Libya into the World Trade Organisation (WTO) and appointing the former Trade and Economy Minister Tayeb Asafay, a supporter of privatisation and liberalisation, as Prime Minister (Igbikiowubo, 2003). In October 2003 the Prime Minister announced a list of 360 firms in a variety of sectors (steel, petrochemicals, cement and agriculture) which were to be privatised (Energy Information Administration/Department of Energy, 2005). This was executed in three stages. The ownership of 260 factories was to be transferred from the public to the private sector by the end of 2005. The second stage, was to transform 46 factories into shareholding companies in which employees and others would own shares by the end of 2007. The third stage aimed to privatise 54 factories during the period 2007/2008.

Further improvement in relations between Libya and the West will allow more of a focus on boosting investment in the oil and gas sector, where competition is increasing between EU and US companies. With an investment target of some 30 billion US\$ by 2010, the National Oil Company (NOC) launched several rounds of Exploration and Production Sharing Agreements, which include incentives for both oil and gas development companies (St John, 2008). Furthermore, "Libya is also looking at internal political reform, on the basis that it cannot combine a foreign policy open to the international capitalist system with a domestic political system based on post-Cold War revolutionism. Just how far Libya is prepared to go in terms of reforming its internal affairs, however, remains to be seen [...] Hence political reform will require structural changes and time to advance" (Arbab, 2005: 6).

Since August 2003 Libya, UK and the US have reached an agreement that Libya will accept responsibility for the 1988 bombing of Pan Am flight 103 whereupon Libya agreed to pay 2.7 billion US\$ in compensation to the families of the victims of the Lockerbie (Scotland) air crash. In response, the UN Security Council voted on 12 September 2003 to lift its 11-year economic sanctions' regime and the arms embargo which had been in place for 18 years. This represented a further step that allowed EU countries to export arms and other military equipment to the Libyan government. The Libyan Foreign Minister announced on 19 December 2003 that the government had decided of its own *free will* to completely divest of internationally banned weapons (Paul, 2004). In 2004 Libya agreed to pay compensation to the French airline UTA for the loss of a plane which crashed over the west African state of Niger in 1989, to pay 170 million US\$ compensation, with each of the victims' families receiving 1

million US\$, and to pay 35 million US\$ to relatives of the victims of a 1986 attack on a discotheque in West Berlin (Time, 2004). On 23 April 2004 US President Bush made the decision to remove most of the remaining sanctions on connected business in Libya, with the establishment of a diplomatic mission in Tripoli. Six months later he revoked the trade embargo on Libya, but the nation remains on the US list of state sponsors of terrorism. By 27 April Al-Qaddafi commenced a visit to Europe after 15 years of the Euro-Mediterranean Partnership (EMP or Barcelona Process) for preliminary talks, with an announcement in Brussels that he would again support Freedom Fighters if the west refused his offer of peace (Tim, 2004; Fernández, 2006). Since September 2004, when the UN and US gradually removed Libya-specific trade sanctions and the implementation of measures intended at enhancing the role of the private sector in the economy, the pace of reform in Libya has accelerated (IMF, 2005).

On 1 November 2004 Prime Minister Shukri Ghanem announced that Libya intended to abolish some 5 billion US\$ worth of subsidies on electricity, fuel and basic food items in a move to liberalise the economy. One month later the freeze on 1 billion US\$ in assets in US banks which had been imposed in 1986 was withdrawn. In mid-November 2004 President Bush asked Congress to lift the US ban on export and imports bank loans to Libya, arguing that action was very important to facilitate US investment (St John, 2006). In January 2005 the Libyan government announced the first of a series of new tenders for Exploration and Production Sharing Agreements (EPSA) and awarded 11 of 15 new EPSAs to American oil companies operating in partnership with other companies. The other winners, Petrobras of Brazil, Indian oil, Medco Energy International of Indonesia, Oil Search of Australia, Verenex Energy of Canada, and Sonatrach of Algeria, reflected global interest in Libyan hydrocarbon deposits (St John, 2006). Italian businesses, particularly investment companies, had interests in Libyan oil and gas, which provided about 25 *per cent* of Italy's total energy imports and with activation of the "Green Stream" gas pipeline, its share was increased to 30 *per cent*.

On 3 February 2005 the US Boeing Corporation announced the sale of up to six 737-800s to Buraq Airlines, a privately owned Libyan company. The sale to Buraq, Libya's first private airline, was valued at 367 million US\$ (St John, 2006). In March 2005 the General People's Congress approved a measure allowing foreign banks to establish branches in Libya and to contribute as shareholders in local banks. In August 2005 Libya lifted all import duties (except for those on cigarettes) in a bid to spur trade and render the country a free trade area. These

duties were replaced by a 4 *per cent* tax called the Service Import Tax. On 23 May 2005 the organisation Human Rights Watch said that, in their first visit to the country, Libya had taken significant steps to improve its human rights' record over the past years. However, there were still serious problems, including the use of violence against detainees, restrictions on the freedoms of expression and assembly, and the continued detention of political prisoners. Sarah Whitson, who works as MENA director at Human Rights Watch said "*The Libyan Government's invitation reflected a welcome degree of transparency. We're looking forward to further human rights improvements in the country, as Libyan and international laws demand*" (Human Rights Watch 2005). In May 2006, the US restored full diplomatic relations, removing Libya from the US State Department list of State Sponsors of Terrorism and omitting it from the annual certification of countries not fully cooperating with American antiterrorism efforts. In October 2007 Libya was voted onto the UN Security Council as a non-permanent member. Although Washington did not endorse Libya's application, it did not block it either (as it had with Libya's previous attempts to join in 1995 and 2000) (Kaplan, 2007).

These facts are consistent with the Libyan geography and natural resources (oil and gas) and are helping countries take a step further in relations with most of the global economic countries. Libya is located in the heart of the North Africa region, within the Arab Muslim world, having a long coastline with the Mediterranean Sea of about 2,000 km long. Much of it is still in a natural undeveloped state and its total area is 1,759,540 sq km (679,358 sq ml), with a very small population estimated to be 6,089 million in 2007 (see Appendix A). In the east lie Egypt and Sudan as neighbours, whilst Algeria and Tunisia border to the west with Chad and Niger to the south. Given what has been mentioned, one of the most important subjects on the agenda for government policy is economic reform and Libya's reintegration into the international community, which would make it a gateway for an African Union-pan-African body. Libya is a founder member of the 50-state African Union Organisation with strong political and economic ties with its members and is a leading participant in its activities. In December 2007 French President Sarkozy was the first western leader since 2004 to welcome Col. Qaddafi, when the latter took the decision to rebuild diplomatic bridges with a view to ending the country's isolation from the US and Europe. The two parties signed a number of business agreements, such as an accord to develop the peaceful use of nuclear energy and the supply of one "or several" nuclear reactors to desalinate sea water, with deals agreed potentially worth 10 billion € for French firms (BBC News, 2007). After Qaddafi had

made this five-day trip to France he visited Spain (his second official visit since 1984), which included meetings with Prime Minister Zapatero and King Juan Carlos. A number of key agreements were signed with possibilities for investments by Spanish firms in Libya totalling over 17 billion US\$ (11.8 billion €). Beyond this there is the possibility of 2 billion € investment in defence and aeronautics and over 5 billion US\$ in the Libyan energy sector. Other investment enterprises with the Libyan government are anticipated in bids for infrastructure contracts worth in excess of 50 billion € in Libya (AFP, 2007). Finally, Libya and Spain signed a political cooperation accord, whereby the Spanish will help train the Libyan police in migration and maritime border control. After a long period of isolation, Al-Qaddafi opened the Libyan economy to business and investment. Libya, a member of OPEC, holds the largest proven oil reserves in Africa, accounting for greater than 3 per cent of the world's total reserves. By 2012 Libya seeks to double its oil output capacity to 3 million barrels a day. Furthermore, it is also seeking to invest about 50 billion US\$ in the US Sovereign Wealth Fund (Kessler, 2008). On 5 September 2008 the relationship between the US and Libya entered an historic new phase when Condoleezza Rice become the first US Secretary of State to visit Libya since 1953. She pointed out that: "We have had a long and bad history with Libya, and that began to turn around when they turned away from nuclear weapons and terrorism. That country has radically changed its behaviour [...] The relationship between [the US and Libya] has been moving in a good direction for some time but we have a long way to go [...] we are working on a trade investment framework which will improve the climate for investment, which I know many American firms want to do". She made a further point: "We are co-operating on Libya's membership of the Security Council of the UN" (Kellerhals, 2008 and BBC News, 2008). David Welch Assistant Secretary of State, concluded that "This is a relationship that has had a troubled past, but now it is on a much firmer foundation". A further point was made by the president of the Ploughshares Fund, Joe Cirincione: "We've seen, under the Bush administration, Qaddafi go from a poster child of a rogue state leader to a man that President Bush calls a model that others should follow [...] that is a remarkable transformation. If a Democrat did that, the Republicans would undoubtedly be accusing him of appeasement. Sometimes, it takes a Bush to go to Libya" $(Kelemen, 2008)^{2}$.

² Cirincione aded that "The Iraq war has proved to be very difficult, costly and unnecessary [...] The Libya model has been cheap, 100 per cent effective, and nobody died [...] This is the right message to be sending to Iran and to North Korea, though those cases are far more complicated than Libya" (Kelemen, 2008).

From this summary it can be seen that Libya, however, after the lifting of UN sanctions in 1999, opened the oil and gas field to international investors. This increased Libyan income from oil and gas exports, which led the government to introduce a series of liberalisation economy measures including a significant role for deregulation and private sector ownership. As a result, Libya's economic growth rate during the economic reform period was one of the highest in the world. GDP increased at an average rate of 22.6 per cent per annum. Furthermore, gross fixed investment had increased at an average rate of 15.6 per cent per annum with the share in non-oil GDP reaching 63 per cent (CBL, 2007). Therefore, within these financial and political changes in Libyan history, having begun to change from a socialist command economy to a Western-style free market system in which its citizens were increasingly expected to fend for themselves, it is understandable that the newly-established stock exchange has had expectations placed upon it to create buoyancy and growth in the national economy. Even so, as the political history of Libya is somewhat dissimilar from other developing countries, there are certain influences which are peculiar to Libya and which, consequently, require an early evaluation of the performance of its stock exchange. As a result, Libya has launched an economic reform programme as an attractive development option to help sustain the national economy and financial investment.

1.3.2 The Need and Viability

The question arises here as to why Libya has been selected as the case study. In this respect it is vital to appreciate the fact that, like most developing countries, Libya possesses a stock market. However, the Libyan stock market has only very recently been established, in 2006, as a major step in the development of the country's economic reform programme. Its presence will facilitate a better operation of a valuable resource through the development of sound capital market performance. The stock market was established in order to move Libya from being a bank-based to a market-based economy and to contribute towards the capital-raising and capital-allocating process which is critical to increase the economy's growth. In some market-based economies the stock market is a key means of mobilising saving and reallocating resources, acting as a back-up and an assurance for domestic and foreign investment promotion and a significant source of capital formation and business financing.

Many countries of the MENA, such as Lebanon, Israel, Jordan, and Egypt, have established their own stock markets, therefore, it is not surprising that there has been an increasing demand to develop an organised stock market in Libya as well (see Appendix A, Table 3.19).

Without doubt, its overall economic reform initiative has situated the Libyan economy in the closest proximity to a market economy, transforming vital sectors within it. Increased privatisation, liberalisation, further investment, lower inflation and fewer financial regulations mean that the economic growth rate is far better than it was prior to 1999. In 2003-2004, reflecting the favourable developments in the world oil market, Libya's fiscal and current account balances registered large surpluses (IMF, 2005). Privatisation has indeed succeeded in attracting more and more investors, especially from the local market, thus alleviating public spending and adding liquidity. In this context, the need for a stock market to manage this situation is very apparent.

1.3.3 Justification for the Research

Libya's continuing economic reform will lead to a deepening of the Libyan stock market and will aid its maturity, but the absence of an effective stock exchange will deter the transformation of the country from being a bank-based to a market-based economy. Since it has been demonstrated that other emerging economies, whose economic features are very similar to those of Libya, prove the positive outcomes of developing stock markets, there is a rationale for exploring their approaches and learning from their experience, with the intention of proposing a route towards enhancing the development of the newly-instituted stock exchange in Libya. Therefore, the research is justified on economic and national development grounds. The reason for this is based on the growth model of the emerging economy as observed in MENA, Eastern Europe, Asia and Latin America which is similar in many respects to the Libyan economy (see Chapter Nine). Therefore, this critical review explores the emerging economy experience and concentrates on the part played by these countries' respective stock markets. From this review it is observed that, by reforming their stock markets, these countries have achieved significant levels of economic growth. Thus, the case for the development of an effective stock exchange in Libya is made on the assumption that stock markets contribute indirectly to economic growth. Several empirical examples are utilised to support this argument. Egypt, for instance, has developed positively over the last few years, after the Cairo Stock Exchange made significant progress. Consequently, it could be argued that a healthy relationship exists between Egypt's economic development and the advancement of its stock exchange. This hypothesis is supported by Omran's (2002) research findings in Egypt demonstrating that the economy prospered as a result of the development of the stock market along with other factors. These facts are interesting for the proposed study,

since Libya and Egypt, being neighbours in North Africa, possess several historical, cultural and climatic similarities and are connected by transportation and highway networks. Both altered their economic ideology from socialism to capitalism, although they effected this at different times. Libya is lagging behind Egypt in terms of the development of a stock exchange (the Cairo Stock Exchange was created in 1903), but a sharp focus on the relevant developments in Egypt will provide much valuable information and some beneficial lessons for Libya to assimilate in its own economic and financial progress.

Given what has been said in the previous discussion about Libya's unique position as a country, this study can be considered as a step towards the theoretical and empirical building of research methods which will provide a significant contribution to knowledge and understanding of Libya's economic reform programme and the case for a stock market performance for the subject area. The review of literature in this research reveals several gaps and the need for more empirical studies to be conducted. Therefore, this study has given more focus to extend the empirical work in this area with the intention of filling some of the gaps in the literature of Libyan economy in particular and other developing nations in general. This study can also be described as a significant exploratory study that includes crucial issues which can be barriers to understanding or a temptation/requirement to judge some practices as 'better' than others for adoption of effective approaches to the development of economic reform programmes and implementation of successful stock market performance and economic growth. As such, this research is based on two theoretical models. The first one concerns economic reform programmes, macro-economic reform and financial maturity variables related to stock market performance. The second model is based on the best practice transferable model from other emerging stock markets to the Libyan economy. These models are modified to develop the proposed research framework, as illustrated in Chapter Six. Based on the results of this study, the expected outcome of the research may be of value to other countries that have experienced political isolation and/or disturbance and that can provide direction to those embarking upon decisions relating to the development of stock market performance for increase the economic growth. In this respect, the results of the research may operate as a guide.

• This study is one of the first of this type to explicitly investigate the direction of causality (developed model chosen) between Libya and other emerging economies.

- Libya has some similar characteristics to other developing countries that have established their own stock exchanges very recently (for instance Sudan, Yemen, Syria, Iraq and UAE) and this study will make two contributions in this respect. Firstly, the methodology used in this thesis (including the statistical measurement of economic variables and qualitative analysis of the progress of other similar countries) can offer a method for these countries to follow. Secondly, the questionnaire developed as one of the research methods will be available for researchers to adopt with the need to make only slight variations to reflect national situations.
- Libya is still in the early stages of its financial liberalisation and reform, but it is moving quickly as a result of the removal of UN and US sanctions and there are signs of rapid development. However, as yet, there has been no study to explore the readiness of the Libyan financial market for the stock exchange, although this would seem to be an inevitable outcome.
- It provides recommended solutions for stock market problems through suggested effective economic reform programmes to enhance market-based financial systems where the influence of Arab it investment on a stock market would be minimised or improved.

Finally, this study's findings will be of help to other researchers and academics who have an interest in Libya's economy, stock market and other facts which could be considered as directions for further work. As a result, the following aims and objectives were determined for this study.

1.4 RESEARCH AIM, OBJECTIVES AND QUESTIONS

The aim of this research is to determine the most appropriate model for the continued viability of a stock market mechanism in Libya and to consider an appropriate strategy for the Central Bank of Libya to undertake successful continuation of stock market development. To achieve this aim, the following research objectives were formulated.

- **Objective 1:** To assess the success of the economic reform programme in Libya, specifically the deregulation, corporatisation, privatisation and liberalisation that has led to the inception of Libyan stock exchange.
- **Objective 2:** To identify the benefits of a stock market to the Libyan economy. This involves examining the relationship between macro-economic reform variables and stock market performance variables.

- **Objective 3:** To assess the performance of market-based economies, with particular reference to the emerging economy of Libya, and to evaluate current and best practice in financial deregulation.
- **Objective 4:** To determine whether best practice from other emerging stock markets is transferable to the Libyan situation and context.

In order to realise the primary aim and objectives of the study, several research questions were designed. The following questions are to be answered by this research.

- **Question 1:** Has Libya's reform programme prepared the country for developing a stock market and how effective are the post-sanction economic reforms (1999-2007) in changing the Libyan economy from a bank-based into a market-based one?
- **Question 2:** Are the emerging economies' models transferable to the Libyan situation and context?
- **Question 3:** What weaknesses in the structure of the Libyan economy may appear in the stock market? What are the main prerequisites for successful stock market performance in Libya?



Figure 1.1 Questions Related to the Aim and Objectives

Notice a: Research Objective (RO), Research Question (RQ).

Source: Developed for this research from Section 1.4: Research Aim, Objectives and Research Questions.

In order to achieve the first objective, the economic reform programme of deregulation, corporatisation, privatisation and liberalisation is described via the analysis of descriptive statistics. The findings are then explained in terms of correlations or differences between an

CHAPTER 1: INTRODUCTION

economic reform programme and stock market variables. Semi-structured interviews are also used to explore and explain themes that emerge from the analysis of self-administered questionnaire data. The findings are then compared, when appropriate, with those reported in other countries. In order to achieve the second objective, the relationship between macroeconomic reform and stock market performance variables is described in terms of the analysis of descriptive statistics. The findings are additionally explained in terms of correlations or differences between macro-economic reform and stock market performance variables. Semistructured interviews are used to explore and explain themes that have emerged from the selfadministered questionnaire survey. The Libyan findings are then compared, when appropriate, with those reported from various other countries.

In order to achieve the third objective, the aim is to explain and compare the impact of a bankbased system on the market-based system in Libya's transforming economy. This is also described via the analysis of descriptive statistics. The findings are explained in terms of correlations or differences between financial maturity and with particular reference to emerging economies. The semi-structured interviews were utilised to explain why respondents conduct the market-based system in a particular way. The findings are subsequently compared, when possible, with those reported in different contexts. With respect to the fourth objective, regression models were adopted to conduct a comparison between the Libyan context and those in other economies. Moreover, in many cases such a comparison is problematic. A comparison can still be interesting and may be illustrative.

1.5 RESEARCH SCOPE

Libya is still in the early stages of its financial liberalisation and reform, but it is moving quickly as a result of the removal of UN and US sanctions in the last few years, and there are signs of rapid development. However, as yet, to the best of the researcher's knowledge, there has been no study which explores the readiness of the Libyan financial market for the establishment of a stock market. There is currently a lack of published material and, in some areas, the existing literature is rather limited. Therefore, this study provides a contribution to the subject of Libyan economic growth by analysing the data necessary evaluate the success of the economic reform programme and Libya's readiness to complete its transition to a market-based economy. The focus on Libya is of interest for several key reasons. Firstly, the economic reform programme is now at the forefront of the government's agenda. Secondly, the Libyan government has been willing to develop the stock market and its recent policies of

engaging employees and managers alike in striving for reform, ownership change and foreign policy investment. Thirdly, the Libyan experience could provide new insights into the debate on the effectiveness of economic reform programmes and the underlying conditions for success. Finally, a comparison is made between Libya and the other emerging economies in introducing financial deregulation, including in the stock exchange. Ultimately, it is hoped that the study will identify and suggest more about the Libyan economic reform programme and the role of stock market performance in the nation's economic growth.

1.6 EMPIRICAL STUDY

The empirical component of this research consisted of two parts. A field study was implemented via a mixed quantitative and qualitative approach, as presented in Table 1.1. For certain aspects of the study, a quantitative strategy was used by the administration of a survey in order to achieve both the research aim and objectives (Cohen, 1992; Miles and Huberman, 1994; King, 2006). For other aspects of the research, qualitative data was obtained through semi-structured interviews (Remenyi et al., 1998; King, 2004; Saunders et al., 2007). A comparison between the situation in emerging market economies and the Libyan stock market was conducted³. The data are based on forty-two emerging markets which were obtained from the Emerging Market Database (EMDB) provided by the International Financial Corporation (IFC). Glen and Singh (2003) argue that a comparison of the financing patterns between countries is extremely valuable in analysing the economic reform programmes in developing nations. Comparison of the financing patterns between countries might provide empirical guidelines to execute such economic reform programmes. In this regard, assessing the Libyan economic reforms requires a body of empirical knowledge for providing a comparison with emerging market countries in order to achieve designed levels of stock market performance and to develop Libyan business behaviour.

³ Kim and Singal (2000) explained that emerging stock markets are defined by the International Finance Corporation as consisting of stock markets in developing countries (low-and middle-income economies).
Bessensh Methodology ^b	Objectives ^a						
Kesearch Miethodology	1	2	3	4			
Data collection methods							
• Literature	\checkmark	\checkmark					
Self-administered questionnaire	\checkmark	\checkmark					
Semi-structured interviews	\checkmark	\checkmark					
Cross-sectional regression (Datastream Database)							
Data statistical analyses							
• Descriptive statistics (<i>Means</i> , <i>SD</i>)	\checkmark	\checkmark		\checkmark			
• Skewness	\checkmark	\checkmark					
Kurtosis	\checkmark	\checkmark					
Chi-square-based measures	\checkmark	\checkmark					
• Spearman's Rho		\checkmark					
Hypotheses test	\checkmark	\checkmark					
Multiple regression model	\checkmark	\checkmark					
ANOVA	\checkmark	\checkmark					
Coefficient	\checkmark	\checkmark					
Collinearity diagnostics							
Two stage least-squares model							

Table 1.1 Research Methodology and Objectives

Notice a: The objectives of this research are discussed in Section 1.4 including the research aim.

Notice b: A tick $(\sqrt{)}$ indicates that the data collection method or statistical analysis was used in achieving a particular result from research objective.

Source: Developed for this research from Chapter Six: Research Methodology.

1.7 STRUCTURE OF THE STUDY

The thesis contains nine chapters, including this one, and is divided into three sections, as shown in Figure 1.2. *The first part*, a review, is theoretical in nature and continues with an introduction and literature review which provides extensive coverage of the theory and literature concerning economic reform programmes and macro-economic performance, stock market performance and economic growth. *The second part* is more empirical in nature and discusses the research methodology that was applied in undertaking this research. *The third part* describes the empirical studies and comprises the last three chapters, offering a discussion and analysis of the research, its conclusions and suggestions for further research.



Figure 1.2 Flowchart for Thesis Navigation

Key Chapters



Source: Developed for this research from Section 1.7: The Structure of the Thesis.

CHAPTER TWO

AN OVERVIEW OF LIBYA'S ECONOMY, STRUCTURAL CHANGES AND DEVELOPMENT PATTERNS

2.1 INTRODUCTION

After highlighting the purpose of this research study, discussed its background and explained the structure of the thesis in Chapter One, this chapter introduces the main characteristics of the Libyan economy and the changes that have taken place within it since the country became independent in 1951. There is also a focus on the 20-year period prior to the current time, when Libya was affected by economic sanctions imposed by the US and the UN. An assessment is made of the extent to which the Libyan bank-based economy has been transformed into a capital market. The chapter is set out as follows: Section 2.2 provides the historical background and geography of Libya, and in section 2.3 the basic characteristics of the country's economic reform programme are presented. Section 2.4 addresses the financial system, placing special emphasis on the Libyan stock market under the economic reform programme umbrella and, finally in section 2.5, a summary and conclusion are provided.

2.2 HISTORIC BACKGROUND IN LIBYA

The historical background in Libya can be discussed from two angles: geography and population; politics and economy. These two perspectives are now considered subsequently.

2.2.1 Geography and Population

The Great Socialist People's Libyan Arab Jamahiriya, still generally known as Libya, is located in the heart of North Africa besides the Mediterranean Sea, between Egypt to the east, and Tunisia and Algeria to the west. Sudan, Chad and Niger are to the south. The area of Libya is about 1,759,540 sq km (679,358 sq ml) and there is a coastline of some 2,000 km. It is the fourth largest country in Africa, the seventeenth largest in the world and is seven times the size of the UK. The climate is moderate in spring and autumn, summer tends to be very hot, whilst the winter is cold. The country is subject to a diverse climate with the Mediterranean Sea in the north and the desert climate of the south. Likewise, Libyan comprises desert (45 *per cent* with 25 *per cent* sand dunes), the rest being arable land (19 *per cent*), permanent crops (17 *per cent*), pasture (20 *per cent*) and forest (4 *per cent*) (UNEP, 2002). In fact, Libya is rich in natural resources (oil and natural gas) and other areas of iron ore, Sebeka salts and pastures.

Agriculture and fishing are the key industry sectors outside the oil and gas sectors (see Appendix A, Table 3.2). Finally, Libya's geographical location has aided the growth of trade networks and exchange of goods and products between Europe and Central Africa via Libyan local areas, such as coastal towns and oases (see Figure 2.2). The geographical location of Libya renders it a strategic link between eastern and western nations (see Figure 2.1).



Figure 2.1 The Libyan Geography and Location

Source: Libya Location Map (2008).

The population increased during the last decade from 3,225.1 thousand in 1980 to 4,524.4 thousand in 1990, 5,426.8 thousand in 2000 and, finally, reached an estimated 6,089 million in 2007 (IMF, 2007). The annual population growth rate during the period 1975-2005 was about 2.9, due to improved standards of living, and is expected to be 1.9 *per cent* during the period 2005-2015, as reported in Appendix A, Table 3.1.

2.2.2 Political and Economic

As mentioned previously, historical records indicate that the name 'Libya' is indigenous (i.e. Berber), the country being known as *Libu* in ancient Egyptian texts, a name referring to one of the Berber tribes living west of the River Nile in Egypt thousands of years ago. Given Libya's location at the strategic crossroads of Europe, Africa and the Middle East, it should come as little surprise that its history has been one of successive invasions. From the 10th century BC the Phoenicians inhabited the west of the country. The Greeks referred to the tribesmen as *Libyes* and the country became 'Libya'. It was also mentioned by ancient Egyptians during the era of Ramses II (1298-1232 BC). In the 6th century BC Carthage rose to prominence as a mighty state, and continued so until the 2nd century BC, when it was overthrown by the Romans. In the 5th century Libya was seized by the Vandals and in the 6th century it fell under the control of the Byzantines. In the 7th century Muslim Arabs conquered Libya. By the beginning of the 15th century the Libyan coast had minimal central authority and its harbours

were havens for pirates. Khdduri (1963) explained that, in 1550, in order to protect the Islamic faith from Christian Spain, the Muslim lands of Libya, Tunisia and Algeria invited the Ottoman Sultan to extend his control to their countries. The Turkish occupation began in 1551 and lasted until 1835. Simultaneously, Algeria was occupied by the French and Egypt almost achieved independence. In 1843 the Sanussi Order settled in Cyrenaica at the heart of the Bedouin tribes. By 1894 they moved to Al-Kufrah oasis from the Jaghbub oasis to control the Sahara trade routes, which passed through Zawaya. As Figure 2.2 shows, the highest degree of political and historical movement occurred within this period.



Figure 2.2 Political and Historical Landmarks

Source: Developed for this research from Subsection 2.2.

The weakening of the Ottoman Empire led to the first Italian War against the Turks in 1911. In 1912 Italian sovereignty was recognised and control of most of the land was established. Libyans, however, continued to fight the Italians until 1914. In 1949 the UN voted that Libya should become an independent country and consist of three separate areas: Tripolitania, Cyrenaica and Fezzan⁴. On 24 December 1951 the political policy became part of the United Kingdom of Libya. The new nation became a monarchy ruled by King Idris between 1951 and 1969. His political leadership derived from his role as a leader of the religious movement established by his family under the name of the Senussi Order. During the early years of

⁴ In this context, Tripolitania is today called Tripoli; Cyrenaica is Burka, and Fezzan is unchanged in name.

independence in 1951, the Libyan economy was under-developed and dependent upon British and US military treaties. For instance, per-capita income was between 30 and 40 million US\$ per annum. Given this low rate the levels of nutrition and health-care were severely affected. The Libyan birth rate was additionally low at about 5.3 *per cent* in 1952. The natural rate of population decreased to 1.1 *per cent* due to a death rate of 4.2 *per cent*, which was affected by the low standards of healthcare. In 1958 oil was discovered in Libya and by 1962 exports had commenced. Appendix A (Table 3.2) indicates the significant related contribution of the economic sector to GDP during the first nine years following independence and the ten years preceding revolution. On 1 September 1969, 27-year-old Muammar Al-Qaddafi deposed King Idris and revolutionised the country. The following main sections explain how the Libyan political situation altered and the manner in which the economy improved.

2.3 BACKGROUND TO THE LIBYAN ECONOMIC REFORM PROGRAMME PERIOD

After twenty years of political chaos⁵, and nearly three decades of central planning control of its economic system, Libya launched an economic reform programme under the framework of economic privatisation (UN, 2002). Since 1999 the Libyan government has been gradually implementing measures to reform and to follow an *open door policy*. These efforts are aimed at altering the country's state-controlled economy to a more market-oriented one in order to integrate it into the global economy, restructuring the economic mechanism from central administrative planning to mostly market-driven pricing and exposing the economy to the world through trade (oil exports and imports) and foreign investment.

⁵ On 27 March 1982 the US declared an embargo of the Libyan economy. The Reagan administration banned imports of Libyan oil to the US and halted all export services to Libya, other than food and medical supplies. In 1986 the embargo was expanded. In January Reagan ordered the freezing of Libyan assets in the US. In April 1986 Washington launched a series of air raids against targets in both Tripoli and Benghazi, which killed 101 people and struck Qaddafi's house, killing his adopted daughter, partially in an attempt to kill him. In June 1986 the US government banned exports to third nations of goods and technology destined for use in the Libyan oil industry. With the expansion of the embargo during 1986 all US companies withdrew from Libya with their assets frozen by Libyan authorities. 27 March 1987 saw liberalisation of the economy and loosening of the socialist structure policy. In September 1987 Libya lost its occupied territories in north Chad. In December 1988 Pan Am flight 103 exploded over Lockerbie in Scotland. In the 1990s the disagreement between Libya and the US escalated. Furthermore, as a result of changing international circumstances of the post-Cold War area, the US organised internationalisation of the conflict. In April 1992, the UN Security Council, led by the US, adopted Resolution no.731 and 748 which imposed a ban on civil aviation, a worldwide embargo on arms purchase and a reduction in Libyan diplomatic missions abroad. On 11 November 1993, the UN Security Council voted by eleven to zero to tighten the sanctions against Libya with Resolution no.883. For further information, see (Altunisik, 1996 and Appendix A 2.2).

2.3.1 Lessons and Challenge

Economic reform programmes have become the new phenomenon during the past two decades. Libya, Egypt, Tunisia, Jordan and Saudi Arabia, akin to most developing countries, shared similar experiences whilst striving to achieve the stabilisation of their economies. They did so because they found themselves incapable of earning sufficient foreign exchange to protect their currencies. Indeed, many of these nations began transforming their situations from socialist, state-owned entities to market-based economies (Frances and Garnsey, 1996). During the early 1990s for instance, Libya launched a market-based system which transformed it from being centrally planned as the public sector grew under the umbrella of socialist transformation and bureaucratic procedures controlled by laws and resolutions issued during the early 1970s and late 1980s. Leasing, however, provided opportunities for the introduction of managerial and technical skills that would allow the use of knowledge, ability and vitality value in a more efficient way within the lease duration. Libya, therefore, is developing an economic policy to improve management of the most productive public enterprises and to address the privatised ownership enterprises. The primary challenges facing Libya's are discussed in the following subsection. This is followed by an explanation of the outcomes of an economic reform programme.

2.3.1.1 Acceleration of the Transition to a Market Economy

In light of the previous discussion, Libya requires a comprehensive medium-term strategy policy to reform its economy and to make better use of its economic advantage and financial potential, through diversification of the economy and reducing the country's reliance on the oil sector. The proposed strategic policy aims to maintain macro-economic performance stability and rational use of the country's oil wealth, accelerating the transfer to a market-based economy and laying a solid foundation for development of the non-oil economy. The essential steps towards the restoration of confidence of the private sector and enhancing the country's economic potential, requires a sound macro-economic framework sending clear signals to the market underlying the commitment of the authorities to reform. IMF technical assistance has been provided⁶ in order to implement the necessary reform measures. To achieve this, priority should be given to the major steps leading towards an economic reform programme. In this research, as shown in Figure 2.3, these three characteristics are investigated, along with a

⁶ The IMF has already provided technical assistance in the areas concerned with monetary policy and bank restructuring, banking supervision, tax policy, revenue administration and statistical activities (IMF, 2006).

liberalisation programme, a variable which was not taken into consideration in any of the economic reform programme research.



Figure 2.3 Potential Economic Reform Programme

Source: Developed for this research from Subsection 2.3.1.1: Acceleration of the Transition to a Market Economy.

2.3.1.1.1 Deregulation Programme

The Libyan government initiated a change by passing new laws and regulations to motivate and organise the activities of the private and co-operative ownership of productive enterprises, as well as the shareholding companies⁷. This new form of ownership (i.e. in medium and small-scale industries) is known as "tashrukiya", an arrangement between private sector and state ownership, whereby each worker would share in capital and profit. This commenced with the significant reform law no.8 of 1988 concerning a number of economic activities which were allowed for the first time in 1977, when the private sector and investment limited and the economy also was under control. It was followed by law no.9 in 1992 on the practice of the Libyan business environment which allowed the establishment of joint-stock companies called "sharikah musahimah". Nevertheless, neither of these laws was sufficiently adequate to motivate the private sector as they merely caused a small growth in the service and crafts sector, while more critical sectors remained on hold. By the second half of the 1990s and the beginning of this century, another group of laws and regulations were passed by the GPC with the aim of expanding the role of the private sector, establishing private banks, adopting the liberalisation of the wholesale trade and directing its activities towards economic reform. These laws were additionally aimed at attracting foreign capital and achieving the required financial stability of the Libyan economy, as discussed. The following regulations provided most significant contribution to the Libyan economy.

⁷ According to La Porta et al. (1998) most MENA nations, particularly the North African countries, adopted the French law principles in their commercial law, as agreed by Kilani (1988) wherein Libyan commercial law has been based on the principles of French law. In Libya, therefore, during 1953 commercial law was issued in order to regulate economic and business activities. Libyan commercial law divided business into four types: simple limited partnership; general partnership; limited partnership with shares; joint stock companies.

- a) Law no.8 of 1988 is based on sentences related to economic activity. This laid the basis for restructuring the Libyan economy. The first clause allowed individuals to separately or co-operatively practise economic activities, including the distribution of products and services, which were in the past limited to public companies. The second clause addressed individuals, families, land ownership and factories, etc. The fourth clause asserted the application of commercial law and its affiliated regulations and decisions taken on issues which were not covered in law.
- b) Law no.1 of 1993 concerns banks and credit finance, replacing the banking law of 1963. This permitted the private ownership of commercial banks, besides allowing citizens to establish co-operative companies (banking companies), on condition that a company's capital was about 10 million LD⁸. It also permitted foreign banks to establish their representative agencies and offices in Libya and allow foreigners to acquire and maintain bank accounts in a foreign currency. Clause 13 of this law dictated that the Central Bank would be responsible for buying, selling and guaranteeing bonds and stock issued by the general treasury, payable within a 15-year period (at least).
- c) Law no.21 of 2001 defines the minimum limit of shareholders in shareholder companies, where it had become permissible to establish a shareholder company with 25 shareholders, if the total capital did not exceed one million LD. This positive step facilitated the establishment of shareholder companies for individuals, as cancelled law no.8 in 2001 had assumed a minimum of 500 shareholders for the establishment of a shareholders' company, which presented an obstacle against the formation of such companies.
- d) Law no.1 of 2004 focuses on the addition and adjustment of a number of rulings of law no.21 of 2001 on various economic activities. The former facilitated the establishment of shareholding companies by enabling them to issue carrier stock not only in name. It is further expected that these companies will be allowed to lower the minimum limit of shareholders and increase the percentage of individual holdings, in contrast to previous numbers decreed by law no.100 for 2001.
- e) GPC Resolution 134 of 2006 concentrates on the establishment of the Libyan Stock Market (LSM) and the issuance of the assets system. The establishment of the LSM is a major step on the path of developing financial resources via the development of a sound

⁸ Libyan Dinar (LD) denotes the Libyan currency and US\$ denotes the US Dollar.

capital market which is a critical issue and is discussed in detail in Subsection 2.4.3: Initiatives for a Libyan Stock Market.

On the whole, as shown in the previous discussion, the new economic reform stipulated the establishment of shareholding companies which are relied upon to lead economic reform in all sectors. The entire process lasted more than 12 years before reaching the current state, commencing with law no.9 in 1992, followed by law no.8 and no.21 in 2001, and law no.1 in 2004. All the aforementioned laws facilitated the establishment of shareholding companies, allowed the issuance of stock in their carriers' names, lowered the minimum limit of shareholders and increased the individual's share as opposed to levels previously dictated by law no.21 in 2001.

2.3.1.1.2 Privatisation Programme

Privatisation became one of the primary policies adopted by the IMF and the World Bank as part of its economic reform and structural adjustment programme to remedy deteriorating economic conditions, especially in developing nations. However, transfer of ownership in Libya began in 1987 by sharing the returns on investment with employees; profits were divided among the various elements of production (worker, machine, capital) according to a specific accounting system. The transfer was governed by the People's Committee Resolution no.447 of 1987, based upon a chapter from the Green Book (Partners not Workers), law no.9 in 1985 and GPC Resolution no.313 of 2003. The number of transformed units reached 295; the value of their assets equalled 98,000,000 LD. The transformation process continued from 1987 to 1992, as shown in Table 2.1. This saw many positive results, such as the retrieval of 75 *per cent* of the assets' value owned by these units, including the transfer of 30,000 products to the national sector.

Economic Sector	Number of Units	Value of Assets
Industrial	150	52,000,000
Agricultural	50	6,000,000
Marine fisheries	50	29,000,000
Livestock	45	11,000,000
Total	295	98,000,000

Table 2.1 Transfer of Ownership of a Number of Factories and Production Units, 1987-1992

Source: GPC Decision no. 313 in 2003 approving the public sector's restructuring programme.

Following cessation of the 1992 transfer process there was an eight-year gap until it resumed with the subsequent GPC decree no.313 in 2003 approving the public sector's restructuring

programme and enhancement of the ownership base. The Committee additionally issued the list of designated public companies and economic units no.31 of 2003, including a time frame for the transfer process guided by a legal and administrative framework concerning partial and overall levels. Ownership was transferred from 360 economic units into agricultural, industrial, livestock and marine fishery sectors; the value of their total assets equalled 8 million LD, and 100,000 plus employees worked in these economic units. This programme was executed over a period of time between 2004 and 2008 in three stages, as illustrated in Table 2.2.

Table 2.2 Number and Classification of Units that will be Transformed between 2004and 2008

Economic Sector	First Stage	Second Stage	Third Stage	Total
Industrial	145	41	18	204
Agricultural	28	4	24	56
Marine fisheries	11	0	11	82
Livestock	16	1	1	18
Total	260	46	54	360

Source: Production affairs transfer of ownership in Libya (2004).

2.3.1.1.3 Liberalisation Programme

During the past five years Libya has witnessed many changes. Until recently the country had no private sector and foreign entities could not own or lease property from individuals and there was a state-controlled monopoly on the purchase and distribution of consumer goods (US Commercial Service, 2006). In 1987 the initial signs of economic liberalisation allowed the Libyan government to quietly found the *Tacharukiyat* activity (Joint Stock Company). Subsequent regulations allowed for *'economic practice activity'* and the establishment of private banks. Since 1992, when the regulation on privatisation was issued, the sale of state property of the Libyan government was permitted. Many sources point to Act no.5 of 1997 which encouraged foreign investment, a landmark event that defined the conditions for foreigners investing in Libya.

On 1 January 1998 the Libyan government signed several agreements for foreign trade. One of these signalled the great free Arab market, which started to eliminate customs fees on Arab commodities. Within a decade it was expected that there would be free movement of trade among the countries that signed this treaty. Libya then began to exempt Arab commodities from all customs fees and officially submitted an application in 2001 to join the World Trade Organisation (WTO) (Alfaris, 2003). The WTO's aims are to continue the world General

CHAPTER 2: AN OVERVIEW OF LIBYA'S ECONOMY

Agreement on Tariffs and Trade (GATT) in agreeing international trading rules and furthering the liberalisation of international trade (Goldstein et al., 2007). WTO membership assures access to the entire world market for Libyan manufactured products. This will encourage Libyan production in industry, agriculture or services under competition with foreign products in the Libyan market including intellectual property rights and institutions to promote greater trade and investment. The aim of this opening to the external world is to increase the efficiency of foreign production facilities and create competition within the Libyan market.

By 1999 considerable changes in the economy started to occur, specifically within the banking sector. The most critical step was the adoption of the BPC resolution no.1 in 1999 concerning banking law and credit currency. However, laws were issued from the trade economy organisation. Most recently, in 2003, the Libyan President, Colonel Al-Qaddafi, described the socialist experiment as a failure, calling for privatisation of the public sector, including the oil industry and the banking sector. He also called for the application of the alternative, social capitalism, in the form of companies owned by Libyan citizens. These new entities had the right to utilise foreign expertise in order to liberate the Libyan economy. In September 2005, he asked the public *"With the exception of oil, show me what other goods are produced by Libyans and which can draw money to the country?"* (Porter and Chairman, 2006: 10). He further stated *"We have to produce and make a profit."* (Porter and Chairman, 2006: 10). However, he arrived at the conclusion that *"the stability of the regime was threatened"* (Altunisik, 1996). Hence, the economic reform programme was concomitant with political policy reform.

2.3.2 MACRO-ECONOMIC REFORM PERFORMANCE

The challenges of devising a strategic policy were aimed at achieving a sustainable development of the Libyan economy. One of the main factors that consolidate this fact is that annual population growth rate during the period 1970-2000 was about 3.8 *per cent* due to improved living standards (CBL, 2000). However, measuring the success of the economic reform programme in the Libyan economy is via macro-economic reform variables: exchange rate, interest rate, inflation rate, budget deficit, per capita income and GDP growth, as Table 2.3 displays.

Macro-economic Reform Variables	1999	2004	Change
Exchange Rate (million LD)	3.1	1.31	0.81
Interest Rate (per cent)	2.7	5.5	2.8
Inflation Rate (per cent)	3.5	-1.1	-4.6
Budget Deficit (billion LD)	3.1	2.4	-0.7
Per-capita Income (thousand LD)	3.1	4.2	1.1
GDP Growth Rate (per cent)	1.1	4.3	3.2

Table 2.3 Measuring the Macro-economic Reform Programme in Libya, 1999-2004

Source: IMF, Country Report no.05/83, Washington D.C. March 2005.

As shown in Table 2.3, the rate of changes in these variables was dramatic between 1999 and 2004. For instance, there is a longer budget deficit and real GDP rose by 3.2 *per cent* as a result of the privatisation of many state-owned companies. The economic reform programme was fruitful in Libya in that these changes of variables transformed the nearly-paralysed economy of the sanction period to one that is better performing. Further, the classification of macro-economic reform development extends to six main types: exchange rate; interest rates; inflation rate; budget rate; balance of payment; per-capita income; GDP growth rate. How do these variables have a positive impact on stock market performance variables and how significant is the sound performance of the stock market to any economy? Was growth rate improvement assisted? The indication variables will be addressed in Chapters Seven and Eight.

2.3.3 The Results of Economic Reform Programme

As already indicated, the challenges mentioned are beginning to be addressed. For instance, the pace of economic and structural reform has improved since the lifting of the UN and US trade sanctions when other developing countries' rates grew worldwide with an average annual growth rate of real GDP around 0.3 *per cent* since 1999. This means that real Libyan GDP was 5.5 *per cent* times its 1999 value in constant prices by 2007 (see Figure 2.4). Libyan real GDP per-capita also increased by 4.7 *per cent* during this period. At the same time, the increasing openness of the national economy meant that domestic growth was accompanied by oil exports. In 1999 total exports were relatively small at less than 36,822 million LD. Reform accomplishments include the passing of laws to encourage private investment, both domestic and foreign, the abolition of customs duty exemptions enjoyed by public enterprises, and the reduction of customs duties. Clearly, Libyan exports increased at an average annual growth rate of 13 *per cent* largely shifting from primary products to manufacturing. By 2004, total Libyan exports had reached over 208,483 million LD, while imports during this period, grew at much the same rate, on average. These increased from 19,286 million LD in 1999 to 82,552

CHAPTER 2: AN OVERVIEW OF LIBYA'S ECONOMY

million LD in 2004 in real volume terms. The opening of its doors to the outside world, especially to European countries, prepared Libya for thorough involvement in the global economy, thus rendering Libyan economic policy more relatively significant in the international economy framework.



Figure 2.4 Percentage of Real GDP Growth and GDP Per-capita in US\$, 1980-2007

Source: IMF, WEO Database, April 2007.

2.3.4 Foreign Direct Investment in the Libyan Economy

Since 1997, the Libyan government has actively sought and encouraged an increase in foreign private investments. Law no.5 of 1997 and its amendment in 2003 by law no.7, which offered many incentives and numerous guarantees to attract foreign business to participate in realising economic development in Libya, created a new website *www.investinLibya.com*. The aims of this law, specified by article one, included: transfer of modern technology to expand the Libyan technical system; diversification of income sources; contribution to development of national products for international markets. Table 2.4 illustrates the distribution of Foreign Direct Investment (FDI) among the various economic activities. In 2000 the total FDI in Libya reached 189,150 million LD and then declined to 121,323 million LD in 2002. In 2005 it had reached 521,333 million LD, with an annual growth rate of 143.9 *per cent* from 2000 to 2005. As for the distribution of these investments among the various non-oil sectors, the following facts can be depicted.

The industrial sector occupies first place in attracting FDI. In 2000, 65 million LD was invested in this sector, which comprised 34.4 *per cent* of total direct foreign investment. This figure rose to 73,636 million LD or 60.7 *per cent* of total investment in 2002 and, finally, amounted to 399,136 million LD or 76.6 *per cent* by 2005.

- The tourism sector ranks second in attracting FDI. In 2000 the total incoming investment for this sector was 124,150 million LD. In other words, this represented 65.6 *per cent* of total FDI that year. By 2003 incoming investment decreased to 112,498 million LD, the equivalent of 60.3 *per cent* of total FDI.
- In third position for attracting FDI is the health sector. Investment here reached 29,636 million LD, or 24.4 *per cent* of total FDI in 2000 and rose to 65,600 million LD, or 12.65 *per cent*, in 2005.
- FDI in the service sector reached 17,401 million LD, or 14.4 *per cent* of total FDI and increased to 51,872 million LD, or 9.9 *per cent* of total investment, in 2005.
- FDI in the agricultural sector was 65 million LD in 2002, which was equivalent to 0.5 *per cent* of that year's total FDI. By 2005 this figure rose to 4,725 million LD, or 0.9 *per cent* of total FDI.

Table 2.4 Development in the Size of FDI in Economic Sectors, 2000-2005

(Million LD)

Sectors	Indust	rial	Agricu	ltural	Touri	sm	Heal	th	Servi	ce	Total
Years	Value	%	Value	%	Value	%	Value	%	Value	%	Value
2000	65,000	34.4	NA	NA	124,125	65.6	NA	NA	NA	NA	189,125.0
2001	14,763	100	NA	NA	NA	NA	NA	NA	NA	NA	14,763.0
2002	73,636	60.7	0,650	0.5	NA	NA	29,636	24.4	17,401	14.4	121,323.0
2003	33,979	18.2	NA	NA	112,498	60.3	30,175	16.2	3,557	3.3	108,209.0
2004	155,047	83.4	4,960	2.7	NA	NA	10,556	5.7	15,190	8.2	185,753.0
2005	399,136	76.6	4,725	0.9	NA	NA	65,600	12.6	51,872	9.9	521,333.0

Source: Investment Enhancement Agency, Annual Reports (2000-2005).

Finally, the size of FDI is still limited for the various economic sectors. This is related to the investment environment that is constricted by the burdens of administrative organisation and bureaucracy; privatisation is very slowly progressing and the infrastructure is inadequate. This is in addition to the fact that the local market has yet to develop appropriately.

2.4 FINANCIAL SYSTEM

Demirgüc-Kunt and Levine (1999) examined the relationship between the financial development and financial structure of more than 150 countries. They used a large selection of indicators that measure the activity, size and efficiency of financial intermediaries and stock markets. Their studies provide international comparisons regarding economic development and banking, non-banking and stock market development. According to their analysis, Libya's financial sector is still forming among countries that have financially and economically

developed⁹. An analysis of the financial sector is now provided in the following three sections which consider respectively: the development of the banking system (currency market), monetary policy, and the establishment of the Libyan stock market.

2.4.1 Development of the Banking System

The establishment of banks in Libya began gradually until an agriculture bank was founded in Tripoli in 1901 during the rule of Hafidh Pasha (1900-1905) with several branches inside the Tripoli governorate; the Othman Bank established two branches in Libya, the first one commenced business in Tripoli in 1906 and continued until 1913, while the second began business in Benghazi in 1911. These two branches stopped practising their banking activities in 1912 because of the Italian occupation of Libya in 1911 (Masoud, 1989). In 1951 the CBL was established, called the Monetary Authority (MA), under the supervision of the Ministry of Finance (MF). However, in 1956 the MA changed its name to the CBL (CBL, 2005). The CBL had its headquarters in the capital city of Libya (Tripoli) and included three branches nationally: Benghazi, Sirte and Sebha. By the end of 2006 the capital of CBL was 500 million LD and total assets of 83.682 billion LD were recorded (CBL, 2006). In the late 1950s and early 1960s Libyan bankers learnt from other international banks, such as those of the UK, the US and Italy who represented the successful banking operators in Libya at that juncture.

Since the 1969 Revolution most of Libya's financial institutions were subsidiaries of foreign organisations, mainly overseas banks. In December 1970, liberalisation of the Libyan banking system occurred. The Libyan sector is mainly in the hands of the public sector, which represents 90 *per cent* of national banking activities and business. Prior to the start of privatisation, there were three banks: the National Commercial Bank (NCB), Jamhouria Bank (JB) and Umma Bank (UM) including shares in the capital of Wahade Bank (WB) of 87 *per cent* and Sahara Bank (SB) 82.7 *per cent* (IMF, 2006). The private sector owns four banks: the Bank of Commerce and Development (BCD), Aljma'a Alarabi Bank (AAB), Alwafa Bank (AB), Aman Bank (AB) and one mixed bank, Regional Banking Corporation (RBC), plus a further 48 small regional banks. The specialised banks are, Libyan Foreign Bank (LFB), Agricultural Bank (AB) and Saving Real Estate Investment Bank (SREIB). Table 2.5 reveals

⁹ Samuels (1981) provided five characteristics to identify the repressed financial sector: interest rate ceilings; high reserve requirement on bank deposits; compulsory credit allocations by way of quantitative ceiling; subsidised loans; unaccountable lending institutions.

the distribution of approximately 269 branches and agencies of the commercial banks in Libyan financial institutions.

Municipalities	NCB	WB	UB	JB	SB	BCD	Total
Benghazi	5	14	3	7	8	2	39
Jabel Algebee	5	9	7	7	1	NA	29
Jabel Akhdar	14	10	1	6	5	1	37
Sebha	7	1	5	7	4	NA	24
Sirte	7	12	9	11	4	2	45
Tripoli	15	16	18	22	11	2	84
Zawia	5	7	7	6	6	1	32
Total	58	69	50	66	39	8	290

Table 2.5 Number of Libyan Commercial Banks' Branches and Agencies

Source: Alqadafi (2002).

Development of the commercial banks operating in Libya during the period 1970-2003 is shown in Figure 2.5, which shows that liquid assets increased from 31.9 million LD in 1970 to reach 32,651 million LD at the end of 2003. The balance of promissory notes and permission of the local treasury amounted to 14,377 million LD at the end of 1995; the value of fixed and other assets amounted to 312 million LD in 1970 and grew to 13,955 million LD at the end of 2003.

Figure 2.5 Indicators of the Development of Commercial Banks in Libya, 1970-2003



Source: CBL (1980, 1999, 2005)

2.4.2 Monetary Policy

The CBL is equipped with the usual monetary instruments. It is the sole issuer of currency; it re-discounts bills for banks, administers gold and foreign exchange reserves, regulates all financial institutions, establishes the level and structure of interest rates and controls the quantity and quality of bank credit. Growth rates over the period 1962-1982 involved a swift expansion in bank credits to the private sector. This increased (in real terms) at an average rate of 15.6 and 21.5 *per cent*, respectively, over the two oil boom periods (CBL, 1985). Specialised Credit Institutions (SCI) were established by the CBL to provide medium and

long-term credit to the government institutions and liquidity credit to commercial banks for the financing of development projects in various Libyan economic sectors, particularly agriculture, industry and housing. The specialised credit institutions comprise the Libyan Arab Foreign Bank (LAFB), the Agriculture Bank (AB), the Savings and Real Estate Bank (SREB) and the National Development Bank (NDB). Throughout this period the goals of monetary policy were to reduce inflation, promote economic growth, mobilise domestic savings into productive investments and stabilise the LD. In order to promote economic growth and combat inflation, the CBL began the policy of limiting credit to consumer goods and increasing credit to productive projects. As these favourable developments and the monetary policy were adopted, it was a narrow M1 money supply, which included the currency in public circulation, and the demand deposits in LD, which had declined from 46.6 *per cent* in 1970 to 27.5 *per cent* in 1982 (see Table 2.6).

 Table 2.6 Changes in Money Supply and Factors Affecting, 1970-1990

Factors	1970	1973	1975	1980	1982	1988	1990	1970 1982	1983 1990
Net Claims on the Public Sector	277.0	172.0	256.0	382.0	479.0	376.0	1201.0	29.02	30.0
Net Claims on the Private Sector	94.4	246.1	645.3	618.0	677.1	668.1	658.3	1.5	0.5
Net Foreign Assets	564.7	580.3	772.2	2439.0	987.6	347.6	402.4	17.8	-6.9
Money Supply	237.3	455.7	867.6	1565.0	1380	852.2	1145	17	-1.0
Currency (per cent) of Money Supply	46.6	39.4	39.9	56.9	27.5	29.90	32.80	NA	NA

(Million LD; Constant Price)

Source: Calculations from CBL, statically series and monthly bulletin, various issues.

2.4.3 INITIATIVES FOR A LIBYAN STOCK MARKET

As hitherto mentioned, with the introduction of an "open door policy" during 1999, the Libyan economy considered how economic reform programmes may benefit businesses and how they may be managed. They reactivated private sector ownership and aimed to attract foreign investment including mobile domestic investment and allocation of resources. Additionally, the Libyan economy sought guidance from emerging countries, those which have undergone the experiment of financial markets. These issues are discussed subsequently whilst Appendix A (A.2) shows more details about the Libyan stock market.

2.4.3.1 Umbrella: Libya's Stock Market under the Economic Reform Programme

As discussed earlier, in Chapter One, since 1999, economic reform measures have been adopted encompassing almost all economic areas: restructuring of banking rules and bureaucracy, fiscal and monetary reform, foreign investment law and dispute resolution. The objective of the reforms was to achieve the much-needed skills of citizens and enhance the knowledge of workers to better meet the demands of the marketplace. Furthermore, the planned economic reforms include comprehensive economic liberalisation and mobilising savings by removing any constraints hindering stock market development. Libya's stock market was established with the crucial remit of rising capital under an economic reform programme among corporations and investment abroad that has made significant contribution to the Libyan economy's growth. With reference to the encouragement of savings and investment, the market's role is responsible for offering investors the opportunity to participate, in particular, in the significant development of the private sector in Libya and to share in the growth of its first companies' ownership, thus realising economic goals and objectives.

2.4.3.2 Establishment of an Organised Stock Market in Libya

Similarly, there has been an increasing demand to establish an organised stock market in Libya. Historical and political circumstances indicate that the establishment of the first stock company occurred during independence in 1951. Later, in 1953, when the Libyan commercial code (LCC) was enacted, the trading of stocks was allowed. More specifically, the stock markets did not really have any significant impact until the early 1990s, when a substantial increase in the number of companies occurred, and most Libyan companies were unable to sell additional shares issued for financing further capital. In 1992/93, the state started to restructure the major sectors in the economy; several industries were privatised and investment was encouraged. This created a great number of new shareholders. Libya, however, needed to develop an efficient stock market in order to handle the large volume of share trading likely to arise. New regulations concerning commercial activities, bank currencies, credit and privatisation of many of the public units and companies were passed (Decision no.9 in 1992, no.1 in 1993, and the decision of the GPCs issued in 1993) for the freedom of ownership of general economic units and establishments. Such steps stressed the need to establish a stock market in Libya.

1. GPC Resolution 198 for 2001 on the establishment of the General Agency of Ownership of Companies (GAOC) and Public Economic Units (PEU): this association undertook responsibility for transferring ownership of economic units from the public sector to the private. The tenth clause of this law dictated that commercial banks and other legally authorised agencies might act as mediators to handle the buying and selling of economic units of stock owned by the worker, without the right to own any of the traded stock. The

agency would arrange the setting of the necessary rules and regulations with the CBL to put in motion the sentences of this clause.

2. This law presented a distinct message to allow the trading of pre-owned economic units of stock and commence the establishment of a financial market. A clearer invitation to establish a stock market in Libya came with law no.21 in 2001, defining the practice of economic activities and execution of regulations. The tenth clause of this law dictated that the Board of the General Committee (BGC) would decide upon certain agencies to form a stock market according to the conditions and procedures defined by this law's regulations.

On 3 June 2006, GPC Resolution 134 regarding the revised establishment of a stock market in Libya determined its Memorandum Agreement. The fourth clause of the stock market law dictated market membership, including the following entities:

- Commercial and specialised banks
- Insurance companies
- Financial funds
- Subscribed companies, with capitals that are no less than the limit stated by the market's administration committee.

The Libyan stock market consists of nine departments:

- 1. Trading Department
- 2. Central Listing and Depository Department
- 3. Membership and Administrative Affairs Department
- 4. Financial and Administrative Affairs Department
- 5. Research Studies and Development Department
- 6. Control Oversight and Supervision Department
- 7. Legal Affairs Department
- 8. Public and Quality Relations Department
- 9. Communications and Information Systems Department.

The establishment of the Libyan stock market as a stock exchange was a major step in the contribution towards improving financial institutions within the economy, in order to enable it to realise a better utilisation of financial resources by mobilising domestic savings, encouraging foreign savings and channelling such resources towards productive projects through the development of a sound capital market.

2.4.3.3 The Stock Market Structure

The current structure consists of two types of stock market: a primary and a secondary market. The former is one where shares are issued by corporations or other institutions to the public. In Libya, securities being offered to the public are governed by the Companies Act, 1976. This Act, administered by the Ministry of Commerce, is a crucial piece of stable legislation in the Libyan financial institution, which also defines the joint-stock company which issues negotiable shares of equal value to their members. The secondary market is a locus where shares already in circulation are traded once they have been issued in the primary market. GPC no.100 in 2004 and Resolution no.180 of 2004, as shown in Appendix A, Table 3.11, were issued and transferred to ownership companies for selling by the Libyan government as follows:

- Issuing the company's shares at market value in the stock market at an initial price determined by re-evaluation results besides the number of shares
- Adjusting the company's capital for net ownership rights according to re- evaluation results and issuing it in the stock market in the form of nominal shares
- Establishing new corporations to which the net asset value of the company is transferred after evaluation and issuing shares in the stock market for exchange at face value.

During 2006 the total trading volume in the stock market was only 355,840 LD related to the Sahara Bank, Suq Al-Gumaa Regional Bank, Hay Al-Andalus Regional Bank, Cement Regional Company and Muttahida Insurance Company. Other listed companies were not traded. The Sahara Bank, for instance, traded 18,898 shares on the stock market in 2006, with an average price ranging between 8 and 10 LD per share, which led to increasing the total value of shares traded to about 170,720 LD. In the meantime, shares of the Muttahida Insurance Company amounted to 1,822 with a total value of 182,200 LD with an average price of 10 LD in 2006. Likewise, Tripoli Regional Bank shares traded totalled about 277 with an average price of 10 LD and a total value of 2770 LD. Table 2.7 presents companies listed in the Libyan stock market, whose shares have been traded by buying and selling over time.

Company	Data	Subscribed Capital (LD)	Shares
Assaray Bank	20/08/2005	3,000,000	300,000
Sahara Bank	13/12/2005	126,000,000	12,600,000
Suq Al-Gumaa Regional Bank	19/01/2006	3,000,000	300,000
Hay Al-Andalus Regional Bank	30/05/2006	3,000,000	300,000
Cement Regional Company	80/06/2006	600,000,000	6,000,000
Libyan Insurance Company	24/06/2007	50,000,000	10,000,000
Muttahida Insurance Company	29/06/2007	10,000,000	100,000

Table 2.7 Companies Listed in the Libyan Stock Market

Source: Libyan Stock Market Report (2007).

The Libyan stock market, upon which this research focuses, can be considered an auction market, not a negotiated market, where the arena in Libya is concerned with trading stocks of companies listed on the stock exchange. All transactions take place through financial intermediaries called brokerage firms and brokers collect commissions for their efforts; no type of transaction can occur in the national stock market which involves dealers directly. Indeed, the banking sector plays a key role in the stock market in Libya in both the primary and secondary markets. In the former, most of the initial public offerings are conducted via the banking sector. They suggest appropriate details to the relevant companies and all phases of subscription are conducted by banks. This differs from the system in the US where commercial and not investment banks are responsible for such transactions. In fact, these banks, and not their investment counterparts, are responsible for such transactions and they offer the individual investors financial assistance to purchase securities. They participate in increasing activity by providing more liquidity to investors in the secondary market.

2.4.3.4 The Main Obstacles to a Sound and Efficient Stock Market in Libya

As explained in early discussion, Libya, like many other countries, had long suffered from deficiencies within its economy, especially in areas such as inflation, balance of payment deficit, low rates of employment and growth, all of which created an imbalance in the economy. Problems of irregularity in supply and demand had a negative impact on prices and balance of payments. Similarly, fiscal policy tended to exceed the level of expenditure causing an imbalance between population density and growth and employment, resulting in structural distortion in the employment market. Likewise, severe decline in international oil prices during the 1980s and the effect of UN and US sanctions against the Libyan economy during the early 1990s have been explained in order to clarify the financial crises that Libya faced in the early 1980s and 1990s.

Cobham (1995) explains that stock markets in developing countries can suffer from problems. He identifies problems such as an imperfect flow of information and high transaction costs because the economies are too small to benefit from the economies of scale that characterise the major stock markets of the world. Cobham (1995) emphasises that in the developing countries there must be a strong preference for a bank-based rather than a market-based system. Furthermore, Fry (1997: 754) points out that, "*At best stock markets play a minor role; more often they resemble gambling casinos and may actually impede growth in developing countries*". Despite the recognition of the importance of the stock market for Libyan economic development, and the limitations which the absence of a stock creates, it was noted by respondents to the questionnaire survey that many obstacles exist to the development of an efficient stock market in Libya (see, Appendix K: Subsections 3.4.4, 3.4.5 and Table K 4.4). Among the limitations and obstacles identified by the respondents are:

- Difficulty of prediction and domination of uncertain circumstances, owing to the issue of many rules and regulations without feasibility studies being conducted;
- The weakness of public awareness of the role of investment and the role of the stock market in the economy;
- The Libyan banking system is not sufficiently well developed;
- Libyan privatised companies were valued by unfair evaluation techniques due to the loss of basic trading information;
- Most Libyan companies are unable to sell additional shares issued for financing further capital investment;
- The short and long-term investments opportunities for Libyan companies are hindered;
- Lack of efficiency and development of financial information institutions and a financial press in this field.

2.5 SUMMARY AND CONCLUSION

The purpose of this chapter was to shed light on the characteristics and development of the Libyan economy since the revolution of 1969 in order to provide a framework within which the study's findings are to be interpreted and understood. Following the 1969 Revolution, the economy witnessed three major systems: the nationalism and socialism period, an open door policy period and, latterly, the economic reform programme. The economy experienced rapid expansion during the 1970s and early 1980s as real GDP grew by more than 10 *per cent* on average. This expansion was mainly financed by the oil revenue sector. In the mid-1980s economic growth slowed and the Libyan government started to experience reversionary trends. These later trends saw the collapse of oil prices and the Gulf war in 1990/1991.

CHAPTER 2: AN OVERVIEW OF LIBYA'S ECONOMY

Following the collapse of its economy, Libya was forced to re-examine its policies and redirect its development strategies. The country was struggling with economic and financial crises by the mid-1980s and international UN sanctions following US Libya-specific trade sanctions in the early 1990s. At this time, traditional remedies were becoming ineffective. The challenges of devising a strategic policy aimed at achieving the sustainable development of the Libyan economy are particularly linked to current official tendencies for restructuring the economy through increasing the role of the private sector and by continuing the process of privatising the public sector within the country.

Finally, it can be argued that, since implementing its economic reform programme, Libya's wealth has grown strongly amongst the country's small population due to oil price increases. Economic development has two significant factors: a high value/low employment energy sector and a low value/high employment non-energy sector. The latter contributes about 40 *per cent* of GDP employing 97 *per cent* of the formal workforce ranked at a very low level of productivity. Oil revenues accounted for about 90-95 *per cent* of exports and over 60 *per cent* of total GDP with employment of about 3 *per cent* of the population. The size of the informal economy is estimated at 30-40 *per cent* of GDP. Also, under an economic reform programme called "umbrella" and an open door policy, the stock market was established. This, subsequently, became successful and entered a new economic phase bearing the labels of economic openness and freedom. The following chapter will discuss the previous literature in the economic reform programme which related to the first and second research objectives.

CHAPTER THREE THE ECONOMIC REFORM PROGRAMME

3.1 INTRODUCTION

Having discussed the main characteristics of the Libyan economy and the changes that have taken place within it since gaining independence, the thesis now moves in this chapter, to present an inclusive discussion of the literature that has been reviewed to help develop the economic reform model required for the development of a stock market and economic growth. In addition, the chapter indicates the reasons for adopting a programme of economic reform, pointing out the urgency for most developing economies to accelerate their growth rates and control the problem of poverty. Other aspects are also discussed and the main variables of economic reform and macro-economic performance, and strategies for assessing these when evaluating the performance of such programmes are considered. Furthermore, attention is drawn to the comparison between the experimenting countries from which Libya could learn.

Thirty years ago the world was divided into two main economic blocs: the first followed the direction of capitalism, which emphasises a market-based economy, freedom and deregulation and allows for privatisation and investment; the other developed centrally-planned, state-owned economies. Within the past 15 years the capitalist economy had proved to be the most effective method of managing any economy in the country of the world. Most of the old socialist and communist economies, such as the Soviet Union and Eastern Europe (e.g. Czech Republic, Hungary, Slovakia and Poland),¹⁰ are now conducting partial or entire economic reform programmes that will last for decades. With the collapse of communist regimes in late 1989 in Eastern and Central Europe, symbolised by such events as the fall of the Berlin Wall and the execution of President Ceauşescu of Romania, and the disintegration of the Soviet Union in 1991 newly independent countries in Africa, the Middle East and Latin America realised the importance of such programmes, which were mainly supported by international organisations such as the IMF and World Bank (Williamson, 1994). When the first post-communist government in Poland came to power in August 1989 it adopted a radical strategy

¹⁰ The four countries (Czech Republic, Hungary, Slovakia and Poland) are referred to as the Visegrad countries after the town in Nothern Hungary that held a meeting of the presidents of Czech Republic, Slovakia, Hungary and Poland on 15 February 1991. They signed a declaration of close co-operation during their post-communist transition economy and on the path towards EU membership (Fidrmuc et al., 2002: 2).

CHAPTER 3: THE ECONOMIC REFORM PROGRAMME

for rapid transformation from socialist-planned to market economy following the capitalist model of Western Europe. This strategy has also been adopted in modified forms throughout the rest of Eastern Europe and the former Soviet Union, including Russia post-1991 (Sachs, 1994). According to Williamson (1994), economic reforms have offered a global comment on the world economy, namely the social and political dimensions of transformation policy. The majority of developed and developing countries, regardless of geographic entities, income levels and ideologies, have joined mainstream economic development. As Wood (1990) remarked:

"It is necessary to distinguish between nominal reform, meaning official policy initiatives, and real reform, meaning changes in how the economy actually works. Nominal reform can contribute to real reform (price decontrol being an example), but does not do so if it outstrips microeconomic learning capacity. In other words, the real pace of reform may be limited by how rapidly people and institutions can learn to play new economic roles."

This chapter is divided into seven main sections. Section 3.2 deals with the various definitions of economic reform. Section 3.3 draws attention to the importance, aims and objectives of economic reform programmes. Section 3.4 assesses economic reform programme variables and macro-economic reform variables are also described. Section 3.5 deals with the process of economic reform. In Section 3.6 many points of view from the literature are explained, drawing on the experiences of numerous countries worldwide which have already effected reforms to improve their economies. In Section 3.7 attention is drawn to the comparison between experiment countries. This is followed by a summary and conclusion in Section 3.8.

3.2 DEFINITIONS OF ECONOMIC REFORM

At the end of the Second World War it was not by accident that, in referring to the process of post-communist transformation, the word "revolution", formed by combination of the words "reform" and "evolution", was utilised (Gil, 2003 and Ferrero, 2004). There are numerous terms which possess the same meaning as "economic reform programme" and many authors use the same phrases of "economic reform programme" (for example, Marks, 1995; Cronin, 2001; Krueger, 2005). However, expressions are also used such as "economy and transition" or "transitional economy" (e.g. Kolodoko, 1993; Nove, 1994; Lavigne, 1995; Jurajda and Terrell, 2000; Karsten, 2005). Even so, other authors prefer the expression "transformation" which has the same meaning as economic reform programme and transition (Blommest and Marrese, 1991; Francis and Kompas, 2001; Akin and Demircioglu, 2005; Daniel and Yang, 2006). Similarly, some authors define this process as "economic adjustment, stabilisation, and

CHAPTER 3: THE ECONOMIC REFORM PROGRAMME

regime" (Balcerowicz and Gelb, 1994; Sarma and Gupta, 2002). Nevertheless, most economists are accustomed to using the first two expressions "economic reform programme" and "transition" (e.g. Gordon, 1991; Hughes and Lovei, 1999). Accordingly, it was determined via the literature review that most articles and books concerning developing nations, such as China and Eastern European regions, adopt the expression "transition economies", which determines the programme of reform as the transition from a capitalist to socialist model or to a capitalist market economy (Jeffries, 1993; Bishop et al., 2004; Bouev, 2004 and Stein, 2005) and to the transition from capitalism to post-industrial, post-economic society (Buzgalin, 1994 and Brown, 1995), the transition from Stalinism to the socialist market economy (Nolan, 1980) and the transition to the drastic growth of production functions in the US (Felstein, 1980). Similarly, for Middle Eastern countries, such as Egypt, Israel and Jordan, the expression of "economic reform programme" is generally applied. On the other hand, in China and Eastern European countries the "process of transition" has involved, besides economic changes, shifts in the political system. Here, political factors led to economic change in these nations. One further example is the general approach to the "process of transition" based on the "institutional analysis of the evolution of transition to development" applied to disparate countries (Intriligator et al., 1999). Since this research concentrates on the Libyan economic situation, the expressions "economic reform programme" and "transition" will be adopted referring to various kinds of changes involving economic, political and social factors.

In addition, there are many definitions of the economic reform programme and transition. According to the Oxford American English Dictionary and Thesaurus, 2006, the meaning of reform is to "make changes" in something (typically a social, political, or economic institution or practice) in order to improve it, such as an opportunity to reform and restructure an antiquated schooling model. The Organisation for Economic Co-operation and Development (OECD, 1996) defined the economic reform programme as "an attempt to eliminate a number of market rigidities". The Oxford English Dictionary (2006) defines transition thus: "the process or a period of changing from one state or condition to another". Finally, Fanelli (2007) defined market reform as the implementation of changes in the rules of the economic game by a legal authority to widen the role of markets and private sector ownership, to deliver a more open-door policy to the entire economy.

3.3 THE IMPORTANCE, AIMS AND OBJECTIVES OF ECONOMIC REFORM PROGRAMMES

The measures of economic reform have been developed by the Freedom House Country Rating, the Heritage Foundation and the Fraser Institute. Each of these organisations independently reaches the same conclusion, as do several academic studies using different proxies for institutional conditions or the policy regime (De Haan and Sturm, 2000). In the late 1980s social revolutions in Central and Eastern Europe (CEE) initiated a new political and economic situation and a cultural process of restructuring without precedent in history. It became evident that nations which had based their development on centrally planned directives and inward-looking programmes were not only failing to provide their citizens with rising living standards but, in some cases, were actually retrogressing (Schipke and Taylor, 1994). In many developing countries this was the decade of debt crisis and high inflation. Rich nations and those with natural resources such as Libya, as one example of a developing country, found themselves, in effect, unable to earn sufficient foreign exchange to satisfy their people's needs in particular, and the economic system in general.

Bgcolor (1996) reported that the majority of developing nations were attempting to transform their societies by adopting a new model of market-oriented mechanisms, which attempted to compensate for the lost period of transition. This occurred especially among the countries of East and South Asia, Central and Eastern Europe and the former Soviet Union, and the Latin American nations, which undertook broad economic reform programmes. In the same way, these countries transformed their economies and societies from ones dominated by the public sector to ones where they could burgeon in response to market signals. Discussing the East European economies, Bird (1992) found various economic problems, such as shortages and general weaknesses in the supply side of economic growth, soft budget limitations (at the level of both individual enterprises and the economy as a whole), repressed inflation (which resulted in other manifestations of excess aggregate demand), hidden unemployment (which was reflected in low labour productivity) and a substantial level of unproductivity. These variables led such countries to adopt programmes of economic reform in order to treat said deficiencies in their economic activities.

From another perspective Gomulka (1994) observed that, to achieve its objectives, any economic reform programme package usually consists of the following measures: consideration of government budget; price liberalisation and launching privatisation; control

monetary policy; convertibility of domestic currency; foreign trade liberalisation; establishing a market environment laws that support privatisation; dismantling of former state monopolies. While studying the economic reform programme in the case of the transition economy of Czech Republic, Colijn (1991), Grosfied and Hare (1991) indicated certain achievable steps in an economic reform programme:

- Introduction of deregulation so that the private sector can play its role;
- Introduction of laws to achieve a budget surplus via a decrease or, at least, a restraint on government expenditure; and
- Introduction of laws so that the exchange rate of the currency can be controlled; this can be accompanied by an increase in level of exports and a decrease in level of imports.

3.4 ASSESSING THE ECONOMIC REFORM PROGRAMME VARIABLES AND MACRO-ECONOMIC REFORM VARIABLES

What is always fundamental in any economic reform programme is its success in improving the performance of economy activity. In order to evaluate the performance of an economic reform programme it is crucial to establish basic variables that can reveal performance. According to many authors, such as Charemza (1991), Hare and Hughes (1991), Hrneir and Klacek (1991) and Landesmann (1991), economic reform variables can be measured mostly via the following factors:

- Privatisation, together with new measures in fiscal and monetary policy and income policy;
- Price liberalisation;
- Exchange and trade liberalisation.

Lavigne (1999), though, stated that more specific variables should be taken into consideration in measuring macro-economic performance. These include: inflation rate; interest rate; government budget; current exchange rate; rate of growth. Williamson (2004) considered ten reforms of the "*Washington Consensus*" concerned with economic institutions' policy: fiscal discipline, public expenditure priorities, tax reform, financial liberalising, exchange rates, trade liberalisation, foreign direct investment, privatisation of state owned enterprises, deregulation and property rights. Clearly, these 10 points should be relevant to the economic policy situation of the transition countries, which included other points reported by the World Bank and IMF (IMF, 1996) as caution against corruption in the public sector and money laundering in the financial sector, which will take fiscal and monetary policy very seriously. Nuti (1999), however, indicated that the Washington Consensus permitted the following explanation of the transformation of former socialist economy: transition = liberalisation + privatisation (State Owned Enterprises) + open international trade in order to have an efficient market economy.

3.5 THE PROCESS OF ECONOMIC REFORM

Guitian (1982) presented a set of standards which can be used to measure or evaluate the performance of an economic reform programme. These standards or principles can be classified into three categories, namely (1) positive or practical standards, (2) normative standards and (3) conjectural or judgmental standards. The first category measures the performance of the countries' economies under an economic reform programme by comparing their results to the situation that prevailed prior to introduction of this programme. Normative standards measure the performance of the countries' economies under an economic reform programme by comparing the actual results achieved by the economic reform programme period with the targets which were already specified when the economic reform programme was introduced. Conjectural or judgmental standards focus upon what has been done and what should be done. In fact, the third standard aims to compare the actual performance of adopting an economic reform programme with the absence of the economic reform.

In reality, these standards provide the opportunity to understand and to plan so that those implementing economic reforms can improve the results of any programme. Obviously, the second standards are quite easy to understand as they measure output against objectives. Khan and Langmead (1990) considered the third group of standards to be the most controversial from many angles. This may be because the objective of most assessment is to compare what has happened with what would have happened in the absence of economic reform. Besides this, the first standard can be considered as the least contentious because it simply compares sets of realistic information concerning the economic reform programme. The Libyan economy in the early stage of reform, after the resolution of the Lockerbie question and the UN Security Council, permanently lifted its 11-year-old sanctions by 12 September 2003, and tried to re-establish the relationship with the US. This clearly suggests that positive standards should be applied by this research. The crucial point of consideration is why we apply standards in economic evaluations. The reasons are to improve the economy from both sides. It is understood that, to study the economic reform programme, there are many common

parameters through which objectives of the Libyan economy were achieved, the first standard (positive standard), compared with the other two standards, can give a more straightforward assessment, especially since this research aims to state the actual performance in the economic situation under the economic reform compared with its situation prior to adopting this programme (for more explanation see Chapters Seven and Eight).

3.6 THE ECONOMIC REFORM PROGRAMME EXPERIMENTS EXPERIMENT

Since the late 1970s transition from socialist planned to market economies began in Asia and in the late 1980s, in Europe, its success or failure had the most significant impact upon the population of these economies and the world at large as a general event in the 20th century (after the Great Depression and until at least the beginning of the 1980s) (Levigne, 1999). The process and outcome of the transition economy will assist the scope of the 21st century global system. Transformation of centrally-planned economies is the introduction of new financial policy, new ownership structures, new political systems (Knell and Rider, 1992). In their economies, starting from the post-communist countries with emerging markets in the CEECs and the former Soviet Union, China, India, Latin America, MENA nations within the space of a few years, and in which Libya is located, tend to be in a similar situation politically and economically.

3.6.1 Central and East European Countries Experiment

In the early 1990s the economic performance of the reform process in South East Asia and Latin America suggests that the post-communist countries may have difficulty in joining liberal democratic rule and fundamental economic restructuring (Staehr 2006). Lipton and Sachs (1990: 75-99) argued that: "[...] Eastern Europe should work hard to create the common core of market institutions found in all of western Europe: private ownership protected by a commercial law, a corporate structure for industry, an independent financial system, and so forth [...] structural reforms cannot work without a working price system; a working price system cannot be put into place without ending excess demand and creating a convertible currency [...] the macroeconomic shock must be accompanied by other measures, including selling off state assets, for bankruptcy, preparing a social safety net, and undertaking tax reform. Clearly, the reform process must be comprehensive."

Recently, Tridico (2006) reported that both Hungary and Slovenia implemented the gradual approach programme of macro-economic stabilisation, whilst Poland and the Czech Republic, under the shock therapy programme, achieved similar results. Other countries, though, such as Russia and Bulgaria, which also adopted a shock therapy programme, had very negative performances in their economic reform. Lipton and Sachs (1990) provided five steps related to the transition of market economy requiring implementation:

- 1. Macro-economic stabilisation,
- 2. Liberalisation of foreign trade and ensuring convertibility of national currency,
- 3. Companies' reform through privatisation,
- 4. Establishing a social security system,
- 5. Developing institutional and legal infrastructure of market economy and market-based financial system.

There are, according to King (2003), three reforms to be considered, price liberalisation, stabilisation and privatisation, as preconditions for a successful transition to a market economy. These reforms should all be completed at the same time or in the short-term and, provided that the negative shocks to the domestic economy, would succeed rapid liberalisation of price and foreign trade as the shock associated with the austerity of a "stabilisation" policy programme (fiscal and monetary austerity). King referred to rapid privatisation of large entry prices by a mass privatisation technique policy as "shock privatisation" which creates severe direct and indirect supply and demand. Few enterprises, for instance, Russia in 1994, had a mass privatisation programme, where investment was only 30 per cent of its 1990 level. Table 3.1 suggests that the depth of recession in transition countries is based on classification with respect to liberalisation strategy. Whilst the Czech Republic undertook rapid reforms and suffered from transitional depression, the national decline in GDP was 8 per cent, 17 per cent for Poland and 13 per cent for Slovakia. Other countries which followed such an approach experienced decline in growth rates following reforms in the generally positive outcome for those countries adopting the gradual approach. Considering growth performance was relatively large, with growth rates from 1.9 per cent in Hungary to 6.5 per cent in Estonia, and 4.1 per cent for Slovenia, growth performance following recession stood at 2.4 per cent and 3.9 per cent in Bulgaria (see Table 3.1).

	L	iberalisation	Three-year Average Growth Rates of GDP						
Countries	Reform	rm Voors of	Prior to	Following Start	Recovery S	stage			
Countries	Strategy	Recession	Reforms 1986-1988	of Reforms 1990-1992	Turning point	Growth			
Pulgorio	Fact	1080 1002	12	0.4	1994	2.4			
Dulgaria	rasi	1989-1995	4.5	-9.4	1998	3.9			
Czech Rep.	Fast	1990-1992	1.6	-4.4	1994	4.1			
Estonia	Fast	1991-1995	N/A	-12.0	1996	6.5			
Hungary	Gradual	1990-1993	1.8	-6.2	1994	1.9			
Latvia	Fast	1992-1994/1996	N/A	-16.1	1997	5.6			
Lithuania	Gradual	1990-1995	N/A	-10.1	1996	5.0			
Poland	Fast	1990-1991	3.4	-5.3	1992	3.9			
D '	Г (1000 1000	0.0	0.1	1993	4.2			
Romania	Fast	1989-1992	0.9	-9.1	2000	4.3			
Slovakia	Fast	1990-1993	2.8	-7.9	1994	6.0			
Slovenia	Gradual	1989-1993	N/A	-5.1	1994	4.1			

Table 3.1 Growth Rates in CEECs Following Reforms

Source: Foster and Stehrer (2005).

Poland and Hungary undertook a gradual approach to reform their GDP level in 1996 and 2000, respectively, but Hungary (2000) and Slovenia (1998) took longer to return to prereform GDP levels than Poland (1996) and the Czech Republic (1997). Both countries followed the fast approach. Slovenia also followed faster reforms because it recovered more quickly than Slovakia (1999). Fisher et al. (1996a, 1996b, and 1998) concluded that the transition experience for both macroeconomic stabilisation and structural reforms contribute growth and that the more structural reform occurred the more rapidly an economy expanded.

3.6.2 China Experiment

From the early 1950s to the early 1970s the centrally planned economic system was based on the Soviet model as endogenous to development strategy and external environment. It has the following features described by Chai and Roy, 2006: collective farms in the agricultural sectors; state-owned enterprises; central planning; control economy by government. In early 1978 China's economy was a closed system with very limited trade and no financial interaction with the rest of the world (Zhao, 2006). At the end of 1978 and early in 1979 China adopted reform and an open door policy with an alternative gradual, evolutionary approach to the transition reform to the rest of the world. The success of China in its approach to the process of transition produced many challenges from a closed to an open economy to conventional knowledge in economic theory (Chow, 1997; Qian and Wu, 2000). Grose and Steinherr (1995) argue that the Chinese economy has been successful in a gradualism situation in introducing economic reforms compared with Eastern Europe. They summarised the points. Firstly, China had no political "big bang"; it was under control, whereas Eastern Europe has experienced a political "big bang" and people expect freedom of choice of

employment or consumption incompatible with gradual reform. Secondly, the key sector of China is highly labour-intensive and the small size of the basic unit permits quick liberalisation that has employed more than half the population. This means that China has experienced a greater "big bang" effect than Eastern European farming, which has been much less important for the whole economy and much more capital-intensive. Thirdly, China decentralised decision-making to provincial and communal authorities which embarked on developing local industry on a competitive basis. The increasing share of non-state enterprise is not the result of privatisation of existing state-owned enterprises (SOEs), but more the result of rapid growth of non-state enterprises. Finally, China has had successful gradualism of "open economy zones". China's introduction of economic reform has grown faster than during the pre-reform period, as Table 3.2 demonstrates. China's real average GDP has been estimated as growing by 9.6 *per cent* during 1979-2007.

Time period	Average Annual Growth (<i>per cent</i>)	Time period	Average Annual Growth (<i>per cent</i>)
1960-1978 (pre-reform)*	5.3	1998	7.8
1979-2007 (post-reform)**	9.6	1999	7.6
1990	3.8	2000	8.4
1991	9.3	2001	8.3
1992	14.2	2002	9.1
1993	14.0	2003	10.0
1994	13.1	2004	10.1
1995	10.9	2005	9.8
1996	10.0	2006	9.5
1997	9.3	2007	9.0

Table 3.2 China's Average Real Annual GDP Growth Rates, 1960-2007

Source: *Data from 1960 to 1978, Das (2006); **Data from 1979 to 2007, IMF (2007) World Bank Outlook Database.

3.6.3 India Experiment

India, after over sixty years of independence and economic planning, is the second largest country in terms of population, the seventh largest in terms of geographical area and is one of the world's poorest countries with per-capita income of 350 US\$ in 1998 (World Bank, 2000). Since 1991 India has adopted a public sector under centralised investment planning with flexible controls on private investment, which compelled the restructuring of manufacturing firms. Since the 1990s India's economy has been one of the fastest growing worldwide (see Table 3.3).

Variables	1950-1980	1980-1990	1990-2000
Annual Real GDP Growth	3.7	5.9	6.2
Annual Real GDP Growth Per-capita growth	1.5	3.8	4.4

Table 3.3 Indian Rates of Economic Growth, 1950-2000

Source: Delong (2001).

(In nor cont)

Delong (2001: 5-6) wrote, in relation to Indian economic growth: "What are the sources of India's recent acceleration in economic growth? Conventional wisdom traces them to policy reform at the start of the 1990s. Yet the aggregate growth data tells us that the acceleration of economic growth began earlier, in the early or mid-1980s, long before the exchange crisis of 1991 and the shift of the government of Narasimha Rao and Manmohan Singh toward neoliberal economic reforms. Thus apparently the policy changes in the mid-and late-1980s under the last governments of the Nehru dynasty were sufficient to start the acceleration of growth, small as those policy reforms appear in retrospect." Srinivasan (2003), Chai and Roy (2006), and Girdner and Siddiqui (2008) argued that the structure of the economic reform programme was aimed at addressing one of the root causes of India's economic growth problems, i.e. government over regulation.

3.6.4 Latin American Experiment

Following Williamson (2004), as mentioned in section 3.4, economic reform in most of Latin America occurred within the past two decades. Reforms in the first generation included labour markets that are more flexible, free trade liberalisation and deepening privatisation. Similarly, the second-generation included regulatory, judicial social security, education, tax and political reforms which established the institutional conclusion that these could be sustained in their economic growth. Holden and Rajapatirana (1995) presented economic reform in Latin America via two types of generation: a first generation of economic variables and a second generation of reforms at macro-economic level, which are required to ensure that first generation changes are sustainable.

First Generation Reforms and Outcomes

Since the early 1970s Latin America began its economic reform with the initial case of Chile and was implemented from the mid-1980s for the majority of Latin American countries. Chile, for instance, has the longest period of reform having been able to complete its first and second-generation reform agenda (Inter-American Development Bank, 1996). Table 3.4 summarises economic reform in Latin American states. Mexico, for instance, has had foreign debt crises and huge balance-of-payment deficits. Inflation has been brought under control by Argentina, Bolivia, Brazil and Peru where the foreign crisis caused hyperinflation and current account deficits became unsustainable (see Table 3.4).

Country	Starting Economic Reform	Recovery ^a	Macro-economic Situation	International Context
Argentina	1978 (beginning) 1981-1989 (suspension) 1991 (resumption)	1991-1993	Economic instability High inflation Stabilisation programme (previous hyperinflation : 4,924 <i>per</i> <i>cent</i> in 1989)	Financial liquidity Financial liquidity Slight improvement in terms of trade
Bolivia	1985	1988-1990	Stabilisation programme (previous hyperinflation : 11,750 per cent in 1985)	Financial liquidity Sharp decline in terms of trade
Brazil	1990 (beginning) 1994 (new phase)	1994-1995	Successive failure of stabilisation programmes Real Plan: price stabilisation (1994)	Financial liquidity Decline in terms of trade
Chile	1974 (beginning) 1982-1985 (crisis in model) 1986 (new reform phase)	1976-1980	Stabilisation programme (previous hyperinflation: 559 <i>per cent</i> in 1973) Adjustment and stabilisation programme Stability	Financial liquidity Decline in terms of trade Financial liquidity Improved terms of trade
Colombia	1990	1994-1995	Stability Reform preceded by sharp devaluations	Financial liquidity Decline in terms of trade
Mexico	1986	1991-1993	High inflation Stabilisation program	Financial liquidity Sharp decline in terms of trade
Peru	1990	1992-1994	Stabilisation programme (previous hyperinflation: 7,482 per cent in 1990)	Financial liquidity Decline in terms of trade

Table 3.4 The Economic Reform in Latin America

Notice a: Inter-American Development Bank (1996).

Source: Moguillansky and Bielschowsky (2001).

Second Generation and Reform Strategy

The key aim of the second-generation agenda is to define and re-evaluate the structural role of the state in economic policy relating to legal systems, property rights and financial institutions, (Holden and Rajapatirn, 1997). Holden and Rajapatirn (1997) re-defined the role of the state in second-generation agenda as follows:

- Ensure that regulation leads to competitive solutions;
- Let public goods undertake activities freely; and
- From those activities, there is no comparative advantage.

According to the Inter-American Development Bank, perception of economic reform in Latin American countries is too pessimistic. Deregulation, free trade and consistent attack on inflation have contributed long-term to increase economic growth and investment. Camdessus (1997) concluded that Latin American economies have witnessed significant increases in growth rates. Average growth in Latin America increased from less than 1.5 *per cent* in 1988-1989 to 5 *per cent* in 1994. Additionally, the rate of inflation declined from 567.6 *per cent* in 1990 to 5.8 *per cent* in 2006 (see IMF, 2007); private capital inflows rose from close to zero in 1989 to 70 billion US\$ over the same period.
3.6.5 Middle Eastern and North African Countries Experiment

From the early 1970s up to the late 1980s the Arab state system had a high degree of dependency on external capital inflows which comprised a significant portion of central government revenues. This inflow consisted of international economic and political rents (Pawelka and Boeckh, 2004). In the early 1980s world oil prices decreased and failed to recover until the end of the 1990s, except for a short period during the second Gulf War. Inflows increased to more than 20 per cent, for instance in Algeria, Egypt, and Jordan, whilst, in the United Arab Emirates (UAE), inflation was under control, running at 3 per cent in 1991 (Pawelka and Boeckh, 2004). The majority of Arab countries had implemented liberal reform in economic conditions as suggested by the International Financial Organisation. The Middle East developed its own region of Gulf States, i.e. Bahrain, Oman, Qatar, Saudi Arabia, Kuwait and the UAE. These countries made higher levels of economic progress due to oil revenues, but economic and political reform was required for horizontal integration to occur (Pawelka and Boeckh, 2004). Philippe (2007) indicates: "As is the case with Central and Eastern European countries, these actions would greatly benefit the Mediterranean countries. While the European framework is neither perfect nor fully flexible, legal issues that affect intellectual property, the right to compete, and other rules are indispensable complements to liberalisation that would both increase communication between economies and long-term security. This would encourage investment and thus growth and employment."

Figure 3.1 displays the most significant events and real world oil prices situation during the 1970-2006 period. According to the IMF, during mid-2006 Libya's macro-economic indicators remained strong which might indicate the impact of the main steps of economic reform due to the high international price for Libya's crude oil exports. From among this oil price increases per barrel, the average amount was 38 US\$ for 2004, 50.64 US\$ for 2005, 63.05 US\$ for 2006 (CBL, 2005 and OPEC September, 2006). Subsequently, it increased to 93.33 US\$ for 2007 and 126.74 US\$ during early 2008 (EIA, 2008). On 27 March 2008 oil prices reduced to 141.71 US\$ per barrel amid Libya's threat to cut output and the prediction by OPEC's president that prices could reach 170 US\$ by the summer (Schmollinger, 2008). Libyan economic growth rates during the early and late 1990s were either negative or considerably lower. The negative and lower growth rates during that period, however, reflect an era of political instability and poor adjustment to the reversal of boom conditions of higher oil prices. Regarding data from the IMF (2003, 2008), in 2003 there was rapid GDP growth during the post-sanction period at 9.1 *per cent* and non-oil sector growth by 2.2 *per cent* due to

increasing oil prices. The fiscal stance continued to be expansionary, with a non-oil fiscal deficit widening to 36 *per cent* of GDP.



In 2004 real GDP grew by 4.5 per cent, while, in 2005, macro-economic reform remained relatively strong. There was real GDP growth of 35 per cent, inflation remained low (2.5 per cent) and agriculture had remained modest at 2.5 per cent, whilst the manufacturing sector grew positively by 1.8 per cent for the first time for 5 years. In external terms, the current account surplus widened to 41 per cent of GDP with increased hydrocarbon exports wroth about 29 billion US\$, at the end 48 per cent. Imports grew about 11 billion US\$ of a 24 per cent, boosted by increased demand. Furthermore, gross international reserves rose to about 32 months of projected 2006 imports (IMF, 2006). Appendix A, Table 3.14 illustrates crude oil production in MENA nations including Libya during the 1980-2005 period. The recent economic reform programme in Libya, which involved the financial sector in several ways, has been essential for growth of the private sector and the elimination of illegal black market transactions previously practiced before 2002. Thus, deregulation directed national and foreign investment of production in Libya, to broaden its capacity and expand the role of the private sector, especially with the issuance of law 134 of 2006 to establish the stock market following the privatisation process and the increase in national economic growth rates (see Chapter Two).

3.7 COMPARISON BETWEEN EXPERIMENT COUNTRIES

Many nations have adopted economic reform, such as Africa, Middle East, Central and Eastern Europe, South and East Asia and Latin America. For instance, Central and Eastern European countries, as mentioned earlier in this chapter, adopted this programme reflecting

CHAPTER 3: THE ECONOMIC REFORM PROGRAMME

economic change and political policy. Moreover, most Asian countries achieved favourable results when compared with other countries, which may have already had the benefit of human resources in their economy. For instance, Korea had a literacy rate of 30 per cent in the mid-1950s, which had increased to over 95 per cent by the early 1990s (Camedssus, 1997a). The Appendix A, Tables and Figures summarise real GDP rate, inflation rate, central government fiscal balance and the difference between savings and investment in various countries internationally during the 1985-2006 period. As these tables and figures illustrate, every country has achieved distinct progress in their economy after adopting the economic reform process, but there are varying degrees of economic performance from one country to another. It is evident that South and East Asia achieved a significant economic performance when compared with other countries, as Table A 3.15 and Figure A 3.1 show. In fact, there was a strong rate of growth over this period and a lower rate of inflation, as presented in Table A 3.16 and Figure A 3.2. Furthermore, there was a small deficit in its central government fiscal balance (see Table A 3.17 and Figure A 3.3) and it had a higher score of variance between savings and investment compared to other countries, as Table A 3.18 and Figure A 3.4 demonstrate. Sub-Saharan countries additionally achieved sound performance when faced with major problems in their economies because the ratio of external debt to GDP growth rate grew from 111.7 in 1994 to 101.1 in 1996 (World Bank Outlook, 1995). Nonetheless, Central and Eastern European countries can be considered as the worst performing in the world as they achieved very poor economic indicator performance during the 1985-2006 period when compared with other nations. Indeed, the most crucial points in the various countries internationally can be due to variance between savings and investment. For instance, the highest percentage for difference between savings and investment as a percentage of GDP in South and East Asia was 35.6 per cent in 2006 compared to 22 per cent for Africa, 19.7 per cent for Sub-Saharan Africa, 21.9 per cent for Middle East, 24.5 per cent for Central Eastern Euro and 21 per cent for Latin American states (see Table A 3.18 and Figure A 3.4). At the same time, exports from these countries increased gradually throughout the last decade (see World Economic Outlook, 2007).

As observed, research in respect of the transitional economies has been mainly conducted in Eastern Europe and Asia, where economic reform programmes began earlier than in Libya. The first practical action of the Libyan state towards economic reform started in 1999 as the sanctions pushed the economy towards collapse. The reforms continued with deregulation, privatisation and liberalisation with another landmark being the establishment of the stock

CHAPTER 3: THE ECONOMIC REFORM PROGRAMME

market during 2006. Consequently, given Libya's late development in this area, the country's policy-makers should have been able to examine similar experiments in other developing countries, with a view to, for instance, assessing their strengths and weaknesses. Specifically, the economic reform programmes in the MENA nations such as Algeria, Egypt, Bahrain, Jordan, Oman, Qatar, Saudi Arabia, Kuwait and the UAE are of note, since many lessons can be learnt from the experiments of these other developing countries, taking into consideration certain similarities in the environment and characteristics of their various economics. Tardos (1988: 69) emphaised that lessons can be learned from the experiences of Hungary and Yugoslavia"as well as from monetarist and neo- and post- Keynesian economists and practical experts, who have studied experiences of collected experts, who have studied experiences of collected in market economies over the past decades."

On 1 January 1987 Hungary, introduced banking reform which established commercial banking operations. However, within a year of this reformation, Hungary began to experience the same frustration with monetary policy regulations that plagued Yugoslav countries for endogenously growing money supply and the inability of central monetary authorities to control spending and inflation (Gedeon, 1986). Despite these observations, there are some differences when compared with the Libyan experience.

- Libya has experienced huge reform, moving from a centrally-controlled to a market-based system.
- Libya has the distinctive characteristic of having been reliant on the oil industry for its economy. Oil is the major source of income and almost all foreign currency. Oil revenues comprise over 95 *per cent* of Libya's hard currency earnings (75 *per cent* of government receipts and 30 *per cent* of the gross domestic product).
- Libya investment capital is available in abundance.
- Libya's economic reform movement has not been inspired by the desire to join the European Union, as is the case with Eastern European countries that have had to formulate and implement initial economic reforms as a pre-condition of joining.

3.8 SUMMARY AND CONCLUSION

This chapter has provided an economic review of the literature of economic reform programmes on macro-economic performance. Key points of this chapter are summarised as follows. There are diverse expressions of economic reform programme such as transition,

CHAPTER 3: THE ECONOMIC REFORM PROGRAMME

transformation, economic adjustment, stabilisation and regime. In this research, the expressions "economic reform programme" and "transition" will be used.

- Economic reform programmes and macro-economic reform generally seek combativeness of the economy system, satisfaction of demands, solvency of the state and welfare of citizens. They may differ in the particular mix of these goals and in the instruments they use. Inflation, budgetary and current-account deficits, unemployment and loss of competitiveness are serious challenges for new democratic government policy.
- 2. The variables of reform are assessed in terms of the economic reform programme and macro-economic reform performance, such as deregulation, corporatisation, privatisation and liberalisation, budget deficit, the exchange rate, rate of GDP growth, per-capita income, real interest rates and inflation rate.
- 3. In the last twenty-five years Latin America has undergone a period of restructuring market reforms. Chile is the only case where a Latin American reform programme can certainly be classified as a reform success. The other states reveal that reforms have only partially realised the desired target (Brazil, Mexico, Peru and Uruguay) or resulted in absolute failure (Argentina, Paraguay and Venezuela).
- 4. Kostadinova (2004) believes that geography plays an indirect role in terms of levels of economic reform by affecting the outcomes of elections. For instance, countries physically closer to Western Europe were more receptive to it in terms of political and economic structures than countries more distant. Libya, in the centre of North Africa, is very close to the European countries which may be useful in respect of obtaining positive experience from these nations, both politically and economically.
- 5. The economic transition experience in other countries has provided reform policy close to the Libyan situation. For instance, the nature and objectives of the former Soviet economies and their economic situations are quite dissimilar to those of the Libyan economy. Libya, however, may learn from their errors.

This chapter has argued that the economic reform programme plays an important role in stock market performance. It has explored the nature of the reforms and challenges associated with those economies that have embarked upon such reform. The next chapter will provide a review of theoretical and empirical perspectives on the effect of both stock market development and economic reform on economic growth.

CHAPTER FOUR ECONOMIC GROWTH THEORY

4.1 INTRODUCTION

The preceding chapter reviewed the literature concerning economic reform, within the framework of moving from a closed or socialist economy to an open or capitalist economy. Now, in this chapter, a simple framework is developed having considered the theoretical and empirical literature on economic growth and financial development, with reference to the endogenous growth theory. The chapter begins by considering how the traditional growth theories failed to recognise the importance of financial factors in the growth process, and it proceeds to demonstrate how the emergence of the endogenous growth literature changed the profession's views on factors affecting growth by allowing technical change to be modelled endogenously. Within this recent approach, previous theoretical and empirical studies have attempted to identify the relationship between the development of financial markets and economic growth.

Growth and development theorists at both the micro and macro-economic level are concerned with the level of national output and its growth over time. They study problems of economic stagnation, unemployment, inflation, interest rates, economic growth, wages, the exchange rate, the stock market and cyclical instability, also the fiscal and monetary policies adopted by governments to deal with these problems. Economic conditions abroad, the price of oil and the balance of payments within other countries are also examined. They may also be concerned with the economic welfare of both poor and wealthy households affected by fluctuations in interest rates or inflation, which are called endogenous variables. The other endogenous variables are the objects of analysis in the economic and financial model, as Figure 4.1 shows. The relationships between endogenous and exogenous variables are linked, as exogenous variables are not strictly independent of the endogenous variables (Burda and Wyplosz, 2001). Within the emergence of the endogenous growth literature model, the focus would be shifted from relying on the unknown exogenous technological progress to explaining economic growth by different theoretical and empirical models where the engine of growth also includes human capital or knowledge that is accumulated during a period of time. However, periods of

high unemployment and stagnation occur from time to time throughout the world (such as in the 1930s in the US and the early 1980s and early 1990s in South East Asia).



Figure 4.1 Relationships between Endogenous and Exogenous Variables

Source: Developed for this research from Section 4.1: Introduction.

During the 1990s Libya experienced the same economic problems of stagnation and unemployment. For these reasons, theorists of the late 1950s and early 1960s interpreted the process of development as a series of successive types of economic growth, in which the right quantity and mixture of savings, investment and foreign aid were all that was essential to enable the development of nations that had historically been followed more by developed countries (Bourne, 2006). Through the 1980s and early 1990s there was a counterrevolution in economics concentrating on issues such as the beneficial role of free markets, open-door economies and the privatisation of inefficient public enterprises (Todaro and Smith, 2006).

This chapter is divided into four main sections. Section 4.2, discusses neoclassical growth theory and the use of production functions. This is followed by an explanation of the new growth theory, or endogenous growth theory in Section 4.3. In Section 4.4 the way in which the effects of the stock market on economic growth can be identified is discussed. In Section 4.5 a summary and the conclusion of this chapter is provided.

4.2 NEOCLASSICAL GROWTH THEORY

The history of the economic theory growth of output and the distribution of income between wages and profits were presented by Adam Smith in 1776 in the "*Wealth of Nations*". This most important contribution was to introduce the notion of increasing returns, based on the division of labour. His major contribution was to understand the fundamental forces which underlie the development of economic policy (Farmer, 1997). Modern growth theory started

with a classic article by British economists Harrod and Domar, "An Essay in Dynamic theory", now called the Harrod-Domar Growth Model. This model describes the economic mechanisms whereby more investment leads to more growth. Both Harrod (1939 and 1948) and Domar (1946) used the production function to support their claim that the capitalist system was inherently unstable. However, they explained how aggregate supply expands, which means that investment has two effects; one on the aggregate demand side such as when business spends more; the other on the aggregate supply side, whereby more investment increases capital stock and produces more business. However, in Harrod-Domar's growth model, which has come under attack from the new growth theory, investment does not matter for long-term growth. The assumptions and perspective of *neoclassical growth theory* will now consider how to understand the source of growth used in empirical models for developed and developing countries. *Neoclassical growth theory* was born as a result of Harrod-Domar's model and the new growth theory developed as the result of *neoclassical growth theory*.

During the early 1960s, neo-classical growth theory was practiced and people generally accepted its approach to modelling growth in the long-term, which has been driven by increasing returns: Solow (1956) and Swan (1956), Arrow (1962), Cass (1965), Koopmans (1965). This kind of framework assumed the neo-classical model production of consumption rising as a function of the stock of knowledge increasing within constant return to scale, which returns to each input (labour and capital) as well as smooth elasticity of the substation between the inputs¹¹. For instance, Arrow (1962), in his model of "learning-by-doing", argued that new machines are improved and more productivity will result as the function of the cumulative which will also increase investment for the industry because new knowledge should be discovered as the result of investment. However, Arrow's model meant that two problems could be encountered which would increase any rates of growth model of increasing returns: existing competitive equilibrium and the function of capital and labour which increase returns to scale. Smith (1776) pointed out the technological improvement in the form of "learning by doing" or "learning by using" with economies of scale through to the concept of division of labour in the process of the wealth of nations. Furthermore, according to some studies (Lucas, 1988; Stiglitz, 1987; Romer, 1986) it has been argued that the major difference between the

¹¹ A recent review by Cowton (2008: 1) stated that: "The neoclassical theory of the firm presents an impressive analysis in which the primacy of owners' interests is asserted. It offers powerful insights into the contracting and monitoring challenges that arise when owners, as principals, delegate control to executive managers, their agents. This perspective has had a major impact on debates regarding corporate governance and upon programs of corporate governance reform, especially in the wake of various corporate 'scandals'".

more and less developed countries increased by learning-by-doing. Thus, "*learning-by-doing*" increases the stock of knowledge and human capital, and other factors such as yield quality¹².

According to Mankiw et al. (1992), the predictions of the Solow model are approximately consistent with the evidence, and they found that savings and population growth affect income in the direction of the Solow model¹³. They explained why the approximate influences of savings and population growth appear too large, giving two reasons. First, the rates given to human capital accumulation, higher saving or lower population growth lead to higher levels of income and, thus, higher levels of human capital. In other words, the elasticity of income with respect to the stock of physical capital is no different from capita or from capitalist share income; this result is supported by Romer's suggestion that approximates its social return. Second, human-capital accumulation may be correlated with saving rates and population growth rates. This would simply imply that omitting human-capital accumulation biases the estimated coefficients of saving and population growth. Gwartney and Lawson (2003) argued that the Solow model, in physical and human capital, is the key of the economic growth. Mankiw et al. (1995) mentioned three problems relating to the neoclassical theory against Solow's model. Finally, according to the *neo-classical growth model*, as Figure 4.2 shows, the impact of an increase in savings rate leads to higher output, consumption and capital perworker in the long-term. Strategic policy should then have to make the savings rate as high as possible. Also, more output will not only raise living standards but also lead to more human capital formation and innovation, which will increase productivity and output, which will then increase human capital and innovation again.

¹² For more explanation see, Chang (1997).

¹³ Hansen and Prescott (2002: 1205) point out that "[...] there is no fixed factor in the Solow production function, population growth has less influence on the growth rate of per capita income and living standards begin to improve. In the limit, the economy behaves like as standard Solow growth model, which displays many of the secular of modern industrial economies". See also, Michael (1993).



Figure 4.2 The Living Standards for the Long-term Process

Source: Abel et al. (1998).

In conclusion, the neo-classical growth model is simply a general equilibrium model that emphasises the process which leads the economy to its a steady-state crucial role in ensuring convergence to such a steady-state but also implies that, in the steady state, the productivity of capital is zero, the capital labour ratio is fixed and that growth-generated endogenous factors (i.e., capital accumulation or capital transaction) are zero. Hence, in this model the steady-state rate could be positive if some exogenous force (usually technological) is acting upon the system the expression. Furthermore, this model assumes that the rate of technological change is given as exogenous, which does not provide a useful framework for understanding economic forces and policies, such as the financial factors effect. Therefore, this model of financial factors can explain only the equilibrium level of capital stock per-work but, however, not the rate of economic growth. Following the work of Romer (1986, 1990) and Lucas (1988), the endogenous growth model in the next section gives more detail on these channels.

4.3 NEW ENDOGENOUS GROWTH THEORY AND THE MACRO-DETERMINANTS OF GROWTH

Since the mid-1980s there has been a new wave of literature and research on the applied economic of growth. This has led to the development of explanations of the differences in the rates of output growth and per-capita income growth in the long-term across the world, according to the so-called new growth theory. For instance, the richest 5 *per cent* of the world's nations averaged a per-capita income over the period 1960-1985 that was about

twenty-nine times the corresponding figure for the poorest 5 *per cent*. In 1985, the richest state in the US was Connecticut, the poorest was Mississippi and the ratio of per-capita incomes worked out at approximately 2 (Ray, 1998). The per-capita income of the aforementioned eight East Asian economies over the period 1965-1990, excluding China, increased at an annual rate of 5.5 *per cent*. For the entire data set between 1980 and 1993, China's per-capita income grew at an annual rate of 8.2 *per cent* which is truly phenomenal (Ray, 1998). Furthermore, over the period 1960-1985, per-capita growth averaged 1.9 *per cent* per year of 102 nations studied by Parente and Prescoott (1993) and, in wealthy nations, the productive potential of the economy has been increased by capital accumulation, the opening up of new territories and increased supplies of better quality labour. In addition, so much more has been learnt about how to optimise output from resources through increasing efficiency for each other (Apart and Donaldson, 1971).

The birth of the endogenous growth theory has enabled the development of tractable growth models where the long-term rate can be affected by elements such as technology, government policies and institutional arrangements. This has rekindled interest in the role of financial development in economic growth. The new models of endogenous growth theory began with the works of Romer (1986, 1990), Lucas (1988) and Rebelo (1991), who developed models of non-decreasing returns to a broad class of capital goods, including human capital. The difference between Rebelo's model and Solow's is simply in the specification of the production function, where output and capital in each period (t) is expressed by the following equation:

$$\mathbf{Y}_t = \mathbf{A} \mathbf{K}_t \tag{4.1}$$

From this equation there is no exogenous technological change. Therefore, this type of framework, implied by equation 4.1, has been built upon by Arrow (1962), who developed models characterised by non-decreasing returns. Lucas (1988) used the "*multi-good*" model, which was adapted from Krugman (1987). He commenced his model with an aggregate production function in the following equation:

$$\boldsymbol{Y}_{t} = \boldsymbol{A}_{t} F(\boldsymbol{K}_{t}, \boldsymbol{H}_{t}) \tag{4.2}$$

where Y, K and H are output, physical capital and human capital with different types of investment at t and where the parameter A represents the level of technology. However, in

general, school enrolment rates are used as simple proxy measures for human capital. One of the main contributions was presented by Romer, who published a series of papers on this area in the mid-1980s.

- 1. In the endogenous growth model, the assumptions of constant returns to scale and diminishing returns of individual functions are abandoned, which means that the economy grows without bounds and no convergence around economies is predicted.
- Technological change becomes endogenous. This implies that a change of capital input has both direct effects on output and indirect effects on technology change. From this it could be seen that the endogenous growth model may better reflect the reality of economic policy and practices of the transmission of advanced technology and new knowledge.

Romer (1986: 1003) assumed that "the creation of new knowledge by one firm is assumed to have a positive external effect on the production possibilities of other firms, because knowledge cannot be perfectly patented or kept secret". His model stressed integration within a competitive framework by rediscovering the simple Y = AK, dependent on production of knowledge and function of physical capital. The knowledge stock (*A*) is a public good, which, like (*A*) in Solow's model, introduces positive spill-over into the competitive framework for increasing returns to scale to the production function. This model, therefore, treats "learning-by-doing" as "learning-by-investment". Consider the following equation, where production function depends only on the capital stock:

$$\mathbf{Y}_{t} = f(\mathbf{K}_{t}) \tag{4.3}$$

where \mathbf{Y}_t and \mathbf{K}_t denote the output and stock of capital at time *t*, respectively. However the following equation is different from equation 4.4.

$$G_{y} = \frac{dk_{t}}{y_{t}} f(k_{t}) = s_{t} \varphi_{t}$$
(4.4)

where G_t is the growth rate of output, (s) is the savings rate and φ is the marginal productivity of capital. In this case, if decreasing φ , output growth will go to Zero as capital stock and K grows over time. Keller (1998: 1470) has discussed the general class of models developed by Grossman and Helpman (1991) and argued that cross-country R&D spill-overs

are important sources of productivity growth, pointing out that productivity spill-over exists if *"the importing country pays less than the intermediate good's full marginal product"*¹⁴.

Barro (1991) found human capital to be a significant contribution to growth rate but, in his regression analysis model, he left unexplained a good deal of the weak performance, for example, of Sub-Saharan African and Latin American countries. Romer (1990), however, extended his model to include a framework of competition to increasing returns of scale, through to fixed cost financial elements in the intermediate goods sector, in which the treatment of knowledge stock is usually similar to physical capital and assumed to be dependent on cumulative Research and Development (R&D) activities. Thus, if the output of capital is doubled, the amount of output is doubled too, as follows:

$$\dot{\mathbf{K}} = s\mathbf{Y} - \delta\mathbf{K} \tag{4.5}$$

This equation, together with the Y = AK production function gives:

$$\dot{\mathbf{Y}}/\mathbf{Y} = \dot{\mathbf{K}}/\mathbf{K} = s\mathbf{A} - \boldsymbol{\delta}$$
 (4.6)

So, as $sA > \delta$, income will grow forever, even without the assumption of exogenous technological improvement. From this equation it can be seen that savings lead to growth forever, but, in the neoclassical model, savings lead to the rate of growth temporarily. Levine (1997) examined two financial function factors of capital accumulation and technological innovation which could affect economic growth. Basically, in this situation, the first class of growth models used capital accumulation (capital externalities or capital goods produced) as discussed by Romer (1986), Lucas (1988) and Rebelo (1991), using constant returns to scale without using non-reproducible factors to generate steady-state per-capita growth models. The result of these models affects the steady-state growth by influencing the rate of capital formation. On the other hand, Romer (1990), Grossman and Helpman (1991), and Aghion and Howitt (1992) used second class growth models to focus on the invention of new production processes and goods. The results obtained from these models showed that the function performed by the financial system was affected by the steady-state growth when the rate of technological innovation altered.

¹⁴ "Spill-over" means that the aggregate of trading in the financial market is related to the liquidity of individual equities (Levine and Schmukler, 2006).

Empirical Framework in Variation Study

In the previous section, the literature review focused on the theoretical consideration "Theoretical Framework" of the endogenous economic growth model and the effects of stock market development on saving and investments. This section reviews the previous empirical studies that analyse the effects of financial development along with the stock market mechanism. In endogenous growth theory, the growth rate is a positive function of exogenous technical innovation with the best chances of successfully initiating new goods and production processes as the result of some empirical studies which have been conducted (see King and Levine, 1993b; Galetovic, 1996; Blackburn and Hung, 1998; Morales, 2003; Acemoglu et al., 2003). Recent endogenous growth models presented diverse approaches to explain the effect of financial markets on growth rate in the economic system situation. For instance, Bencivenga and Smith (1991) and Levine (1991) were the first to propose "endogenous growth" framework models to identify channels through which financial markets affect longterm economic growth. Their studies emphasised that financial markets will help to diversify an agent's liquidity risk and capital accumulation to attract more savings into productive investment and prevent the early removal of physical capital invested in long-term projects. Within the existence of financial markets there will be more capital intended to use rates of growth which could be maintained in productive investments. Hand in hand, endogenous growth theories illustrate that the rate of economic growth is related to financial development, technology and level of income. Greenwood and Jovanovic (1990) argued that income percapita will assist financial intermediaries to process the transmission channel between financial structure and economic growth to produce better information, enhance resource allocation within positive growth and improve investment decisions and the impact of economic growth rate¹⁵. They further mentioned that, as income level increases, financial structure becomes more extensive and economic growth becomes more rapid. Furthermore, income in equality will be across the board for rich and poor nations.

The group of authors who considered the economic growth theory, as Table 4.1 demonstrates, include Barro (1991), Levine and Renelt (1992), Mankiw et al. (1992), Barro and Lee (1993) and Levine and Zervor (1993). Barro (1991) investigated the new growth theory by using neoclassical growth models such as those of Cass (1956), Koopmans (1965) and Solow

¹⁵ See also Townsend (1979), Gale and Hellwig (1985), Diamond (1984), Diamond and Dybvig (1983), Gertler (1988) and Levine (1997).

(1956), measuring human capital to examine the growth of per-capita income from 1960 to 1985 in a cross-section of 98 countries using data from Summers and Heston (1988). Barro discovered that there is no significant relationship between the initial level of GDP and the per-capita growth rate; the correlation coefficient was 0.09. This finding accords with recent models, such as Lucas (1988) and Rebelo (1991), in which non-diminishing returns to capital are assumed, while the growth rate of per-capita GDP is positively related to initial human capital peroxiod by school enrolment rates. Countries with high ratios of human capital also seem to have lower fertility rates and a higher ratio of physical investment to GDP. Furthermore, growth rates are positively related to political stability and inversely related to a proxy for market distortions. Mankiw et al. (1992) examined whether the predictions of the Solow (1956) growth model are consistent with international variations in the standard of living, which used three samples of 98 non-oil countries in the period 1960-1985 in intermediate 76 developing countries and 22 OECD countries with populations of more than one million. The authors claimed that the data supported Solow's neoclassical model against the new endogenous growth models, which means that the assumption of non-diminishing returns to capital could predict the variations in initial per-capita income between countries.

Knight et al. (1993) tested the model by examining samples of 76 developing countries and 22 OECD countries, using panel data including the rate of technical progress, of trade policy and the stock of infrastructure investment *proxied by the flow variable*, and government-fixed investment as a proportion of GDP. The result of the growth is that output per-worker is positively related to the saving ratio and negatively to the growth of population and the initial level of GDP, while the human capital investment is significant and increases the productivity of physical investment. Barro and Lee (1993) tested data from 116 countries during the 1965-1985 period and found that five factors or variables explained 80 *per cent* of the differing growth rates between countries (see Table 4.1).

- 1. The initial level of real per-capita GDP measured by education and health has a negative effect.
- 2. The investment ratio has a positive effect.
- 3. The ratio of government consumption to GDP has a negative effect.
- 4. Market-distortions, as measured by the black market rate of foreign exchange, have a negative effect.

5. Political instability, as measured by the number of political revolutions per year, has a negative effect.

Factors	Barro (1991)	Mankiw et al. (1992)	Knight et al. (1993)	Barro and Lee (1993)	Levine and Renelt (1992)	Levine and Zervos (1993)
Case study	98 countries 1960-1985	98 countries 1960-1985	98 countries 1960-1985	116 countries 1965-1985	119 countries 1960-1989	98 countries 1960-1985
Dependent variable	Growth of per- capita income	Level of per- capita income	Growth of output per-worker	Growth of GDP	Growth of per - capita income	Growth of per - capita income
Convergence	Conditional	Conditional	Conditional	Conditional	Conditional	Conditional
Savings-investment ratio	Not considered	Significant positively	Significant positively	Significant positively	Significant positively	Not considered
Population growth	Not considered	Significant positively	Significant positively	Not considered	Not robust	Not considered
Education	Significant positively	Significant positively	Significant positively	Significant positively	Significant positively	Significant positively
Government consumption distortions	Significant negatively	Not considered	Not considered	Significant negatively	Not robust	Not considered
Political instability	Not considered	Not considered	Not considered	Significant negatively	Not robust	Significant negatively
Monetary and fiscal variables	Not considered	Not considered	Not considered	Not considered	Not robust	Weak
Trade variables	Not considered	Not considered	Significant positively	Not considered	Not robust	Weak
Inflation	Not considered	Not considered	Not considered	Not considered	Not robust	Not significant
0	TT1 1 11 (00)					

Table 4.1 The Macrodeterminants of Growth

Source: Thirlwall (2006).

The "*new*" endogenous growth theory, however, like many theories which consider various approaches to the marginal productivity of capital, does not converge to zero as capital grows over time. As already indicated, it is possible for real per-capita output to grow endogenously, even in the obstacles of exogenous productivity growth rate by altering the rate of human capital accumulation or technological development. Furthermore, investment in physical and human capital, respective of the endogenous growth theory, appropriate policies and options, assists private agents which could influence long-term steady growth. Therefore, in short, the overall policy regime of a country, including its taxes, financial structures, market and regulatory regimes, and levels of liberalisation and macro-economic distortions, could alter savings and investment allocation in various ways that would shape long-term growth rates. Solow (1956) argued that technological progress is the exogenous variables that most affects the growth rate in the long-term. He also argued that financial markets could only affect the equilibrium level of capital stock per-worker and not the rate of economic growth.

4.4 HOW STOCK MARKETS AFFECT ECONOMIC GROWTH

The endogenous growth literature and recent theoretical studies have tried to provide a link between the literature of endogenous growth theory and financial markets. Gurley and Shaw (1955) were the first to study the relationship between financial markets and rates of economic growth after Fisher (1933). They argued that the difference between developed and developing

countries is that the financial sector is accorded greater status in developed countries than is the case in developing countries. Gurley and Shaw found that financial markets contribute to economic development by enhancing physical capital accumulation. Their conclusions were supported by Friedman and Schwartz (1969). Although prior to the 1970s knowledge of the relationship between financial markets and real growth rates suffered due to the lack of evidential studies, the reason behind this may be that no model has been supplied to explain the mechanism by which the financial market affected economic growth. Moreover, limited data was available in this period. This remained the case until economists began to introduce models with which to analyse financial markets and economic growth behaviour. For instance, Goldsmith (1969), McKinnon (1973) and Shaw (1973) found that the development of financial markets has been significantly correlated with the rate of economic growth. Goldsmith (1969) argued that, as real income and wealth increase, then this, in turn, will lead to the size of financial markets growing as well¹⁶. Shaw (1973) and McKinnon (1973) were the first to study the importance of financial sector development, examining the effects of government intervention on the development of the financial sector to achieve a higher rate of economic growth. These studies suggest that financial intermediation has a positive effect on the steadystate growth (see Greenwood and Jovanovic, 1990; Bencivenga and Smith, 1991) and government intervention has a negative effect on the growth rate in the financial sector.

Bencivenga and Smith (1991) argued that financial markets promote growth to reduce capital liquidity risk, which leads to increasing funds available for productive capital¹⁷. Also, Greenwood and Jovanovic (1990) argued that financial intermediaries influence technological choice in their risk sharing and pooling services by facilitating high yield investment and, thereby, growth using the pooling of idiosyncratic investment risks across a large number of investments. Fulgieri and Rovelli (1998), Levine (1991), Levine and Zervos (1998a), Bencivengo et al. (1995) and others have argued that the greater liquidity of stock markets will provide shifts for long-term growth and higher return technologies process. Nevertheless, more recent research on the role of the stock market in the economy has argued from a different perspective. The emphasis is that having developed stock markets could enhance economic growth through the productivity of investment but not via the savings function.

¹⁶ Goldsmith (1969: 400) pointed out that: "[the financial sector] accelerates economic growth and improves economic performance to the extent that it facilitate s the migration of funds to the best user, i.e. to the place in the economic system where the funds will yield the highest social return".

¹⁷ Also Bencivenga and Smith (1991: 207) indicate that "differences in the extent of financial markets across countries seem to depend primarily on legislation and government regulation".

Levine (1992) argues that economic growth even influences the type of financial intermediation systems which the economy could afford. Therefore, when real per-capita income is low, the economy will select simple forms of financial intermediary whose main purpose will be to mobilise savings, diversify productivity risks and control liquidity risks. Furthermore, an increase in per-capita income enables the economy to develop more sophisticated financial intermediaries, whose financing will be correspondingly more costly as they will be involved in monitoring investment projects and the identification of the most costeffective innovations. Levine (1997) examined the two channels, capital accumulation and technology innovation, which affect economic growth. He also examined five financial functions that affect saving and allocation decisions and influence economic growth rates, as Figure 4.3 demonstrates.





Some authors tend to argue that stock markets do not raise much capital accumulation. They are also relatively insignificant in the development process. Atje and Jovanovic (1993)¹⁸ used a reduced form of equation driven by the argument of Greenwood and Jovanovic (1990) to determine empirically whether stock market development has a transitory level effect or permanent growth effects. They established a significant correlation between growths over the 1980-1988 period and the value of the stock market for 72 countries divided by GDP. In addition, they concluded that a stock market can raise a typical developing country's economic growth by around 2.5 per cent per annum. Atje and Jovanovic succeeded in obtaining a result for the hypothesis that development of the stock market has positive effects on growth but this

¹⁸ Atje and Jovanovic used the cross-section model to explain growth in per-capita output as follows:

 $G_i = \alpha_1 + \alpha_2 I_i + \alpha_3 S_i + \alpha_4 N_i$

where G is growth in per-capita output, I is investment as measured by output, S is the product of the level of stock market activity within investment, and N is growth in labour force. They assume that investment and stock market activity are both endogenous growth theory (Additionally, see Harris (1991)).

is, after all, in the short period between 1980-1988 which does seem to bear a significant impact upon long-term trends. Levine and Zervos (1998a) examined whether stock markets and banks promote economic growth by following the methodology of Atje and Jovanovic (1993) utilised for 47 countries during the 1976-1993 period and by enlarging the sample size and time coverage in diverse measures of stock market development. Thus, they recognised that stock market liquidity is strongly related to growth, capital accumulation and productivity, but stock market size does not seem to be correlated to economic growth. Consequently, the banking sector lending to the private sector has a robust impact upon economic growth. Harrison et al. (1999) presented a theoretical model of banking and growth, a feedback effect which is generated between financial development and economic growth rates. They argue that economic growth increases the average distance between borrowers to encourage regional specialisation and reduces the cost of intermediation. This, in turn, will increase investment and economic growth.

Empirical Framework

Many authors emphasise the links between the state of development of a country's financial sector and the level and rate of economic growth. Essentially, the argument is that the functions of the financial sector provide a conceptual framework of economic growth rates¹⁹. This type of empirical study started with Goldsmith (1969) and McKinnon (1973, 1989) and more recent studies have been conducted by Ghani (1992), King and Levine (1993a, b), Degregorio and Giudotti (1995) Rousseau and Wachtel (1998), Beck et al. (1999a) Levine et al. (2000), Levine (2000, 2004), Deidda (2006) and others. Goldsmith (1969) used data from 35 developed and less developed countries during the 1860-1963 period to examine the value of financial intermediary assets as a share of economic output. He found that the size of the financial intermediary sector and growth is positively correlated with the quality of financial function measured by financial sector. King and Levine (1993) and Levine and Zervos (1995) concluded that stock markets offer different services from those provided by banks. They used data from 49 countries over the period 1976-1993 to investigate the development of stock markets and rates of economic growth. They discovered that stock market development is strongly correlated with growth rates of real GDP per-capita and real physical capital per-

¹⁹ Barro and Sala-I-Martin (1995: 443) point out that "it is unclear whether the relation between growth and financial sophistication isolates the effect of an exogenous improvement in financial system on the growth rate, or, inversely reflects the impact of good growth prospects on the incentive to develop the financial sector."

capita. In addition, they discovered that both stock market liquidity and banking development predicted future economic growth rates when they were entered into the growth regression. However, they did not elucidate upon the precise nature of this difference.

Singh (1992a, b; 1996 and 1999) and Singh and Weisse (1998) argued that stock market development in developing countries during the 1980s and 1990s is unlikely to have resulted in achieving quicker industrialisation and faster long-term growth rates and has also not led to an increase in aggregate savings as a result of greater new-issue activities on the stock markets in the countries studied (Turkey and Mexico) where aggregate savings fell during the 1980s. The World Institute for the Development of Economic Research (WIDER) stated in their annual report (1990: 6) that: "*The need to affect foreign capital in non-debt creating forms is only one reason, and not the most important reason, why developing countries should wish to foster their emerging equity markets. Equity markets are a vital part of economic development; they encourage savings, help channel savings into productive investment and encourage entrepreneurs to improve the efficiency of investment."*

Liu (2002) examined the direction of causality between financial development and economic growth, using Geweke's (1992) study as his model to test pooled data for 109 developing and industrialised countries during the 1960-1994 $period^{20}$. He recognised five interesting points. First, financial development in general led to economic growth in 109 developing and industrialised countries. He proposed that financial deepening in many countries has yielded economic structure. Second, there is evidence of bi-directional causality when the sample is split into developing and industrialised nations. This illustrates that financial depth stimulates economic growth and, at the same time, economic growth propels financial development. Third, financial deepening contributes more to the causal relationship in developing countries than in industrial countries, thus implying that financial intermediaries have large relative effects on less-developed economies compared to more developed countries. In other words, developing countries have more room for financial and economic improvement. Fourth, the sampling has a gap in the long-term effect of financial development on economic growth, which suggests that the impact of financial deepening on the real financial sector takes greater time. Fifth, financial development may enhance economic growth both through more rapid capital accumulation and technological change, enhancing the channel productivity far more

²⁰ However, most empirical studies use Granger causality modelling analysis. For more comprehensive investigation, see Line (2002).

strongly. This suggests that the causal relationship between finance and TFP growth is stronger for developing than for industrial economies.

Al-Awad and Harb (2005) examined the link between financial development and the stock market in Middle East countries during the 1969-2000 period, using panel cointegration with time series methodologies. They found that, in the long-term, financial development and economic growth seem to be related to some level of growth but, in the short-term, it demonstrates that causality runs from economic growth to financial development. They suggest that neighbouring countries should adopt measures to reduce financial repression and to help increase financial development; this view is supported by Lucas (1988). Atindéhou et al. (2005) examined the relationship between financial intermediation and economic growth, using time series data for the 1961-1997 period for the Economic Community of West African States (ECOWAS). They discovered that, in three countries, there was no significant causal relationship between economic growth and all the financial variables used. In the remaining eight countries a one-way relationship could be observed from the variable, depending on the countries and the variables used.

4.5 SUMMARY AND CONCLUSION

This chapter has provided a comprehensive review of theoretical and empirical literature that links together endogenous growth theory and current theory of the functions of financial market and institutions, in order to study how financial market development affects economic growth rates. The argument that the stock market influences real economic growth rate depends on liquidity bands, the risk between sharing and pooling and, finally, the information and monitoring functions. Additionally, there is a link between the financial market and endogenous growth economic theory model, which is made by some authors, such as Romer (1986), Lucase (1988), Rebelo (1991) and Chick and Dow (2001, 2005) etc, in order to include the effect of stock market development. Other significant studies have used endogenous growth models to identify the channels through which financial markets affect the long-term growth rate of the economy. These include Greenwood and Jovanovic (1990), Berthelemy and Varoudakis (1996), Greenwood and Smith (1997) and Boyd and Smith (1998), all of which predict a two-way causality between financial development and economic growth.

Nevertheless, the birth of the new endogenous growth theory has facilitated the development of improved growth models where the long-term rate could be affected by a number of variables. These include technology, education and health policies, the process of economic development, capital accumulation, government policies and institutional activities and the extent of financial development. Additionally, the role of financial factors in the steady-state of the long-term rate in the neo-classical model could be related to the level of capital stock per-worker or to the level of productivity, but not to their respective rates. Notwithstanding this, external to the steady-state, financial elements could affect the transitional growth rate where it is not in the long-term growth rate. During development of the endogenous growth theory, available literature and the current research provided insights into how the reform of economics on long-term growth rate related to the macro-economic and financial institution performance. This framework has been reviewed by Stiglitz and Ellerman (2001), Dabrowski et al. (2001), Currie (2003) and other sources from the World Bank and IMF.

The argument of this chapter is that economic growth is a function of stock market development and economic reform indicators, with the main determinants of growth being set as the control variables. A comprehensive theoretical framework linking stock market development to economic growth has been presented, from which a theoretical consideration has been made of how the financial system and stock market development could affect real economic growth in the neoclassical Solow-Swan model (1956) and the endogenous growth model of Lucas (1988) and Romer (1986). A theoretical link between finance, stock market development, saving, investment and economic growth has been identified. The next chapter theoretically and empirically investigates the stock market in emerging countries, exploring the issue of whether the stock market and banking system are complementary or substitutes for each other. The relationship between economic reform programmes, macro-economic reform and stock market performance variables will be introduced.

CHAPTER FIVE

EMERGING STOCK MARKETS

5.1 INTRODUCTION

The main outcome of Chapter Four was the establishment of a significant relationship between economic growth and the stock market. Building upon this understanding, Chapter Five comprehensively considers a number of theoretical and empirical studies by Levine (1991), Demirgüc-Kunt (1992), Atje and Jovanovic (1993), Levine and Zervos (1995), Demirgüc-Kunt and Levine (1996), Harris (1997) and Demetriades and Andrianova (2004), among others, who have contributed to an understanding that stock market development affects economic growth in developing countries. Stulz (1997) explained that economic reform in emerging markets has led to an increase in equity that flows rapidly from industrial or developed countries to developing markets; this was supported by empirical evidence in Bekaert and Harvey (1995) and Henry (2000a) where financial capital is classified as flowing from industrial countries to emerging economies, to their mutual benefit. In addition, endogenous growth models demonstrate that the rate of economic growth is related to financial development, level of technology and income distribution.

Financial systems are, today, classified as bank-based or market-based (stock markets). This division is exemplified by the Anglo-Saxon market-based models of capitalist economies which allow for private investment and private ownership, and the other model, Germany, where the bank-based model has been practised more widely by Eastern European countries. These latter models were centrally-planned or, to be politically correct, non-socialist economies (Hall and Soskice, 2001). The UK and US are market-based and these countries have similar long-term growth rates²¹. Throughout the world, the type of financial model practised by sovereign countries reflects the type of government as a regime in power. Many Eastern European, Middle Eastern and African countries, including Libya, have practised socialism for a long time. However, in the light of recent trends, and under the direction of the

²¹ Demirgüc-Kunt and Levine (1999: 2) argue that in "bank-based" financial systems such as those of Germany and Japan, banks play a significant role in mobilising savings, overseeing the investment decisions of corporate managers, allocating capital and providing risk management vehicles. In "market-based" financial systems such as in the UK and US, securities' markets share centre stage with banks in terms of transferring society's savings to firms, exerting corporate control and easing risk management.

CHAPTER 5: EMERGING STOCK MARKETS

IMF and World Bank, many countries are now reforming their economies and gradually adopting capitalism in order to rescue their economies. In this context the World Bank (1994, 1989) has argued for the establishment and promotion of stock markets in developing countries in line with those existing in developed countries.

As mentioned in Chapters One and Two, Libya has been subject to many economic sanctions in recent years. It has now started to reform its economy akin to most developing countries. Today, the Libyan state is gradually transforming the economy from being centrally planned to a free market orientation, with the aim of creating more scope for private ownership in the retail trade and for small and medium-scale business and agricultural sectors to investment sourced from both local and foreign companies. The attempt to consolidate efforts towards liberalisation is intended to solve the current economic problems and provide more employment opportunities for Libyan nationals by increasing the economic growth rate. The key objective behind the establishment of the Libyan stock market was to contribute, and to support, its allocation process in order to strengthen the national economy (see Chapter Two: Subsection 2.4.3). The purpose of this is based on the growth theory model of the emerging economy which is similar in many aspects to the Libyan economy. Therefore, this explanation will consider the emerging market experience, attribute part of the success of developing countries to the fact that the majority of developed nations have stock markets and that, by developing its stock market, Libya achieved significant levels of economic growth. Thus, the case for the development of a stock exchange in Libya is founded upon the assumption that stock markets contribute to economic growth. This assumption is tested by several empirical examples (see Chapters Seven and Eight). An analysis of the role of the stock exchange in emerging economies is also undertaken.

This chapter, therefore, is divided into ten main sections, related to stock market performance. Section 5.2 commences with various definitions of stock markets. In Section 5.3 the functions of the stock market are explained, whilst Section 5.4 describes the importance of the stock market. Section 5.5 addresses the emerging market and economic growth. Five potential stock market performance variables are identified in Section 5.6 and, in Section 5.7, financial maturity measuring variables are identified. In Section 5.8 financial crises in developed and emerging markets are discussed. Section 5.9 deals with the relationship between economic reform and stock market performance. Section 5.10, is concerned with the relationship between macro-economic reform and stock market performance variables are provided in Section 5.11.

5.2 DEFINITION OF A STOCK MARKET

A stock market can be considered a very sophisticated market place, where stocks and shares are the traded commodity. At the same time, it is central to the creation and development of a strong and competitive economy. It is a key to structural transformation in any economy from traditional, rigid and insecure bank-based systems to a more flexible, more secure economy that is immune to shocks, fluctuations and lack of investors' confidence (Stapley, 1986). According to Arnold (2004), stock markets are places where governments and industry can raise long-term capital and investors can purchase and sell securities. Typically, whether markets trade in shares, bonds, cattle or fruit and vegetables, they are simply mechanisms to allow the possibility of trade between individuals or organisations. Whilst some markets (e.g. for livestock) are physical spaces where buyers and vendors meet to trade, others (e.g. for foreign currency) are national networks, based on communication using telephone lines and computer links with no physical meeting place. Additionally, very few stock exchanges around the world still possess a physical location where buyers and sellers meet to trade. Stock markets can be classified into debt and equity markets, or may involve short-term debt instruments (money markets) and longer-maturity financial assets (capital markets), including cash or spot market and futures markets, as shown in Figure 5.1.



Figure 5.1 Classification of Stock Markets

5.3 THE FUNCTION OF STOCK MARKETS

Many people may not see the stock markets and their economic functions as being important to them but, in fact, they represent a relationship between the disparate sectors in social society and between savers and producers as the saving sector needs to employ their savings in more beneficial and ambitious projects. Additionally, the productive sectors always require financial sources to assist them in continuing to perform more in the function of economy, in which stock market performance and functions of basic economy transfer funds from people who have amassed surplus to those who have a paucity of funds (Henry, 1997).





Figure 5.2 shows that borrower-spenders must borrow funds from lender-savers to finance their spending in two ways. The first is an indirect transfer by financial intermediary institutions, such as banks and other commercial organisations; the second is a direct transfer where borrowers obtain funds directly from lenders from the sale of financial instruments (securities). As the financial intermediary holds the largest part of the investment to reduce the risk to the economy, and low interest rates would lead to increased investment, financial intermediary institutions work on the transfer of funds from lenders to borrowers more

CHAPTER 5: EMERGING STOCK MARKETS

efficiently compared with the absence of these institutions. In other words, intermediary institutions have a better incentive to assume the risks resulting from the possession of investment instruments because they have huge financial possibilities allowing them to diversify their portfolios and reduce investment risk rate, which is assisted by the availability of its financial experts. Their specialists take advantage of a rising surplus from the management of purchasing and selling these investment tools. In addition to the previous discussion, the economic function of these markets can be identified in the following ways.

- 1. To provide or increase the amount of financial resource available. The stock market offers many opportunities for both creditors and civilians via the provision of multiple investment channels.
- 2. To provide financial information and projects relating to various financial assets available in the stock market, regarding information of the financial situation of companies, thus reducing the cost of access to such information in terms of effort, time and risk.
- 3. To provide liquidity for owners of various financial assets.
- 4. To assist in the development of diverse methods of financing in the short, medium and long term for the projects.

In point of fact, it can be observed how important a stock market is for the economy, since it allows movement of funds from persons who possess them and have no investment opportunities to those who enjoy these opportunities, by using the stock market function to increase production and to achieve economic efficiency and improve the level of prosperity in society. Financial intermediaries include commercial banks, savings banks, investment banks and specialised lending institutions, insurance companies and pension funds. These institutions play the crucial role of mediator in transferring funds from lenders to borrowers. The common factor between these intermediaries is the possibility of accessing funds through the creation of the debt on the same (in deposits coffers), then borrowing from the public to invest these funds via the purchase of investment instruments "stocks, bonds".

5.4 THE IMPORTANCE OF STOCK MARKETS

Stock markets perform a central role in economic development worldwide. Theorists and researchers present diverse opinions whilst arguing the importance of stock markets in increasing economic development in recent years by attracting and consolidating savings and

other forms of capital, which is so critical for the development and growth of both the private sector and trade in general. This has increased the stock market's role in commerce, information technology, communication and management. Singh (1999: 347) suggests three critical elements of a stock market which can improve economic growth:

- Increasing savings and investments;
- Improving the productivity of investments;
- Raising the profitability of existing capital stock.

The most significant contribution to the theoretical literature of stock market development²² and economic growth has been made by Levine (1991), Bencivenga and Smith (1991), Diamond (1996) and Fulghieri and Rovelli (1998) who supported the stock market for economic growth and further held that stock market liquidity is necessary for economic growth. Levine (1996) conducted an empirical study on 38 cases from developed and developing countries and inferred that stock markets might affect economic activity via the increase of liquidity. At this point an addition of increased liquidity could be the reason for a poor or slower growth rate in economic growth. However, this needs to be further investigated to establish what level of liquidity could guarantee economic growth. It has more to do with the secondary market, where stocks are traded, than the primary market, where stocks are issued, which can assist the new firms listed in increasing their capital. To alleviate their worries, liquid equity markets make investment less risky and more attractive because they allow savers to acquire an asset equity and to sell it rapidly and inexpensively to facilitate access to their savings or in case they wish to alter their portfolios.

Based on these arguments Singh (1992a, 1996, 1997 and 1999) and Singh and Weisse (1998) argue that stock markets in developing countries do more harm than good and that a bank-based financial system is more suitable for the developing countries. Singh (1997: 780) points out that, "Stock markets are potent symbols of capitalism but paradoxically capitalism often flourishes better without their hegemony. [...] Developing countries simply cannot afford the luxury of stock markets. As Keynes noted [...], when the capital development of a country becomes the by-product of the activities of a casino, the job is likely to be ill-done." On the other hand, Kenny and Moss

²² See a more comprehensive list of studies concerning development of stock markets in an early study by Cross (1973), French (1980), Gibbons and Hess (1981), Lakanishok and Levi (1982), Keim and Stambaugh (1984), Jaffe and Westerfield (1985), Smirlock and Starks (1986), Anup and Kishore (1994), and Agrawal and Tandon (1994), among others.

CHAPTER 5: EMERGING STOCK MARKETS

(1998) have gone to some lengths to reject Singh's suggestion that stock markets will cause such problems as to make them do more harm than good for developing economies, and to develop a justification for their opinion that the focus should be on stock markets rather than on banks in developing countries. They argue that, first, short-terms becomes less necessary with the creation of investment instruments that no longer have to mach the preferred liquidation dates of equity holders. As stock can be sold at any time, much of the liquidity risk will be reduced. Second, stock markets can make equity finance increasingly cheaper for firms. Stock markets can do that by encouraging new supplies of capital. This, in turn, reduces the average cost of capital and new investors entering the market will raise price-to-earnings (P/E) ratio. Finally, companies, by financing their activities through issuing equities, will have the opportunity to take risks much greater than they could with finance from a bank loan, they will also be more willing to take risks than company owners because that risk has been diversified among many investors. They also argue that because of the need for reserve requirements, moving capital from the informal market to the banking sector crowds credit; stock markets are free from this problem.

In conclusion, there are many factors which have helped financial markets to gain prominence during the 1990s in many countries. These include an increase in the private sector's role in various economic activities, innovative technological advances and the phenomenal speed of access to financial information. In addition, there have been advances in modes of communication and the emergence of new and effective financial tools besides new legislation that guarantees freedom of trade and movement of capital. All of these indicate the significance of stock markets in providing further opportunities for growth. Fortunately, the last few years have witnessed an increase in the importance of developing and establishing stock markets across the Arab world, whatever the reason underlying the establishment of a particular stock market, which bears many advantages such as reducing the costs of financing and vital contributions to a nation's capital market (Abdelnour, 2003). As a result, stock markets have been established in Bahrain, Egypt, Jordan, Kuwait, Lebanon, Libya, Morocco, Oman, Saudi Arabia and Tunisia.

5.5 EMERGING STOCK MARKETS IN DEVELOPING COUNTRIES

In recent years emerging markets have had much greater importance in the worldwide investment community. For instance, Korajczyk (1996) argued that emerging markets have become more integrated with the worldwide capital markets which have been changed in the particular time of the preceding seven years. In 1996, however, the number of stock markets

throughout the developing countries continued to rise to constitute almost 60 per cent of the entire world stock markets with a total capitalisation of about 1.9 trillion US\$. It is suggested that, in these countries, liquidity has increased even faster and turnover leapt from 26 to 85 per cent of emerging market capitalisation between 1985 and 1999 (Levine, 2003). According to the International Finance Cooperation (IFC, 1997), all stock markets in developing countries are considered to be emerging economies as classified by the World Bank, which are either low- or middle-income economies (low- and middle-income is defined as those in 1995 with per-capita GDP of 9,385 US\$ or less) regardless of their particular level of development. By the end of 1999, among all economies with an annual income per-capita of less than 9,266 US\$ and with populations of greater than 30,000 people, 157 nations were designated as being less developed countries and 49 as being industrial nations (Füss, 2002). According to the Global Stock Market Fact-book (2007), Libya (in 2005) was classified as being in mecategory or developed economies with an upper- or higher- middle- income per-capita of 3,466 to 10,725 US\$ with populations of greater than 6 million people, were designated as being less developed countries. Table 5.1 and Figure 5.3 present a classification of developed and developing countries after income per-capita during 2005.

 Table 5.1 The Population and Income Per-capita Indicators of the World including Emerging

 Market Countries, 2005

Income Category	No. of Economies	Population (millions)	GDP (billions US\$)	GDP Per-capita (US\$)
Low Income Countries (755 US\$ and Less)	54	2,352	1,377	585
Middle Income Countries (MIC)				
 Low Middle Income (876-3,465 US\$) 	58	2,475	4,760	1,923
• High Middle Income (3,466-10,725 US\$)	40	600	3,379	5,634
High Income Countries (10,726 US\$ or more)	56	1,011	35,643	35,264
Emerging Markets	152	5,427	9,515	1,753
Developed Markets	56	1,011	35,643	35,264
World	208	6,438	45,135	7,011

Source: Standard and Poor's (2007).



Figure 5.3 Distribution of World GDP and Population, 2005

Source: Standard and Poor's (2007).

CHAPTER 5: EMERGING STOCK MARKETS

The growing number of empirical studies by Kirkpatric (2005), with respect to emerging markets, tend to support the link between stock market development and economic growth. Levine and Zervos (1996), using data from 41 countries over the period 1976-1993, show that there is a positive correlation. In addition, King and Levine (1993a, b) show that there is a strong link between financial development and the economic growth rate in the long-term. Furthermore, it has been suggested that there is increased awareness of the relationship between the financial market, financial development and economic growth (Singh, 1996). Demmogrüc-Kunt (1995) found a positive correlation between stock market development and the development of financial intermediaries. In order to identify the effect of liberalisation on stock markets, Levine and Zervos (1998b) used data from 16 emerging markets. They examined the impact on liberalisation of restrictions on capital and an international flow on stock market development. They discovered that liberalised stock markets became larger, more internationally integrated, more liquid and more volatile. Table 5.2 shows that stock market liquidity increased significantly in 12 out of 14 countries following liberalisation controls in international capital flows, which can assist economic growth to increase more rapidly by encouraging stock market liquidity. Table 5.2 further indicates that stock market volatility increased in 7 out of 11 countries without significant decrease in any of the 11 nations following liberalisation, where it would be easy for international capital flow restrictions to increase liquidity and volatility.

Country	Year of Liberalisation	Size	Liquidity	Volatility
Argentina	1989	\mathbf{A}	\wedge	\wedge
Brazil	1983			\mathbf{A}
Chile	1988		\mathbf{A}	
Colombia	1989-91		$\overline{\mathbf{A}}$	\wedge
India	1990-91	\wedge	$\overline{\mathbf{A}}$	$\overline{\mathbf{A}}$
Jordan	1987		$\overline{\mathbf{A}}$	
Korea	1981-92	\wedge	$\overline{\mathbf{A}}$	
Malaysia	1986			n.a.
Pakistan	1990	\wedge	\rightarrow	\mathbf{A}
Philippines	1988		$\overline{\mathbf{A}}$	
Portugal	1988		$\overline{\mathbf{A}}$	n.a.
Thailand	1988		$\overline{\mathbf{A}}$	\wedge
Turkey	1990	\wedge	$\overline{\mathbf{A}}$	n.a.
Venezuela	1988		$\overline{\mathbf{A}}$	\wedge

Table 5.2 Stock Market Performance after Liberalisation in Emerging Markets

Notice a: Arrows indicate increases; dashes indicate no significant change; n.a. indicates data unavailable.

Source: Levine (1996).

Consequently, the following two sections (5.6 and 5.7) are developed with reference to the Libyan context. Section 5.6 considers the stock market variables presented in the hypotheses testing in respect of the dependent variables, while section 5.7 introduces the banking system (independent) variables.

5.6 MEASURING STOCK MARKET VARIABLES

The following variables are those suggested for measuring market (size, liquidity and activity), market efficiency and market volatility related to stock market performance. Since this study focuses on determining Libyan stock market performance after the implementation of the economic reform programme, the first three variables in this research will be tested as the Libyan stock market was founded very recently (see Chapters Seven and Eight). These variables are, therefore, of particular analytical interest.

5.6.1 Market Size

The size of the stock market from the volume of capital and listed companies is determined, and can be measured, by the number of listed companies. The rationale behind this measure is that most of the stock market size is positively correlated with the ability to mobilize capital and diversify risk on the worldwide economy. Hartmann and Khambata (1993), Demirgüc-Kunt and Maksimovic (1998), and Subrahamanyam and Titman (1999) have calculated market size as an indicator for stock market development. It consists of two dimensions. Total market capitalisation is measured by the total value of listed shares on the stock exchange and the number of listed companies in the stock market can indicate to what extent that value is increased among the listed firms. In addition to these two dimensions, market capitalisation per GDP can be considered as an indicator of market size. Bekaert and Harvey (1995, 1997) determined that the ratio of equity capitalisation to GDP is a useful tool in characterising the time-series of market integration. Furthermore, Morck et al. (2000) and Wargler (2000) contend that larger market size relative to economic activity contains more informative prices which assist investors and managers to differentiate between sound and poor investments through more precise measures of stocks.

5.6.2 Market Liquidity

Stock market liquidity is the ability to buy and sell securities easily. Greater stock market liquidity can encourage investors to obtain more information about firms which prefer to

CHAPTER 5: EMERGING STOCK MARKETS

change portfolios quickly and reasonably. Market liquidity is measured by the total value of shares traded on the stock market divided by GDP and the turnover ratio is the value of the total shares traded divided by market capitalisation. Furthermore, liquidity is a key characteristic of stock market development because more liquid stock markets theoretically improve the allocation of capital to their optimum use and influence investment in long-term growth. For instance, the critical issue of liquidity provision is stressed theoretically in Levine (1991) and empirically in Levine and Zervos (1998). Considerable research has focused on stock market liquidity size. Levine (1991), Bencivenga and Smith (1991), Bencivenga et al. (1995, 1996), Diamond (1996), Greenwood and Smith (1997), Fulghieri and Rovelli (1998) and others contest that stock market liquidity is necessary for economic growth. Levine (1997) suggests that stock market liquidity encourages, or at least strongly forecasts, corporate investment even though much of this is financed through reserved earnings and bank loans, rather than equity issues. Miller (1991) argues that greater liquidity has a direct impact on the effectiveness of the government function of the stock market which increases market activity to reaching information which, in turn, increases the content of share prices. Additionally, the effective use of the stock market for corporate control is required for the market to be liquid. Ahimud et al. (1997) and Henry (2000a, b) contest that increased stock market liquidity can also reduce the cost of equity capital via a reduction in the expected return that investors require when investing in equity to compensate them for associated risks i.e. risk premium.

5.6.3 Market Activity

Stock market activity is determined by two factors: the value and volume of trade divided by GDP and the turnover ratio and number of transactions. The works of Levine (1991) and Bencivenga et al. (1995, 1996) claim that higher transaction costs are associated with lower value-traded ratios, which may affect saving rates through to economic growth rates. Using cross-country regressions during the period 1976-1993, Levine and Zervos (1998) argued that stock markets might improve growth through liquidity. They found that equity market activity was positively correlated with measures of real activity. Atje and Jovanovic (1993) adopted a similar approach for forty countries over the period 1980-1988. They discovered significant correlation between the value of stock market trading as a percentage of GDP and economic growth. Peterson and Fialkowski and Petersen (1994) and Dactar et al. (1998) further illustrated that the quoted spread is a poor proxy for actual transaction costs faced by investors and called for an alternative proxy such as turnover rate, which may perform a better job of

capturing an asset's liquidity. Brennan et al. (1998) and Chordia et al. (2001) found a negative and significant relationship between expected returns and the level of liquidity measured by trading volume and share turnover.

5.6.4 Market Efficiency

A valuable addition to the literature in this subject is the definition of efficient financial markets introduced by Malkiel (2003: 60). He states that efficient financial markets are markets which do not allow investors to earn above-average returns and arbitrage opportunities without accepting above-average risk. Malkiel (2003) argues, that despite this, some anomalous behavior of stock prices can create portfolio trading opportunities which enable investors to earn extraordinary risk adjustment return. This means that stock markets are still efficient even if some interruption or impediments happen. Moreover, Malkiel (2003) criticizes some studies which find that, if the reaction for new information, or important news announcement, is only grasped over a period of time, stock prices will exhibit the positive serial correlation found by investors. He shows that the evidence of efficiency occurring systematically in the stock market is often rather thin (see Fama (1988) and Fama and French (1998)).

The weak form efficiency hypothesis has been tested by a number of authors, for example, Fama (1965) for USA; Dryden (1970) for UK; Solnik (1973) for eight European markets; Conrad and Juttner (1973) for Germany; Jennergren and Korsvold (1975) for Norway and Sweden; Laurance (1986) for Malaysia and Singapore; Aybar (1992) for Turkey, etc. These studies provide mixed results. Butler and Malaikah (1992) examine the behaviour of individual returns in two stock markets, Saudi Arabia and Kuwait, over the period 1985-1989. They used serial correlation and ran several tests to evaluate the weak form efficiency in these two stock markets. Their study tries to investigate the similarities and dissimilarities of these stocks, regarding exchange mechanisms and efficiency. They conclude that institutional factors contribute to operational inefficiency in the Saudi Arabia stock market. This inefficiency includes trading delay, illiquidity, market fragmentation and the absence of official markets. Similar results for Arab markets were provided by (Gandhi et al. (1980) and Al Ajami (1994)). In particular, Al Ajami (1994) investigated the efficiency of four Gulf states' markets. The countries studies were, Kuwait, Saudi Arabia, Oman and Bahrain. Using a serial correlation test and a runs test, Al Ajami concluded that the finding of this study provides evidence to substantiate the claim that a small and thin stock market does not *conform to the efficient market hypothesis.* Even in the more developing markets the evidence does not strongly support the efficient market theory. Dryden (1970) studied the price changes of 15 stocks listed on the LSE, testing that the results of some statistical tests agree with the random walk model, despite the presence of small systematic changes *which suggests that perhaps there are some patterns present in share prices*.

5.6.5 Market Volatility

An average of 12 months' records could guide and may even provide concrete ways to comprehend the issues involved. The literature review revealed that different authors focused empirical evidence upon stock market liberalisation and volatility. This includes the work of De Santis and Imrohoroglu (1997), Aggarwal et al. (1999), Huang and Yang (1999), Kim and Singal (2000), Kassimatis (2002), Bekaert and Harvey (2003a, 2003b) and Bekaert et al. (2005). Chou (1988) contends that, in 1974, high stock market volatility led to the crash of the US stock market which indicates the importance of classifying the sources of volatility. Therefore, stock market volatility can mirror changes in money supply and oil prices (Engle, 2004).

Koutoulas and Kryzanowski (1996) provided a detailed empirical analysis of the major determinants of stock market volatility and how these responded significantly to exchange rate volatility in Canada. Kearney (1998) obtained similar results for Ireland. Fang (2000) found a negative reduction effect in the stock market return process related to the Asian crisis in Taiwan. Various studies have found a lower volatility in developed markets than in emerging markets. For instance, Harvey and Ferson (1993, 1994) supply evidence that emerging markets have higher volatility than their industrial or developed counterparts. Also, Stone (1990), Divecha et al. (1992), Wilcox (1992) and Harvey (1994a, 1995a) document high volatility in emerging market nations. Low correlation between developed and emerging markets reduces volatilities in portfolios from these markets (Hassan et al., 2006).

5.7 THE MEASUREMENT OF FINANCIAL MATURITY

Many of the previous literature studies focused on the relationship between the financial system environment and economic growth. Banks are involved in uplifting economic growth and, historically, play quite a balanced role within the markets. Similarly, economists focused upon banks. In this context, Schumpeter (1912) and Patrick (1966) highlighted the role of banks and argued that the services provided by the banking system are essential for

CHAPTER 5: EMERGING STOCK MARKETS

technological innovation and economic growth. They additionally indicated that banks provide assurances to companies so that the latter could conduct business across borders. Goldsmith (1969) and McKinnon (1973) discussed conceptual descriptions of how the financial system affects economic growth; it is simply that the financial systems provide the foundation where the GDP of respective companies/countries increases or decreases or even remains stable. Diamond (1984), Greenwood and Jovanovic (1990), King and Levine (1993a, b) and Arnold and Walz (2000) determined that better capital allocation and productivity growth are interrelated. Fama (1985) argued that banks are more competent at reducing informational asymmetries by collecting information, because bank borrowers are also depositors and deposit history provides background information appertaining to the quality of the borrower. Boyd and Prescott (1986) argue that the banking sector has an uncomplicated ability to obtain information and, thus, allocate resources. Stiglitz (1985) and Bhide (1993) defend the idea that banks are more efficient than equity markets. Levine (2004) argued that the exogenous module of both stock market development and bank development help predict economic growth using two similar methods (panel GMM and OLS) over the 1975-1998 period. Through their empirical result he found that if Mexico's turnover ratio had been 68 per cent (the average of an OECD country) instead of the actual 36 per cent, it would have grown 0.6 per cent points faster per year. Similarly, if bank credit had been 71 per cent at the average of OECD countries instead of the actual 16 per cent, then the exogenous components of both bank and stock market development would have a significant impact on economic growth rate. Levine and Zervos (1998a) established that stock development is healthy with economic growth and that the stock market provides different financial services from those provided by banks. Gerard and Demirgüc-Kunt (1997: 26) mention that "banking systems and stock markets development are complementary, most likely because each produces and demands better information". In an important empirical study Levine (2002) argued that financial services provided by banks and markets are both critical to a country's economic development process. Furthermore, Beck and Levine (2000) provide evidence that industries which are heavy users of external finance grow faster in countries with higher overall rates of financial development; they then conclude that banks and markets might act as complements to one another in providing financial services and increasing economic development. Another challenge to finance and growth literature (Allen et al., 2005, 2007) comes in the form of individual country outlines. For instance, China is often mentioned as a counter example to the findings in finance and growth literature. Despite weaknesses in its formal banking system, China is one of the faster growing economies worldwide (Demirgüc-Kunt and Levine, 2008).
5.8 FINANCIAL CRISIS IN DEVELOPED AND EMERGING MARKETS

The discussion to this point has concentrated mainly on financial crises leading to major disruptions in financial markets²³; a phenomenon which began in advanced economies and then spread to emerging markets and low-income countries. The four types of factor that lead to such crises are increases, such as interest rates, in uncertainty, asset market effects on balance sheets and bank panics (Mishkin and Eakins, 2003). Fama (1989), Roll (1989) and Javed and Ahmed (1999) presented a summary of the literature on the US stock market crash of 1987, and Kaminsky et al. (1997) reviewed the results of 25 selected studies pertaining to currency crises. Harris (1987) provides the index of stock market and Kleidon (1992) identified the limit between buying and selling for the behaviour of the S&P 500 index.

Mishkin and Eakins (2003) contend that crises are caused by an increase in adverse selection and moral hazard problems that prevent financial markets from channelling funds to people with productive investment opportunities, leading to a sharp contraction in economic activity. The stock market crash in October 1987 inspired national links between stock markets worldwide. In recent years, however, many emerging markets have encountered financial crises. In Latin America, for instance, the Mexican crisis began in December 1994, when stock prices on the Bolas stock exchange had fallen almost 20 per cent from their September 1994 peak. Commencing in February 1994, the financial crisis saw a rise in interest rates overseas. Among East Asia a crisis began in July 1997 in Thailand, Malaysia, Indonesia, Philippines and South Korea. As Higgott (1998: 349) argued: "the Asian crisis is a contest of ideology between Asian and Anglo-American ways of organising capitalist production". In Thailand a foreign debt default in early February 1997 required over 8 billion US\$ of loans from the central bank to bolster the government. The financial markets of Thailand and South Korea both experienced declines in their securities' markets. This increased uncertainty in the financial markets of both nations and substantial declines were experienced in their securities' markets. As the IMF (1999: 17) indicate: "the Asian crisis differed from previous financial crises in which the IMF's assistance has been needed [...] Conventional fiscal imbalances were relatively small and only in Thailand were significant real exchange rate misalignments

²³ According to the IMF (1998: 74-75), a currency crisis can be defined as a sudden devaluation of a currency by speculative attacks or a massive reduction of foreign currency reserves as a result of defending the value of a currency. A financial crisis is a disruption of financial markets which is often caused by actual or potential failures of financial institutions that compel the government to rescue them on a large scale. Whilst a financial crisis is occasionally accompanied by a currency crisis, the latter does not necessarily evolve into a financial crisis.

CHAPTER 5: EMERGING STOCK MARKETS

evident". Furthermore, the collapse of currencies led to a rise in actual and expected inflation in these countries. Market interest rates rose to dramatic levels (over 100 per cent in Mexico). As a result, the increase in interest payments caused cash flow reductions for households and companies, which led to further deterioration in their balance sheets. This negative shock was most severe in Indonesia, Thailand, the Philippines, Malaysia and South Korea where the value of the currency in their stock markets declined between 50 and 80 per cent; the Mexican market declined 50 per cent from its peak value. Consistent with the US experience in the nineteenth and early twentieth centuries, stock market declines and increases in uncertainty were additional factors precipitating the full-blown crises in Mexico, Thailand and South Korea. In 1994 the Mexican economy was hit by political shocks that created great uncertainty. Finally, in the aftermath of crisis, Mexico began to recover in 1996. The East Asian nations, on the other hand, were recovering by 1999. In all these countries the economic hardship caused by the financial crises was incredible. Unemployment rose sharply, poverty increased significantly and the social fabric was stretched thinly. Mexico City, for instance, became one of the most crime-ridden zones in the world, while Indonesia witnessed waves of violent ethnic conflict.

Lehman Brothers was the fourth-largest US investment bank and 158-year-old firm which had survived the railroad bankruptcies of the 1800s and the Great Depression of the 1930s. It filed for bankruptcy protection on 15 September 2008, becoming a victim of the global credit crunch. The company listed more than 613 billion US\$ (343 billions UK£) of debt and made this the biggest crises in US history dwarfing World Com's insolvency 6 years earlier (BBC News, 2008; Onaran and Scinta, 2008; Taylor, 2008). On the Friday, Lehman's was an investment-grade bank with 42 billion US\$ of liquid assets. By the following Monday it was bankrupt. The New York firm employed 25,000 people worldwide including 5,000 staff in Britain (Boden, 2008 and Wood, 2008). Andrew Goodwin, a senior economist at Oxford Economics states: "There is a danger there could be more bad news in the next few weeks. We would tend to see an acceleration in job losses right about now" (Master, 2008). Lehman shares dropped 92 per cent in New York to 29 per cent from their 3.65 US\$ close on 12 September 2008. UBS AG, HBOS plc and Axa Sale saw a decline of more than 3 per cent for European stock markets (Onaran and Scinta, 2008). Finally, equity markets in Europe and the US fell to their lowest levels since mid-July. By mid-afternoon the Pan European FTSE Eurofirst 300 index fell 3.6 per cent while, in New York, the S&P 500 was down 2.6 per cent. Markets in Japan, Hong Kong, China and South Korea lost between 5 and 7 *per cent*, when they opened for business and fell sharply that were closed for holiday (see Figure 5.4).



Figure 5.4 The World Equities Crisis in Europe, American and Asia Pacific in September 2008

However, the stock market on this particular Monday witnessed its worst day's trading decline since the 11 September 2001 terrorist attacks. European markets dropped by 3.5 *per cent*. Similarly, oil prices declined to less than 100 US\$ a barrel for the first time since March 2008 (Wood, 2008). The benchmark FTSE Eurofirst 300 fell by about 2 *per cent* to 1097.85, with its losses for the week amounting to 5.5 *per cent*. Also, the index had lost over 30 *per cent* since the beginning of 2008. Some analysts believe that the Lehman collapse could lead to a reorganisation of the entire US financial system which has been stuck in the economic depression since the mortgage crisis struck more than a year previously (Aljazeera, 2008).

As with other developing countries, Libya has also experienced a financial crisis, as mentioned in Chapters One and Two. This started during the late 1980s and early 1990s and originated with the reduction of oil prices and exports. It was, additionally, a time of political uncertainty, due to the US sanctions against the Libyan economy beginning in 1992. Recovery then required more than seven years. Early in 1980 the balance of trade had reached 4,483 million LD, falling to 1,925 million LD the following year. In mid-1986 oil prices collapsed; the price

Source: Financial Times (2008).

CHAPTER 5: EMERGING STOCK MARKETS

of a barrel fell to less than 10 US\$ in sharp contrast to the average price of 27 US\$ in 1985. In fact, Libya experienced cash flow problems that impacted adversely on economic growth, when oil revenues plummeted from 22 billion US\$ in 1980 to 5 billion US\$ in 1986, whilst GDP also declined from 35,500 US\$ to 24,000 US\$ during the same period, and average GDP per-capita fell from 10,900 US\$ in 1980 to 6,404 US\$ (Altunisik, 1995). This was due to the decline in oil prices and export quantities from 23.3 billion US\$ to 5 billion US\$ in 1988. In this regard, Fisher (1990) concluded that, during the 1980s, the Libyan economy was heavily affected by the low price and quantity of oil exported, whereby GDP recorded a negative growth rate of around 13.6 per cent in 1987²⁴. Mahamud and Russel (1999) point out that the trade ban and economic sanctions introduced by the US government against Libya during 1981 and 1986, respectively, froze all Libya's assets in America. Libyan exports subsequently decreased 1.6 million US\$ by 96.4 per cent. American imports of Libyan oil were blocked and there was a withdrawal of US companies based in the country, with their assets correspondingly frozen by the Libyan government. Furthermore, Mahamud and Russel (2002) explained that the US embargo and sanctions affected all infrastructure development programmes and plans. Resultingly, there was a decline in the production capacity of crude oil and the natural gas sector as almost all Libyan oil fields consist of ageing American-made equipment and operating companies were unable to obtain spare parts. Libya, however, began to search for other business markets and was able to compensate for this loss by exporting elsewhere via the National Oil Corporation (NOC). The Libyan mission to the UN reported in a letter the implication of Security Council resolutions 748 (1992) and 883 (1993) for Libya and quantified the cost of the sanctions as 33.602 billion US\$ from 15 April 1992 to 5 April 1999. This included 1.430 billion US\$ in the health sectors, 1.495 billion US\$ in agriculture, 6.610 billion US\$ in the livestock sector, 3.713 billion US\$ in transport and communications, 5.850 billion US\$ in mining and industry, 8.627 billion US\$ in trade and 5.877 billion US\$ in the energy sector.

As indicated by Fayad (2000), the decline in GDP growth rate meant that these mounting economic problems were attributed to various political factors: oil production; reducing imports; OPEC operations to control prices and oil export quantities; decreasing the number of

²⁴ After the sharp decrease in oil prices in 1986, Qaddafi stated "Analysts believe that oil cannot be dispensed with and (oil prices) will not slump continuously. They will reach a certain level and they will start to rise again. [While], every family plans how to manage the situation. The important thing is we will not be hungry, thirsty, without clothing until the end of the oil problem, when the situation improves. It means it is possible, in a simple way, to overcome any crisis. No matter how much we lost in price drop, in my opinion the loss was only in terms of luxury and extravagant items we used to buy." (Altunisik, 1996: 8).

CHAPTER 5: EMERGING STOCK MARKETS

foreign workers in Libya. Serious cash-flow problems and increased debt repayment difficulties resulted (Abuarroush, 1996). A decade later, during 1990, the balance of trade rose to 1,925.4 million LD due to the second Gulf war. With the collapse in the oil market during the 1990s, the trade surplus dropped to 766.3 million LD in 1993. Since the lifting of the UN and US trade sanctions against Libya in 2003 and 2004, respectively, an open-door economic policy and structural reforms began and a series of policy measures were undertaken through the government agenda of transition to enhance the role of the private ownership sector and reform programme of deregulation including liberalisation of the economy. The trade surplus again reached 9,208.7 and 29,402.2 million LD in 2003 and 2007, respectively. In 2000 GDP achieved a growth rate of 42.1 per cent or 7761.9 million LD to increase to 30848.4 million LD or 64.1 per cent in 2004, before reaching 69.3 per cent or 61834.2 million LD in 2007. This rise in the economic growth rate in 2000 is attributed to the growth in non-oil sectors by 57.9 per cent, which then dropped to 30.7 per cent in 2007 due to the oil price increases, while the oil grew by 69.3 per cent (see Appendix A, Table 3.12). Figure 5.5 illustrates that, during the 1980-2007 period, structural changes in Libya's GDP growth rate via economic activity resulted from fluctuations in supply and demand in international oil prices. Rodrik (1996) argues that the idea is tautological in the explanation: "That reform should follow crisis, then, is no more surprising than smoke following fire". Bates and Krueger (1993: 454) point out that: "[...], reforms have been undertaken in circumstances in which economic conditions were deteriorating. There is no recorded instance of the beginning of a reform programme at a time when economic growth was satisfactory and when the price level and balance of payment situations were stable". Related to this is the work by Sturzeneqger and Tommasi (1998), wherein crises could be "useful" for welfare, if the indirect but beneficial effect of reducing delay outweighs the direct but adverse effect of the crisis.



Figure 5.5 Structural Changes in GDP by Economic Activity (1980-2007)^a

Notice a: More data see (Appendix A, Table 3.12).

Source: General Planning Board, Economics and Social Indicators (1962-2000), IMF (2008) World Bank Outlook Database.

5.9 THE RELATIONSHIP BETWEEN ECONOMIC REFORM PROGRAMMES AND STOCK MARKET PERFORMANCE

As mentioned previously in Chapter Three, any programme of macro-economic reform or, in other words, consolidating the economic reform, aims to target the rate of growth to improve society's living standards. To achieve this, and to increase the level of investment in terms of stock market performance in order to play a vital role in attracting both local and foreign investment, there should be a direct link between economic reform programmes and the performance of a stock market in order to transforming the economy from a socialist to market-based economy. The development of research technique by Granger (1986) and Engle and Granger (1987) has encouraged many studies on the short and long-term to scrutinise the relationship between economic variables and stock market performance. There are various economic aspects, such as those of economic reform variable deregulation and privatisation.

5.9.1 Relationship between Deregulation and Stock Market Performance

Many previous studies have focused on the relationship between deregulation and stock market performance. Pardy (1992) highlighted the need for legal, institutional and regulatory changes in order to realise the necessary desired results of financial liberalisation in developing countries. Others, meanwhile, argue that there should be no regulatory discrimination for particular interest groups in an economy (Chuppe and Atkin, 1992). In this context, Claessens and Rhee (1994) performed an empirical study of significant decrease in the risk-adjusted cost of capital in developing countries, if foreign investors have encountered fewer restrictions in emerging markets. As recognised by Vittas (1992), investments in emerging markets have recently been of great interest to the worldwide investment community. Pratten (1993) described the alterations to the UK stock market since the 1920s and 1930s. These include institutional changes to the relative importance of categories of participants and market performance. The most significant changes were the relaxation of rules governing investment by trustees introduced by Trustee Investments Act of 1961 and the increase in the number of markets plus the reduction in transaction costs for large deals associated with the "Big Bang" transformations to the London market since 1986, where screen-based trading replaced trading on the floor of the stock exchange. Behind the regulatory changes and shifts in the pattern of ownership, there have been alterations in the role, importance and perception of the stock market. In Britain its role has been enhanced by an extensive programme of privatisation.

5.9.2 Relationship between Privatisation and Stock Market Performance

Privatisation became one of the primary policies adopted by the IMF and the World Bank as part of its economic reform and structural adjustment programme to remedy deteriorating economic conditions, especially in developing nations. They suggested the introduction of market-oriented policy reforms to deregulation, liberalisation, nationalisation, budget deficit and balance of trade, to cut public expenditure, reduce state intervention and trade controls, devalue currencies, public sector inefficiencies and privatise or liquidate public ownership enterprises. In developing countries, between the early 1980s and late 1990s, forms of privatisation were adopted; for instance, in Libya, India, South Korea, Thailand, Malaysia and the Philippines, within their its all-pervasive economic sectors or activities, including the sales of equity and of assets, electricity, management contracts, banking, insurance, water supply, oil and gas, telecommunications, airlines, hotels, food processing, tourism and corporatisation (Mohamed, 1995 and World Bank, 1995a). In Asia the number of privatisation transactions increased from 108 to 367 during the periods 1980-1987 and 1988-1993, from 136 to 561 in Latin America and from 210 to 254 in Africa (World Bank, 1995a). During the 1988-1993 period, in Asia, the total value of privatisation transactions was 19.7 billion US\$, in Africa 3.2 billion US\$ and in Latin America 55.1 billion US\$ (World Bank, 1995a). By the mid-1990s, Malaysia privatised about 357 projects, including numerous large companies in telecommunications and energy. In June 1993 the Philippines replaced 310 companies out of 419 non-performing assets and 78 out of 122 state ownership enterprises and similar programmes of privatisation were launched in Bangladesh and Pakistan (i.e. see Sader 1993, World Bank, 1994 and 1995a; Kelegama, 1995). The report of the Organisation for Economic Co-operation and Development (OECD) for 2000 stated that total revenues of privatisation in all nations had increased by 10 per cent during the previous 10 years, which equals 145 billion US\$. The largest sole operation was in Italy. This was worth 14 billion US\$, representing 34.5 per cent of the shares of the largest national public electric power company.

Figure 5.6 shows that, when privatisation and financial regulation increases, the deepening of the stock market contributes to the increase of the efficiency of the whole financial system. Thus, the efficient capital allocation with higher factor productivity will contribute to bring the country towards a faster economic growth rate. Bekart et al. (2002) argue that privatisation and regulatory reform programmes help create more efficient domestic capital markets which can channel savings to investment by private ownership and the liberalisation of their financial

market. The aforementioned authors suggest that economic reforms reduce governmentfunded systems by private social security which may result in an increase in private savings (Edwards, 1995).



Figure 5.6 Conceptual Framework of Privatisation and Regulatory Reforms

In theory, a privatisation programme requires a well-organised stock market, which can help to increase market capitalisation and market liquidity. Al-Robiai (2004) proposed that two kinds of private sector exist: the organised market system of accounts and the disorganised private sector. In Libya it had become evident by the late 1980s that the nationalisation of business had failed to improve the economy, and the government began to introduce a programme of privatisation. An extremely limited part of the public sector was privatised, including some small and medium-sized enterprises which were transferred to their existing employees. This research will pay greater attention to examine the impact of economic reform programmes on stock market performance as applied to the Libyan economy in terms of one of the world's emerging markets.

5.10 THE RELATIONSHIP BETWEEN MACRO-ECONOMIC REFORM AND STOCK MARKET PERFORMANCE

Changes in macro-economic variables, such as interest rates, inflation rates, exchange rates, the budget deficit, per-capita income and rate of growth, affect stock returns and prices. This

Source: Gentzoglanis (2007).

part of the research allows for the natural arguments of macro-economic variables which should have a considerable impact upon stock market performance.

5.10.1 Relationship between Interest Rates and Stock Market Performance

Rates of interest must be positive to encourage society to save and to establish new businesses, because these directly affect everyday lives and bear a critical impact upon the health of the economy which accelerates the rate of growth. Interest rates determine personal decisions about whether to consume to save or to buy, and whether to purchase bonds or to deposit funds in savings accounts. They affect the economic decisions of businesses and households (Mishkin and Eakins, 2003). In the mid-1960s interest rates increased due to the liquidity effect being dominated by price-levels, income and expected-inflation effects. By the 1970s interest rates reached levels unprecedented in the post-World War II period. The liquidity effect indicates that a rise in money supply growth will lead to a decline in rates of interest. Despite this argument, the literature review in many financial studies examined alterations to the relationship between stock returns, interest rates, inflation and real activity (Fama, 1981 and 1990; James et al., 1985; Mandelker and Tandon, 1985; Asprem, 1989; Schwert, 1990; Lee, 1992; Canova and De Nicolo, 2000). For instance, Shiller and Beltratti (1992) found an indirect relationship between stock returns and interest rates. Meanwhile, others have considered the relationship between stock returns and macro-economic variables (Chen et al., 1986; Fama and French, 1989; Ferson and Harvey, 1991; Cheung and Ng, 1998; Bredin and Fountas, 2005).

Fry (1980) recognised a positive relationship between savings and the rate of interest for fourteen developing countries. Yusuf and Peters (1984) recognised this in South Korea, whilst Leite and Vaez-Zadeh (1986) do the same for six African countries and Ostry and Reinhart (1992) for 13 developing countries. Assorted results were obtained by Gupta (1987), Lahiri (1989) and Villagomez (1997). Other studies that have found a negative or insignificant relationship include Giovannini (1983, 1985), Oshikoya (1990) and Marwan and Ahmad (2005). Seppala (2000) classifies one of the oldest problems in economic theory as being the understanding of the "*term structure of interest rates*" (see e.g. Fama (1979, 1990) and other research by Asprem (1989) and Konan (2008) discover that there is a negative relationship between interest rates and stock returns in Korea. Abken (1993) and Blough (1994) provide literature surveys. Other studies, by Svensson (1994a, b), Fendel and Frenkel (2005) argue that monetary policy and the role of the "*term structure of interest rates*" as information in

financial resources alter motives. Ritter et al. (1997) presented a model of discounted cash flow to explain the relationship between interest rates and stock prices. Peavy (1990) discovered that there is a negative relationship between interest rates and stock prices. Additionally, Baye and Jansen (1995) note that there is an unchanging relationship between interest rates and stock prices.

From all the previous discussion in the empirical study, the relationship between interest rates and stock market performance was scrutinised to determine stock prices and returns. It is contended that changes in rates of interest do not affect all stocks to the same degree but, in general, the relationship between interest rates and stock prices is negative. This impact will be examined in the empirical part of this research (Chapter Eight).

5.10.2 Relationship between Inflation Rates and Stock Market Performance

In an early empirical analysis, Fisher (1930) argued that the nominal interest rate is mainly related to the information available concerning possible future values of the inflation rate. This hypothesis featured the conclusion known as the "*Fisher effect*", which holds that the nominal rates of interest on financial assets should move effortlessly one-to-one with expected inflation. Fama (1981) identified a negative relationship between inflation and economic activity, but a positive relationship between economic activity and stock prices; Benderly and Zwick (1985) supported these results. Other studies, by Fama (1976), Linke and Zumwalt (1980) and Sharpe (1999), have demonstrated a negative relationship between expected inflation and stock price returns. Various other studies have shown that there is a negative unexpected inflation effect in stock returns (Bae, 1991 and Flannery, 1994). Rapach (2002) also found a negative relationship between inflation and stock prices; Shiller and Beltratti (1992) identified a small correlation between inflation and stock returns.

Using annual data from 1875 to 1970 and monthly data from January 1953 to December 1971 Jaffe and Mandelker (1976) demonstrated a positive and statistically insignificant relationship between returns on stock and inflation rate. Nelson (1976) and Boudoukh and Richardson (1993) investigated the relationship between the rate of inflation and returns on common stocks over the post-war period using the Fisher effect for stock market returns. It is a common empirical finding that stock returns and inflation rates have a mutually negative relationship. Cagan (1974), however, discovered a positive relationship between common stock returns and the inflation rate for several countries over the period 1939-1969. Boyd et al. (2001) highlighted the relationship between macro-economic measures via inflation rates and

CHAPTER 5: EMERGING STOCK MARKETS

financial market performance. They established that there is a statistically negative relationship between inflation and the stock market, wherein it is very evident that stock market development is not a completely exogenous determinate.

The economic reform programme that Libya has been implementing aims to tackle inflation. If this is high it will affect interest rates negatively. Libya has focused on maintaining inflation at a low rate in order to prevent it from having a negative impact upon interest rates. Significant measures which can be implemented include lifting limits on interest rates and eliminating lending limits to the private and public sectors (Handy and Subramanian, 1997). Accordingly, interest rates will depend on market forces, thus, the gap between nominal interest rates and inflation rates becomes narrower. Here, banks and other financial institutions are able to attract more savings and to invest these in numerous development projects. Low inflation and interest rates are sufficiently positive to attract both money supply and demand.

5.10.3 Relationship between Exchange Rates and Stock Market Performance

Several interesting studies have investigated the relationship between stock market performance and exchange rates. Empirical studies by Dornbusch (1975) and Boyer (1977) suggested that changes between exchange rates and stock prices related to capital movements. Dimitrova (2005) argued that the relationship between stock prices and exchange rates is important for three reasons: firstly, it may affect decisions about monetary and fiscal policy; secondly, the link between the two markets may be used to predict exchange rate stability; thirdly, there is a link between currency rates and other assets in investment fund portfolios. According to Dornbusch and Fisher (1980), classical economic theory suggests the relationship between exchange rate behaviour and stock market performance as a "flow oriented" example for that model of exchange determination. Bahmani-Oskooee and Sohrabian (1992) investigated the relationship between stock prices and exchange rates in the US economy. They established that there is no relationship between these variables in the long-term. Ratner (1993) utilised monthly data from March 1973 to December 1989 to examine whether US dollar exchange rates affected US stock prices. His findings indicated that, in the long-term, there was no link between the US stock index and foreign exchange rates.

Kasman (2003) examined the relationship between stock prices and exchange rates in one emerging economy, the Turkish market. He discovered that there was a stable relationship

CHAPTER 5: EMERGING STOCK MARKETS

between stock prices and exchange rates in the long-term. The results additionally indicated that the causal relationship existed only from the exchange rate to industry sector index, which, in turn, affects currency. In general terms, an empirical finding suggests that there is no relationship between these three financial variables of exchange rates, stock prices and returns in most countries in long-term equilibrium via international investments or capital inflows and outflows. Many other studies have indicated that there is a statistically significant relationship in the short-term. The underlying reason for this may be found in the early empirical studies focusing on the short-term impact, where stock markets may take lengthy periods to respond to exchange rate modifications. Clearly, as detailed in the previous discussion, many studies have determined that these variables have some "*predictive ability*" for each other and the direction of causality seems to depend on the specific characteristics of the nation analysed.

5.10.4 Relationship between Budget Deficits and Stock Market Performance

Many theoretical and empirical studies have tested the link between monetary and fiscal policy and the nature of its impact on stock prices. On the whole, there are mixed results. In theoretical studies Tobin (1969), Blanchard (1981) and Shah (1984) contended that fiscal policy bears an impact upon stock prices. Certain other studies, such as those by Modigliani (1961) and Blinder and Solow (1973), have proposed that the level of government expenditure will raise interest rates and reduce private investment. Studies conducted by Dewald (1982), Dwyer (1982), Evans (1985) and Hoelscher (1986) claim that interest rates are unaffected by budget deficit size. For instance, Hoelscher (1986) investigated the relationship between budget deficit and long-term interest rates. Using annual data from 1953 to 1984, the study found a strong significant relationship between budget deficit and interest rates. Darrat (1988) investigated the fundamental effect of the relationship between the stock market and the budget deficit of quarterly data from the Canadian market covering the period 1960 to 1984, adopting a technique suggested by Granger (1980). He reached the found evidence that a significant government budget deficit is a determinant of stock prices.

In order to assess economic performance, rates of imports, exports, public spending, degree of privatisation and investment the overall budget should be considered. In cases of budget surplus or deficit this is a crucial index of the rate of growth. The Libyan economy solely depends upon the oil sector, whilst imports are diversified between all manner of consumer goods and technical equipment. This dependence has led to a budget deficit which is, and has been, a major weakness which the Libyan government has had to tackle. A reduction in the

overall budget deficit could be achieved either by increasing revenues or by reducing expenditure. Direct methods would include privatisation to reduce public spending, imposing taxes on international trade and raising exchange rates.

5.10.5 Relationship between Per-capita Income and Stock Market Performance

A number of theoretical studies have begun to show links between financial markets and rates of economic growth and it is proposed that higher per-capita incomes may affect many aspects of the economy and stock market performance. Gurley and Shaw (1955, 1960 and 1967) argued that financial development is a positive function of real income and wealth. This study supports the quantitative work of Goldsmith (1969), who discovered that, in most of the 35 developed and developing countries investigated, the ratio of the market size to GDP tends to increase with higher real income and wealth. This relationship between growth and the size of the stock market is further supported by more recent evidence from the World Bank (1989). Much of the empirical research shows that finance is strongly associated with the economic growth rate. In addition, it can be contested that there is a positive relationship between percapita income and stock market performance; this relationship is examined within this research.

5.10.6 Relationship between GDP Growth Rate and Stock Market Performance

As mentioned in Chapter Four, the theory of finance shows that a high rate of growth in output affects the average real rate of capital return. It also encourages capital expenditure, as high capital expenditure will boost stock markets. Empirical studies by authors such as Carroll and Weil (1993), Weil (1993) and Edwards (1995) have traced a positive relationship between savings and economic growth. De Long and Summers (1991, 1993) utilised cross-country regressions for eighty-eight countries over the period 1960-1985. They discovered evidence of a significant relationship between high levels of physical investment and high levels of GDP growth per-works. A number of subsequent studies have adopted the growth regression framework related to stock market performance (see for example, Dowrick, 1992; King and Levine, 1993a; Atje and Jovanovic, 1993; Levine and Zervose (1995) used cross-country regressions for 49 countries and identified a significant correlation between GDP growth rate and stock market development. Using data from 40 countries and a simple cross-country regression, Ajie and Jovanovic (1993) discovered a significant relationship between growth rates over the period 1980-1988 and the value of stock market trading divided by GDP.

5.11 SUMMARY AND CONCLUSION

This chapter has considered theoretical perspectives on the importance of well-functioning stock markets for economic growth, since the study examines the effect of both stock market development and economic reform on economic growth. It has identified and discussed the main functions of the stock market within the economy. Additionally, it has analysed trends and quantitative measures of financial development in general and stock markets in particular, including the market size as measured by market capitalisation, market activity by value traded, and turnover ratio for stock market liquidity. The global financial crisis that started in advanced economies spreading to emerging markets and low-income countries has been reviewed, as have various empirical studies which have examined the relationship between economic indicators and stock market indicators. This empirical study proposed that an economic reform programme, in terms of deregulation, privatisation and liberalisation, has a positive significant effect upon stock market performance which increases market size, market liquidity, market activity and market concentration. Additionally, empirical studies illustrate that there is a positive relationship between macro-economics in connection to per-capita income and GDP growth and stock market performance with impact of changes in exchange rates on stock prices and returns which differ from one economy to another and from one industrial economy to another. However, the empirical studies revealed that there is a negative relationship between other macro-economics reform vis a vis interest rates, inflation rates and budget deficit on stock market performances related to stock prices and returns.

In Chapters Three, Four and Five a theoretical foundation upon which the research is based has been provided. The research objectives have also been identified. In order to collect the data required to answer the research objectives, an appropriate methodology needs to be developed, consequently, the overall research process, from its theoretical underpinnings to the collection and analysis of the data, is discussed in the next chapter.

CHAPTER SIX RESEARCH METHODOLOGY

6.1 INTRODUCTION

The discussion in Chapter Two highlights the major changes in the Libyan economy since the independence of the country in 1951, providing evidence that, since the late 1980s and early 1990s, this economy has been subjected to a considerable shift from a centrally planned to a market economy system. Moreover, the survey of the literature in Chapters Three, Four and Five have provided an in-depth theoretical background relevant to the research problem, thereby offering a general overview of the literature and assisting in providing a framework for the theoretical and empirical study. With this knowledge, and an appreciation of the results from previous research studies pertinent to the general problem, an appropriate research methodology and procedure to investigate the problem of this thesis has been formulated. The aim of this chapter is to explain and justify the methods used in this study of economic reform and stock market performance in Libya, from which it will be possible to generate empirical evidence and assess the volatility and reliability of findings. Figure 6.1 illustrates the research design whilst Appendix B (Figures 1 and 2) shows a more detailed conceptual framework of the research.

The research philosophy and design applied in this research was a multi-method/mixed triangulation approach, since both quantitative and qualitative approaches are combined. The logic behind this is that it will allow more general conclusions to be drawn. Quantitative data was acquired through a questionnaire and qualitative data was acquired through semi-structured interviews. This approach would find favour with authors such as: Gacdamer (1976); Howe (1985); Cohen (1992); Gresswell (2003); Leedy and Omrod (2005); Miles and Huberman (1994); Remenyi et al. (1998); King (2004); King (2006) and Saunders et al. (2007). This multi-method approach provides the level of detail needed to both explore the economic reform programme and stock market performance and allow a judgement to be made on whether it assists in developing Libyan stock market performance. With regard to ethical issues, many respondents were concerned about matters of confidentiality and anonymity. It was made very clear to all interviewees that any names and places would remain totally confidential and anonymous and would be omitted from any published data. It is for this reason that the researcher guaranteed respondent confidentiality, which was a promise that the respondent would not be identified or presented in an identifiable form and anonymity

122

which is a promise that even the researcher will not be able to tell which responses came from which respondent (Robson, 2002).



Figure 6.1 Major Steps Influencing Research Design and Process



The chapter commences with Section 6.2, which highlights the research aim and objectives. This is succeeded by a brief case study justification adopted in Section 6.3. Section 6.4 offers an overview of the process of choosing an appropriate research philosophy. This consideration is followed by a detailed justification for the chosen research philosophy in Section 6.5. Section 6.6 presents the research framework. Section 6.7 discusses the research design-survey strategies. Section 6.8 presents methods of data collection. Section 6.9 focuses upon the validity and reliability evaluation. Section 6.10 focuses on the data analysis statistics. Finally, in Section 6.11, a summary and conclusion of this chapter is provided.

6.2 RESEARCH AIM AND OBJECTIVES

Methodology refers to "how research should be undertaken, including the theoretical and philosophical assumptions upon which research is based and the implications of these for the method or methods adopted" (Saunders et al., 2007: 481). It is concerned with the overall research process, from the theoretical underpinning to the collection and analysis of the data (Collis and Hussey, 2003). Therefore, researchers should be careful when they choose the methodology which must be appropriate for the researcher's objectives and able to answer the relevant research questions. Consequently, in order to choose an appropriate methodology, the research aim and objectives should be identified. The main aim of conducting this research is to determine the most appropriate model for the continued viability of a stock market mechanism in Libya and to consider an appropriate strategy for the Central Bank of Libya to undertake successful continuation of the stock market development. To achieve this, the following research objectives were formulated.

- **Objective One:** To assess the success of the economic reform programme in Libya, specifically the deregulation, corporatisation, privatisation and liberalisation that has led to initiation of the Libyan stock exchange.
- **Objective Two:** To identify the benefits of a stock market to the Libyan economy. This involves examining the relationship between macro-economic reform variables and stock market performance variables.
- **Objective Three:** To assess the performance of market-based economies, with particular reference to the emerging economy of Libya, and to evaluate current and best practice in financial deregulation.
- **Objective Four:** To determine whether best practice from other emerging stock markets is transferable to the Libyan situation and context.

6.3 CASE STUDY AND JUSTIFICATIONS FOR ITS USE

In respect of case study research, Yin (1989: 23) defines this method as "an empirical inquiry that investigates a contemporary situation within its real-life context when boundaries between phenomenon and context are not clearly evident and in which multiple sources of evidence are used". As such, combining the questionnaire survey and the case study approaches may help to better understand the economic reform programme and stock market development in Libya. Furthermore, Yin (1994) argues that the ability of surveys to investigate the context is extremely limited, which can be compensated for using the case study method. In contrast, the case study methodology is often criticised for relying on a single case, making it incapable of providing a general conclusion (Tellis, 1997). The latter incorporates description, illustration and explanation as it thoroughly investigates the past and present situation regarding the economic development in order to expand what can be learned in the period of time available for the study (Tellis, 1997). To achieve the research aim and objectives herein, the overall approach will be that of a case study focusing on the Libyan economic reform programme and stock market performance. This is in order to develop the research framework which is used to analyse and interpret the data collected from a quantitative approach and to investigate the relationships between economic reform factors and stock market performance factors. As mentioned in Chapters One and Two, the main factors of the Libyan economic reform programme and stock market performance include: oil sector dependency as a major source of income and foreign exchange earnings; limited stock market data; the freezing of UN sanctions in 1992. Since then, Libya has been gradually implementing measures to reform and open its economy. But, it is only following the lifting of UN sanctions in 1999 and all US Libya-specific trade sanctions in September 2003 and 2004, respectively, that the pace of reform has accelerated somewhat, with the implementation of measures aimed at enhancing the role of the liberalisation stability and private sector in the economy.

To reach objective four it was necessary to review studies of stock market determinates in emerging developing countries (see Subsection 6.6.3) to determine the issues used to explain financing behaviour and to ascertain which of these have an impact upon an economic reform programme and stock market as major factors in economic growth rate. As indicated by researchers (Michaelas, 1998 and Creswell, 2003), the mixture of different methods of data collection and analysis in stock market studies provides a "*triangulation*" approach, as outlined in Section 6.4. Cresswell (2003) argues that a rich research context requires more than one data collection method; the present approach satisfies that requirement since it

includes quantitative data from questionnaires and interviews as well as introduction on 42 emerging economies for comparison to Libya. Therefore, this approach was adopted as this study's major method because it has the potential to guarantee collection of sufficient information to allow the study's aim and objectives to be achieved.

6.4 RESEARCH PHILOSOPHY

Research philosophy refers to the procedural framework within which the chosen methodology (e.g. case study, survey, experiment and grounded theory) is located. It is the particular paradigm (e.g. the view or approach) that a researcher adheres to, when conducting his or her study (Easterby-Smith et al., 2002; Cresswell, 2003; Leedy and Omrod, 2005). The two paradigms that dominate social sciences literature in general, and business management in particular, are 'positivism' and 'phenomenology' (Hussey and Hussey, 2003), and each derives from the way the researcher considers the development of knowledge (Saunders et al., 2007). It is necessary to take a philosophical position when deciding on what approach to adopt in research because, as Easterby-Smith et al. (2002: 27) point out: "There are at least three reasons why an understanding of philosophical issues is very useful. First, because it can help to clarify research designs. Second, knowledge of philosophy can help the researcher identify, and even create, designs that may be outside his or her past experience".

Positivist (Quantitative) Paradigm	Phenomenological (Qualitative) Paradigm					
Basic Beliefs						
• The world is external and objective	• The world is socially constructed and subjective					
Observer is independent	• Observer is part of what is observed					
Researcher Should						
• Focus on facts	 Focus on meanings 					
• Look for causality	 Try to understand what is happening 					
 Reduce phenomena to simplest elements 	• Look at the totality of each situation					
• Formulate hypotheses then test them	 Develop ideas through inducting from data 					
Preferred	Methods Include					
• Operationalising concepts so they can be measured	• Using multiple methods to establish different views of phenomena					
• Taking large samples	 Investigating samples in-depth or over time 					
• The location is artificial	• The location is natural					
Reliability is high	• Reliability is low					
Validity is low	 Validity is high 					
• Generalises from sample to population	• Generalises from one setting to another					

Table 6.1 Difference between Positivist and Phenomenological Philosophies

Source: Gaber (1993); Easterby-Smith et al. (2002); Hussey and Hussey (2003).

As Table 6.1 demonstrates, the positivistic paradigm seeks to develop a theory and hypothesis (or hypotheses) about the relationship between two or more variables to be measured

CHAPTER 6: RESEARCH METHODOLOGY

quantitatively from available data, which is then tested empirically by analysing data concerning the relevant variables, using statistical tests to identify significant results. Findings confirm the theory or relationship between variables in the modification of the theory (Hussey and Hussey, 2003; Saunders et al., 2007). In investigating a phenomenon, the positivist approach takes an ontological assumption of 'reality' as being external and objective (Easterby-Smith et al., 2002). It assumes that a given phenomenon will happen regardless of who is observing and interpreting it. Hussey and Hussey (2003) explain that positivism relies on the belief that the study of human behaviour should be conducted in the same manner as studies undertaken in natural sciences, to establish cause and effect links between variables. On the other hand, the phenomenological paradigm is characterised by a focus on meanings that research subjects attach to social phenomena. It does not consider 'reality' as objective, but rather as being socially-constructed, with its meaning supplied by those involved. Phenomenological researchers attempt to understand what is happening and why. This involves focusing on the way (why and how) people view different experiences, rather than searching for external causes and fundamental laws to explain their behaviour (Leedy and Ormrod, 2005).

6.5 JUSTIFICATION FOR THE RESEARCH PHILOSOPHY

The choice of research methodology in social research has become a debated and problematic issue because it depends on the research questions, objectives and research philosophy; neither approach is appropriate or inappropriate in itself until it is applied to a specific problem. However, the positivistic paradigm is the most commonly-adopted philosophy in business research, as noted by Hussey and Hussey (2003: 73): "It can be argued that the dominant paradigm in business research is the positivistic paradigm. If this is acceptable in your discipline, and to your supervisor, you will not need to expend much energy in justifying the methodology you adopt for your research. If you decide on a phenomenological approach, you may have to spend more time expanding and justifying your methodology." The justification of the research philosophy lies in the following issues.

- a. It is the most appropriate when considering the nature of the research questions and objectives (Punch, 2005).
- b. It is a design and approach that has been used by other Arab researchers who have made progress through various studies in contributing to the literature related to business and management in Arab cultures (e.g. Abdalla and Al-Homoud, 1995; Agnaia, 1996;

Albahussain, 2000; Al-Athem, 2000). These researchers found that a multi-method approach combining questionnaires and interviews are the most popular and acceptable research methods in Arab environments. Since this research was conducted in Libya, application of the same research design employed by other Arab researchers strengthens the study's consistency, validity and reliability.

- c. Time available to the researcher is a determining factor, because the deductive approach can be quicker and a lower-risk strategy than the inductive approach (Creswell, 2003).
- d. The nature of the research questions and objectives is such that many subjective variables are involved which need to be investigated and measured by both quantitative and qualitative measures. Statistical analysis of quantitative data collected will allow for comparisons and generalisations to be relative and accurate, while qualitative data will provide a forum for elaboration, explanation and description of events, actions, attitudes, and behaviour of the subjects being investigated (Bryman, 1993).
- e. Analysts of financial structure models should incorporate strategic management preference, decision sciences and social psychology to a conceptual model to achieve a better understanding of capital structure policy (Matthews et al., 1994; Barton and Gordon, 1987). In this context, Rajan and Zingales (1996) suggest that future research should proceed in different ways: continuing to develop the relationship between both theoretical model and empirical findings by using models in diverse positions.
- f. The combination of questionnaire survey and technical models can provide a significant contribution to clarify and comprehend the economic reform in terms of stock market performance in emerging countries. Collection and empirical analysis in this research adopted a survey questionnaire and a regression analysis model for a systematic comparison of other emerging economies.

6.6 RESEARCH FRAMEWORK

The aim of this section is to develop the research framework theoretically and empirically and, thus, investigate relationships between economic reform factors and stock market performance factors. This framework is designed to test research hypotheses to achieve the research objectives. It is largely based upon key findings from a literature review of an economic reform programme, a macro-economic reform (Chapter Three), economic and financial structure theory (Chapter Four), emerging stock markets (Chapter Five). Therefore, this study applies quantitative and qualitative approaches for the purpose of gaining a more comprehensive picture of the issues in question. It is valuable for the researcher to employ

CHAPTER 6: RESEARCH METHODOLOGY

multiple methods for primary data collection since this provides for more scientific rigour and allows for the first three research objectives to be pursued effectively. The quantitative approach was adopted in order to examine the hypotheses by utilising data extracted from survey questionnaires that were forwarded to Libyan financial institutions and companies in order to test these hypotheses using the liner regression model (Chapter Eight). In this regard, due to the fact that the economic reform programme and financial market in Libya requires a body of empirical knowledge, this stage attempts to further that knowledge by providing a comparison with emerging market countries in order to put the Libyan financial market into perspective. The empirical analysis in this study uses a systematic combination of regression analysis models and econometric techniques. Consequently, the fourth objective was formulated in the context of data, obtained from Datastream, on 42 emerging markets and were utilised in an attempt to analyse and, where possible, to understand and explain the financing behaviour of the Libyan financial market and to analyse the differences, if any, of financing patterns between Libya and the other 42 emerging market countries (Chapter Nine). This was done in order to test the fourth hypothesis using the econometrics model. The quantitative approach is described as an extensive approach, which employs less detail and more generality; it is believed that the quantitative approach is an appropriate methodology to link the models applied in this research area.

Four hypotheses have been formulated based on the research framework to be tested (see Figures 6.3 and 6.4 below). The primary sources of building or designing the research framework were undertaken.

• The theoretical review of literature on the research topics which provided a theoretical background of research factors or variables. This network of relationships is known as a theoretical framework (Sakaran, 2003).

The extracted variables in this study, as depicted in Figure 6.3, and tested in Chapter Eight are:

- Dependent variable: stock market performance;
- Independent variables: economic reform programme, macro-economic and financial maturity.

6.6.1 Building the Research Framework

Building the research framework is concerned with detailed clarification of literature review arguments and justifications that are used to construct it. These arguments and justifications are major findings that represent a research framework based on the following research factors.

- The first part consists of economic reform programme factors including deregulation, corporatisation, privatisation and liberalisation. Theoretical studies in terms of these factors or variables were investigated by several researchers (e.g. Fisher and Gelp, 1991; Clague and Rausserc, 1993; Laban and Sturzenegger, 1994; Lissowska and Swaan, 1996; Easterly and Bruno, 1996; Fisher et al., 1996; Stulz, 1997; Kim and Singal, 2000; Williamson, 2004; Gwarnt and Lawson, 2005). Empirical work has determined the importance of the economic reform programme has options to alter the transformation of any economy from one that is closed or socialist to one which is an open-door policy or capitalist economy (see, Cho, 1986; Zingales, 1998; Demirgüc-Kunt et al., 2004; Baltagi et al., 2006).
- The second part is concerned with macro-economic reform factors. These include six components (interest rates, exchange rates, inflation, budget deficit, per-capita income and real GDP growth rate). The review of empirical studies revealed that macro-economic factors differ from one economy to another (De Melo et al., 1997; Krueger and Ciolko, 1998; Lavigne, 1999). Other studies have further results (e.g. Wolf, 1997; Berg et al., 1999; Fisher and Sahay, 2000; Kasman, 2003; Omran, 2004; Bolbol and Omran, 2005).
- The third part relates to financial maturity factors. These are concerned with testing bankbased aspects such as banking assets, domestic credit and stock market capitalisation divided by GDP. It is a simple matter that financial systems provide a foundation where GDP of respective companies/countries increases or decreases or even remains stable. Levine (1997), Beck et al. (1999b) and Levine et al. (2000) used the ratio of total bank assets to GDP which are relative to the economy size and value of credit provided by banking sector to private sector divided by GDP. Whilst some authors stress the significant relationship between banks and stock markets (Stiglitz, 1985; Boyd and Prescott, 1986; Bhide, 1993), other empirical studies consider the impact of banks and stock market development on economic growth. These include Atje and Jovanovic (1993), Jappelli and Pagano (1994), Harris (1997), Levine (1997), Levine and Zervos (1998), Boyd and Smith (1998), Huybens and Smith (1999), Rousseau and Wachtel (2000), Demirgüc-Kunt and Levine (2001), Levine (2001), Bekaert et al. (2002), Beck and Levine (2002). Dow and Gorton (1979), Stiglitz (1989), Collier and Maryer (1989) and Cobham (1995) who argued from a contrasting perspective and supported banks rather than stock markets, particularly with regard to the situation in developing countries.

• The fourth part concerns stock market performance factors. These include market activity, size, liquidity and concentration. Following early research by Robinson (1952), Gurley and Shaw (1955), Fisher (1933), Goldsmith (1969), McKinnon (1973) and Shaw (1973) discovered development of financial markets has been significantly correlated within the level of rate of economic growth. Several theoretical and empirical studies by Greenwood and Jovanovic (1990), Berthelemy and Varoudakis (1996), Levine (1991), Kunt (1992), Atje and Jovanovic (1993), Levine and Zervos (1995), Kunt and Levine (1996) Harris (1997), Bossone (2000) and Currie (2003), among others, have contributed to the realisation that stock market development affects economic growth in a developing nation. For instance, Greenwood and Smith (1997) demonstrate that large capitalised stock markets can facilitate investments in the most productive technology projects by lowering the cost of mobilising and savings. Similarly, Levine (1991), Bencivenga and Smith (1991), Diamond (1996) and Fulghieri and Rovelli (1998) supported the notion that stock market liquidity is critical for such growth to occur.

Figure 6.2 illustrates the research's theoretical framework developed to test the relationships between economic reform programme, macro-economic reform, financial maturity and stock market performance factors.



Figure 6.2 Conceptual Framework for the Impact of Stock Market Performance

Source: Developed for this research Subsection 6.6.1: Building the Research Framework.

6.6.2 The Primary Research Model

This section provides a conceptual framework of this research based on a linear regression model to test the first three of the research objectives (Section 6.2 and Chapter Eight: Figure 8.2), which can be used for identification and integration to represent potential factors of macro-economic reform, which are treated as independent variables when measuring their effect on stock market performance. The factors of stock market performance, in turn, are treated as dependent. Performance factors of a stock market, therefore, depend heavily on the success of the economic reform programme's factors. When stock market factors experience a significant increase, stock market performance becomes more effective. As has already been shown, a well-performing stock market sustains developing economic growth (see Subsection 6.6.1). The factors of financial maturity act independently by linking to each other. These are considered to be stock market performance. Banks and corporate financing were considered as decision-makers for understanding the relationship between bank and market-based systems. Figure 6.3 displays the research model for the impact of stock market performance over an economic reform programme, macro-economic and financial maturity. This model is adopted to develop a framework that can test factors or variables of independent variables.



Figure 6.3 The Primary Model for the Impact of Stock Market Performance

Source: Developed for this research from Subsection 6.6.2: The Primary Research Model.

6.6.3 Background Statistical Technical Growth Model

This section provides a conceptual framework of the current research based on an econometric model in order to investigate the fourth research objective (Section 6.2 and

Chapter Nine: Figure 9.2) between emerging market economies²⁵ and the Libyan stock market to display diverse financing behaviour from that of other emerging markets included in the sample. Descriptive statistics are techniques to describe the sample distribution and enable a researcher to decide the appropriate analytical statistical method that may be used to demonstrate the extent of adoption of the growth model to indicate performance variables, such as original income level, population growth, education, investment levels, trade openness (exports and imports divided by GDP), political stability and economic freedoms, inflation rate, banking development indicators and stock market development indicators (see Chapter Nine). This totalled 42 emerging developing market countries: Algeria, Argentina, Bahrain, Bangladesh, Brazil, Bulgaria, Chile, China, Czech Republic, Egypt, Fiji, Hungary, India, Indonesia, Iran, Israel, Jordan, Kuwait, Lebanon, Malaysia, Mexico, Morocco, Nigeria, Oman, Pakistan, Philippines, Poland, Qatar, Romania, Russia, Saudi Arabia, South Africa, Sudan, Syria, Thailand, Tunisia, Turkey, United Arab Emirates, Venezuela, Yemen, Zimbabwe plus Libya. These nations were used to examine the financing model in place among these emerging developing nations (see Figure 6.4).

Figure 6.4 Theoretical Model from other Emerging Markets



Source: Developed for this research Subsection 6.6.3: Background Statistical Technical Model Cross-country Comparison.

Booth et al. (2001) explained that developing countries have many institutional similarities to developed countries (e.g. existence of financial market). In this context, it is crucial to study the stock market within the Libyan business environment. However, the Libyan economy differs from other emerging developing countries included in the sample, as its stock market is relatively new in this locus. Libya further differs in terms of economic reform, regulation ownership, companies and in terms of its macro-economic conditions which may differ significantly from other countries in terms of GDP growth rates, exchange rates, interest rates, inflation rates and in culture and corporate governance. Moreover, decline in oil prices had a vital impact upon the Libyan economy and its financing policy, affected by UN sanctions in

²⁵ Kim and Singal (2000) explained that emerging stock markets are defined by the International Finance Corporation as consisting of stock markets in developing countries (low-and middle-income economies).

the early 1990s²⁶ (see Appendix A for more details). Glen and Singh (2003) argue that comparison of financing patterns between countries is extremely valuable. They added that, as the economic reform programme in developing countries is on the national and international agenda, the comparison of financing patterns between countries might provide empirical guidelines to execute such economic reform programmes. Accordingly, the Libyan economic reform programme, in fact, requires a body of empirical knowledge which may pay greater attention to further that knowledge by providing a comparison with emerging market countries to achieve stock market performance to develop Libyan business behaviour. For that reason, this framework is used to analyse and interpret the data collected from these countries under investigation (see Chapter Nine).

6.6.4 Background Quantitative and Qualitative Data Analysis

The third philosophy is known as triangulation (other names: mixed and multi-method approach) between quantitative and qualitative based on triangulating methods of primary data collection. The use of multi-method (questionnaires and interviews) is based on the assumption that quantitative and qualitative methods are complementary rather than competitive (Saunders et al., 2007). Adcock et al. (1998) identified that the approach represents less than 25 per cent of research in marketing and more than 75 per cent of all marketing research data are provided as quantitative research. Based on the content analysis for research published in the five fields of social since during period of 1994-2003, Bryman (2006) found that 57.3 per cent of all articles published during that period employed a combination of a survey of instrument techniques and qualitative interviewing techniques. Another study by Mason (2006) indicated that, while the qualitative research techniques are able to answer the questions of "how and why" related to social phenomena, the analytical tools of quantitative research techniques are able to numerically demonstrate wide relationships of patterns and changes in social phenomena. This research has predominantly employed the quantitative approach by conducting a questionnaire survey, and less dominantly employed the qualitative approach by conducting a number of in-depth interviews as a second primary data collection (see Section 6.8). Even so, the preferred research approach according to Saunders et al. (2007) depends on the research subject, the time available to the researcher and audience favourites. The wealth of available literature assists researchers in developing

²⁶ US sanctions were introduced in 1986, prohibiting US companies from any trade or financial dealings with Libya, while freezing Libyan assets in the US. UN sanctions were imposed in 1992, suspended in 1999 and finally lifted in 2003. US sanctions were lifted in 2004.

CHAPTER 6: RESEARCH METHODOLOGY

theoretical frameworks and hypotheses leading to the adoption of the deductive approach, whilst a new subject with little extant literature results in adopting the inductive approach.

The aforementioned authors found a multi-method approach combining questionnaire survey and semi-structured interview technique to be the most applicable and acceptable research methods within Arab organisations. This approach was adopted by other Arabic researchers who have conducted studies related to business research studying in disparate content areas (e.g. Al-Faleh, 1987; Abdalla and Al-Homoud, 1995; Agnaia, 1996; Albahussain, 2000; Al-Athem, 2000; Ahmad, 2006). Applying the same research techniques adopted by such research to identify all the problems and challenges that might face this research's objectives and current knowledge of the topic under investigation led to this research study adopting a mixedmethod approach to research design and data collection activities. This combines both quantitative and qualitative methods, within the overall framework of a single case study (that of Libya) that depends upon collection of both primary and secondary data. However, within that design, a series of smaller case studies will be conducted on countries' progress since implementing stock exchanges, and of organisations in Libya, thereby providing a comprehensive data bundle. Based on this debate, the rationale behind choosing a multimethod approach to research is now given.

- The choice of multi-methods of data collection enables triangulation to take place (Leedy and Ormrod, 2005). Triangulation, as defined by Saunders et al. (2007), is *"the use of two or more independent sources of data or data collection methods within one study in order to help ensure that data are telling you what you think they are telling you"*. Such triangulation, as suggested by Creswell (2003), can achieve three advantages: it provides a kind of convergence of findings; it provides complementary between facts; it adds scope and breadth to a study.
- Different methods can be used for different purposes in a study (Hair et al., 2005) as a consequence; the choice of multi-method data collection follows directly from the research objectives which are also consistent with the research approach and strategy. Describing the stock market in the Libyan economy requires a questionnaire, while understanding practices requires flexible personal interviews with a limited number of respondents.
- Using of multi-method (questionnaires and interviews) in this research is due to the lack of availability of such high-quality databases published, which might constitute the major barrier to conducting a study of stock market performance within the Libyan economy. Data, then, gathered directly from the questionnaire survey responses to test the research

hypotheses (Chapter Eight). They also refer to a means of objective related to the degree and character of the relationship between dependent and independent variables.

6.7 RESEARCH DESIGN - SURVEY STRATEGY

Research design is a basic plan or strategy to be adopted for a particular study that will make it possible and legitimate to draw more general conclusions from the study (Oppenheim, 2000). Thus, the research design indicates how a research sample will be drawn, what sub-groups will be formed, what comparisons will be made, whether or not there is need to control research groups, what variables must be measured and how these measures will be related to external events. Remenyi et al. (1998) note that the survey is a research strategy involving collection of data from a large number of people or a sizeable population. In general, the main criteria adopted when selecting a particular survey strategy are determined by research approach (Saunders et al., 2007) which depends, in turn, on the study's objectives. Figure 6.5 illustrates that research typology contains quantitative methods measured as experimental and non-experimental, mixed method and model research, whilst qualitative research is measured as phenomenology, ethnography, case study, historical and grounded theory (Pamerino, 1999).



Source: Palmerino (1999).

The main criteria in choosing a particular survey strategy is the research approach adopted in undertaking the research (Saunders et al., 2007), which in turn, depends on research aim, objectives and questions. For instance, Hussey and Hussey (2003), adopting the deductive approach, leads the researcher to employ the experimental or survey strategies. In contrast, adopting the inductive approach leads the researcher to employ the strategies of case study, grounded theory and action research. According to the deductive approach, researchers deduce

CHAPTER 6: RESEARCH METHODOLOGY

their studies' hypotheses based on known facts (theories), translate them into operational terms and test them in empirical ways by using statistical methods (Bryman, 2007, Sunders et al., 2007). Consequently, the aims of each approach are also different. The aim of a deductive approach is to involve moving from the general (theoretical position) to the specific (inquiry of the research), while the inductive approach starts from a set of observations to build a theory (Anderson, 2004). Generally, the inductive approach involves moving from the specific to the general. Figure 6.6 shows the two approaches to the relationship between theory and research.



Figure 6.6 Deductive and Inductive Approaches

This research is mainly deductive despite being based on triangulation methods of primary data collection. The rationale of this survey choice is for the following reasons.

- They are usually associated with the deductive approach and the most popular and commonly used strategies in business studies or management research (Saunders et al., 2007).
- The research topic, related to a wealth of literature, helps develop a theoretical framework and hypotheses leading to the adoption of the deductive (quantitative) approach, whereas, when the research topic is new with scant existing literature, it may be more appropriate to generate data and analysis thoroughly to formulate a theory (inductive qualitative approach) (Creswell, 2003).
- This is the most common survey strategy when data must be collected from a sizeable population at a particular time (Leedy and Ormrod, 2005).
- A sample of the findings from the survey can be generalised in order to understand the population better.
- Generalisation: survey-based research (quantitative approach) seeks to identify relationships that are common across organisations and, therefore, provide a general statement upon, or a theory about, the phenomenon being researched (Bryman, 1993 and Eldabi et al., 2002).

Source: Bryman (2004).

• This study can be classified as descriptive and as one involving hypothesis testing, in which relationships between certain variables are to be verified and phenomenological strategies are not appropriate.

6.8 DATA COLLECTION METHODS

Data collection methods are an integral part of research design and depend primarily on the research objectives, approach and strategy (Sekaran, 2003). Data can be gathered from secondary and primary sources, both of which have been extensively used in social and business research. Both methods of data collection were adopted in this study as suggested by Malhotra and Birks (2006), who strongly recommended this strategy, together with the combination of quantitative and qualitative methods, in order to provide opportunity for triangulation where possible. Primary and secondary data sources are now examined in the following subsections.

6.8.1 Secondary Data

Secondary data refers to all sources of information that are available before a research project is undertaken. Sekaran (2003: 63) mentions that: "[...] Secondary data can be extracted from various sources, including books and periodicals, government publications and information sources, the media, census, stock market reports, and mechanised and electronic information of all kinds such as the bar code, scanner data, and the internet. Secondary data can be culled from the historical records of the organisation itself, from information already available on the internet, or from external sources such as the ones mentioned above, either through the internet or otherwise". In this study, secondary data has been obtained from recentlypublished research and articles on the relationship between economic reform and the stock market. This involved a review of the literature and the most recent publications related to economic indicators in general, and in the financial sector in particular. Quite specifically, literature relating to this system's application in Libya has been featured. Data covers the 1970-2007 period, with special focus on the post-sanction economic reform programme (1999-2007) and, because of the availability of data collection sources, annual figures are used; unpublished data, for instance, theses and other relevant material from the General Statistics department in Libya, the General People's Committee Secretary of Economy and Administration Record Management, Tripoli branch. The World Bank and the IMF are, additionally, useful sources of statistics and other data relating not only to the Libyan situation, but to the market and emerging economic indicators besides. Saunders et al. (2007) explained that using secondary data within organisations may additionally have the advantage that, because the information has already been collected, its use provides an unobtrusive measure. This is supported by Cowton (1998) who emphasises the benefits of using secondary data in sensitive situations and refers to this advantage as eavesdropping.

6.8.2 Primary Data

Primary data, on the other hand, is gathered personally from subjects by the researcher; this can be done in various ways utilising different (both quantitative and qualitative) key methods (Silverman, 2001). As indicated previously, this study adopted a mixed-method approach to the research design and data collection activities, combining quantitative and qualitative methods within the overall framework of a single case study (Libya) that depends upon the collection of both primary and secondary data. However, within that design, a series of smaller case studies were conducted of countries' progress since introducing stock exchanges, thereby providing a comprehensive data bundle. A questionnaire survey was used to collect information from all financial sector organisations in Libya, relating to changes in their performance since the introduction of economic reform measures. In addition to the use of the *self-administered questionnaire, semi-structured interviews* are employed. These two methods are empirically examined in the next subsections.

6.8.2.1 Self-Administered Questionnaire

Questionnaire can be defined as "a reformulated written set of questions to which respondents record their answers, usually within rather closely defined alternatives" (Sekaran, 2003: 236). For this research, the use of a questionnaire method follows directly from the research objectives. Hussey and Hussey (2003) argued that the questionnaire is the core method in survey research and serves to ask a list of carefully structured questions, chosen after considerable testing to achieve a reliable response from a selected sample. Bouchard (1979: 269) states: "[...] Methods are means to ends, no more, no less. The key to good research does not only lie in choosing the right method, but rather in asking the right question and picking the most powerful method for answering that particular question. Methods are neither good nor bad, but rather more or less useful for answering particular questions of a particular time and place." The questionnaire survey has the advantage of being cheaper and less timeconsuming than certain other methods (such as interviewing) and also allows completed responses to be collected within a short period, which gives the opportunity to introduce the research topic and motivate respondents to supply answers truthfully. Many researchers (e.g. Oppenheim, 2000; Easterby-Smith et al., 2002; Sekaran, 2003) emphasise that questionnaires are the most popular method of collecting data and can be administered personally, mailed

CHAPTER 6: RESEARCH METHODOLOGY

through the post, or electronically distributed. In this study, the questionnaire method was appropriate considering the research objective since were chosen as the main data collection method. In general, the rationale for using the questionnaire method for this research is now explained.

- According to Sekaran (2003) there are a number of advantages in using self-administered questionnaires: all the completed responses can be collected within a short period of time and a good rapport can be established to motivate respondents; there is an opportunity to introduce the research topic and motivate the respondents to give their answers honestly; any doubt or misunderstanding about the questionnaire will be clarified; it is less expensive when distributed to group of respondents; it gains an almost 100 *per cent* response rate.
- The research objectives are to describe behaviour concerning economic reform and the stock market in Libya and to explain relationships between certain variables. A description of relationships needs the use of a statistical package, in this case the Statistical Package for the Social Sciences (SPSS software). The questionnaire can provide the required information relatively quickly given that the findings from the sample financial institutions can be generalised to all financial institutions. Cohen et al. (2000) explained that questionnaires can be used to determine points and future plans. They can be further adopted to explore effectively the relationships between variables.
- The questionnaire approach has been extensively used in previous research into management and business research at Huddersfield University (i.e. Akous, 2003; Casserly, 2003; Khaldi, 2005; Alnsour, 2006; Al-Hussari, 2006; Ahmad, 2007).
- It was expected that many Libyan financial institutions and companies would respond to the questionnaire because this research would provide one of the first studies of economic reform and performance of stock market trading in the Libyan context.

6.8.2.1.1 Population and Samples

The research population is the entire group of people, events, or things of interest that the researcher wishes to investigate in order to collect the required data for the research problem (Sekaran, 2003). The process of identification of a research population depends on the type of problem being explored and the purpose of the main research objective, besides other major concepts or issues involved (Awda and Malkawi, 1992). As stated earlier, a questionnaire survey was used to collect information from all financial sector organisations in Libya, relating to changes in their performance since the introduction of economic reform measures.

It is appreciated that such a strategy can suffer from a low response rate (Starbuck, 2004) but, as this method is considered the most effective in the circumstances, it was used cautiously by the researcher and, in an attempt to improve the response, the study sample included more than one person in each financial sector. As noted in the Introductory Chapter and Chapter Two, this research concentrates upon five sectors within thirty-four financial and non-financial organisation representing the major part of the Libyan economy in terms of stock market, as reported in Table 6.2.

Code	Groups	Description	Population size	e	Sample size
1	A. Capital Market Libyan Stock Market	A stock market is part of the capital market in that they provide a market for the shares and loan stock that can represent the capital once it has been raised/ a stock market is central to the creation and development of a strong and competitive economy.	145	-	68
2 3 4 5 6	<i>B. Currency Market</i> Central Bank of Libya Commercial Bank Ummah Bank Jamhoriya Bank Commerce and Development Bank	Currency market divided to the central bank principle and independences serve as control on government policy through specific rules leading government borrowing from central bank. Other banks perform general financial activities, such as financing working capital, term finance, import and export activities, and deposit services.	309 289 268 287 308	.05 .05 .05 .05 .05	15 14 13 14 15
7 8 9 10 11 12 13 14 15 16 17 18 19 20 21	C. Private Sector Sahara Bank Wahda Bank Tripoli Al-Ahaly Bank Ahly Souq Alquma Bank Agricultural sector Libyan Insurance Company United Company for Insurance National Company for Mill and Animal Feed The General Company for Chemical Industries Pharmaceuticals and Medical Supplies The Libyan Iron and Steel Company Arab Cement Company Saria investment and commerce bank Public company for Cement Hotels	The part of an economy that is not under government control. Private sector, however, included private companies, such as: banks; savings; loan associates and also services. According to the Libyan economic reform, law no.1 1993 was issued in order to allow the establishment of private banks, also, concerning the engagement in economic activities such as corporations, partnerships, and joint stock companies, in addition to the ownership of micro public economic activities. Followed by law no.9 for 1992 concerning the engagement in economic activities such as corporations, partnerships, and joint stock companies, in addition to the ownership of micro public economic activities. In 2003, the Prime Minister announced a list of 361 firms in a variety of economic sectors.	129 108 89 85 88 128 89 87 109 88 89 107 87 88 88	.05 .05 .05 .05 .05 .05 .05 .05 .05 .05	6 5 4 4 6 4 4 5 4 4 5 4 5 4 5 5
21 22 23 24 25 26 27 28 29 30 31 32 33 34	Telecommunication D. Public Sector Industry (Manufacturing) Construction Wholesale Transportation Services Petroleum E. Financial Institutions Libyan Arab Foreign Bank The Agriculture Bank Savings and Real Estate Bank National Development Bank Pension and Social Security Institution Libyan Arab Company for Foreign Investment	The part of an economy that covers the activities of the government and local authorities. Nonetheless, this includes education, health and social services, local public services, the police, etc Although, the Libyan economy consists of two main sectors, oil and non-oil sectors, in 1992/93 the state began to re-structure major sectors in the economy, several industries were privatised and investment was encouraged. Financial institution: an organisation whose core activity is to provide financial services or advice in relation to financial products. Libyan financial institutions offer long term lending to a wide variety of economic sectors. They finance equity for startups and thus contribute to enlarging business activities within the Libyan economy.	100 108 229 195 188 205 209 249 199 206 189 228 205 222	.05 .05 .05 .05 .05 .05 .05 .05 .05 .05	5 5 11 10 9 10 10 11 9 10 9 11 8 10

Table	6.2	Study	Po	nulation	(Sam	nle	Size) a
Lanc	U.	Study	IU	pulation	(Sam	pic	SILC	,

The total workforce numbered 5778 employees distributed among the sectors. The avoidance of Notice a: sampling error was accomplished by using a stratified sampling technique. The first step in obtaining a representative sample from the sectors was to select all employees in each sector and then to classify them separately. Then in order to acquire a level of confidence of 95 per cent a random sample of 5 per cent was chosen from each sector to represent that sector in the sample group, based on the number of workers in each sector.

Source:

Developed for this research Subsection 6.8.2.1.1: Population and Samples.

CHAPTER 6: RESEARCH METHODOLOGY

Collis and Hussey (2003: 56) defined a sample as "*a subset of population*". Hair et al. (2005) stated that it is very important to determine the appropriate sample size prior to data collection, from either a large (infinite) population or a small (finite) population. They also added that the researcher can determine the sample size by referring to factors such as time available, budget and previous similar studies. The sample of this research can be defined as a sub-set of the population, for instance a collection of population units drawn from a frame or frames (Scherffer et al., 1990). The sample in respect of primary data sources comprised 328 questionnaires (see Subsection 6.8.2.1.3) distributed to the study sample and 203 questionnaires that were returned and usable. Respondents all worked for Libyan financial institutions and companies located in the city of Tripoli²⁷. The Libyan stock market, central bank and manufacturing companies are classified as public sector shareholding entities, including the Association of Banks, finance and insurance companies. Participants were all in management and from various departments (e.g. managing directors, deputies, investment managers, budget, finance and accounts personnel, computer analysts etc).

In the same vein, Malhotra and Birks (2006) pointed out that sample size in business and marketing research projects is influenced and determined by many factors, which can be summarised as follows: the average size of samples in similar studies; the nature of the research (e.g. in qualitative research, the sample size is typically small, meanwhile with quantitative research, large samples are required); the nature of analysis, (if sophisticated analysis of the data using multiariate techniques is required, the sample size should be large); resource constraints, because any marketing research project, money and time are limited. Taking these factors into consideration, Table 6.3 provides an overview of the sample sizes used in previous studies with details of response rates and usable responses received, as related to this research study (see Subsection 6.8.2.1.3). It can be seen that there is no certain sample size in such studies. Lehmann (1989) mentioned that sample sizes, which have been used by others conducting similar studies in the past is an acceptable approach. Generally, however, quantitative methodologies are associated with large sample size to obtain a high degree of accuracy and to ensure that findings are representative of the population being studied and unbiased (Collis and Hussey, 2003; Sekaran, 2003; Saunders et al., 2007).

²⁷ The city of Tripoli is the capital of the country and has most of the headquarters of the financial institutions and companies and a higher population compared to other cities in the country.

Authors	Year	Purpose of Study	Research Approach	Size of Sample	Response
Goode and Moutinho	1995	Banking/UK	Survey/ Questionnaire	451	380 (42 %)
Levesque and McDougall	1996	Retail banking/Canada	Survey/ Questionnaire	400	325 (81 %)
Edris and Almahmeed	1997	Banking/Kuwait	Survey/ Questionnaire	500	304 (61 %)
Daniel	1999	Banking and building societies/UK and Republic of Ireland	Survey/ Questionnaire	44	25 (57 %)
Lassar et al	2000	Banking/USA and South America	Survey/ Questionnaire	300	65 (22 %)
Abdelaziz	2001	Banking/Egypt	Survey/ Questionnaire	300	247 (82 %)
Al-Hussaini	2001	Quality of financial market/Kuwait	Survey/ Questionnaire	400	306 (77 %)
Polatoglu and Ekin	2001	Banking/Turkey	Survey/ Questionnaire	724	114 (15.7 %)
Appiah-Adu et al	2001	Banks and building societies	Survey/ Questionnaire	121	52 (43 %)
Jamal and Naser	2002	Retail banking/USA	Survey/ Questionnaire	200	167 (83.5 %)
Jabnoun and Al- Tamimi	2003	Commercial banks/UAE	Survey/ Questionnaire	800	462 (58 %)
Yavas et al	2004	Banking/Germany	Survey/ Questionnaire	500	226 (45 %)
Akinci et al	2004	Banking/Turkey	Survey/ Questionnaire	1228	140 (11.4 %)
Al-Kizza and Akbar	2005	Impact of business environment/Libya	Survey/ Questionnaire	164	86 (52.4 %)
Al-Hroot.	2007	Destination Marketing/Jordan	Survey/ Questionnaire	1120	168 (15 %)

Table 6.3 Sample Size: Previous Studies ^a

Notice a: Further information relating to these surveys/questionnaires can be found under the author's name in the research reference.

Source: Developed for this research Subsection 6.8.2.1.1: Population and Samples.

6.8.2.1.2 Designing and Developing the Questionnaire

In order to achieve the aims of the questionnaire, several basic issues should be considered. Sudman and Bradburn (1983) argue that a well-designed questionnaire can make the tasks of both participants and researchers easier and reduces errors. In this regards, a questionnaire should be done with the aim of providing answers to certain questions that will allow the stipulated research objectives to be achieved. A questionnaire with good flow is easier to use; it motivates respondents to progress through it and assists them to remember and provide accurate information (Mangipne, 1995). Several researchers (e.g. Oppenheim, 2000; Aaker et al., 2001 and Saunders et al., 2007) have provided guidelines in respect of questionnaire design, but the final approach and sequencing of questions should be determined not only by the research objectives, as mentioned already, but also by pilot work results (Oppenheim, 2000). Therefore, the following steps were taken.

6.8.2.1.2.1 Planning what to Measure

Following Aaker et al. (2001), questions were designed to achieve research objectives and to enable validation of answers obtained. Firstly, objectives were clarified to keep them firmly in
mind and to be achieved. Secondly, questions relating to research variables were developed in order to provide sufficient material for analysis. Thirdly, questions relating to the economic reform and stock market were designed to obtain general information and, finally, testing of preliminary versions was undertaken.

6.8.2.1.2.2 Formatting the Questions

This study's questionnaire utilised four scales: the nominal scale, used to obtain information about respondents and their companies according to questions in the questionnaire, relating to demographic data, such as type of financial market, banking sector and share ownership patterns; the ordinal scale which was employed to analyse some demographic data, such as the level of institutionalisation; the interval scale which was used to measure managerial perspectives of sources and information; the rating scale, which was used to obtain such data for this research. Given that these scales do have some disadvantages, the rationale for using them is threefold.

- In this research it is necessary to use a variety of statistical techniques and conduct powerful statistical analysis (e.g. using linear regression and econometric techniques). Attitude-scaling is the most widely used technique in such business research.
- It is easy for respondents to comment on such scales as they are simple.
- Rating scales allow respondents some degree of flexibility to reflect intensity of their views, thinking, feelings and understanding.

For this study, it was decided to adopt a five-point Likert scale throughout the questionnaire in order to indicate the degree of agreement or disagreement with each statement included in the questionnaire, in which the range was from 1 "strongly disagree" to 5 "strongly agree".

6.8.2.1.2.3 Question Wording and Layout Decisions

Question design is a creative process that should be linked to a study's conceptual framework (Oppenheim, 2000). Actual wording is one of the most critical aspects of questionnaire design, requiring a clear and unambiguous language of communication to be established with respondents to collect data. According to Aaker et al. (2001), the wording of particular questions can bear a considerable influence on how a research respondent interprets them. Additionally, question layout is of paramount importance, especially when dealing with a self-completion questionnaire. Oppenheim (2000) explained that the layout should, in addition, be attractive to encourage respondents and provide convenience for the data processors who have

to enter data on their computers. Several recommendations regarding choice of layout and wording of questions have been offered by Oppenheim (2000), Easterby-Smith et al. (2002), Sekaran (2003) and Saunders et al. (2007). All these recommendations were taken into account in the design of this study's questionnaire (see Appendix C and D).

6.8.2.1.2.4 Translation of the Research Questionnaire

Although English is the official language of business and commerce in Libya, and widely spoken in business sectors, especially in mixed companies (e.g. oil and foreign banking), it was anticipated that respondents would feel more comfortable using their native language. Hence, the questionnaire, whilst originally in English in the UK (see Appendix G), was translated into Arabic in Libya to render the questions very clear and avoid the possibility of misunderstandings. It was decided to distribute the questionnaire in both English and Arabic; the reasons being that some respondents prefer to answer the English copy. Malhotra and Birks (2006) proposed methods for translating a research questionnaire for use in an international research context. The method of back translation consists of the questionnaire being translated again by a second bi-lingual person from the target language into the original language in order to remove any errors or misinterpretation. In the current study back translation was used but, in addition, the final copy of the questionnaire in Arabic was sent to an Arabic language expert who checked grammar and wording in order to ensure that the final version was clear and understandable.

6.8.2.1.2.5 Pilot Testing the Questionnaire

Pre-testing the questionnaire prior to data collection is a vitally important step in research since the process has the practical benefits of ensuring that the final version contains questions that are specific, understandable and capable of obtaining a response from those answering (Oppenheim, 2000 and Saunders et al., 2007). In this respect, Malhotra and Birks (2006) recommend piloting since this allows for improvement of the instrument. Saunders et al. (2007: 386) state: *"The purpose of the pilot test is to refine the questionnaire so that respondents will have no problems in answering the questions and there will be no problems in recording the data. In addition, it will enable you to obtain some assessment of the questions' validity and the likely reliability of the data that will be collected". Clearly, there are several advantages, as indicated by various researchers such as Hussey and Hussey (2003), Sekaran (2003) and Saunders et al. (2007), who have explained the purpose and nature of pilot testing.*

- 1. The pilot test enables the researcher to obtain some assessment of the questions' validity and likely reliability of data that will be collected.
- 2. Pre-testing may be conducted with just a small number of respondents to establish comprehension.
- 3. Pre-testing may involve friends, colleagues, experts or academics and people with diverse insights and ideas.
- 4. Pilot-testing almost guarantees that respondents in the study will have no problems in answering questions as these have been refined during the process, and that there will be no problem in recording data.
- 5. Pilot-testing allows for research variables, questions and ideas to be verified in terms of their adequacy and ability to investigate the research problem.

Figure 6.7 illustrates the research elements and activities required for the pilot studies undertaken, developed on the basis of information gathered from literature and other documentation.



Figure 6.7 Research Activities and the Pilot Test

Many researchers, for instance, Bryman and Cramer (2001) and Banced and Mittoo (2002), suggested that short questionnaires receive higher response rates. This study's questionnaire consists of five sections; nominally, each has a different number of questions and the

Source: Hardaker (1997).

questionnaire was limited to 10 pages. Consequently, PhD students in the University of Huddersfield Business School took an average of 25 to 30 minutes to complete this questionnaire. Once the opinion of the researcher's supervisory team had been taken into consideration to ensure that the questionnaire's length was reasonable, understandable and the layout as precise and attractive as possible, the aforementioned suggestions concluded that this was the best method of testing the questionnaire. In this research, the pilot study was conducted in four stages.

The first stage: piloting the questionnaire involved providing the draft version to four PhD students within the University of Huddersfield Business School and another three PhD students from each of Manchester and Liverpool Business Schools. They proffered many useful comments relating to the wording of questions, the clarity and ability to understand its contents. Most of their comments were, accordingly, taken into consideration. This method of questionnaire pre-testing was proposed by Remenyi et al. (1998: 151): "Approaches to pretesting can be fairly informal where one consults friends, colleagues, experts and people of adverse opinions, or it could be more informal, involving a pilot study which is a replication, on a small scale, of the main study". The second stage: the questionnaire was given to five members of academic staff at the University of Huddersfield Business School. Three universities in Libya also received two versions of the questionnaire, one in English and one in Arabic (see Subsection 6.8.2.1.2.4). They were asked to provide feedback concerning the questionnaire in terms of design, format and layout, besides the addition and deletion of certain questions, words, contents and measurements. Useful feedback was received from the academic staff which resulted in amendments to the wording and scales of questions. The third stage: a copy of the questionnaire was e-mailed to 16 academic researchers internationally and to IMF staff members dealing with the Arabic situation, specifically pertinent to the Libyan economy. Positive feedback was received from these groups including suggestions for alterations to the wording and measurement of a number of questions related to the balanced scorecard. The final stage: the pilot survey was delivered to a number of experts in Libya, among them fifteen members of financial and non-financial sectors. These were selected randomly from the sample and used for the survey. Ten questionnaires were completed and returned. This reflects a response rate of 66.7 per cent of the pilot survey's total sample. Feedback was received from respondents, which suggests that the questionnaire was clear in nature and straightforward to complete.

Many modifications were made to the questionnaire as a result of pre-test and pilot test stages. In response to the comments received, the number of pages was reduced without excluding any questions. Most of the questionnaire modifications related to organisation comprehensibility, layout, instructions and improvements to the clarity of content to make it more intelligible to readers.

6.8.2.1.2.6 Fieldwork

At the initial stage of the fieldwork a personal visit was made to the Libyan financial market, the central bank of Libya, companies and other financial organisations in order to build trust with participants and to determine a deadline for completing the questionnaire and interviews. In addition, through these visits the researcher was able to determine which managers are in charge and to be interviewed. This stage started in September 2007 and ended in January 2008. For fieldwork in Arab countries, hence, in Libya, to be successful, certain requirements must be met.

- The researcher must be competent in Arabic (first language of the researcher), be aware of cultural influences, appreciate the time available to both researcher and participants and understand all the ethical issues concerned with conducting research in that particular environment.
- The researcher must be aware of the importance of human and social relationships (family support, friends, relatives) and be able to capitalise upon these since social networking is very important and helpful.
- The researcher must be able to encourage the participants' full co-operation and, hence, be capable of ensuring that they completely understand both purpose and nature of the research.
- The researcher must be able and prepared to learn from the experiences gained in the process and to accept that these may change the research's intended plans as it progresses. In other words, it is necessary to appreciate that there is much to be learnt about the subject area in Libya, both theoretically and empirically, since it is a developing country where there has been little empirical work concerned with the financial market.

6.8.2.1.3 Response Rate

The main survey was carried out in 34 (the entire population) financial and non-financial organisations between September 2007 and January 2008 (see Table 6.2). A total of 330 questionnaires were distributed and of these 203 were returned completed. A total of 205 questionnaires were returned, two of which were ineligible for consideration because they

were not completed in a satisfactory manner (see Table 6.5). The total response rate for this research is 61.5 *per cent*. Such an adequate response rate, which is, additionally, a clear indication of business management interests in the topic, can be attributed *inter alia* to following Oppenheim's (2000) and Robson's (2002) recommendations of increasing the response rate.

- Sending an official letter to potential respondents prior to questionnaire distribution (see Appendix D) in order to introduce the research to respondents.
- A covering letter should accompany the questionnaire (see Appendix C) to explain research objectives and emphasise that the respondent's co-operation is a major contributory factor to the research's success.
- The researcher reminded respondents via telephone calls and personal visits to answer the questionnaire.
- All the information provided by the participants and their answers will be treated in confidence and will not mention their names or companies in the written documentation.

A total of 125 respondents did not respond due to several reasons as listed in Table 6.4. Non-response was attributed to: being busy or work pressure 34.4 *per cent*; financial sectors not underestimating people's research information 15.2 *per cent*; apology, with no reason offered 27.2 *per cent*; questionnaires unreturned 23.2 *per cent*.

Dessons	Non-responses (N=125)			
Reasons	Ν	Percentage		
Busy/work pressure	43	34.4		
Financial sectors not releasing information	19	15.2		
Apology without reasons	34	27.2		
Other reasons	29	23.2		
Total	125	100		

 Table 6.4 Number of Respondents who did not Respond

Source: Developed for this research from fieldwork data analysis.

The chi-square test was used to test whether or not the sample was representative and unbiased. Table 6.5 displays the output of chi-square statistics as indicating no statistically significant difference between respondents and non-respondents within the financial and non-financial sectors ($x^2 = 3.872$, P = 0.424, 2-sided) (more details see Chapter Seven: Sample Description Characteristics).

Treatment Crown		Sample (valid N=330)				
I featment Group	Responses	Non-responses	Total N	Per cent		
Capital Market (CLM)	49	19	68	20.6		
Currency Market (CYM)	43	27	70	21.2		
Private Sector (PVS)	41	28	69	20.9		
Public Sector (PCS)	38+2 ^a	22	60	18.8		
Financial Institutions (FLI)	32	29	61	18.5		
Total	205 (61.9%)	125 (38.1%)	330	100		

Table 6.5 Sample Representative by Financial and Non-financial Sectors

Notice a: Total number of responses from the public sector was forty of which two were unusable.

Source: Developed for this research from questionnaire survey data analysis.

6.8.2.2 Semi-structured Interviews

The interview can be described as a purposeful discussion between an interviewer and interviewee (a respondent), with the intention that the researcher can elicit certain information from the interviewee which is of relevance to the research questions (Frankfart-Nachmias and Nachmias, 2007). Hussey and Hussey (2003) define an interview as a method of collecting data in which participants are asked questions in order to ascertain the solution or what they think or feel. This technique can be adopted whether the research uses a positivist or phenomenological approach. In addition, several Arab researchers have used semi-structured interviews for their data collection. For instance, Al-Faleh (1987), Al-Rasheed (1996), Al-Ali (1999), Al-Bahusseir (2000) and Al-Duaij (2004) all found this technique to be advantageous in Arab organisations, where managers prefer to talk rather than to complete a questionnaire. According to Knox and Masilela (1990), the opinions of such managers and professionals are significant as they may be active as *social gatekeepers* and within their distinctive values, attitudes and role orientations, assist to set the agenda and ultimate direction of policy development related to unplanned settlements. According to Leedy and Ormrod (2005), use of interviews as a data collection method depends on research objectives. To help in this decision, it must be considered that interviews can be divided into the following three types: structured, unstructured and semi-structured. In this research the semi-structured interview technique was selected for the following reasons.

- a. Semi-structured interviews have the highest degree of flexibility of all qualitative methods and are able to reach specified subjects more effectively than the questionnaire method (Leedy and Ormrod, 2005).
- b. Usage of interviewer-administered questionnaires requires the researcher to interview respondents and, as this research uses semi-structured interviews as well as a questionnaire approach, it is not necessary to interview respondents twice.

- c. It allows the researcher to ask more complex questions and to ask follow-up questions not possible in the questionnaire. Moreover, it takes into account the non-verbal communication such as the feeling, behaviour, attitudes and facial expression of the interviewee. Thus, it may allow a higher degree of confidence in the replies than in the questionnaire responses (Hussey and Hussey, 2003).
- d. The use of the interviews increase the certainty. Due to the direct contact between the interviewer and interviewee it allows the researcher to explain purpose of the study more freely and to clarify any doubt or to avoid misunderstanding of the questions or the concepts (Oppenheim, 2000).

In this research, semi-structured interviews were conducted after the questionnaire had been administered. It is important to state that they were used to obtain a clearer understanding of the whole problem, to allow the researcher to focus on particular themes that emerged in the questionnaire findings and to explore them.

6.8.2.2.1 Pre-testing and Piloting the Interview

To ensure the questions in the interview schedule were valid in terms of research objectives, it was necessary to pilot them. Consequently, the interview questions were pre-tested with a small number of managers (six from the Libyan stock market and financial sector general managers) mainly to determine that the wording of the questions was unambiguous and to recognise what had to be effected to achieve a satisfactory response during the main study.

Test interviews were conducted to study the reaction of managers to particularly sensitive questions and to investigate the likely duration of interviews, as well as to estimate the time-scale for the research, taking into account peculiarities of the behaviour of the majority of Libyan business people. Interviewees were chosen on the basis of favourable access conditions and were used to assess both proposed data collection procedures and type of data needed to address the research questions. The subsequent subsection discusses key components of the interview pilot study.

6.8.2.2.2 Interview Sample

It was decided that semi-structured interviews would be conducted with the manager of the Libyan stock market, along with the senior managers of banking sector organisations, companies and other financial institutions (two from each of five case study organisations, including one executive or senior manager who was one of the organisation's decision-

CHAPTER 6: RESEARCH METHODOLOGY

makers) (see Appendix J). The individuals who participated in the interviews were not respondents to the questionnaire survey but were chosen because they are the sole individuals with the requisite information about Libyan stock market issues, banking, management and finance, and because they occupied critical roles in their financial institutions. Interviews were conducted on the premises of the organisations concerned and either Arabic or English was involved as the medium of discussion, according to the interviewee's preference and to ensure that misunderstandings were avoided. Interviews lasted between 30 and 60 minutes and were tape-recorded, with the interviewee's prior permission. The semi-structured interview is very beneficial in providing an in-depth analysis of points under investigation. In this regard, Sekaran (2003) argues that, in organisational settings, the opinions of leaders who are very knowledgeable and who can provide the desired information are included in the sample, either because they are the only ones who possess it, or to conform to some criteria set by the researcher. However, in choosing a sample for semi-structured interviews, it is additionally necessary to remember several issues. In this research targeted general managers and senior managers were interviewed rather than asked to complete a questionnaire. The reasons behind that are now given.

- In the organisational setting the opinions of leaders who are very knowledgeable are included in the sample. Enlightened opinions, views and knowledge constitute a rich data source (Sekaran, 2003).
- General managers, who are found in the headquarters of each organisation and companies, are the most knowledgeable people in terms of financial market practise.
- People in the financial sector who had refused to respond to the questionnaire would probably not agree to grant an extensive personal interview
- Using the questionnaire respondents as a sample frame has the advantage, in both selection and analysis, of being able to draw on substantial information provided by the questionnaire.

6.8.2.2.3 Interview Process

Interviews utilised as a data collection method assume that participants' perspectives are significant and knowable (Kvale, 1996). During the first stage of using the semi-structured interview method, as shown in Appendix I, a list of eight questions to be covered varying from interview to interview was included in the pilot study. It found that interviews depend not solely on the quality of the questions asked, but on the knowledge of, and control over, the

CHAPTER 6: RESEARCH METHODOLOGY

interaction involved. Interviews were conducted with respondents from financial and nonfinancial organisations and companies in order to account for respondents' specific organisational background. Interviews took the form of conversations between the researcher and the respondents and required approximately 30-60 minutes each. Most semi-structured interviews were held at the managers' official offices for their convenience. Every interviewee was explicitly asked if they would give permission for the interview to be recorded or taped. The researcher aimed to make 18 semi-structured interviews; a total of 14 were achieved, while 4 were not achieved because 2 of the people were very busy and 2 others cancelled the appointments many times until it was too late. Three of them were general managers and one of them was a senior manager. Six interviews were tape-recorded with permission; this was a significant action, which allowed interviewees the freedom to express their views. Eight interviews were not recorded and only partial note-taking was used. For those interviewees who did not give permission for tape-recording, notes were taken immediately the interviewees replied to their questions. Furthermore, interview skills required consciousness of the likely impact of the interviewer upon the respondents. It was necessary to maintain sensitivity to the cultural aspects of managers' sophistication, non-verbal as well as verbal, because interaction was not only structured by the questions but also by personal feelings. Thus, interviews with insiders (see Appendix J for an interview sample) and questionnaire surveys were performed at the second stage of the fieldwork and during the follow-up period.

6.9 VALIDITY AND RELIABILITY

Validity and reliability of a research study are crucial elements that function as basic criteria for evaluating its accuracy and precision. Validity is required in order to ensure that the same results are gained on re-measurement and reliability is necessary to ensure that the same results or observations are obtained or made by different researchers on separate occasions (Sekaran, 2003 and Saunders et al., 2007). Similarly, Leedy and Ormrod (2005) argued that a data collection method must be valid, accurate and reliable because both validity and reliability can affect the probability that a researcher obtains statistical significance in his data analysis and the extent to which a researcher can draw meaningful conclusions from data, as Table 6.6 displays.

Table 6.6 Data Quality Issues Related to Questionnaires and Semi-structured Interviews

Data Quality Issue	Questionnaires	Semi-structured Interviews
Validity	Low	High
Reliability	High	Low

Source: Hussey and Hussey (2003); Saunders et al. (2007).

Since the researcher adopted a multi-method approach using two data collection methods (personally-administered questionnaire and semi-structured interview), before, during and after collecting data, many procedures were undertaken to ensure validity and reliability of study findings. These procedures are now described.

6.9.2 Validity

Validity is the extent to which research findings accurately represent what is happening in the situation (Hussey and Hussey, 2003). It is not raised as a data issue with regard to qualitative research (qualitative interviews) (Healey and Rawlingson, 1994; Saunders et al., 2007) but, rather, refers to the degree to which a test measures what it is intended to measure. There are two different types of validity. The first is predictive validity, which refers to the ability of the measuring instrument to differentiate among individuals with reference to predict a future criterion variable (Sekaran, 2003). The second is concurrent validity which refers to the extent to which a measurement scale relates to other well-validated measures of the same subject (Oppenheim, 2000). This is established when results obtained from the scale used are consistent with results from other scales utilised to measure the same object (Oppenheim, 2000). Moreover, it is possible to talk of external and internal validity. External validity of research findings refers to the data's ability to be generalised across people, settings and times (Cooper and Schindler, 2006) thus, it depends on the selection of a representative sample. In order to meet the requirements of validity, as suggested by Malhotra and Birks (2006) and Saunders et al. (2007), the following procedures were undertaken in this research.

- An extensive literature review was undertaken to define and clarify the questions used in the questionnaire. Many questions used in the questionnaire were adapted or adopted from related previous studies which, in turn, made possible the comparison of the research's findings with those findings of other studies.
- The use of the entire population was targeted as the sample for this research which raises the external validity; furthermore, there was a high response rate for the questionnaire (61.5 per cent) and for the semi-structured interviews (78 per cent).
- 3. To meet content validity requirements an extensive literature review was undertaken to define and clarify scales and measures adopted in the study. Sekaran (2003) stressed the need to use valid and reliable measures to ensure that research is scientific and indicated the importance of developed measures and scales since their reliability and validity have been established by previous research and are already known to be good.

- 4. Prior to using a questionnaire to collect data, the pilot study for the questionnaire was conducted in three stages: colleagues, academic experts and target companies. This test was to judge the content and face validity of the questionnaire. They suggested that the content validity of the questionnaire was established.
- 5. Usually, the validity of in-depth and semi-structured interviews is very high because of the flexible and responsive interactions possible between interviewer and interviewee, allowing the meaning to be probed, the topic to be covered from a variety of angles and questions to be made clear to respondents (Saunders et al., 2007).

6.9.3 Reliability

The reliability of a measure is an indication of the stability and consistency of the instrument which is without bias (*error free*) and, hence, ensures consistent measurement across time and across various items in the instrument (Sekaran, 2003). The *high level* of reliability that is possible in relation to questionnaires is made evident by Robson (2002) and Hussey and Hussey (2003). In this study questions were designed to be clear and understandable and the fact that they were all piloted suggests they were able to collect accurate data. Furthermore, the researcher attempted to ensure that it was the desired respondents who had completed the questionnaire by having an informal conversation with them where possible. There is an argument about the recommended minimum acceptable limit of *alpha*, Hair et al. (2005) and Sekaran (2003) argue that if the alpha coefficient scores are less than 0.60 they are considered to be poor. Nunnally (1978) and Sharma (2002) explained that the alpha or *cronbach alpha's* generally accepted minimum level is 0.50. In this study the *cronbach alpha* scores for the multiple-item questions that were adopted to measure the independent variables are presented in the research data analysis.

However, reliability is raised as a quality data issue when using semi-structured interviews, as it is not certain that alternative interviews conducted by other researchers would reveal similar information (Robson, 2002). Reliability can further be damaged by other factors, such as interviewer bias and interviewee or response bias (Saunders et al., 2007). The latter can be caused when interviewees are asked sensitive questions that they do not wish, or are not empowered, to discuss, with the outcome that the interviewee may provide only partial or biased information.

6.10 DATA ANALYSIS STATISTICS

Empirical analysis of quantitative data will be conducted by using *SPSS* software and will concentrate on economic variables identified as indicators of progress. Analysis of data from the questionnaire exercise and interviews will be standardised in a way that reduces raw data to categories and forms appropriate to the research themes (Myers, 1999). Malhotra and Birks (2006) proposed that choice of employing particular techniques depends on the type of data regarding the study's objectives and the nature of data (i.e. nominal, ordinal, interval and ratio) plus distribution of data (i.e. normal versus, non-normal). Hussey and Hussey (2003) suggested three areas for consideration when choosing and applying a suitable statistical technique:

- The number of variables to be used in the analysis
- The nature of the data, i.e. whether normally-distributed or not
- The measurement scale of the current data (nominal, ordinal, interval or ratio).

Two main statistical techniques can be used to explore the differences between groups and techniques to explore the relationship between variables to conduct the data analysis, parametric and non-parametric statistical tests; choice depends on the type of data (Malhotra and Birks, 2006). Parametric statistics are utilised when data *reflect an interval or ratio scale* and, secondly, *fall in a normal distribution* (Leedy and Ormrod, 2005: 263). Hair et al. (2005: 259) emphasise other assumptions when deciding to use parametric or non- parametric, stating that: "*in general, when the data are measured using an interval or ratio scale and the sample size are large, parametric statistics are appropriate.*"

Several arguments have been forwarded regarding usage of parametric and non-parametric tests. Siegel and Castellan (1988) and Sekaran (2003) contested that a parametric test can be used under two conditions when the level or scale of measurement is of equal interval or ratio scaling and when distribution of population scores is normal. But, if data fails to satisfy these conditions, the cautious decision should be to employ non-parametric statistical tests. Non-parametric tests are distribution-free tests and do not require the average score of each variable, as is the case in their parametric counterparts. Hair et al. (2005) explained, however, that, in business research, it is appropriate to treat the ordinal scale statistical treatment as if it were interval. Consequently, non-parametric statistical methods were used to conduct the analysis. The rationale for using these statistical methods is presented in the following subsections are provided in Chapters Seven to Nine. In addition, the content of the following Subsections,

6.10.1 and 6.10.2, will be drawn off to interpret and explain the statistical findings discussed in those chapters.

6.10.1 Quantitative Data Analysis

In order to achieve the research aim and objectives, a number of statistical techniques were used in primary data analysis. A brief discussion of these statistic is given in the following subsection.

6.10.1.1 Inferential and Differences Statistics Test

Inferential statistics have two major functions: firstly, to estimate a population parameter from a sample *e.g. whether the sample is representative of the total population*; secondly, to test statistically-based hypotheses *e.g. whether they are organisational variables related to the level of institutionalisation* (Leedy and Ormrod, 2005). Moreover, inferential statistics can be involved using a drawn sample of a population to the characteristics of the larger population (Leedy and Ormrod, 2005). However, Hair et al. (2005) contested that, if a researcher conducts a census of the entire population, statistical inference is unnecessary, because any difference or relationship, however small, is *true* and does exist. In this category, two of forms statistics were adopted. It was decided to utilise a number of statistical methods in analysing that data: descriptive statistics (i.e. frequency, mean, percentages and standard deviation); kurtosis and skewness. They were employed to test the first, second and third objectives. To carry out this analysis the five point likert scale was collapsed into three categories to facilitate the analysis of the large number of items included in the questionnaires. The rationale for using these statistical methods is presented below, and were in the tests presented in Chapter Seven in order to obtain a comprehensive picture of the research sample and issue.

6.10.1.1.1 Descriptive Statistics

Descriptive, or exploratory, statistics involve the transformation of raw data into a form that provides information to describe (and compare) a set of variables in a situation (Sekaran, 2003). Descriptive statistics, using frequencies, central tendency and standard deviation, were utilised to achieve explorative objectives of this research to determine sample characteristics and to develop primary data distribution. The frequencies procedure provides statistics and graphical displays that are useful for describing many types of variables. For a frequency report and bar chart, researchers can arrange the distinct values in ascending or descending order, or order the categories by their frequencies. The central tendency can be used to (1) provide a simple description of the entire population or sample and (2) make comparisons

between categories of individuals or between sets of figures (Gravetter and Wallnau, 2004). Central tendency can be measured by the *mean* or *median*. Leedy and Ormrod (2005: 258) state that: "the [mean is] appropriate only for interval data, because it makes mathematical sense to compute an average only when the numbers reflect equal intervals along a particular scale." The means are further used to describe types of non-financial performance measurements according to their importance for measurement and evaluation purposes. On the other hand, standard deviation (SD), is used to establish the extent to which the values for a variable differ from the mean. Thus, SD is a measure of how well the mean represents the data. Field (2005) argued that small SD (relative to the value of the mean) indicates that the data points are close to the mean, whereas, a large SD (relative to the mean) indicates that the data points are distant from the mean (i.e. the mean is not an accurate representation of the data).

6.10.1.1.2 Kurtosis and Skewness

Kurtosis is a measure of a distribution's peaked-ness. Skewness is a measure of symmetry of a distribution. Both values were adopted to test and check the normality of each variable included in the research. Normality refers to the degree to which the distribution of sample data corresponds to a normal distribution. The latter is a theoretical contribution to the characterisation, in which the horizontal axis represents all possible values of a variable and the vertical axis represents probability of those values occurring (Hair et al., 2005). When skewness values are larger than +1 or smaller than -1 this indicates a significantly skewed distribution. A positively skewed distribution has relatively few large values and tails off to the right, whilst negatively skewed distribution has relatively few small values and tails off to the left. Thus, skewness values within the range of -1 to +1 and kurtosis values within the range of -3 to +3 indicate an acceptable range. Positive kurtosis is associated with distributions having long, thin tails; whereas, negative kurtosis is associated with shorter, fatter tails relative to the normal curve (West et al., 1995).

6.10.1.2 The Statistical Techniques used to Test the Research Hypotheses

This section describes the statistical methods used to test the research hypotheses (H1, H2, H3 and H4) using the survey data. The first three hypotheses examine the relationship between the independent variables and dependent stock market performance variables (see Figure 6.3); a regression model will be used to test the relationship in the environmental condition of the Libyan economy. On the other hand, the data analysis process conducted to test the fourth hypothesis will be based upon econometric techniques from other emerging markets (see

Figure 6.4). Due to the nature of the research hypotheses, as mentioned previously, two types of statistical techniques are used. The first technique can be used to explore the relationship between variables and the second can be used to explore the differences between groups. Consequently, the statistical tests used to investigate the hypotheses and analyse the results are mostly non-parametric tests. The tests that were used to examine the first three research hypotheses H1, H2 and H3 (Chapter Eight) and the statistics used are discussed below.

6.10.1.2.1 Linear Multiple Regression Models

6.10.1.2.1.1 Multiple Regression Analysis Model

Multiple regression analysis is a statistical technique that can be implemented to analyse the relationship between a single "dependent variable" and several "independent variables" whose values are known to predict the single dependent value selected by the researcher (Hair et al., 2005). Each dependent variable is weighted by an equation of multiple regression to ensure maximal prediction from independent variables (Hair et al., 2005). The weights denote ability to quantify precisely the relative significance of each proposed variable. Burton et al. (1989) define regression analysis as a statistical tool which may be adopted to learn more about the relationship between independent or explanatory variable(s) and dependent variables. The model of multiple regression takes the form of an equation (6.1) that predicts the value of outcome variable Y from a combination of prediction variables, each multiplied by its own respective coefficient, plus residual term:

$$\mathbf{Y} = \boldsymbol{\alpha} + \boldsymbol{\beta}_1 \mathbf{X}_1 + \boldsymbol{\beta}_2 \mathbf{X}_2 + \dots + \boldsymbol{\beta}_n \mathbf{X}_n + \boldsymbol{\varepsilon}_i \tag{6.1}$$

where Y represents the "dependent variable" and $X_1 \dots X_n$ represent the "explanatory" or "predictor variables", and α and β are the "parameters" of the model. Random error term, ε_i , represents the collective unobservable influence of any omitted variables. If β values replace into the equation, the model can be defined. Flexibility of multiple regression ensures that the researcher examines the *truth* of the relationship based on the linearity relationship (Hair et al., 2005) which gives an insight into the inter-relationship between independent variables through the correlation matrix. In this research regression analysis was utilised to achieve the central objective of examining the relationship between the extent of the stock market as the dependent variable and other factors, in terms of economic reform programme, macro-economic reform and financial maturity as the independent variable. This analysis highlights the extent of each independent variable's impact on the variation of stock market dependent variables. It seeks to determine whether or not the extent each variable can explain variation in the level of stock market performance in the Libyan economic situation.

6.10.1.2.1.2 Correlation Statistics

Correlation analysis is used to describe the strength and direction of the linear relationship between two variables (Pallant, 2003). It is also employed to test the effect of moderator variables on the relationship or association between economic reform variables and stock market performance variables. The resulting statistic, called a correlation coefficient represented by (r), is utilised since it gives a number between -1 and +1; a correlation of -1 or +1, which indicates a perfect correlation, negative, respectively, between two variables. Figure 6.8 illustrates a relationship between two variables.

Figure 6.8 Values of the Correlation Coefficient

-1	-0.7	-0.3	0	+0.3	+0.7	+1
Perfect	Strong	Weak	Perfect	Weak	Strong	Perfect
Negative	Negative	Negative	Independent	Positive	Positive	Positive

Source: Saunders et al. (2007).

If the correlation coefficient (r) is zero, then there is no relationship between variables, meaning that they are perfectly independent. But, if the probability (p) of correlation coefficient (r) value is significant (i.e. $p \le 0.05$), there is a significant relationship between the two variables and this is statistically meaningful. In contrast, Cohen (1992) suggested that a value of approximately 0.10 represents a small correlation, 0.30 a medium correlation and 0.50 or greater, a large correlation. Williams (1992) proposes the following values as guidelines for interpreting the significance of the correlation coefficient (r), as shown in Table 6.7.

Table 6.7 Values of Correlation Coefficient (r)

Correlation Coefficient (r)	Descriptive	Interpretation
0.00-0.20	Slight	Almost no relationship
0.20-0.40	Low	Small relationship
0.40-0.70	Moderate	Moderate relationship
0.70-0.90	High	Marked relationship
0.90-1.00	Very high	Solid relationship

Source: Williams (1992).

According to Kohout (1974), interpreting Spearman's Rho correlation coefficient (r), the (r) values indicate the amount of *rank change* in the correlation of Y variable when the X variable increases by one rank. Moreover, because Spearman's Rho (r) coefficient is

completed based on ranked data, it can be demonstrated that the (r) coefficient indicates the proportion of variation in Y – *rank* which is explained by variation in the X – *rank*. If the probability (p) of correlation coefficient value is significant (i.e. $p \le 0.05$), there is a statistically significant relationship between two variables. There is a simpler formula for calculating Spearman's Rho correlation coefficient (r):

$$r_{s} = 1 - \frac{6\sum d_{i}^{2}}{n(n^{2} - 1)}$$
(6.2)

where *n* is the number of period ranks and d_i is the difference between period ranks of X - rank of Y - rank (i.e. a difference score). If there are no tied scores, Spearman's Rho correlation coefficient will be closer to the Person product moment correlation. Also note that the sum of squares from 1 to *n* can be expressed as n (n+1)(2n+1)/6.

6.10.1.2.1.3 Chi-square-based Measures

Chi-Square (χ^2) , or cross-tabulation, sometimes referred as contingency table, is adopted to assess differences between groups using nominal or ordinal variables (Hair et al., 2005). It allows for testing significance of differences between a set of observed frequency tables arranged such that each cell in the resulting table represents a unique combination of specific values of cross-tabulated variables. The resulting chi-square test for independence enables the researcher to test whether there is a significant association between two categorical variables (Field, 2005). So, if the *p*-value for the resulting person chi-square is less than 0.05, hypothesis of a correlation is accepted.

6.10.1.2.1.4 Statistical Significance

Statistical significance (p - level) refers to the degree of difference or association being tested that would only occur by chance alone (*SPSS* 2003) package. Rowntree (1991) has described the various levels of significance to identify the real difference between the means of two samples. If the observed significance level is small enough, usually less than 0.05 or 0.01, the null hypothesis is rejected (H₀). Cooper and Schindler (2006) explained that type I error, which is sometimes referred to as alpha (α), indicates accepting the designed or alternate hypothesis when the null hypothesis is actually *true*. Type II error, sometimes referred to as Beta (β), refers to accepting the null hypothesis when the alternate hypothesis is *true* (Figure 6.9). "It is therefore generally more important to minimise type I than type II errors", Saunders et al. (2007: 443).

Chance of Making			
Type I error	Type II error		
	Significance level at 0.05		
Increased	Decreased		
Significance level at 0.01			
Decreased	Increased		

Figure 6.9	Type 1	I and Typ	e II Errors
------------	--------	-----------	-------------

Source: Saunders et al. (2007).

Bearing this discussion in mind, this research sets $p-level \le 0.05$ as its significance level. Rationale for this choice is threefold:

- a. $(p level \le 0.05)$ is the accepted level when using the Statistical Package for the Social Sciences (SPSS 2003)
- b. $(p-level \le 0.05)$ is proposed by many researchers, such as Hair et al. (2005) and Saunders et al. (2007)
- c. $(p-level \le 0.05)$ is adopted by many previous related business researchers such as Kobrin (1982) and Yazid (2001).

The following test will be used to present and analyse the fourth research hypothesis H4 (see Chapter Nine) by using an econometric model to investigate the difference in determents of financial market between Libya and emerging market countries from the data relating to the period over 1995-2006.

6.10.1.2.2 Econometric Growth Model

6.10.1.2.2.1 Two-Stage Least-Squares (2SLS)

This section discusses the process used to test the fourth research objective, which is presented in Chapter Nine. The Two-Stage Least-Squares (2SLS) approach was used to correct the endogeneity of independent variables, particularly the level of stock market development. Where the cross-country regressions suffer from three kinds of error, statistical, measurement and conceptual, statistical errors occur when regression analysis assumes that observations are drawn from the same population, whereas measurement errors occur when the variables are defined, collected and measured incorrectly across various countries. However, published data sometimes fail to capture the true picture. That means different countries appear in crosscountry regression. Harberger (1987: 256) pointed out: "[...] regressions cross country do not resolve causal issues, nor do the regressions describe a single piece of machinery over time. They should be viewed as evaluating the strength of the partial correlation and not behavioural relationships that suggest how much growth will change when the right-hand-side variable changes. The results unearth suggestive empirical regularities and should not be *interpreted as stylised facts or as behavioural relationships*". In consideration of the remarkable limitations of the cross-sectional method of studying growth, which is a truly dynamic concept, it can be argued that the results of empirical work by Atje and Jovanovic (1993), Harris (1997), Levine and Zervos (1996, 1998b) and Arestis et al. (2001) are subject to criticism and it is not true that the equation of causality is addressed satisfactorily in these studies.

Overall, as mentioned early in Section 6.5, despite the fact that the use of survey questionnaire together with econometrics models is unusual in the field of finance research, Michaelas (1998) argues that the use of the survey based analysis with mathematical techniques models may introduce a new research methodology in the financial market structure research, which can overcome some of the disadvantages inherent with each individual technique. This combination is sometimes called a 'triangulation method', which involves viewing the evidence from different angles or viewpoints (see for instance, Fielding and Fielding, 1986). Consequently, the combination of survey questionnaire and econometrics models can provide a significant contribution to the current understanding of financial market differences. The empirical work in this study uses a systematic combination of linear regression analysis models (Chapter Eight) and econometrics model (Chapter Nine).

6.10.2 Qualitative Data Analysis

In relation to the differences between the quantitative and qualitative research, Rust (1993: 70) has argued that the difference between the techniques is based on the purpose of the analysis rather than the method of data gathering. While qualitative research answers the question of how things happen, quantitative research answers the question of how often things happen. However, qualitative research methods comprise interpretive techniques which seek to describe, decode, translate and explain the meaning, not frequency, of the phenomena in a social world (Van Maanen, 1983). Researchers are not bound by any particular methodology but rather use diverse tools and methods according to the requirement of problem resolution (Denzin and Lincoln, 2000). Qualitative research is most suitable when 'rich' information about a small number of subjects is needed, when a flexible or informal approach is desirable, or when members of the sample population involved are better researched by encouraging description and analysis of situations in their own words (Ticehurst & Veal, 2000).

Although different approaches are not mutually exclusive, they can be combined. It can be argued that the template analysis is the process of organising and analysing textual data according to themes. The data involved are usually interview transcripts, but may be any kind

of textual data. Template analysis involves the development of a coding 'template', which summarises themes identified by the researcher(s) as important in data set and organises them in a significant and useful manner²⁸. Coding is the process of identifying where themes occur in accounts and attaching labels 'codes' to a section of text to index them (King, 2004). Related codes with only small amounts of data might be more appropriately merged into a single code (see Chapter Seven and Appendix K: Section K 5). Based on this discussion, template analysis statistics were used for the following reasons.

- Template analysis is, on the whole, a more flexible technique with fewer specified procedures, permitting researchers to tailor it to match their requirements (King, 2004: 275).
- To reduce large amounts of unstructured text to that which is relevant and manageable for the evaluation.
- To compare perspectives of different participant groups about their experience of the learning context.
- To make inferences about what is influencing participants' communication and behaviour and the impact of their communication and behaviour.

The process of analysis of the qualitative data collected in the current study started when the researcher finished collecting the data. Taking into account the large amount of qualitative information, the process of interview transcription was carried out by the research without outsourcing this technical work. The major reason for this was that the interviews were conducted in Libya and often contained multiple expressions which might prove difficult for interpretation by non-native transcribers. Secondly, it was more logical to intertwine the interview data with the observation notes at the transcription stage to draw better conclusions on the subject of research. Condensing of the data was enhanced through thorough work on the fieldwork materials in printed and audio form. All the data were coded according to the generic corporate restructuring types, leaving space for new strategies which could occur during this research.

The data obtained from the interviews was processed and analysed using a number of steps. It started by listening carefully to the tapes and then transcribing them by writing down each participant's response on a separate sheet of paper. Processing and analysing interviews normally takes time because the researcher needs to listen carefully to the tape recorder,

²⁸ A "template" technique of analysis was described by Crabtree and Miller (1992), and the approach has been further developed by (King, 1998; 2004).

CHAPTER 6: RESEARCH METHODOLOGY

transcribe, read and re-read the data repeatedly (Drever, 1995). Therefore, the researcher read transcribed data accurately in order to identify the topic area related to the study aim. Each question's responses were put together and re-written on different sheets of paper so that all the answers to a particular question were together. Categories of the responses to each question were described in the form of broad themes, e.g. competitive strategies and relevant data was placed under each category. The responses were described in terms of the topics or categories and quotes were used to illustrate them. Ethical consideration suggested that all interviewees had to be coded in order to ensure confidentiality of the provided data (see Appendix J: Respondent Interviews). This was also useful in terms of the categorisation of the information in relation to different institutions. Once a final version of each transcript is defined and all transcripts have been coded to it, the template serves as the basis for the researcher's interpretation or illumination of the data set and the writing-up of findings. The analysis of the interviews is summarised in Tables K.5 and described in more detail in Appendix K, Section K.5.

6.11 SUMMARY AND CONCLUSION

In order to achieve the research aim, distinct objectives were formulated and, in light of these, the main methodological approach has been detailed. In doing this, various methodologies available for researchers were discussed and justifications for employing the positivistic paradigm were provided. Since economic reform and stock market performance literature has enabled the researcher to define a theoretical framework and develop hypotheses, the deductive approach was selected. Among various strategies within this approach, the survey strategy was adopted since it is the most popular and commonly used strategy in business research. The chapter concluded with an explanation of the statistical methods framework used in this research. The results of the regression analysis were, then, merged with the results of the survey questionnaire in order to shed more light of financial as well as non-financial and behavioural issues that affect the Libyan stock market. The combination of different research methods in collecting and analysing data is one of the contributions of this study. A variety of statistics, differences and correlation statistics were applied after due consideration of the assumptions of each. The cross-country comparison stage exploited data from 42 developing countries including Libya. The result of this section was analysed and, subsequently, differences were highlighted in the financing model between the Libyan economy and other emerging financial markets. In addition, the result provided a framework of empirical knowledge which might be adopted in the current economic reform programme within the Libyan stock market performance.

165

CHAPTER 6: RESEARCH METHODOLOGY

After presenting methodological procedures implemented in the present study and various resources for data collection, the following chapters discuss the presentation of findings in this study arising from the research instruments which have been discussed in this chapter. This is displayed and discussed within the three main chapters which follow. Chapters Seven, Eight and Nine relate to the research aim and objectives. For example, Chapter Nine provides an empirical investigation analysis technique between the financing patterns of the developed Libyan stock market and other emerging economy markets to empirically test the final research hypothesis. The econometric growth model discussed in the current chapter will be examined in Chapter Nine. The next chapter (Chapter Seven) will provide the presentation of the findings from quantitative and qualitative data analysis by using the questionnaire survey and semi-structured interviews. Examination of the first three research hypotheses, tested by the linear regression analysis models for Libyan stock market, will be provided in Chapter Eight.

CHAPTER SEVEN

PRESENTATION OF THE FINDINGS

7.1 INTRODUCTION

Having outlined the collection and preliminary screening processes in respect of the data in the previous chapter, the thesis now focuses upon the analysis of that data. Consequently, this chapter considers the data obtained from the self-administered questionnaires and from the semi-structured interviews, exploring the findings in accordance with the study's objectives which are concerned with providing an initial investigation into changes in the Libyan economy since the adoption of an economic reform programme in 1999 and the effect of this programme upon stock market performance. In order to discuss the results obtained from the questionnaire responses, frequency and descriptive statistics were used as a means of analysing the first part of the questionnaire survey. Findings in this respect are summarised and tabulated, leading to an initial *analysis* of the sample and results²⁹. After the presentation of this information, a more *in-depth analysis* and *discussion* of issues involved in the findings obtained from both the questionnaire survey and the subsequent interviews (face-to-face semistructured in nature) is offered. This enables an exploration of some of the research questions and provides more understanding. In presenting the findings, the chapter is divided into four main sections. Section 7.2 describes the data sources and analysis. Section 7.3 represents the general demographic analysis data. Section 7.4 deals with practices of the Libyan financial sector and economic characteristics. Thereafter, a summary and conclusion is provided in Section 7.5.

7.2 DATA SOURCES AND ANALYSIS

7.2.1 Primary Data Sources (Self-administered Questionnaires and Semi-structured Interviews)

The aim of this section is to present and analyse the primary data obtained from the use of self-administered questionnaires and semi-structured interviews to assess the relationship between the independent and dependent variables³⁰. Each interview lasted between 30 and 60 minutes and all interviews were tape-recorded and later transcribed. The strategy was to begin

²⁹ The rationale and assumptions of using each statistic are discussed in Chapter Six: Section 6.10.

³⁰ In the next chapter, a linear regression model is provided to test the relationship between these variables.

with broad questions and then, as the interviews progressed, to concentrate more on the research objectives. During the process the interviewees were encouraged to expand on any point they considered relevant and/or important.

7.2.2 Data Analysis

The data collected from the self-administered questionnaires and semi-structured interviews at the research site were analysed in the light of the research framework. Each question and interview was analysed and the significant issues raised were identified. Statements representing these issues were also summarised and listed. Furthermore, the main concepts highlighted by each question and interview were used as key points.

7.3 GENERAL DEMOGRAPHIC ANALYSIS

Discussion of the general background of respondents, as revealed in the demographic questions, offers a clear perspective of Libyan economic issues gathered in the initial part of the questionnaire survey (Appendix G: Section A). The data was collected in order to discover whether the response of the sample population was representative of ideas about the Libyan economic reform programme and performance of stock market. It was also collected to explain, in some detail, the concepts and percentages frequently utilised in the data analysis. Other statistical tables and figures are located in Appendix K.

7.3.1 Sample Description Characteristics

As discussed in Chapter Six, a total of 330 questionnaires were distributed, of which 205 were returned (a response a rate of 61.9 *per cent*). Of these, two were unusable giving a number of 203 (61.6 *per cent*) usable questionnaire responses. Non-responses numbered 125 (38.1 *per cent*) (see Chapter Six: Subsection 6.8.2.1.3, Response Rate). Table 7.1 compares the responses and non-responses of the sample for each group (capital market, currency market, private sector, public sector, financial sector and financial institution) and the eventual sample size as a result of non-responses. The survey population consists of people believed to be familiar with economic reform and financial markets. This includes those who might have professional and technical ability in the economic reform field, those who are expected to be involved in one way or another in stock market performance in the Libyan economy and those who might assist in change. Such a heterogeneous group has been surveyed in previous studies relating to economic reform programmes, macro-economic reform, financial maturity and stock market performance.

~	Respon	nses	Non-responses		
Groups	Frequency	Percent	Frequency	Percent	
Capital Market (CLM)	49	24.1	19	15.2	
Currency Market (CYM)	43	21.2	27	21.6	
Private Sector (PVS)	41	20.2	28	22.4	
Public Sector (PCS)	38	18.7	22	17.6	
Financial Institutions (FLI)	32	15.8	29	23.2	
Total	203	100.0	125	100.0	

 Table 7.1 Distribution of Responses by Sample Size of Population Groups

Source: Analysis of questionnaire survey data.

7.3.2 Characteristics of Respondents

This section describes the characteristics of the sample, consisting of general managers, senior managers and employers from other financial institutions. Appendix K contains tables and figures showing how these characteristics have been broken down into six main groups: gender, age, education levels, academic background, current position and work experience.

Gender and Age

Appendix K, Table 1.1 indicates the demographics for each of the capital market and groups. These groups have a gender distribution of approximately 75.6 *per cent* male, with the exception of the public sector group, which was 42.1 *per cent* female. The sample was biased towards the higher ratio of 40 *per cent* aged between 41 and 50 years. Among the youngest participants, 10 *per cent* were aged under 30 and 35 *per cent* were aged between 30 and 40, followed by 17 *per cent* who were aged between 51 and 65. The group sample sizes were not equal. The average age of participants was 45. Nonetheless, multivariate analysis of covariance allows for differences in group size (Hair et al., 2005). A higher ratio of managers belonging to the older age group, between 41 and 50, was found in the Currency Market group with prior experiences, followed by the Capital Market group (39 *per cent*).

Education Levels and Academic Background

Appendix K, Table 1.2 illustrates that the majority of respondents in the groups (63.2 *per cent*) held a Bachelor's degree, mostly in business and related science. 30.2 *per cent* of the participants from the Currency Market group held a Master's degree and 18.4 *per cent* from the Capital Market group were PhD-holders. These results suggest that most respondents had achieved reasonable levels of education, and hence, would be able to discuss their problems and provide clear opinions about their knowledge. The findings are supported by an empirical survey conducted by Porter and Chairman (2006) that documented literacy levels in the Libyan education system as being significantly high, and among the highest in the MENA countries, at 82 *per cent*, with youth literacy reaching 100 *per cent* and female literacy considerably better than in many MENA countries. Appendix K, Table 1.3 shows that a high

percentage of respondents held a degree in accounting and finance (38.8 *per cent*). The second highest percentage (20.9 *per cent*) held degrees in banking. Management and marketing degrees were held by 36.8 *per cent*, and 28.1 and 12.5 *per cent*, respectively, were in economics and business studies.

Current Position and Work Experience

Appendix K, Table 1.4 displays a numerical overview of the types of job held by all managers. Administrative managers accounted for 38.8 *per cent* of the sample, human resource managers 35 *per cent* and public relations managers 24 *per cent*. General and senior managers accounted for 22 *per cent*. Responses to the questionnaire suggest that respondents do not switch careers frequently. Appendix K, Table 1.5 shows that approximately 70.9 *per cent* of the participants had more than 10 years of work experience. Average duration of work experience in the capital market, currency market, private sector, public sector and financial institution groups, respectively, was 20, 17, 14, 13 and 10 years. Fewer than 9.9 *per cent* of the sample possessed less than three years' experience.

7.3.3 The Characteristics of Participants' Companies

The main purpose of this section is to provide a brief outline of the companies' characteristics. It covers seven main features: bank category, employee numbers, ownership, size, current market region, industry and company revenue.

Bank Category and Employee Numbers

Appendix K, Figure 1.1 presents the Libyan banks classified into five main categories, as participants from commercial banks represented the majority of the sample (76). Almost all commercial and private banks participated in this study, representing 37.4 *per cent* and 22.2 *per cent*, respectively, while other foreign and investment banks accounted for 14.8 *per cent* and 15.8 *per cent* of the study's participants. Appendix K, Figure 1.2, demonstrates that 24.1 *per cent* believed that their capital market employed fewer than 850 people because it had only started in recent years. The current market employed over 3,210 employees, while the private sector employed, on average, between 851 and 1,340 people. The public sector had 18.7 *per cent* of 2,106-3,210 employees, and the fifth group had, on average, 1,341-2,105 employees. Following Khalifa and Aspinwall (2000), the researcher classified companies in two main categories by size: large companies with 100+ employees and small companies with less than 100 employees.

Ownership

Allocating the responses according to type of ownership, as outlined in Appendix K, Figure 1.3, indicates that the sample consists of 95 public (state-owned) companies (46.8 *per cent*) and 108 privately-owned companies (53.2 *per cent*) in various industries. The Chi-square test confirmed significant differences among company ownership in terms of public and private. Two main reasons lie behind the growth in the number of companies during the last few years.

- The benefit from the increased oil boom of the late 1970s to mid-1980s and between 2003 and 2007: these periods were considered as higher economic growth periods (Appendix A, Section A 1.2: Growth and Structure of the Libyan Economy)
- The economic reform programme, which started in 1999 and aimed to liberalise and modernise the Libyan economy (Appendix A, Section A.1.2: Growth and Structure of the Libyan Economy).

Size and Current Market Region

Respondents were divided into five groups depending on the company's total assets in million LD. Appendix K, Figure 1.4, illustrates that there are 84 small companies of less than 10 million LD (23.2 *per cent* of respondents), 47 large companies (more or = 30 million LD - 41.4 *per cent* of respondents), the remainder being medium-sized companies (10-30 million LD - 35.5 *per cent* of respondents). As shown in Appendix K, Figure 1.5, most markets are local, accounting for 50.7 *per cent*. The explanation for this high percentage was the Libyan economy's closed door policy prior to 1992. Regional markets represented 20.2 *per cent*, local and regional 15.3 *per cent*. For instance, the Libyan market operates within North African countries. The lowest percentage was that of the international market, at 13.8 *per cent*. This can be explained in two ways. Firstly, the government's policy encourages the establishment of local companies rather than regional or international companies. Secondly, the Libyan economy is relatively poor in the international marketplace and, therefore, it is unable to compete (Awamleh, 2002).

Industry and Company Revenue

Allocating responding companies according to industry classification, as shown in Appendix K, Figure 1.6, reveals that manufacturing and mining constitutes 24.1 *per cent* of the sample, while non-manufacturing constitutes 75.9 *per cent*. Antoniou et al. (2002) argued that

manufacturing and non-manufacturing³¹ companies have few differences among their assets' structure and their degree of weakness *vis a vis* changes in capital markets. For this reason, factors that may affect capital market decisions may differ in these two sectors. Allocating the responding companies according to a company's percentage of revenue in terms of income, as displayed in Appendix K, Figure 1.7, reveals that 36.7 *per cent* of samples are low income, 22.7 *per cent* are medium-income and 40.4 *per cent* are high-income, which can additionally be regarded as high-productivity.

7.4 PRACTICES OF THE LIBYAN ECONOMIC CHARACTERISTICS

The aim of this section is to *present* and *analyse* data obtained from both a self-administered questionnaire survey and semi-structured interviews regarding research objectives relating to the process of Libyan economic reform and stock market performance factors, including financial maturity and macro-economic reform variables. The analysis of the general results about Libyan economic activity and Libyan businesses is discussed according to the following sequential steps. In order to conduct the necessary calculations and cross-references required for tabulation, data was entered into an *SPSS* 2003 package. Closed questions used a 5-point scale "Likert scale"³² to illustrate disagreement and agreement factors. These were summarised using numerical values of the scale running from 1 "strongly disagree" to 5 "strongly agree". The tables and all other statistical figures are located in Appendix K for clear presentation.

The section is divided into four main subsections. Subsection 7.4.1 deals with the economic reform programme independent variables, while subsection 7.4.2 illustrates macro-economic reform independent variables. Financial maturity independent variables are discussed in subsection 7.4.3. Subsection 7.4.4, which concludes the chapter, deals with stock market performance dependent variables.

7.4.1 Economic Reform Programme Independent Variables

This group of variables covers the second part of the questionnaire survey (Appendix G: Section B) and some interview questions (Appendix K). Fulfilment of economic reforms led to the transformation of the social economy into a market economy within the Libyan context.

³¹ Akbar (2001) defines manufacturing companies as those which produce goods, items for use and trade, skills and resources, and non-manufacturing companies as those involved in transportation, finance, insurance and the service industry.

³² The rationale and assumptions of adopting a 5-point scale "Likert scale" were discussed in Chapter Six: Subsection 6.8.2.1.2.2.

Deregulation, corporatisation, privatisation, liberalisation of prices, trade and capital movement were among the main supports of the transformation. This section focuses on the relationship of Libyan economic reform programme independent variables. It was critical to determine the approach to the deregulation framework in the Libyan economy and discover the identity of corporatisation, privatisation and liberalisation.

7.4.1.1 Deregulation

Questionnaire Findings

Questionnaire respondents were presented with a list of eight questions concerning deregulation and were requested to rate degrees of concern on a 5-point rating scale. The average of the Likert scale is 3 ((1+2+3+4+5)/5) and hence, a mean above 3 shows an agreement with the statements while a mean below 3 shows overall disagreement. Based on the means, skewness and kurtosis³³, the output of tests, as presented in Table 7.2, indicates similarity among some means, for example, 'Regulation stability encourages companies to invest in the stock market' (mean 4.29, Skewness -1.82, Kurtosis 2.13) and 'The laws of the stock market assist in promoting investment in the capital market' (mean 4.16, Skewness -1.55, Kurtosis 2.08). These results referred to the regulatory and institutional environment that helped to define the development stage of the stock market in order to secure a successful movement towards the new markets. This movement involved activities such as encouraging private ownership of economic activities, reducing the role of the state to be limited to some public activities such as health, education and security, and privatising state-owned interests and liquidating unprofitable business units. This is consistent with the argument proposed by La Porta et al. (1997) that the legal environment influences the size of the capital market, which, in turn, affects capital structure decisions. In this regard, La Porta et al. (1998) state that the legal protection to investors, which includes the content of the law and the quality of its enforcement, is one of the most significant treatments to mitigate agency problems.

The statement: 'The laws of the stock market play a crucial role in limiting the value of companies' shares' (*Mean* 4.01, *Skewness* -1.30, *Kurtosis* 1.21) was ranked lowest and, therefore, these laws were considered the least important source of deregulation of those listed. The loss of market share and poor financial performance were the catalysts of change, which

³³ Skewness values within the range of -1 to +1 and Kurtosis values within -3 to +3 indicate an acceptable rate for normality, whereas values decreasing outside the range of Skewness and kurtosis indicate a substantial exit from a normal distribution. The rationale and assumptions of using *Skewness* and *Kurtosis* were discussed in Chapter Six: Subsection 6.10.1.1.2.

could be due to the change in the organisational strategic goals in the state regulations and/or the increase in market competition. Respondent L supports this result saying: "*The deregulation of the economy has negatively affected the company's performance. The worst impact was made by the change in the foreign currencies exchange rates in 1999, which has increased the production costs*". The overall mean of 4.11 indicates the respondents' agreement with the statements in relation to the deregulation factors and the application of the economic reform programme.

Code	Sode Dank Statements -			Valid	N=203	
Code	Kalik	Statements	Means ^a	SD	Skewness	Kurtosis
B.1	3	Economic reform is an important factor in the regulation of the stock market	4.1429	0.98719	-1.569	2.375
B.2	4	Regulation of the stock market is good for investment	4.1232	1.00474	-1.403	1.662
B.3	1	Regulation stability encourages companies to invest in the stock market	4.2857	0.87152	-1.815	2.133
B.4	2	The laws of the stock market help to promote investment in the capital market	4.1626	1.02840	-1.545	2.077
В.5	6	Regulation is an important element to market and promote investment for the stock market	4.0542	1.07262	-1.324	1.161
B.6	5	Regulation tools contribute to improvement of the stock market	4.0985	1.06713	-1.482	1.676
B.7	8	The laws of the stock market play a crucial role in limiting the value of companies' shares	4.0099	1.06701	-1.304	1.213
B.8	7	Simplified procedures affect investment in the stock market	4.0394	1.03810	-1.259	1.155
			-			

Table 7.2 Descriptive Statistics for Deregulation

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree".

Source: Analysis of questionnaire survey data.

Interview Findings

Managers interviewed in the financial institutions were asked about change in the financial deregulation context. Since interviewees were not required to rank in order or rate the aspect, the relevant measure was simply the number of times any given deregulation was mentioned. One financial regulations manager said:

"The deregulation structure policy of the Libyan economy is shaped by the following deregulation objectives: creation of appropriate measures for investor protection; support for a truthful and competent financial services sector and industry; control and monitoring of systemic risk, supervision of management risk in deregulation firms, and the assurance of consistency of deregulation with international standards, and to support the government's is competitive position in the financial market" (Respondent L).

Interviewees were further required to explain concerns relating to the deregulation framework of the Libyan economy. Respondent F indicated that, firstly, this framework had allowed foreign products and investments to penetrate the Libyan market which, in turn, caused a decline in the companies' domination of the market in respect of their products. Secondly, it had terminated the special foreign currencies exchange rates assigned to the state-owned companies, thereby causing a fivefold increase in the cost of materials and foreign labour within a year (see Subsection 7.4.2.2: Exchange Rate). Consequently, product prices have increased, which has placed both companies in a weaker competitive position and caused

deterioration in their financial performance. Many managers gave specific responses, however, regarding deregulation and the legal framework governing the work of the basic components operating in the financial market, brokers, shareholders and bonds.

7.4.1.2 Corporatisation

Questionnaire Findings

Respondents to the questionnaires were presented with six questions concerning potential corporatisation processes and were requested to rate degrees of concern, in relation to the economic reform programme, on a 5-point rating scale. The corporatisation factors all had a *mean* score of 4 or above which contributed to development of the stock market by increasing the number of private companies as presented in Table 7.3. In descending rank order of concern the results are: 'Managers of corporate companies play a crucial role in encouraging prospective customers to invest in the market-based economy' (mean 4.24, Skewness -1.46, Kurtosis 3.12), as during the economic transition capital structure, decisions are governed by managers' preferences, perceptions and beliefs towards external finance as business develops and more funds are needed (Michaelas, 1998). The second highest ranking is: 'The corporatisation of companies plays a vital role in contributing to the stock market' (mean 4.22, Skewness -1.70, Kurtosis 3.31). This finding mirrors that in 1992, when the government passed Act number 9 of 1992 to enhance and regulate the private sector activities in the nation. The Act permits the establishment of private business activities owned and managed by families and individual entrepreneurs. It also allows the selling of publicly-held companies to private investors, which has resulted in the emergence of some private companies. According to this Act, the state needs to transfer its role as sole owner to that of a shareholder with limited liability and limited responsibility in order to fully privatise the state-owned companies. This was followed by statements concerning the importance of different strategic targets for the future of the firm. The highest mean was for 'Providing job opportunities' (mean 4.17, Skewness -1.47, Kurtosis 2.56) which was followed by 'Increasing profitability' (mean 4.16, Skewness -1.61, Kurtosis 2.85). Bevan and Danbolt (2002) explained this finding in that the more profitable firms should hold less debt, because high levels of profits provide a high level of internal funds. The average mean in the whole section is 4.14, which indicates overall agreement with the corporatisation statements.

Code	Donk	Statements		Vali	d N=203	
Coue	Kalik	Statements	Means ^a	SD	Skewness	Kurtosis
B.9	3	Corporatisation is an important factor in a market-based economy	4.1478	0.94807	-1.462	2.316
B.10	2	The corporatisation of companies plays a vital role in contributing to the stock market	4.2217	0.93614	-1.700	3.314
B.11	5	Public sector assigns laws for endorsing investment in the stock market	4.0148	1.03157	-1.150	0.853
B.12	4	The Libyan government encourages corporate companies to invest managers in the stock market	4.1133	0.98603	-1.450	2.058
B.13	1	Managers of corporate companies play a crucial role in encouraging prospective customers to invest in the market-based economy	4.2414	0.81803	-1.459	3.119
B.1	14	The following strategic targets are important for the future of the firm:				
B.14.1	1	Providing job opportunities	4.1724	0.90915	-1.466	2.561
B.14.2	5	Providing domestic market goods and services	4.0345	1.04066	-1.320	1.471
B.14.3	2	Increasing profitability	4.1626	0.96379	-1.605	2.853
B.14.4	4	Repaying borrowing	4.1478	0.92160	-1.372	2.161
B.14.5	3	Expanding the firm	4.0936	1.04176	-1.436	1.766

Table 7.3 Descriptive Statistics for Corporatisation

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree".

Source: Analysis of questionnaire survey data.

Interview Findings

Respondents interviewed in the companies were requested to indicate techniques they most relied on in corporatisation programmes involved in their economic reform. As interviewees were not required to tick or rate a pre-formulated written set of answers, the relevant measure was the number of instances where any given technique was mentioned. Much of the evidence from the interviewees supports the findings of the questionnaire. One financial manager explained that the main aim behind the introduction of the corporatisation programme was to encourage the workers to take control in both the state-owned and private enterprises in which they worked. Based on this encouragement, the ownership of many private companies was changed in 1979 such that they became state-owned. Respondent H stated that, "By the end of 1981, with the exception of the agriculture sector, all private ownership in Libya was abolished, housing ownership was restricted to one house per family, business enterprises were replaced by government agencies, and workers took over private and public factories". Despite this evidence supporting the questionnaire data, there were some responses which contradicted the argument that the running of enterprises by workers made effective management almost impossible and enterprises had no clear role in the economy, since workers' committees rarely accepted economic efficiency or profitability as valid objectives Respondent I.

7.4.1.3 Privatisation

Questionnaire Findings

Questionnaire respondents were presented with a list of ten questions concerning the potential privatisation processes, and requested, with reference to the economic reform programme, to

CHAPTER 7: PRESENTATION OF THE FINDINGS

rate degrees of concern on a 5-point rating scale. Some similarities based on the *mean* of the privatisation score, as presented in Table 7.4, occurred but generally the results varied between attracting high and moderate interest. In highest place was: 'The private sector can co-operate with stock market authorities to increase competitive advantages' (*mean* 4.24, *Skewness* -1.61, *Kurtosis* 2.99). The next highest rank was given to: 'The privatisation moves of the Libyan economy towards a stock market' (*mean* 4.24, *Skewness* -1.57, *Kurtosis* 3.25). This finding is in line with the view suggested by Bekart et al. (2002) who argued that the stock market can also play a significant role in the planned privatisation of the public sector by evaluating the stocks, providing hire purchase funding for the small investor and transforming part of the debt of public sector enterprises into equity shareholding. This will rectify financial structures and enable the promotion of these organisations' shares when they are offered for sale, as well as assisting in the establishment of investment funds in stock and widening ownership of shares.

The third highest ranking was: 'Privatisation transforms the economy from a socialist to a market economy' (mean 4.22, Skewness -1.57, Kurtosis 2.99). By way of illustration, the General People's Committee resolution number 300 in 1993 recommends that joint enterprises could be sold to the private sector. During 1993 and 1997, 124 state-owned business units were privatised, about 10,250 collective-ownership companies were established in different sectors (Alsharif, 2002) and many companies were liquidated or merged with other companies (Alsharif, 2000). Alfaitori (2004) describes the state's role at present and in the forthcoming stage as being to transfer the state-owned sector into the private sector, motivate the private sector and monitor the overall economy. He demonstrates that 361 state-owned enterprises were listed for privatisation before 2008 in a three-stage-plan (261 enterprises by the end of 2005, 46 by the end of 2007, and 54 by the end of 2008). However, Binyon (2005) reports that only 48 state-owned firms were privatised in 2004 out of the 360 listed for privatisation before 2008. These findings suggest that, first, privatisation is expected to reduce the financial burden on the government budget caused by regular subsidies to public sector enterprises. By releasing scarce financial resources, the government will be able to increase investment allocations to sectors that directly benefit the poor, such as education, nutrition and health. Secondly, privatisation will improve the efficiency of public sector enterprises, which have not only been performing poorly, but have also been highly resistant to change despite almost continuous reform efforts. Finally, privatisation will bring in the investment resources needed for the modernisation of the public sector. An overall average of 4.09 is recorded indicating that the respondents agreed with the privatisation programme statements.

Code	Donk	Statements		Valid	N=203	
Coue	Nalik	Statements	Means ^a	SD	Skewness	Kurtosis
B.15	5	Private sector is important in strengthening the corporate market	4.0887	1.06335	-1.426	1.569
B.16	7	Privatisation is a new term in the stock market	4.0493	1.11139	-1.278	0.978
B.17	2	Privatisation moves the Libyan economy towards a stock market	4.2365	0.85784	-1.569	3.254
B.18	3	Privatisation transforms the economy from a socialist to a market economy	4.2217	0.89836	-1.570	2.986
B.19	10	Privatisation of commercial banking sector	3.8522	1.17643	-1.073	0.258
B.20	9	Your company encourages you to borrow money for a business deal as long as it is profitable	3.9015	1.00992	-1.053	0.732
B.21	4	The Libyan government motivates the private sector to invest in the stock market	4.1675	0.91820	-1.424	2.342
B.22	6	Privatisation is the real reason for contribution of productivity growth	4.1576	0.93065	-1.473	2.527
B.23	8	Technology tools such as internet telecommunications are crucial in the privatised system	4.0099	1.08541	-1.286	1.139
B.24	1	The private sector can co-operate with stock market authorities to increase competitive advantages	4.2414	0.90972	-1.613	2.989

Fable 7.4 Descriptive	Statistics for	Privatisation
-----------------------	----------------	---------------

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree".

Source: Analysis of questionnaire survey data.

Interview Findings

Respondents interviewed within companies were asked to indicate the techniques they most relied upon in the privatisation programme involved in their economic reform. Since interviewees were not given a pre-formulated written set of answers, the relevant measure was the number of times any particular use was mentioned. Respondent G indicated that the Libyan government planned to privatise 361 companies by the end of 2007, and that to date more than 260 companies, with a value greater than 2 billion LD, had gone through the privatisation process. However, the government introduced significant companies, such as cement, iron and steel mills, feed, Libyan insurance and five new private commercial banks: Aman Bank for Commerce and Investment, Alijmae Alarabi Bank, Wafa Bank, Waha Bank (owned by the Sahil and Sahara Bank) and Weheda Bank. Nonetheless, the owners of these institutions were witnessing a reluctance to purchase their shares by citizens (Respondents E and H). Presently, the private sector constitutes about 35 per cent of Libyan economic activities (Respondent I). Libyan citizens paid a heavy price for the economic liberalisation policy because of the large gap between private and public sectors and the extent of the pervasive phenomenon of corruption (Respondent G). These findings suggest a need to: (1) facilitate the transformation from the public to the private sector by formulating appropriate policies to deal with the problems associated with this transformation; (2) motivate the private sector to invest in new projects as a means to decrease the number of unemployed people, (3) assist the process of reform by correcting certain policies and proceedings regarding protection, supervision and control.

7.4.1.4 Liberalisation

Questionnaire Findings

Questionnaire respondents were presented with five questions concerning potential liberalisation processes and requested to rate their degrees of concern regarding the economic reform programme on a 5-point rating scale. Based on the *means* of the liberalisation score, as presented in Table 7.5, the issues that concerned the respondents the most were: 'Liberalisation may affect specific components of capital flows including debt (financial and non-financial borrowing and lending), equity (portfolio) investment in stock markets) and foreign direct investment' (*mean* 4.17, *Skewness* -1.41, *Kurtosis* 2.23). The underlying argument for this reform is that the government has realised that greater opening of the stock market to foreign investors has many benefits³⁴, representing an opportunity to attract foreign capital to finance economic growth. By raising the demand for shares on the stock market, foreign investment liberalisation also lowers the cost of capital for local firms and adds to their incentives for going public which, in turn, makes the market more liquid and efficient and increases the market size. This, in turn, increases local investors' opportunities for portfolio diversification, which raises their incentive to invest in shares.

The second highest ranked concern was: 'Liberalisation restricts movement of product and factor flows across borders involving goods, services, investment, financial capital, technology and labour' (*mean* 4.16, *Skewness* -1.51, *Kurtosis* 3.39). As mentioned early in Chapter Two, during the 1970s and 1980s, the Libyan economy had been dominated by the state-owned sector. This domination had led to major economic crises that prompted the government to open the door to the private sector and start a policy of privatising a large number of state-owned companies (Bait-Elmal, 1999). Alsharif (2000) demonstrates that, since the late 1980s and early 1990s, a number of laws and resolutions have been issued, aiming to transform the Libyan economy from a centrally planned system to a more productive and flexible market-based economy by encouraging the private sector and reducing the role of the state to limited involvement in some public activities such as health, education and security, by privatising the state-owned companies and by giving priority to those projects that use domestic raw materials. An overall mean of 4.08 gives an indication that the whole section is agreed.

³⁴ For more a detailed discussion of the benefits of opening the stock market to foreign investors see, for instance, Bekaert and Harvey (2000); Bekaert et al. (2000); Hargis (2000) and Kim and Singal (2000).
Cala	Daula	Rank Statements		Valia	l N=203	
Code	капк		Means ^a	SD	Skewness	Kurtosis
B.25	5	Liberalisation of trade and investment regimes as well as deregulation and privatisation of government business activities generate opportunities for expansion of trade investment and technology flows	3.9606	1.08929	-1.151	0.819
B.26	2	Liberalisation restricts the movement of product and factor flows across borders involving goods, services, investment, financial capital, technology and labour	4.1645	0.95520	-1.512	2.391
B.27	3	Liberalisation has led to a lower cost of transactions and global business improvements in transport, communication and information technology networks	4.0936	1.06059	-1.370	1.467
B.28	4	Rapid liberalisation (dismantling and lowering of tariff/non-tariff barriers) and deregulation, both at national and international levels, aids trade	4.0345	1.06417	-1.364	1.475
B.29	1	Liberalisation may affect specific components of capital flows including debt (financial and non-financial borrowing, and lending), equity (portfolio) investment in stock markets), and foreign direct investment	4.1724	0.92535	-1.410	2.228

Table 7.5 Descriptive Statistics for Liberalisa	tion
---	------

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree".

Source: Analysis of questionnaire survey data.

According to the CBL, during the period 2001-2006, several steps related to monetary policy and aimed at achieving general price level stability and maintaining the banking system were taken. Beginning in 2002, after unification of the exchange rate, many procedures and operations in respect of monetary policy control were revealed as constraints on current operations and personal purposes related to foreign currency. Libya accepted the provisions of Article Eight of the IMF agreement related to liberalisation of the current account of operations and did not impose any restrictions upon it.

Interview Findings

Respondents interviewed in the liberalisation programme were requested to detail techniques involved in their economic reform. As they were not asked to tick or rate a pre-formulated written set of answers, the relevant measure was the number of times any given technique was mentioned. The financial and monetary authorities have several procedures and significant changes have had positive effects upon the whole economic arena in achieving vital steps for stability and payment of the liberalisation reform programme. These steps are: standardisation of the LD exchange rate and the consequent effect of correcting the distorted cost of many goods and services (Respondent A); the passing of new laws to regulate banking work, strengthening the role of the CBL and opening banking sector policy to domestic and foreign competition (Respondent E); simplification of procedures for customs tariffs, divided into two rates, 10 *per cent* for tobacco products and zero *per cent* for all other products, with subordination of all goods imported to new charges named "service fees" rated at 4 *per cent*. This fee or tax increased production and consumption of between 25 and 50 *per cent* on imported goods and reduced to 2 *per cent* for goods produced locally (Respondents C and D).

CHAPTER 7: PRESENTATION OF THE FINDINGS

To summarise, all the data which has been presented and discussed previously was collected by a questionnaire built on a 5-point Likert scale. The collected data identifies that the highest rank was gained in respect of corporatisation with a mean of 4.13, while the lowest level was recorded for liberalisation with a mean of 4.08 which could be a reasonable sign that there is a clear understanding of the value of economic reform programme factors. The finding is in line with the related literature. It can be argued that the economic reform programme was motivated by different factors, each of which has combined with the others, to initiate the need to develop the Libyan stock market. These findings show that, to a very great extent, the deregulation of the economy is considered as the most influential factor affecting the economic reform programme. Thereafter came other issues like the need for more economic liberalisation, the impact of other reform via privatisation through public share offerings which can also spur the development of stock markets, directly through higher equity valuations and trading liquidity and indirectly through increasing opportunities for portfolio diversification, accelerating the development of an institutional framework, and increasing investors' confidence.

7.4.2 Macro-economic Reform Independent Variables

This group covers the third part of the questionnaire survey (Appendix G: Section C) and some interview questions (Appendix K). The aim of this section is to *present* and *analyse* data related to macro-economic reform and financial conditions for developing a stock market within the framework of the Libyan economic context. These independent variables include interest rates, exchange rates, inflation rates, budget deficit, per-capita income and real GDP growth rate in Libya's economy.

7.4.2.1 Interest Rate

Questionnaire Findings

Respondents to the questionnaires were presented with five statements concerning the interest rate and were required to tick those which they felt were relevant to macro-economic reform in Libya. The findings are presented in Table 7.6, from which it can be seen that the statement: 'Interest rates affect the stock market' was ranked most relevant (*mean* 3.89, *Skewness* -1.01, *Kurtosis* 0.30) with the second highest being: 'Interest rate liberalisation' (*mean* 3.85, *Skewness* -0.96, *Kurtosis* 0.14). These findings may indicate the effect of Law no.16 in 2004 related to reducing the minimum interest rate on loans and credits granted for productive purposes from 7.0 to 3.0 *per cent*. Additionally, law no.15 in 2005 related to cutting the

interest rate given by the CBL on commercial bank deposits with the bank from 2.5 to 1.7 *per cent* in order to encourage them to seek other domestic investment and finance areas that aid the achievement of desired economic growth. In developing countries, as McKinnon (1973) argues, governments should use appropriate policy in the domestic capital market to achieve economic growth. Raising the rate of interest to equilibrium level is known to play an important role in improving quality and the quantity of investment.

The statement: 'The policy of interest rate' (*mean* 3.44, *Skewness* -0.38, *Kurtosis* -1.01) was ranked lowest and, therefore, policy was considered the least important source of interest rate factors of those listed. It can be argued that the policy of fixing interest rates suffered from a major problem, that the interest rates were set regardless of the rate of inflation prevailing in Libya and, in most cases, they were below the prevailing inflation rates. This led investors to look for a quick profit by investing in unproductive projects. As a result, the policy of investing in long-term to short-term projects, i.e. trade, real estate and the housing sector, emerged. In early 2006, the CBL took the first step towards liberalising the interest rate by floating interest rates on deposits. Overall, a mean of 3.73 is recorded, showing that there is agreement of the interest rates statements.

Cada	Donk	S Statements -	Valid N=203				
Code	канк		Means ^a	SD	Skewness	Kurtosis	
C.1	5	The policy of interest rate	3.4384	1.28198	-0.381	-1.009	
C.2	4	When interest rates increase, the required rates of return will increase, then stock prices will decrease	3.6946	1.22083	-0.748	-0.380	
C.3	1	Interest rates affect the stock market	3.8916	1.12491	-1.007	0.304	
C.4	3	The flow of money depends upon the interest rate either to the stock market or the banking sector	3.7980	1.17465	-0.859	-0.105	
C.5	2	Interest rate liberalisation	3.8473	1.16097	-0.963	0.140	

Table 7.6 Descriptive Statistics for Interest Rates

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree".

Source: Analysis of questionnaire survey data.

Interview Findings

Bank managers who were interviewed were also asked where they obtained the techniques involved in their macro-economic reform initiatives. As interviewees were not required to tick or rate a pre-formulated written set of answers, the relevant measure was the number of instances where any given technique was mentioned. The interview findings confirm those obtained from the questionnaires. The general manager of the Central Bank explained that partial liberalisation of the interest rate was considered as significant, where interest rates had been consolidated on deposits with CBL. In addition, there was the cancellation of multiple interest rates has been on deposits with commercial banks (Respondent E).

Interviewees were additionally required to disclose reasons for relying on particular techniques. There is much deregulation but, despite this, a number of laws have been introduced by the CBL related to interest rates. For instance, Resolution no.16, 2004 concerned a reduction in the maximum rate of interest on loans and credit facilities for productive purposes from 7.0 to 3.0 *per cent*, which represented citizens and encouraged the public sector to borrow from banks to finance their productivity (Respondent A). Resolution no.15 in 2005 related to cutting the interest rate granted by the CBL on commercial bank deposits from 2.5 to 1.75 *per cent*, urging them to seek other areas of investment and financing local aid to achieve desired economic growth (Respondent A). Resolution no.36 of 2005 related to the liberalisation of credit interest rates (on deposits) and allowed negotiations between banks and their customers (Respondent L). Finally, Respondents E and N concluded that the effect of Resolution no.39, 2005 related to the standardisation of interest rates across all areas of facilities and loans granted by commercial banks and the equivalent of rediscount rates of the CBL with a percentage of not more than 2.5 *per cent*.

7.4.2.2 Exchange Rate

Questionnaire Findings

Respondents to the questionnaire were presented with six questions concerning the exchange rate processes and were asked to rate degrees of concern on a 5-point rating scale. Based on the *means* of exchange rate scores, as presented in Table 7.7, the statement that most concerned respondents was: 'The exchange rate might alter investment in the local currency market compared with other markets' (*mean* 4.05, *Skewness* -1.35, *Kurtosis* 1.60). The CBL argued that the majority of the country's foreign exchange rate earnings stem from oil and gas sector sales, were denominated in US\$ and, were unaffected by local exchange rate movements. The Bank further asserted that any devaluation would have little effect on foreign exchange rate earnings. This devaluation would depress foreign investment and, therefore, make the Libyan markets unattractive. It can be argued that the CBL was right, as the devaluation in exchange rates has had a bad effect on the Asian and Russian crises, causing foreign investors to liquidate their investment. Aljerrah (1993) tried to find the optimal exchange rate method in some Middle Eastern countries. It was suggested that the economies of these countries could have enjoyed more advantages from the exchange rate system if they used the exchange rate regime.

The statement: 'It is important to reduce the fluctuations between domestic and foreign currency' was ranked second (*mean* 4.03, *Skewness* -1.33, *Kurtosis* 1.34) and third was:

'Exchange rate fluctuations have an influence on stock prices' (mean 3.96, Skewness -1.18, Kurtosis 0.65). These findings are in line with the view suggested by the General Manager of the Central Bank, who argued that, in early 1998, the CBL introduced a new policy for determining the Libyan Dinar exchange rate. Under this amendment priority was given to keeping the Dinar's exchange rate stable against the US dollar and allowing it to fluctuate against other foreign currencies. The CBL took this measure in order to inject greater transparency, to improve the stability of the Libyan Dinar exchange rate and to bolster the interest rate policy in its efforts to enhance the attractiveness of keeping assets denominated in Libya Dinar against those denominated in US dollars. It is important to note here that, during the period 1986-1995 and the de-stabilisation of the currency arising from the collapse of oil prices and the failure of oil revenues, it was decided to implement measures to safeguard the Libyan Dinar from the negative effects of fluctuations in the value of the US\$ and the whims of the international financial markets, resulting from the deficit in the balance of payments. In fact, the Libyan Dinar was pegged to the Special Drawing Rights (SDR). The average mean of the whole section is 3.94, which indicates overall agreement with the exchange rates statements.

Code	Donk	k Statements	Valid N=203				
Coue	Nalik		Means ^a	SD	Skewness	Kurtosis	
C.6	6	There are no distinct monetary policies in relation to the exchange rate policy	3.8227	1.19325	-0.835	-0.262	
C.7	4	The movement in exchange rate may change foreign investors' demand for domestic stock prices compared to other countries	3.9261	1.10778	-1.088	0.563	
C.8	2	It is important to reduce fluctuations between domestic and foreign currency	4.0345	1.07344	-1.330	1.336	
C.9	1	The exchange rate may influence investment in the local currency market compared with other markets	4.0542	0.98605	-1.350	1.599	
C.10	3	Exchange rate fluctuations have influence upon stock prices	3.9557	1.13583	-1.182	0.646	
C.11	5	The exchange rate plays a vital role in investing in the stock market	3.8867	1.13538	-1.025	0.340	

 Table 7.7 Descriptive Statistics for Exchange Rates

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree".

Source: Analysis of questionnaire survey data.

Interview Findings

Respondents among the bank managers interviewed were requested to indicate the techniques they most relied on in implementing the exchange rate policies involved in their macroeconomic reforms. Since interviewees were not provided with a pre-formulated written set of answers, the relevant measure was the number of times any given use was mentioned. Law no.49 of 2001 of the CBL's Board of Directors, which considered the equivalent value of the Libyan Dinar against foreign currencies, determined one Libyan Dinar as being equal to 0.608 units of the Special Drawing Rights (SDR) (Respondent A). Moreover, Law no.17 in 2003 decreed that 15 *per cent* of exchange rates related to the foreign exchange earnings and credits transfers to finance the Great Man Made River equated the Libyan Dinar with 0.5175 units of the SDR (Respondent A). This exchange rate is fixed against this unit, but fluctuates against other foreign currencies at the same exchange rate against the SDR unit (Respondent E). The senior general marketing managers interviewed were asked about the exchange rate which concerned them. After more than five years of operating with the currently used exchange rate, it was evident that this rate was suitable in its time for the Libyan economic conditions. They further emphasised the report of the consultations as confirmed by the IMF which visited Libya in 2005 and was discussed by the Fund's Board of Executive Directors in 2006.

7.4.2.3 Inflation Rate

Questionnaire Findings

Questionnaire respondents were presented with six questions about the inflation rate processes and requested to rate degrees of concern in respect of macro-economic performance, on a 5point rating scale. Based on the means of inflation rate score, as presented in Table 7.8, the statements that concerned respondents, in descending order of rank, were: 'Any increase in inflation rate reduces expected real returns to investors from holding stocks' (mean 4.17, Skewness -1.55, Kurtosis 2.52). Thereafter, there were no substantial differences between: 'Inflation increases the price of goods and services' (mean 4.03, Skewness -1.33, Kurtosis 1.11) and 'When the rate of inflation rises, stock prices fall due to the effect of inflation on the historic cost method of depreciation and the taxation of nominal capital gains' (mean 3.97, Skewness -1.13, Kurtosis 0.61). These findings are evidenced by this current study. During 1991 and 1992 the inflation rate increased dramatically from 9.4 per cent to 11.7 per cent, respectively, stemming from the devaluation of the Libyan Dinar and the consequent UN suspension of certain sanctions upon Libya in that period. Therefore, the control of inflation was the most significant success story within the past few years when the US lifted further sanctions and Libya adopted new economic reforms through a modern open door policy. The rate declined sharply from -2.9 per cent in 2000 to -2.2 per cent in 2004 (IMF, 2007).

In fact, it was crucial for the Libyan economy to depress the inflation rate in the first stage of the economic reform programme period. The finding is consistent with McKinnon's (1988) argument that economic stability is a vital condition for financial development to take place. Countries that suffer from high inflation are less likely to adopt successful financial development plans. Complementary to that, Cooley and Hansen (1989) and De Gregorio (1993) highlight the fact that higher inflation has the effect of reducing labour supply and, consequently, reducing growth. High rates of inflation reduce investment spending and cause a substitution from labour supply to leisure, which directly and negatively affects growth and

185

capital accumulation (Bittencourt, 2006). The average mean of the whole section is 3.96, which indicates overall agreement with the inflation rates statements.

Cada	Donk	Statements	Valid N=203					
Code	Kalik		Means ^a	SD	Skewness	Kurtosis		
C.12	3	When the rate of inflation rises, stock prices fall due to the effect of inflation on the historic cost method of depreciation and taxation of nominal capital gains	3.9704	1.10315	-1.125	0.608		
C.13	1	Any increase in the inflation rate reduces the expected real returns to investors from holding stocks	4.1724	0.96720	-1.546	2.515		
C.14	6	The inflation rate impacts strongly on social life	3.8177	1.15668	-0.859	-0.056		
C.15	2	Inflation increases the price of goods and services	4.0296	1.11653	-1.330	1.111		
C.16	4	Inflation decreases currency value	3.9458	1.12224	-1.102	0.514		
C.17	5	Inflation causes unemployment and reduction of job opportunities	3.8424	1.15817	-0.924	0.034		
Notice	a: T	he percentage of respondents who scored 4 "agree" and 5	"strongly	agree".				

 Table 7.8 Descriptive Statistics for Inflation Rates

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree".

Source: Analysis of questionnaire survey data.

Interview Findings

Interviewees were additionally asked about concerns regarding inflation rates in which their macro-economic reform operated. Since they were not required to rate or tick a pre-formulated written set of answers, the relevant measure was the number of times any given information was mentioned. Much of the evidence from the interviews supports the findings of the questionnaire. Respondent M spoke about inflation saying, "*This issue is highly significant at this particular time and has become a problem for limited income and inflation. The continuing rise in prices and low purchasing power has made this a crucial issue*".

Respondent E said that many nations suffer from the problem of high prices due to the economies of either debt or production and that it is a problem encountered by many global economies. However, in Libya there is no inflation within the national market. The problem is not the economic policies, but that there is no proper co-ordination within economic policy, meaning that monetary policy and fiscal policy, customs' policy and wage policy are not linked. This result suggests that if there is no increase in production, prices rise automatically and this results in depressed salaries. Despite this evidence supporting the questionnaire data, there were some responses which disagreed. However, there is no co-ordination between the CBL policy and the Secretariat of Treasury and Customs. Each refuses to agree upon a certain policy, and hence, it is the citizens who are affected. Here we are discussing the rise in money supply within the Libyan market (Respondent A). Two key reasons for not relying on the inflation rate were mentioned by interviewees: the price increase is due primarily to the exchange rate; the internal policies in the domestic market have a critical relationship with the level of salaries. This has played a significant role in the increase in costs of some commodities including food and building materials, which rose suddenly (Respondents E, J and M).

7.4.2.4 Budget Deficit

Questionnaire Findings

Questionnaire respondents were presented with a list of six questions concerning budget deficits processes. They were asked to rate degrees of concern about the macro-economic reform, on a 5-point rating scale. Based on the mean of the budget deficits score, as presented in Table 7.9, the statement about budget deficit issues that concerned respondents the most was: 'Budget deficit is considered to be the excess of government expenditure over time' (mean 3.01, Skewness -1.25, Kurtosis 0.85). The interviews support this result, three of the fourteen interviewees considering that rates of increase in total expenditure reached high proportions, hitting more than 45 per cent in 2001, falling to 25.2 per cent in 2003, then rising to more than 29 per cent in 2006 before returning to stabilise at 6.3 per cent in 2005 (Respondent K). Respondents A and G also indicated that, despite this, increases were not reflected in per-capita public expenditure in the real sense that these increases in public spending are apparent where the per-capita public expenditure in 2002 amounted to about 1068.6 LD at a lower total 14.7 per cent of its level in 2001. It fell by 17.3 per cent in 2005 from the level reached in 2004. This is an indication that, despite increased spending levels during the period 2000-2005, this has not been reflected in the increase of goods and services for individuals.

The next highest ranked statement was: 'The exploitation of private sector tools contributing to reduction of the budget deficit' (*mean* 3.87, *Skewness* -0.95, *Kurtosis* 0.15). A possible explanation could be that most private firms are small and their top managements play a key role, controlling most of the operational and managerial activities. The third highest ranking statement was: 'A low value in budget deficit reduces government spending' (*mean* 3.73, *Skewness* -0.79, *Kurtosis* -0.20). To achieve this, many developing countries, i.e. Bulgaria, Egypt, Hungary, Jordan, Turkey and Latin America, began their economic reform programmes by cutting or reducing both subsidies and investment and, at the same time, those countries used taxes as an instrument to raise their revenues. Although these methods have a bad effect on the investment climate, this can be considered an appropriate procedure in the short-term until stability in the government budget is achieved. Empirically, Heitman (1969) attempted to analyse the effect of oil income on the Libyan economy before the oil embargo. He concluded that government expenditure depends heavily on oil revenues and oil exports. An overall average of 3.75 is recorded indicating that the respondents agreed with the budget deficit statements.

187

Cada	Donk	Statements	Valid N=203				
Coue	Kalik		Means ^a	SD	Skewness	Kurtosis	
C.18	1	Budget deficit is considered to be the excess of government expenditure over time	3.9951	1.11470	-1.246	0.853	
C.19	5	Reducing the budget deficit overcomes unemployment problems	3.6601	1.19725	-0.664	-0.474	
C.20	4	Budget deficit prohibits the increase of public resources	3.6847	1.19368	-0.710	-0.398	
C.21	6	Budget deficit can lead to increased inflation and interest rates	3.5862	1.25320	-0.624	-0.644	
C.22	3	A low value in budget deficit reduces government spending	3.7291	1.16487	-0.785	-0.203	
C.23	2	The exploitation of private sector tools contributes to reduction of the budget deficit	3.8719	1.12283	-0.953	0.146	

Table 7.9 Descriptive Statistics Budget Deficit

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree".

Source: Analysis of questionnaire survey data.

Interview Findings

Interviewees were additionally asked about concerns regarding the budget deficit and were requested to indicate the techniques upon which they relied the most in budget deficit decisions involved in their macro-economic reforms. Since interviewees were not given a preformulated written set of answers, the relevant measure was the number of instances any given technique was mentioned. A public sector manager stated, "During the period 2000-2007, oil revenues represented more than 79.5 per cent of budget revenues and, therefore, the proportion of non-oil revenues to cover total public expenditure of about 21 per cent during the same period, which rendered non-oil deficit greater than 32 per cent of GDP and the average for the period 2000-2007". In addition, the financial manager and accounting disclosure manager claimed that the general situation of the public budget can be observed as the continuation of the budget deficit during the period 2000-2006. The highest rates reached in excess of 2 billion LD. Furthermore, non-oil revenues cover averaged 19.7 per cent of the total expenditure. This percentage amounted to a minimum 9.6 per cent in 2003 and reached 12.6 per cent in 2006. In this regard, Respondent N stated that the government authority adopted a fiscal policy regarding management of aggregate demand via a reduction of public spending and the continued policy of salaried workers' organisations, secretariats and intensifying tax collection efforts in various ways, with some adjustments to tax legislation and tax administration reform. In spite of this, there remain outbreaks of tax evasion not keeping pace with developments within contemporary tax administration and IT technical fields.

7.4.2.5 Per-capita Income

Questionnaire Findings

Questionnaire respondents were presented with a list of five statements relating to per-capita income processes and asked to rate their degrees of concern with reference to the macro-economic reform, on a 5-point rating scale. Based on the *mean* of the per-capita income score,

as presented in Table 7.10, the per-capita income statements that concerned respondents the most were, firstly: 'Per-capita income is one of the driving forces towards a stock market' (mean 3.83, Skewness -0.84, Kurtosis -0.40), a finding that is consistent with Goldsmith (1969), Shaw (1973) and Mckinnon (1973) who all found that development of financial markets was significantly correlated with the level of per-capita income. The next highest rank was the statement: 'Per-capita income is reflected in the development of the Libyan economy' (mean 3.81, Skewness -0.88, Kurtosis -0.04). The national economy finding, oil discovery and exportation moved the country to the forefront of world economies. The standard of living rose as a result of the increasing number of foreign oil companies and the establishment of new Libyan economic activities. The average income per capita was 20 LD per annum before 1950, about 100 LD in 1960, 600 LD in 1970 and 8,000 LD in 1984, while the development expenditure from 1970 to 1984 was equivalent to 18.5 billion LD, or 62.5 billion US\$ (Agnaia, 1996). By the end of the 1980s the Libyan economy was near to collapse. In turn, this caused the per-capita income to be placed under fiscal discipline with structural and economic reforms including the investor-friendly laws of the 1990s which contributed towards a satisfactory recovery in per-capita GDP. Indeed, per-capita income increased from 59,339 US\$ in 1999 to 7,589 US\$ in 2006, which was attributed to improved performance in the construction, trade and financial services sector (IMF, 2007). This finding suggests that apart from experiencing a huge alternation to the market system mechanism, the developing economy of Libya, has the distinctive characteristic of being reliant, to a very great extent, on the oil industry as the main source of income of the economy.

The third highest ranking was the statement: 'The human development index is strongly affected by per-capita income' (*mean* 3.73, *Skewness* -0.73, *Kurtosis* -0.59). This finding is in contradiction to that reported by the UNDP (2001: 13), that pointed out that, "*Rankings by HDI and by GDP per capita can be quite different, showing that countries do not have to wait for economic prosperity to make progress in human development [...]. Costa Rica and Korea have both made impressive human development gains, reflected in HDIs of more than 0.800, but Costa Rica has achieved this human outcome with only half the income of Korea. Pakistan and Vietnam have similar incomes, but Vietnam has done much more in translating that income into human development [...]. So, with the right policies, countries can advance faster in human development than economic growth." This tends to support the view of those who argue that per capita income is a reasonable proxy for ranking nations as their relative and absolute level of development. Overall, a mean of 3.73 is recorded, showing that there is agreement on the per-capita income statements.*

Code	Donk	Statemente	Valid N=203				
Coue	Nalik	Statements	Means ^a	SD	Skewness	Kurtosis	
C.24	2	Per-capita income can be reflected in development of the Libyan economy	3.8128	1.11446	-0.881	-0.035	
C.25	1	Per-capita income is one of the driving forces towards a stock market	3.8325	1.19881	-0.839	-0.395	
C.26	3	The human development index is strongly affected by per-capita income	3.7340	1.21385	-0.734	-0.590	
C.27	4	The per-capita income of individuals depends upon growth population	3.6847	1.08048	-0.652	-0.479	
C.28	5	Fiscal discipline determines the per-capita income that satisfies GDP growth rate	3.6059	1.13555	-0.634	-0.531	

Table 7.10 Descriptive Statistics for Per-capita Income

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree".

Source: Analysis of questionnaire survey data.

Interview Findings

Interviewees were asked from where they obtained technical information about per-capita income in which their macro-economic reforms operate. Since they were not required to rate or tick a pre-formulated written set of answers, the relevant measure was the number of times any given aspect was mentioned. Respondent L said that the Libyan government has worked on many of the reforms that would increase citizens' incomes where the law has reconsidered salaries which are associated with Law no.15 in 1981. This resulted in an increase of 50 per cent and is expected to increase an employee's salary to more than 600 LD per month. In addition, the government gradually removed their support for commodity supply (Respondents G and M). Moreover, the Libyan authorities worked to take action to regulate foreign labour, which will assist in increasing national employment by about 14 per cent and obliges foreign companies wishing to operate in Libya to make major investments (Respondent C). Interviewees were further required to explain concerns of per-capita income. Many had difficulty in responding, however, having witnessed the extremely competitive global companies' investment in Libya in 2007, especially in the areas of oil and gas, where there has been intense competition following the arrival of major American and European companies to compete alongside Asian entities (Respondent D).

7.4.2.6 Real GDP Growth Rate

Questionnaire Findings

Respondents to the questionnaires were presented with a list of eight statements concerned with real GDP growth rate processes and, were asked to indicate their degrees of concern with reference to macro-economic reform on a 5-point scale. Based on the *mean* of the real GDP growth rate score, as presented in Table 7.11, the statements about real GDP growth rates that concerned respondents the most were, firstly: 'Oil prices impact on GDP of the stock market' (*mean* 3.88, *Skewness* -1.05, *Kurtosis* 0.64). This finding is consistent with the conclusion of Abbas (1987) that oil prices increased from 2.3 US\$ per barrel in 1969 to 25 US\$ per barrel in

1979 and, after this increase in oil price, as stated by Giurnaz (1985), Libyan oil revenues increased from 2.4 billion LD to about 6.5 billion LD by 1980. The GDP growth rate was 38.8 *per cent* in 1980. This explanation suggests that this situation gave the government the ability to increase spending on development in all sectors. More data available (see Chapter Two: Subsection 2.3.3 and Appendix A, Table A 3.12) shows the GDP growth rate by type of economic activity.

The next highest rank was given to the statement: 'Higher GDP growth leads to an increase in companies' cash flows' (*mean* 3.74, *Skewness* -0.86, *Kurtosis* -0.27). Similar results can be observed for the large companies as these may also have lower risk through diversification, more stable cash flows and established operating and credit histories. These factors provide large firms with greater access to alternative sources of finance in times of financial distress and may encourage them to take on relatively high debt burdens. In the third highest ranking there was some similarity among the means for two statements, these being: 'New improved and varied products, goods and services which increased in the economic growth rate' and 'Growth in GDP reduces unemployment' (*mean* 3.69, *Skewness* -0.67, *Kurtosis* -0.51). This follows the work by Levine and Zervos (1993), Atje and Jovanovic (1993), Levine and Zervos (1998), Roussean and Wachtel (2000), and Beck and Levine (2003), which shows that stock market development is strongly correlated with growth rates of real GDP per capita. The overall mean of the section was 3.63, indicating agreement with the statements regarding continuous improvement in the economic growth rate.

Cada	Donk	Statements	Valid N=203				
Code	капк		Means ^a	SD	Skewness	Kurtosis	
C.29	2	Higher GDP growth leads to increase in companies' cash flows	3.7389	1.19231	-0.863	-0.266	
C.30	5	Real GDP growth rate is necessary to increase output	3.5419	1.23960	-0.554	-0.871	
C.31	6	When increasing the per-capita of GDP, the Libyan economy must be competitive in the stock market	3.4680	1.25970	-0.475	-0.976	
C.32	4	There is a strong growth rate in export and import trade	3.6601	1.17639	-0.657	-0.628	
C.33	1	Oil prices impact on GDP of the stock market	3.8818	1.03685	-1.052	0.638	
C.34	3	New improved and varied products, goods and services which increased in the economic growth rate	3.6897	1.11126	-0.673	-0.511	
C.35	3	Growth in GDP reduces unemployment	3.6897	1.11126	-0.673	-0.511	
C.36	7	Investment return growth is high in their economy	3.3596	1.28731	-0.346	-1.110	

 Table 7.11 Descriptive Statistics for Real GDP Growth Rate

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree".

Source: Analysis of questionnaire survey data.

Interview Findings

Interviewees were asked from where they obtained technical information related to the real GDP growth rate in which their macro-economic reform operates. Since interviewees were not required to rate or tick a pre-formulated written set of answers, the relevant measure was the number of times any given factor was mentioned. The Director of the Economic Research

CHAPTER 7: PRESENTATION OF THE FINDINGS

Centre stated, "During 2007 the Libyan economy increased by more than 5.6 per cent following the largest economic and administrative reform programme by the government during its five-year plan. Development plans at all levels exceeded some 350 billion US\$ in infrastructure, housing, airports and railways". This results from a continued increase in world oil prices upon which 95 per cent of the Libyan economy was dependent (Respondent M). This was further explained by Respondent C who said that Libyan oil wealth ranked 5th among Arab countries and 9th globally, with an estimated 42 billion barrels annually, and that it was aimed, in the coming years, to increase this to 100 billion barrels. In addition, the wealth amounts to 1,672 billion cubic metres, ranking as 8th among Arab states and 24th internationally. Oil company managers were asked which oil and gas production aspects most concerned them. Their responses revealed that rate of production is currently 1.7 million with the intention of attaining 2 million barrels by the end of 2008, and then 3 million barrels by the end of 2012, as it seeks through exploration to attract new investments in this area, which so far exceeds more than 60 billion US\$ and gas production of about 45 trillion square metres.

Based on this discussion, the collected data identifies that the highest ranking statement relates to the inflation rate, which has a mean of 3.96, while the lowest ranking was recorded for real GDP growth rate with a mean of 3.63 which could be a reasonable sign that there is a clear understanding of the value of macro-economic reform factors. Respondents further indicated that interest rates affect the stock market and that liberalisation of the interest rate also has an influence, but to a lesser extent. Exchange rates which may alter investment in the local currency market compared with other markets are considered a significant factor of macro-economic reform. Most respondents felt that any increase in inflation rate reduces expected real returns to investors from holding stocks, and that when inflation increases, so too does the price of goods and services. Additionally, the responses indicate that budget deficit is considered to be the excess of government expenditure over time, and that per-capita income is one of the driving forces towards a stock market, as also are oil prices which impact on stock market performance. The analysis provided evidence to suggest that the most important strategy is to increase the rate of economic growth of output and to make a strong contribution to capital and labour.

7.4.3 Financial Maturity Independent Variables

This group covers the fourth part of the questionnaire survey (Appendix G: Section D) and some questions from the interviews (Appendix K). The aim of this section is to *present* and *analyse* data related to financial maturity as the financial sector is very important for capital

CHAPTER 7: PRESENTATION OF THE FINDINGS

allocation and financial intermediation. Transformation of savings into investments includes risk sharing, as reforms in the financial sector were a key element of Libya's structural reform programme. The relationship between banking intermediation and that performed by stock markets is becoming increasingly observable within the framework of the Libyan financial development context. These independent variables include ratio of banking assets to GDP (banking assets ratio), ratio of domestic credit to GDP (domestic credit ratio) and ratio of stock market capitalisation of GDP (stock market capitalisation).

7.4.3.1 Banking Assets Ratio

Questionnaire Findings

Respondents to the questionnaires were presented with four statements about the ratio of banking assets to GDP processes and were asked to rate degrees of concern relating to financial maturity, on a 5-point rating scale. Based on the *means* of banking assets to GDP ratio score, as presented in Table 7.12, the statement concerning respondents the most was: 'Financial intermediaries get larger measured by total assets relative to GDP' (*mean* 3.92, *Skewness* -0.01, *Kurtosis* 0.35). This ratio gives evidence of the importance of the financial service performed by the banking sector relative to the size of the economy. It has been reported in the literature that businesses will have different financial needs as they develop and they will increasingly rely on more funds from the banks (Michaelas, 1998; Churchill and Lewis, 1983). Empirically, Demirqüe-Kunt and Levine (1996) have found that most stock market indicators are highly correlated with the financial intermediaries (banks, insurance companies and pension funds). This finding suggests that countries with well-developed stock markets tend to have well-developed financial intermediaries.

The second highest ranked concern was expressed in the statement: 'The low ratio of broad money to GDP indicates a low monetary depth within the economy' (*mean* 3.85, *Skewness* - 0.97, *Kurtosis* 0.05) while 'Banking assets of GDP can improve growth rate in the stock market' was the statement recording the third highest rank: (*mean* 3.83, *Skewness* -0.91, *Kurtosis* 0.16). In addition, the Financial Times newspaper (1996) in one of its studies, indicated that liquidity within the stock market encourages banks to lend, giving investment a further boost. This, in fact, is especially important in poor countries since successful stock markets can dramatically improve the information that is available about companies, which ought to give creditors confidence. That may be one reason why the ratio of bank loans to GDP tends to rise as the level of investment increases and this should lead to raising the

economic growth. The overall mean of 3.85 indicates respondents' agreement to the statements in relation with the banking assets ratio of GDP.

Code	Rank	k Statements	Valid N=203				
Coue			Means ^a	SD	Skewness	Kurtosis	
D.1	4	Deposit money bank assets/total banking sector assets	3.8030	1.17759	-0.842	-0.197	
D.2	3	Banking assets of GDP can improve growth rate in the stock market	3.8325	1.11760	-0.911	0.162	
D.3	1	Financial intermediaries get larger measured by total assets relative to GDP	3.9212	1.10521	-0.998	0.352	
D.4	2	The low ratio of broad money to GDP indicates low monetary depth of the economy	3.8473	1.17790	-0.966	0.051	
Notice a: The percentage of respondents who scored 4 "agree" and 5 "strong			rongly ag	rree".			

Table 7.12 Descriptive Statistics for Banking Assets Ratio of GDP

Source: Analysis of questionnaire survey data.

Interview Findings

The main reason for not relying on the banking assets ratio was mentioned by interviewees as being the ratio of banking assets to GDP, confirming that levels of financial intermediation in the Libyan economy are comparatively low in relation to economic activity (Respondents A and E). However, the higher ratio of banking assets to GDP is not evidence of a healthier banking system (Stanislav, 2004).

7.4.3.2 Domestic Credit Ratio

Questionnaire Findings

Respondents were presented with nine statements related to the ratio of domestic credit to GDP processes and were requested to rate their degree of concern in relation to the financial maturity, on a 5-point rating scale. Based on the *means* of domestic credit to GDP score, as presented in Table 7.13, the associated statement that concerned respondents the most was: 'Domestic credit of GDP is one of the measurements of financial maturity of the stock market systems' (*mean* 3.73, *Skewness* -0.84, *Kurtosis* -0.22). In order for banks to fund investors, investors must fulfil at least two kinds of covenant: they must meet minimum debt-service coverage ratios and meet prudent bank-stipulated "debt-equity" ratios. General Managers of the Central Bank and the stock market suggest that firms prefer to finance their investments using debt since the cost of debt (due to its tax deductibility) is lower than the cost of equity. However, debt cannot exist as a replacement for equity beyond a certain level of leverage since lenders want the collateral of equity as a cushion. More equity, therefore, allows for more debt.

The following statement was ranked highest by the respondents: 'There is limited access to loans and credit facilities' (*mean* 3.67, *Skewness* -0.75, *Kurtosis* -0.31). This was followed by statements concerning the problems with lenders regarding loans or overdraft facilities in

Libvan companies³⁵. Some differences based on the public and private sector emerged. Private companies, for instance, seem to have more problems with lenders regarding loans or overdraft facilities than public companies. Manufacturing companies have fewer problems with lenders than non-manufacturing ones, while there is little difference between smaller and larger companies about facing problems with lenders regarding loans or overdraft facilities. Smaller companies are more influenced by red tape, banking charges and the rejection of loan applications than are large companies, whereas the large companies are more affected by the interest rate than smaller companies. The average mean of the whole section is 3.72, which indicates overall agreement with the statements regarding the domestic credit ratio of GDP.

Code D.5 D.6.1 D.6.2 D.6.3 D.6.4 D.6.5 D.6.6 D.7 D.8	Dank	Statements		Valid	N=203	
	канк		Means ^a	SD	Skewness	Kurtosis
D.5	2	There is limited access to loans and credit facilities	3.6749	1.15300	-0.747	-0.311
D	.6	Have you experienced any of the following problems?				
D.6.1	4	Loan application rejected	3.7143	1.17614	-0.792	-0.212
D.6.2	2	Charges too high	3.8177	1.15239	-0.931	0.047
D.6.3	5	Relationship difficulties	3.6897	1.12454	-0.651	-0.355
D.6.4	6	Bank errors	3.6404	1.17049	-0.633	-0.445
D.6.5	1	Red tape (Bureaucracy)	3.8719	1.11841	-1.031	0.355
D.6.6	3	Interest rates too high	3.7783	1.15832	-0.890	-0.010
D.7	1	Domestic credit of GDP is one of the measurements of financial maturity of the stock market systems	3.7340	1.16389	-0.837	-0.216
D.8	3	Domestic credit of GDP ability is considered to represent financial maturity in the stock market	3.5616	1.17730	-0.544	-0.749
Notic	e a.	The percentage of respondents who scored 4 "agree" and 5 "	'strongly a	oree"		

Table 7.13 Descriptive Statistics for Domestic Credit Ratio of GDP

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree".

Source: Analysis of questionnaire survey data.

Interview Findings

Interviewees were additionally asked how the domestic credit ratio influenced their decisionmaking processes. Since interviewees were not given a pre-formulated written set of answers, the relevant measure was the number of times any given use was mentioned. A decrease in the percentage of total bank credit to private sector and public institutions was mentioned, and to partnerships and individuals during the period 1995-2005, which was reported as dropping from 41 per cent in 1995 to 16.7 per cent in 2005. The total volume of credit increased from 4,372 billion LD in 1995 to 6,069 billion LD in 2000 and then rose to 9,358 billion LD in 2005. The reason was that the Libyan government adopted a policy of giving credits to various sectors in order to enhance and improve the social and economical situation of the country. Despite this evidence supporting the questionnaire data, there were some contradictory responses, indicating the weakness of investments, particularly if the increase in the apparent

³⁵ Keister (2000) argues that during an economic transition, the capital structure of companies might be affected due to the shortage of financing from the state. In this regard, Libyan companies may have to begin gradually to borrow from non-state capital sources in order to cover the shortage of financing from the state.

size of general liquidity within the economy, which exceeded 20,585 billion in 2006, is considered (Respondents E and M). Therefore, it can be argued that Libyan companies are more likely to be financed by debt over equity and especially short-term debt financing.

7.4.3.3 Stock Market Capitalisation Ratio

Questionnaire Findings

Respondents to the questionnaires were presented with four statements concerning the stock market capitalisation to GDP processes and requested to rate degrees of concern in relation to the financial maturity, on a 5-point rating scale. Based on the means of stock market capitalisation ratio score, as presented in Table 7.14, the statement that most concerned respondents was: 'Stock markets become larger, as measured by market capitalisation relative to GDP' (mean 3.70, Skewness -0.77, Kurtosis -0.38). This result suggests that, as stock markets become more developed and as imperfections in the capital markets are removed, corporations will have greater access to a broader range of financial instruments. This increased access would be expected to affect the financial policies of firms, thus affecting firms' investment, profitability and productivity growth, thereby establishing evidence about the relationship between the stock market development and firms' growth. The second highest ranked statement was: 'Stock market capitalisation of GDP represents one aspect of financial maturity in the market-based economy' (mean 3.69, Skewness -0.77, Kurtosis -0.45). In the same vein, Rousseau and Wachtel (2000) use two measures of stock market development: these being, the ratio of market capitalisation to GDP and the ratio of total value traded to GDP. Both have a positive coefficient, but only the latter is significant. The results show that the development of a liquid and highly capitalised equity market accelerates growth.

The statement: 'Independence of the Central Bank' (*mean* 3.37, *Skewness* -0.29, *Kurtosis* - 1.07) was ranked lowest and, therefore, this factor was considered the least significant source of stock market capitalisation of GDP of those listed. This response can be attributed to the fact that the CBL has made significant improvements in the supervisory system by instituting full financial disclosure by the banks. It has already adopted a number of measures to replace direct control with indirect monetary instruments in order to encourage competition among banks and, as a confidence-building measure removed the credit-to-deposit ratio, thus encouraging the evolution of an efficient inter-bank money market and resulting in the opinion of most respondents that the stock market should be independent and not controlled by the CBL (see Appendix K: Section K 4.1). An overall average of 3.59 is recorded, indicating that the respondents agreed with the stock market capitalisation ratio of GDP statements.

Cada	Donk	k Statements	Valid N=203					
Coue	Nalik		Means ^a	SD	Skewness	Kurtosis		
D.9	2	Stock market capitalisation of GDP represents one aspect of financial maturity in the market-based economy	3.6897	1.20117	-0.766	-0.451		
D.10	3	Stock market capitalisation of GDP increases investment in the stock market	3.5764	1.28504	-0.623	-0.801		
D.11	1	Stock markets become larger, as measured by market capitalisation relative to GDP	3.7044	1.16523	-0.770	-0.382		
D.12	4	Independence of the central bank	3.3695	1.27677	-0.289	-1.070		
Notic	Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree".							

Table 7.14 Descriptive Statistics for Stock Market Capitalisation Ratio of GDP

Notice a. The percentage of respondents who scored 4 agree and 5 strong

Source: Analysis of questionnaire survey data.

Interview Findings

Participants who were interviewed from within the marketing sector were requested to indicate the techniques they most relied on in stock market capitalisation. They were not required to tick or rate a pre-formulated written set of answers. The relevant measure was the number of times any given technique was mentioned. The findings from these interviewees, as depicted in Table 7.15, confirmed the questionnaire findings.

Table 7.15 Respondents' Interviews to Stock Market Development

Stock Market Development	Valid N=9 interviews ^a				
Stock Market Development	Number of Mentions	Per cent			
Market capitalisation ratio	8	88.9			
Share value traded	6	66.7			

Notice a: Details totalled more than N = 9 and 100 *per cent* due to duplicate responses.

Source: Analysis of the data obtained from semi-structured interviews.

The primary reasons for relying on stock market development were stated as market capitalisation ratio (Respondent B) and share value traded ratio defended by GDP (Respondent D).

To summarise, all the data which has been presented and discussed previously was collected by a questionnaire built on a 5-point Likert scale. Analysis of the results of the critical variables of financial maturity shows that the highest rank has been marked for banking assets ratio with a mean of 3.85, while the lowest level was recorded for stock market capitalisation ratio with a mean of 3.59 which could be a reasonable sign that there is a clear understanding of the value of financial maturity factors. The ratio of banking assets defended by GDP concerned respondents as a most important factor, since higher ratios of bank credit to GDP indicate better functioning of the country's banking sector and a growth in stock markets, as measured by market capitalisation relative to GDP. We argue that banking and the stock market are complimentary in the provision of financial development because they reflect the quality of bank credits, which is just as important as the size of bank credits.

7.4.4 Stock Market Performance Dependent Variables

This group covers the fifth part of the questionnaire survey (Appendix G: Section E) and some questions from the interviews (Appendix K). The aim of this section is to *present* and *analyse* data related to stock market performance within the framework of the Libyan economic context, where dependent variables include market (size, liquidity, activity, concentration).

7.4.4.1 Market Size

Questionnaire Findings

Respondents to the questionnaires were presented with five statements concerning market size processes and were asked to rate degrees of concern relating to stock market performance on a 5-point rating scale. Based on the *mean* of the market size score, as presented in Table 7.16, market size issues that concerned respondents the most were expressed firstly in the statement that: 'The number of listed companies is a critical factor for determining market size' (*mean* 3.79, *Skewness* -0.89, *Kurtosis* -0.18). This may indicate that the limited number of companies listed on the Libyan stock market is related to the fact that many Libyan companies remain family-owned businesses with no interest in going public. Their refusal to issue publicly-held equity is based on a fear that they will lose control and be forced to disclose what is believed to be personal information. The average size of the companies listed in stock markets³⁶ provides an idea about the number of shares available for trading. When small sized companies are listed on the stock market, this implies that a smaller number of shares are available for trading, which initially deters investors from entering the market and, at a later stage, when trading is active, results in violent price movements.

The next highest ranking statement was: 'The number of financial intermediaries increases the market size in the stock market' (*mean* 3.75, *Skewness* -0.77, *Kurtosis* -0.53). This finding was the most important result because the Libyan economy needs to develop and promote the creation of specialised financial institutions: pension funds, insurance companies and mutual funds. These institutions can play an important role in the modernisation of the stock market and the development of an efficient trading and settlement system. The third highest ranking statement was: 'The volume of shares listed enhances the stock market size' (*mean* 3.74, *Skewness* -0.83, *Kurtosis* -0.34), which refers to the total number of shares for all companies listed in the market. It is assumed that the size of the stock markets is positively correlated

³⁶ The average company size is calculated by dividing the total market capitalisation by the number of listed companies.

with the ability to mobilise capital and to diversify risk³⁷. The statement: 'Market capitalisation is considered to be a measure of market size in the stock market' (Mean 3.69, Skewness -0.79, Kurtosis -0.37) was ranked lowest and therefore market capitalisation was considered the least important source of market size of those listed. If the current research were to be repeated it may be better to change this question to give greater differentiation; for example, to ask respondents to rank the importance of the factors. The overall mean the section was 3.73, indicating agreement with the statements regarding continuous improvement in the companies.

Cada	Donk	Statements	Valid N=203				
Coue	Nalik	Statements		SD	Skewness	Kurtosis	
E.1	5	Market capitalisation is considered to be a measure of market size in the stock market	3.6946	1.17537	-0.790	-0.366	
E.2	4	Market capitalisation is a percentage of GDP in the stock market	3.6995	1.24811	-0.756	-0.549	
E.3	1	The number of listed companies is an important factor for determining market size	3.7882	1.18131	-0.891	-0.181	
E.4	3	The volume of shares listed enhances the stock market size	3.7438	1.21598	-0.832	-0.341	
E.5	2	The number of financial intermediaries increases the market size in the stock market	3.7537	1.22613	-0.772	-0.535	
Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree"							

 Table 7.16 Descriptive Statistics for Market Size

The percentage of respondents who scored 4 "agree" and 5 "strongly agree".

Source: Analysis of questionnaire survey data.

Interview Findings

Respondent interviewees in the stock market were asked about the market size which most concerned them. Since interviewees were not required to rank order or rate the size, the relevant measure was simply the number of times any given market size was mentioned. The interview findings, as depicted in Table 7.17, illustrate that the number of financial intermediaries was reported by nine interviewees, the number of listed companies was reported by eight and the volume of shares listed was reported by six interviewees.

Table 7.17 Respondents' interviews to Market Size							
Monkot Sigo	Valid N=12 interviews ^a						
Warket Size	Number of Mention	s Per cent					
Number of listed companies	8	66.7					
Volume of shares listed	6	50.0					
Number of financial intermediaries	9	75.0					
	10 1100	1 , 1 1 ,					

Table 7 17 D • to Marlad C:

Details totalled more than N = 12 and 100 per cent due to duplicate responses. Notice a:

Source: Analysis of the data obtained from semi-structured interviews.

Interviewees were further required to explain their concerns regarding market size. The Libyan stock market has witnessed an increasing improvement in the level of its performance, where

³⁷ Sirri and Tufano (1995) argue that by enhancing risk diversification, liquidity and size of feasible firms, resource allocation is improved. In addition, I argue here that the main impact of capital mobilisation is one of promoting technological innovation. Through easing the burden of risk to capital contributors as well as to the entrepreneurs, markets and institutions enable the undertaking of risky technological innovations.

the volume of shares listed reached about 4,942,533, as a total value of 254,543,455.000 LD in 2007. The number of financial intermediaries has risen from only two in 2006 to 14 in 2008, which means a 650 *per cent* increase. It can be argued that one reason for this is the large list of companies which implies a more developed stock market and that, in mature markets, it is relatively easy to list the companies on the exchange. Cumbersome listing rules (as are prevalent in undeveloped markets) deter many companies from listing.

7.4.4.2 Market Liquidity

Questionnaire Findings

Questionnaire respondents were asked to rate four statements concerned with market liquidity processes, on a 5-point rating scale. Based on the *mean* of the market liquidity score, as presented in Table 7.18, the statement that concerned respondents the most was: 'The total ratio of value-traded to GDP which impacts on the stock market' (*mean* 3.70, *Skewness* -0.81, *Kurtosis* -0.43). The next highest ranking statement was: 'Turnover ratios which are indicators of the market's ability to trade significant positions' (*mean* 3.67, *Skewness* -0.73, *Kurtosis* - 0.47). These findings may indicate that both the value-traded and turnover ratios are significantly correlated with economic growth. This implies that increases in the stock prices are not driving the liquidity results and this price effect does not influence the turnover ratio.

The third highest ranking statement was: 'The total value-traded in market capitalisation contributes to viability and accessibility of the stock market' (mean 3.58, Skewness -0.87, Kurtosis -0.55). Previous literature has revealed that stock markets in countries with betterdeveloped financial systems tend to grow faster than in countries with poor and relatively less liquid stock exchanges and weak financial and banking sectors (Levine, 2003). Nevertheless, empirical work by Levine and Zervos (1998) has demonstrated that the mere size of the stock market is not enough to foster economic growth. Simply listing privatised companies on the stock market is not enough to spur further real economic activity. It is rather liquidity of stock markets that fosters resource allocation and growth. The relatively low liquidity of the Libyan stock market compared with some emerging markets is primarily due to the following factors: (1) the current method of trading on the exchange: there are no markets and that tends to limit the size of trades that can be executed; (2) a substantial government's share portfolio, a significant portion of which is not traded; (3) the relative lack and under-development of institutional investors, e.g. pension funds, insurance companies and mutual funds. An overall average of 3.63 is recorded indicating that the respondents agreed with the market liquidity statements.

200

Code	Donk	Statamonta	Valid N=203				
Coue	Nalik	Statements	Means ^a	SD	Skewness	Kurtosis	
E.6	3	The total value-traded in market capitalisation contributes to viability and accessibility of the stock market	3.5813	1.34129	-0.816	-0.550	
E.7	1	The ratio of total value-traded to GDP impacts on the stock market	3.6995	1.24015	-0.811	-0.425	
E.8	4	The ratio of total value-traded shares helps in increasing volume of shares listed of market liquidity in the stock market	3.5714	1.34554	-0.819	-0.553	
E.9	2	Turnover ratios are indicators of the market's ability to trade significant positions	3.6650	1.19243	-0.725	-0.470	
Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree".							

Table 7.18 Descriptive Statistics for Market Liquidity

Source: Analysis of questionnaire survey data.

Interview Findings

Interviewees in the stock market were asked about the market liquidity aspects which most concerned them. As they were not required to rank order or rate the liquidity, the relevant measure was simply the number of times that any given market liquidity was mentioned. Total value traded to GDP, as reported in Table 7.19, headed the list and was reported by ten interviewees, followed by volume of shares traded to volume of shares listed that was reported by eight individuals and total value traded to market capitalisation reported by six interviewees.

 Table 7.19 Respondents' Interviews to Market Liquidity

Manhat Liquidity	Valid N=12 interviews ^a				
Market Liquidity	Number of Mentions	Per cent			
Total value traded to GDP	10	83.3			
Total value traded to market capitalisation	6	50.0			
Volume of shares traded to volume of shares listed	8	66.7			

Notice a: Details totalled more than N = 12 and 100 per cent due to duplicate responses.

Source: Analysis of the data obtained from semi-structured interviews.

Much of the evidence from the interviews supports the findings of the questionnaire. Respondents G and M argued that the value per share in the companies contributes no more than 100 LD and determines the proportion of the contribution per-capita ratios not exceeded, according to the following regulation:

"4 per cent of the total shares of companies do not exceed one million LD capital and must not exceed assets owned by the individual and its branches where there are 8 per cent of the total number of shares; 2 per cent of the total shares of companies in excess of their capital million LD, or more than two million LD and must not exceed assets owned by the individual, and its branches at 4 per cent of the total number of shares; 0.5 per cent of the total shares of companies that have capital exceeding four million LD and must not exceed assets owned by the individual and its branches at 1 per cent of the total number of shares".

This finding indicated the opinion that a more liquid stock market can encourage investors to equity market more information about firms, and allows them to alter their portfolios quickly

and cheaper. Thus, in more liquid stock markets, investment is less risky and facilitates longterm, more profitable investment.

7.4.4.3 Market Activity

Questionnaire Findings

Questionnaire respondents were presented with six statements concerning market activity processes and were asked to rate degrees of concern in respect of stock market performance on a 5-point rating scale. There were similarities in the *means* based on the market activity score, as presented in Table 7.20. The statement about market activity issues that concerned respondents the most was: 'The increased number of trading companies raises market activities in the stock market' (*mean* 3.79, *Skewness* -0.87, *Kurtosis* -0.34). Increased market activity induces information acquisition which, in turn, increases the information content of share prices which enable firms to make better investment decisions. The next highest ranking statement was: 'The increased value of new issues, including capital, generates market activities in the stock market' (*mean* 3.78, *Skewness* -0.81, *Kurtosis* -0.31). As the value of a new issue increases, besides including companies most representative of their sector, factors such as market capitalisation and activity are taken into consideration. This finding also means that new issues can be considered as an indicator of value, taking prices into consideration.

The third highest ranking statement was: 'A number of transitions are considered to be market activities that affect the stock market' (*mean* 3.70, *Skewness* -0.80, *Kurtosis* -0.54). The results confirm the findings by Levine and Zervos (1998) and Rosseau and Wachtel (2000) that there is a significant relationship between stock market development measured by share value traded ratio (i.e., market activity) and economic growth. The results are also consistent with the argument put forward by Levine and Zervos (1998) and Rosseau and Wachtel (2000) that market activity is related to economic growth more significantly than market size. One might argue that the value-traded ratio, which measures the value of shares traded as ratio to national output, should be expected to have a significant relationship with economic growth. A country could have a relatively large stock market in terms of size, yet this might constitute a small proportion of its GDP. The overall *mean* of the section of market activity is 3.74, which shows that market activity factors are implemented in the stock market.

Cada	Dank		Valid N=203				
Coue Kalik		Statements	Means ^a	SD	Skewness	Kurtosis	
E.10	6	The value of trade is part of market activity	3.6897	1.17618	-0.777	-0.386	
E.11	4	Volume of trade plays an important role in the stock market	3.7488	1.23117	-0.814	-0.468	
E.12	3	A number of transitions are considered to be market activities that affect the stock market	3.7044	1.24731	-0.796	-0.542	
E.13	1	The increased number of traded companies increases market activities in the stock market	3.7882	1.21029	-0.872	-0.338	
E.14	2	The increased value of new issues, including capital, generates market activities in the stock market	3.7833	1.14853	-0.814	-0.306	
E.15	5	The value of new issues acts as a percentage of GDP in the stock market	3.7291	1.25883	-0.753	-0.541	

Table 7.20 Descriptive Statistics for Market Activity

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree".

Source: Analysis of questionnaire survey data.

Interview Findings

Respondent interviewees in the stock market were asked about the market activity which most concerned them. Since interviewees were not required to rank order or rate the activity, the relevant measure was simply the number of times any given market activity was mentioned. Findings of the interviews, as depicted in Table 7.21, illustrate that the number of transactions was reported by eleven interviewees, whereas volume of trade and number of traded companies were each reported by ten interviewees. These findings support those reached by Atje and Jovanovic (1993), who show that trading volume (stock market development indicator) has a strong influence upon on economic growth, while bank credit does not.

 Table 7.21 Respondents' Interviews to Market Activity

Mowhat Antivity	Valid N=12 interviews ^a			
Market Activity	Number of Mentions	Per cent		
Volume of trade	10	83.3		
Value of trade	8	66.7		
Number of traded companies	10	83.3		
Number of transactions	11	91.7		
Value of new issues (including capital increases)	6	50.0		
Value of new issues (including capital increases) as a % of GDP	9	75.0		

Notice a: Details totalled more than N = 12 and 100 *per cent* due to duplicate responses.

Source: Analysis of the data obtained from semi-structured interviews.

Interviewees were additionally required to explain their concerns surrounding market activity. The head of Inspection and Control Market Securities stated, "*The stock market in Libya is witnessing a marked improvement*". He added, "*Despite the difficulties working to address the market the management is working on the preparation and processing of the rules and regulations governing the operation of the market with competent departmental staff, and the market will support sophisticated electronic systems to meet market needs in the Libyan environment*". With regard to the aspirations sought by the securities market, "the markets will strongly improve in the management of the progress of national economy if the necessary conditions are met".

7.4.4.4 Market Concentration

Questionnaire Findings

Questionnaire respondents were presented with two statements regarding market concentration processes and were requested to rate degrees of concern in respect of stock market performance on a 5-point rating scale. Based on the mean of the market concentration score, as presented in Table 7.22, the market concentration issue that concerned respondents the most was expressed in the statement: 'The percentage of large companies affects the value traded in the stock market' (mean 3.75, Skewness -0.84, Kurtosis -0.30). The second highest ranked concern was in the statement: 'The percentage of the largest company's share in market capitalisation increases the market capitalisation share in the stock market' (mean 3.72, Skewness -0.78, Kurtosis -0.46). Such a high concentration may not be desirable, since it affects liquidity. Furthermore, the large volume of trading activity may be mistakenly interpreted as a sign of a developed stock market. Therefore, any measure of stock market development should include a proxy for market concentration. Duyn (1993) carried out a large study on 50 emerging stock markets from various regions in the world in order to rank them. The study measured the market concentration by considering the share of market capitalisation of the ten largest stocks and the share of the value traded by them. The higher the *percentage*, the higher the market concentration and the lower the score. The results indicated that Taiwan emerges top with a total score of 164.14, followed by the Republic of Korea with a total score of 161.52 and then Malaysia with a total score of 158.89. In fact, these results show that the emerging stock market in South and East Asia could be considered as an advanced emerging stock market compared with other regions. In contrast, Tunisia was in the last position with a total score of only 22.42, close to Bangladesh with a total score of 27.59. An overall average of 3.74 is recorded indicating that the respondents agreed with the market concentration statements.

Code	Rank	Statements -	Valid N=203				
			Means ^a	SD	Skewness	Kurtosis	
E.16	2	The percentage of the largest company's share in market capitalisation increases market capitalisation share in the stock market	3.7192	1.20024	-0.779	-0.456	
E.17	1	The percentage of large companies affects the value traded in the stock market	3.7537	1.20987	-0.837	-0.295	
Notice	a: '	The percentage of respondents who scored 4 "agree" and 5 "stro	ngly agree				

Table 7.22 Descriptive Statistics for Market Concentration

Source: Analysis of questionnaire survey data.

Interview Findings

Interviewees from the stock market were asked about the market concentration aspects which most concerned them. Since they were not required to rank the order or rate the concentration, the relevant measure was simply the number of times that any given market concentration was mentioned. The findings of the interviews, as depicted in Table 7.23, reveal that the rate of the biggest companies' share in value traded was reported by eleven interviewees, whilst the rate of the largest companies' share in market capitalisation was reported by eight interviewees. The explanation suggests that stock market concentration refers to that select little scrip that dominates market activity. In certain emerging markets, while there may be both high capitalisation and volume, a few companies dominate the market.

Table 7.23 Respondents' Interviews to Market Concentration

Market concentration	Valid N=12 interviews ^a							
Market concentration	Number of Mentions	Per cent						
Rate of the largest companies' share in market capitalisation	8	66.7						
Rate of the largest companies' share in value traded	11	91.7						
Notice a: Details totalled more than $N = 12$ and 100 <i>per cent</i> due to duplicate responses.								

Source: Analysis of the data obtained from semi-structured interviews.

In general, the findings of the interviews seem to be congruent with the result of the questionnaire survey findings and the mainstream of the empirical research in the field of economic reform programmes and developing stock markets. The establishment of the Libyan stock market was a major step on the path of developing financial resources through the creation of a sound capital market. According to Libyan stock market law, the objectives of the market are threefold: firstly, to mobilise savings by encouraging investments in the securities, thereby channelling savings to serve the interests of the national economy. Secondly, the aim is to regulate and control the issuance of securities and dealings in a bid to ensure soundness, ease and speed of transactions in accordance with the financial interests of the country. Finally, there is the aim of collecting and publishing information that will help investors evaluate all the quoted companies. These factors are related specifically to the transitional economies and the move from a centrally-controlled approach to a market mechanism through deregulation. In these circumstances, the fundamental role played by the stock markets is that of facilitating the reallocation of funds from agents (individuals) with an excess of capital to projects that have a shortage of funds, through the provision of investment opportunities. This implies that stock market development, by reducing the transaction costs of savings and investment, lowers the overall cost of capital in the economy.

7.5 SUMMARY AND CONCLUSION

This chapter has presented and analysed the findings of the data collected by questionnaires and semi-structured interviews. The descriptive analysis of the data is the first statistical technique used to provide a summary of the respondents' and their companies' demographic characteristics using means, frequencies and standard deviation. The data obtained from interviews was processed and analysed using a thematic approach in order to explain and support the results that emerged from the questionnaire.

The results revealed that economic reform programmes, macro-economic reform and financial maturity plays a significant role in the progress of the Libyan stock market in terms of its performance and contribution towards the Libyan economy. Thus, it is important to liberalise the economy, thereby mobilising savings and increasing the ratio of investment to GDP when considering the economic reform programme. This strategy will promote better development of the stock market and, hence, its performance. Libya can encourage national economic growth by means of an appropriate policy reform programme. The respondents in this research study additionally point out that the most important requirement of a bank-based system or a market-based system is better growth for the long-term. In a bank-based system, this implies mobilising savings, imposing a regulatory structure, addressing problems associated with excessively powerful banks and structuring activity. In a market-based system it is necessary to identify sound investment, allocate capital, increase transparency and accountability and finance activity. Findings suggest that market size, as measured by the value of listed shares divided by GDP, was considered the most significant indicator factor. This was followed by the total value traded ratio which measures the organised trading of firm's equity as a share of national output and, which therefore, is considered as the most significant positive factor related to the liquidity market. To this end, the increased numbers of traded companies increase market activity in the stock market where the percentage of the largest companies affects the value traded in the stock market.

The next chapter presents the results from the analysis of the data collected by the questionnaires, and which serve to examine the research hypotheses. These latter results examine the economic reform programme, macro-economic reform and financial maturity independent variables, tested by using the linear regression models for developing the Libyan stock market as dependent variables (as were mentioned in preceding chapter). The last hypothesis regarding best practice from emerging the stock market and its transferability to the Libyan situation and its context will be examined in Chapter Nine.

206

CHAPTER EIGHT

TEST OF RESEARCH HYPOTHESES

Bivariate multivariate analyses and results

8.1 INTRODUCTION

This is the second of three chapters on data analysis. Chapter Six outlined the methods used in the collection of data for this research. Chapter Seven reported the descriptive analyses of the economic reform programme and the stock market performance in Libya including macroeconomic reform and financial maturity. The aims of this chapter are to present statistics of the research *model* and results relating to the *hypothesis* testing. In order to test the research model hypotheses, cross-sectional³⁸ regression analysis is based on models of the correlation matrix analysis, multiple regression analysis statistical techniques, simple linear regression model technique statistical test are used³⁹ to ascertain whether both Libyan economy and stock market has changed following introduction of the economic reform programme in 1999 (see Chapter Two). Several factors were hypothesised to impact on an economic reform programme and performance of the stock market from a review of the relevant literature and previously published capital market studies, which adopted questionnaire surveys. A hypothesis is an unproven statement or proposition about a factor or phenomenon that is of interest to the researcher (Malhotra et al., 2003). One reason for Libya's adoption of the economic reform programme was the stagnating status of the country's economy during the previous decade. In the late 1990s macro-economic stability was the sole aim of such reforms. More specifically, it presents and discusses the results of the research model relating to the aim and first three objectives of this research.

The main aim of conducting this research is to determine the most appropriate model for the continued viability of a stock market mechanism in Libya and to consider an appropriate strategy for the Central Bank of Libya to undertake successful continuation of the stock market development. To achieve this aim, the following research objectives were formulated.

³⁸ Bryman (2001) argues that cross-sectional data is usually provided with two variables or more, which are tested to determine the significance and directions of the association.

³⁹ The rationale and assumptions of using each statistic are discussed in Chapter Six: Subsection 6.10.1.2.

- **Objective One:** To assess the success of the economic reform programme in Libya, specifically the deregulation, corporatisation, privatisation and liberalisation that has led to the inception of the Libyan stock exchange.
- **Objective Two:** To identify the benefit of a stock market to the Libyan economy. This will involve examining the relationship between macro-economic reform variables and stock market performance variables.
- **Objective Three:** To assess the performance of market-based economies, with particular reference to the emerging economy of Libya, and to evaluate current and best practice in financial deregulation.
- **Objective Four:** To determine whether best practice from other emerging stock markets is transferable to the Libyan situation and context.

Within the Libyan context, hypotheses that are derived according to the economic reform programme are tested. Characteristics of the Libyan macro-economic reform, financial maturity and impact of stock market performance have not been investigated to date. The main purpose of this chapter is, therefore, to test the researcher's hypotheses. By carrying out this investigation, the researcher hopes that further investigation regarding stock market performance has been successfully accomplished. In this chapter, Section 8.2 reviews the descriptive statistics for research variables used in testing the hypotheses. This is followed by a brief justification of statistical techniques adopted to test the research models that were used to analyse data and examine the research hypotheses in Section 8.3. Section 8.4 presents the results of the modelling economic reform programme. Section 8.5 focuses upon the results of modelling macro-economic reform. Section 8.6 additionally focuses on the results from modelling financial maturity. Finally, in Section 8.7, a summary and conclusion to this chapter is provided.

8.2 DESCRIPTIVE STATISTICS FOR RESEARCH VARIABLES USED IN TESTING THE HYPOTHESES

Descriptive statistics of the research *model* that can be adopted for identification and integration to represent the potential variables were formulated to examine the impact of various independent variables of the economic reform programme (deregulation, corporatisation, privatisation and liberalisation), macro-economic reform such as interest rate, exchange rate, inflation rate, budget deficit, per-capita income, real GDP growth rate and

CHAPTER 8: TEST OF RESEARCH HYPOTHESES

financial maturity, such as banking assets' ratio, domestic credit ratio and stock market capitalisation ratio on the dependent variable stock market performance. These dependent variables are proxied by the market activity, market size, market liquidity and market concentration (see Chapter Five). The research aim and objectives of the cross-sectional study on the structure of the economic reform programme and the performance of the stock market (see Chapters Three, Four and Five) related to the research hypotheses in Section 8.4 were additionally tested. The structure of the research hypotheses testing is illustrated in Figure 8.1 showing the framework of research data analysis.



Figure 8.1 Framework of Research Data Analysis Excluding the Measurement Analysis

Source: Developed for this research from Section 8.2.

Table 8.1 illustrates that descriptive statistics for the research variables were measured in this research, followed by the coding process, data were subjected to several statistical tests. The table also includes the *mean* as a measure of central tendency, *standard deviation* as a measure of distribution spread, *minimum* and *maximum* values and *skewness* and *kurtosis* values to check for each variable's normality⁴⁰.

⁴⁰ Normality provides the degree to which distribution of sample data corresponds to *a normal distribution*, where normal distribution is a theoretical probability distribution in which the horizontal axis represents possible values of variables and the vertical axis represents the probability of those values occurring. Scores on the variables are clustered around the mean in a symmetrical, abnormal pattern known as the symmetrical bell-shaped or frequency curve (Hair et al., 2005: 38).

N 1 W • 11	Code	Valid N=203					
Research Variables	Variables	Mean	SD	Min	Max	Skewness	Kurtosis
Economic Reform Programme	ERP						
Deregulation	ERPD	4.1145	0.36724	3.00	4.88	-0.532	0.216
Corporatisation	ERPC	4.1350	0.33642	2.90	4.70	-0.831	1.167
Privatisation	ERPPV	4.0926	0.34996	2.80	4.70	-0.834	0.881
Liberalisation	ERPL	4.0847	0.48822	2.60	5.00	-0.829	0.431
Macro-economic Reform	MER						
Interest Rate	MERIR	3.7340	0.70765	1.40	5.00	-0.365	-0.151
Exchange Rate	MERER	3.9417	0.45901	2.17	4.83	-0.709	0.773
Inflation Rate	MERIFR	3.9631	0.50425	2.33	4.83	-0.768	0.526
Budget Deficit	MERBD	3.7545	0.47873	2.50	5.00	-0.465	-0.009
Per-capita Income	MERPCI	3.7340	0.51606	2.20	5.00	-0.304	0.141
Real GDP Growth Rate	MERGDP	3.6287	0.58879	1.88	4.75	-0.351	-0.413
Financial Maturity	FM						
Banking Assets Ratio	FMBA	3.8510	0.68145	1.75	5.00	-0.616	0.177
Domestic Credit Ratio	FMDC	3.7203	0.84795	1.44	4.89	-0.738	-0.090
Stock Market Capitalisation Ratio	FMSMC	3.5850	0.62079	1.50	5.00	-0.320	0.069
Stock Market Performance	SMP						
Stock Market Performance Variables	SMPV	3.7104	36607	2.47	4.47	-0.708	0.650

Table 8.1 Descripti	ve Statistics for	Research	Variables
---------------------	-------------------	----------	-----------

Source: Developed for this research from questionnaire survey data analysis.

Figure 8.2 shows that the structural statistical model test is divided into three models for the impact of stock market performance variables over an economic reform programme's variables, macro-economic reform variables and financial maturity variables. This model is adopted to develop a framework that can test the factors or variables of the independent variables (see further details in Chapter Six, Section 6.6). In the first model the hypothesised independent variables were tested in a single model based on the hypothesised relationship between this independent variable ERP and the dependent variables MER and the dependent variable SMP. The third stage of the structural model analysis tests the independent variable FM and the dependent variable SMP.





Source: Developed for this research from Chapter Six, Subsection 6.6.2: The Primary Research Model.

Section 8.3 explains and discusses "how" and "why" the research variables specified in Figure 8.2 are mentioned and, additionally, the procedure framework undertaken to develop the structural model using the relationships between exogenous and endogenous variable analysis. Sections 8.4, 8.5 and 8.6 test the first, second and third models, respectively, referred to in Figure 8.2 and other subsection reports on the research hypotheses' subsidiary.

8.3 STRUCTURAL MULTIPLE REGRESSION MODEL

All other statistical tests are located in Chapter Six for clear presentation. In order to achieve the research objective the researcher has applied a quantitative approach (questionnaire survey responses). The quantitative approach involves multiple regression analysis models that have been applied to test the research hypotheses using the *SPSS* Statistical Package for Social Science version 12.0. The purpose behind employing multiple regression analysis is to identify how much of the variance in the dependent variable will be explained when several independent variables are theorised to influence it simultaneously. Consequently, a multiple regression analysis test is conducted, by which the independent *predictor* variables are jointly regressed against the dependent *outcome* variable, in an effort designed to explain the variant and its predictions. Individual correlations collapse into what is termed a multiple *R* square value or multiple correlations. The square of multiple *R* is the amount of variance explained in the dependent variable by the predictors. When the R^2 value, the *F*-test statistic and its significance level are known, it is possible to interpret the results from a multiple regression analysis (Sekaran, 2003; Hair et al., 2005; Field, 2005).

8.3.1 The Simple Regression Linear Model

Multiple regression analysis model or both ANOVA and MANOVA, is a statistical technique that can be used to analyse the relationship between a single *dependent variable* and several *independent variables* and the objective of multiple regression analysis is to use the independent variables to predict the single dependent value selected by the researcher. Each dependent variable is weighted by the equation of the general model of multiple regressions to ensure maximum prediction from independent variables (Hair et al., 2005). The weights denote the ability to quantify precisely the relative significance of each proposed variable. Burton et al. (1989) define regression analysis as a statistical tool which may be utilised to learn more about the relationship between independent or explanatory variable(s) and dependent variables. To determine whether the assumption of linearity has been met, it is

necessary to examine the relationship between dependent and independent variables by utilising regression scatter plots.

From the previous discussion it is possible to test how the linear relationship between two variables could be measured using a simple linear correlation coefficient. Regression analysis, on the other hand, attempts to determine the functional relationship between two or more variables. In this case, one variable is being predicted by another. Thus, the relationship could be tested using correlation analysis. The best probabilistic general linear regression model, after several estimation trials, can be expressed as:

$$\boldsymbol{Y}_{i} = \boldsymbol{\beta}_{0} + \boldsymbol{\beta}_{1} \boldsymbol{X}_{i1} + \boldsymbol{\beta}_{2} \boldsymbol{X}_{i2} + \dots + \boldsymbol{\beta}_{k} + \boldsymbol{X}_{ik} + \boldsymbol{\mathcal{E}}_{i}$$

Figure 8.1 presents Y_i and X_i a variable, where Y_i is the outcome variable being predicted and is the dependent variable or the *regressand*. The predictor variable, denoted by X_i , is *i* th participants' score on the predictor variable and known as either the independent variable or the explanatory variable, β_1 is the coefficient of the first predictor X_1 , β_2 is the coefficient of the second predictor X_2 and β_0 is the intercept of that line. β_k is the coefficient of the *n* th predictor X_k . The portion $\alpha + \beta X_i$ represents the deterministic type of the probabilistic model. The random error term, ε_i , represents the collective unobservable influence of any committed variables (for more details see Appendix L: Section L.4).

8.3.2 The Correlation Matrix

Field (2005) observes that the correlation matrix is one of the most common and useful statistic analysis tests used for obtaining a general idea of the relationships between variable *predictors* and *outcome*, and for a preliminary search for *multicollinearity*. Notwithstanding, *multicollinearity* is a term reserved for describing the case when the intercorrelation of predictor variables in a regression model known as the independent variables is high. The model could have a high R^2 , low \sqrt{MSM} and highly significant *F*-test statistics. The problem with collinearity is not with the fit of the model, but with estimates of the β_i . Another approach technique for testing the hypotheses in the critical t-value is estimated based on a level of significance (α); Berenson and Levine (1999) argue that critical values can be expressed on whether the level of statistics test is one-tailed or two-tailed. If a positive or negative relationship is hypothesised, then a one-tailed test of significance can be employed.

Hair et al. (2005) state that there are three suggested methods for assessing *multicollinearity*: the presence of high correlation (generally 0.90 and over); tolerance values; the variance inflation factor values (*VIF*). However, the variance inflation factor (VIF) showed no values that exceed the generally accepted maximum level of 10 (an indication of high levels of *multicollinearity*) and the tolerance values showed no values less than the maximum level of 0.2^{41} . The VIF values range from 1.060 to 1.231, all well below 10, the value suggested by Myers (1990). Tolerance values range from 0.813 to 0.944 (see Table 8.5). None should be below 0.1, since tolerance = 1/VIF, also, Menard (1995) suggests that values below 0.2 are cause for concern. The average of the VIF values = 1.128. It is suggested by Bowerman and O'Connel (1990) that this should be no greater than 1. Therefore, this is indicating that *multicollinearity* problems may occur in this backward elimination model.

8.4 RESULTS FROM MODELLING ECONOMIC REFORM PROGRAMME

8.4.1 Testing of Hypothesis Theorising the Relationship between Economic Reform Programme and Stock Market Performance

Empirical Results

The hypotheses of this subsection are to investigate the relationship between the independent variables in terms of the economic reform programme (ERPD, ERPC, ERPP, ERPL) and SMP dependent variables. Achievement of the study's first objective can be realised via the survey questionnaire, which has been adopted to test Libya's stock market performance and whether or not it was affected following introduction of the economic reform programme. See more details on analysing data test in Appendix L; Section L.1. The following hypotheses are derived:

- H_0 There is no statistically significant relationship between economic reform variables and the stock market performance variable
- *H*₁ *There is a statistically significant relationship between economic reform variables and the stock market performance variable*

In this research, H_1 does not represent a single hypothesis, but a series of sub-hypotheses which are developed in Subsection 8.4.2.

⁴¹ Field (2005:196) explained that for testing these two scores:

[•] If the largest VIF is greater than 10 then there is cause for concern (Myers, 1990; Bowerman and O'Connell, 1990).

[•] Tolerance under 0.1 indicates a serious problem.

[•] Tolerance under 0.2 indicates a potential problem.

8.4.1.1 Correlation Matrix Statistics

Correlation matrix statistics tests were applied to explore the correlation between economic reform programme and stock market performance variables. The output of this test, as presented in Table 8.2, indicated three points: it demonstrates the value of the Spearman's Rho correlation coefficient between every pair of variables (e.g. the ERPP and ERPL had a positive correlation with SMP variables, R = 0.347; 0.425); the two-tailed significance of each correlation is displayed (e.g. the correlation of the ERPP and ERPL are significant, p < 0.01); the number of cases contributing to each correlation (N = 203) is displayed.

Variables		Spear	rman's Rho (2-taile	d) (N=203)	
variables	SMPV	ERPD	ERPC	ERPP	ERPL
SMDV		r = 0.169*	r = 0.147*	r = 0.347 **	r = 0.425 **
SMPV	-	p = 0.016	p = 0.036	p = 0.000	p = 0.000
EDDD			r = 0.083	r = 0.214 **	r = 0.139*
LKFD		-	p = 0.238	p = 0.002	p = 0.048
EDDC				r =0.203**	r = 0.071
LKFC			-	p = 0.004	p = 0.312
EDDD					r = 0.519 * *
EKPP				-	p = 0.000
ERPL					-
tice a:	* Correlation	on is significan	t at the 0.05 leve	el (2-tailed): ** (Correlation is
	41 - 0 01 1	···1 (2 ···1··1)		× //	

Table 8.2 Correlation of Descriptive Statistics for Regression Analysis

the 0.01 level (2-tailed).

Source: Analysis of *SPSS* output from questionnaire survey data.

With regard to the relationships between predictors and the outcome, two out of five economic reform programme variables had a significant positive correlation with stock market performance variables. Between these predictor variables, ERP and the outcome variables SMP ranged from Spearman's Rho = 0.083 and 0.071 with correlation of all four ERP being significant, p < 0.01, 2-tailed. The output Spearman's Rho test, in terms of the correlation matrix regression analysis did not reveal any significant correlation coefficient between two predictors which was p > 0.9. However, the highest correlation is between the ERPP, ERPL and SMP variables, respectively, which can be accepted at a significance level of p < 0.01 (Spearman's Rho = 0.347 and Spearman's Rho = 0.425, p < 0.01, 2-tailed).

8.4.1.2 Overall Fit of the Summary Model

The aim of this section is to *present* and *analyse* the overall multiple regression model and, thus, explain whether the model is successful in illuminating and presenting what might contribute to the ERP variables in better responding and adapting to the influence in the SMP variables. The output of the *SPSS* test, as depicted in Table 8.3, reveals that the model summary included all four predictors being used, which provides some very significant information about the model fit option. From Table 8.3, the column labelled R is the value of

the multiple correlation coefficients among the *predictors* and the *outcome*. Therefore, large values of the multiple R represent a considerable correlation among predicted and observed values of the outcome. In fact, a multiple R of 1 represents a situation in which the model perfectly predicts observed data. As such, multiple R is a measure of how well the model predicts the observed data analysis.

		р	Adjusted	Std Error of		Durbin				
Model 1	R	Square	P Square	the Estimate	R Square	F	df1	df2	Sig. F	Watson
		Square	K Square	the Estimate	Change	Change	un	u12	Change	w atsoli
1	0.689 ^(a)	0.439	0.423	1.19381	0.439	15.450	4	198	0.000	2.214
Notice a: a Predictors: (Constant), ERPD, ERPC, ERPP, ERPL										

Table 8.3 Multiple Regression Model Summary ^(b)

b Dependent Variable: SMP

Notice b:

Source: Analysis of SPSS output from questionnaire survey data.

The value of multiple R repetition for this model is 0.689, which is an indication that the model provides a reasonably good explanation of the observed values of the outcome variables of SMP variables. The next column in Table 8.3 provides the value of R^2 , which is a measure of how much of the variability in the outcome is accounted for by the predictors. This value is 0.439, which means that the four ERP variables included as predictors in the model account for 43 per cent of the variation SMP variables. The adjusted- R^2 , however, affords an idea of how well the model generalises and, ideally, it is preferable if the value of the adjusted- R^2 is the same as, or close to, the value of R^2 . With regard to this model, the difference between R^2 and adjusted- R^2 is 0.02 (0.439-0.423 = 0.016). This decrease means that if the model was derived from the population rather than a sample, it would account for approximately 2 per cent less variance in the outcome. The output of the change in the F-test, resulting from constructing the multiple regression model, is demonstrated in Table 8.3. The F-test is a measure of how much the model has improved the prediction of the outcome compared to the model's level of inaccuracy. In this context, a good model should have a large F-test greater than one at least⁴². As such, the model causes R^2 to alter from zero to 0.439 and this change in the amount of variance explained gives rise to an F-test of 15.450, which is significant (P < 0.01). At this point, the changed statistics exemplify variance incurred by adding new predictors to the model. Finally, the Durbin-Watson statistic tests whether the assumption of independent errors is tenable. The DW statistic, though, is only used for time series data, i.e. the data set being the values of the independent variables at fixed intervals of time. The value greater than 2 indicates a negative correlation between adjacent residuals and, for these data

⁴² According to (Field 2005) *F*-test is a measure of the goodness-of-fit of the model, adjusted R^2 , and a good model should possess more than 1 at least.
the value is 2.214, whereas a value $2 < DW \le 4$, negative residual autocorrelation. MS_R is mean squares for the regression model and MS_E the residual mean squares. Based on analysis of variance ANOVA test of SPSS output, as presented in Table 8.4, findings were that the *F*test is calculated by dividing average improvement in prediction by model MS_E by the average difference between the model and observed data MS_R .

Sum of Mean F Model 1 df Sig. Square Square 0.000^(a) 88.076 4 22.019 15.450 Regression Residual 280.760 198 1.425 Total 368.837 202

Table 8.4 ANOVA for Multiple Regressions ^(b)

Notice a: a Predictors: (Constant), ERPD, ERPC, ERPP, ERPL

Notice b: b Dependent Variable: SMP

Source: Analysis of SPSS output from questionnaire survey data.

If the improvement, due to fitting the regression model, is much greater than the inaccuracy within the model, then the value of F will be greater than 1. For this model, the F-test being 15.450, which is significant (P < 0.01). These results can be interpreted as meaning that the model significantly improved the ability to predict the outcome variable due to the "F-test is significant". Table 8.4 shows the value of the sum of squares for model SS_M which represents the improvement in prediction resulting from fitting a regression line to the data rather than using the mean as an estimate of the outcome. This value is 88.076. The table additionally indicates the value of the residual sum of squares SS_R , which represents the total difference between the model and observed data with the value of 280.760. For SS_M and SS_R , the degrees of freedom df for each term is provided. In the case of improvement due to the model, this value is equal to the number of predictors (four predictors in the case of this model) and, for the SS_R , it is the number of observations (203) minus the number of coefficients in the regression model. The model has five coefficients one for the constant and four the predictors. Furthermore, the model has 198 degrees of freedom related to SS_{R} . The average sum of squares or mean squares MS is then calculated for each term, by dividing each SS by its df. The F-test is calculated by dividing the average improvement in prediction by the model MS_M by average difference between the model and observed data MS_R .

8.4.1.3 Model Parameters

The section provided the output of parameters of the mutable regression model, which is presented in Table 8.5.

Model	Unsta Co	andardised efficient	Standardised Coefficients	ed nts t Sig.		95 % C Interva	confidence al for $oldsymbol{eta}$	Correlations			Collinearity Statistics	
1	β	Std. Error	Beta		U	Lower Bound	Upper Bound	Zero- order	Partial	Part	Tolerance 1/VIF	VIF
(Constant)	-3.415	1.356		-2.518	0.013	-6.090	4.045	-0.741				
ERPD	0.253	0.239	0.069	1.060	0.290	-0.218	0.259	0.724	0.170	0.066	0.920	1.087
ERPC	0.386	0.266	0.096	1.454	0.148	-0.138	0.113	0.910	0.159	0.090	0.884	1.132
ERPP	0.317	0.129	0.170	2.462	0.015	0.063	0.242	0.570	0.301	0.153	0.813	1.231
ERPL	0.684	0.115	0.380	5.940	0.000	0.457	0.192	0.911	0.423	0.369	0.944	1.060

Table 8.5 Coefficient of the Multiple Regression Model^(a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

In the current analysis this general model becomes:

$$SMP_{i} = \beta_{0} + \beta_{1}ERPD_{i} + \beta_{2}ERPC_{i} + \beta_{3}ERPP_{i} + \beta_{4}ERPL_{i}$$

= -3.415 + (0.253ERPD_{i}) + (0.386ERPC_{i}) + (0.317ERPP_{i}) + (0.684ERPL_{i})

The β -values inform the relationship between dependent variables SMP and each predictor. If the value is positive, this indicates a positive relationship between predictor and outcome, whereas a negative coefficient represents a negative relationship. Viewing the β -values presented in the first column, ERPD has significant positive relationship with the outcome variable SMP variable ($\beta = 0.253$). Similarly, ERPC ($\beta = 0.386$), ERPP ($\beta = 0.317$) and ERPL ($\beta = 0.684$) respectively. In these contacts, the β – values reveal to what degree each predictor affects the outcome if the effects of all other predictors are held constant. The output from t-test is considered, as presented in Table 8.5, in order to establish whether a β -value is substantially different from zero. Thus, t-test is measured by whether or not the predictor is offering a significant contribution to the model. Accordingly, if the t-test associated with a β -value is significant (e.g. if the value in the column labelled Sign. is less than 0.01 or 0.05) then the predictor is creating a significant contribution to the model. Similarly, the value of Sig. and the larger the value of t-test, the greater the contribution of that predictor is. From this model, only one predictor, which is liberalisation programme (t = 5.940, P < 0.01), emerged as a significant predictor of the ERP influencing the SMP variables. Standardised versions of the β -values are, in many ways, easier to comprehend because they are not dependent on the units of measurement of the variables. The standardised beta-values provided by SPSS and presented in Table 8.5 reflect the number of standard deviations that the outcome will alter, as a consequence of one standard deviation change in the predictor. All the standardised β - values are measured in standard units and so are directly comparable. The standardised β is representing the direct effect of its indicator on the outcome regardless of the amount of relationship *correlation* between this indicator and other indicators in the model, which cause the amount of indirect effect of this indictor on the outcome via the other indicators correlated with it. Hence, they provide a superior insight into the importance of a predictor in the model. In the finding from Table 8.5, it is noted that liberalisation has the highest standardised β -value 0.684, indicating that this variable has the highest degree of significance in the model, compared with other predictors. This supports the result obtained from the t-test, in that the deregulation programme also emerged as the most significant predictor of SMP variable.

8.4.2 Subsidiary Hypotheses

The relationship between each variable of independent variables of economic reform programme and dependent variables in stock market performance, four secondary hypotheses were formulated to be tested. Subsections 8.4.2.1, 8.4.2.2, 8.4.2.3 and 8.4.2.4 test, respectively, the first model referred to in the research. Variables specified in Figure 8.2 are measured between ERP independent variables and SMP dependent variable.

8.4.2.1 Deregulation

Test Hypotheses of Deregulation

H_0 Deregulation has a significant negative effect on the stock market performance

$H_{1,1}$ Deregulation has a significant positive effect on the stock market performance

Deregulation programme is the first dimension of ERP variables, as presented in Table 8.6, multiple regression model, which is the ERPD independent variables and SMP variables as dependent variables. Reviewing the hypothesised regression model revealed $\beta = 0.644$, b = 0.174 and t-value = 2.503 resulting in the completely standardised coefficient of ERPD as positive significant at 0.05 level, P < 0.05. Controlling for all other variables, the correlation matrix coefficient between the deregulation programme and stock market performance variables, was positively significant (Spearman's Rho = 0.169, p = 0.016, 2-tailed). Results of the coefficient correlation analysis indicated that the ERPD effect is indirect, and the results of SMP variables confirmed the direct relationship (i.e. *ERPD* \rightarrow *SMP*). Results of the regression analysis revealed a positive effect deregulation programme on stock market performance. Thus, H_0 is rejected and the alternative hypothesis $H_{1.1}$, which states that deregulation has a significant positive effect on stock market performance is accepted.

Model	Unsta Coe	ndardised fficient	Standardised Coefficients	t	Sig.	95 % Co Interval	nfidence for $oldsymbol{eta}$	Collinearity	Statistics	Durbin-
1	β	Std. Error	Beta		e	Lower Bound	Upper Bound	Tolerance 1/VIF	VIF	Watson
(Constant)	0.485	1.062		0.457	0.648	-1.610	2.580			2 221
ERPD	0.644	0.257	0.174	2.503	0.013	0.136	1.151	1.000	1.000	2.231
R^{2}	0.331									
Adjusted R ²	0.320									
F	6.263	P < 0.05								

Table 8.6 Coefficient of the Multiple Regression Model^(a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

It was argued in the literature (McKinnon, 1973; Shaw, 1973; Fry, 1982; Cho, 1988) that, in theory, deregulation of financial sectors in most developing countries is vital for an efficient distribution of economic resources, which has increased considerable leverage. Lombardo and Pagano (2000) argue that the legal and institutional environment has a significant impact on expected returns. Laporta et al. (2005) found that legal systems affect financial systems and political change. Arlman (2003: 236) mentions: "[A] strong and modernised EU financial legislation should be beneficial not only for the securities markets but more importantly for the EU economy as a whole. An efficient and integrated securities market is vital in the process of raising the level of competitiveness, the efficient allocation of capital, mobilising savings and disciplining management". Rajan and Zingales (1998), therefore, claim that countries with poor legal financial systems benefit from a bank-based system, as an enhanced legal financial system improves the financial market-based system. Barth et al. (2004, 2005) demonstrate that regulatory systems and supervisory practice is crucial in promoting information disclosure and private sector enterprise. In the Libyan context, deregulation was the first step related to the change with overall function of financial institutions and economic activity. These change origins are related specifically to the transitional economies, such as the deregulation of the economy from a centrally controlled to a market mechanism and the change of companies' objectives from maintaining national goals to profit-making.

8.4.2.2 Corporatisation

Test Hypotheses of Corporatisation

- H_0 Corporatisation has a significant negative effect on stock market performance
- $H_{1,2}$ Corporatisation has a significant positive effect on stock market performance

Corporatisation programme is the second dimension of ERP variables, as presented in Table 8.7, multiple regression model, which is the ERPC independent variables and SMP variables as dependent variables. Corporatisation of economic reform programme, when entered into the hypothesised regression model, revealed $\beta = 0.650$, b = 0.161 and t-value = 2.310 resulting in

the completely standardised coefficient of ERPC being positive significant at 0.05 level, P < 0.05. Controlling for all other variables, the correlation matrix coefficient between corporatisation programme and stock market performance variables, was positively significant (Spearman's Rho = 0.147, p = 0.036, 2-tailed). Results of the coefficient correlation analysis indicated that the ERPC effect is indirect, and the results of SMP variables confirmed the direct relationship (i.e. $ERPC \rightarrow SMP$). Regression analysis results in Table 8.7 revealed a positive effect corporatisation programme on stock market performance. Hence, H_0 is rejected and the alternative hypothesis $H_{1.2}$ which states that corporatisation has a significant positive effect on stock market performance is accepted.

Model	Unsta Coe	ndardised efficient	Standardised Coefficients	t	Sig.	95 % Co Interval	nfidence I for eta	Collinearity	Statistics	Durbin-
1	β	Std. Error	Beta		0	Lower Bound	Upper Bound	Tolerance 1/VIF	VIF	Watson
(Constant)	0.446	1.167		0.382	0.703	-1.855	2.748			2 207
ERPC	0.650	0.281	0.161	2.310	0.022	0.095	1.205	1.000	1.000	2.207
R^{2}	0.271									
Adjusted R ²	0.258									
F	5.334	P < 0.05								

 Table 8.7 Coefficient of the Multiple Regression Model
 (a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

As earlies explained in Chapter Two, the Libyan economy has experienced two major changes during the last 35 years. The first change started in the mid-1970s with the banning of private ownership and the nationalisation of all economic activities. The second wave was in the opposite direction and started officially in 1988, which was followed by further steps over the early years of 1990s⁴³: privatisation of many state-owned companies; decline in the state support of those companies, represented by the imposition of equal foreign exchange rates⁴⁴, allowing foreign products to penetrate the domestic market; etc., aiming to reform the economy towards a market-based system and to provide an opportunity for the private sector to contribute to the economy and overcome the difficulties that the economy had encountered. Most Libyan companies have, positively or negatively, been affected by these economic changes, which have resulted in a dramatic deterioration in the financial performance of most of the state-owned companies in particular.

⁴³ Law No. 8 was issued in 1988 aiming to reorganise the private ownership of economic activities such as corporations, partnerships and joint stock companies.

⁴⁴ Special foreign currency exchange rates were applied to the state-owned sector, allowing them to exchange one LD for about 3.50 US\$, while the situation was opposite for the private companies, which were exchanging 3.5 LD for one US\$. This situation gave the state-owned companies advantages in costs when they imported foreign raw materials and paid foreign employees (the cost was ten times more for private companies than for the state-owned companies).

8.4.2.3 Privatisation

Test Hypotheses of Privatisation

 H_0 Privatisation has a significant negative effect on stock market performance

 H_{13} Privatisation has a significant positive effect on stock market performance

The privatisation programme is the third dimension of ERP variables, as presented in Table 8.8, multiple regression model, which is the ERPP independent variables and SMP variables as dependent variables.

Model	Unstandardise	d Coefficient	Standardised Coefficients	t	Sig.	95 % Co Interval	nfidence for eta	Colline Statis	arity tics	Durbin-
1	β	Std. Error	Beta		6	Lower Bound	Upper Bound	Tolerance 1/VIF	VIF	Watson
(Constant)	0.960	0.498		1.927	0.055	-0.022	-0.022			2 259
ERPP	0.560	0.126	0.351	4.457	0.000	0.312	0.312	1.000	1.000	2.238
R^{2}	0.301									
Adjusted R ²	0.290									
F	19.861 <i>P</i> < 0.01									

Table 8.8 Coefficient of the Multiple Regression Model^(a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

Privatisation of the economic reform programme, when entered into the hypothesised regression model, revealed $\beta = 0.560$, b = 0.351 and t-value = 4.457 reflecting that the completely standardised coefficient of ERPP is positively significant at 0.01 level, *P*<0.01. Controlling for all other variables, the correlation matrix coefficient between the privatisation programme and stock market performance variables, was highly positively significant (Spearman's Rho = 0.347, *p* <0.01, 2-tailed). The results of the coefficient correlation analysis indicated that the ERPP effect is indirect and the results of SMP variables confirmed the significant direct relationship (i.e. *ERPP* \rightarrow *SMP*). Results of the regression analysis in Table 8.8 revealed a positive effect privatisation programme on stock market performance. Thus, *H*₀ is rejected and the alternative hypothesis *H*_{1.3}, which states that privatisation has a significant positive effect on the stock market performance, is accepted.

In Chapter Five it was argued that a privatisation programme assists in increasing both the market value of stocks and the number of listed companies. The privatisation programme should, therefore, be allocated, according to a number of authors (i.e. El-Naggar, 1989; Boubakri and Cosset, 1998; Claessens and Djantov, 2000; Megginson and Netter, 2001; Cook and Unhida, 2003; Omran, 2004), as one of the most critical issues in the field of current economy in developed and developing countries. Empirical work by Perotti and Van Oijen

CHAPTER 8: TEST OF RESEARCH HYPOTHESES

(2001) highlights that the effect of privatisation on market development may be indirect. This may provide some explanation to Euromoney (1993, 1994), which indicated that Malaysia's 1992 market capitalisation increased about 30 *per cent*, contributed to by the privatised sector. Mexico's market's capitalisation in 1993 had major listed firms, representing about 18 *per cent* of the total. In 1994, in Argentina, the shares of YPF, Telecom and Telephonic contributed some 50 *per cent* of total market capitalisation. Within the same year Chile listed three of its largest companies on the exchange which were all privatised firms, with a market value of over 10 billion US\$, representing almost 25 *per cent* of market capitalisation.

With respect to Libya, transfer of ownership began in 1987 by sharing returns on investment with employees: profits are divided among production elements (worker, machine, capital) according to a specific accounting system. Transfer of ownership was governed by the People's Committee Resolution no.447 of 1987, which was the first step of the Libyan privatisation programme. In 1994, 145 government plants were transferred to the private sector with immediate effect. In 1995, a further 295 government plants were transferred to employee ownership. Based upon a chapter from the Green Book (partners not workers), law no.9 in 1985 and the GPC Resolution no.313 of 2003, the Libyan government announced its intention to privatise a further 360 ownership plants in the industrial and agricultural sectors (see Appendix A). This suggests that, in the Libyan context, privatisation has shown many similarities to privatisation programmes that have been implemented in other emerging economies.

8.4.2.4 Liberalisation

Test Hypotheses of Liberalisation

 H_0 Liberalisation has a significant negative effect on stock market performance

 H_{14} Liberalisation has a significant positive effect on stock market performance

The liberalisation programme is the fourth dimension of ERP variables, as presented in Table 8.9, multiple regression model, which is the ERPL independent variables and SMP variables as dependent variables. Liberalisation of economic reform programme, when entered into the hypothesised regression model, revealed $\beta = 0.746$, b = 0.413 and t-value = 6.425 resulting in the completely standardised coefficient of ERPL as positive significant at 0.01 level, P < 0.01. Controlling for all other variables, the correlation matrix coefficient between liberalisation programme and stock market performance variables, was highly positively significant (Spearman's Rho = 0.425, p<0.01, 2-tailed). Results of coefficient correlation analysis

indicated that the ERPL effect is indirect, and the results of SMP variables confirmed the significant direct relationship (i.e. $ERPL \rightarrow SMP$). The results of the regression analysis, in Table 8.9, revealed a positive effect liberalisation programme on stock market performance. Thus, H_0 is rejected and the alternative hypothesis $H_{1,4}$, which states that liberalisation has a significant positive effect on the stock market performance, is accepted.

Model	Unstar Coe	ndardised fficient	Standardised Coefficients	t	Sig.	95 % Co Interval	nfidence for $oldsymbol{eta}$	Collinearity	Statistics	Durbin-
1	β	Std. Error	Beta		U	Lower Bound	Upper Bound	Tolerance 1/VIF	VIF	Watson
(Constant)	0.201	0.465		0.433	0.665	-0.715	1.117			2 257
ERPL	0.746	0.116	0.413	6.425	0.000	0.513	0.975	1.000	1.000	2.257
R^2	0.520									
Adjusted R ²	0.512									
F	41.283	P < 0.01								

 Table 8.9 Coefficient of the Multiple Regression Model
 (a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

The results support the findings of an assessment of studies by Cho (1986) who examined the link between stock market development and financial liberalisation, using data for 32 developing countries during the period 1978 to 1990. He discovered a significant positive correlation between the level of stock market development and financial liberalisation; the results indicate that the correlation is strong among Asian economies but not within other emerging economies. Alternatively, studies by Bekaret and Harvey (1998), Henry (2000a), Froot et al. (2001) found there to be a positive correlation between liberalisation and stock market performance. Stulz (1997) and Mishkin (2001) argue that liberalisation can help improve financial infrastructure, reducing problems of moral hazard and raise further the availability of credit. Additionally, liberalisation and the consequent development of a country's financial sector tends to greatly facilitate economic growth, as shown in papers such as those of King and Levine (1993), Jayaratne and Strahan (1996) and Levine (2000).

The Libyan economy, which was one of the most promising among MENA countries during the 1970s, began to experience some structural problems towards the end of the 1980s. As mentioned earlier in greater detail in Chapter Two, the government had attempted to liberalise its economy twice before, from 1987 to 1990, and then in a second wave in the later 1990s and the initial years of the new century. After ten years of international economic sanctions related to the Lockerbie case of 1988, the Libyan authorities decided to remove significant control on economic and financial institutions. This may provide some explanation for the liberalisation that has generally accompanied a fundamental deregulation of domestic banking sector, including the financial market and foreign trade to achieve increasing economic growth.

8.4.3 Discussion: Findings from Testing the Economic Reform Programme Hypotheses

The aim of this section is to *present* the overall multiple regression model and hypotheses, that is, to explain whether there is a relationship between the economic reform programme (independent/predictor variables) and stock market performance (dependent/outcome variables). The output of the *SPSS* test, as depicted in Table 8.10, reveals the hypotheses of an economic reform programme relating to the first research objective. These results offer very significant information about the model fit under study with those accepted and rejected hypotheses.

Нурс	otheses	χ^2	df	Relations va	hip with SMP riables	F-test	Sig.	Beta	t-test
				Expected	Comment				
EDDD	H_0	126 527	15	-	Rejected	6 262	0.012	0.174	2 502
EKFD	H _{1.1}	120.337	15	+	Accepted	0.203	0.015	0.174	2.505
EDDC	H ₀	110 286	16	-	Rejected	5 334	0.022	0 161	2 3 10
EKIC	H _{1.2}	110.280	10	+	Accepted	5.554	0.022	0.101	2.510
EDDD	H ₀	106 201	19	-	Rejected	10.861	0.000	0.201	1 157
LKFF	H _{1.3}	100.891	10	+	Accepted	19.001	0.000	0.301	4.457
EDDI	H ₀	145 103	14	-	Rejected	41 283	0.000	0.413	6 4 2 5
LIXEL	H _{1.4}	145.105	14	+	Accepted	41.203	0.000	0.415	0.423

Table 8.10 Structural Model Results for Scope of Testing Hypotheses

Source: Analysis of SPSS output from questionnaire survey data.

The results of the majority of hypotheses are similar for using statistical methods. Acceptance or rejection of each hypothesis is based upon the result being statistically significant for at least two of the methods. Therefore, if one method is statistically significant at 1 *per cent* (P < 0.01) or 5 *per cent* (P < 0.05) level, the hypothesis will be accepted. In this context, each of the economic reform programme hypotheses is tested and justification that might explain reported outcomes is discussed. The hypothesised regression model resulted in a beta of 0.174 (t-value = 2.503) for the completely standardised coefficient of ERPD. Therefore, the regression path is significantly different from zero when t > 1.645 (P < 0.05), t > 1.96 (P < 0.01) for null hypothesis, (i.e. research questionnaire) beta estimate is considered significantly different from zero when t > 1.645 (P < 0.05), t is result supported H_{1.1} of ERPD at 0.01 significant level P < 0.01, since the calculated t-test is greater than the critical t-test = 1.645.

These results suggest that several economic reform programme variables have various effects on stock market performance. Findings indicate that the *F*-test (F = 5.334) was at significant level (Sig. = 0.000), hypothesis of ERPC which is accepted. The results of the t-test showed that ERPP has positive relations with the SMP variables at significant level of (Sig. = 0.000) since the calculated t-test is more than the critical t-test = 1.645. The t-test demonstrated a very strong relationship between ERPL and SMP variables (t = 6.4, F = 41.3, Sig. = 0.000), hypothesis H_{1.4} is accepted for ERPL. To conclude, the findings of four hypothesis models indicate that economic reform programme variables have a positive impact on the stock market performance variables. It would appear that deregulation, corporatisation, privatisation and liberalisation have become vital to long-term Libyan economic growth.

8.5 RESULTS FROM MODELLING MACRO-ECONOMIC REFORM

8.5.1 Testing of Hypothesis Theorising the Relationship between Macro-economic Reform and Stock Market Performance

Empirical Results

It is worth mentioning that the macro-economic reform variables are treated as independent variables when measuring their effect on the variables of stock market performance which are, in turn, treated as dependent variables comprising: market activity, market size, market liquidity and market concentration. Therefore, the performance of stock market variables depends heavily on the success of economic reform variables. In return, a well-performing stock market sustains economic growth and development. See more details on analysing data test in Appendix L; Section L.2. On testing the second objective in this study, that being whether or not the impact of macro-economic reform and the performance of stock market was affected, the null hypothesis H_0 and associated secondary hypotheses are:

- *H*₀ *There is no statistically significant relationship between the macro-economic reform variables and stock market performance variables*
- H_2 There is a statistically significant relationship between the macro-economic reform variables and stock market performance variables

In this research, H_2 does not represent a single hypothesis, but a series of sub-hypotheses which are developed in Subsection 8.5.2.

8.5.1.1 Correlation Matrix Statistics

Correlation matrix statistics tests were applied in order to explore the correlation between macro-economic reform variables and stock market performance variables. The output of this test, as presented in Table 8.11, illustrated that in the relationships between predictors and outcome; one out of seven macro-economic reform variables had a significant positive correlation with stock market performance. Between these predictor variables MER and the outcome variables, SMP ranged from Spearman's Rho = -0.005 and 0.182 with the correlation of all five MER being insignificant, P > 0.05, 2-tailed. Only one of five MERIFR found negative correlation with the outcome of SMP variables (Spearman's Rho = -0.042, P > 0.05, 2-tailed).

Variables			Spearm	nan's Rho (2-tailed)	(N=203)		
variables	SMPV	MERITR	MERER	MERIFR	MERBD	MERPCI	MERGDP
SMPV	-	r = 0.109 p = 0.122	r = 0.172* p = 0.014	r = -0.042 p = 0.556	r = 0.049 p = 0.483	r = 0.044 p = 0.534	r = 0.086 p = 0.224
MERITR		-	r = 0.016 p = 0.823	r = 0.061 p = 0.388	r = 0.044 p = 0.533	r = 0.070 p = 0.320	r = 0.013 p = 0.854
MERER			-	$r = 0.182^{**}$ p = 0.009	r = 0.017 p = 0.814	r = -0.017 p = 0.809	r = -0.018 p = 0.803
MERIFR				-	r = 0.008 p = 0.906	r = 0.131 p = 0.062	r = -0.005 p = 0.939
MERBD					-	r = 0.079 p = 0.260	r = -0.077 p = 0.273
MERPCI						-	r = 0.082 p = 0.245
MERGDP							
Notice a:	* Correlati	ion is significar	nt at the 0.05 lev	vel (2-tailed); **	Correlation is	significant at	the 0.01 level

Table 8.11 Correlation of Descriptive Statistics for Regression Analysis

(2-tailed).

Source: Analysis of SPSS output from questionnaire survey data.

The output of Spearman's Rho test, in terms of the correlation matrix regression analysis, did not reveal any significant correlation coefficient between two predictors that was p > 0.9. However, the highest correlation is between the MERIFR and MERER variables which can be accepted at a significance level of p < 0.01 (Spearman's Rho = 0.182, p < 0.01, 2-tailed).

8.5.1.2 Overall Fit of the Summary Model

The aim of this section is to *present* and *analyse* the overall multiple regression model and, accordingly, explain whether the model is successful in illuminating and presenting what might contribute to the MER variables in better responding and adapting to the influence among the SMP variables. The output of the *SPSS* test, as depicted in Table 8.12, reveals that the model summary illustrated all six predictors are being used, which provides some very significant information about the model fit option.

Table 8.12 Multiple Regression Model Summary ^(b)

						Chan	ge Statist	ics		
Model 1	R	R Square	Adjusted R Square	Std. Error of the Estimate	R Square Change	F Change	df1	df2	Sig. F Change	Durbin- Watson
1	0.735 ^(a)	0.640	0.622	1.33554	0.640	2.225	6	196	0.042	2.013
Notice a:	a Predi	ctors: (Con	stant), MER	ITR, MERER, M	IERIFR, M	ERBD, ME	ERPCI,	MERG	DP.	

Notice b: b Dependent Variable: SMP

Source: Analysis of SPSS output from questionnaire survey data.

From Table 8.12, the value of multiple *R*-value for this model is 0.735, which is an indication that the model provides a reasonably good explanation of the observed values of the outcome variables of SMP variables. The next column in Table 8.12 provides the value of R^2 , which is a measure of how much of the variability in the outcome is accounted for by predictors. This value is 0.640, which means that the four MER variables included as predictors in the model account for 64 per cent of the variation SMP variables. Adjusted- R^2 , however, affords an idea of how well the model generalises and, ideally, it is preferable if the value of the adjusted- R^2 is the same as, or close to the value of R^2 . In relation to this model, the difference between R^2 and adjusted- R^2 is 0.018 (0.640-0.622 = 0.018). This decrease means that, if the model was derived from the population rather than a sample, it would account for approximately 1.8 *per cent* less variance in the outcome. As such, the model causes R^2 to alter from zero to 0.640 and this change in the amount of variance explained gives rise to an F-test of 2.225, which is significant with a probability at 0.05 level. If improvement due to fitting the regression model is much greater than the inaccuracy within the model, then the value of Fwill be greater than 1. For this model, the *F*-test is 2.225, which is significant (P < 0.05). These results can be interpreted as meaning that the model is significantly positive to predict the outcome variable due to the *F*-test being significant. Table 8.13 additionally indicates the value of the residual sum of squares SS_{R} , which represents the total difference between the model and observed data with the value of 349.596.

	Model	Sum of Square	df	Mean Square	F	Sig.
1	Regression	23.813	6	3.969	2.225	0.042 ^(a)
	Residual	349.596	196	1.784		
	Total	373.409	202			

Table 8.13 ANOVA for Multiple Regressions^(b)

Notice a: a Predictors: (Constant), MRPITR, MRPER, MRPIFR, MRPPBD, MRPCI, MRPGDP. Notice b: b Dependent Variable: SMP

Source: Analysis of SPSS output from questionnaire survey data.

8.5.1.3 Model Parameters

This section provides the output of the parameters of the mutable regression model, which is presented in Table 8.14.

Model	Unsta Coe	ndardised fficient	Standardised Coefficients	t	Sig.	95 % C Interva	onfidence al for $oldsymbol{eta}$		Correlations		Collinearity Statistics	
1	β	Std. Error	Beta		U	Lower Bound	Upper Bound	Zero- order	Partial	Part	Tolerance 1/VIF	VIF
(Constant)	-0.772	1.502		-0.514	0.608	-3.734	2.191					
MERITR	0.050	0.033	0.106	1.522	0.130	-0.015	0.116	0.113	0.108	0.105	0.990	1.011
MERER	0.110	0.038	0.204	2.896	0.004	0.035	0.186	0.194	0.203	0.200	0.963	1.039
MERIFR	-0.035	0.037	-0.068	-0.957	0.340	-0.108	0.037	-0.014	-0.068	-0.066	0.945	1.058
MERBD	0.017	0.037	0.032	0.456	0.649	-0.056	0.090	0.047	0.033	0.032	0.984	1.017
MERPCI	0.076	0.040	0.063	1.964	0.036	-0.043	0.114	0.068	0.064	0.062	0.971	1.030
MERGDP	0.094	0.086	0.076	1.983	0.017	-0.035	0.124	0.073	0.078	0.076	0.988	1.012

 Table 8.14 Coefficient of the Multiple Regression Model
 (a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

In the current analysis this general model becomes:

$SMP_{i} = \beta_{0} + \beta_{1}MERITR_{i} + \beta_{2}MERER_{i} + \beta_{3}MERIFR_{i} + \beta_{4}MERBD_{i} + \beta_{5}MERPCI_{i} + \beta_{6}MERGDP_{i}$ $= -0.772 + (0.050MERITR_{i}) + (0.110MERER_{i}) + (-0.035MERIFR_{i}) + (0.017MERBD_{i}) + (0.076MERPCI_{i}) + (0.094MERGDP_{i})$

The β -values inform the relationship between dependent variables SMP and each predictor. If the value is positive, this indicates a positive relationship between predictor and outcome, whereas a negative coefficient represents a negative relationship. Viewing the β -values presented in the first column, MERGDP has the significant negative relationship with the outcome variable SMP variable ($\beta = -0.772$). In these contacts, the β -values reveal to what degree each predictor affects the outcome if the effects of all other predictors are held constant. The output from t-test is considered, as presented in Table 8.14, in order to ascertain whether a β -value is significantly different from zero. Thus, t-test is measured by whether the predictor offers a significant contribution to the model. If the t-test associated with a β -value is significant (e.g. if the value in the column labelled Sign. is less than 0.05), then the predictor is creating a significant contribution to the model. Similarly, the larger the value of Sig. and the larger the value of t-test, the greater the contribution of that predictor is. From this model, only one predictor, MERER (t = 2.896, P < 0.01), emerged as a significant predictor of the MER influencing SMP variables. In addition, the collinearity statistics show (Table 8.14) VIF values ranging from 1.011 to 1.058, all well below 10, and tolerance values ranging from 0.963 to 0.990. None should be below 0.1. The average of the VIF values = 1.028 again indicates that multicollinearity problems may occur in this backward elimination model, so the coefficients of the variables cannot be interpreted through the regression model.

8.5.2 Subsidiary Hypotheses

Additionally, in order to achieve the first research aim, and the second of the study's objectives, several secondary hypotheses were formulated, all of which relate to the relationship period between each variable of macro-economic reform in dependent variables of stock market performance in the Libyan economy.

8.5.2.1 Interest Rate

Test Hypotheses of Interest Rate

- H_0 There is no statistically significant relationship between the interest rate variables and stock market performance variables
- $H_{2.1}$ There is a statistically significant relationship between the interest rate variables and stock market performance variables

Interest rate is the first dimension of MER variables as presented in Table 8.15, multiple regression model, with the MERITR independent variables and SMP variables as dependent variables. Reviewing the hypothesised regression model revealed $\beta = 0.054$, b = 0.113 and t-value = 1.607 for the completely standardised coefficient of *MERIR* \rightarrow *SMP*. Therefore, the regression path is not significant since the correlation coefficient between interest rate and stock market performance was small and negligible (Spearman's Rho = 0.109, p = 0.122, 2-tailed). Accordingly, this result supported H_0 and rejected $H_{2.1}$ at 0.05 significant levels, P > 0.05.

Model	Unstan Coef	dardised ficient	Standardised Coefficients	t	Sig.	95 % Co Interval	nfidence for $oldsymbol{eta}$	Collinearity	Statistics	Durbin-
1	β	Std. Error	Beta		e	Lower Bound	Upper Bound	Tolerance 1/VIF	VIF	Watson
(Constant)	2.294	0.531		4.323	0.000	1.248	3.341			1.005
MERITR	0.054	0.033	0.113	1.607	0.110	-0.012	0.120	1.000	1.000	1.905
R^{2}	0.282									
Adjusted R ²	0.269									
F	2.581	P > 0.05								

 Table 8.15 Coefficient of the Multiple Regression Model
 (a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of *SPSS* output from questionnaire survey data.

Findings suggested that interest rates have an insignificant relationship to stock market performance variables, which state that the latter increases as interest rates decrease. This cannot be rejected as interest rates affected all areas of stock market variables. This may provide some explanation for the relatively well-developed banking system in Libya; i.e. it does not play a prominent role involving interest rate problems. CBL Resolution no.16 in 2004

CHAPTER 8: TEST OF RESEARCH HYPOTHESES

cut the interest rate on loans and credits granted for productive purpose from 7.0 to 3.0 *per cent*. This was followed by Resolution no.15 in 2005, related to reducing interest rates on all loans and credits granted by commercial banks to be equivalent to the rediscount rate of CBL plus a percentage not exceeding 2.5 *per cent*.

As a conclusion, it seems that interest rates have a significant impact on the stock market. When interest rates increase stock prices decrease, as investors prefer to invest in the stock market rather than banks. For instance, culturally, 97 per cent of Libyans are Muslim (Ward, 2008) and Islamic law (Sharia) interprets the Koran's principles for borrowing from commercial banks or ethical economy and investments due to associated interest rates "Reba" (or usury). Libya does not have Islamic banks to encourage its people to invest. "Reba", therefore, is defined in Islamic Sharia as exchanging cash of the same kind, with one party paying back extra at once or at later date. This type of transaction is forbidden in Islam and further applies to saving money in the bank and receiving interest in return. As a result, this offers investors a sound opportunity to transform their way of investment from banking to stock market. Moreover, it may be suggested that the new companies were established in recent times as the cost of capital became lower with the decrease in interest rates and, thus, could obtain required finance via the stock market. The results support findings of the literature review in many financial studies which examined alterations to the relationship between interest rate models and stock returns (i.e. Fama (1975, 1990); James et al. (1985), Asprem (1989), Schwert (1990), Lee (1992) and Canova and De Nicolo (2000)). For instance, Shiller and Beltratti (1992) find an indirect relationship between interest rates and stock returns. Other studies that have found a negative or insignificant relationship include Giovannini (1983, 1985), Reichel (1991) and Oshikoya (1992). Peavy (1992) discovered a negative relationship between interest rates and stock prices. Additionally, Baye and Jansen (1995) note that there is an unchanging relationship between interest rates and stock prices.

8.5.2.2 Exchange Rate

Test Hypotheses of Exchange Rate

- H_0 There is no statistically significant relationship between the exchange rate stability and stock market performance
- $H_{2,2}$ There is a statistically significant relationship between the exchange rate stability and stock market performance

Exchange rate stability is the second dimension of MER variables, as presented in Table 8.16, multiple regression model, which is the MERER independent variables and SMP variables as dependent variables.

Model	Unstar Coef	ndardised fficient	Standardised Coefficients	tandardised Coefficients t		95 % Co Interval	nfidence for eta	Colline Statis	earity stics	Durbin-
1	β	Std. Error	Beta		C	Lower Bound	Upper Bound	Tolerance 1/VIF	VIF	Watson
(Constant)	0.984	0.771		1.276	0.203	-0.536	2.503			1.097
MERER	0.105	0.037	0.194	2.810	0.005	0.031	0.179	1.000	1.000	1.967
R^{2}	0.384									
Adjusted R ²	0.368									
F	7.898	P < 0.05								
	1									

Table 8.16 Coefficient of the Multiple Regression Model^(a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

The review of the hypothesised regression model revealed ($\beta = 0.105$, b = 0.194 and t-value = 2.810) for the completely standardised coefficient of *MERER* \rightarrow *SMP*. Therefore, the regression path is positive significant at 0.01 level, P < 0.01. This result supported $H_{2.2}$ since the calculated t-value is greater than the critical t-value = 1.645. Controlling for all other variables, the correlation coefficient between exchange rate and stock market performance variables was positively significant (Spearman's Rho = 0.172, p = 0.014, 2-tailed). Results of the coefficient correlation analysis indicated the direct relationship between MERER and SMP variables, in which it can be concluded that the notion that stock market performance increases as the exchange rate increases cannot be rejected.

The relationship between exchange rate and stock market depends upon whether the country is import-dominant or export-dominant. There are several interesting studies in the literature which investigate this particular type of relationship. Most empirical studies, however, failed to establish a significant relationship between the exchange rate and stock prices (for example, Jorion, 1991 and Amihud, 1993), whilst there are other studies (Chow, 1997; Choi et al., 1998; De Santis and Cerard, 1998; Doukas et al., 1999; Patro et al., 2002) which discovered that stock returns are significantly affected by exchange rate fluctuations. Similar findings in the literature were evident in Kasman (2003) who examined the relationship between exchange rates and stock prices in emerging economies, specifically the Turkish market. He established that there is a stable relationship between exchange rate and stock prices in the long-term. The results additionally indicate that the causality relationship exists only from the exchange rate to the industry sector index which, in turn, affects currency.

In the Libyan context, the LD has been tied to the Special Drawing Rights (SDR) since June 2003, with LD 0.5175 per SDR unit, but it is altering against other foreign currencies. After more than five years of operating with this system, it was found that the exchange rate is quite appropriate during this period for the Libyan economic policy and conditions, as agreed by the report of the Consultative Mission of the IMF that visited Libya in December 2005 (CBL report, 2002, 2007). Thus, as the exchange rate becomes more stable in the Libyan economy following government introduction of the economic reform programme, this may encourage foreign investors and financial institutions to invest in the stock market. The exchange rate stability affects stock market performance variables, which indicates that the fact that the stock market increases as the exchange rate stability increases cannot be rejected.

8.5.2.3 Inflation Rate

Test Hypotheses of Inflation Rate

- H_0 There is no statistically significant relationship between the inflation rate and stock market performance
- $H_{2.3}$ There is a statistically significant relationship between the inflation rate and stock market performance

Inflation rate is the third dimension of MERIFR variables, as presented in Table 8.17, multiple regression model, which is the MERIFR independent variables and SMP variables as dependent variables. Review of the hypothesised regression model revealed $\beta = -0.037$, b = -0.014 and t-value = -0.994 for the completely standardised coefficient of *MERIFR* \rightarrow *SMP*. The value of R^2 for the MERIFR variable is 0.290 which means that this variable is explained through such a model, with about 0.3 per cent of the variance of inflation rate. Hence, the regression path is not significant. The results of the correlation coefficient indicated an indirect and insignificant relationship between inflation rate and stock market performance variables, which was small and negative significant (Spearman's Rho = -0.042, p = 0.556, 2-tailed). In fact, this relationship was negative and the results can support the hypothesis, which stated that stock market performance increases as inflation rate decreases. Consequently, H_0 cannot be rejected as the inflation rate affected stock market performance variables.

Model	Un (standardised Coefficient	Standardised Coefficients	t	Sig.	95 % Co Interva	onfidence 1 for $oldsymbol{eta}$	Collinearity	Statistics	Durbin-
1	β	Std. Error	Beta		0	Lower Bound	Upper Bound	Tolerance 1/VIF	VIF	watson
(Constant)	3.278	0.755		4.340	0.000	1.789	4.768			1 000
MERIFR	-0.037	0.036	-0.014	-0.994	0.406	-0.079	0.065	1.000	1.000	1.000
R^{2}	0.290									
Adjusted R ²	0.278									
F	2.693	P > 0.05								
Nation at	. Dan	a dame Manialala.	CMD							

Table 8.17 Coefficient of the Multiple Regression Model^(a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

The results support the findings of Fama (1981, 1982) which established a negative relationship between inflation and economic activity, but a positive relationship between economic activity and stock prices. Benderly and Zwick (1985) support Fama's results. Other studies by Fama and Schwert (1977), Chance and Lane (1980), Lynge and Zumwalt (1980) and Wahlroos and Berglund (1986), among others, discover an evident negative relationship between expected inflation and stock price returns. Various other studies have shown an unexpected negative inflation effect in stock returns (Flannery and James, 1984 and Bae, 1990). Rapach (2002) also traced a negative relationship between inflation and real stock prices. Shiller and Beltratti (1992) identify a small correlation coefficient between inflation and stock returns. This finding additionally contradicts the idea that the level of inflation rate can depress or encourage investment in the stock market. In Libya, for instance, the inflation rate is not stable at the current time due to an increase in the proportion of public expenditure (especially the wages sector), besides the higher liquidity surplus and import costs, particularly of foodstuffs. In addition, decrease in inflation rate may offer a good incentive to investors to invest in the stock market. Furthermore, it will anticipate that investors will seek out new companies and secure the required finance for their business with less cost. On the other hand, investors may leave their investment in the bank, as savings there will be less profitable and attractive as interest rates are expected to decrease, as mentioned previously. This may lead some investors to prefer making the correct decision and invest in the stock market.

8.5.2.4 Budget Deficit

Test Hypotheses of Budget Deficient

- H_0 There is no statistically significant relationship between the budget deficit and stock market performance
- $H_{2.4}$ There is a statistically significant relationship between the budget deficit and stock market performance

Budget deficit is the fourth dimension of MERBD variable as presented in Table 8.18, multiple regression model, which is the MERBD independent variables and SMP variables as dependent variables. The review of the hypothesised regression model revealed $\beta = 0.025$, b = 0.047 and t-value = 0.661 for the completely standardised coefficient of *MERBD* \rightarrow *SMP*. Therefore, the regression path is not significant at 0.05 level, P < 0.05. Accordingly, this result did not support $H_{2.4}$, since the calculated t-value is less than the critical t-value = 1.645. Controlling for all other variables, the correlation coefficient between budget deficit and stock market performance variables was small and not significant (Spearman's Rho = 0.049, p = 0.483, 2-tailed). The results of the coefficient correlation analysis models indicated an indirect relationship between MERBD and SMP variables. Thus, the hypothesis which can be stated, that stock market performance increases as budget deficit decreases, can neither be accepted nor rejected.

Model	Unstand Coeff	lardised ficient	Standardised Coefficients	t	Sig.	95 % Co Interval	infidence I for $oldsymbol{eta}$	Collinearity	Statistics	Durbin-
1	β	Std. Error	Beta		U	Lower Bound	Upper Bound	Tolerance 1/VIF	VIF	Watson
(Constant)	2.653	0.733		3.621	0.000	1.208	4.098			1.904
MERBD	0.025	0.038	0.047	0.661	0.509	-0.049	0.099	1.000	1.000	1.894
R^{2}	0.283									
Adjusted R ²	0.269									
F	2.437	P > 0.05								

 Table 8.18 Coefficient of the Multiple Regression Model
 (a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

From Chapter Five empirical studies, in previous discussion of the framework, the relationship between budget deficit and stock market performance is expected to be negative in terms of macroeconomic reform. In addition, the budget deficit may directly or indirectly affect stock market prices and returns via its effect upon both interest rates and inflation rate. Findings clarify that the budget deficit affects stock returns and prices negatively through the stock market performance as a whole. In theoretical studies, Tobin (1969), Blanchard (1981) and Shah (1984) contended that fiscal policy bears an impact upon stock prices. Similar findings in the literature were found in Darrat (1988), who investigated the fundamental effect of the relationship between stock market and budget deficit of quarterly data from the Canadian market covering the period 1960 to 1984, adopting a technique suggested by Granger (1980). He concluded that there is some evidence that a government's significant budget deficit is a determinant of stock prices. This may provide some explanation for the Libyan contexts. In 2006 public expenditure was about 23119.1 million LD, compared to 19547.8 million LD in 2005, which increased to approximately 4171.3 million LD or 21.3 *per cent*. This increase

mainly assigned for salaries and bonuses but, additionally, it was an increase in the provisions of development budget for fundamental productive and service infrastructure projects (CBL, 2006). As a result, the Libyan government should adopt a serious decision policy to control public expenditure in future, which may affect the stock market price via economic growth in the long-term.

8.5.2.5 Per-capita Income

Test Hypotheses of Per-capita Income

- H_0 There is no statistically significant relationship between the per-capita income and stock market performance
- $H_{2.5}$ There is a statistically significant relationship between the per-capita income and stock market performance

Interest rate is the first dimension of MER variables as presented in Table 8.19, multiple regression model, which is the MERPCI independent variables and SMP variables as dependent variables.

Model	Unstar Coef	idardised ficient	Standardised Coefficients	t	Sig.	95 % C Interva	95 % Confidence Interval for $oldsymbol{eta}$		Collinearity Statistics		
1	β	Std. Error	Beta			Lower Bound	Upper Bound	Tolerance 1/VIF	VIF	Watson	
(Constant)	2.525	0.638		3.959	0.000	1.268	3.783			1.007	
MERPCI	0.038	0.040	0.068	1.964	0.036	-0.040	0.171	1.000	1.000	1.907	
R^{2}	0.398										
Adjusted R ²	0.376										
F	4.928	P < 0.05									

 Table 8.19 Coefficient of the Multiple Regression Model ^(a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

The review of the hypothesised regression model revealed $\beta = 0.038$, b = 0.068 and t-value = 1.964 for the completely standardised coefficient of *MERPCI* \rightarrow *SMP*. Consequently, the relationship between per-capita income as an independent variable and stock market performance as dependent variables is not significant with a standard coefficient beta at 0.05 levels (P < 0.05) of the variance of per-capita income. The result of the correlation coefficient analysis between MERPCI variable and SMP variables was small and significant (Spearman's Rho = 0.044, p = 0.534, 2-tailed). Results of the coefficient correlation analysis models indicated the indirect relationship between per-capita income and SMP variables. Thus, H_0 is rejected and the alternative hypothesis $H_{2.5}$, where it can be stated that there is a statistically significant relationship between the per-capita income and stock market performance, cannot be rejected, since the calculated t-value is greater than t-value = 1.645.

CHAPTER 8: TEST OF RESEARCH HYPOTHESES

It was argued in the literature that higher per-capita income may affect many aspects of the economy and stock market performance. As noted, Gurley and Shaw (1955, 1960 and 1967) argued that financial development is a positive function of real income and wealth. Boyd and Smith (1996, 1998), however, contested that financial structure alters with income per-capita. Therefore, it can be concluded that, as per-capita income increased in the Libyan economy after it adopted the economic reform programme due to the increase in world oil prices and additional growth in production, this may well assist in increased interest by foreign investors. Oil production is expected to double to about 3 million barrels per day during 2012 resulting from modern techniques usage by international oil companies in research and prospecting. Indeed the total volume of major investment projects announced in the various sectors (of hydrocarbons and tourism) is some 35 billion US\$. As a result, there is a greater supply of money within society to save and invest, which ultimately benefits the stock market. Market size and activity may be affected positively, too, as the more money people have, the greater ability there is to invest and inject money in stock market investments. To summarise, the evidence emphasises that stock market performance increases as per-capita income in terms of macro-economic reform increases cannot be accepted or rejected. Instead, it can be argued that significantly differing per-capita income affects levels across the stock market performance area.

8.5.2.6 GDP Growth Rate

Test Hypotheses of GDP Growth Rate

- H_0 There is no statistically significant relationship between the real GDP growth rate and stock market performance
- $H_{2.6}$ There is a statistically significant relationship between the real GDP growth rate and stock market performance

GDP growth rate is the sixth dimension of MER variables, as presented in Table 8.20, multiple regression model, which is the MERGDP growth rate independent variables and SMP variables as dependent variables. The review of the hypothesised regression model revealed $\beta = 0.094$, b = 0.076 and t-value = 1.983 for the completely standardised coefficient of *MERGDP* \rightarrow *SMP*. Confidence limits demonstrate that it can be assumed, with 95 per cent confidence, that the true population regression coefficient for GDP growth rate is between - 0.034 and 0.123. Therefore, the regression path is not significant at 0.05 level, P < 0.05. Subsequently, this result was rejected at 0.05 significant levels. Controlling for all other variables, the correlation coefficient between GDP growth rate and stock market performance

variables was insignificant (Spearman's Rho = 0.086, p = 0.224, 2-tailed). The results of the coefficient correlation analysis indicated that GDP growth rate effect is indirect. Nevertheless, the results of regression analysis revealed a positive effect of independent variable GDP growth rate on dependent variables of stock market performance. Findings suggested that H_0 is rejected and the alternative hypotheses $H_{2.6}$, which states that there is statistically significant relationship between the real GDP growth rate and stock market performance variables, is accepted.

Model	Unstan Coef	dardised ficient	Standardised Coefficients	t	Sig.	95 % Confidence Interval for β		Collinearity Statistics		Durbin-
1	β	Std. Error	Beta		0	Lower Bound	Upper Bound	Tolerance 1/VIF	VIF	Watson
(Constant)	2.473	0.647		3.823	0.000	1.198	3.749			1 0 0 7
MERGDP	0.094	0.086	0.076	1.983	0.017	-0.034	0.123	1.000	1.000	1.007
R^{2}	0.401									
Adjusted R ²	0.390									
F	4.621	P < 0.05								

Table 8.20 Coefficient of the Multiple Regression Model^(a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

In this finding, it can be argued that there is a positive relationship between the real GDP growth rate and stock market performance, which cannot be rejected. This variable requires greater attention from the Libyan government in order to achieve a high standard of living for its citizens. The results support the findings of Chapter's Five literature review. This can be explained by a number of subsequent studies that have adopted the growth regression framework related to stock market performance (see e.g. Dowrick, 1992; King and Levine, 1993a; Atje and Jovanovic, 1993; Levine and Zervos, 1995, 1996 and 1998; Harris, 1997; Levine et al., 2000). Levine and Zervos (1995) used cross-country regression for 49 countries. They identified significant correlation between GDP growth rates and stock market development. Demirqüe-Kunt and Levine (1999) found that the correlation between GDP percapita and market capitalisation is about 0.3 and is significant at the 0.05 level. The findings additionally identified found the correlation between GDP per-capita, and both total value trade as a share of GDP and turnover ratio are almost 0.4 and significant at the 0.01 level. In the context of the IMF visiting staff report of 12 November 2007, it was anticipated that real GDP in Libya would have increased by 6.8 per cent in 2007 compared to a rate of 5.2 per cent in 2006, due to rapid growth in non-oil activities of about 7.5 per cent, and robust growth in oil production of 4.7 per cent. Consequently, the Libyan authorities are planning to develop the non-oil sector via the private sector to support the projects of small and medium-sized enterprises, which would contribute to increasing economic growth rates and creating new

jobs. These should focus efforts on training and reduce administrative procedures besides supporting property rights, regulation systems and stock market development.

8.5.3 Discussion: Findings from Testing the Macro-economic Reform Hypotheses

The *SPSS* outcomes in Table 8.21 summarise all previous results found in the multiple regression analysis tests conducted into the relationship between the six macro-economic reform independent variables and the dependent variables stock market performance under study with those accepted and rejected hypotheses. As shown in Table 8.21, there are two steps to using the hypotheses testing model. The first is to run a multiple regression model for each macro-economic reform, as explanatory variables upon all stock market performance, as a response variable. This step's output will demonstrate the results of fitting a multiple regression model, which describes the relationship between one independent variable and all dependent variables. The next step used the correlation matrix and ANOVA statistic analysis to look at the *P*-value and *F*-test. If the result giving independent variables is greater or equal to ten *per cent* level, this means there is not a statistically significant relationship and the variable should be excluded from the model. The general regression model will prove to be very simple. This is effected by eliminating insignificant variables, beginning with the variable that has a higher P-value and so on.

Hypothes	es	χ^2	df	Relationship	with SMP variables	F-test	Sig.	Beta	t-test
				Expected	Comment				
MERITR	H ₀ H _{2.1}	173.56	73	- +	Accepted Rejected	2.581	0.110	0.113	1.607
MERER	H ₀ H _{2.2}	155.172	46	- +	Rejected Accepted	7.898	0.005	0.194	2.810
MERIFR	H ₀ H _{2.3}	154.281	47	- +	Accepted Rejected	2.693	0.406	-0.014	-0.994
MERBD	H ₀ H _{2.4}	167.522	47	- +	Accepted Rejected	2.437	0.509	0.047	0.661
MERPCI	H ₀ H _{2.5}	177.655	48	- +	Rejected Accepted	4.928	0.036	0.068	1.964
MERGDP	H ₀ H _{2.6}	148.690	46	- +	Rejected Accepted	4.621	0.017	0.076	1.983

Table 8.21 Structural Model Results for Scope of Testing Hypotheses

Source: Analysis of SPSS output from questionnaire survey data.

Table 8.21 provides that the results from the research hypotheses model indicated that there is a different significant relationship between macro-economic reform and stock market performance variables. It is useful to mention that this result appears consistent with a number of previous findings, which indicated that explanatory variables tend to have a more significant effect upon stock market performance variables. Nevertheless, the other variable seems to be far from the previous findings as the relationships tend to be insignificant, which reject the hypotheses. For instance, the t-test identified an insignificant relationship between interest rate and stock market performance variables. Similarly, finding in inflation rate that there is no relationship with stock market performance, means these hypotheses cannot be accepted. For this research, given an emerging market, it is inevitable that the population sample is not ideal. Hence, there is a need to develop validated databases in the future as more data becomes available. Consequently, it would be possible to examine and identify any additional relationship between variables that could bear influence upon Libyan economy's financial market and could utilise different significant statistic analysis.

8.6 RESULTS FROM MODELLING FINANCIAL MATURITY

8.6.1 Testing of Hypothesis Theorising the Relationship between Financial Maturity and Stock Market Performance

Empirical Results

The aim of this section is to *present* and *analyse* the financial maturity variables which are treated as independent variables when measuring their effect on the variables of stock market performance. These are, in turn, treated as dependent variables. These latter are: market activity; market size; market liquidity; market concentration. In examining the challenge of the financial maturity in term of their ratios of banking assets, domestic credit and stock market capitalisation of GDP concerning stock market decisions, which may be influenced by business and personal goals. Thus, the survey questionnaire is formulated in order to test this research's third objective. See more details on analysing data test in Appendix L; Section L.3. The null hypothesis H_0 and associated secondary hypotheses are:

- H_0 There is no statistically significant relationship between financial maturity variables and stock market performance
- *H*₃ *There is a statistically significant relationship between the financial maturity variables and stock market performance*

In this research, H_3 does not represent a single hypothesis, but a series of sub-hypotheses which are developed in Subsection 8.6.2.

8.6.1.1 Correlation Matrix Statistics

Correlation matrix statistics tests were applied in order to explore the correlation between economic reform programme and stock market performance variables. This test's output, as presented in Table 8.22, indicated that, in the relationships between predictors and outcome,

two out of four of the financial maturity variables had a significant positive correlation linear relationship with stock market performance. These predictor variables FM and the outcome variables SMP ranged from Spearman's Rho = -0.021 and 0.245 with the correlation of the two FM being significant, *P*< 0.01, 2-tailed. The only FMDC found that there is no correlation with the outcome of SMP variables (Spearman's Rho = 0.011, *P* > 0.05, 2-tailed).

Variables		Spearman's R	ho (2-tailed) (N=20)3)
variables	SMPV	FMBA	FMDC	FMSMC
SMDV		r = 0.245 **	r = 0.011	r = 0.228 **
SIVIE V	-	p = 0.000	p = 0.875	p = 0.001
EMBA			r = -0.021	r = 0.126
TWIDA		-	p = 0.767	p = 0.073
FMDC				r = 0.104
TWIDC			-	p = 0.139
FMSMC				-
Notice a:	** Correlati	on is significan	t at the 0.01 lev	el (2-tailed).

 Table 8.22 Correlation of Descriptive Statistics for Regression Analysis

Source: Analysis of SPSS output from questionnaire survey data.

The output of Spearman's Rho test in terms of the correlation matrix regression analysis did not reveal any significant correlation coefficient between one predictor that was p > 0.9. However, the high correlation is between one predictor, the FMSMC and FMBA, which can be accepted at a significance level of p < 0.05 level (Spearman's Rho = 0. 126, p < 0.05, 2tailed).

8.6.1.2 Overall Fit of the Summary Model

The aim of this section is to *present* and *analyse* the overall multiple regression model and, resultingly, explain whether the model is successful in illuminating and presenting what might contribute to the FM variables in better responding and adapting to the influence in SMP variables. Regarding the output of *SPSS* test, as demonstrated in Table 8.23, the model summary included all three predictors being used, which provides some very significant information about the model fit option.

Table 8.23 Multiple Regression Model Summary^(b)

M- 1-1 1		D	Adjusted	Std. Error			Durhin			
Model 1	R	K Squara	P Square	of the	R Square	F	df1	df2	Sig. F	Watson
		Square	K Square	Estimate	Change	Change	un	u12	Change	vv atsoli
1	0.821 ^(a)	0.742	0.733	1.28576	0.742	8.958	3	199	0.000	1.974
Notice	a: a Pre	edictors: (Constant), FI	MBA, FMDO	C, FMSMC.					

Notice b: b Dependent Variable: SMP

Source: Analysis of SPSS output from questionnaire survey data.

As shown in Table 8.23, the value of multiple *R*-value for this model is 0.821, which is an indication that the model provides a reasonably good explanation of the observed values of the

outcome variables of SMP variables. The adjoining column in Table 8.23 provides the value of R^2 , which is a measure of how much variability in the outcome is accounted for by the predictors. This value is 0.742, which means that the three FM variables included as predictors in the model account for 74 per cent of the variation SMP variables. The output of the change in the F-test resulting from constructing the multiple regression model is demonstrated in Table 8.23. The *F*-test is a measure of how much the model has improved the prediction of the outcome compared to the model's level of inaccuracy. In this context, a good model should have a large F-test (greater than one at least). As such, the model causes R^2 to alter from zero to 0.742 and this change in the amount of variance explained gives rise to an *F*-test of 8.958, which is significant (P < 0.01).

	Model	Sum of Square	df	Mean Square	F	Sig.
1	Regression	44.427	3	14.809	8.958	$0.000^{(a)}$
	Residual	328.987	199	0.134		
	Total	373.409	202			
Notice	a Predic	tors: (Con	stant) FM	BA FMDC	FMSMC	

Table 8.24 ANOVA for Multiple Regressions ^(b)

A, FMDC, FMSMC. b Dependent Variable: SMP Notice b:

Source: Analysis of SPSS output from questionnaire survey data.

If improvement due to fitting the regression model is much greater than the inaccuracy within the model, then the value of F will be greater than 1. For this model, the F-test is 8.958, which is significant (P < 0.01). These results can be interpreted as meaning that the model significantly improved our ability to predict the outcome variable due to the *F*-test being significant at 0.1 levels. Table 8.24 additionally indicates the value of the residual sum of squares SS_R , which represents the total difference between the model and observed data with the value of 328.987.

8.6.1.3 Model Parameters

This section provides the output of the parameters of the multiple regression model, which is presented in Table 8.25.

	Tuble 0.20 Obtilicient of the Multiple Regression Model												
Model	Unstand Coeff	lardised icient	Standardised Coefficients	t	Sig.	95 % Confidence Interval for $oldsymbol{eta}$			Correlation	Collinearity Statistics			
1	β	Std. Error	Beta		C	Lower Bound	Upper Bound	Zero- order	Partial	Part	Tolerance 1/VIF	VIF	
(Constant)	0.221	0.704		0.314	0.754	-1.168	1.611						
FMBA	0.389	0.135	0.195	2.892	0.004	0.124	0.655	0.237	0.201	0.192	0.972	1.029	
FMDC	-0.035	0.102	-0.023	-0.348	0.728	-0.236	0.165	-0.003	-0.025	-0.023	0.985	1.015	
FMSMC	0.447	0.119	0.256	3.772	0.000	0.213	0.681	0.284	0.258	0.251	0.962	1.039	

Table 8.25 Coefficient of the Multiple Regression Model^(a)

Notice a:

a Dependent Variable: SMP.

Source:

Analysis of SPSS output from questionnaire survey data.

In the current analysis this general model becomes:

$$SMP_{i} = \beta_{0} + \beta_{1}FMBA_{i} + \beta_{2}FMDC_{i} + \beta_{3}FMSMC_{i}$$
$$= 0.221 + (0.389FMBA_{i}) + (-0.035FMDC_{i}) + (0.447FMSMC_{i})$$

The β -values inform the relationship between dependent variables SMP and each predictor. If the value is positive, this indicates a positive relationship between predictor and outcome, whereas a negative coefficient represents a negative relationship. Viewing the β -values presented in the first and third column, FMBA and FMSMC have a significant positive relationship with the outcome variable SMP variable ($\beta = 0.389$ and 0.447). Differently, FMDC has an insignificant relationship with outcome variable SMP variables ($\beta = -0.035$). In these contexts, the β -values reveal to what degree each predictor affects the outcome if the effects of all other predictors are held constant. The output from t-test is considered, as presented in Table 8.25, in order to test whether a β -value significantly differs from zero. Resultingly, t-test is a measure of whether the predictor offers a significant contribution to the model. Therefore, if the t-test associated with a β – value is significant (e.g. if the value in the column labelled Sign. is less than 0.05), then the predictor is creating a significant contribution to the model. Similarly, the value of Sig. and the larger the value of t-test the greater the contribution of that predictor is. From this model, only one predictor, which is the ratio of stock market capitalisation (t = 3.772, P<0.01), emerged as a significant predictor of the FM variable influencing the SMP variables. In addition, the collinearity statistics show (Table 8.25) VIF values ranging from 1.015 to 1.039, all well below 10, and tolerance values ranging from 0.962 to 0.985. None should be below 0.1. The average of the VIF values = 1.027 again indicates that multicollinearity problems may occur in this backward elimination model, so the coefficients of the variables cannot be interpreted through the regression model.

8.6.2 Subsidiary Hypotheses

For the relationships between each variable of independent variables of financial maturity and dependent variables in stock market performance, three secondary hypotheses were formulated for testing. Subsections 8.6.2.1, 8.6.2.2 and 8.6.2.3, respectively, test the third model referred to in the research variables specified in Figure 8.2, measured between FM independent variables and SMP dependent variable.

8.6.2.1 Banking Assets Ratio

Test Hypotheses of Banking Assets Ratio

- H_0 There is no statistically significant relationship between banking assets and stock market performance
- $H_{3.1}$ There is a statistically significant relationship between banking assets and stock market performance

The banking assets ratio is the first dimension of FM variables as presented in Table 8.26, multiple regression model, which is the FMBA independent variables and SMP variables as dependent variables. The review of the hypothesised regression model revealed $\beta = 0.472$, b = 0.472 and t-value = 3.452 for the completely standardised coefficient of *FMBA* \rightarrow *SMP*. Consequently, the regression path is positively significant at 0.01 level, P < 0.01. The *F*-test for the FMBA is 11.915, which means that this variable is explained through such a model of banking assets ratio variance. Thus, this result supported $H_{3.1}$, since the calculated t-value is greater than critical t-value = 1.645. Controlling for all other variables, the correlation coefficient between banking assets ratio and stock market performance variables was positively significant (Spearman's Rho = 0.245, *p* = 0.000, 2-tailed). Results of the coefficient correlation analysis indicated the direct relationship between FMBA variables and SMP variables, in which it can be concluded that stock market performance increases as the banking assets ratio increases.

Model	Unsta Coe	ndardised efficient	Standardised Coefficients	t	Sig.	95 % Confidence Interval for $oldsymbol{eta}$		Collinearity	Durbin-	
1	β	Std. Error	Beta		e	Lower Bound	Upper Bound	Tolerance 1/VIF	VIF	Watson
(Constant)	1.315	0.535		2.460	0.015	0.261	2.370			1.004
FMBA	0.472	0.137	0.472	3.452	0.001	0.202	0.742	1.000	1.000	1.904
R^{-2}	0.520									
Adjusted R ²	0.513									
F	11.915	P < 0.01								

Table 8.26 Coefficient of the Multiple Regression Model^(a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

This finding is also in line with the view of Demirqüc-Kunt and Maksimovic (1998), who proposed, empirically, that banking system development and stock market liquidity are positively correlated with development companies. King and Levine (1993) found significant correlation between the development of the banking system over the 1960-1985 period and the growth rate of per-capita GDP. Beck et al. (1999a) and Levine et al. (2000) reiterated that the indicator of the ratio of total bank assets to GDP gives evidence of the significance of financial

services performed by the banking sector associated with the excess size of economic growth, but this measure does not reveal whether the banking system is in the public or private sector.

8.6.2.2 Domestic Credit Ratio

Test Hypotheses of Domestic Credit Ratio

- H_0 There is no statistically significant relationship between domestic credit and stock market performance
- *H*_{3.2} *There is a statistically significant relationship between domestic credit and stock market performance*

The domestic credit ratio is the second dimension of FM variables as presented in Table 8.27, multiple regression model, which is the FMDC independent variables and SMP variables as dependent variables.

Model	Unst Co	andardised	Standardised Coefficients	Standardised Coefficients t		95 % Confidence Interval for $oldsymbol{eta}$		Collinearity Statistics		Durbin-
1	β	Std. Error	Beta		e	Lower Bound	Upper Bound	Tolerance 1/VIF	VIF	Watson
(Constant)	3.150	0.398		7.907	0.000	2.365	3.936			1 00/
FMDC	-0.145	0.107	-0.093	-1.145	0.364	-0.216	0.206	1.000	1.000	1.004
R^{2}	0.281									
Adjusted R ²	0.269									
F	1.272	P > 0.05								

Table 8.27 Coefficient of the Multiple Regression Model^(a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

Review of the hypothesised regression model revealed $\beta = -0.145$, b = -0.093 and t-value = -1.145 for the completely standardised coefficient of $FMDC \rightarrow SMP$. Thus, the regression path is not significant. The results of the correlation coefficient indicated an indirect and insignificant relationship between domestic credit ratio and stock market performance variables, which was small and negligible (Spearman's Rho = 0.011, p = 0.875, 2-tailed). In fact, this relationship was indirect and the results can support the hypothesis, which stated that stock market performance increases as domestic credit ratio decreases. Thus, H_0 cannot be rejected as the domestic credit ratio affected stock market performance variables. These results are consistent with the respondents' findings, as mentioned early in Chapter Two, in that most Libyan public companies suffer from cash flow shortfalls and used bank overdrafts followed by retained earning and trade credit as financial support to cover their payment. The other possible explanation might be that Libyan commercial banks treat public companies more positively due to the banks' policy of obtaining government involvement for producing

CHAPTER 8: TEST OF RESEARCH HYPOTHESES

various liberalisation measures, including ownership companies as more reliable collateral and, therefore, are usually more willing to extend overdraft facilities to public companies. In fact, this leads to reduce public spending and gradual withdrawal of government subsidies. This may provide some explanation for the findings of a study by Kulhánek et al. (2003) in the Czech Republic (the country with the largest banking sector measured by ratio of domestic credit to GDP). This indicator presented about 60 *per cent* of GDP compared with 135 *per cent* of GDP in the euro zone. Furthermore, it found that the ratio of loans to GDP, as compared to the domestic credit ratio, was lower in 2001, at about 36.9 *per cent*. Other observers, however, with regard to the domestic credit ratio were investigated by Fry (1982), who discovered a significant negative relationship for a sample of eighty-five developing countries over the period 1970 to 1995, between economic growth rate and both domestic credit ratio and the ratio of deposit. Levine (2004) found that the estimated coefficient of one-standard-deviation increase in banking credit 0.5 would increase growth 0.7 percentage points perannum.

8.6.2.3 Stock Market Capitalisation Ratio

Test Hypotheses of Stock Market Capitalisation Ratio

- H_0 There is no statistically significant relationship between stock market capitalisation of GDP and stock market performance
- $H_{3.3}$ There is a statistically significant relationship between stock market capitalisation of GDP and stock market performance

The stock market capitalisation ratio is the third dimension of FM variables as presented in Table 8.28, multiple regression model, which is the FMSMC independent variables and SMP variables as dependent variables. A review of the hypothesised regression model revealed $\beta = 0.497$, b = 0.118 and t-value = 4.203 for the completely standardised coefficient of *FMSMC* \rightarrow *SMP*. Hence, the regression path is positively significant at 0.01 level, P < 0.01. The *F*-test for the FMSMC is 17.667, which means this variable can be explained by such a model of the variance of banking assets ratio. Therefore, this result supported $H_{3.3}$, since the calculated t-value is greater than the critical t-value = 1.645. Controlling for all other variables, the correlation coefficient between stock market capitalisation ratio and stock market performance variables was positively significant (Spearman's Rho = 0.228, p = 0.001, 2-tailed). Results of the coefficient correlation analysis indicated the direct relationship between FMSMC variable and SMP variables, by which it can be concluded that the suggestion stock

market performance increases as stock market capitalisation ratio increases, cannot be rejected.

Model	Unstar Coe	tandardised Standardised Coefficients		t	Sig.	95 % Co Interval	for eta	Colline Statist	arity ics	Durbin-
1	β	Std. Error	Beta		6	Lower Bound	Upper Bound	Tolerance 1/VIF	VIF	Watson
(Constant)	1.422	0.417		3.409	0.001	0.600	2.245			1.091
FMSMC	0.497	0.118	0.284	4.203	0.000	0.264	0.730	1.000	1.000	1.981
R^{2}	0.543									
Adjusted R ²	0.536									
F	17.667	P < 0.01								

Table 8.28 Coefficient of the Multiple Regression Model^(a)

Notice a: a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

The results support the findings of Levine and Zervos (1996, 1998a), who built their framework on Atji and Jovanovic's empirical study using various measures of stock market performance. They argue that a well-developed stock market may offer various kinds of impetus to investment and growth rate from development of the banking system. They found that increased stock market capitalisation may improve an economy's ability to mobilise capital and raise further investment. Similar findings in the literature were observed in Demirgüc-Kunt and Levine (1996), where large stock markets measured by market capitalisation ratios to GDP are more liquid, less volatile and more internationally integrated with stronger information disclosure regulations and international accounting standards. Rousseau and Wachtel (2000) adopted two measures of stock market development: the ratio of market capitalisation to GDP and the ratio of total value to GDP. They recognised both as having a significant positive coefficient.

8.6.3 Discussion: Findings from Testing the Financial Maturity Hypotheses

The aim of this section is to *present* the overall multiple regression model and, accordingly, explain the types of financial maturity (independent/predictor variables) and stock market performance (development/outcome variables). The output of the *SPSS* test, as depicted in Table 8.29, reveals very significant information about the model fit under study with those accepted and rejected hypotheses. In this context, the t-test is derived in order to ascertain whether a B value is significantly dissimilar from zero. Consequently, t-tests are considered as measures of whether the predictor is making a significant contribution to the model based on level of significance (α). Critical t-values can be expressed based on the type of test used. Therefore, direction relationships are based upon the hypothesis of addressing the effect on predictors as to the extent of compliance with planning standards. For the 0.05 significant

level, the critical t-values are greater than 1.645 for a one-tailed test and 1.96 for a two-tailed test. Indeed, a two-tailed test of significance was utilised for this study.

Hypotheses		χ^{2}	df	Relationship with SMP variables		F-test	Sig.	Beta	t-test
				Expected	Comment				
FMBA	H_0	114.724	13	-	Rejected	11.915	0.001	0.472	3.452
	H _{3.1}			+	Accepted				
FMDC	H_0	140.963	36	-	Accepted	1.272	0.364	-0.093	-1.145
	H _{3.2}			+	Rejected				
FMSMC	H ₀	06 552	13	-	Rejected	17.667	0.000	0.284	4.203
	H _{3 3}	90.552		+	Accepted				

 Table 8.29 Structural Model Results for Scope of Testing Hypotheses

Source: Analysis of SPSS output from questionnaire survey data.

Results from the multiple regression analysis, as displayed in Table 8.29, indicated that models were reflecting a significant relationship between independent and dependent variables in most areas. The conclusion from this analysis is that financial maturity has affected stock market performance. Therefore, the hypothesis which indicates that there is a significant relationship between financial maturity variables and stock market performance variables cannot be rejected. Besides this, results indicated that the domestic credit ratio divided by GDP has an insignificant relationship with independent stock market performance variables.

8.7 SUMMARY AND CONCLUSION

This chapter has dealt with the empirical work of regression analysis model of this research, beginning with the dataset, research hypotheses and methodology. Findings related to the research's objectives have been *presented* and *analysed*. The subsequent chapter presents results from the regression *analysis* technique *models* for developing the Libyan stock market, *empirically* examining the research *hypothesis* using data collected by the questionnaire survey that may not be tested by utilising financial statements' data. The *hypotheses* examine the independent variables of economic reform programme, macro-economic reform and financial maturity in terms of dependent variables of stock market performance. The research framework model is presented (Chapter Six, Section 6.6) and tested in Chapter Seven. The measurement regression analysis model was conducted with the *SPSS 2003* software package. Data were screened to check for data input errors, distribution, normality, multicollinearity and outliners, given that findings relating to the significant or non-significant relationship between the independent and dependent variables under study, with the acceptance or rejection of each hypothesis, were discussed and are summarised in Table 8.30.

Hypotheses		χ^{2}	df	Relationship with SMP variables		F-test	Sig.	Beta	t-test
				Expected	Comment		, , , , , , , , , , , , , , , , , , ,		
ERPD	$H_0 \\ H_{1.1}$	126.537	15	- +	Rejected Accepted	6.263	0.013	0.174	2.503
ERPC	H ₀ H _{1.2}	110.286	16	- +	Rejected Accepted	5.334	0.022	0.161	2.310
ERPP	H ₀ H _{1.3}	106.891	18	- +	Rejected Accepted	19.861	0.000	0.301	4.457
ERPL	H ₀ H _{1.4}	145.103	14	- +	Rejected Accepted	41.283	0.000	0.413	6.425
MERITR	H ₀ H _{2.1}	173.56	73	- +	Accepted Rejected	2.581	0.110	0.113	1.607
MERER	H ₀ H _{2.2}	155.172	46	- +	Rejected Accepted	7.898	0.005	0.194	2.810
MERIFR	H ₀ H _{2.3}	154.281	47	- +	Accepted Rejected	2.693	0.406	-0.014	-0.994
MERBD	H ₀ H _{2.4}	167.522	47	- +	Accepted Rejected	2.437	0.509	0.047	0.661
MERPCI	H ₀ H _{2.5}	177.655	48	- +	Rejected Accepted	4.928	0.036	0.068	1.964
MERGDP	H ₀ H _{2.6}	148.690	46	- +	Rejected Accepted	4.621	0.017	0.076	1.983
FMBA	H ₀ H _{3.1}	114.724	13	- +	Rejected Accepted	11.915	0.001	0.472	3.452
FMDC	H ₀ H _{3.2}	140.963	36	- +	Accepted Rejected	1.272	0.364	-0.093	-1.145
FMSMC	H ₀ H _{3.3}	96.552	13	-+	Rejected Accepted	17.667	0.000	0.284	4.203

Table 8.30 Structural Model Results for Scope of Research Testing Hypotheses



For the first objective the economic reform programme independent variables, in terms of deregulation, corporatisation, privatisation and liberalisation, are described according to one or more descriptive statistics analysis. Analysis findings are explained in terms of exploring potential correlations or variance between economic reform programme and stock market variables. Clearly, results indicated that all economic reform programme independent variables witnessed a dramatic change within stock market dependent variables. Strong evidence from this hypotheses analysis is that the Libyan government succeeded in implementing its economic reform programme and the fact that Libya's stock market performance improved significantly after adopting the economic reform programme cannot be rejected. The second objective, the macro-economic reform independent variables in terms of interest rates, exchange rates, inflation rate, budget deficit, per-capita income and GDP growth rate, are described and analysed. The results from the previous analysis indicated that most of the macro-economic reform variables tend to affect stock market performance variables. Hence, the hypothesis, which stated that there is a significant relationship between macroeconomic reform and stock market performance, cannot be rejected. In addition, the results indicated that both interest rates and inflation rates have greatest impact upon stock market performance.

CHAPTER 8: TEST OF RESEARCH HYPOTHESES

The third objective, the financial maturity independent variables vis a vis the banking assets ratio divided by GDP, domestic credit ratio divided by GDP and stock market capitalisation ratio divided by GDP, are described and analysed. For this purpose, the multiple regression model was adopted, although this methodology is normally used to predict the success or failure of research hypothesis. The success of the relationship referred to a significant positive change in financial maturity independent variables and the hypothesised dependent stock market performance variables, whilst failure was defined as a non-significant change in these variables. The results of test hypotheses modelling using t-test and beta coefficient have led to the acceptance of some of the research hypotheses and rejection of others. Findings indicated that financial maturity variables tend to affect stock market performance variables, except for domestic credit ratio divided by GDP, where there was an insignificant relationship. Besides emphasising that development of financial structure in Libya is significant for the process of economic growth rate, this finding suggests that the banking system performs various functions against those performed by stock market performance. On the whole, the findings, relating to the research's three objectives have been interpreted, explained and examined in this chapter.

In the following chapter the research's final hypothesis will provide an empirical investigation analysis technique, a comparison between the financing patterns of the developed Libyan stock market and other emerging economy markets to empirically test the fourth objective in the final research hypothesis. The econometric growth model will be *undertaken* and *examined* in the next chapter.

CHAPTER NINE

AN EMPIRICAL INVESTIGATION OF STOCK MARKET PERFORMANCE IN EMERGING MARKET COUNTRIES

9.1 INTRODUCTION

In Chapter Eight three of the research objectives were examined concerning the economic reform programme, macro-economic reform and financial maturity within the stock market, using a multiple regression analysis model. Chapter Nine investigates and analyses the issue of a practice model in developing countries, as stated in objective four. It identifies similarities and differences across these, giving some insight into the situation of the Libyan stock market and its growth within the context of 42 emerging market nations⁴⁵: Algeria, Argentina, Bahrain, Bangladesh, Brazil, Bulgaria, Chile, China, Czech Republic, Egypt, Fiji, Hungary, India, Indonesia, Iran, Israel, Jordan, Kuwait, Lebanon, Malaysia, Mexico, Morocco, Nigeria, Oman, Pakistan, Philippines, Poland, Qatar, Romania, Russia, Saudi Arabia, South Africa, Sudan, Syria, Thailand, Tunisia, Turkey, United Arab Emirates, Venezuela, Yemen and Zimbabwe (plus Libya). However, in terms of its business environment and economic background, Libya differs from some other emerging countries. The stock market was only established as recently as 2006. In order to achieve the fourth research objective in this investigation, a case study (as indicated in Chapters One and Six) focuses on the Libyan financial market. Comparisons are made with other countries bearing some common features similar to the Libyan context where, in each case, the lessons learned for Libya are discussed.

As already indicated in Chapter Two, according to Libyan stock market law, the key objective behind the establishment of the Libyan stock market is to facilitate and contribute to the rising capital structure and to support its allocation process in order to strengthen Libyan economic growth. Consequently, in order to achieve its objective, it is useful to discover whether best practice from other emerging stock markets is transferable to the Libyan situation and context. To this end, this chapter proposes a basic framework for studying the fundamentals of endogenous growth that is also related to key aspects of financial market functions (see

⁴⁵ Emerging markets, as defined by the World Bank, are those with GDP per-capita of 9,256 US\$ or less in 2002. Their bank also classifies economies as low-income measured by GDP (755 US\$ or less), middle-income (756-9,265 US\$) and high-income (9,266 US\$ or more). Low and middle-income economies are usually referred to as developing countries (World Bank, 2002). The emerging stock market, as defined by the IFC, is one that is in transition, increasing in size, activity, or level of sophistication. Mainly, the emerging stock market was based on the World Bank's classification of a fairly narrow list of economies among developing countries.

Chapter Four). Despite this argument, it is possible to make the following two main contributions. First, a simple model of Romer (1986), Lucas (1988), Rebelo (1991), Barro (1991) and Pagano's (1993) kind of endogenous growth economy can be extended in order to fit in with the effects of the financial market. Clearly, it has highlighted the theoretical literature reviewed in Chapters Four and Five, and the expectations of it, and then considered the empirical work arguing that the degree, to which the stock market influences economic growth depends on how effectively it improves capital accumulation, assists capital mobilisation and increases capital investment output. Second, the model is tested on one case study country, Libya. Accordingly, this study illustrated the first attempt to examine the relationship between stock market performance and economic growth in the experience of developing countries. Its results provide relevant empirical knowledge to help identify potential policy reform recommendations for the Libyan economy and for other developing countries with similar economic structures. Some particular questions are also answered in this chapter. Is the emerging economic model transferable to other cultures and/or between cultures of the Libyan situation and context? If it is transferable, what weaknesses in the structure of the economy may appear in the stock market? If they do, what are the key prerequisites for successful stock market performance in Libya? This may suggest that the stock market in question may have an independent empirical link with national economic growth. In order to examine these issues, can the regression analysis of stock market performance variables be the indicators. Simultaneously, control of economic factors may influence the rate of economic growth in order to measure the sensitivity of analysed results to changes in these variables.

This chapter is divided into nine key sections. The second section gives an overview followed by a discussion of the environment of the Libyan stock market within the context of MENA emerging markets and developed countries. Section 9.3 provides previous empirical findings and theoretical framework. Section 9.4 presents additional observations made by the empirical model and variables measure. Section 9.5 contains a brief discussion of data and methods of analysis adopted in this chapter. Section 9.6 presents additional analytical techniques which demonstrate the results of empirical analysis and are reported in Section 9.7. Determinants of banking and financial market development indicators are presented in Section 9.8. Section 9.9 offers a brief summary and conclusion.
9.2 MENA WITHIN THE CONTEXT OF EMERGING MARKETS

As mentioned in previous discussion, strong economic relationships among countries that are in the same region or share the same time zone are expected to reveal higher degrees of similarity in incomes. Thus, it is useful to consider the status and features of emerging markets in the MENA region where most are relatively well in the areas of regulation and supervision as well as in financial openness (Naceur et al., 2008). However, more efforts are needed in order to reinforce the institutional environment and promote non-bank financial sector development (Creane et al., 2004). Specifically, most of these markets have similarities based on the region in which they are located. The analysis, as a quantitative indicator, compares the stock markets in these emerging markets. In MENA countries, for instance, the financial arena is dominated by the banking sector, which represents about 85 per cent of capital market structure⁴⁶. The securities markets within the countries are relatively small despite the region containing some of the developing world's largest institutional investors in international markets. Foreign participation, even in the government securities market, is limited in most nations. Similarly, there have been few direct placements of MENA equities in foreign markets. Since the mid-1980s, the financial system has been gradually liberalised by reducing and eliminating interest rate subsidies to priority sectors, using market-based risk management instruments, for example, by countries in the region. This has been extremely narrow in spite of the relatively limited degree of export diversification. Nevertheless, there are considerable variations across countries, significantly in the equity markets, with the result that the supply of corporate securities remains generally limited, both in absolute and relative terms, to the size of the economies. However, due to significant government credit to the capital market for economic development, the reform programme agenda during the 1990s included plans to revitalise stock markets in some countries and establish them in others. This reflects several factors that have constrained the demand for, and supply of, stocks, including the closed family-owned nature of different companies in the region. While, in several countries, public sector enterprises have continued to play a dominant role in a wide range of business and economic activates, the number of effectively quoted companies has been relatively small and markets have remained thin.

In order to explain the status of the MENA stock market among others, it is worth considering a number of indicators of the market: namely, size, activity, efficiency and performance, in comparison with other countries. Table 9.1 provides data indicators of market capitalisation,

⁴⁶ See IMF, Global Financial Stability Report, September 2004.

market capitalisation ratio of GDP, number of listed companies, value traded besides market performance and turnover ratio for three benchmarked years (1995, 2000 and 2006), which have increased in all countries during the last decade. Aggregate market capitalisation of total emerging markets increased from 2.95 per cent of GDP ratio in 1995 to 7.61 per cent by the end of 2006. In the meantime, the value of trade increased significantly from 9,359.9 million US\$ to 129,245.4 million US\$ in 2005. These overall figures conceal even more individual dramatic stories, regarding both absolute valuation levels and rapid increases in relative valuation. For instance, the markedly high ratios of stock market capitalisation to GDP in 2006 were South Africa (27.79 per cent), Jordan (21.08 per cent), Zimbabwe (18.81 per cent), Malaysia (15.08 per cent), Bahrain (13.35 per cent) and Kuwait (13.06 per cent). Equally revealing are countries with relatively low valuation ratio, including UAE (8.48 per cent) Saudi Arabia (9.36 per cent), Qatar (11.68 per cent) and most other emerging markets. These financial measures may be sensitive to the limited number of transactions on these stock markets since some only open for a few hours daily. Nevertheless, countries that experienced a dramatic increase in market capitalisation relative to GDP between 1995 and 2006 include China (0.58 to 9.17 per cent), Jordan (1.89 to 21.08 per cent), Israel (3.87 to 12.18 per cent), and others. Stock market valuation increases over this period have exceeded any similar growth in corporate profits or national output and, instead, reflect a fundamental reassessment of the value of a nation's common equity. In terms of number of listed companies, India is the leading market with 5937 listed companies, followed by Romania, China and Egypt with 5555, 1086 and 1076 companies, respectively. However, most of these listed companies are not actively trading is mainly due to a relatively new and generally small capital markets when compared with other developing countries, being closed or family-owned enterprises. Table 9.1 further provides two measures of market capitalisation performance and turnover ratio for these countries and regions. While turnover ratio is defined as the total value of shares traded during a year divided by prior year-end market capitalisation, both ratios show market liquidity between 1995 and 2006. Market performance is measured by ratio change in price indexes in US\$, which, in 2006, were Zimbabwe ranked first (943.1 per cent) followed by China (138.3 per cent) and Venezuela (98.1 per cent). The value traded increased in almost all emerging markets between 1995 and 2006 from 9,359.9 to 129,245.4 million US\$ against an increase from 706 to 26392 million US\$ for low-income economies. Several middle-income countries witnessed increases in trading value, including China (49,774 to 1,635,121 million US\$), Egypt (677 to 47,461 million US\$), Iran (741 to 4,886 million US\$) and Jordan (517 to 20,051 million US\$). Although, stock markets in several MENA countries (particularly, the Gulf countries) need to improve liquidity and open their operations to foreign investors.

	Ma	rket Capitali	sation	Marke	t Capitali	sation	Nu	mber of lis	sted		Value Trade	d	Market	Performan	ce (%) b	Turnove	r Ratio (%) p
Country ^a		(Millions US	\$\$)	Ra	tio of GD	P	(Companies	5	(Millions US	\$)	Market	citorinan	ee (70)	Turnove	i Katio (70)
	1995	2000	2006	1995	2000	2006	1995	2000	2006	1995	2000	2006	1995	2000	2006	1995	2000	2006
Low-income	economies																	
Total	3,674	3,609	27,126	1.25	1.52	6.295	238	249	241	706.0	6857.0	26392.0	-10.00	11.00	196.00	10.00	116.00	65.00
Bangladesh	1,338	1,186	3,610	0.34	0.25	0.56	183	221	269	158	768	943	-2.5	28.5	-8.0	13.3	74.8	28.4
Nigeria	2,033	4,237	32,819	0.55	0.92	2.23	181	195	202	14	263	3,559	-26.9	61.4	39.5	0.8	7.3	13.6
Pakistan	9,286	6,581	45,518	1.25	0.89	3.58	764	762	652	3,210	32,974	126,560	-32.6	-13.7	3.2	29.2	486.8	276.8
Yemen ^c	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Zimbabwe	2,038	2,432	26,557	2.85	4.01	18.81	64	69	80	150	279	897	10.6	-22.3	943.1	7.6	11.3	6.2
Lower- middl	le-income ec	conomies																
Total	46,042	86,687	380,852	3.07	2.94	7.55	697	736	662	15,302.8	99,291.3	192,957.2	7.88	-14.33	26.01	24.78	50.69	40.42
Algeria ^c	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
China	42,055	580,991	2,426,326	0.58	4.85	9.17	323	1086	1440	49,774	721,538	1,635,121	-14.5	23.0	138.3	115.9	158.3	102.0
Egypt	8,088	28,741	93,477	1.34	2.9	8.71	746	1076	603	677	11,120	47,461	-10.6	-44.3	12.0	10.9	36.1	54.8
India	127,199	148,064	818,879	3.59	3.22	9.33	5398	5937	4796	13,738	509,812	638,484	-35.2	-28.8	36.3	10.10	306.5	93.1
Indonesia	66,585	26,834	138,886	2.98	1.62	3.81	238	290	344	14,403	14,311	48,831	9.9	-59.3	69.9	25.3	31.5	44.3
Iran	6,552	7,350	37,943	0.72	0.76	1.71	169	304	332	741	1,078	4,886	124.3	44.8	-3.2	15.9	17.9	12.7
Jordan	1,270	4,943	29,729	1.89	5.84	21.08	97	163	227	517	416	20,051	10.6	-20.8	-32.7	11.1	7.9	59.5
Morocco	4,376	10,899	49,360	1.19	2.94	7.55	51	53	55	788	432	1,048	5.8	-20.1	8.9	45.9	8.9	53.3
Philippines	58,859	25,957	68,382	7.79	3.42	5.82	205	228	238	14,727	8,196	11,243	-14.6	-34.1	54.0	26.1	16.4	20.7
Sudan ^c	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Syria ^c	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Thailand	141,507	29,489	141,093	8.42	2.4	6.83	416	381	518	57,000	23,258	100,797	-3.6	-55.7	8.0	41.1	52.9	70.7
Tunisia	3,927	2,828	4,446	2.18	1.45	1.44	26	44	48	663	626	522	30.4	9.0	46.6	19.8	22.6	14.3
Upper- midd	lle-income e	conomies																
Total	72,865	81,119	293,532	5.00	4.00	8.00	194	625	408	18,549.0	33,994.0	103,515.0	-9.00	-16.00	37.00	31.00	33.00	32.00
Argentina	37,783	166,068	79,730	1.46	5.84	3.75	149	127	103	4,594	5,956	4,533	8.6	-25.0	47.9	12.3	4.8	6.4
Brazil	147,636	226,152	711,100	1.92	3.51	6.63	543	459	392	79,186	101,282	254,513	-22.1	-8.4	45.4	47.8	44.6	42.9
Bulgaria	61	617	10,325	0.05	0.49	3.26	26	503	374	4	58	1,509	-15.6	-30.0	65.8	7.7	8.7	19.6
Chile	73,860	60,401	174,556	10.36	8.03	11.97	284	258	244	11,072	6,083	28,753	-2.9	-14.1	29.3	15.3	9.5	18.5
Fiji	67	244	637	0.34	1.45	2.01	4	NA	16	NA	NA	3	NA	NA	NA	NA	NA	0.5
Lebanon	400	1,583	8,279	0.36	0.94	3.64	3	12	11	NA	118	2,035	NA	-18.6	-10.0	NA	6.7	30.8
Libya ^c	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Malaysia	222,729	116,935	235,356	24.7	12.47	15.08	529	795	1027	76,822	58,500	66,904	1.8	-22.8	30.5	35.9	44.6	32.1
Mexico	90,694	125,204	348,345	3.17	2.16	4.15	185	179	131	76,822	45,340	80,095	-27.0	-20.4	46.0	33.0	32.5	27.3
Poland	4,564	31,279	149,054	0.33	1.83	4.36	65	225	267	2,770	14,631	55,041	-9.6	-3.4	58.8	NA	48.1	45.3
Romania	201	1,069	32,784	0.06	0.29	2.67	14	5555	2478	2	236	4,260	NA	-25.3	48.9	71.5	24.3	16.0

 Table 9.1 Stock Market Indicators: Market Size, Activity, Efficiency and Performance during 1995-2006

	Ma	rket Canitali	sation	Market (Capitalicat	ion Patio	Nur	nber of Li	sted		Value Trade	d						
Country ^a	Ivia	(Millions US	Sation SC)	Market	of CDD	IOII Katio	INUI	Componio	sieu		(Millions US	ድ ነ	Market	Performan	$ce(\%)^{b}$	Turnove	r Ratio (%) ^b
Country		(WIIIIOIIS U.S	ο φ)		01 ODF			Companie	8		(Minifolis US	<i>(</i> ه)						
	1995	2000	2006	1995	2000	2006		1995	2000	2006	1995	2000	2006		1995	2000	2006	1995
Russia	15,863	38,922	1,057,189	0.51	1.5	10.69	170	249	309	465	20,312	514,362	-30.6	-24.6	70.8	NA	36.6	46.1
South	280 526	204 052	715 025	18 56	15 /1	27 70	640	616	401	17.048	77 404	312 / 30	1/1.8	173	23.8	6.5	33.2	18.8
Africa	280,520	204,952	/15,025	16.50	13.41	21.19	040	010	401	17,048	77,494	512,459	14.0	-17.5	23.0	0.5	55.2	40.0
Turkey	20,772	69,659	162,399	0.93	2.63	3.07	205	315	314	51,392	179,209	227,615	-13.4	-51.7	-6.2	226.0	196.5	140.5
Venezuela	3,655	8,128	8,251	0.47	0.69	0.45	90	85	53	510	686	666	-31.7	26.5	98.1	9.8	8.8	10.0
High- incon	ne economie:	5																
Total	8,899	21,779	106,336	2.47	3.45	8.59	281	138	136	2881.6	7195.7	194117.3	1.49	-6.93	-4.71	16.77	27.87	74.40
Bahrain	NA	6,624	21,122	NA	8.32	13.35	NA	42	49	NA	247	1,432	NA	-20.7	1.0	NA	3.6	7.4
Czech	15 (()	10.077	49 (04	NT A	1.04	2.4	1 (22	121	20	2 (20	(500	22.975	01.5	0.7	07.0	24.0	577	75 (
Republic	13,004	18,077	48,004	NA	1.94	3.4	1,033	131	29	3,030	0,382	52,875	-21.5	-9.7	21.2	24.9	57.7	/5.0
Hungary	2,399	12,021	41,935	0.54	2.51	3.71	42	60	41	355	12,150	31,183	-29.4	-28.2	34.2	17.3	85.8	83.7
Israel	36,399	64,081	173,306	3.87	5.3	12.18	654	654	612	9,155	23,349	88,771	10.0	14.6	22.3	25.9	36.6	60.5
Kuwait	25,102	20,772	128,940	9.23	5.51	13.06	52	77	163	6,389	4,210	55,886	39.5	-6.5	-11.2	52.8	21.3	43.2
Oman	1,978	3,463	16,158	1.43	1.74	4.52	80	131	124	211	553	321	8.2	-16.0	14.5	14.4	14.2	21.1
Qatar	NA	5,152	61,563	NA	2.9	11.68	NA	22	36	NA	239	20,565	NA	-8.0	-35.5	NA	4.5	27.6
Saudi	40.007	(7.171	226.860	2.97	256	0.26	(0	75	96	6 104	17 212	1 402 027	66	10.1	52 (15.6	07.1	200 4
Arabia	40,907	0/,1/1	320,809	2.87	3.30	9.30	69	15	80	0,194	17,313	1,403,027	0.0	12.1	-32.0	15.0	27.1	288.4
UAE	NA	5,727	138,531	NA	0.86	8.48	NA	54	81	NA	118	113,005	NA	NA	-42.3	NA	NA	62.1
Total	32,870.0	48,298.4	201,961.6	2.95	2.98	7.61	353	437	362	9,359.9	36,834.5	129,245.4	-2.41	-6.57	63.57	20.64	56.89	52.95

Table 9.1 Continued: Stock Market Indicators: Market Size, Activity, Efficiency and Performance during 1995-2006

Notice a: The International Finance Corporation (IFC) defines a stock market as *emerging* if it contains at least one of the following two criteria: it is located in a low or middle-income economy defined by the World Bank and its predictable market capitalisation is low relative to its most recent GDP figures. Since 1995, IFC's definition of an emerging market was based mainly on the World Bank's classification of low and middle-income economies. If a country's GDP per-capita did not achieve the World Bank's threshold for a high-income economy the stock market of that country said to be emerging. Therefore, the IFC has adopted new criteria for a market to graduate from index coverage. For a new market to be located in an economy whose GDP per-capita places in the World Bank's low and middle-income classifications in at least one of the last three years. Based on 2005 data, economies with GDP per-capita of (875 US\$ or less), (876-3,465 US\$), (3,466-10,725 US\$), (10,726 US\$ or more) were classified as low, lower middle, upper middle and high-income economies respectively. For more details, see IFC, 2007 and World Bank's website (http://web.worldbank.org) data and statistics "country groups".

Notice b: Market performance ratio measured by *Per cent* changes in price indexes in US\$. Turnover ratio measured by value shares traded as *per cent* of capitalisation.
 Notice c: NA: Not Applicable.

Source: International Finance Corporation's (IFC's) Emerging Markets' Factbook, now named Standard and Poor's Emerging Market Database, Various Issues.

9.3 EMPIRICAL MODEL AND VARIABLE MEASURE

9.3.1 Empirical Model

In respect of the empirical work, this chapter addresses the gap in this field by providing an empirical analysis of the effect of stock market development on economic growth rate in emerging markets in general, and the Libyan economy in particular, by proposing a simple reasonable framework that suggests that stock market development may influence economic growth rate *vis a vis* improving capital accumulation, increasing productivity of capital investment and facilitation of capital mobilisation. Empirically, therefore, a Two-Stage Least-Squares (2SLS) approach is used to correct the endogeneity of independent variables, particularly level of stock market development⁴⁷. While the 2SLS approach allows this work to address the issue of endogeneity of independent variables by using instrumental variables, this method is the key difficulty in finding suitable tools for endogenous variables. Such tools must, in each case, be sufficiently and highly correlated with the explanatory variables and uncorrelated with the error term. However, in this chapter a one-lagged value of endogenous variables as tools in empirical estimations of the 2SLS regressions is adopted.

As discussed in Chapters Three, Four and Five, centred on the modelling of financial markets and an investigation of their economic growth, claims that the effectiveness of financial markets and level of real activity are closely related are, however, not new. Thus, empirical analysis begins by estimating the following basic approach 2SLS for a sample of 42 emerging markets using base-line variables taken from endogenous growth model:

$$Growth = a + b_1 R + b_2 T + b_3 C + \mathcal{E}$$

$$(9.1)$$

Where the growth indicator, G, is per-capita GDP growth rate averaged over the 1995-2006 period, a is intercept, R is a vector of variables that are generally measured to explain growth such as: per-capita income; investment; human capital and labour including exogenous variables that determine stock market development, T is a vector of variables that are under study where assumption could affect growth rate, C is a vector of selected variables

⁴⁷ 2SLS was widely used during the late 1960s and early 1970s. It was originally provided as a method of estimating parameters of a single structural equation in a system of linear simultaneous equations. It was introduced in early work by Anderson and Rubin (1950), Theil (1953a, 1953b and 1969), Basmann (1957) and Sargan (1958). Farebrother (1999), Anderson (2005), Dhrymes and Muney (2006), Lee (2007), Jallab et al. (2008) and others.

frequently used as controls in estimation (inflation, economic freedom, political stability and the country's openness to trade), and ε is the error term⁴⁸.

To shed greater light on how financial development, particularly stock market development, could influence economic growth rate, a simple theoretical framework based on an endogenous growth model framework of the AK as exemplified by Romer (1986), Lucas (1988), Rebelo (1991) and Pagano (1993) should be presented. Suppose the aggregate of output, y, is produced according to the following constant-to-scale production function:

$$y_t = AK_t \tag{9.2}$$

$$y_{t+1} = AK_{t+1}$$
(9.2*a*)

$$or\left(A = \frac{y_{t+1}}{K_{t+1}}\right) \tag{9.2b}$$

$$y_{t-1} = AK_{t-1}$$
(9.2 c)

$$t = 1, 2, 3....n$$

where y is a scalar dependent variable, K and t are the capital stock and time, respectively, and A is a variable which measures the social marginal productivity of capital or level of total factor productivity which has two components: economic efficiency and level of technological progress. For the sake of simplicity it is assumed that the population is stationary and that the economy produces a single good, which could be invested or consumed: if it is invested it decreases in value at the rate, δ , per period; then gross investment is equal to:

$$I_{t} = K_{t+1} - (1 - \delta)K_{t}$$
(9.3)

$$K_{t+1} = I_t + (1 - \delta)K_t$$
(9.3 a)

where I is the level of investment. In a closed economy with no government policy, capital market equilibrium requires that gross saving S is equal to gross investment I:

$$S_t = \mathbf{I}_t \tag{9.4}$$

In the process of transforming savings into investment, the stock market and other intermediates absorb resources, so a dollar saved by households produces less than one dollar's worth of investment. Since households invest only a fraction of their savings, capital market equilibrium is:

⁴⁸ For more details see Barro, 1991; Hermes and Lensink, 2001; Omran and Bolbol, 2003; Bolbol et al., 2005.

$$\beta S_t = I_t$$

$$or \ \beta = \frac{I}{S_t}$$
(9.5)
(9.5a)

The fraction which produced $(1 - \beta)$ goes to the stock market as commission and fees, and to other financial intermediaries i.e. banks, as the spread between lending and borrowing rates is lost in the process of the stock market and other financial intermediates. The leakages of financial resources may also reflect economic efficiency or what is called X-efficiency of financial markets.

The equation growth rate at time t + 1 can be defined as:

$$G_{t+1} = \frac{\Delta y_t}{y_t} = \frac{y_{t+1}}{y_t} - 1$$
(9.6)

$$G_{t+1} = \frac{y_{t+1}}{y_t} - 1 \tag{9.6a}$$

Since $y_t = AK_{t+1}$, substituting this into (9.6 *a*) yields:

$$G_{t+1} = \frac{AK_{t+1}}{y_t} - 1 \tag{9.6b}$$

Substituting the value of AK_{t+1} from (9.3 *a*) into (9.6 *b*) yields:

$$G_{t+1} = \frac{A[I_t + (1 - \delta)K_t]}{y_t} - 1$$

plummeting all subscripts for a steady state of growth rate:

$$G = \frac{AI}{y} + A \left[\frac{(1 - \delta)K}{y} \right] - 1$$

$$G = \frac{AI}{y} + A \left[\frac{K}{y} - \frac{\delta K}{y} \right] - 1$$
(9.7 c)

Substituting the value of K/y from (9.7 c) yields a steady state growth rate:

$$G = \frac{AI}{y} + A\frac{I}{A} - A\delta\frac{I}{A} - 1$$

= $\frac{AI}{y} - \delta$ (9.8)

Substituting the value of 'I' from (9.5) into (9.8) yields:

$$G = A\beta \frac{S}{y} - \delta \tag{9.9}$$

The equation can be rewritten (9.9) as follows:

$$G = A(\beta s) - \delta$$
(9.10)
Where $s = S/y$

The equation (9.10) suggests that at steady states, the real economic growth rate is some composite of the social marginal productivity of capital, the proportion of total savings that are mobilised to investment and the savings ratio. In identity form, a re-interpretation of the equation (9.10) can be written as follows:

$$G = \ln A + \ln \beta + \ln s \tag{9.11}$$

The key point is that the equation (9.11) represents a composite of the three main mechanisms by financial development which may induce endogenous economic growth:

- 1. It can raise β the proportion of savings channels to investment (i.e. mobilisation and allocation of capital to its efficient use)
- 2. It can contribute to raising s, the savings rate and, thus, physical capital accumulation
- 3. It can increase *A*, the level of total factor productivity by influencing economic efficiency or the level of technological progress.

As financial markets and institutions play a key role in mobilising and allocating resources to their efficient use, Stulz (2000: 11) states: "even though a country has savings, its growth can be stunted because its financial system fails to direct these savings where they can be invested most efficiently". Khasnobis and Bhaduri (2000: 2) commented "a fundamental job of the economy is to allocate capital efficiently. To achieve this, capital is supposed to have high returns and be withdrawn from sectors with poor prospect [...] formal financial markets and associated institutions improve the capital allocation process and thus contribute to economic growth. Financial system collective the small savings of numerous investors for use by agents with entrepreneurial and managerial talents who need funds for large-scale capital investment".

However, financial markets and institutions that are not sufficiently developed may either fall short of receiving all potential savings from households or divert many savings away from investment. With underdeveloped financial markets, individuals may allocate some of their savings to passive tools, such as gold, which cannot be used directly for physical investment. With the development of efficient financial markets and institutions, more savings would be channelled to the financial sector and this would increase the amount of funding directed to the most productive investments and, therefore, lead to economic growth. Nevertheless, financial institutions also provide an information production function. Instead of informed traders producing information *vis a vis* trading and conveying it via prices, banks hire loan officers who produce information for evaluating projects for loan financing. However, most information collected by banks remains as private assets. In the sense of influencing the whole economy, stock markets are certainly more significant than banks. They can, thus, play a unique role in furthering productivity and long-term economic growth.

In summary, the growth rate in output (G) in an economic activity, which is an aggregate measure of performance, is partly determined by effectiveness and efficiency of financial markets and institutions in performing liquidity, sharing information and monitoring functions. Investigation is aimed at analysing and, where possible, explaining variations in the financing model between emerging market economies and the Libyan stock market to display diverse financing behaviour from that of other emerging market economies. This totalled 42 emerging developing market countries: Algeria, Argentina, Bahrain, Bangladesh, Brazil, Bulgaria, Chile, China, Czech Republic, Egypt, Fiji, Hungary, India, Indonesia, Iran, Israel, Jordan, Kuwait, Lebanon, Malaysia, Mexico, Morocco, Nigeria, Oman, Pakistan, Philippines, Poland, Qatar, Romania, Russia, Saudi Arabia, South Africa, Sudan, Syria, Thailand, Tunisia, Turkey, United Arab Emirates, Venezuela, Yemen and Zimbabwe (plus Libya). These nations were selected to examine the financing model in place among these emerging developing nations (see Figure 9.1).





Source: Developed for this research from Chapter Six, Subsection 6.6.3: Background Statistical Technical Model Cross-country Comparison.

The hypothesis tested in this chapter is based on the premise that institutional features of the Libyan business environment may induce the national stock market to display disparate financing behaviour. Therefore, the null hypotheses were formulated.

H4 There is no significant difference in financing behaviour between the Libyan stock market and other emerging market economies

In order to investigate empirically and analyse the strength of the correlation between stock market development and the growth rate of economic activity in developing countries (including Libya), this model can first be used to determine the version of empirical model that can be used to examine this relationship in order to achieve an extension of the endogenous growth model mentioned in previous discussions incorporating stock market development effects. As shown in equation 9.10, the expression for growth is combined with three equations that capture the effect of financial market development on growth: the social marginal productivity of capital (A), or what is termed the level of total factor productivity, savings rate (S) and production of savings channelled to investment (β). Based on the theoretical framework proffered previously, to consider the behavioural nature of these equations it can be assumed that the behaviour of 's' and ' β ' is influenced by some measurements of stock market development as follows:

$$\ln s_t = \varphi_0 + \varphi_1 STOCK_t + \eta_t \tag{9.12}$$

$$\ln \beta_t = \delta_0 + \delta_1 STOCK_t + \mu_t \tag{9.13}$$

where $STOCK_t$ is the vector of stock market development measurement, η_t and μ_t are the white-noise error terms, and t represents the time period. Second, the behaviour of A_t is influenced by stock market development and, following Romer (1986) and Locas (1988) it is influenced by capital-output ratio and human capital accumulation. Hence:

$$\ln A = \rho_0 + \rho_1 STOCK_t + \rho_2 (K_t / Y_t) + \rho_3 (L_t + H_t) + v_t$$
(9.14)

where K/Y is capital-output ratio, L_t is an indicator of labour, H_t is an indicator of human capital and v_t is white-noise error term, population growth (*pop*) is measured and substituted for growth in labour. Population is included as an instrument (determinant of the development a stock market but not economic growth). In review of population growth, human capital including the investment ratio as elements in the vector R usually determines growth. Substituting equations 9.12, 9.13 and 9.14 with 9.11, the following empirical model can be obtained:

$$G_{t} = a + b_{11}pop + b_{12}H_{t} + b_{13}(K_{t}/Y_{t}) + b_{2}STOCK_{t} + b_{3}C + \varepsilon_{t}$$
(9.15)

where ε_t is white-noise error term $(\varepsilon_t \approx iidN(0, \sigma_{\varepsilon}^2))$. The argument in this basic empirical model is that savings and investment activities in financial markets, e.g. stock markets, induce economic growth endogenously. Consequently, with the growing framework, it can be utilised for the growth rate of per-capita real GDP as a measure of real economic growth. That would generate the growth rate of real per-capita GDP as the first variable in the logarithm of the real per-capita GDP series. Investment is included as a measure of development since it indicates that a country has risen beyond subsistence consumption. Ultimately, with most empirical studies in economic growth, secondary school enrolment rates would be required as a proxy for the stock of human capital H_t^{49} . The other tool is a measure of the country's trade openness specifically (ratio of imports and exports to GDP) where the country that is engaged in a large amount of international trade may have more information about the key framework required for a stock market and, more significantly, to help structure the background of the stock market than a closed economy⁵⁰.

Data analyses were not available for a uniform period for each stock market in emerging markets country, as many countries have established their stock markets recently including Libya, thus rendering it quite onerous to estimate a long-term growth equation such as 9.15 and, although no existing methods are reliable for estimating, making simplified assumptions to transfer equation 9.15 into a form could be estimated⁵¹. However recent studies, for instance, Ghura (1997), Beddies (1999) and Bolbol et al. (2005), can be used as the investment output ratio (I/y), rather than using capital output ratio (K/Y). Thus, the basic statistical

⁴⁹ Human capital is empirically proxied by numerous variables including primary and secondary school enrolment rates, total estimated average schooling years and HDI. This was widely used by many studies e.g. Barro, 1991; Mankiw et al., 1992; Levine et al., 2000 and others. Thus, none can perfectly represent human capital, secondary school enrolment rate as mentioned hitherto.

⁵⁰ For more details see, e.g. Minier (2001) for recent discussion of the relationship between trade openness and opening a stock exchange.

⁵¹ One of the most significant difficulties in measuring and/ or estimating accumulation data is in determining initial capital stock k_0 . Most imperial researchers base their measurement or estimate on an assumption of initial capital stock of zero (more details see i.e. King and Levine, 1993a; Levine and Zervos, 1996). Others suggest deriving a guess at initial capital stock in 1950, which assumes that a country was then at its steadystate capital-output ratio (see i.e. Harberger, 1998; Levine and Zervos, 1998a; Beck et al., 1999 and Levine et al., 2000. Most of these assumptions did not offer the true picture of capital stock market accumulation.

model adopted to test the impact of stock market development on the countries' economic growth rate is specified in the following equation:

$$G_{t} = a + b_{11}pop + b_{12}H_{t} + b_{13}(I_{t}/Y_{t}) + b_{2}STOCK_{t} + b_{3}C + \varepsilon_{t}$$
(9.16)

As previously mentioned in Chapter Five, an intensive argument in the literature focuses on the relative virtues of stock markets as opposed to those of the bank. Banks and stock markets, however, may act as complements in providing financial services and promoting economic growth. Therefore, in addition to this chapter's key target, the significant issue of whether the stock market and the banking sector substitute for each other or/are complementary in providing financial services and promoting economic growth in emerging countries, including Libya, is examined. Whether the stock market or the banking sector is better at facilitating economic growth, or if it is the overall level of financial development that is critical for economic growth, is also regarded. In order to examine these issues in this empirical model, accounting is required for the effect of banking sector development. Additionally, another variable (BANK.) that measures banking sector development is added to the basic empirical model as in equation 9.16. If the coefficient estimate of the stock market development indicator (STOCK,) is still significant and relatively stable after including the banking sector development indicator (BANK,), this may imply that the stock market and banking sector are complementary rather than substitutes in providing financial services to the emerging market in general, and the Libyan economy in particular. The scales of the coefficient estimate of stock market and banking sector development indicators provide a measure with an indicator about the significance of stock market development in comparison with banking sector development in enhancing the economy's growth rate.

9.3.2 Variable Measure

This study follows standard practice in the literature findings, both theoretically and empirically, in measuring financial development by aggregate variables that reflect bank-and-market structures. Appendix M, Table M.1 presents a summary of the main previous empirical evidence on finance and economic growth. In this research, as shown in Figure 9.2, these four characteristics were investigated, along with stock market development variables, banking sector development variables, macro-economic reform variables and conditioning information set variables. The following subsections provide a detailed description of said variables.



Figure 9.2 Forms of Variable Measures

Source: Developed for this research from Subsection 9.3.2: Variable Measures.

9.3.2.1 The Stock Market Development Variables

These variables are of particular analytical interest for domestic and international dimensions of stock market development. As mentioned earlier, in Chapter Five, Section 5.6, well-functioning stock markets can play a significant role in economic development processes by performing the following functions: aggregate and mobilise capital; provide risk-pooling and sharing of services; enhance liquidity; monitor managers; exert corporate control. It is difficult, however, to construct precise measures of these functions. In this study indicators are used to suit the purpose of concept of stock market development by constructing proxies for stock market development that are most commonly used as statistics by numerous academics and practitioners (see e.g. Wurgler, 2000). These indicators are associated with size, activity and efficiency of stock market development or market-based system.

- To measure stock market size. *Market capitalisation is used as a share of GDP* ratio, which equals the value of domestic equities (that are traded on domestic market) to GDP. It is assumed that the size of the stock market is positively correlated with the ability to mobilise capital and to diversify risk. However, the size of the stock market does not provide any indication of its liquidity (Naceur e al., 2008).
- To measure stock market activity or liquidity. Value traded is used which is measured by the ratio of *total value traded to GDP* and is related to value of stock market transactions relative to the size of economy. Liquid stock markets are a significant attribute of stock market development because, theoretically, more liquid stock markets improve the share of capital to their best use, influence investment in the long-term and facilitate technological innovation, therefore, enhancing long-term growth rate (see Levine, 1991; Miller, 1991; Bencivenga et al., 1996; Henry, 2000). Liquidity is expected to have a positive impact on stock market capitalisation because the large amount of savings is channelled *vis à vis* stock market.
- To measure stock market efficiency. *Turnover ratio*, which equals the value of trades of shares on national stock markets divided by market capitalisation. More efficient stock markets can faster better resource allocation and spur growth (Bencinvenga et al., 1995).

9.3.2.2 The Banking Sector Development Variables

In this study two indicators of banking sector development or bank-based system are utilised. These have been adopted by Beck et al. (1999a) and Levine et al. (2000) as the ratio of total bank assets to GDP and the value of credit given by the banking sector to the private sector divided by GDP⁵².

• Banking assets. The ratio of *total bank assets to GDP* provides a measure of overall size of the banking sector. Boyd and Smith (1996) suggest that banks and stock markets may behave as complements rather than as substitutes. Empirical findings by Demirgüc-Kunt and Levine (1996) demonstrate that the degree of stock market development is positively related to bank development. Alternatively, Garcia (1989) finds that central banks may create a negative correlation between bank growth and stock market development.

⁵² For Libya, Omran and Bolbol (2003) use both ratio of banking assets and private sector.

• Credit to private sector. Value of credit given by the banking sector to the private sector divided by GDP provides a measure for financial intermediary development following Levine and Zervos (1998), Rousseau and Wachtel (2000), and Beck and Levine (2004). While the first measure does not indicate whether claims of banks are in the public or private sector, the second indicators concentrate on claims to the private sector. Levine (1997: 705) stated "financial systems that allocate more credit to private firms are more engaged in researching firms, exerting corporate control, providing risk management services, mobilising savings, and facilitating transactions than financial systems that simply funnel credit to government or state owned enterprises".

9.3.2.3 Macro-economic Reform Variables

Past and recent empirical findings on growth have identified a large number of variables correlated with growth; therefore, other regressions are included to evaluate the strength of practical correlation between real per-capita GDP growth and each stock market and banking sector development indicator to changes in the conditional information set. Specifically, two variables most generally adopted in growth empirical models are included.

- Income. Real per-capita GDP growth in US\$ is used to measure income level. As income increases, the cyclical component should have a positive influence on the size of stock market development, whereas a higher level of real GDP per-capita means better education, better business environment and wealthier citizens. Accordingly, a positive impact on stock market development is expected.
- Population growth rate. This ratio is the fractional rate at which the number of individuals in a population increases⁵³. While this measure ordinarily refers to change in population over a unit time period, it is often expressed as a percentage of the number of individuals in the population at that period's outset.
- Secondary school enrolment⁵⁴. This ratio is a measure of the size of the level of educational system in the country's various regions and of the demand for teachers, buildings and educational resources.

⁵³ The rate of national growth is expressed as a percentage for each country, commonly between about 0.1 *per cent* and 3 *per cent* (Rosenberg, 2009).

⁵⁴ Enrolment ratios are calculated as the number of students enrolled at a certain level of education as a percentage of the population of the age group that officially corresponds to that level. Enrolment ratios can be higher than 100 *per cent* because some students are younger or older than the corresponding age group.

• Investment ratio. This ratio, which is an important policy instrument, is a measure between the gross fixed capital and gross disposable income. Investment ratio depends on saving.

9.3.2.4 The Conditioning Information Set Variables

In order to examine the sensitivity of results provided, various conditional information sets can be experimented with, as suggested by recent empirical findings on growth⁵⁵. The aforementioned finding on growth rate has identified many variables partially linked with growth. Despite being both technical and time variables, *the degree of freedom* restrictions compel the limit of conditional variables to those most generally used in the previous literature. This includes inflation rate (based on consumer price index), economic freedom measured by business freedom, trade freedom, fiscal freedom, government size, monetary freedom, investment freedom, financial freedom, property rights, freedom from corruption and labour freedom, political stability and the country's opening trade (imports and exports) as a share of GDP⁵⁶ to capture the degree of the economy openness.

9.4 DATA AND SUMMARY STATISTICAL ANALYSIS

9.4.1 The Sample

Data were extracted from various sources. The original intention was to cover all countries in the emerging market region but, given that some established a stock market only in the past few years (Algeria, Libya, Sudan, Syria and Yemen), the sample lists 42 countries, 16 of which are Arab nations. As data were not available for a uniform period for other Arab nations (i.e. Iraq, Mauritania and Palestine) they were excluded from the sample where some countries had no data available on stock market units until recently. Criteria used for selecting emerging market countries, including Libya, were availability and quality of a reasonably large sample of variable measures over a timescale of 12 years from 1995 to 2006. In this study, emerging countries were chosen to be empirically investigated, not only because very few studies have been devoted to the emerging market, but also because size and financial systems' structure differ sensibly between these countries. Most emerging market nations have embarked on farreaching financial reforms since the mid-1980s. Besides, the key findings stemming from emerging market economies could be of interest to other developing countries at the same

⁵⁵ See for instance, Barro (1991), Mankiw et al. (1992), Levine and Renelt (1992), Barro and Sala-I-Martin (1995), Beck et al. (1999), Bolbol et al. (2005) and others.

⁵⁶ As mentioned by Edwards (1993), the literature on endogenous growth argue that economies that are more open to international trade can grow more rapidly by expanding their markets and becoming more efficient.

CHAPTER 9: AN EMPIRICAL INVESTIGATION OF STOCK MARKET PERFORMANCE

stages of financial development; for instance, African, Asian, Eastern European and Latin American countries that are extensively reforming and liberalising their financial systems.

On domestic stock market development indicators, the US\$ amounts of market capitalisation, value traded and turnover ratio on the major database were obtained from the International Finance Corporation's (IFC's) Emerging Markets Factbook (also now named Standard and Poor's Emerging Market Database). The dataset on these stock market measures covers the period 1995-2006 for 37 countries. Data on macro-economic variables per-capita GDP growth, gross domestic product (GDP), consumer price index (CPI) and population growth are retrieved from the International Monetary Fund (IMF), the World Bank Outlook Database and ERS International Macroeconomic Data Set including investment, exports and imports which are retrieved from the World Economic Factbook published by the Central Intelligence Agency, respectively. Banking sector variables, total banking sector assets and total bank claims on the private sector are from annual reports of the Central Banks. Data on secondary school enrolment rates are obtained from the UNESCO Institute for Statistics. Datasets of economic freedom come from the Index of Economic Freedom 2008 and the political stability database from the working paper "Government Matters VII and the Aggregate and Individual Governance Indicators 1996-2007" by Kaufmann et al. (2008). All data are measured in nominal US\$ currency and averaged over the 12 years to smooth growth and explanatory variables.

9.4.2 Summary Statistical Analysis

Table 9.2 provides a summary of statistics for the entire previous variables over the sample period used for estimation during 1995-2006. It can be noted that, prior to estimating the relationship between the economic growth rate and its determinants, it is significant to consider whether or not those data are stationary. This study used testing for stationary to ensure that the variables used in the regressions are not subject to spurious correlation, although the problem of spurious correlations could also emerge when variables are deflated by a stochastic series such as GDP growth. For instance, Phillips (1986) argued that regressions involving non-stationary variables may lead to spurious results demonstrating apparently significant relationships even if variables are generated independently. Additionally, the Phillips-Perron test, which was introduced by Phillips-Perron (1988), was used to determine the order of integration of the variables of interest. The Phillips-Perron test statistics (PP) are modifications of the Augmented Dickey-Fuller t-statistics (ADF) which take

into account the less restrictive nature of the error process⁵⁷. The last two columns, as shown in Table 9.2, show the outcome of unit root test for all the variables adopted. The PP test failed to reject the unit root hypothesis at the 5 *per cent* level for all variables tested. These findings are generally consistent with the hypothesis that all the variables are individually integrated of order one I (1). Therefore, the test rejects the null hypothesis of a unit root for each series in first differences. However, before estimating the regressions the variables need to be transformed by taking the first differences operator to achieve stationary⁵⁸. The *mean* for entering the variables is reported in Table 9.2. Only five variables had the highest mean, such as bank assets ratio (*mean* 71.27), ratio of the sum of export and import (*mean* 62.27), market capitalisation ratio (*mean* 56.88), economic freedom (*mean* 56.68) and domestic credit ratio (*mean* 52.93).

Variables	Code	Mean	Median	SD	Skewness	Kurtosis	PP in Level	PP in 1 st Difference
Real Per-capita GDP Growth Rate	PCYG	2.43	2.67	2.22	-0.59	2.70	-2.723	-5.613
Population Growth	POP	1.51	1.49	1.06	-0.08	0.13	-2.645	-4.558
Secondary School Enrolment Rate	EDUC	44.09	39.89	24.31	0.50	-0.67	-2.745	-5.594
Investment Ratio	INVES	17.60	18.52	7.61	-0.85	0.19	-1.605	-5.401
Market Capitalisation Ratio	MC/GDP	56.88	36.75	80.35	3.24	6.26	-1.671	-3.729
Value Trade Ratio	VT/GDP	28.18	19.14	22.13	0.79	-0.41	-2.745	-5.231
Turnover Ratio	TURNOVER	36.04	25.89	38.22	2.06	4.81	-2.383	-5.072
Bank Asset Ratio	BAR	71.27	72.00	13.15	-0.38	-0.76	-0.893	-6.436
Domestic Credit Ratio	DC/GDP	52.93	49.57	22.22	0.26	-0.66	-2.625	-3.668
Economic Freedom	EF	56.68	57.80	9.44	-0.47	-0.11	-2.010	-5.897
Political Stability	PS	-0.42	-0.36	0.82	0.01	-0.64	-2.511	-5.379
Export + Import/GDP	OPEN	62.27	61.22	16.82	-1.09	2.63	-2.317	-5.572
Inflation Rate	INF	12.30	5.48	17.02	2.43	6.06	-2.543	-4.283

Table 9.2 Descriptive Statistics: Forty Two Emerging Markets (Annual Average 1995-2006)

Notice a: Critical value of PP test is 1 % equal -3.830, 5 % equal -3.029 and 10 % equal -2.655.

Market capitalisation ratio is market capitalisation divided by GDP. Value traded ratio is total value of equity transactions divided by GDP. Turnover ratio is total of equity transactions divided by market capitalisation. Bank asset ratio is total assets of banking sector divided by GDP. Domestic credit ratio is ratio of bank claims on private sector GDP.

Source: Developed for this research from Section 9.4 Data and Summary Statistical Analysis.

⁵⁷ Phillips and Perron (1988) proposed a non-parametric method of controlling for higher-order serial correlation in series. In theory, the PP test is known to be superior, Stock (1994) has shown that the ADF test performs better in practice. Cheung and Lai (1994, 1998 and 2004) and Martinz (1999) have shown that the PP test achieve more favourable evidence than the ADF test. Consequently, in this research only use Phillips and Perron test.

⁵⁸ Another advantage of running the regressions using first differences of the data is to reduce any influence of multicollinearity, which can be improving the robustness of the estimation results.

Table 9.2 additionally displays the entire variables, except for inflation rates, which have no significant values of kurtosis and positive/negative skewness. This indicates that these variables do not have significant departure from a normal distribution⁵⁹. Inflation rates show significantly high values of kurtosis (6.06), indicating that, for the given level of standard deviations, observations for this variable cluster around a central point with a small number of large outliners. Brunner and Hess (1993) argue that the inflation rate was stationary before the 1980s, but it became nonstationary afterward. Table 9.3 presents the correlation matrix among the variables. The first number is the correlation and the second is the *P*-value (a value of less than 0.01 and 0.05 indicates that the correlation is significant at one-per cent and five-per cent levels respectively). Findings from Table 9.3 illustrate that the two stock market development indicators are statistically significant correlated with growth rate (per-capita income real GDP) growth rate). Stock market liquidity measures that value traded and turnover ratios are positively and significantly correlated with growth at the 0.01 level. Correlation between market capitalisation and GDP per-capita is almost 0.18 and is significant at the 0.05 level. In terms of individual countries, rankings can depend significantly, particularly in some countries that are presented as well-developed, in all measures, such as Israel, Jordan, Kuwait, Malaysia and South Africa, as displayed in Table 9.1. Some countries are large and illiquid, such as Bahrain, Chile and Zimbabwe. Other countries have small but active stock markets: China, Pakistan, Saudi Arabia and Turkey.

The correlation between stock market development indicators demonstrates three key points. First, the two measures, namely value traded and turnover ratios, may be substitutes and reveal similar aspects of stock market development. Second, the stock market, since the indicator is measured by the markets' capitalisation ratio, is significantly and positively correlated with liquidity indicators of turnover ratio. This leads to major points that, when the size of stock market increases, it becomes more liquid and efficient. These findings confirm those of Demirgüc-Kunt and Levine (1996) that large stock markets measured by equity capitalisation to GDP are more liquid and less volatile. Finally, liquidity and market capitalisation are significantly correlated where the correlation coefficient among these variables is below the 0.05 level on average. This suggests that the various indicators capture different aspects of stock market to mobilise capital and hedge risk, while market liquidity may reveal information on firms and improve corporate governance (Holmstrom and Tirole, 1993), thereby facilitating growth. On the other hand, the role of information in this process has been questioned by

⁵⁹ Values of *skewness* and kurtosis are 0 and 3 respectively, if observed distribution is normal.

CHAPTER 9: AN EMPIRICAL INVESTIGATION OF STOCK MARKET PERFORMANCE

Stiglitz (1985) who argues that developed stock markets quickly reveal information through price changes, creating a free rider problem and reducing investor incentives to spend resources to conduct costly search. Table 9.3 further shows that the ratio of domestic banks on the private sector to GDP is positively and statistically correlated with growth rate at the 0.01 level. Empirically, Levine et al. (2000) found that the private credit to GDP is statistically significantly correlated with economic growth rate at the *five-per cent level* in the simple, policy and full conditioning information set regressions. They suggested that the exogenous component of financial intermediary development is closely tied to long-term rates of percapita GDP growth. Additionally, there are strong correlations between both the bank sector and stock market development indicators. This means that a possible explanation for this correlation is that the stock market transmits information useful to creditors. This revelation of information may make lending to a publicly quoted company less risky. Consequently, greater development in the stock market may increase the ability of companies to obtain credit and encourage banks to provide more credit. These findings confirm the results of Demirgüc-Kunt and Levine (1996), Levine and Zervos (1996, 1998b) and Garcia and Liu (1999) that banking sector and stock market development are complements rather than substitutes.

	PCYG	POP	EDUC	INVES	MC/GDP	VT/GDP	TURNOVER	BAR	DC/GDP	EF	PS	OPEN	INF
PCVG	1.000	-0.330**	0.092*	0.105*	0.180*	0.154**	0.216**	0.175**	0.179**	-0.037	0.165**	0.100*	0.037
1010		0.000	0.039	0.018	0.017	0.001	0.000	0.000	0.000	0.410	0.000	0.024	0.407
POP		1.000	-0.285**	-0.375**	0.015	-0.120**	-0.228**	-0.175**	-0.105*	0.030	-0.244**	0.062	-0.232**
101			0.000	0.000	0.736	0.009	0.000	0.000	0.020	0.505	0.000	0.169	0.000
FDUC			1.000	0.027	-0.079	0.054	0.095*	-0.128**	-0.004	-0.215**	-0.127*	-0.172**	0.180**
LDOC				0.541	0.076	0.222	0.035	0.004	0.936	0.000	0.006	0.000	0.000
INVES				1.000	0.129**	0.134**	0.086	0.387**	-0.023	0.107*	0.058	-0.024	0.060
IIIVES				1.000	0.004	0.003	0.055	0.000	0.604	0.018	0.210	0.594	0.185
MC/GDP					1.000	0.757**	0.323**	0.346**	0.051	0.514**	0.190**	-0.006	-0.247**
Mic/ODI					1.000	0.000	0.000	0.000	0.258	0.000	0.000	0.899	0.000
VT/GDP						1.000	0.745**	0.435**	0.221**	0.416**	0.151**	-0.110*	-0.116*
V I/ODI						1.000	0.000	0.000	0.000	0.000	0.001	0.014	0.011
TURNOVER							1.000	0.389**	0.345**	0.235**	0.110*	-0.146*	-0.019
TORUSOVER							1.000	0.000	0.000	0.000	0.019	0.001	0.676
BAR								1.000	0.320**	0.425**	0.291**	0.120**	-0.138**
Drift								1.000	0.000	0.000	0.000	0.007	0.002
DC/GDP									1.000	-0.038	0.077	-0.035	-0.046
DO/ODI									1.000	0.406	0.099	0.432	0.305
EF										1 000	0.414**	0.046	-0.438**
										1.000	0.000	0.313	0.000
PS											1.000	0.194**	-0.249**
15											1.000	0.000	0.000
OPEN												1.000	-0.068
												1.000	0.135
INF													1.000

 Table 9.3 Correlation Matrix among Variables

Notice a: * Correlation is significant at 0.05 level (2-tailed). ** Correlation is significant at 0.01 level (2-tailed).

Notice b: (P-Value in Parentheses). Growth is real per-capita GDP growth. POP is population growth rate. EDUC is secondary school enrolment rate. INVES is value investment as percentage of GDP. MC/GDP is market capitalisation ratio divided by GDP. VT/GDP is value traded ratio divided by GDP. TURNOVER is the total of equity transactions divided by market capitalisation. BAR is the total asset of the banking sector divided by GDP. DC/GDP is domestic credit ratio in the ratio of bank claims on private sector GDP. EF is economic freedom. PS is political stability and OPEN is a measure of openness calculated as the sum of imports and exports divided by GDP. INF is inflation rate.

Source: Developed for this research from Section 9.4 Data and Summary Statistical Analysis.

9.5 ANALYTICAL TECHNIQUE

Results obtained from equation 9.16 estimated using 2TSLS with OLS, where these estimates are reported to have been founded upon both bank and market-based indicator development measures from 1995 to 2006. In order to reduce the potential econometric problems, such as heteroscedasticity, GDP growth was used as a deflator in accordance with the suggestion of Demirgüc-Kunt and Levine (1996) and Levine and Zervos (1996, 1998b). In papers, Ketteni et al. (2007) empirically assess the relationship between financial development and economic growth rate in order to explore possible non-linearity. They utilise the same data set as in the previous theoretical study but employ nonparametric estimation techniques. Therefore, they acknowledge that the relationship between finance and growth is linear when account is taken of non-linearity between initial per-capita income and human capital on the one hand and economic growth on the other. This chapter additionally uses heteroscedasticity-consistent standard error test based on the White test, but covariance for mitigating heteroscedasticity in calculating statistics, original regressors and their squares, could not reject the null of homoscedasticity. It is important, before analysing the results, to mention that all regression models estimated in the first difference are stationary. To avoid misspecification of the estimated regressions, two dummy variables have been utilised, dummy intercept and dummy interaction variables, to identify significant differences in the relationships between per-capita real GDP growth and the explanatory variables. Practices established in the literature by used natural log (L) for initial per-capita income IPCY to control for convergence⁶⁰. According to neoclassical theory (Chapter Four), the coefficient associated to per-capita income represents the convergence effect and thus should be negative.

9.6 Empirical Results and Analysis

Table 9.4 illustrates bank-based indicator results relating to BAR and DC/GDP. In panel A: BAR from model 2 represents basic regression, where its explanatory variables appear in all regressions. Dependent variables are economic growth proxied by per-capita real GDP growth. Other variables are natural logarithm of initial per-capita income, population growth rate, secondary school enrolment rate and investment ratio and natural logarithm of banking assets ratio including change of per-capita income. As the results demonstrate, LIPCY is strongly positive and significant, indicating that there is conditional convergence to the higher per-

⁶⁰ If convergence is confirmed, then a country a relatively lower level of initial per-capita GDP will grow faster since it is that much farther away from its steady state and must catch up.

capita income levels of some emerging market nations such as China, Hungary, India, Iran, Poland, Qatar, Russia, Sudan and UAE . The investment ratio is, as expected, positive and significant. Model 3 adds the interaction term between BAR and change of per-capita income, both positive and significant, thus validating the key hypothesis of this study. The regression explains 66.5 *per cent* of the variation in measuring the economic growth of emerging market nations. *F*-statistics for regression, F= 125.137, reject the null hypothesis of no explanatory power for the regression as a whole *at the one-per cent level*.

Models 6-8 include standard control variables: economic freedom ratio (EF), political stability (PS), sum exports and imports ratio (OPEN) and inflation (INF). They provide a robustness *test* for this result since the interaction term, LBAR x Δ PCI, remains positive and significant. Note that EF has a significant negative effect on PCYG, whereas PS has an insignificant effect since it accurately reflects the difference between political stability and national situations. As observed by AON Corporation (2004), many Middle Eastern nations have witnessed their political stability downgraded owing to continued uncertainty in Iraq, possibility of further coalition activity elsewhere in the region and terrorism. AON Corporation (2004) adds that ongoing unrest and terrorist activity in the Middle East continue to pose a high risk to foreign investors. In the future, the AON Corporation (2005) explains, Middle Eastern nations continue to cause concern, particularly Iraq, where the outlook is increasingly pessimistic. This region is a common site for many conflicts, imposition of economic sanctions and wars. The Iraq War, fighting international terrorism, sanctioning Syria, finding a solution for the Palestinian problem and the Iranian nuclear programme are all examples of potential political crises. As revealed by the UN and US, such incidents include Libya (1986-1999), Iraq (1990-2003), Somalia (1992- to data), Sudan (1996-2001) and Yemen (1994). Wars which happened in the region are related to both the Arab-Israel conflict and related crises.

The region has also been host to other disputes and wars not directly involving Israel. Many such conflicts are related to Iraq, including the Iran-Iraq War (1980-1988), the Iraq invasion of Kuwait (1990), the Gulf War (1991) and the Iraq War (2003). Other wars that occurred in the context of the Middle East include the Yemeni War (1979) and the Lebanon War (1982, 1995-2000). Furthermore, the region regularly witnesses coups d'état. Currently-serving leaders who came to power via coups include those of Libya (1969), Tunisia (1987), Sudan (1989), Qatar (1995) and Mauritania (2005, 2008). More surprising is the positive effect of the INF ratio. This could be attributed not only to the negative impact that interest rate has on the growth, but also effectiveness of fiscal policy, given that most emerging markets follow a fixed exchange rate system, which makes monetary policy ineffective, especially among Arab

CHAPTER 9: AN EMPIRICAL INVESTIGATION OF STOCK MARKET PERFORMANCE

nations. The effect of the growth of openness, as measured by ratio of export, plus import to GDP is negatively and statistically correlated in a significant ratio with economic growth. This result is consistent with the empirical findings of Bloom et al. (1999) and Bassanini et al. (2001), besides others. The implication of this result is that when the value of exports and imports relative to a nation's GDP increases, economic growth increases likewise. Exports, therefore, can affect economic growth via two channels, low domestic demand, while exports provide an outlet for this excess production and generate income. Second, in the long-term, exports aid growth because they tend to gather technical progress and further savings. Additionally, they improve the national credit rating by generating hard currency and, thus, facilitating ease of foreign loans.

Table 9.4 reports results for the second bank-based indicator, the DC ratio. They duplicate almost exactly those obtained for BAR, except that the LDC ratio is negative in models 3-4. The INF is also negative in model 7, but investment ratio, as expected, has a strongly significant and effective relationship exhibiting economic growth. The result is consistent with the findings by Kean (1983) and Singh (1992a) that the pricing of shares are critical as to how well the stock market can perform its allocative functions. An efficient pricing process will reward the well managed and profitable firms by valuing their shares more highly than these of unsuccessful and unprofitable firms, which in aggregate has an effect on economic development in the long-term. As can be observed, the regression explains about 69.6 per cent of variation in measuring economic growth of emerging market economies. However, Fstatistics for the regression decline to 75.106, but still reject the null hypothesis of no explanatory power of the regression as whole at the one-per cent level. In panels A and B it can be noted that population growth rate as measured by the labour force has a negative effect on efficiency-related growth rate. This result is related to human capital rate which affects percapita real GDP growth. Individual coefficients on the regression enter with appropriate signs. Human capital stock proxied by secondary school enrolment rate, enters positively, but not statistically significant at the one-per cent level. This could be explained by reference to the fact that, among most emerging Africa and Arab market nations, unemployment has risen. 2006 saw the highest unemployment with Algeria (15.7 per cent), Bahrain (15 per cent), Iran (15 per cent), Lebanon (20 per cent), Libya (30 per cent), Mauritania (20 per cent), Oman (15 per cent), South Africa (25.5 per cent), Sudan (18.7 per cent), Yemen (35 per cent) and Zimbabwe (80 per cent) all suffering from high levels of unemployment, especially the structural type where there is a mismatch between the output of the education system, the labour force and the economic needs.

Dependent Variable: PCYG												
Independent Variables	Model 1	Model 2	Model 3	Model 4	Model 5	Model 6	Model 7	Model 8				
Panel A: BAR												
a	1.194	-0260	0.917	-0.464	-2.645	-2.828	-2.326	-2.306				
Constant	(3.732)**	(-0.151)	(0.565)	(-0.270)	(-1.698)	(-1.809)	(-1.517)	(-1.511)				
LIDCV	4.600	4.592	4.283	4.174	3.875	3.845	3.724	3.724				
LIPCY	(22.398)**	(22.319)**	(21.731)**	(21.019)**	(21.538)**	(21.200)**	(20.760)**	(20.798)**				
DOD	-0.35	-0.031	-0.96	-0.090	-0.143	-0.159	-0.195	-0.195				
POP	(-0.396)	(-0.347)	(-1.155)	(-1.076)	(-1.886) [†]	(-2.065)*	(-2.577)*	(-2.588)*				
FDUC	0.030	0.032	0.034	0.028	0.035	0.040	0.044	0.044				
LDUC	(0.998)	(1.042)	(1.183)	(0.975)	(1.341)	(1.530)	$(1.723)^{r}$	(1.724)*				
INVES	0.030	0.026	0.025	0.025	0.024	0.24	0.019	0.019				
INVES	(2.913)**	(2.412)*	(2.457)*	(2.417)*	(2.575)*	(2.543)*	(2.051)*	(2.077)*				
LBAR		0.817	(0.242	2.030	3.311	3.363	3.371	3.372				
		(0.858)	(0.270)	(1.945)*	(3.499)**	(3.553)**	(3.656)**	(3.663)**				
LBAR x A PCI			0.002	0.002	0.002	0.002	0.002	0.002				
			(7.383)**	(7.292)**	(6.835)**	(6.884)**	(6.857)**	(6.868)**				
EF				-0.032	-0.033	-0.032	-0.039	-0.039				
				(-3.699)**	(-4.180)**	(-4.113)**	(-4.830)**	(-5.007)**				
PS					-0.001	-0.006	-0.009	-0.009				
					(-0.020)**	(-0.217)	(-0.351)	(-0.373)				
OPEN						0.012	0.007	0.007				
	-					(1.237)	(0.738)	(0.731)				
INF							0.001					
							(0.159)					
R^2	0.616	0.616	0.665	0.675	0.711	0.712	0.716	0.716				
Adi. R (%)	0.612	0.611	0.659	0.669	0.704	0.704	0.707	0.708				
F-Ratio	152.516**	122.075**	125.137**	108.940**	104.311**	93.036**	82.654**	92.108**				
D.W.	1.621	1.628	1.425	1.468	1.388	1.389	1.348	1.416				
Panel B: DC/GDP					•							
0	1.194	0.817	1.315	2.906	1.775	1.696	2.168	2.114				
Constant	(3.732)**	(1.199)	(1.998)*	(3.405)**	(2.264)*	(2.151)*	(2.711)**	(2.729)**				
LIDCV	4.600	4.597	4.392	4.332	4.001	3.979	3.845	3.844				
LIPCI	(22.398)**	(22.356)**	(21.986)**	(21.476)**	(21.733)**	(21.438)**	(20.908)**	(20.936)**				
POP	-0.35	-0.025	-0.071	-0.069	-0.122	-0.134	-0.170	-0.169				
FOF	(-0.396)	(-0.278)	(-0.827)	(-0.793)	(-1.544)	(-1.677) [†]	(-2.168)*	(-2.159)*				
FDUC	0.030	0.029	0.032	0.026	0.033	0.037	0.040	0.040				
EDUC	(0.998)	(0.963)	(1.118)	(0.863)	(1.213)	(1.352)	(1.519)	(1.523)**				
INVES	0.030	0.030	0.026	0.031	0.036	0.036	0.032	0.032				
IIII III	(2.913)**	(2.954)**	(2.677)**	(3.145)**	(3.995)**	(3.984)**	(3.570)**	(3.565)**				
LDC/GDP		0.214	-0.018	-0.160	0.373	0.372	0.434	0.442				
EDG/GDI		(0.627)	(-0.054)	(-0.438)	(1.100)	(1.096)	(1.310)	(1.339)				
LDC/GD x APCI			0.002	0.002	0.002	0.002	0.002	0.002				
			(5.995)**	(5.908)**	(5.849)**	(5.875)**	(5.921)**	(5.923)**				
FF								0.025				
				-0.024	-0.019	-0.019	-0.026	-0.023				
				-0.024 (-3.019)**	-0.019 (-2.658)**	-0.019 (-2.573)*	-0.026 (-3.449)**	(-3.498)**				
PS				-0.024 (-3.019)**	-0.019 (-2.658)** -0.004	-0.019 (-2.573)* -0.008	-0.026 (-3.449)** -0.013	-0.023 (-3.498)** -0.013				
PS				-0.024 (-3.019)**	-0.019 (-2.658)** -0.004 (-0.165)	-0.019 (-2.573)* -0.008 (-0.314)	-0.026 (-3.449)** -0.013 (-0.505)	-0.023 (-3.498)** -0.013 (-0.477)				
PS OPEN				-0.024 (-3.019)**	-0.019 (-2.658)** -0.004 (-0.165)	-0.019 (-2.573)* -0.008 (-0.314) 0.010 (0.027)	-0.026 (-3.449)** -0.013 (-0.505) 0.004 (0.400)	-0.023 (-3.498)** -0.013 (-0.477) 0.004				
PS OPEN				-0.024 (-3.019)**	-0.019 (-2.658)** -0.004 (-0.165)	-0.019 (-2.573)* -0.008 (-0.314) 0.010 (0.937)	-0.026 (-3.449)** -0.013 (-0.505) 0.004 (0.409)	-0.023 (-3.498)** -0.013 (-0.477) 0.004 (0.427)				
PS OPEN INF				-0.024 (-3.019)**	-0.019 (-2.658)** -0.004 (-0.165)	-0.019 (-2.573)* -0.008 (-0.314) 0.010 (0.937)	-0.026 (-3.449)** -0.013 (-0.505) 0.004 (0.409) -0.002	-0.023 (-3.498)** -0.013 (-0.477) 0.004 (0.427)				
PS OPEN INF				-0.024 (-3.019)**	-0.019 (-2.658)** -0.004 (-0.165)	-0.019 (-2.573)* -0.008 (-0.314) 0.010 (0.937)	-0.026 (-3.449)** -0.013 (-0.505) 0.004 (0.409) -0.002 (-0.278)	-0.023 (-3.498)** -0.013 (-0.477) 0.004 (0.427)				
PS OPEN INF R ²	0.616	0.616	0.649	-0.024 (-3.019)** 0.656	-0.019 (-2.658)** -0.004 (-0.165) 0.691	-0.019 (-2.573)* -0.008 (-0.314) 0.010 (0.937) 0.692	-0.026 (-3.449)** -0.013 (-0.505) 0.004 (0.409) -0.002 (-0.278) 0.696	-0.023 (-3.498)** -0.013 (-0.477) 0.004 (0.427) 0.696				
PS OPEN INF R^2 Adi, R (%)	0.616	0.616	0.649	-0.024 (-3.019)** 0.656 0.649	-0.019 (-2.658)** -0.004 (-0.165) 0.691 0.684	-0.019 (-2.573)* -0.008 (-0.314) 0.010 (0.937) 0.692 0.683	$\begin{array}{c} -0.026 \\ (-3.449)^{**} \\ -0.013 \\ (-0.505) \\ 0.004 \\ (0.409) \\ -0.002 \\ (-0.278) \\ 0.696 \\ 0.687 \end{array}$	(-3.498)** -0.013 (-0.477) 0.004 (0.427) 0.696 0.688				
PS OPEN INF R ² Adj. R (%) F-Batio	0.616 0.612 152.516**	0.616 0.611 121.897**	0.649 0.644 116.550**	-0.024 (-3.019)** 0.656 0.649 99.504**	-0.019 (-2.658)** -0.004 (-0.165) 0.691 0.684 94.986**	-0.019 (-2.573)* -0.008 (-0.314) 0.010 (0.937) 0.692 0.683 84.499**	$\begin{array}{r} -0.026 \\ (-3.449)^{**} \\ -0.013 \\ (-0.505) \\ 0.004 \\ (0.409) \\ -0.002 \\ (-0.278) \\ 0.696 \\ 0.687 \\ 75.106^{**} \end{array}$	-0.023 (-3.498)** -0.013 (-0.477) 0.004 (0.427) 0.696 0.688 83.677				
PS OPEN INF R ² Adj. R (%) F-Ratio D.W.	0.616 0.612 152.516** 1.621	0.616 0.611 121.897** 1.623	0.649 0.644 116.550** 1.504	-0.024 (-3.019)** 0.656 0.649 99.504** 1.527	-0.019 (-2.658)** -0.004 (-0.165) 0.691 0.684 94.986** 1.392	-0.019 (-2.573)* -0.008 (-0.314) 0.010 (0.937) 0.692 0.683 84.499** 1.393	$\begin{array}{r} -0.026 \\ (-3.449)^{**} \\ -0.013 \\ (-0.505) \\ 0.004 \\ (0.409) \\ -0.002 \\ (-0.278) \\ \hline 0.696 \\ 0.687 \\ 75.106^{**} \\ 1.340 \\ \end{array}$	-0.023 (-3.498)** -0.013 (-0.477) 0.004 (0.427) 0.696 0.688 83.677 1.462				

Table 9.4 Regression	Results:	Bank-based	Indicators
----------------------	-----------------	------------	------------

Notice a: Figures between parentheses are heteroscedasticity-consistent t-statistics. * Significant at 0.05 level, ** Correlation is significant at 0.01 level, [†] significant at 0.001 level.

Notice b: The sample period is 1995-2006. PCYG is per-capita income growth, LIPCY is natural logarithm of initial per-capita income and POP is population growth rate. EDUC is secondary school enrolment rate. INVES is value investment as percentage of GDP. LBAR is the natural logarithm of bank asset, LBAR x Δ PCI interaction with change per-capita income. DC/GDP is domestic credit ratio in the ratio of bank claims on private sector GDP, LDC/GD x Δ PCI interaction with change per-capita income. EF is economic freedom. PS is political stability and OPEN is a measure of openness which is calculated as sum of imports and exports divided by GDP. INF is inflation rate.

R Adjusted is the usual R^2 , adjusted for the degrees of freedom; DW is the Durbin-Watson test for residual serial correlation. *F*-test tests the null hypothesis that all coefficients apart from the intercept are zero.

As for market-based development indicators, their results are listed in Table 9.5, and it can be observed from all panels that the investment ratio remains positive and significant at the oneper cent level but, unlike the comparable result for the MC ratio, it has no significant independent effect on investment efficiency, while its interaction with changing PCI is positive and significant. However, MC ratio can have a positive impact on investment efficiency occurring with more private resource flows. Model 3 in Table 9.5 demonstrates that the most important result in this regression is that stock market indicators as proxied by market capitalisation adjusted for GDP enter an extremely significant at one-per cent level positive correlation with economic growth. A one-percentage point increase in the growth rate of market capitalisation to GDP is associated with a 0.420 percentage point increase in real percapita GDP growth. This result is inconsistent with the findings in Levine and Zervos (1996, 1998b). Their results suggested that market capitalisation is not a sound predictor of economic growth. However, this finding is consistent with that of models 5-6 in Table 9.5. Results in Tables 9.4 and 9.5 mirror, to a large extent, those obtained for bank development indicator measures, with the interaction terms between LIPCY ratio and each of the value-traded and turnover ratios being positive and significant. The fit of the regressions with an R^2 greater than 67 per cent is relatively satisfactory. The F-ratio in each cited regression rejects the null hypothesis of having no explanatory power for the regression as a whole being better than at the one-per cent level.

It should be noted that the coefficient of the banking-based development indicators (0.442) is larger than the coefficient of the stock market-based development indicator (0.018), suggesting that banking sector development has a greater effect upon economic growth of banks than on stock market development. A one percentage point increase in the growth of banks lending to the private sector to GDP is associated with a 0.442 percentage point increase in real percapita GDP growth. The fit of this regression with an R^2 of 69.6 *per cent*, also the *F*-statistics, F=83.677 rejecting the null hypothesis of no explanatory power for the regression as a whole is better than *at the one-per cent level*, while a one percentage point increase in the growth rate of market capitalisation to GDP is associated with a 0.018 percentage point increase in real per-capita GDP growth. This may simply reflect the translation ratio between market capitalisation and eventual finance via new share issues. It may additionally indicate that the banking sector is more significant for real growth than the stock market in a developing country. For an emerging market without a mature entrepreneurial experience, banks will be relatively more significant. This argument appears to be applicable to the Libyan economy during the initial stages of the period under study.

	Dependent Variable: PCYG												
Independent Variables	Model 1	Model 2	Model 3	Model 4	Model 5	Model 6	Model 7	Model 8					
Panel A: MC/GDP													
Constant	1.194	1.199	1.230	2.402	2.316	2.222	2.841	2.831					
Constant	(3.732)**	(3.702)**	(3.848)**	(3.780)**	(4.045)**	(3.815)**	(4.660)**	(4.752)**					
LIPCY	4.600	4.588	4.515	4.440	4.097	4.077	3.936	3.935					
	(22.398)**	(21.890)**	(21./14)**	(20.925)**	(21.011)**	(20.760)**	(20.213)**	(20.253)**					
POP	-0.35	-0.007	-0.022	-0.023	-0.100	-0.117	-0.155	-0.155					
	0.030	0.033	0.034	0.029	0.040	0.044	(-1.800)	(-1.808)					
EDUC	(0.998)	(1.094)	(1.112)	(0.927)	(1 414)	(1.540)	(1.598)	(1.601)					
	0.030	0.033	0.034	0.037	0.039	0.039	0.036	0.036					
INVES	(2.913)**	(3.173)**	(3.276)**	(3.576)**	(4.119)**	(4.117)**	(3.747)**	(3.767)**					
LMC/CDD		-0.302	0.420	0.219	-0.102	-0.115	0.016	0.018					
LMC/GDP		(-1.735) [†]	(2.402)*	(1.073)	(-0.527)	(-0.593)*	(0.081)	(0.090)					
I MC/GD x APCI			0.015	0.015	0.012	0.012	0.012	0.012					
			(3.347)**	(3.429)**	(2.906)**	(2.950)	(3.023)**	(3.029)**					
EF				0.022	-0.018	-0.018	-0.026	-0.026					
				(2.239)*	(-2.112)*	(-2.007)	(-2.878)**	(-2.907)**					
PS					-0.016	-0.020	-0.023	-0.023					
					(-0.389)	0.010	0.004	0.004					
OPEN						(0.889)	(0.370)	(0.376)					
						(0.005)	-0.001	(0.270)					
INF							(-0.081)						
R^2	0.616	0.613	0.624	0.630	0.662	0.663	0.668	0.668					
Adj. R (%)	0.612	0.608	0.618	0.623	0.654	0.654	0.658	0.659					
F-Ratio	152.516**	117.476**	102.170**	87.147**	81.818**	72.769**	65.157**	72.618**					
D.W.	1.621	1.641	1.616	1.629	1.487	1.489	1.434	1.563					
Panel B: VT/GDP				•		•							
	1.194	1.321	1.329	2.321	2.110	2.022	2.555	2.563					
Constant	(3.732)**	(3.811)**	(3.828)**	(3.837)**	(3.833)**	(3.615)**	(4.391)**	(4.594)**					
LIDOV	4.600	4.619	4.613	4.555	4.092	4.067	3.886	3.886					
LIPCY	(22.398)**	(21.581)**	(21.522)**	(20.842)**	(20.846)**	(20.515)**	(19.898)**	(19.932)**					
POP	-0.35	(-0.340)	(-0.341)	(-0.013)	-0.130	$(-1.689)^{\dagger}$	(-1.189) (-2.273)*	(-2, 278)*					
	0.030	0.035	0.033	0.026	0.033	0.038	0.036	0.036					
EDUC	(0.998)	(1.155)**	(1.089)	(0.815)	(1.170)	(1.301)	(1.288)	(1.289)					
INIVEC	0.030	0.032	0.032	0.036	0.039	0.039	0.035	0.035					
INVES	(2.913)**	(3.053)	(3.096)**	(3.439)**	(4.050)**	(4.036)**	(3.680)**	(3.711)**					
I VT/GDP		-0.122	0.125	-0.050	0.043	0.049	0.154	0.154					
ETTIODI		(-1.381)	(1.412)	(-0.485)	(0.462)	(0.526)	(1.618)	(1.620)					
LVT/GD x ΔPCI			-0.004	-0.002	-0.035	-0.036	-0.046	-0.046					
			(-0.509)	(-0.265)	(-2.401)*	(-2.486)**	(-3.212)**	(-3.227)**					
EF				(-2.047)*	$(-1.780)^{\dagger}$	$(-1.718)^{\dagger}$	-0.24	-0.024					
				(-2.047)	-1.156	-0.166	-0.208	-2 208					
PS					(-2.408)*	(-2.527)*	(-3.239)**	(-3.244)**					
OPEN						0.010	0.007	0.007					
						(0.910)	0.000	(0.634)					
INF							(0.047)						
R^2	0.616	0.611	0.611	0.617	0.659	0.660	0.670	0.670					
Adj. R (%)	0.612	0.605	0.604	0.609	0.651	0.651	0.660	0.661					
F-Ratio	152.516**	115.098**	95.494**	81.548**	80.030**	71.193**	65.521**	73.027					
D.W.	1.621	1.599	1.597	1.623	1.417	1.415	1.342	1.382					

Table 9.5 Regression	Results:	Market-based	Indicators
----------------------	----------	--------------	------------

			Dependent	Variable: PCY	ίG			
Independent Variables	Model 1	Model 2	Model 3	Model 4	Model 5	Model 6	Model 7	Model 8
Panel C: TURNOVE	R							
Constant	1.194	1.251	1.353	2.712	2.592	2.514	2.831	2.753
Constant	(3.732)**	(3.455)**	(3.744)**	(4.735)**	(5.002)**	(4.769)**	(5.098)**	(5.160)**
LIPCY	4.600	4.583	4.536	4.443	4.000	3.977	3.908	3.908
LITET	(22.398)**	(21.752)**	(21.611)**	(20.896)**	(21.002)**	(20.646)**	(20.272)**	(20.300)**
POP	-0.35	-0.050	-0.069	-0.045	-0.111	-0.121	-0.141	-0.139
101	(-0.396)	(-0.563)	(-0.781)	(-0.504)	(-1.375)	(-1.475)	(-1.722) ^r	(-1.705) [†]
FDUC	0.030	0.036	0.040	0.028	0.034	0.037	0.039	0.039
LDOC	(0.998)	(1.202)	(1.317)	(0.900)	(1.220)	(1.329)	(1.396)	(1.407)
INVES	0.030	0.028	0.026	0.033	0.037	0.037	0.036	0.035
III VES	(2.913)**	(2.726)**	(2.573)	(3.191)**	(4.035)**	(4.034)**	(3.795)**	(3.767)**
I TURN/GDP		-0.035	-0.088	0.036	0.157	0.127	0.183	0.178
ETORIVODI		(-0.266)	(-0.663)	(0.250)	(1.229)	(1.327)	(1.403)	(1.367)
I TURN/GD x APCI			0.001	0.001	0.001	0.001	0.001	00.1
			(2.630)**	(2.700)**	(2.821)**	(2.865)**	(2.954)**	(2.931)**
FF				-0.027	-0.026	-0.025	-0.030	-0.028
11				(-3.158)**	(-3.296)**	(-3.260)**	(-3.603)**	(-3.601)**
PS					-0.015	-0.019	-0.022	-0.021
15					(-0.564)	(-0.683)	(-0.820)	(-0.767)
OPEN						0.009	0.007	0.007
						(0.809)	(0.657)	(0.681)
INF							-0.003	
							(-0.511)	
R^2	0.616	0.612	0.619	0.629	0.669	0.670	0.672	0.671
Adj. R (%)	0.612	0.607	0.612	0.622	0.662	0.661	0.661	0.662
F-Ratio	152.516**	117.874**	100.574**	87.491**	85.071**	75.614**	67.051**	74.640**
D.W.	1.621	1.604	1.573	1.592	1.417	1.419	1.392	1.554

Table 9.5 Continued: Regression Results: Market-based Indicators

Notice a: Figures between parentheses are heteroscedasticity-consistent t-statistics. * Significant at 0.05 level, ** Correlation is significant at 0.01 level, 'significant at 0.001 level.

Source: Developed for this research from Section 9.4 Data and Summary Statistical Analysis.

9.7 DETERMINANTS OF BANKING AND FINANCIAL MARKET DEVELOPMENT INDICATORS

This section will provide an analysis of the prospect for financial development in emerging market economies. In particular, the investigation model points to the following facts: the stock market is complementary rather than a substitute for the banking sector in providing financial services to the corporate sector in an emerging market, whereas stock market and banking sector development have different effects on a small and a larger economy. Changes in stock market development are likely to bear more significant influences upon a large

Notice b: The sample period is 1995-2006. PCYG is per-capita income growth, LIPCY is the natural logarithm of initial per-capita income, and POP is population growth rate. EDUC is the secondary school enrolment rate. INVES is the value investment as percentage of GDP. LMC/GDP is the natural logarithm of market capitalisation ratio divided by GDP. LMC/GD x Δ PCI interaction with change per-capita income. LVT/GDP is the natural logarithm of value traded divided by GDP, LVT/GD x Δ PCI interaction with change per-capita income. LTURN/GDP is the natural logarithm of turnover ratio is the total of equity transactions divided by market capitalisation. EF is economic freedom. PS is political stability OPEN is a measure of openness which is calculated as the sum of imports and exports divided by GDP. INF is inflation rate. R adjusted is the usual R^2 , adjusted for the degrees of freedom; DW is the Durbin-Watson test for residual serial correlation. *F*-test tests the null hypothesis that all coefficients apart from the intercept are zero.

economy than upon smaller one, while changes in banking sector development are prone to have more influence on a less developed economy than on a more developed one.

Tables 9.6, 9.7 and Figure 9.3 present the banking sector and stock market indicators between four groups of 42 emerging markets for each group including Libya. The findings from Table 9.6 show that the coefficient of banking sector development indicators 1.141 and 0.615 in group three (Libya, Argentina, Brazil, Bulgaria, Chile, Fiji, Lebanon, Malaysia, Mexico, Poland, Romania, Russian, South Africa, Turkey and Venezuela) is larger than the coefficient of 0.321 and 0.010 of group two (Libya, Algeria, China, Egypt, India, Indonesia, Iran, Jordan, Morocco, Philippines, Sudan, Syria, Thailand and Tunisia), but the fit of group two, with an R^2 of 79.2 and 77.1 *per cent*, is superior to that obtained by group three, where banking sector development indicators were proxied by the natural logarithm of bank asset ratio which is the total assets of banking sector and domestic credit ratio to the ratio of bank claims on private sector GDP. The *F*-statistics of group two, F = 132.955 and 121.111, is additionally improved, which rejects the null hypothesis of no explanatory power for the regression as a whole at better than at one-per cent level. This finding is in contradiction to the finding related by Demetriades and Hussein (1996) who argue that in developing countries it is more likely that bank deposit liabilities are rising and credit to the private sector remains unchanged, that is because government increases reserve requirements. According to the Mckinnon/Shaw inside money model, credit to the private sector has a direct impact on the quantity and quality of investment and economic growth (Chapter Four). Sinha and Macri (1999) also examine the causality between economic growth and financial development, measured by M1, M2 and domestic credit, for eight Asian countries. Their results show that there is a two-way causality relationship between growth and financial variables for India and Malaysia, one way causality from financial variables to growth for Japan and Thailand and reverse causality for Korea, Pakistan and the Philippines. The result in Table 9.7 mirrors to a large extent that obtained for bank development indicator measures, with both market development indicators seemingly having a negative and insignificant independent effect on per-capita income growth rate. As results in groups two and three are mainly similar to those included in groups two and three, in Table 9.6 at any rate, given the unbalanced nature of data set and the limited country treatment, the results of stock market development should be viewed as provisional at best.

Stock market development indicators in Table 9.7 have a negative insignificant effect on growth in groups two and three. This result is consistent with Levine and Zervos (1996, 2000) wherein they failed to establish a relationship between stock market capitalisation and growth in per-capita income growth; Rajan and Zingales (1998) failed to identify market capitalisation

as having an effect on the relationship growth of financial development indicators. To illustrate the significance of this result, it would be necessary to consider the "*change in real per-capita income growth*" which indicates how much faster the economic growth is at level percentage. The result in Table 9.7 demonstrates that most groups have a significant effect on the economic growth rate. Test statistics presented in Tables 9.6 and 9.7 indicate that the null hypothesis cannot be rejected stating that there is no correlation between error terms and investment vector and that models are correctly specified.

Independent Variables	Group (1)	Group (2)	Group (3)	Group (4)
Panel A: BAN				
(Constant)	4.759	0.186	2.976	-1.339
(Collstant)	(1.744)*	(0.096)	(1.281)	(-0.377)
LIPCYG	2.401	8.106	4.867	5.376
LFCTO	(8.010)**	(19.675)**	(16.637)**	(11.598)**
INVES	0.021	0.008	0.045	0.048
INVES	(0.804)	(.605)	2.743)**	(1.151)
ΙΒΛΡ	-1.538	0.321	1.141	0.940
LDAK	(-0.974)	(0.292)	(0.869)	(0.455)
ΙΒΔΡ γ ΔΡΟΙ	0.005	0.002	0.001	0.004
	(4.569)**	(3.565)**	(3.691)**	6.692)**
Observation	48	145	142	90
R^2	0.716	0.792	0.726	0.718
Adj. R (%)	0.689	0.786	0.718	0.705
F-Ratio	27.080	132.955	90.949	54.175
Sig. (F-Ratio)	0.000	0.000	0.000	0.000
D-W	1.434	1.809	1.725	1.386
Panel B: DC/GDP				
(Constant)	2.165	-0.491	0.025	-0.833
(Collstant)	(1.187)	(-0.721)	(0.015)	(-0.444)
LECYC	2.510	8.350	4.899	5.522
LICIO	(7.956)**	(19.777)**	(16.540)**	(11.956)**
INVES	0.011	0.006	0.036	0.070
IIIVES	(0.409)	(0.491)	(2.335)*	(1.819)*
I DC/GDP	0.026	0.010	0.615	0.441
LDC/ODI	(0.025)	(0.029)	(0.943)	(0.473)
I DC/GDP x APCI	0.003	0.001	0.001	0.006
	(3.649)**	(1.895)*	(2.517)*	(7.158)**
Observation	48	145	142	90
R^2	0.676	0.777	0.716	0.729
Adj. R (%)	0.646	0.771	0.707	0.716
F-Ratio	22.433	121.111	86.207	57.126
Sig. (F-Ratio)	0.000	0.000	0.000	0.000
D-W	1.660	1.896	1.759	1.523

 Table 9.6 Banking Sector and Growth: Per-capita Real GDP as a Function of Proxy Measures for Banking Development Indicators

Notice a: Figures between parentheses are heteroscedasticity-consistent t-statistics. * Significant at 0.05 level, ** Correlation is significant at 0.01 level.

Notice b: Group (1) Low-income countries: Libya, Bangladesh, Nigeria, Pakistan, Yemen and Zimbabwe. Group (2) Lower-middle-income countries: Libya, Algeria, China, Egypt, India, Indonesia, Iran, Jordan, Morocco, Philippines, Sudan, Syria, Thailand and Tunisia. Group (3) Upper-middle-income countries: Libya, Argentina, Brazil, Bulgaria, Chile, Fiji, Lebanon, Malaysia, Mexico, Poland, Romania, Russia, South Africa, Turkey and Venezuela. Group (4) High-income countries: Libya, Bahrain, Czech Republic, Hungary, Israel, Kuwait, Oman, Qatar, Saudi Arabia and UAE.

Source: Developed for this research from Section 9.4 Data and Summary Statistical Analysis.

As can be noted from Tables 9.6 and 9.7, the regression coefficients are estimated. Findings suggest that countries with common geographic ties (i.e. MENA and Mediterranean Rim countries) may co-integrate with one another. Common economic and geographic ties may lead the national stock markets to follow the same stochastic trend.

Independent Variables	Group (1)	Group (2)	Group (3)	Group (4)
Panel A: MC				
(Constant)	2.447	-0.382	0.953	-0.164
(Collstant)	(6.654)**	(-1.162)	(2.875)**	(-0.171)
LPCYG	2.404	8.395	5.080	5.669
Liefo	(8.184)**	(19.622)**	(17.567)**	(10.426)**
INVES	-0.011	0.003	0.053	.067
	(-0.336)	(0.259)	(3.375)**	(1.388)
LMC/GDP	0.201	-0.197	-0.680	0.484
	(0.307)	(-0.007)	(-3.165)***	(0.879)
LMC/GDP x ΔPCI	(4 504)**	(1.936)*	(0.954)	(5.324)**
Observation	48	145	141	79
	0.719	0.779	0.710	(70)
R^{2}	0.718	0.778	0./19	.070
Adj. R (%)	0.691	0.771	0.711	.653
F-Ratio	27.308	121.524	87.083	37.639
Sig. (F-Ratio)	0.000	0.000	0.000	0.000
D-W	1.589	1.888	1.996	1.476
Panel B: TV/GDP	2,529	0.4(1	1.200	1 105
(Constant)	2.528	-0.401	1.200	1.125
	2.608	(-1.4 <i>37)</i> 8.658	5 306	(1.004)
LPCYG	(7 791)**	(20 434)**	(17 576)**	(11 917)**
	0.028	0.005	0.055	0.062
INVES	(0.844)	(0.381)	(3.445)**	(1.782)*
	-0.300	-0.081	-0.401	-0.276
LV1/GDP	(-1.225)	(-0.742)	(-3.069)**	(-1.564)
LVT/CDD - ADCL	.006	0.840	0.002	0.007
LV I/GDP X ΔPCI	(.818)	(0.015)	(0.328)	(1.091)
Observation	48	145	136	78
R^2	0.592	0.772	0.719	0.683
Adj. R (%)	0.554	0.766	0.710	0.665
F-Ratio	15.623	117.781	83.634	39.290
Sig. (F-Ratio)	0.000	0.000	0.000	0.000
D-W	1.942	1.882	1.999	1.636
Panel C: TURO				
(Constant)	2.754	-0.498	1.181	0.472
	6.400)**	(-1.586)	(3.019)**	(0.491
LPCYG	2.752	8.303	3.387 (17.616)**	3./08 (10.156)**
	0.014	0.006	0.046	0.050
INVES	(.450)	(0.473)	(2.961)**	(1.048)
LTUDNOVED	-0.475	-0.051	-0.416	-0.176
LIUKNOVER	(-1.849)*	(-0.342)	(-1.932)*	(-0.472
LTURNOVR x ΔPCI	0.002	0.012	0.021	0.005
Observation	48	145	141	84
D ²	0.609	0.772	0.710	0.506
	0.008	0.772	0.710	0.590
Adj. R (%)	0.572	0.766	0.701	0.576
F-Katio	16.703	117.743	83.226	29.160
Sig. (F-Katio)	0.000	0.000	0.000	0.000
D-W	16.703	1.889	1.893	1. /61

 Table 9.7 Stock Market and Growth: Per-capita Real GDP as a Function of Proxy Measures for Stock Market Development Indicators

Notice a: Figures between parentheses are heteroscedasticity-consistent t-statistics. * Significant at 0.05 level, ** Correlation is significant at 0.01 level.

Source: Developed for this research from Section 9.4 Data and Summary Statistical Analysis.

CHAPTER 9: AN EMPIRICAL INVESTIGATION OF STOCK MARKET PERFORMANCE

Scatterplots are utilised in Figure 9.3 to display average annual banking and stock market development indicators against real GDP per-capita growth rate during the period 1995-2006 for 42 emerging market economies. This allows a more subjective method of identifying influential observations of the partial relationship between each of the banking and stock market development indicators to identify the outliners that may be influencing the slope and estimated regression line. Thus, this sample rejects the hypothesis of absolute convergence. For each observed value of banking and stock market indicators, a range of predicted values can be seen. If the assumption of equal variances is met, the range should generally be the same for all observed banking and stock market development indicators. For a large observed stock market value, the range of predicted values is much narrower than elsewhere in the plot, because there is a natural limit on banking and stock market development indicators which most of the highly developed nations have reached. The effect is within the group of nations suggesting that a more active stock market is associated with growth ratios for higher significant compared to the presence of less active stock markets.





Source: Analysis of *SPSS* output from statistics data.

Therefore, low-income nations could increase their growth rates by creating an active stock market. Rioja and Valev (2004) argued that the relationship between financial development

and economic growth has received much attention in recent years. They adopted a broad sample of 74 countries over the 1960-1995 period and found that the effect of financial markets on growth is not usually positive but, where it is, its size differs. They suggested that financial development has a strong positive effect on economic growth when it reaches the threshold of the so-called "*middle region*". Under this threshold, called the "*low region*", financial development suggests a negative, zero or positive effect. In the "*high region*" the growth effect declines as financial development increases.

The problem of using a probable sample selection is related to econometric issues. For example, heteroscedasticity might be attributable to regression scale differences as a result of having mixed small and large stock market indicators in the same sample. Only countries with sufficient stock market activities should be included to permit addition in data bases. This study has data on all non-stock market activities for an additional five nations (Algeria, Libya, Sudan, Syria and Yemen), where overt observations on stock market transactions in these emerging markets do not exist. Thus, for the aforementioned nations, values were also entered for market capitalisation, value trading and turnover ratios. Zero is not an extreme assumption. Table 9.2 explains that the mean values for market capitalisation, total value traded and turnover ratios including standard deviations may create statistical problems, where weighted least square regressions are used. These procedures are adopted to improve reliability of this study's results, while a weight of 1 is imposed on all nations where direct stock market data is added. Alternatively, weights of 0, 0.25, 0.50 and 0.75 are to be imposed, including 1 on the countries where a zero is added for stock market development indicators⁶¹. However, it is not evident how much weight they are keen to afford to zero assumptions. Table 9.8 demonstrates results on stock market and banking development indicators in growth regression with different weights. In consideration of these countries and weight results, however, they merely suggest there is an exogenous component of stock market development linked to economic growth. Thus, these results are insufficient, in themselves, to lead us to conclude that the stock market as a leading sector in emerging markets is significant in its process of economic development; this finding indicates the banking sector performs dissimilar functions from those performed by the stock market.

⁶¹ One of the greatest difficulties is in estimating the value of stock market indicators data solving sample selection problem. Some researchers' base their estimate on an assumption of stock market indicators using weighted last squares regressions (see for example, Levine and Zervos, 1996). Minier (2001) used control group to solve that problem as compared with countries that open a stock market to a priori a similar countries that do not. Others suggest using dummy variables equalling 1 to 1 for stock market development indicators for no available data for countries, and 0 otherwise.

Independent Variables	Weig	ht = 1	Weight	t = 0.75	Weight	= 0.50	Weight	= 0.25	Weigh	nt = 0
Stock Market Indicator (SMI)	BANKI	SMI	BANKI	SMI	BANKI	SMI	BANKI	SMI	BANKI	SMI
MC/GDPI	0.018	0.028	0.018	0.021	0.018	0.021	0.018	0.021	0.018	0.027
	(0.532)	(1.168)	(0.525)	(0.858)	(0.530)	(0.876)	(0.535)	(0.906)	(0.540)	(1.117)
VT/GDPI	0.001	0.001	0.001	0.001	0.001	0.001	0.001	0.001	0.001	0.001
	(1.452)	(1.622)	(1.448)	(1.604)	(1.452)	(1.602)	(1.455)	(1.599)	(1.459)	(1.492)
TURNOVERI	0.002	0.001	0.002	0.001	0.002	0.001	0.002	0.001	0.002	0.001
	(0.819)	(0.234)	(0.824)	(0.322)	(0.825)	(0.371)	(0.826)	(0.440)	(0.828)	(0.579)

Table 9.8 Initial Stock Market Development, Banks and Growth, 42 Country Sample(Dependent Variables: Real Per-capita GDP Growth 1995-2006)

Notice a: Weighted Least Squares: Indicated weights are given to countries without direct measures of stock market development indicators.

Notice b: BANKI is measured by the value of banking assets ratio and the value of the ratio of bank credit to private entries divided by GDP. MC/GDPI is market capitalisation ratio measured by market capitalisation divided by GDP. VT/GDP is value traded ratio by GDP. TURNOVER is turnover ratio is the total of equity transactions divided by market capitalisation. BANKI measured by: BAR is the total asset of the banking sector divided by GDP and DC/GDP is domestic credit ratio in the ratio of bank claims on private sector GDP. Besides BANKI and the indicated stock market indicator SMI, the previous regression in Table 9.4 also included as explanatory variable logarithms of initial real per-capital GDP, of initial secondary school enrolment, of initial population growth rate and the logarithm of initial investment ratio. Initial values of economic freedom, political stability, the sum of imports and exports divided by GDP and inflation.

Source: Developed for this research from Section 9.4 Data and Summary Statistical Analysis.

Discussion of Results

This study has attempted to investigate empirically the relationship of independent correlation between stock market development and economic growth in the emerging market, including Libya. Given the data's still incomplete condition, conclusions reached in this chapter based on the simple econometric testing could only be regarded as provisional and as providing preliminary answers to research questions. However, this is the first study to incorporate such a broad collection of indicators to measure a transferable model from a stock market in emerging markets into a small developing country like Libya and to test its correlation with the economic growth process. It applies option estimation and a testing procedure that departs from former studies in this subject. Tables 9.4 and 9.5 illustrate the importance factors of investment proxied by gross capital formulation divided by GDP, emphasised by the strongly positive and statistically significant at the one-per cent level relationship that is exhibited in economic growth. This result is consistent with the findings in Sinha and Macri (1999). A onepercentage point increase in growth rate on investment to GDP is associated with the arrangement between 0.030 to 0.039 percentage point increases in the real per-capita GDP growth. It is consistent with the theoretical view and, too, with empirical evidence that high output growth is associated with high investment rates (see i.e. Fielding, 1997 and Bassanin et al., 2001). As mentioned early in Chapter Four, the suggestion behind Solow's (1956) framework is that higher investment over savings rates leads to more accumulated capital perwork, resulting in an increase in per-capita output, but at decreased rates. Under the endogenous growth theory an emphasis on broader concepts of capital, such as that of Rebelo (1991) as per-capita real GDP growth and investment ratio, tend to move in concert. The highly significant total investment coefficient may reflect the effects of entrenched technological change on aggregate supply working via the new investment.

Whilst this study has empirically established the link between stock market and economic growth of emerging markets including Libya, the entire analysis is based on stock market development indicators and market capitalisation ratio. To ascertain the *robustness* of results, this study tested whether they are particular to the choice of stock market development indicators or whether the various proxies that capture different facets of the level and the degree of the stock market give similar conclusions. The question then resembles *the various other proxies of the market development indicators also correlated with the economic growth rate of emerging market* Model 2-8 of Table 9.5 attempts to answer this question. In these models stock market or market-based development is proxied by total value traded and turnover ratios. Since the DC/GDP ratio appeared insignificantly correlated with economic

growth in models 3 and 4, this indicator was replaced with the ratio of bank assets to GDP as a proxy for banking-based development in all these regressions. As this result showed, the ratio of banks' claims to private sector to GDP ratio is a finer measure of banking sector development and a better predictor of growth than banking asset ratio. As argued by Beck et al. (1999a) and Levine et al. (2000), the ratio of the banks' claims to private sector to GDP has a distinct advantage over the measure of banks' asset ratio, in that it more accurately represents actual volume of funds channelled into the private sector. As is evident from Table 9.5 in each of the cited regressions, the indicator for stock market development is statistically and significantly correlated with economic growth. As can be seen from models 5-7, all liquidity indicators-value-traded and turnover ratios are statistically significant at the one and five-per cent level correlated with economic growth rate. Besides being statistically significant, the estimated coefficient additionally suggests that the relationship between stock market liquidity and growth is economically large. The estimated coefficient on value-traded ratio implies that with an increase of a one-percentage point in growth rate, this variable would increase economic growth by a 0.154 percentage point. The estimated coefficient of turnover ratio further suggests a similarly large economic relationship between stock market liquidity and economic growth. Consequently, an increase of a single-percentage point in the growth rate of turnover ratio would increase economic growth by a 0.183 percentage point.

Regarding the banking sector case, as can be observed from Table 9.4, in each of the previouscited regressions, bank-based development indicators proxied by the ratio of bank assets to GDP and the ratio of bank claims on private sector to GDP enter statistically at the one-per cent level correlated with economic growth rate. Empirically, DeGregorio and Guidotti (1995) attempted to examine the relationship between financial development and economic growth. The ratio of bank credit to GDP is used to measure the degree of financial development during 1960-1985 for 98 countries. The result suggests that economic growth can be positively affected by bank credit. They also found that the impact of financial intermediation on economic growth seems to be higher in high-income countries than in low-income countries. They also found that there is a strong negative correlation between bank credit and growth in twelve Latin American countries and concluded that the negative correlation among those twelve Latin American countries is due to the lack of regulation that to a fragile financial system. Mandelker and Tando (1985) find that money growth has a positive impact on real stock returns in six major industrialised countries. Aspern (1989) also, finds that money supply is positively related to stock prices in ten European countries. This result confirms that both the banking sector and the stock market are significant in the growth process of emerging
CHAPTER 9: AN EMPIRICAL INVESTIGATION OF STOCK MARKET PERFORMANCE

markets. This finding is in contradiction to the finding related by Demetriades and Hussein (1996), who use causality test between financial development (as proxied by bank deposit liabilities to GDP and the ratio of bank claims on the private sector to GDP) and real GDP. They found that finance is playing a limited role in the economic growth process. The study also found that there is two way causality between financial development and economic growth. The study highlighted the dangers of statistical inference based on cross-country studies alone, which implicitly treat different economies as homogeneous entities. A possible explanation could be that this study suffers from poor sample size, as the authors used only 27 observations per country, and they also ignored the significant role of stock market in determining economic growth.

The result further validated the argument that stock market performance of various sets of functions is not provided entirely by the banking-based system. The fit of regression in Table 9.4 with an R^2 greater than 70 *per cent* is relatively satisfactory. The *F*-statistics in each of the cited regressions rejects the null hypothesis that no explanatory power for the regression as a whole is better than *at the one-per cent level*. However, most of the evidence supports the view that the relationship between stock market development and economic growth in the emerging market is bi-directional. Greater development in the stock market causes higher real economic growth. This, in turn, promotes development in the stock market. As income increases, its cyclical components such as the volume of savings should impact stock market development. This reflects that a stock market open to both domestic and foreign investors may be beneficial to economic growth.

Tables 9.6 and 9.7 display the results of groups one to four, where the stock market and banking sector in emerging markets play different, but complementary, roles. A possible explanation of this result can be found in theoretical literature where banks primarily ameliorate information asymmetries, while the stock market mostly enhances liquidity and facilitates risk diversification. Thus, policies undertaken to develop the stock market need not adversely affect the existing banking system. The best practice can be found in groups 2 and 3 in emerging markets under 2SLS test (Figure 9.3), which have the potential to inform improvements to the Libyan stock market. It was observed that the results and conclusions are similar to those obtained for the whole sample. So, similarities between the two groups of emerging countries could be confirmed. Despite the difference of the economic structure of these two groups, the financial systems seem to impact homogenously on economic growth. Therefore, many of the emerging countries issued new capital market laws which aimed at encouraging private investment, increasing investor protection and enhancing the banks' role

CHAPTER 9: AN EMPIRICAL INVESTIGATION OF STOCK MARKET PERFORMANCE

in stimulating capital markets through the establishment of mutual funds. Specifically, their core provisions included establishing a new legal framework to govern specialised capital market companies, strengthening financial disclosure, giving foreign investors full access to the market and increasing investors' rights through provisions prohibiting unfair market practices. However, security markets in the MENA nations are still, in general, underdeveloped, with a limited number of listed companies, low free-float of shares and thin trading (see Section 9.2).

The findings are also consistent with the findings in this study of previous findings and the conclusions of Demirgüc-Kunt and Levine (1996), and Demirgüc-Kunt and Maksimovic (1995, 2000) that stock market and financial intermediary development proceed simultaneously. Consistent with the King and Levine (1993) provide the starting point in the empirical research on the relationship between finance and economic growth. They study four financial indicators (the size of the financial intermediary sector relative to GDP, the importance of banks relative to the central bank, the percentage of credit allocated to private firms out of total credit and the ratio of credit issued to private firms to GDP) and the growth indicators (real per capita GDP growth, the rate of physical capital accumulation, and the ratio of domestic investment to GDP). The empirical results of this paper show that the indicators of financial development are strongly and robustly correlated with economic growth indicators, using cross-country analyses for 77 countries over the period 1960-1989. Finally, Garcia and Liu (1999) empirically explored the determinants of stock market development, particularly market capitalisation. They also examined the association between financial intermediary development and stock market development using a sample of fifteen industrial and developing countries from 1980 to 1995. They concluded that real income level, saving rate, financial intermediary development and stock market liquidity are important predictors of market capitalisation, while macroeconomic stability does not have any explaining power. They confirmed that banks and markets are complement instead of substitutes.

9.8 SUMMARY AND CONCLUSION

Apart from further analysing the findings presented and discussed in Chapters Seven and Eight, this chapter has considered some significant outcomes of the *endogenous growth model* in the light of various contributions reviewed earlier from the literature. This chapter's findings contribute towards a finer understanding of differences in the financial market between the Libyan economy and other emerging markets by providing a simple, theoretical framework that links endogenous growth and current theory on functions of financial markets

and institutions, in order to investigate how the former and banking sector development indicators affect economic growth in these countries. Levine (1997: 688-689) argues: "Although conclusions must be stated hesitantly and with ample qualifications, the preponderance of theoretical reasoning and empirical evidence suggests a positive, first-order relationship between financial development and economic growth...There is even evidence that the level of financial development is a good predictor of future rates of economic growth, capital accumulation, and technological change." Additionally, the study framework may have identified three mechanisms of the research by which financial market development may encourage endogenous economic growth⁶². Firstly, financial markets could encourage a more efficient allocation of new investment-additions through to capital stock from less relatively to more relatively productive uses by financial intermediation between savers and entrepreneurial investors. Secondly, financial markets could encourage an increase in the rate of capital accummulation by providing increased incentives between saving and investment. Thirdly, financial markets could, in addition, improve productivity of capital by providing the necessary function of controlling and monitoring managerial actions. While this study has argued the degree to which the stock market influences real economic growth depends on how effectively it provides liquidity, information and monitoring function, risk sharing and pooling, it has extended a model of the Romer (1986), Lucas (1988), Rebelo (1991) and Pagano (1993) type of endogenous growth economy in order to achieve and incorporate the effect of stock market development.

Analyses indicate that the stock market has played a significant role in emerging markets. The relationship between most stock market development indicators and economic growth have been recognised as *robust*, even after controlling the economic variables associated with growth rate. The stock market size indicator as measured by the market capitalisation ratio enters positively and statistically correlated with real per-capita GDP growth rate. Another two liquidity indicators, value traded and turnover ratios, enter positively and statistically correlated with real per-capita GDP growth rate. It additionally emerged from this work that banking sector development indicators enter significantly in growth regressions. Besides emphasising that financial sector development in an emerging market is significant in its process of economic growth, this finding suggests that the bank-based performs different functions from those performed by the stock market-based. This is consistent with the view

⁶² Singh (1999:347) explained the stock market can promote economic growth through, "1- increasing savings and investments, 2- improving the productivity of investments and 3- raising the profitability of existing capital stock."

CHAPTER 9: AN EMPIRICAL INVESTIGATION OF STOCK MARKET PERFORMANCE

that the stock market offers opportunities primarily for trading risk and boosting liquidity, whereas banks focus mainly on establishing long-term relationships with companies because they seek to acquire information about projects and managers in order to enhance corporate control. The result, in general, shows strong support for the model's theoretical prediction that stock market and banking sector development indicators have played significant and complimentary roles in the growth process.

Overall, this chapter's findings have significant policy implications for the Libyan economy and other emerging markets beyond a similar economic structure. Regarding policy implications, it is important that we draw some policy proposals based on the results of the best practice model found for emerging economies. For one thing, policies that support both banks and stock markets development are more favourable, rather than in favour of either banks or stock markets. Knowing the determinants of banks and stock markets development is an important next step for policy makers since higher financial markets development would facilitate economic growth in all countries, as shown by the extensive empirical literature. The results stress the importance of creating the conditions for an efficient provision of financial services. Efforts should be concentrated to promote well-developed banking sector and stock market as well as in identifying key conditions such as good quality of institutions. Realising several aspects of improving institutional quality for instance, enhancing rule of law and quality of financial regulation infrastructure, securing property right, reducing uncertainty, enhancing investors' confidence and playing a key role in the functioning of financial markets, is the key to both financial development and economic growth. It follows that an improvement of the performance of financial system in the Libyan stock market is crucial in order to stimulate growth through financial development. Evidence indicates that economic growth has a vital role in stock market development. Thus, it is significant for liberalisation and, in order to promote development of stock market indicators, Libya can increase economic growth via appropriate reform policy. To diversify the economy away from central control by government policy and to be successful, existing problems in all these areas need to be solved. In some instances that will require more private ownership, deregulation and reform of financial liberalisation. Key findings are stated in this chapter. Previous chapters will be summarised together with conclusions stating practical implications for reform policy, practice, research contribution, theory and future research in the next chapter.

CHAPTER TEN

CONCLUSIONS, CONTRIBUTION AND IMPLICATIONS

10.1 INTRODUCTION

Essentially, this research proposed to provide a theoretical framework of an economic reform programme, macro-economic reform performance, financial maturity and stock market performance as a whole, as a step towards filling the gap in the theoretical background of researchers. In a bid to test aspects of this framework, and confronted by new developments, an empirical model was conducted to measure the extent of the economic reform programme and stock market performance practised by the Libyan economy. This chapter focuses upon the *summaries* and the findings, and *states* the conclusions of the objectives in terms of their contribution to existing research and their implications for academics and practitioners alike. The chapter is divided into six main sections. Section 10.2 presents the overview of the current research. Section 10.3 deals with key findings that emerge from the theoretical and empirical analysis undertaken. In section 10.4 the contribution to knowledge is addressed. Section 10.5 presents the implication policy and within Section 10.6 the potential limitations affecting the research are addressed. Finally, in Section 10.7, the outline for future research, building on the literature of the economic reform programme and stock market performance and the findings of this research, are suggested.

10.2 OVERVIEW OF THE CURRENT RESEARCH

This research provides considerable knowledge about the relationship between the economic reform programme and stock market performance with special emphasis upon the developing country of Libya. Some of this knowledge might appropriately be generalised for other developing countries that have a similar economic structure. The general aim and specific objectives of this research were to provide a model that could be used to identify critical variables of an economic reform programme and stock market performance and apply this model to the Libyan context. Chapter One introduced the research, defined the problem and considered the aim and objectives of the research. The aim of this research is to determine the most appropriate model for the continued viability of a stock market development in Libya and to consider an appropriate strategy for the Central Bank of Libya to undertake the successful continuation of the stock market mechanism. To achieve this, the research objectives were formulated: firstly, to identify the success of the economic reform programme

in Libya, specifically the deregulation, corporatisation, privatisation and liberalisation that have led to the initiation of the Libyan stock exchange; secondly, to examine the relationship between macro-economic reform variables and stock market performance variables; thirdly, to assess the performance of market-based economies, with particular reference to the emerging economy of Libya, and to evaluate current and best practice in financial deregulation; finally, to determine whether best practice from other emerging stock markets is transferable to the Libyan situation and context. Theoretical, methodological and empirical studies were undertaken to provide the conceptual framework of the research model in order to achieve the research objectives. In addition, Chapters Three, Four and Five represent the theoretical part of the study, while Chapter Six discusses the methodological part. Chapters Seven, Eight and Nine represent the empirical data analysis, which is one of the major contributions of this study. Chapter Ten discusses the research summary, conclusions, contribution, implications and recommendations. The theoretical, methodological and empirical parts of this research are discussed subsequently.

10.2.1 The Theoretical Part of the Study

The primary theoretical part of the study was to develop a model framework that could be used to identify the critical variables in developing countries, in Libya in particular. Chapter Two offers a comprehensive analysis of the macro environment of Libya and the performance and evolution of commencing financial sector development within its economy. Following the 1969 revolution, the economy witnessed three major systems: the nationalism and socialism period, an open door policy period and, latterly, the economic reform programme. The economy experienced rapid expansion during the 1970s and early 1980s as real GDP grew by more than 10 per cent on average. This expansion was mainly financed by oil sector revenue. In the mid-1980s economic growth slowed and the Libyan government started to experience reversionary trends. These later trends resulted from the collapse of oil prices and the Gulf War in 1990/1991. The Libyan economic reform programme was inaugurated during 1999 stemming from many factors such as: economic crisis in terms of oil price collapse by the mid-1980s and the early 1990s; the country's political isolation resulting from international UN sanctions following US Libya-specific trade sanctions in the early 1990s. By the end of the twentieth-century, after three decades or so of its leader's self-proclaimed revolution, Libya found itself at a significant connection. The country was able to emerge to enjoy its diplomatic reform and stave off its long economic crossroads. In addition, after economic sanctions were suspended in 1999, European and other foreign investors slowly returned to Libya as a central arena among North African nations.

The theoretical underpinning of this study is based on literature divided into three main chapters. Chapter Three presents a review of the relevant literature on economic reform programme. In Chapter Four theoretical aspects of economic growth theory are discussed. Emerging stock markets are illustrated in Chapter Five. Chapter Three offers an overview of how the economic reform programme has become a global phenomenon adopted by many developing countries in a transition economy in the last two decades such as Chile, China, Czech Republic, Egypt, Estonia, Hungary, Lithuania, Poland, Romania, Slovak Republic, Ukraine, Venezuela and Vietnam. Various classifications are used to define the economic reform programme, such as transition, transformation, economic adjustment, stabilisation and regime. In this research the expression "economic reform programme" and "transition" were considered in relation to the Libyan economy. It was argued that the key target of any economic reform programme represents the rate of economic growth and increase per-capita income in order to enhance the welfare of the economic society as a whole. However, the method which should be followed to implement such programmes was not determined and a comparison was made between the costs and benefits of a direct impact, which was the deciding factor in choosing between the two through experiments and learning from other regions which have implemented an economic reform programme. It could be argued that the South and East Asian regions with similar economies were the most successful when compared with other regions, with higher national savings, more investment and exports as the leading successes of these areas. In this context, Libya, with respect to IMF and World Bank advice, chose the gradualism method and to follow a gradual implementation of its programme in order to avoid any negative effects for its society.

Chapter Four provided the theoretical literature reviewing relevant good practice in order to identify a framework that links together the economic growth theory and the current theory on the function of the financial market and institutions was reviewed explicating developments made since Solow's work (1956). A link between the financial market and an endogenous growth economic theory model is made by some authors such as Romer (1986), Lucase (1988) and Rebelo (1991), among others, in order to include the effect of stock market development. Bencivenga and Smith (1991) and King and Levine (1993) considered additional theoretical support for relationships between financial markets and the rate of economic growth rate. King and Levine (1993) and Levine and Zervos (1995) believed that capital accumulation is an important conduit. Other studies, however, by Rajan and Zingales (1996) and Jayaratne and Straahan (1996) did not support this view. Deidda (2006) argued that the relationship between

financial development and economic growth is generally weak or has insignificant correlation at low levels of per-capita income. This stems from the result of the cross-section analyses by Deidda and Fattouh (2002) and Harris (1997) or the negative outcome resulting from the studies of Xu (2000). The birth of the *new endogenous growth theory* has facilitated the development of improved growth models where the long-term rate could be affected by a number of elements. These included technology, education and health policies in the process of economic development, capital accumulation, government policies and institutional activities in the role of financial development in economic growth. During the development of the *endogenous growth theory*, information was gained from literature and research how the reform of economics on long-term growth rate related to the macro-economic and financial institution performance. This framework has been reviewed by Stiglitz et al. (2001), Dabrowski et al. (2001) and Currie (2003) besides resources from the World Bank and IMF. Other works have revised financial institutions: Levine (1997) and Tsuru (2000).

Chapter Five has laid out evidence from theoretical and empirical studies, of how the stock market has played a significant role within both the advanced economy and the emerging market. Most empirical studies deal with emerging stock markets in diverse regions of the world. The South and Eastern-Asia countries, for instance, could be considered as foremost emerging stock markets compared with other emerging markets, such as the Sub-Saharan African countries, which can be regarded as occupying the last position within that market, and whether these markets can provide potential investors with the opportunity to invest. Furthermore, Chapter Five deals with various aspects of empirical studies which examine the relationship between economic and stock market performance indicators. This empirical study proposed that an economic reform programme, in terms of deregulation, privatisation and liberalisation, has a positive significant effect upon stock market performance which increases market size, market liquidity, market activity and market concentration. Additionally, empirical studies illustrated that there is a positive relationship between macro-economics in connection with per-capita income and GDP growth and stock market performance with impact of changes in exchange rates on stock prices and returns which differ from one economy to another and from one industrial economy to another. However, the empirical studies revealed that there is a negative relationship between other macro-economics vis a vis interest rates, inflation rates and budget deficit on stock market performances related to stock prices and returns.

10.2.2 The Methodological and Methodical Part of the Study

The main methodological and methodical part of the study, Chapter Six, discusses the selected research methodology. This is briefly justified along with the methods of data collection and analysis applied when undertaking this research. In order to achieve the research aim, clear objectives were formulated and, in light of these, the main methodological approach has been detailed. In doing this, various methodologies available for researchers were discussed and justifications for employing the positivistic paradigm were also given. Since economic reform and stock market performance literature has enabled the researcher to define a theoretical framework and develop hypotheses, the deductive approach was selected. The study utilised secondary and primary data and combined quantitative and qualitative methods. The multimethods approach was used to achieve different research purposes, and to enable triangulation to occur. The questionnaires were designed and distributed to the entire target population within the Libyan stock market, and other financial sector organisations, while the researcher attempted to explore the nature and origin of respondents' viewpoints, in semi-structured faceto-face interviews including the sample, the sample population, the sampling frame, the sampling size and the translation of the questionnaire. Finally, the chapter concludes with an explanation of the statistical methods framework model adopted for this research. A variety of descriptive statistical analyses, differences, correlation statistics and regression analysis techniques were applied after due consideration of the assumptions of each.

10.2.3 The Empirical Part of the Study

The aim of the empirical part of the study is to provide a better understanding of relationships between economic reform programmes, macro-economic reform performance, financial maturity and stock market performances. Self-administered questionnaire and semi-structured interviews were used as the key method of data collection. Data obtained from the self-administered questionnaires is used in describing and exploring the correlation between the economic reform programme and the stock market performance. Data derived from semi-structured interviews is utilised to explore and explain the themes that have emerged from using self-administered questionnaires. The statistical package for social sciences (*SPSS 2003*) was used in this research. Spearman's rank correlation test was employed to test general research objectives and hypotheses. The direction of the correlation coefficient (r) can be examined to establish whether it is positive or negative. A positive correlation indicates that there is a positive effect of economic reform programme and stock market performance. If the result of the correlation coefficient r is zero, then this indicates that there is no relationship

between variables. The linear regression analyses model examined were used to test the relationship between variable models in order to ascertain whether they are relevant for explaining and identifying the economic reform programme and stock market performance in the Libyan context. Consequently, there is a need to conduct stock market research by using different types of data sources. This is because regression analysis of the final Libyan data only answers some of the objectives of this research and is also due to the problem of unavailability of *hard* data collection. The transferable model was tested related to the relationship between stock market development, and economic growth was tested by using various specifications of an empirical model extended from a simple endogenous growth theory, where the stock market is assumed to effect economic growth via increasing capital accumulation mobilising capital to its efficient use and improving total factor productivity growth in an economy. To control the endogenous determination of stock market development (a subset of the stock market development may not be exogenous) all regressions were run using the two stage least-squares technique.

Chapter Seven examined the critical factors influencing the economic reform programme, macro-economic reform, financial maturity and stock market performance in the Libyan economy and financial sector according to their characteristics. The findings in this chapter have a significant policy reform implication for Libya and other developing countries with similar economic structure and growth. The evidence indicates that economic reform programmes, macro-economic reform and financial maturity play a significant role in the stock market performance development of the Libyan economy. Thus, it is significant to liberalise the economy, mobilising savings and increasing the ratio of investment to GDP when considering the economic reform programme in order to promote better development of the stock market performance. Libya, however, can encourage the economic growth rate by means of an appropriate policy reform programme. The findings suggest that stock market size, as measured by the value of the listed share divided by the GDP, is considered the significant indicating factor. The total value of traded ratio measures the organised trading of a firm's equity as a share of national output and, therefore, is considered as the most significant positive factor related to the liquidity market. To this end, the increased numbers of traded companies increases market activity in the stock market, followed by the market concentration that concerns respondents, where the percentage of the largest companies affects value traded in the stock market.

Chapter Eight dealt with the empirical work of the regression analysis model of this research, preliminary with the dataset, research hypotheses and methods. Findings related to the

research's objectives have been *presented* and *analysed*. The subsequent chapter presents the results from the regression *analysis* technique *models* for developing the Libyan stock market, *empirically* examining the research *hypothesis* using data collected by the questionnaire survey that may not be tested by utilising financial statements' data. The *hypotheses* examine the independent variables of economic reform programme, macro-economic reform and financial maturity in terms of dependent variables of stock market performance. The research framework model is presented in Section 6.6 in Chapter Six and tested in Chapter Seven. The measurement regression analysis model was conducted with the *SPSS 2003* software package. Data was screened to check for data input errors, distribution, normality, multicollinearity and outliers. Findings relating to the significant or non-significant relationship between the independent and dependent variables under study, with acceptance or rejection of each hypothesis, were discussed and summarised in Tables 8.10, 8.21 and 8.29.

Chapter Nine attempted to provide a better understanding of the transferable model of a financial market between Libya and other emerging markets. The evidence presented in this chapter challenges some economists' belief that stock market development is not significant in the growth process of nations, especially for developing countries, since other financial intermediaries can provide a relationship between stock market development and economic growth. This has a significant implication; it provides a high degree of confidence that stock market development is an effective policy for promoting the emerging market economic growth. This chapter's findings have significant policy implications for the Libyan economy and other emerging markets beyond a similar economic structure. The evidence indicates that economic growth plays an important role in stock market development. Thus, it is significant for liberalisation and, in order to promote the development of the stock market indicators, Libya can engorge economic growth by means of the appropriate reform policy. In an attempt to diversify the economy away from central control by government policy and to be successful, the existing problems in all these areas should be solved. In some cases this will require more private ownership, deregulation and reform of financial liberalisation.

10.3 THE KEY FINDINGS

Based on the literature review and theoretical and empirical analysis, several key findings emerged. These are summarised and discussed throughout the research. However, to capture the major results might be of interest for both economic reform programme and stock market performance in developing countries that have a similar economic structure. The next four subsections of this chapter present a summary of the main findings.

10.3.1 The Relationship between Economic Reform Programme and Stock Market Performance

The Results of Descriptive Statistics

The questionnaires have several implications for economic reform and developing stock market performance. The economic reform programme was measured by deregulation, corporatisation, privatisation and liberalisation stability. Libyan respondents indicated that the deregulation of the economy was the most influential factor affecting the economic reform programme which encourages companies to invest in the stock market. A manager of corporate companies playing a crucial role in encouraging prospective customers to invest in the market-based economy is considered to be a significant source of corporatisation. The private sector can co-operate with stock market authorities to increase competitive advantages of the most important factors in a privatisation reform programme. The private sector, as has been previously mentioned, consists of the service, industrial and agricultural sectors. The contribution of the companies within the service and industrial sector to GDP is 43.5 per cent and their contribution to the labour force is 37 per cent. Liberalisation, consequently, may affect specific components of capital flows including debt (financial and non-financial borrowing, and lending), equity (portfolio), investment (stock markets) and foreign direct investment. Liberalisation of the economy ensued with Libya joining the WTO membership during 2001 and via free trade agreements, which helped the country's competitive economic development to encourage foreign investment and provide consumers with superior quality goods and lower prices across a wide range of products.

An economic reform programme is, nevertheless, conducted along with other recommended reform. Therefore, financial policy fulfilling legal requirements regarding the capital is considered as the most critical factor indicated by respondents. Firms' decisions regarding the issue of shares were the most significant to maintain an increase in market share to satisfy the prospective customer. Adding strength to the company's reputation is considered as an unimportant aspect of a firm's decisions even among private companies.

The Results of Hypotheses Testing

The most significant part in this subsection is the modelling analysis hypotheses of the impact of economic reform programme variables upon stock market performance variables by using multiple regression analysis models. To examine the relationship between the economic reform programme and stock market performance, the theoretical work (as mentioned earlier in the literature review) of Fisher and Gelp (1991), Clague and Rausser (1993), Williamson (1994, 2004), Bgcolor (1996), Fisher et al. (1996) and King (2003) was followed by the empirical study Cho (1986), Zingales (1998), Demirgüc-Kunt et al. (2004) Baltagi et al. (2006), among others, to observe four economic reform programme indicators: deregulation, corporatisation, privatisation and liberalisation. For the economic reform programme empirical findings, however, the result supported the hypothesis that there is a significant positive relationship between economic reform programme variables which cannot reject the hypotheses. It is useful to mention that this result tends to be consistent with stock market behaviour in the Libyan economy after introducing the economic reform programme, as indicated in the literature review. Thus, the impact of each independent economic reform programme variable upon stock market performance variables tends to follow the same direction found in the previous literature review.

10.3.2 The Relationship between Macro-economic Reform and Stock Market Performance

The Results of Descriptive Statistics

In the context of the current research, Libyan macro-economic reform in general, and the economic reform programme in particular, have been designed to increase investment and to strengthen the economy. Macro-economic reform was measured by interest rates, exchange rates, inflation rates, budget deficit, per-capita income and real GDP growth rate in the Libyan economy. The responses further indicated that interest rates affect the stock market, followed by liberalisation of the interest rate that the exchange rate may alter investment in the local currency market compared with other markets, is considered to be a significant factor of macro-economic reform variables, as any increase in inflation rate reduces the expected real returns to investors from holding stocks, followed by when inflation increases the price of goods and services increase. The responses additionally indicate that budget deficit is considered to be the excess of government expenditure over time, as the respondents indicated that per-capita income is one of the driving forces towards a stock market, followed by the impact of oil prices upon GDP and the stock market.

The Results of Hypotheses Testing

This part featured the modelling of the macro-economic reform variables upon stock market performance variables by using the regression analysis model. Testing the relationship between the macro-economic reform and stock market performance followed work by Ajie and Jovanovic (1993), Boundoukh and Richardson (1993), Levine and Zervos (1995), Harris (1997), Fama and French (1989), Levine and Zervos (1996), Levine (1997), Beck et al. (1999a), Levine et al. (2000), Kasman (2003), Omran (2004), and Bolbol and Omran (2005). The overall results indicated that the impact of each independent macro-economic reform variable upon stock market performance variables tends to follow the same direction found in the empirical study in line with the literature review, except for interest rates and inflation rate hypotheses. The interest rates and inflation rate, however, had significant negative relationship with stock market performance, which cannot accept the hypotheses. Also, the results illustrate exchange rate and budget deficit which stated that stock market performance increases as both exchange rate stability increase and budget deficit decrease can either accept or reject the hypotheses, as the exchange rate and budget deficit affected the majority of the stock market performance area. Ultimately, the results supported the hypotheses which stated that the Libyan government should pay greater attention to developing its stock market with success in implementing its macro-economic reform performance as the result does not lend strong support to the hypotheses that there is a fundamental relationship between macro-economic reform and stock market performance variables.

10.3.3 The Relationship between Financial Maturity and Stock Market Performance

The Results of Descriptive Statistics

Three ratios were used to measure financial maturity: banking assets ratio, the ratio of domestic credit to GDP and the ratio of market capitalisation to GDP. Analyses of the results of the critical variables of financial maturity were the ratio of banking assets to GDP that concern respondents as the most critical factors. Domestic credit of GDP was additionally considered as one of the measurements of financial maturity of stock market systems, followed by stock market enlargement as measured by market capitalisation relative to GDP. Responses also highlight the most important factors of a bank-based system or market-based system to be better growth for the long-term. In a bank-based system, mobilising savings, regulatory structure, justifying the problems associated with excessively powerful banks and structure activity are held to be the most important factors within a market-based system

identifying sound investment, allocating capital, increasing transparency and accountability and finance activity as being paramount considerations.

The Results of Hypotheses Testing

In this part of testing the hypotheses model, use was made of two indicators of the banking sector in terms of financial maturity variables that have been used recently following the work by Beck et al. (1999a) and Levine et al. (2000). This involves the rate of total bank assets to GDP, which provides evidence of the significant services performance by the banking sector related to the size of economy, whilst the second indicator used the value of credit given by the banking sector to the ratio of domestic credit divided by GDP. The third indicator used ratio of market capitalisation divided by GDP as measured by the stock market performance development, following studies by King and Levine (1993), Levine and Zervos (1996), Levine (1997), Beck et al. (1999a) and Levine et al. (2000). It was argued that the stock market is no better than banks in motivating an economic growth rate. Both banks and the stock markets work together to influence economic growth. The results support findings by Levine and Zervos (1996), Levine (2004) and the argument by Kunt (1994) and Samuel (1996) amongst others, that there is a strong relationship between the stock market and the rate of economic growth. Indication of the size of the banking system using the total bank sector assets divided by GDP ratio, with another indicator of domestic credit from the banking sector divided by GDP ratio, found that the stock market continues to remain positively and significantly correlated with total bank assets and stock market capitalisation ratio divided by GDP, which cannot reject the hypotheses. Nonetheless, the result does not lend strong support to the hypothesis that there is a crucial link between the stock market and domestic credit divided by GDP ratio. This result has a significant suggestion. It confirms that stock market performance represents a different set of functions not entirely provided by banks. The empirical finding bears a significant policy implication; it argues that the Libyan government should not necessarily abandon the development of one component of the financial sector. Despite the stock market performance and the banking sector playing complementary roles in promoting economic growth rate, the sector should be developed in tandem.

10.3.4 Comparison between Emerging Countries and the Libyan Stock Market

The Results of Descriptive Statistics

Respondents indicated that the most significant lessons learned emanated from other MENA nations followed by those of Central and East European countries, Far Eastern countries and

Latin American nations. The Libyan stock market establishment is very recent in comparison with those of most countries. However, the Libyan stock market should have been able to examine similar experiments in other countries, for instance by assessing the strength and weakness of stock markets in MENA countries.

The Results of Hypotheses Testing

The analysis in Chapter Eight offered comprehensive evidence on the strength of the relation between the stock market development and economic growth in emerging market countries. The findings from the estimation of basic regression, which consisted of per-capita real GDP growth as dependent variable and the labour, capital and stock market development as independent variables, showed that stock market capitalisation is a statistically significant correlated with per-capita real GDP growth. A one-percentage point increase in the growth rate of market capitalisation to GDP is associated with a 0.420 percentage point increase in per-capita real GDP growth. This result is highly consistent with the theoretical view that the ability of the stock market to mobilise capital and risk sharing is intimately associated with economic growth. This appreciated result, however, is inconsistent with previous studies which failed to establish a relationship between stock market capitalisation and economic growth. It was also found that, when the banking sector development indicator was added to the basic model, measured as the total assets of the banking sector divided by GDP and the regression was re-run, then the stock market capitalisation coefficient remained positively and statistically correlated with per-capita real GDP growth. The banking sector development variable, on the other hand, inters positively but insignificantly. The findings merely suggest that there is an exogenous component of stock market development that positively influences per-capita real GDP growth.

Overall, empirical results supported the hypotheses in these findings and, hence, confirmed that the significance of stock market performance in the development process of the Libya has been affected. These findings might appropriately be generalised to other developing or emerging economies that have similar economic and financial structure. Thus, when applied to the important role to developing countries that have similar economic structure to the Libyan economy, it could be beneficial for their financial market to liberalise and mobilise a higher amount of national savings. Thus, it was also confirmed that improving and developing investment raises the efficiency of investment, increases output and, consequently, develops the economic growth rate.

10.4 CONTRIBUTION TO KNOWLEDGE

This study seeks to make an original contribution to knowledge by investigating the impact of an economic reform programme on the stock market performance in an emerging economy. These contributions have combined to construct an applied study of a type that does not appear elsewhere, and the framework offered is, therefore, not only appropriate to Libya as a case study, but also to other countries in similar circumstances such as Algeria, Bahrain, Brazil, Chile, Czech republic, Egypt, Iran, Kuwait, Mexico, Qatar, Saudi Arabia, Syrian, United Arab Emirates, Venezuela and many other countries. The research provides an important introduction to this area and has attempted to explore its significance for the economy and business. A critical review of this research has added to the existing body of literature and assists the researcher in obtaining new ideas and perspectives, exploring the significant variables regarding development and in the application of a series of research models of economic reform and stock market performance related to economic growth in the following areas.

10.4.1 Theoretically

- 1. This is one of the first studies of its type in this critical area. It has explored many new ideas by targeting a large population, undertaking a statistically detailed data analysis and interpreting and justifying the study results and evidence which could be considered as directions for future research.
- 2. The literature was discussed, summarised, categorised and analysed using a methodical and comprehensive technique; the results provided are in a database which includes not only details of the articles such as title, author, and date, but also abstracts of published academic articles in the research. It is hoped that this resource will, additionally, be made widely available by publication, probably as some kind of web-based resource, similar to that of the University of Connecticut economics database, http://ideas.repec.org/f/pma467. html, yet be more extensive, comprehensive and up-to-date. The contribution of this piece of the study extends beyond the descriptive by measuring aspects of the literature and offering suggestions for improving the quality of empirical work investigated and/or undertaken in the area of economic reform and stock market performance.
- 3. Libya is still in the incipient stages of its financial liberalisation and reform, but it is moving quickly as a result of the removal of UN and US sanctions within the last two years, and there are signs of rapid development. However, hitherto, there has been no

study to explore the readiness of the Libyan financial market for stock market performance, although this would seem to be an inevitable outcome.

- 4. The Libyan stock market, in the long-term, will benefit from increasing economic growth. It will determine the impact of the economic reform programme and whether financial theories, concepts and practices have been influenced by environmental factors (cultural, economic, political, and social), as in many other MENA countries.
- 5. It is hoped that such contributions will be beneficial, both academically and professionally. Academically, this research aims to focus attention upon a neglected area in the context of this research. Professionally, managers will further seek out the practical implications offered by this research in their actual relationships with their stock market performance.
- 6. The study contributes to the existing limited studies on economic reform programmes and stock market performance conducted on the developing nation of Libya and within the emerging economies in general.
- 7. The review of literature in this research has revealed gaps and provides a case for more empirical studies to be conducted, where future research integrates, refines, extends the empirical work conducted in this study and fills some of the gaps in this research's literature.
- 8. This study could be considered as a step towards raising and improving the comprehension of current practices and management of economic reform programmes and stock market performance in the Libyan context and fill any gaps of the literature in other developing countries.

10.4.2 Empirically

- Few empirical studies have been conducted that contribute to the understanding of the relationship between economic reform programmes and stock market performance of specific developing countries in both the quantitative and qualitative approach to explore their relationship, where previous research efforts have not been made to investigate such relationships.
- 2. This study was the first of its kind to explicitly investigate the direction of causality (developed model chosen) between Libya and other emerging economies and, hopefully,

will prompt others into replicating the study in other contexts, that is to say, applying the model to other emerging economies.

- 3. Stock market performance was tested as a dependent variable where other variables, in terms of economic reform programmes, macro-economic reform and financial maturity were tested as independent. Stock markets may develop in the anticipation of the financial needs of banking and industries or companies. Factors, such as quality of the deregulation system, quality of goods and services, transparency and internal information problems, modern technologies, taxation issues, capital investment, accounting standards, financial disclosure, liberalisation of restrictions on international flows, macro-economic environments, direct foreign investment, saving investment, the development of both banking sector and non-banking financial institutions (i.e. insurance companies, pension funds, mutual funds and investment banks), are significant considerations to be eliminated for the development of stock market performance.
- 4. This study provides recommended solutions for stock market problems through suggested effective economic reform programmes to enhance market-based financial systems where the influence of Arab culture on a stock market would be minimised or improved.

10.4.3 Methodologically

- This study is the first of its kind to offer a critical research by providing analysis of data that will evaluate the success of the economic reform programme and Libya's readiness to complete its transition to a market-based economy. As a consequence, the Libyan experience could provide new insights into the debate of the effectiveness of stock market performance and the underlying conditions for its success.
- 2. Libya has some similar characteristics to other developing countries that have also established their own stock exchanges by the end of 2000 (UAE: Abu Dhabi Stock Exchange and Dubai Financial Market). A number of other nations only established stock markets very recently, for instance during 2005-2006 (Comoros, Djibouti, Mauritania, Somalia, Syria and Yemen) and this study will make two contributions in this respect. Firstly, the methodology used in this thesis (statistical measurement of economic variables and qualitative analysis of the progress of other like countries) can offer a method for these countries to follow. Secondly, the questionnaire developed as one of the research methods will be available for researchers to adopt with the need to make only slight variations to reflect national situations.

- 3. This research provided a combination of various methods of data collection and analysis testing in the financial and banking sectors of the Libyan economy which was considered to be richer in theoretical models. The richness of the data collected reflected the advantage of adopting the case study method in conducting the research.
- 4. Due to the lack of high-quality published databases, this research has embarked on a significant development of the application of data collection (questionnaire survey and semi-restructured interviews) of the economic reform programme and stock market performance in the developing country of Libya. The critical review and findings can, accordingly, be of great assistance to researchers and policy-makers alike.

Ultimately, this study contributed to the limited studies on economic reform programmes and stock market performance in developing countries in general. In either case, this study contributes to the significant lessons learnt. Experience is related to the doctoral research fulfilling the important aspect of making informed choices and decisions, in that alternative possible courses of action undertaken, investigated and evaluated comprehensiveness, focusing on the main issues addressed by research and looking to fulfil them. This is what finally delivers the research, before committing to a particular path, method, or technique. This is because any decision-making might influence the subsequent course of events and, possibly, impinge on the final outcome. Hopefully, the outcomes emerging from any study are necessarily shaped and influenced by the nature of the study's design.

10.5 THE IMPLICATION OF THE STUDY FOR POLICY REFORM

Based on the information obtained in the existing literature and the empirical findings, as the situation is developing in Libya, the following significant implications can be made to the Libyan financial and business policy reforms' context. This could be transformed, so that Libyans are able to be more open about sensitive issues and constraints which may help introduce and manage strategic shift towards genuine economic reform and stock market performance. In this respect, there still remain outstanding critical issues that address a challenge which should encourage the activities of the stock market performance in the Libyan economy and social wellbeing. Therefore, without careful planning and management, problems will occur and they will be serious. Overall, the Libyan government should aim to achieve the adoption of appropriate plans and strategies which are required for the development of this area.

- 1. Some Arab nations (Bahrain, Kuwait, Qatar, and Saudi Arabia, United Arab Emirates) have huge oil wealth and a small population, whilst others (Egypt, Morocco, Sudan, Syrian and Tunisia) are poor with an enormous population. Libya lies between these extremes and, as such, "*must create*" its own model. Given its available resources, it should increase its wealth still further and "*catch up*" with the income levels of the most developing nations in the world (refer to 3.6.5).
- 2. In order to ensure long-term development of stock market performance and to solve the current problems, it is essential to have management planning that explores the positive and negative effects of financial institutions, including the companies, on the environment of economic reform and deregulation policy (refer to K.4.2 and K.3.4.2).
- 3. Promoting independence and transparency of the Libyan government should create trust in the credibility of the country's deregulation. These will remain the key issues in influencing investors' perceptions of transfer risk (refer to 7.4.1.1).
- 4. It was evident from analysis that stock market development influences policy framework, thus encouraging Libya's economic growth rate. By implanting the measurement of stock market performance, the Libyan government may be successful in establishing a strategy that enhances the role of stock market development in the process of the rate of economic growth but which should promote the activities of stock market performance in a globally competitive arena (refer to K.3.2.1).
- 5. There is reasonable confidence that a politically stable system is making progress on social indicators in the region which can attract foreign capital inflows to realise greater comprehension nationally. This could create the appropriate environment for economic activities and various social groups in Libya to act out their full roles (refer to K.3.4.3).
- 6. Increased awareness and knowledge of the successful implementation of privatisation and liberalisation programmes and putting key deregulation governing taxation in place in order to produce significant direct and indirect benefits for stock market development should be effected. This would engender a significant alteration in the government's mode and control towards ownership of national assets (refer to 7.4.1.2 and 7.4.1.3).
- 7. The findings of this study demonstrate that the stock market will perform a significant role in the long-term, relating to Libyan economic growth by increasing the rate of GDP, promoting the national economic reform programmes, providing that modern

technologies are introduced to develop the production and support other reform programmes to proceed faster (refer to 7.4.2.6 and K.4.1). This may provide some explanation for the expressed intent of Saif al-Islam, the son of President Al-Qaddafi that:

"the old times are finished and Libya is ready to move onto a new stage of modernisation [...] which will be conducted in a well organised manner that ensures new ownership and ownership by the people of Libya, not just a small class of oligarchs like Russia and Egypt ..." Vandewalle (2006: 189).

- 8. The findings of this study also show that the stock market will play a significant role in the long-term relating to Libyan financial institutions and companies. Thus, it might provide the opportunity to acquire finance from investors and provide potential investors with new investment opportunities to increase their portfolio in order to restore desired levels of liquidity (refer to 7.4.4.2).
- 9. The Libyan stock market is expected to implement changes: financial market reform; raising individual income levels; controlling circulation of financial market transitions; new foreign investors to invest in securities; investment policy reform; building on previous measures such as passing of new deregulation pertaining to the dealing of securities; providing the realistic competition between national ownership companies; improving the level of financial disclosure within companies (refer to 9.7 and K.4.2).
- 10. Libya's macro-economy provides a positive context for reform efforts, significantly better than prior to the early 1990s, when Libya required liberalisation of the economy. The latter should create much-needed new job opportunities in the private sector and result in a reduction in unemployment rates. To achieve this, Libya needs to build a physical infrastructure of ports and roads; a telecommunication networks and administrative infrastructure are also required to improve the country's position are greater as knowledge and skills levels to prepare its indigenous workforce to work in the financial markets and to alter the prevailing approach (refer to 7.4.1.4 and 7.4.2.5).
- 11. Libya needs to develop and promote the establishment of specialised financial institution intermediaries such as pension funds, mutual funds, credit unions, savings and loan associations and insurance companies. These will all play a very significant role in increasing depth and liquidity in the stock market and market deregulation, especially *vis a vis* protection of minority shareholders' rights, besides protection of insider information. The development of specialised financial institutions can additionally extend

the development of shares and the level of professional fund management in order to increase market capitalisation and the value traded both relative to GDP ratio (refer to 7.4.3.2 and 7.4.3.3).

- 12. Libya's banking system has a significant role to play in the development of the stock market. This derives from its role as an intermediary between the business sector and investors. Investment banks are most needed, which might be developed either from within or through the existing commercial banking system to provide brokerage services, financial analysis, fund management, portfolio, trading and underwriting of share issues at their own risk. A similar consideration should also be given to the role of banks as market makers, in order to adopt positions and create markets by buying and selling shares (refer to 2.4.1 and K.3.2.3).
- 13. Restructuring monetary policy should include a new role for the Central Bank of Libya and its functions in order to reduce inflation rates and restore the credibility and reliability of Libyan economic growth rates. The proper application of these policy mechanisms and methods consistent with the privacy of the Libyan economy will render it possible to achieve the goals of economic stability and economic growth, rationalise public expenditure and increase efficiency and productivity of non-oil economic sectors (refer to 2.4.2 and 7.4.2.1).
- 14. The corporation of small and medium enterprises within the privatisation programme in Libya is designed to promote small businesses, and young people in particular, by offering them public funding and government support. Improvement of the economic environment that supports the work description and performance measurement of the stock market should ensue. Listing of large privatised companies leads to a significant increase in market capitalisation; it additionally provides a substantial influence on trading liquidity in stock market performance. Listing new companies due to privatisation sales reduces the non-systematic risk of equity portfolio and increases investment opportunities for local investors to increase their portfolios (refer to 7.4.1.2).
- 15. Organising and developing Libyan cooperation with the outside world requires more diligent effort. Motivation of companies to improve the investment climate, organise tax-offices to impose and collect taxes and rates and enhance access to the issues of corporate government, are areas of advice recommended to Libya by the investment programme of the Organisation for Economic Cooperation and Development (OECD) (refer to K.3.4.2).

- 16. Corruption remains in various forms, whether political, administrative, legal or economic. This is one of the most serious problems facing the development of economic, political and social arenas in Libya, as indeed elsewhere. Countenance remains one of the most significant functions of the government, so as not to bring a well-established culture of working to undermine statehood and tearing apart the social fabric of society to thrive and flourish. Such a result will only come through an evident desire for change and a comprehensive vision of reform, strengthening the institutional capacity of the governance and, the transparency of policies, procedures and fairness (refer to 9.6 and K.3.2.3).
- 17. Libya does not need local investment to provide liquidity, but there is a requirement to develop its energy industry, particularly the oil and gas companies, which suffered greatly under US and UN economic sections. The new website, *www.investinlibya.com*, targets the promotion of foreign investment in industry, education, health care, agriculture and services, and will keep investors informed via new deregulation of the Libyan Foreign Investment Board. The economic reform programme, which began in 1999, has already witnessed new policies beginning to bear fruit, including restructuring banking rules and bureaucracy to build on previous measures such as passing new deregulation on foreign investment in 1997 and restructuring the real market which dates back to the late 1980s (refer to 2.3.4 and K.3.3.1).
- 18. Learning from some of the successful experiences in other developing countries regarding the role of the stock market performance in the economic reform programme is essential. As presented by the experience of emerging economies such as Chile, China, Egypt, India, Malaysia, Qatar, Singapore and United Arab Emirates, approaches to enterprise reform and ownership change might be appropriate models to follow. Conducting deregulation, corporatisation, privatisation and liberalisation programmes through the financial market may produce significant direct and indirect benefit for developing stock market performance (refer to 3.6 and K.3.4.6).

In the case of Libya, as illustrated by the research findings, the government should make efforts to promote the successful creation and implementation of economic reform, macroeconomic reform performance, financial maturity system and stock market performance. This can only occur with the mobilisation of the full range of human, capital and physical resources, since these are required to develop the particular area of economic policy and financial market on offer, to organise its marketing, control the sector and ensure its

integration into the global economy. Foreign, private sector undertakings will be the exception rather than the rule. Liberalising the civil society of deregulation that avoids freedom of association and liberalising media and encouragement of economic reform programme agenda. All of this will enable them to contribute to the control of public and private institutions and follow-up their activities and allow the detection of corruption. Nevertheless, some of the reforms will be problematic and there will certainly be resistance to change among some quarters. Libya, however, should reform if it is to complete its vision. The Basic People's Congress would need to play a critical role in reaching decisions as to which reforms should be embraced. Leadership will be needed from committed individuals who have the ability, ideas and courage to carry through the reform agenda and policy decisions. They will require support and resources from government to succeed in implementing reforms. Consequently, the government will need to ensure the results of the change programme and update the reform agenda on an annual basis, as its aims are successively achieved and new objectives arise.

10.6 LIMITATIONS OF THE RESEARCH

The research has successfully achieved its objectives and answered its questions to explore and study the economic reform programme and stock market performance in developing countries. Although almost all research has limitations, these might be explored in future studies based on the scope of the research. Keeping in mind the following limitations, this study should merely be perceived as a contribution to continue the economic reform programme and stock market performance in emerging economy research process in Libya and not as an end in itself. The limitation of this research can be achieved and discussed threefold: theoretically; empirically; methodologically.

10.6.1 Theoretically

- The major complexity with this research is that no studies have been previously performed regarding the implementation of economic reform programme influencing the stock market performance in the Libyan context. This would add to the originality and value of this research, as this research will not have the added benefit of learning from others' mistakes.
- There was a lack of literature concerning the economic reform programme and stock market performance in Libya in particular, and other MENA nations in general. Most of the available literature was related to developing countries including Eastern Europe, Latin America and developed countries for example, UK and US.

10.6.2 Empirically

- There were some problems with the distribution of the research questionnaire; the same method could not be adopted for all Libyan financial institutions and companies. Some respondents, for instance, refused to accept the questionnaire survey; they believed that the financial market is unethical, because they saw the stock markets as a form of gambling and in conflict with Sharia law. Others accepted the survey and volunteered to distribute it to the research sample, while others allowed the researcher to deliver the survey to the research respondents individually. This might have created inconsistency in the data collection process.
- There was a scarcity of empirical work previously conducted in the research area. This meant a lack of scales and measurements that could be used in order to establish a cause and influence study to examine the relationship between economic reform programme and stock market performance. Additionally, most of the available studies revealed contentious findings that do not encourage future research, as hitherto mentioned.

10.6.3 Methodologically

The data collection obtained from the questionnaires' survey suggested the use of nonparametric technical statistics. Thus, according to Field (2005), these were applied as "less powerful" than their parametric counterparts in that, if there is a significant effect in data analysis, a parametric test is more likely to detect it than a non-parametric one. By design, the non-parametric is a distribution-free test; it does not require the average score of each variable, as is the case in its parametric complement. Such a methodology confirms the ordinal scale statistical treatment due to the non-parametric test solely ordering or ranking responses in a certain way. There is, though, an increased chance of a Type II error, sometimes referred to as Beta (B), which involves the opposite occurring (i.e. accepting the null hypothesis when the alternate hypothesis is true) (see Chapter Six, Subsection 6.10.1.2.1.4). Furthermore, and unlike regression tests, this type of test is unable to provide an including of causality between variables. Employing non-parametric statistics with valid assumptions is methodologically "safer" than employing "robust" statistics with invalid or violated assumptions (Hollander and Wolfe, 1999). The "robustness" of using particular statistics lies in whether or not a statistics assumption is connected; if so, the statistic yields generally valid results (Leedy and Ormrod, 2005).

Regardless of the fact that it is recognised that this study has some inherent limitations, it is believed that the findings and adopted methodology of this study might form the basis for future research.

10.7 OUTLINE FOR FUTURE RESEARCH

This thesis has been intended throughout as a comprehensive study of the Libyan economic reform programme and its implication upon the country's economy and stock market performance. It is, therefore, important to determine where the field study finds Libya apparent today and the trends that are emerging. The current research effort is in line with such a direction (see Appendix B, Figure 2: The Structure of Research Framework Model). Whilst this study has significantly contributed research to the understanding of the growing empirical literature in developing countries, much still needs to be done as the research cannot cover the entire economic reform programme or stock market variables. With this in view, possible guidance and a number of ideas are offered for future extensions to this study, which could be undertaken following the results and theoretical framework stated in this research. This is discussed, subsequently in a number of areas.

10.7.1 Research Analysis

Future research could expand the analysis of this research to study other countries, extend the sample survey and examine the effects of other financial disclosure, financial intermediaries and the degree of openness to foreign portfolio. Other aspects for analysis would be to test various forms of the efficient market theory in the MENA countries, including Libya. Although, as more data becomes available in future, such material can be used for testing and identifying additional variables that could have influence on financial institutions and, therefore, more long-term research is needed in order to investigate the determinants of stock market performance over an extended period of time. Furthermore, study of the proposed model in other MENA countries could be performed in order to raise further explanation of the model and to reveal more generalised findings. In addition, comparison could be made with other MENA countries at different stages of development or between MENA economies and those of Sub-Saharan Africa. To this end, therefore, the recent liquidity crisis in world markets has had little impact on Libya due to the closed nature of its economy. As the situation changes over time and Libya's markets become more liberalised and international rather than internal-facing, further research would be needed to identify the impact additional liquidity crises may have within the nation's economy.

10.7.2 Economic Reform Programme

As this research is one of the first to include such variables of economic reform programme, additional research is required in order to confirm the Libyan findings. Due attention should be paid to examine the relative significance of, and the relationships among, stock market performance, a measurement that has been underestimated by most of the previous research. A study of deregulation, corporatisation, privatisation and liberalisation stability is still required to test more fully how far Libyan stock performance has achieved success and effectiveness as a consequence of applying an economic reform programme.

10.7.3 Macro-economic Reform

This study provides comprehensive empirical tests of macro- and micro-levels of relations with stock market performance. In this area more research is required.

- At macro-level, future research could measure the effect of stock market performance on capital accumulation, human capital, private savings rates and productivity growth. It additionally examined the foreign direct investment (FDI), which can be stated to be a significant variable to reveal its effect on the Libyan economy.
- At micro-level, future research could investigate the causal relationship between stock market and firms' investment rates, profitability ratio, technical changes and economic efficiency.

10.7.4 Financial Maturity

There is a need for further studies that examine the relation between a banking-based system and a market-based system by using M2 divided by GDP ratio as a common indicator of bank development (depth of bank intermediation). To measure a bank's development, Rousseau and Wachtel (2000) use M3 divided by GDP ratio. King and Levine (1993a, 1993b) used the total liquid liabilities of financial intermediaries M3 divided by GDP ratio. In the near future researchers could undertake further studies on financial maturity in Libya's banking system as an example from a developing country.

10.7.5 Stock Market Performance

• In this study there are other measures for stock market performance, such as volatility and stability measure. Further research could consider separate measures and tests for the

relationship between stock market performance and rate of economic growth. How changing those measures affects the rate of economic growth is an interesting topic for further research.

- Where all stock market performance variables are dependent could be a topic if examination of the interaction between these variables is conducted. Other independent variables in terms of economic reform programme, macro-economic reform and financial maturity may be used to evaluate the performance of individual companies upon a stock market.
- This study does not consider the effect of stock market liberalisation. Opening the stock market to foreign investment by raising demand for shares on the stock market increases liquidity and efficiency; this increases local investors' opportunities for portfolio diversification which, in turn, raises their motivation to invest in shares. It additionally increases liquidity, having an indirect effect the monitoring and control functions of stock markets. Furthermore, it should be expected that a more international integrated stock market has a potent effect on economic growth.

10.7.6 Comparison between Emerging Countries and the Libyan Stock Market

Chapter Nine analysised approaches adopted by some emerging countries in their moves towards establishing a framework for the positive impact of stock market performance. In this regard, the cases of emerging economies are observed, identifying similarities and differences across countries of different backgrounds. Examples of successful emerging markets are given, particularly between Libya and other emerging market countries. Including stock markets from other developed countries (such as Australia, Finland, France, Germany, Greece, Japan, Luxembourg, Singapore, Spain, UK and US) may have provided a wider and/or different understanding of other model practices in the Libyan context. Access, time and cost constraints and the sheer amount of work involved prevented the researcher from expanding the research setting.

On the whole, much remains to be learned about the relationship between economic reform programmes and stock market performance. This research represents a novel approach since it addresses four key aspects (economic reform programme, macro-economic reform performance, financial maturity and stock market performance). Moreover, this research is one of the first to investigate empirically the prime relationship between several contextual variables, namely, the economic reform programme, macroeconomic reform performance,

financial maturity and stock market performance in developing countries. The research is also one of the first to test the impact of several variables on the extent to which financial markets in developing countries can be informed by the Libyan case. Additionally, it contributes to the body of knowledge by providing some guidance for future economic reform programmes and can be applied to stock market performance, given its great potential for testing theories, controlling measurement error and validating research constructs. Consideration of whether to establish such financial markets depends on whether there is a strong demand for their services by the private sector. Emanating from the current research question, further development points are that the current financial crisis requires a change in the management policy in the financial sector in order to stimulate economic activities and the establishment of new regulatory frameworks. This requires a new theory of economics; for example, a change from equilibrium theory to reflexivity theory which requires a change in the underlying model of the economic activity framework. However, as mentioned in Chapter Five, Section 5.8, the main global economy was influenced by four crises, mutually feeding on each other, viz. other climate change, oil and energy, food and financial and economic crises. The global financial crisis, for instance, started to show its effects in the middle of 2007 and into 2008, when the stock markets around the world fell and large financial institutions collapsed. As a result, governments in even the wealthiest nations have had to come up with rescue packages to bail out their financial systems. On the other hand, many people are concerned that those responsible for the financial problems are the ones being bailed out. Whilst the current research effort concerns deregulation in Libya, it is still timely, since the question about the appropriate degree of regulation can be seen as part of the same debate as to what is the appropriate level of *regulation*. To quote Timothy Geithner, United States Secretary of the Treasury, "It's time now for us to move together and to begin to act to put in place a stronger framework of reforms" (Puzzanghera and Reynolds, 2009).

Further research is needed in order to provide a clear understanding of the framework that managers face under current regulatory rules in the banking system and financial markets in general. There is also a need to identify remedies for the developing regulatory framework in order to improve current practices. President Obama outlined two main goals for the G-20 *"The first is to make sure that there is concerted action around the globe to jumpstart the economy. The second goal is to make sure that we are moving forward on a regulatory reform agenda that ensures that we don't see these systemic risks and the potential for this kind of crisis again in the future"*. As a result of the current crisis, regulation is rising and is creating new challenges that need managing. Further concerns mentioned by Obama include *"We're*

moving forward in stabilising the financial system through a whole host of steps that have already been taken and a number of steps that we intend to take in the future to make sure that the financial system is solvent, that our banks are strong, and that we start lending again to businesses and consumers" (Sherman, 2009).

Regulation changes in the UK may help banks by improving liquidity, relating to the ability of the banks to manage cash and stimulate the UK economy. These funding needs may differ from those in other countries or those advanced by the Basel Committee. The discussion highlighted shows that this is not a normal cyclical crisis of capitalism but a global crisis, which needs to be tackled with new regulatory frameworks for financial institutions. Future research is needed to meet up-to-date information regarding the nature of capital markets and financial institutions. It is hoped, however, that this research will stimulate the economic reform programme, macro-economic reform performance, financial maturity and stock market performance in Libya in particular, as well as other comparable nations. Achieving these goals, and others, remains dependent on serious and comprehensive restructuring programmes.

REFERENCES ⁶³

- Aaker, D.A., Kumar, V. and Day, G.S. (2001) *Marketing Research*. 7th ed., New York: John Wiley and Sons.
- Abbas, H. (1987) *Industrial Development and Migrant Labour in Libya*. PhD. Unpublished Thesis, UK: Manchester University.
- Abdalla, I. and Al-Homoud, M. (1995) "A Survey of Management Training and Development Practices in the State of Kuwait", *Journal of Management and Development*, Vol. 14, No. 3: pp. 14-25.
- Abdelaziz, G. (2001) Service Quality in Egyptian Banking: Dimensions and their Relative Importance. PhD. Unpublished Thesis, UK: Nottingham University.
- Abdelnour, Z.K. (2003) "Democratisation of Capital in the Arab World", *Middle East Intelligence Bulletin*, May, Vol. 5, No. 5: p.1.
- Abel, A., Bernanke, B. and McNabb, R. (1998) *Macroeconomics*. Europe-Economic Conditions, Harlow: Addison-Wesley Longman Inc, New York.
- Abken, P.A. (1993) "Inflation and the Yield Curve", *Economic Review*, pp. 13-31.
- Abuarroush, M. (1996) A Cross-cultural Study of Locus of Control and Self-esteem as Related to Socio Cultural Factors among Libyan and British Postgraduate Students. PhD. Unpublished Thesis, UK: Manchester University.
- Acemoglu, D., Johnson, S.H. and Robinson, J.A. (2003) "The Rise of Europe: Atlantic Trade, Institutional Change and Economic Growth", *Working Papers*, No. 4269-02.
- AFP (2007) Gaddafi Spain Visit Could Yield Billions in Investment [online] Available at: http://www.france24.com/france24Public/en/news/europe/20071218-Libya-Spain-diplomacy-Kadhafi-contracts-billions.php [Accessed 15 May 2008].
- Aggarwal, R., Gray, I. and Singer, H. (1999) "Capital Rising in the Offshore Market", *Journal of Banking and Finance*, Vol. 23, Issue 8: pp. 1181-1194.
- Agnaia, A. (1996) "Assessment of Management Training Needs and Selection for Training: The Case of Libyan Companies", *International Journal of Manpower*, Vol. 17, No. 3: pp. 31-51.
- Ahmad, A. (2007) The Impact of Competitive Environment on Service Marketing Mix Strategy of Health Organisations in Developing Countries: Jordanian Private Sector Hospital Senior Managers Perspective. PhD. Unpublished Thesis, UK: Huddersfield University.
- Ahimud, Y., Mendelson, H. and Lantrebach, B. (1997) "Market Microstructure and Securities Values: Evidence from Tel Aviv Stock Exchange", New York University, Leonard N. Stern School Finance Department, Working Paper Series, Vol. 45: pp. 365-390.
- Akin, D. and Demircioglu, E. (2005) "Usability of Large Urban Facilities in Spatial Transformation-Case Study of Regional Shopping Centers in Istanbul", *ERSA*, Conference Papers, No. 05.

⁶³ This research used Harvard System referencing, as explained by the University of Huddersfield's Referencing Handbook, September 2007. University of Huddersfield, the Library and Commuting Centre, UK.

- Al-Ali, A. (1999) Human Resource Development Training and Development Practices and Related Organisational Factors in Kuwaiti Organisations. PhD. Unpublished Thesis, UK: Bradford University.
- Al-Athem, A. (2000) *The Impact of Multimedia Based Training on Employee Training Effectiveness* and Organisational Performance. PhD. Unpublished Thesis, UK: Bradford University.
- Al-Awad, M. and Harb, N. (2005) "Financial Development and Economic Growth in the Middle East", *Applied Financial Economics*, Vol. 15, Issue 15: pp. 1041-1051.
- Albahussain, A. (2000) Human Resource Development: An Investigation into the Nature and Extent of Training and Development in the Saudi Private Manufacturing. PhD. Unpublished Thesis, UK: Bradford University.
- Al-Duaij, A. (2004) An Investigation of the Current use of Fingerprints in the Saudi Justice System and the Barriers to their Admission as Evidence. PhD. Unpublished Thesis, UK: Huddersfield University.
- Alfaitori, E. (2003) "Nationalisation and Economic Development in Libya, Difficulties and Expected Economic Effects", *Economic Magazine for Economical Researches*, (in Arabic), June, Vol. 4: pp. 1-25.
- Al-Faleh, M. (1987) "Cultural Influences on Arab Management Development: A Case Study of Jordan", Journal of Management Development, Vol. 6, No. 3: pp. 19-33.
- Alfaitori, A. (2004) "The State Role in Privatisation Process", *The Privatisation in Libyan Economy*, Conference Hold on 29-30 December, *Centre of Economic Sciences*, (in Arabic), Benghazi: Libya.
- Al-Hroot, A.H. (2007) *Marketing of a Destination Jordan as a Case Study*. PhD. Unpublished Thesis, UK: Huddersfield University.
- Aljazeera Net (2008) *Talks to Sell Lehman Collapse* [online] Available at: http://english.aljazeera.net/news/americas/2008 [Accessed 16 September 2008].
- Aljerrah, M.A. (1993) *Trade Balance Instability and the Optimal Exchange Rate Regime: The Case of OPEC Countries.* United States-Nebraska. The University of Nebraska- Lincoln.
- Al-Khalifa, K. and Aspinwall, E. (2000) "The Development of Total Quality Management in Qatar", *The TQM Magazine*, Vol. 12, No. 3: pp. 194-204.
- Al-Rasheed, A. (1996) *Managers Motivation and Job Satisfaction*. PhD. Unpublished Thesis, UK: University of Kent at Canterbury.
- Al-Robiai, F. (2004) *The Impact of Privatisation on Development in the Developing Countries*. Cairo: Madbli Library (in Arabic).
- Alsharif, A. (2000) "Expansion in Education and Labour Market Requirement", *Journal of Economic Research*, (in Arabic), Vol. 11, Issue 1-2: pp. 75-105.
- Alsharif, A. (2002) "The Role of Small Enterprises in the Libyan Economy", *The Economic Development in Libya: Past and Present*, Conference (in Arabic), Tripoli: Libya.
- Altunisik, M.B. (1995) External vs. Internet Debate Revisited: The Political Economy Reform Policies in Libya (1987-1993). PhD. Unpublished Thesis, US: Boston University.

- AON Corporation (2004) Political and Economic Risk map Underlines the Need for Investors to Understand Emerging Market Risks [online] Available at: http://aon.mediaroom.com/index.php?s=43&item=434> [Accessed 8 November 2008].
- AON Corporation (2005) Business Should Beware of the Understandable Political Risks of Investing in
Emerging Market Risks [online] Available at:
<http://www.aon.com/uk/en/about/press_office/politicalap-2005.Jsp> [Accessed 8 November
2008].
- Arbab, F. (2005) *Engaging Libya: From Confrontation to Compliance* [online] Available at: www.issi.org.pk/journal/2005_files/no_3/article/a7.html 50k> [Accessed 12 August 2006].
- Arestis, P. and Demetriades, P.O. (1997) "Financial Development and Economic Growth: Assessing the Evidence", *Economic Journal*, Vol. 107, Issue 442: pp. 783-799.
- Arestis, P., Demetriades, P.O. and Luintel, B. (2001) "Financial Development and Economic Growth: The Role of Stock Markets", *Journal of Money, Credit and Banking*, Vol. 33, No. 1: pp. 16-41.
- Arlman, P. (2003) "All in the Name of Competition: European Securities Exchange", Journal of Financial Regulation and Compliance, Vol. 11, No. 3: pp. 236-240.
- Arnold, H. (2004) "Schumpeter and Methodological Individualism", *Journal of Evolutionary Economics*, Vol. 14, Issue 2: pp. 153-156.
- Arnold, L. and Walz, U. (2000) "Financial Regimes, Capital Structure, and Growth", European Journal of Political Economy, Vol. 16, Issue 3: pp. 491-508.
- Arrow, K. (1962) "The Economic Implications of Learning by Doing", *Review of Economic Studies*, Vol. 29, No. 3: pp. 155-173.
- Asprem, M. (1989) "Stock Prices, Asset Portfolios and Macroeconomic Variables in Ten European Countries", *Journal of Banking and Finance*, Vol. 13, Issue 4-5: pp. 589-612.
- Atje, R. and Jovanovic, B. (1993) "Stock Markets and Development", *European Economic Review*, Vol. 37, Issue 2-3: pp. 632-640.
- Awamleh, K. (2002) "Public Officials' Attitudes Towards Privatisation in Jordan A Field Survey", International Journal of Public Sector Management, Vol. 15, No. 3: pp. 237-256.
- Awda, A. and Malkawi, F. (1992) Principles of Scientific Research in Education and Human Sciences Irbid. Jordan: Al-Kittani Library.
- Aybar, C.B. (1992) Descriptive Analysis of Stock Return Behaviour in an Emerging Equity Market: The Case of Turkey. PhD. Unpublished Thesis, US: Ohio State University.
- Bae, H. (1990) "Multithreaded Bargaining", Journal of Economic Theory, Vol. 55, Issue 1: pp. 213-219.
- Bagehot, W. (1873) Lombard Street. 1962 ed., Irwin: Homewood IL.
- Bahmani-Oskooee, M. and Sohrabian, A. (1992) "Stock Prices and the Effective Exchange Rate of the Dollar", *Applied Economics*, Vol. 24, Issue 4: pp. 459-464.
- Balcerowicz, L. and Gelb, A. (1994) "How to Stabilize: Policy Lessons from Early Reformers", *Transition*, May-June, Vol. 5, No. 6: pp. 3-14.

- Baltagi, H.B., Demetriades, P. and Hook. L.S. (2007) "Financial Development, Openness and Institutions: Evidence from Panel Data", *World Economy and Finance, Working Paper*, No. 0022.
- Baltagi, H.B., Li, D. and Li, Q. (2007) "Transaction Tax and Stock Market Behaviour: Evidence from an Emerging Market", *Empirical Economics*, Vol. 31, Issue 2: pp. 393-408.
- Banziger, H. (2008) "Setting the Right Framework for Modern Financial Markets: Lessons Learned from the Recent Crisis", *Financial Stability Review*, No. 12: pp. 1-8.
- Baines, A. and Langfield-Smith, K. (2003) "Antecedents to Management Accounting Change: A Structural Equation Approach", Accounting, Organisations and Society, Vol. 28, Issue 7-8: pp. 675-698.
- Bait-Elmal, A. (1999) The Role of Management Control Systems in Libyan Organisations: A Libyan Development Policy Case Study with Special Reference to the Industrial Sector. PhD. Unpublished Thesis, UK: Manchester Metropolitan University.
- Barro, R. (1991) "Economic Growth in a Cross-section of Countries", *Quarterly Journal of Economics*, Vol. 106, No. 2: pp. 407-443.
- Barro, R. and Sala-i-Martin, X. (1995) Economic Growth. New York: McGraw-Hill.
- Barro, R. and What Lee, J. (1993) "Losers and Winners in Economic Growth", Proceedings of the World Bank Conference on *Development Economics*, World Bank, 1994, Washington, D.C. [online] Available at: htt://www.nber.org/papers/w4341.pdf> [Accessed 13 July 2006].
- Barton, S. and Gordon, P. (1987) "Corporate Strategy. Useful Perspective for the Study of Capital Structure", *Academy of Management Review*, Vol. 3, No. 1: pp. 1-7.
- Bassanini, A., Scarpetta, S. and Hemmings, P. (2001) "Economic Growth: The Role of Policies and Institutions: Panel Data. Evidence from OECD Countries", OECD Economics Department, Working Papers, No. 283: pp. 1-70.
- Bates, K. and Krueger, A. (1993) Generalisations Arising from the Country Studies. In: Bates, R. and Krueger, A., eds. Political and Economic Interactions in Economic Policy Reform: Evidence from Eight Countries. Oxford: Basil Blackwell.
- Baye, M. and Jansen, D.W. (1995) *Money, Banking and Financial Markets: An Economic Approach.* New Jersey: Houghton Mifflin College Div.
- BBC News (2007) *Gaddafi Visit Seals French Deals* [online] Available at: http://news.bbc.co.uk/2/hi/africa/7135788.stm> [Accessed 12 May 2008].
- BBC News (2008) US Libya Relations in "New Phase" [online] Available at: http://news.bbc.co.uk/1/hi/world/middle_east.stm> [Accessed 6 September 2008].
- BBC News (2008) *Global Market Turmoil Continues* [online] Available at: http://news.bbc.co.uk/1/hi/world/asia-pacific/7617976.stm> [Accessed 16 September 2008].
- Bechri, M.Z. and Naccache, S. (2003) *The Political Economy of Development Policy in Tunisia* [online] Available at: < http://www.gdnet.org/pdf2/gdngrowth/final.pdf> [Accessed 2 August 2008].
- Beck, T. and Levine, R. (2000) "New Firm Formation and Industry Growth-does having a Market or bank-based System Matter?", *Policy Research, Working Paper Series*, No. 2383: pp. 1-50.

- Beck, T., Demirgüc-Kunt, A. and Levine, R. (1999a) "A New Database on Financial Development and Structure", *World Bank Policy Research, Working Paper*, No. 2146.
- Beck, T., Demirgüc-Kunt, A. and Maksimovic, V. (2006) "The Influence of Financial and Legal Institutions on Firm Size", *Journal of Banking and Finance*, Vol. 30, Issue 11: pp. 2995-3015.
- Beck, T., Levine, R. and Loayza, N. (1999b) "Finances and the Sources of Growth", *Journal of Financial Economics*, Vol. 58, Issue 1-2: pp. 261-300.
- Beddies, C.H. (1999) "Investment, Capital Accumulation, and Growth-some Evidence from the Gambia 1964-98", *IMF*, *Working Papers*, No. 99/117.
- Bekaert, G. and Harvey, C. (1995) "The Cost of Capital in Emerging Markets", *Working paper*, Duke University and Stanford University.
- Bekaert, G. and Harvey, C. (1997) "Emerging Equity Market Volatility", *Journal of Financial Economics*, Vol. 43, Issue 1: pp. 29-77.
- Bekaert, G. and Harvey, C. (2003a) "Emerging Markets Finance", *Journal of Empirical Finance*, Vol. 10, Issue 1-2: pp. 3-65.
- Bekaert, G. and Harvey, C. (2003b) "Market Integration Contagion", NBER, Working Papers, No. 9510.
- Bekaert, G., Harvey, C. and Landblad, C. (2002) "Dating the Integration of World Equity Markets", *Journal of Financial Economics*, Vol. 65, Issue 2: pp. 203-247.
- Bekaert, G., Harvey, C.R. and Lundblad, C. (2005) "Does Financial Liberalisation Spur Growth?", *Journal of Financial Economics*, Vol. 77, Issue 1: pp. 3-55.
- Bencivenga, V.R. and Smith, B.D. (1991) "Financial Intermediation and Endogenous Growth", *Review* of Economic Studies Limited, Vol. 58, Issue 2: pp. 195-209.
- Bencivenga, V.R., Smith, B.D. and Starr, R.M. (1995) "Transaction Costs, Technological Choice, and Endogenous Growth", *Journal of Economic Theory*, Vol. 67, Issue 1: pp. 153-177.
- Bencivenga, V.R., Smith, B.D. and Starr, R.M. (1996) "Equity Markets, Transactions Costs, and Capital Accumulation: An Illustration", World Bank Economic Review, Vol. 10, Issue 2: pp. 241-265.
- Benderly, J. and Zwick, B. (1985) "Inflation, Real Balances, Output, and Real Stock Returns", *American Economic Review*, Vol. 75, No. 5: pp. 1115-1123.
- Berg, A. and Berg, E. (1997) "Methods of Privatisation (Privatisation and Economic Challenges)", Journal of International Affairs, Vol. 50: p. 1.
- Berg, A., Borensztein, E., Sahay, R. and Zettelmeyer, J. (1999) "The Evolution of Output in Transition Economies: Explaining the Differences", *International Monetary Fund*, Working Paper WP/99/73.
- Berthelemy, J.C. and Varoudakis, A. (1996) "Economic Growth, Convergence Clubs, and the Role of Financial Development", *Oxford Economic Papers*, Vol. 48, Issue 2: pp. 300-328.
- Bevan, A. and Danbolt, J. (2002) "Dynamics in the Determinants of Capital Structure in the UK", *Working Paper*, University of Glasgow.
- Bgcolor, B. (1996) "The Economic Reform Survey", Paper Presented at the *Center for International Private Enterprise*, World Bank, Washington, D.C.
- Bhide, A. (1993) "The Hidden Cost of Stock Market Liquidity", *Journal of Financial Economics*, Vol. 34, Issue 1: pp. 31-51.
- Binyon, M. (2005) "A Pragmatic Leader Now", The Times Focus Report, May 16, No. 68388: p. 3.
- Bird, G. (1992) Economic Reform in Eastern Europe. Edward Elgar: Publishing Ltd.
- Bittencourt, M.F.M. (2006) "Financial Development and Inequality: Brazil 1985-99", *Bristol Economic Discussion Papers*, No. 06/582: pp. 1-25.
- Blackburn, K. and Hung, V.T.Y. (1998) "A Theory of Growth, Financial Development, and Trade", *Economica*, New Series, Vol.65, No.257: pp. 107-124.
- Blanchard, O.J. (1981) "Output, the Stock Market, and Interest Rates", *American Economic Review*, Vol. 71, No. 1: pp. 132-143.
- Blinder, A.S. and Solow, R.M. (1973) "Does Fiscal Policy Matter", *Journal of Public Economics*, Vol. 2, Issue 4: pp. 319-337.
- Blommest, H. and Marrese, M. (1991) *Transformation of Planned Economies: Property Rights Reform and Macroeconomic Stability*. Organisation for Economic Co-operation and Development, Centre for Co-operation with European Economies in Transition, Paris: OECD.
- Bloom, D.E., Canning, D. and Malaney, P.N. (1999) "Demographic Change and Economic Growth in Asia", *CID*, *Working Papers*, No. 15.
- Blough, S. (1994) "Interest in the Future", Regional Review, p. 25.
- Boden, N. (2008) Black Monday: FTSE Plunges 212 Points as Global Markets Tumble Following Lehman Collapse [online] Available at: http://www.dailvmail.co.uk/news/worldnews/articles [Accessed 16 September 2008].
- Bolbol, A., Fatheldin, A. and Omran, M. (2005) "Financial Development, Structure, and Economic Growth: The Case of Egypt, 1974-2002", *Research in International Business and Finance*, Vol. 19, Issue 1: pp. 171-194.
- Bogdewic, S.P. (1992) Participant Observation. In: Crabtree, B.F. and Miller, W.L., eds. *Doing Qualitative Research*. Newbury Park: Sage Publications.
- Booth, L., Aivazian, V., Demirgüc-Kunt, A. and Maksimovic, V. (2001) "Capital Structures in Developing Countries", *Journal of Finance*, Vol. 56, No. 1: pp. 87-130.
- Bossone, B. (2000) "What Makes Banks Special? A Study on Banking, Finance and Economic Development", *Policy Research, Working Paper*, the World Bank, Washington, D.C.
- Boubakri, N. and Cosset, J.C. (1998) "The Financial and Operating Performance of Newly Privatised Firms: Evidence from Developing Countries", *Journal of Finance*, Vol. 53, No. 3: pp. 1081-1110.
- Boudoukh, J. and Richardson, M. (1993) "Stock Returns and Inflation: A Long-horizon Perspective", *American Economic Review*, Vol. 83, No. 5: pp. 1346-1355.

Bowerman, B.L. and O'connell, R.T. (1990) *Linear Approach*. 2nd ed., Belmont, CA: Duxbury.

- Boyd, J.H. and Prescott, E. (1986) "Financial Intermediary-coalitions", *Journal of Economic Theory*, Vol. 38, Issue 2: pp. 211-232.
- Boyd, J.H. and Smith, B.D. (1998) "The Co-evolution of the Real and Financial Sectors in the Growth Process", *Working Papers*, No. 541: pp. 1-43.
- Boyd, J.H., Levine, R. and Smith, B.D. (2001) "The Impact of Inflation on Financial Sector Performance", *Journal of Monetary Economics*, Vol. 47, Issue 2: pp. 221-248.
- Boyer, R.S. (1977) "Devolution and Portfolio Balance", *American Economic Review*, Vol. 67, No. 2: pp. 54-63.
- Bredin, D. and Fountas, S. (2005) "Macroeconomic Uncertainly and Macroeconomic Performance: Are they Related?", *Manchester School, University of Manchester*, Vol. 73, Issue 1: pp. 58-76.
- Brennan, M., Chordia, T. and Subrahmanyam, A. (1998) "Alternative Factor Specifications, Security Characteristics, and the Cross-section of Expected Stock Returns", *Journal of Financial Economics*, Vol. 41, Issue 3: pp. 341-364.
- Brunner, A.D. and Hess, G.D. (1993) "Are Higher Levels of Inflation Less Predictable? A Statedependent Conditional Heteroscedasticity Approach", *Journal of Business and Economic Statistics*, Vol. 11, Issue 2: pp. 187-197.
- Bruno, M. and Easterly, W. (1996) "Economic Reform and Growth Inflation's Children: Tales of Crisis that Beget Reform", *American Economic Review*, Vol. 86, No. 2: pp. 213-217.
- Bryman, A. (1993) Quantity and Quality in Social Research. London: Unwin Hyman.
- Bryman, A. (2007) *Qualitative Research*. 2nd ed., London: SAGE.
- Bryman, A. and Cramer, D. (2001) *Quantitative Data Analysis with SPSS Release 10 for Window: A Guide for Social Scientists.* London: Routledge.
- Burda, M. and Wyplosz, C. (2001) *Macroeconomics: A European Text*. Oxford: Oxford University Press, New York.
- Burton, G., Carrol, G. and Wall, S. (1989) *Quantitative Methods for Business and Economics*. London: Longman.
- Butler, K.C. and Malaikah, S.J. (1992) "Efficiency and Inefficiency in Thinly Traded Stock Markets: Kuwait and Saudi Arabia", *Journal of Banking and Finance*, Vol. 16, Issue 1: pp. 197-210.
- Cagan, P. (1974) "Inflation and Market Structure 1967-1973", *Explorations in Economic Research*, Vol. 2, No. 2: pp. 203-216.
- Cameron, R. (1967) Banking in the Early Stages of Industrialisation: A Study in Comparative Economic History. New York: Oxford University Press.
- Camdessus, M. (1997) "Old Battles and New Challenges: A Perspective on Latin America", *International Monetary Fund*, October 18, World Bank, Washington, D.C.
- Canova, F. and De Nicolo, G. (2000) "Monetary Disturbances Matter for Business Fluctuations in the G-7", *International Finance*, Discussion Papers, No. 660.
- Cass, D. (1965) "Optimum Growth in an Aggregative Model of Capital Accumulation", *Review of Economic Studies*, Vol. 32, No. 3: pp. 233-240.

- CBL Central Bank of Libya, Economic Reports, Various Issues.
- CBL Central Bank of Libya (1966-2000) Research and Statistics, Economic Bulletin, Money and Statistics, Tripoli, Libya.
- CBL Central Bank of Libya (1979) Research and Statistics, Economic Bulletin, 23rd Annual Report, Tripoli, Libya.
- CBL Central Bank of Libya. (2000) "Financial and Banking Statistic: 1966-2000", *Economical Researches Administration*, Tripoli, Libya.
- CBL Central Bank of Libya (2002) *Research and Statistics, Economic Bulletin, 46th Annual Report,* Tripoli, Libya.
- CBL Central Bank of Libya (2006) Research and Statistics, Economic Bulletin, for the Fourth Quarter of 2006, Tripoli, Libya.
- CBL Central Bank of Libya, (2007) *Research and Statistics, Economic Bulletin, 51st Annual Report,* Tripoli, Libya.
- CBL Central Bank of Libya, (2008) *Research and Statistics, Economic Bulletin, 1st Quarter of the year 2008*, Tripoli, Libya.
- CEPS Centre for European Policy Studies (2005) "From Barcelona Process to Neighbourhood Policy Assessments and Open Issue Michael Emerson and Gergana Noutcheva", *CEPS, Working Document*, No. 220/ March, pp. 1-23.
- Chai, J. and Roy, K. (2006) *Economic Reform in China and India: Development Experience in a Comparative Perspective*. Edward Elgar: Publishing, Limited, UK.
- Chakraborty, I. (2008) "Does Financial Development Cause Economic Growth? The Case of India", *South Asia Economic Journal*, Vol. 9, No. 1: pp. 109-139.
- Charemza, W. (1991) "Alternative Paths for Macroeconomic Stabilisation in Czechoslovakia", *European Economy*, Special Edition, No. 2: pp. 41-56.
- Chen, N., Roll, R. and Ross, S.A. (1986) "Economic Force and the Stock Market", *Journal of Business*, Vol. 59, No. 3: pp. 383-403.
- Cheung, Y.W. and Ng, L. (1998) "International Evidence on the Stock Market and Aggregate Economic Activity", *Journal Empirical Finance*, Vol. 5, Issue 3: pp. 281-296.
- Cho, Y.J. (1986) "Inefficiencies from Financial Liberalisation in the Absence of Well-functioning Equity Markets", *Journal of Money, Credit and Banking*, Vol. 18, Issue 2: pp. 191-199.
- Chordia, T., Subrahmanyam, A. and Anshuman, V. (2001) "Trading Activity and Expected Stock Returns", *Journal of Financial Economic Review*, Vol. 43, Issue 1: pp. 3-32.
- Chow, C. (1997) "Challenges of China's Economic System for Economic Theory", American Economic Review, Vol. 87, No. 2: pp. 321-327.
- Churchill, N. and Lewis, V. (1983) "Five Stage of Small Business Growth", *Harvard Business Review*, Vol. 61: pp. 30-50.

Churchill, G. (2001) Marketing Research. 4th ed., London: The Dryden Press.

- Claessens, S. and Rhee, M. (1994) "The Effects of Barriers on Equity Investment in Developing Countries", *NBER, Working Paper*, No. W4579.
- Claessens, S., Djankov, S. and Larry, H.P. (2000) "The Separation of Ownership and Control in East Asia Corporation", *Journal of Financial Economics*, Vol. 58, Issue 1-2: pp. 81-112.
- Clague, C. and Rausserc, G.C. (1993) The Emergence of Market Economies in Eastern Europe. Oxford: Blackwell.
- Cobham, D. (1995) "Financial Systems for Developing Countries, with Particular Reference to Egypt, Iraq, Jordan, Lebanon, and Syria. In Economic Research Form for the Arab Countries, Iran, and Turkey (eds)", *Development of Financial Markets in the Arab Countries, Iran and Turkey*. Cairo-Egypt.
- Cohen, J. (1992) *Statically Power Analysis for the Behavioural Sciences*. 2nd ed., New Jersey: Lawrence Erlbaum: pp. 360-379.
- Colijn, L. (1991) "East Central European Economies in Transition: Long-term Performance, New Aspirations and the Costs of Realising Them", Background Material for the Institute of East West Security Studies Task Force Workshop on Western Assistance, *The European Studies Centre at Czechoslovakia*, 27-29 September.
- Collier, P. and Maryer, Y. (1989) "The Assessment: Financial Liberalisation, Financial Systems, and Economic Growth", *Oxford Review of Economic Policy*, Vol. 5, No. 4: pp. 1-12.
- Conrad, K. and Juttner, D. (1973) "Recent Behaviour of Stock Market Prices in Germany and the Random Walk Hypothesis", *Journal Kyklos*, Vol. 26, Issue 3: pp. 576-599.
- Cook, P. and Uchida, Y. (2003) "Privatisation and Economic Growth in Developing Countries", *Journal of Development Studies*, Vol. 39, Issue 6: pp. 121-154.
- Cooley, T.F. and Hansen, G.D. (1989) "The Inflation Tax in a Real Business Cycle Model", *American Economic Review*, Vol. 79, Issue 4: pp. 733-748.
- Cooper, D.R. and Schindler, P.S. (2006) Business Research Methods. 9th ed., London: Irwin Ethics, Boston.
- Creane, S., Goyal, R., Mobarak, A.M. and Sab, R. (2004) "Financial Sector Development in the Middle East and North Africa", IMF, *Working paper*, No. 04/102, Washington, D.C.
- Cowton, C.J. (1998) "The Use of Secondary Data in Business Ethics Research", *Journal of Business Ethics*, Vol. 17, No. 4: pp. 423-434.
- Cowton, C.J. (2008) "Corporate Governance and the Theory of the Firm: A Re-assessment of Shareholder Primacy in the Light of Limited Liability and the Position of Creditors", *Financial Ethics and Governance Research Group, Working Paper*, University of Huddersfield, Huddersfield, UK, (Unpublished), pp. 1-20.
- Creswell, W.J. (2003) *Research Design: Qualitative, Quantitative and Mixed Methods Approaches.* 2nd ed., Thousand Oaks, Calif.
- Cronin, B. (2001) "Productive and Unproductive Capital: A Mapping of the New Zealand System of National Accounts to Classical Economic Categories, 1972-95", *Review of Political Economy*, Vol. 13, Issue 3: pp. 309-327.
- Curley, J.G. and Shaw, E.S. (1955) "Financial Aspects of Economic Development", American Economic Review, Vol. 45, No. 4: pp. 515-538.

- Currie, C.V. (2003) "The Need for a New Theory of Economic Reform", UTS, Working Paper, No. 131: pp. 1036-7373.
- Dabrowski, M., Gomulka, S. and Rostowsti, J. (2001) "Whence Reform? A Critique of the Stiglitz Perspective", *Journal of Policy Reform*, Vol. 4, No. 4: pp. 1-33.
- Dactar, V., Naik, N. and Radcliffe, R. (1998) "Liquidity and Stock Returns: An Alternative Test", *Journal of Financial Markets*, Vol. 1, Issue 2: pp. 203-219.
- Daniel, F. (1999) "Provision of Electronic Banking in the UK and the Republic of Ireland", *International Journal of Bank Marketing*, Vol. 17, No. 2: pp. 72-82.
- Daniel, W.B. and Yang, Y. (2006) "Understanding China's Economic Transformation", Article Overview, Vol. 7, No. 2: pp. 73-95.
- Darrat, A.F. (1988) "On Fiscal Policy and the Stock Market", *Journal of Money, Credi and Banking*, Vol. 20, No. 3, Part 1: pp. 353-363.
- Das, D. (2006) China and India: A Tale of Two Economies. 1st ed., London: Routledge.
- Ray, D. (1998) Development Economics. Princeton: Princeton University Press.
- De Gregorio, J. (1993) "Inflation, Taxation and Long-run Growth", *Journal of Monetary Economics*, Vol. 31, Issue 3: pp. 271-298.
- De Haan, J. and Sturm, J. (2000) "On the Relationship between Economic Freedom and Economic Growth", *European Journal of Political Economy*, Vol. 16, Issue 2: pp. 215-241.
- De Long, B.J. (1988) "Productivity Growth, Convergence and Welfare: Comment", American Economic Review, Vol. 78, No. 5: pp. 1138-1154.
- De Long, J.B. and Summers, L.H. (1991) "Equipment Investment and Economic Growth", *Quarterly Journal of Economics*, Vol. 106, No. 2: pp. 445-502.
- De Long, J.B. and Summers, L.H. (1993) "How Strongly do Developing Economies Benefit from Equipment Investment?", *Journal of Monetary Economics*, Vol. 32, Issue 3: pp. 395-415.
- De Melo, M., Denizer, C. and Gelb, A. (1997) Transition to Date: A Comparative Overview. In: Salvatore, Z., ed. Lessons from the Economic Transition. Central and Eastern Europe in the 1990s, Dordrecht: Kluwer Academic Publishers, pp. 59-78.
- Demetriades, P.O. and Andrianova, S. (2004) Finance and Growth: What we Know and What we Need to Know. In: Goodhart, C., ed. *Money, Finance and Growth*. Routledge, forthcoming.
- Denzin, N.K. and Lincoln, Y.S. (2000) Handbook of Qualitative Research. 2nd ed., London: SAGE, UK.
- De Santis, G. and Imrohoroglu, S. (1997) "Stock Returns and Volatility in Emerging Financial Markets", *Journal of International Money and Finance*, Vol. 16, Issue 4: pp. 561-579.
- DeGregorio, J. and Giudotti, E. (1995) "Financial Development and Economic Growth", *World Development*, Vol. 23: pp. 433-448.
- Deidda, L.G. (2006) "Interaction between Economic and Financial Development", *Journal of Monetary Economics, Elsevier*, Vol.53, Issue. 2: pp. 233-248.

- Deidda, L.G. and Fattouh, B. (2002) "Non-linearity between Finance and Growth", *Economic Letters*, Vol. 74, Issue 3: pp. 339-345.
- Delong, J. (2001) *India since Independence: An Analytic Growth Narrative* [online] Available at: http://ksghome.harvard.edu/~drodrik/Growth%20volume/DeLong-India.pdf [Accessed 3 March 2007].
- Demetriades, P. and Hussein, K. (1996) "Does Financial Development Cause Economic Growth? Time-series Evidence from 16 Countries", *Journal of Development Economics*, Vol. 51, Issue 2: pp. 387-411.
- Demirgüc-Kunt, A. and Huizinga, H. (2000) "Financial Structure and Bank Profitability", *Policy Research, Working Paper Series*, No. 2430: pp. 1-30.
- Demirgüc-Kunt, A. (1992) "Developing Country Capital Structures and Emerging Markets", *Policy Research Working Paper Series*, No. 933.
- Demirgüc-Kunt, A. and Detrgiche, E. (1998) "The Determinants of Banking Crises in Developing and Developed Countries", *International Monetary Finance, Staff Papers*, Vol. 45, No. 1: pp. 1-29.
- Demirgüc-Kunt, A. and Detrgiche, E. (1999) "Does Deposit Insurance Increase Banking System Stability? An Empirical Investigation", *Policy Research, Working Paper Series*, No. 2247.
- Demirgüc-Kunt, A. and Levin, R. (2004) Financial Structures and Economic Growth: A Cross-country Comparison of Banks, Markets and Development. Cambridge: MA, MIT Press.
- Demirgüc-Kunt, A. and Levine, R. (1995) "Stock Markets Development and Financial Intermediaries: Stylized Facts", *World Bank Economic Review*, Vol. 10, No. 2: pp. 291-321.
- Demirgüc-Kunt, A. and Levine, R. (1996) "Stock Market, Corporate Finance and Economic Growth: An Overview", *World Bank Economic Review*, Vol. 10, Issue 2: pp. 223-239.
- Demirgüc-Kunt, A. and Levine, R. (1999) "Bank-based and Market-based Financial System: Crosscountry Comparisons", *World Bank Policy*, *Working Paper*, No. 2143: pp. 1-72.
- Demirgüc-Kunt, A. and Levine, R. (2008) "Finance, Financial Sector Policies and Long-run Growth", *World Bank Policy, Working Paper*, No. 4469: pp. 1-82.
- Demirgüc-Kunt, A. and Maksimovic, V. (1995) "Stock Market Development and Firm Financing Choices", *Policy Research, Working Paper Series*, No.1461: pp. 1-48.
- Demirgüc-Kunt, A. and Maksimovic, V. (1996) "Financial Constraints, Uses of Funds, and Firm Growth: An International Comparison", *Policy Research, Working Paper Series*, No.1461: pp. 1-54.
- Demirgüc-Kunt, A. and Maksimovic, V. (1998) "Law, Finance and Firm Growth", *Journal of Finance*, Vol. 53, No. 6: pp. 2107-2137.
- Demirgüc-Kunt, A. and Maksimovic, V. (1999) "Institutions, Financial Markets and Firm Debt Maturity", *Journal of Financial Economics*, Vol. 54, Issue 3: pp. 295-336.
- Demirgüc-Kunt, A. and Maksimovic, V. (2000) "Funding Growth in Bank-based and Market-based Financial Systems: Evidence from Firm Level Data", *Policy Research, Working Paper*, No. 2432, World Bank, Washington, D.C.

- Demirgüç-Kunt, A., Laeven, L. and Levine, R. (2004) "Regulations, Market Structure, Institutions, and the Cost of Financial Intermediation", *Journal of Money, Credit and Banking*, Vol. 36, No. 2, Part 2: pp. 593-622.
- Derrabi, M. and Leseure, M. (2005) Global Asset Allocation: Risk and Return on Emerging Stock Markets. In: Motamen-Samadian, S., ed. *Risk Management in Emerging Markets*. UK: London, Palgrave Macmillan.
- Dewald, W.G. (1982) "Disentangling Monetary and Fiscal Policy", Economic Review, pp. 7-18.
- Diamond, D.W. (1984) "Financial Intermediation and Delegated Monitoring", *Review of Economic Studies*, Vol. 51, No. 3: pp. 393-414.
- Diamond, D.W. (1996) "Financial Intermediation as Delegated Monitoring: A Simple Example", *Federal Reserve Bank of Richmond Economic Quarterly*, Summer, Vol. 82-3: pp. 51-66.
- Dimitrova, D. (2005) "The Relationship between Exchange Rates and Stock Prices: Studied in A Multivariate Model", *Political Economy*, Vol. 14: pp. 1-25.
- Divecha, A.B., Drach, J. and Stefek, D. (1992) "Emerging Markets: A Quantitative Perspective", Journal of Portfolio Management, Vol. 19, No. 1: pp. 41-50.
- Domar, E. (1946) "Capital Expansion, Rate of Growth, and Employment", American Economic Review, Vol. 14, No. 2: pp. 343-355.
- Dornbusch, R. (1975) "Exchange Rates and Fiscal Policy in a Popular Model of International Trade", *American Economic Association*, Vol. 65, No. 5: pp. 859-871.
- Dornbusch, R. and Fisher, S. (1980) "Exchange Rates and the Current Account", *American Economic Association*, Vol. 70, No. 5: pp. 960-971.
- Dowrick, S. (1992) "Enterprise Bargaining, Union Structure and Wages", *The Economic Record*, Vol. 69, No. 207: pp. 393-404
- Drever, E. (1995) Using Semi-structure Interviews in Small-scale Research. Glasgow: The Scottish Council for Research in Education.
- Dryden, M. (1970) "A Statistical Study of UK Share Prices", *Scottish Journal of Political Economy*, Vol. 17, Issue 3: pp. 369-389.
- Duyn, A.V. (1993) "Taiwan Emerges Top", Euromoney, December, pp. 68-70.
- Dwyer, G.P. (1982) "Inflation and Government Deficits", *Economic Inquiry*, Vol. 20, Issue 3: pp. 315-329.
- Easterby-Smith, M., Thorpe, R. and Lowe, A. (2002) *Management Research: An Introduction*. 2nd ed., London: Sage Publications.
- Easterly, W. and Levine, R. (2003) "Tropics, Germs, and Crops: How Endowments Influence Economic Development", *Journal of International Money and Finance*, Vol. 27, Issue 1: pp. 749-776.
- Edris, T. and Almahmeed, M. (1997) "Services Considered Important to Business Customers and Determinants of Bank Selection in Kuwait: A Segmentation Analysis", *International Journal of Bank Marketing*, Vol. 15, No. 4: pp. 126-133.

- Edwards, S. (1993) "Openness, Trade Liberalisation and Growth in Developing Countries", *Journal of Economic Literature*, Vol. 31, Issue 3: pp. 1358-1393.
- Edwards, S. (1995) "Why are Saving Rates so Different Cross Countries?: An International Comparative Analysis", *NBER, Working Paper*, No.5097, National Bureau of Economic Research, Inc.
- Eldabi, T., Irani, Z., Paul, R. and Love P. (2002) "Quantitative and Qualitative Decision-making Methods in Simulation Modelling", *Management Decision*, Vol. 40, No. 1: pp. 64-73.
- El-Ftouh, A. (1997) "Guide to Morocco's Privatisation Programme", *Journal of Economic Cooperation Among Islamic Countries*, Vol. 18: pp. 61-106.
- El-Naggar, S. (1989) "Privatisation and Structural Adjustment in the Arab Countries," *International Monetary Fund*, World Bank, Washington, D.C.
- El-Qumati, S.M. (1993) "The Role of Institutions and Financial Markets in Restructuring the Libyan Economy", *Conference*, 29-30 December, (in Arabic), Benghazi, Libya, pp. 1-25.
- Energy Information Administration/Department of Energy (2005) *Libya: Country Analysis Brief* [online] Available at: http://www.eia.doe.gov/emeu/cabs/Libya.html [Accessed 5 May 2006].
- Engle, R.F. (2004) "Risk and Volatility: Econometric Models and Financial Practice", *American Economic Review*, Vol. 94, Issue 3: pp. 405-420.
- Engle, R.F. and Granger, C.W.J. (1987) "Co-integration and Error Correction Representation, Estimation and Testing", *Econometrica*, Vol. 55, No. 1: pp. 251-276.
- Evans, P. (1985) "Do Large Deficits Produce High Interest Rates", *American Economic Review*, Vol. 75, No. 1: pp. 68-87.
- Faiadh, A. and Almejairy, M. (2003) "Public Expenditure and GDP Growth of the Non-oil Sector Maintains Libyan Experiment 1980-1999", Paper Presented at the Conference on Economic Development in Libya: Past and Future, 14-16 December, Benghazi: Libya, Journal of Economic Research, (in Arabic), Vol. 14, No. 1: pp. 145-156.
- Fama, E.F. (1965) "The Behaviour of Stock Market Prices", *Journal of Finance*, Vol. 38, No.1: pp. 383-417.
- Fama, E.F. (1975) "Short-term Interest Rates as Predictors of Inflation", *American Economic Review*, Vol. 65, No. 3: pp. 269-282.
- Fama, E.F. (1976) "Efficient Capital Markets: Reply", *Journal of Finance*, Vol. 31, Issue 1: pp. 143-145.
- Fama, E.F. (1981) "Stock Returns, Real Activity, Inflation, and Money", *American Economic Review*, Vol. 71, Issue 4: pp. 545-565.
- Fama, E.F. (1985) "What's Different about Banks?", *Journal of Monetary Economics*, Vol. 15, Issue 1: pp. 29-39.
- Fama, E.F. (1989) "Permanent and Temporary Components of Stock Prices", *Journal of Political Economy*, Vol. 96, Issue 2: pp. 246-273.
- Fama, E.F. (1990) "Term-structure Forecasts of Interest Rates, Inflation and Real Returns", Journal of Monetary Economics, Vol. 25, Issue 1: pp. 59-76.

- Fama, E.F. and French, K.R. (1989) "Business Conditions and Expected Returns on Stocks and Bonds", *Journal of Financial Economics*, Vol. 25, Issue 1: pp. 23-49.
- Fanelli, J.M. (2007) Understanding Market Reform in Latin America: Similar Reforms, Diverse Constituencies, Varied Results. 1st ed., Palgrave Macmillan.
- Fang, Y. (2000) "Seasonality in Foreign Exchange Volatility", *Applied Economics*, Vol. 32, Issue 6: pp. 697-703.
- Farley, R. (1971) *Planning for Development in Libya, in the Exceptional Economy in the Developing World.* New York: Praeger Publishers.
- Fayad, M. (2000) *Government Expenditure and Growth in Libya*. PhD. Unpublished Thesis, UK: Liverpool John Moores University.
- Farmer, D. (1997) "Contemporary Conceptual Space: Reading Adam Smith", *Journal of Management History*, Vol. 3, No. 3: pp. 246-255.
- Fendel, R. and Frenkel, M. (2005) "Monetary Policy Rules and the Information Content of the Term Structure of Interest Rates", *Applied Economics Letters*, Vol. 12, Issue 15: pp. 933-936.
- Ferrero, M. (2004) "Revolution or Reform? Socialism's Dilemma as Rational Choice Problem", *Journal of Homo Oeconomicus*, Vol. 21: pp. 251-282.
- Fernández, A.H. (2006) Libya's Return: Between Change and Continuity, Area: Mediterranean and Arab. World-ARI 58/2006. Translated from Spanish [online] Availabl at: http://www.realinstitutoelcano.org/analisis/Libya.pdf [Accessed 8 February 2007].
- Ferson, W.E. and Harvey, C. (1991) "The Variation of Economic Risk Premiums", *Journal of Political Economy*, Vol. 99, Issue 2: pp. 385-415.
- Fialkowski, D. and Fetersen, M.L. (1994) "Posted Versus Effective Spreads: Good Prices or Bad Quotes?", *Journal of Financial Economics*, Vol. 35, Issue 3: pp. 269-292.
- Field, A. (2005) Discovering Statistics Using SPSS. 2nd ed., London: Sage, UK.
- Fielding, D. (1997) "Adjustment, Trade Policy and Investment Slumps: Evidence from Africa", Journal of Development Economics, Vol. 52, Issue, 1: pp. 121-137.
- Fielding, N. and Fielding, J. (1986) Linking Data. London: Beverly Hills, Sage.
- Financial Times (2008) *Markets Data Equities* [online] Available at: http://markets.ft.com/ft/markets/worldEquities.asp [Accessed 16 September 2008].
- Fisher S., Sahay, R. and Vega, C. (1998) "From Transition to Market: Evidence and Growth Prospects", *International Monetary Fund, Working Paper*, WP/98/52, Washington, D.C.
- Fisher, I. (1933) The Theory of Interest. New York: Macmillan.
- Fisher, S. (1990) *Libya, the Middle East and North Africa Report.* 36th ed., London: Europe Publications Limited.
- Fisher, S. and Gelp, A. (1991) "The Process of Socialist Economic Transformation", *Journal of Economic Perspectives*, Vol. 5, No. 4: pp. 91-105.
- Fisher, S. and Sahay, R. (2000) "The Transition Economies After Ten Years", *International Monetary Fund*, *Working Paper*, WP/00/30.

- Fisher, S. and Sahay, R. (2004) "Transition Economies: The Role of Institutors and Initial Conditions", *International Monetary Fund*, Calvo Conference, 15-16 April, pp. 1-29.
- Fisher, S., Sahay, R. and Vegh, C. (1998) "How far is Eastern Europe from Brussels?", *International Monetary Fund, Working Paper*, WP/98/53, Washington, D.C.
- Fisher, S., Sahay, R. and Vegh, C. (1996a) "Stabilization and Growth in Transition Economies: The Early Experience", *Journal of Economic Perspective*, Vol. 10, No. 2: pp. 45-66.
- Fisher, S., Sahay, R. and Vegh, C. (1996b) "Economies in Transition: The Beginnings of Growth", *American Economic Review*, Vol. 86, No. 2: pp. 229-233.
- Flannery, M.J. (1994) "Debt Maturity and the Deadweight Cost of Leverage: Optimally Financing Banking Firms", *American Economic Review*, Vol. 48, No. 1: pp. 320-331.
- Flannery, M.J. and James, C.M. (1984) "Market Evidence on the Effective Maturity of Bank Assets and Liabilities", *Journal of Money, Credit and Banking*, Vol. 16, Issue 4: pp. 435-445.
- Foster, N. and Stehrer, R. (2005) "Modelling GDP in CEECs Using Smooth Transitions", Vienna Institute for International Economic Studies, Working Papers, No. 36: pp. 1-47.
- Frances, J. and Garnsey, E. (1996) "Supermarkets and Supplies in the United Kingdom, System Integration, Information and Control", *Accounting, Organisation and Society*, Vol. 21, Issue 6: pp. 591-610.
- Francis, J.L. and Kompas, T. (2001) "Uzawa's Transformation and Optimal Control Problems with Variable Rates of Time Preference", *Microeconomics, Working Paper*, No. 576.
- Friedman, M. and Schwartz, A.J. (1963) "The Definition of Money: Net Wealth and Neutrality as Criteria", *Journal of Money, Credit and Banking*, Vol. 1, Issue 1: pp. 1-14.
- Froot, K.A., O'Connell, G.J. and Seasholes, M. (2001) "The Portfolio Flow of International Investors", *Journal of Financial Economics*, Vol. 59, Issue 2: pp. 151-193.
- Fry, M.J. (1980) "Money, Interest, Inflation and Growth in Turkey", *Journal of Monetary Economics*, Vol. 6, Issue 4: pp. 535-545.
- Fry, M.J. (1982) "Models of Financially Repressed Developing Economies", *World Development*, Vol. 10, Issue 9: pp. 731-750.
- Fulghieri, P. and Rovelli, R. (1998) "Capital Markets, Financial Intermediaries and Liquidity Supply", *Journal of Banking and Finance*, Vol. 22, Issue 9: pp. 1157-1180.
- Füss, R. (2002) "The Financial Characteristics between 'Emerging' and 'Developed' Equity Markets in: EcoMod 2002", *International Conference on Policy Modelling*, 4-6 July, Proceedings Brussels, Brussels, CD-Rom.
- Gaber, J. (1993) "Reasserting the Importance of Qualitative Methods in Planning", *Landscape and Urban Planning*, Vol. 26, No. 1-4: pp. 137-148.
- Galal, G.H. (2001) "From Context to Constructs: The Use of Grounded Theory in Operationalising Contingent Process Models", *European Journal of Information Systems*, Vol. 10, No. 2: pp. 2-14.
- Galetovic, A. (1996) "Specialisation, Intermediation, and Growth", *Journal of Monetary Economics*, Vol. 38, Issue 3: pp. 594-559.

- Garcia, G. (1989) "The Lender of Last Resort in the Wake of the Crash", *American Economic Review*, Vol. 79, Issue 2: pp. 151-155.
- Garcia, V. and Liu, L. (1999) "Macroeconomic Determinants of Stock Market Development", *Journal* of Applied Economics, Vol. 2, Issue 5: pp. 29-56.
- Gandhi. D.K., Anthony, S. and Richard, S.W. (1980) "Thin Capital Markets: A Case Study of the Kuwaiti Stock Market", *Applied Economics*, Vol. 12, Issue 3: pp. 341-349.
- Gedeon, S. (1986) "Yugoslav Socialism, Theory and Practice: Oxford: Clarendon Press", *Journal of Comparative Economics*, Vol. 10, Issue 3: pp. 361-362.
- Gentzoglanis, A. (2007) "Financial Integration, Regulation and Competitiveness in Middle East and North Africa Countries", *Managerial Finance*, Vol. 33, No. 7: pp. 461-476.
- Gerard, J. and Demirgüc-Kunt, A. (1997) "The Role of Long Term Finance: Theory and Evidence", *Policy Research, Working Paper Series*, No. 1746.
- Gertler, M. (1988) "Financial Structure and Aggregate Economic Activity: An Overview", *Journal of Money, Credit and Banking*, Vol. 20, Issue 3: pp. 559-88.
- Geweke, J. (1992) "Priors for Macroeconomic Time Series and Their Application", *Institute for Empirical Macroeconomics, Discussion Paper*, No. 64.
- Ghani, E. (1992) "How Financial Markets Affect Long-run Growth: A Cross Country Study", *Policy Research, Working Paper Series*, No. 843.
- Ghani, H. (1992) "How Financial Markets Affect Growth: A Cross-country Study", *Policy Research*, *Working Paper*, No. 848, World Bank, Washington, D.C.
- Ghura, D. (1997) "Private Investment and Endogenous Growth-evidence from Cameroon", *IMF*, *Working Papers*, No. 97/165.
- Giurnaz, A. (1985) *Economic of Oil: A Case of Libya*. PhD. Unpublished Thesis, UK: University of Keel.
- Giovannini, A. (1983) "The Interest Elasticity of Savings in Developing Countries: The Existing Evidence", *World Development*, Vol. 18, Issue 7: pp. 601-607.
- Giovannini, A. (1985) "Saving and the Real Interest Rate in LDCs", Journal of Development Economics, Vol. 18, Issue 2-3: pp. 197-217.
- Girdner, E.J. and Siddiqui, K. (2008) "Neoliberal Globalisation, Poverty Creation and Environmental Degradation in Developing Countries", *International Journal of Environment and Development*, Vol. 5, No. 1: pp. 1-27.
- Glen, J. and Singh, A. (2003) "Capital Structure, Rates of Return and Financing Corporate Growth: Comparing Developed and Emerging Markets, 1994-2000", *Working Paper Series*, March.
- Goldsmith, R.W. (1969) Financial Structure and Development. New York: Yale University Press.
- Gomulka, S. (1994) "Economic and Political Constraints during Transition", *European-Asia Studies*, Vol. 46, No. 1: pp. 89-106.
- Goode, M. and Moutinho, L. (1995) "The Effects of Free Banking on Overall Satisfaction: The USE of Automated Teller Machines", *International Journal of Bank Marketing*, Vol. 13, No. 4: pp. 33-40.

- GPC General People's Committee Resolution 242 (2003) Establishment of Capital Market would be the Responsibility of the Central Bank of Libya.
- GPC General People's Committee for the Production (2004) Tripoli, Libya: pp. 4-11.
- GPC General People's Committee Resolution 134 (2006) Regarding the Revised Establishment of a Stock Market in Libya.
- Granger, C.W.J. (1980) "Long Memory Relationships and the Aggregation of Dynamic Models", *Journal of Econometrics*, Vol. 14, Issue 2: pp. 227-238.
- Granger, C.W.J. (1986) "Developments in the Study of Cointegrated Economic Variable", Oxford Bulletin of Economics and Statistics, Vol. 48, Issue 3: pp. 213-228.
- Gravetter, F. and Wallnau, L. (2004) *Statistics for the Behavioural Sciences*. 6th ed., Belmont, Calif, London: Wadsworth.
- Greenwood, J. and Jovanovic, B. (1990) "Financial Development, Growth and the Distribution of Income", *Journal of Political Economy*, Vol. 98, No. 5, Part 1: pp. 1076-1107.
- Greenwood, J. and Smith, B. (1997) "Financial Markets in Development and the Development of Financial Markets", *Journal of Economic Dynamic and Control*, Vol. 21, Issue 1: pp. 145-181.
- Gros, D. and Steinherr, A. (1995) Winds of Change Economic Transition in Central and Eastern Europe. Longman Group, UK Limited.
- Grosfied, I. and Hare, P. (1991) "Privatisation in Hungary, Poland and Czechoslovakia", *European Economy*, Special Edition, No. 2: pp.129-156.
- Grossman, G. and Helpman, E. (1991) "Quality Ladders in the Theory of Growth", *Review of Economic Study*, Vol. 58, Issue 2: pp. 43-61.
- Guitian, M. (1982) "Conditionality: Access to Fund Resources", *International Monetary Fund*, *World Bank*, Washington, D.C.
- Gupta, K.L. (1987) "Aggregate Savings, Financial Intermediation, and Interest Rate", *Review of Economics and Statistics*, Vol. 69, No. 2: pp. 303-311.
- Gurley, J.G. and Shaw, E.S. (1955) "Financial Aspects of Economic Development", *American Economic Review*, Vol. 45, No. 4: pp. 515-538.
- Gwartney, J. and Lawson, R. (2003) "The Concept and Measurement of Economic Freedom", *European Journal of Political Economy*, Vol. 19, Issue 3: pp. 405-430.
- Hair, J.F., Babin, B., Money, A. and Samouel, P. (2005) *Essentials of Business Research Methods*. USA: John Wiley and Sons.
- Hall, P.A. and Soskice, D. (2001) Varieties of Capitalism the Institutional Foundation of Comparative Advantage. Oxford: Oxford University Press.
- Handay, E.M. (1997) Securities and Capital Market. Alexandria: Monshat Almarf, Egypt (in Arabic).
- Handy, H. and Subramanian, A. (1997) "The Egyptian Stabilisation Experience", International Monetary Fund, Middle East Department, Washington, D.C.
- Harberger, A.C. (1987) "Foundations for Applied Welfare Economics: Some Open or Not-so-open Issues", *Mathematical Social Science*, Vol. 13, Issue 3: pp. 298-298.

- Hardaker, G. (1997) Technological and Organisational Integration for Time to Market of New Product Development: An Empirical Investigation. PhD. Unpublished Thesis, UK: Huddersfield University.
- Hargis, K. (2000) "International Cross-listing and Stock Market Development in Emerging Economies", *International Review of Economic and Finance*, Vol. 9, Issue 2: pp. 101-122.
- Harrise, R.G. (1997) "Stock Markets and Development: A Re-assessment", *European Economic Review*, Vol. 41, Issue 1: pp. 139-146.
- Harris, R.D.F. and Kucukozmen, C.C. (2001) "The Empirical Distribution of Stock Returns: Evidence from an Emerging European Market", *Applied Economics Letters*, Vol. 8, No. 6: pp. 367-371.
- Harris, S. (1987) "Current Issues in the World Sugar Economy", *Food Policy*, Vol. 12, Issue 2: pp. 127-145.
- Harrison, B. and Paton, D. (2004) "Transition, the Evolution of Stock Market Efficiency and Entry into EU: The Case of Romania", *Economic of Planning*, Vol. 37, No. 3-4: pp. 203-223.
- Harrison, P., Sussman, O. and Zeira, J. (1999) "Finance and Growth: Theory and New Evidence", *Federal Reserve Board, Working Paper*, Washington, D.C.
- Harrod, R. (1939) "An Essay in Dynamic Theory", *The Economic Journal*, Vol. 49, No. 139: pp. 14-33.
- Hartmann, M. and Khambata, D. (1993) "Emerging Stock Markets: Investment Strategy of the Future", *Columbia Journal of World Business*, 28-104 Summer.
- Harvey, C.R. and Ferson, W.E. (1993) "An Exploratory Investigation of the Fundamental Determinants of National Equity Market Returns", *NBER, Working Paper*, No. 4595.
- Harvey, C.R. and Ferson, W.E. (1994) "Sources of Risk and Expected Returns in Global Equity Markets", *NBER, Working Paper*, No. 4622.
- Harvey, C.R. (1994) "Conditional Asset Allocation in Emerging Markets", National Bureau of Economic Research, Working Paper, No. W4623.
- Harvey, C.R. (1995) "The Cross Section of Volatility and Autocorrelation in Emerging Stock Markets", *Financial Market and Portfolio Management*, Vol. 9: pp. 12-34.
- Hassan, K., Haque, M. and Lawrence, B. (2006) "An Empirical Analysis of Emerging Stock Markets of Europe", *Quarterly Journal of Business and Economics*, Vol. 45, No. 1-2: pp. 31-52.
- Healey, M.J. and Rawlinson, M.B. (1994) Interviewing Techniques in Business and Management Research. In: Wass, V.J. and Wells, P.E., eds. *Principles and Practice in Business and Management Research*. Aldershot, Dartmouth, pp. 123-169.
- Henry, P. (2000a) "Stock Market Liberalisation, Economic Reform, and Emerging Market Equity Prices", *Journal of Finance*, Vol. 55, No. 2: pp. 529-564.
- Henry, P. (2000b) "Do Stock Market Liberalisation Cause Investment Boom?", *Journal of Financial Economics*, Vol. 58, Issue 1-2: pp. 301-334.
- Heitman, G. (1969) "Analysis of the Oil Economy", *Journal of Modern African Studies*, Vol. 7, Issue 2: pp. 249-263.

- Hermes, N. and Lensink, R. (2001) "Capital Flight and the Uncertainty of Government Policies", *Economics Letters*, Vol. 71, Issue 3: pp. 377-381.
- Hicks, J. (1969) A Theory of Economic History. Oxford: Clarendon Press, UK.
- Higgott, R. (1998) "The Asian Economic Crisis: A Study of the Politics of Resentment", *New Political Economy*, Vol. 3, No. 3: pp. 333-356.
- Hoelscher, G. (1986) "New Evidence on Deficits and Interest Rates", *Journal of Money, Credit and Banking*, Vol. 18, Issue 1: pp. 1-17.
- Holden, P. and Rajapatirana, S. (1995) Unshackling the Private Sector: A Latin American Story, Directions in Development Series. World Bank, Washington, D.C.
- Holden, P. and Rajapatirana, S. (1997) *Emerging Economic Issues in Latin America: A Second Generation Agenda.* World Bank, Washington, D.C.
- Holmstrom, B. and Tirole, J. (1993) "Market Liquidity and Performance Monitoring", *Journal of Political Economy*, Vol. 101, Issue 4: pp. 678-709.
- Hopkins, W.G. (2001) *Qualitative vs. Quantitative Research Designs* [online] Available at: < http://www.sportscience5/sportsci.org/jour/0103/inbrief.htm> [Accessed 18 May 2007].
- Howe, K.R. (1985) "Two Dogmas of Educational Research", *Educational Researcher*, Vol. 14, No. 8: pp. 10-18.
- Hrneir, M. and Klacek, J. (1991) "Stabilisation Policies and Currency Convertibility in Czechoslovakia, Hungary and Poland", *European Economy*, Special Edition, No. 2: pp. 65-83.
- HRW Human Rights Watch (2005) *Libya: Reforms Welcome, But Concerns Remain* [online] Available at: < http://hrw.org/english/docs/2005/05/23/libya10983.htm > [Accessed 28 May 2008].
- Huang, Y. and Yang, Y. (1999) "How Important is APEC to China?", *Australian Economic Paper*, Vol. 38, Issue 3: pp. 328-342.
- Hussey, J. and Hussey, R. (2003) Business Research: A Practical Guide for Undergraduate and Postgraduate Students. 2nd ed., New York: Palgrave.
- IDB Inter-American Development Bank (1996) "Annual Report", World Bank, Washington, D.C.
- IEA International Energy Annual (2005) Energy Information Administration: Official Energy Statistics from the US Government [online] Available at: < http://www.eia.doe.gov> [Accessed 19 August 2007].
- IEF Index of Economic Freedom (2008) Index of Economic Freedom of the World Sector Average [online] Available at: < http://www.heritage.org/Index/country.cfm?Estonia> [Accessed 18 August 2008].
- IFC International Finance Corporation, Emerging Stock Markets Fact-book, Various Issues.
- IFC International Finance Cooperation (1997) *Emerging Stock Markets Fact-books*. World Bank, Washington, D.C.
- Igbikiowubo, H. (2003) *Libya's Blood Money Moves to Revive Oil Industry* [online] Available at: < http://www.globalpolicy.org/security/sanction/libya/2003/1014blood.htm> [Accessed 24 July 2008].

- IMF International Monetary Fund (2005) International Monetary Fund Country Report, No. 05/83, IMF, March, Washington, D.C.
- IMF International Monetary Fund (2006) World Economic Outlook: World Economic and Financial Survey, IMF, May, Washington, D.C.
- IMF International Monetary Fund (2007) World Economic Outlook: World Economic and Financial Survey, IMF, April, Washington, D.C.
- IMF International Monetary Fund (2007) IMF Country Report, No. 07/149, May, Washington, D.C.
- Intriligator, M., Braguinsky, S., Bowen II, J., Tullock, G. and Root, H. (1999) "Role of Market Institutions in Pacific Rim Development and Transition", *Contemporary Economic Policy*, Vol. 17, Issue 1: pp. 109-137.
- IPE International Private Enterprise (1992) *The Economic Reform Survey: Case Studies* [online] Available at: http://www.cipe.org/publications/ert/index.php [Accessed 8 November 2006].
- Jennergren, L. and Korsvold, P. (1975) "Price Formation in the Norwegian and Swedish Stock Markets: Some Random Walk Tests", *The Scandinavian Journal of Economics*, Vol. 76, No. 2: pp. 37-54.
- Jabnoun, N. and Al-Tamimi, H.A. (2003) "Measuring Perceived Service Quality at UAF Commercial Bank", International Journal of Quality and Reliability Management, Vol. 20, No. 4: pp. 458-472.
- Jaffe, J.F. and Mandelker, G. (1976) "The "Fisher Effect" for Risky Assets: An Empirical Investigation", *Journal of Finance*, Vol. 31, No. 2: pp. 447-458.
- Jamal, A. and Naser, K. (2002) "Customer Satisfaction and Retail Banking: An Assessment of Some of the Key Antecedents of Customer Satisfaction in Retail Bank", *International Journal of Bank Marketing*, Vol. 20, No. 4: pp. 146-160.
- James, C., Koreisha, S. and Partch, M. (1985) "A VARMA Analysis of the Causal Relations Among Stock Returns, Real Output, and Nominal Interest Rates", *Journal of Finance*, Vol. 40, No. 5: pp. 1375-1384.
- Javed, A.Y. and Ahmed, A. (1999) "The Response of Karachi Stock Exchange to Nuclear Detonation", *The Pakistan Development Review*, Vol. 38, Issue 4: pp. 777-786.
- Jayaratne, J. and Strahan, P. (1996) "The Finance-growth Nexus: Evidence from Bank Branch Deregulation", *Quarterly Journal of Economics*, Vol. 111, Issue 3: pp. 639-670.
- Jick, T.D. (1979) "Mixing Qualitative and Quantitative Methods: Triangulation in Action", *Administrative Science Quarterly*, December, Vol. 24, No. 4: pp. 602-11.
- Johnson, T.E. (1983) Investment Principles. New Jersey: Printice-Hall, Inc.
- Jurajda, S. and Terrell, K. (2000) "Optimal Speed of Transition: Micro Evidence from the Czech Republic", *William Davidson Institute, Working Papers Series*, No. 355.
- Kaminsky, G.L. and Reinhart, C.M. (1999) "The Twin Crises: The Causes of Banking and Balance-of-Payments Problems", *American Economic Review*, Vol. 89, No. 3: pp. 473-500.
- Kaminsky, G.L., Lizondo, S. and Reinhart, C.M. (1997) "Leading Indicators of Currency Crises", *Policy Research, Working Paper Series*, No. 1852.

- Kaplan, E. (2007)How Libya Got off the List [online] Available at: <http://www.cfr.org/publication/10855/how Libya list.html? got off the breadcrumb=%2Fissue%2F458%2Fstate sponsors of terrorism> [Accessed 3 May 2008].
- Karsten, S. (2005) "Reforms and Economic Growth in Transition Economies: Complementarily, Sequencing and Speed", *The European Journal of Comparative Economics*, Vol. 2, Issue 2: pp. 177-202.
- Kasman, S. (2003) *The Relationship between Exchange Rates and Stock Prices: A Causality Analysis* [online] Available at: http://www.sbe.deu.edu.tr/Yayinlar/kasman.pdf> [Accessed 18 January 2007].
- Kassimatis, K. (2002) "Financial Liberalisation and Stock Market Volatility in Selected Developing Countries", *Applied Financial Economics*, Vol. 12, Issue 6: pp. 389-394.
- Kaufmann, D., Kraay, A. and Mastruzzi, M. (2008) "Governance Matters VI: Aggregate and Individual Governance Indicators 1996-2006", World Bank Policy Research, Working Paper, 4280, July, World Bank, Washington, D.C.
- Kearney, I.M. (1998) "Is There A Stable Migration Equation for Ireland?", *Working Paper Series*, No. WP097.
- Kelemen, M. (2008) *Rice's Libya Trip A Signal of Restored Ties* [online] Available at: http://www.npr.org/templates/story/story.php [Accessed 8 September 2008].
- Keller, W. (1998) "Are International R&D Spillovers Trade-related?: Analyzing Spillovers Among Randomly Matched Trade Partners", *European Economic Review*, Vol. 42, Issue 8: pp. 1469-1481.
- Kellerhals, M.Jr. (2008) *Libya: Rice Makes Historic Visit* [online] Available at: http://allafrica.com/stories/200809090190.html /> [Accessed 9 September 2008].
- Kessler, G. (2008) *Rice Sees Gadhafi in Visit to Libya* [online] Available at: http://www.philly.com/philly/news/nation_world/ [Accessed 8 September 2008].
- Ketteni, E., Mamuneas, T., Savvides, A. and Stengos, T. (2007) "Is the Financial Development and Economic Growth Relationship Nonlinear?", *Journal Economics Bulleting*, Vol. 15, Issue 14: pp. 1-12.
- Keister, L.A. (2000) "Capital Structure in Transition: Financial Strategy in China's Emerging Economy", Working Paper, The Ohio State University.
- Kelegama, S. (1995) The Impact of Privatisation on Distributional Equity: The Case of Sri Lanka. In: Romanadham, V., ed. *Privatisation and Equity*. London: Rutledge.
- Kenny, C.J. and Moss, T.J. (1998) "Stock Market in Africa: Emerging Lions or White Elephants?", *World Development*, Vol. 26, No. 5: pp. 829-843.
- Khan, H. and Langmead, J. (1990) An Introduction to Modern Economy. London: Longman, UK.
- Khasnobis, B.G. and Bhaduri, S.N. (2000) "A Hallmark of India's New Economic Policy: Deregulation and Liberalisation of the Financial Sector", *Journal of Asian Economics*, Vol. 11, Issue 3: pp. 333-346.
- Khdduri, M. (1963) *Modern Libya: A Study in Political Development*. John Hopkins University Press, Baltimore.

- Kim, E.H. and Singal, V. (2000) "Stock Market Openings: Experience of Emerging Economies", *Journal of Business*, Vol. 73, Issue 1: pp. 25-66.
- King, L. (2003) "Explaining Procommunist Economic Performance", William Davidson, Working Paper, No. 559, May, pp. 1-71.
- King, N. (1998) Template Analysis. In: Symon, G. and Cassell, C., eds. *Qualitative Methods and Analysis in Organizational Research*. London: Sage.
- King, N. (2004) Using Interviews in Qualitative Research. In: Cassell, C. and Symon, G., eds. *Essential Guide to Qualitative Methods Organisational Research*. London: Sage, pp. 11-22.
- King, N. (2006) *Introduction to Qualitative Research*. University of Huddersfield: Primary Care Research Group.
- King, R.G. and Levine, R. (1993a) "Financial, Entrepreneurship and Growth: Theory and Evidence", *Journal of Monetary Economics*, Vol. 32, Issue 3: pp. 513-542.
- King, R.G. and Levine, R. (1993b) "Finance and Growth: Schumpeter Might be Right", *Quarterly Journal of Economics and Statistics*, Vol. 3, No. 3: pp.717-737.
- Kleidon, A.W. (1992) "Arbitrage, Nontrading, and Stale Prices: October 1987", *Journal of Business*, Vol. 65, Issue 4: pp. 483-507.
- Knell, M. and Rider, C. (1992) Socialist Economies in Transition: Appraisals of the Market Mechanism. England: Edward Elgar.
- Knight, M., Loayza, N. and Villanueva, D. (1993) "Testing the Neoclassical Theory of Economic Growth", *International Monetary Fund*, *Working Papers*, No. 92/106.
- Knox, P.L. and Masilela, C.O. (1990) "Role Orientations of Third World Urban Planners", *Environment and Planning B: Planning and Design*, Vol. 17, No. 1: pp. 9-22.
- Kobrin, S. (1982) Managing Political Risk Assessment: Strategic Response to Environmental Change. California. University of California Press.
- Konan, N.L. (2008) "The Effects of Interest Rates Volatility on Stock Returns and Volatility: Evidence from Korea", *International Research Journal of Finance and Economics*, ISSN 1450-2887, Issue 14: pp. 1-6.
- Koopmans, T. (1965) "On the Concept of Optimal Economic Growth", *Cowles Foundation Discussion Papers*, No. 163, pp. 226-300.
- Korajczyk, R.A. (1996) "A Measure of Stock Market Integration for Developed and Emerging Markets", *World Bank Economic Review*, Vol. 10, Issue 2: pp. 267-289.
- Kostadinova, P. (2004) *Initial Conditions and Economic Reform in Eastern Europe* [online] Available at: < http://plaza.ufl.edu/petiak/initialconditions.pdf> [Accessed 18 May 2007].
- Koutoulas, G. and Kryzanowski, L. (1996) "Macrofactor Conditional Volatilities, Time-varying Risk Premia and Stock Return Behaviour", *Financial Review*, Vol. 31, Issue 1: pp. 169-195.
- Krueger, A. (2005) "De Tocqueville's 'Dangerous Moment': The Importance of Getting Reform Right", *The World Economy*, Vol. 28, No. 6: pp. 749-763.
- Krueger, G. and Ciolko, M. (1998) "A Note of Initial Conditions and Liberalisation during Transition", *Journal of Comparative Economics*, Vol. 26, Issue 4: pp. 718-734.

- Krugman, R. (1987) "The Narrow Moving Band, the Dutch Disease, and the Consequences of Mrs. Thatcher: Notes on Trade in the Presence of Scale Economies", *Journal of Development Economic*, Vol. 27, Issue 1-2: pp. 41-55.
- Laban, R. and Sturzenegger, F. (1994) "Fiscal Conservatism as a Response to the Debt Crisis", *Journal* of Development Economies, Vol. 45, No. 2: pp. 305-324.
- Landesmann, M. (1991) "Industrial Restricting and the Reorientation of Trade Czechoslovakia", *European Economy*, Special Edition, No. 2: pp. 23-41.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A. and Vishny, R.W. (1997) "Legal Determinants of External Finance", *Journal of Finance*, Vol. 52, Issue 3: pp. 1131-1150.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A. and Vishny, R.W. (1998) "Law and Finance", *Journal* of *Political Economy*, Vol. 106, Issue 6: pp. 1113-1155.
- Lassar, W., Manolis, C. and Winsor, R.D. (2000) "Service Quality Perspectives and Satisfaction in Private Banking", *Journal of Services Marketing*, Vol. 14, No. 3: pp. 244-271.
- Lavigne, M. (1995) The Economic of Transition: From Socialist Economy to Market Economy. London: Macmillan Press, Ltd.
- Lavigne, M. (1999) *The Economics of Transition: From Socialist Economy to Market Economy*. 2nd ed., London: Houndsmills, Macmillan.
- Lee, B. (1992) "Causal Relations among Stock Returns, Interest Rates, Real Activity and Inflation", *The Journal of Finance*, Vol. 47, No. 4: pp. 1591-1630.
- Lee, L.F. (2007) "GMM and 2SLS Estimation of Mixed Regressive, Spatial Autoregressive Models", *Journal of Econometrics*, Vol. 137, Issue 2: pp. 489-514.
- Lee, K., Pesaran, M. and Smith, R. (1997) "Growth and Convergence in a Multi-country Empirical Stochastic Solow Model", *Journal of Applied Economics*, Vol. 12, No. 4: pp. 357-392.
- Leedy, P.D. and Ormrod, J.E. (2005) *Practical Research: Planning and Design*. 8th ed., New Jersey: Prentices Hall.
- Leite, S.P. and Vaez-Zadeh, R. (1986) "Credit Allocation and Investment Decisions: The Case of Manufacturing Sector in Korea", *World Development*, Vol.14, Issue 1: pp. 115-126.
- Levesque, T. and McDougall, G. (1996) "Determinants of Customer Satisfaction in Retail Banking", *International Journal of Bank Marketing*, Vol. 14, No. 7: pp. 12-20.
- Levine, R. (1990) "Financial Structure and Economic Development", *International Finance Discussion Papers*, No. 381.
- Levine, R. (1991) "Stock Market, Growth, and Tax Policy", Journal of Finance, Vol. 46, No. 4: pp. 1445-1465.
- Levine, R. (1992) "Financial Intermediary Services and Growth", *Journal of the Japanese and International Economies*, Vol. 6, Issue 4: pp. 383-405.
- Levine, R. (1995) "Stock Market Development and Long-run Growth", *World Bank Economic Review*, Vol. 82, No. 4: pp. 942-963.

- Levine, R. (1996) "Stock Markets: A Spur to Economic Growth", Article Based on 12 Papers Presented at a *World Bank* conference, "Stock Markets, Corporate Finance, and Economic Growth", *World Bank*, 1995, Washington D.C.
- Levine, R. (1997) "Financial Development and Economic Growth: Views and Agenda", *Journal of Economic Literature*, Vol. 35, No. 2: pp. 688-726.
- Levine, R. (1998) "The Legal Environment, Banks, and Long-run Economic Growth", *Journal Proceedings*, Vol. 30, No. 3, Part 2: pp. 596-620.
- Levine, R. (2002) "Bank-based or Market-based Financial System: Which is Better?", *Journal of Financial Intermediation*, Vol. 11, Issue 4: pp. 398-482.
- Levine, R. (2003) "More on Finance and Growth: More Finance, More Growth?", *The Federal Reserve Bank of St. Louise Review*, July-August, pp. 31-46.
- Levine, R. (2004) "Financial and Growth: Theory and Evidence", NBER, Working Papers, No. 10766.
- Levine, R. and Rentlt, D. (1992) "A Sensitivity Analysis of Cross-country Growth Regressions", *American Economic Review*, Vol. 82, No. 4: pp. 942-963.
- Levine, R. and Schmukler, S.L. (2006) "Internationalisation and Stock Market Liquidity", *Review of Finance*, Vol. 10, Issue 1: pp. 153-187.
- Levine, R. and Zervos, S. (1993) "What we have Learned about Policy and Growth from Cross-country Regressions", *American Economic Review*, Vol. 83, No. 2: pp. 926-430.
- Levine, R. and Zervos, S. (1996) "Stock Market Development and Long-run Growth", *World Bank Economic Review*, Vol. 10, Issue 2: pp. 323-339.
- Levine, R. and Zervos, S. (1998a) "Stock Market, Banks, and Economic Growth", *American Economic Review*, Vol. 88, No. 3: pp. 537-558.
- Levine, R. and Zervos, S. (1998b) "Capital Control Liberalisation and Stock Market Development", *World Development*, Vol. 22, Issue 7: pp. 1169-1184.
- Levine, R., Loayza, N. and Beck, T. (2000) "Financial Intermediation and Growth: Causality and Causes", *Journal of Monetary Economics*, Vol. 46, Issue 1: pp. 31-77.
- Libya Location Map (2008) *Maps of World* [online] Available at: http://www.mapsofworld.com/libya/libya-location-map.html [Accessed 18 October 2008].
- Libya Country Profile Economic Trade Issues (2007) [online] Available at: http://www.fita.org/countries/economie_92.html [Accessed 15 November 2007].
- Libya Energy Data, Statistics and Analysis (2007) "Country Analysis Briefs", *Energy Information Administration* [online] Available at: <www.eia.doe.gov> [Accessed 8 December 2007].
- Lim, K.P., Brooks, R.D. and Kim, J.H. (2007) "Financial Crisis and Stock Market Efficiency: Empirical Evidence from Asian Countries", *International Review of Financial Analysis*, Vol. 17, Issue 3: pp. 571-591.
- Linke, C.M. and Zumwalt, J.K. (1980) "More on Public Information and Stock Price", *Journal of Business Research*, Vol. 8, Issue 1: pp. 81-87.
- Lipton, D. and Sachs, J.D. (1990) "Creating A Market Economy in Eastern Europe: The Case of Poland", *Brookings Papers on Economic Activity*, Vol. 21, Issue 2: pp. 75-148.

- Lissowska, M. and Swaan, W. (1996) "Capabilities, Routines, and East European Economic Reform: Hungary and Poland before and after the 1989 Revolution", *Journal of Economic Issues*, Vol. 30, No. 4: pp. 1031-1057.
- Liu, L.Y. (1999) "Approximate Portfolio Analysis", *European Journal of Operational Research*, Vol. 119, Issue 1: pp.35-49.
- Liu, L.Y. (2002) "Reforming China's Urban Health Insurance System", *Health Policy*, Vol. 60, Issue 2: pp. 133-150.
- Lucas, R. (1993) "Making a Miracle", *Econometrica*, Vol. 61, No. 2: pp. 251-271.
- Lucas, R. (1988) "On the Mechanics of Economic Development", *Journal of Monetary Economics*, Vol. 22, No. 1: pp. 3-42.
- Mahmud, M.B. and Russel, A. (2002) "Evidence that the Terms of Petroleum Contracts Influence the Rate of Development of Oil Fields", *OPEC Review*, Vol. 26, Issue 1: pp. 21-43.
- Malhotra, N.K. and Birks, D.F. (2006) *Marketing Research: An Applied Approach*. 2nd ed., European: Prentice Hall, Harlow.
- Malkiel, B.G. (2003) "The Efficient Market Hypothesis and its Critics", *Journal of Economic Perspective*, Vol. 17, Issue 1: pp. 59-82.
- Malthus, T. (1798) An Essay on the Principle of Population. London: Electric Book Co, C 2001.
- Mandelker, G. and Tandon, K. (1985) "Common Stock Returns, Real Activity, Money and Inflation: Some International Evidence", *Journal of International Money and Finance*, Vol. 4, Issue 2: pp. 267-286.
- Mankiw, N., Romer, D. and Weil, D. (1992) "A Contribution to the Empirics of Economic Growth", *Quarterly Journal of Economics*, Vol. 107, No. 2: pp. 407-437.
- Mankiw, G., Phelps, E.S. and Romer, P.M. (1995) "The Growth of Nations", *Brookings Papers on Economic Activity*, Vol. 1995, No. 1, 25 Anniversary Issue, pp. 275-326.
- Marks, J. (1995) "Rapid Reform Paves the way for Investors", *Middle East Digest*, Vol. 39, No. 16: pp. 14-16.
- Marwan, T. and Ahmad, B. (2005) "What Happened to Savings during the Financial Crisis-a Dynamic Panel Analysis of Asian-5 Countries", *Economic Change and Restructuring*, Vol. 38, Issue 3: pp. 257-275.
- Massoud, B. (2004) "Discussed Privatisation and then to Where", Paper Presented at The Conference on the General People's Committee for Economy and Trade, 19-20 Jun, Economic Sciences Research Center, (in Arabic), Benghazi: Libya.
- Masoud, N. (1998) *The Role of Commercial Banks in the Creation Bank Credit In Libya*. MSc. Unpublished Thesis, Tripoli, Libya: Academic of Higher Studies and Economic Research.
- Matthews, C., Vasudevan, D., Barton, S. and Apana, R. (1994) "Capital Structure Decision Making in Privately Held Firms: Beyond the Finance Paradigm", *Family Business Review*, Vol. 7, No. 4: pp. 349-367.
- Myers, S. (2001) "Capital Structure", Journal of Economic Perspective, Vol. 15, Issue 2: pp. 81-102.

- McDougall, G. and Levesque, T. (2000) "Customer Satisfaction with Services: Putting Perceived Value into the Equation", *Journal of Services Marketing*, Vol. 14, No. 5: pp. 392-410.
- McKinnon, R. (1973) *Money and Capital in Economic Development*. Brookings Institution: World Bank, Washington, D.C.
- McKinnon, R. (1989) "Financial Liberalisation and Economic Development: A Reassessment of Interest-rate Policies in Asia and Latin America", Oxford Review of Economic Policy, Vol. 5, Issue 4: pp. 29-54.
- McLindon, M.P. (1996) Privatisation and Capital Market Development: Strategies to Promote Economic Growth. Westport: Greenwood Press.
- Megginson, W.L. (2005) "The Economic of Privatisation", *Journal of Banking and Finance*, Vol. 29, Issue 8-9: pp. 1931-1980.
- Megginson, W.L. and Netter, M.J. (2001) "From State to Market: A Survey of Empirical Studies on Privatisation", *Journal of Economic Literature*, Vol. 39, No. 1: pp. 321-389.
- Mendoza, R. (1998) Serial Autocorrelation: Why it's There, What it does, and How to Get Rid of it. Department of Economics, New York: Fordham University, Mimeo.
- Merlevede, B. (2003) "Reform Reversals and Output Growth in Transition Economics", *The Economic* of *Transition*, Vol. 11, Issue 4: pp. 649-669.
- Michael, K. (1993) "The O-ring Theory of Economic Development", *Quarterly Journal of Economics*, Vol. 108, Issue 3: pp. 551-575.
- Michaelas, N. (1998) *Financial Policy and Capital Structure Choice in UK Privately Held Companies*. PhD. Unpublished Thesis, UK: Manchester University Business School.
- Miller, M. (1991) Financial Innovations and Market Volatility. Cambridge: Blackwell.
- Mishkin, F.S. and Eakins, S.G. (2003) *Financial Markets in Institutions*. 4th ed., New York: Boston San Francisco.
- Modigliani, F. (1961) "Long-run Implications of Alternative Fiscal Policies and the Burden of the National Debt", *Economic Journal*, Vol. 71, No. 284: pp.730-755.
- Moguillansky, G. and Bielschowsky, R. (2001) *Investment and Economic Reform in Latin American*. New York: United Nations.
- Morales, M.F. (2003) "Financial Intermediation in a Model of Growth Through Creative Destruction", *Macroeconomic Dynamics*, Vol. 7: pp. 363-393.
- Morck, R., Yeung, B. and Yu, W. (2000) "The Information Content of Stock Markets: Why do Emerging Markets Have Synchronous Stock Price Movements?", *Journal of Financial Economics*, Vol. 58, No. 1: pp. 215-260.
- Mohamed, R. (1995) Public Enterprises. In: Jomo, K.S., ed. *Privatising Malaysia: Rents, Rhetoric, Realities.* Boulder, Colorado: Westview Press.
- Municipalities of Libya (2008) *From Wikipedia the free Encyclopaedia* [online] Available at: http://en.wikipedia.org/wiki/Municipalities_of_Libya Former_baladiyah> [Accessed 25 November 2008].

- Myers, M.D. (1999) *Investigating Information Systems with Ethnographic Research*. This is the PDF Version of a Paper Published in Communications of the AIS, pp. 231-243.
- Naceur, S.B., Ghazouani, S. and Omran, M. (2008) "Does Financial Market Liberalisation Spur Economic and Financial Development in the MENA Region?", *Journal of Comparative Economics*, Vol. 36, Issue 4: pp. 673-693.
- Nachmias, D. and Frankfort, N.C. (2007) *Research Methods in the Social Sciences*. 9th ed., Worth Publishers.
- Nankani, H. (1990) "Lessons of Privatisation in Developing Countries", *Finance and Development*, Vol. 27: pp. 43-45.
- Nelson, C.R. (1976) "Inflation and Rates of Return on Common Stock", *Journal of Finance*, Vol. 31, No. 2: pp. 471-483.
- Nove, A. (1994) "A Gap in Transition Models? A Comment on Gomulka", *Europe-Asia Studies*, Vol. 46, No. 5: pp. 863-869.
- Nunnally, J.C. (1978) Introduction to Psychometric Theory. 2nd ed., New York: McGraw-Hill.
- Nuti, D.M. (1999) 1989-1999: La Grande Transformation Dell'Europa Centro Orientale. In: Tridico, P., ed. *Institutional Change and Human Development in Transition Economics*. Paper to be Presented at the EAEPE Conference, 10-12 November 2005, Bremen, pp. 1-33.
- Omran, M. (2002) "Testing for a Significant Change in the Egyptian Economy under the Economic Reform Programme Era", *World Institute for Development Economics Research Discussion Paper*, Vol. 2002/59, WIDER, United Nation University.
- Omran, M. (2004) "The Performance of State-owned Enterprises and Newly Firms: Does Privatisation Really Matter?", *World Development*, Vol. 32, Issue 6: pp. 1019-1041.
- Omran, M. and Bolbol, A. (2003) "Foreign Direct Investment, Financial Development, and Economic Growth: Evidence from the Arab Countries", *REV*, *Middle East Economic Finance*, Vol. 1, No. 3: pp. 231-249.
- Omran, M. and Pointon, J. (2004) "The Determinants of Capital by Industry within an Emerging Economy: Evidence from Egypt", *International Journal of Business*, Vol. 9, No. 3: pp. 237-258.
- Onaran, Y. and Scinta, C. (2008) *Lehman Files Biggest Bankruptcy Case as Suitors Balk* [online] Available at: http://www.bloomberg.com/apps/news?pid=awh5hRyXkvs4 [Accessed 16 September 2008].
- Oppenheim, A.N. (2000) *Questionnaire Design, Interviewing and Attitude Measurement.* London: New ed. Continuum.
- Oshikoya, W. (1990) "Oil Exports, Agriculture and Growth Prospects of the Nigerian Economy: A Macroeconomic Analysis", *World Development*, Vol. 18, Issue 5: pp. 753-758.
- Ostry, J. and Reinhart, C. (1992) "Saving and Terms of Trade Shocks: Evidence from Developing Countries", *MPRA*, *Working Paper*, No. 6976.
- Oxford Reference Online (2006) *The New Oxford American Dictionary in English Dictionaries and Thesauruses* [online] Available at: http://www.communication.com (Accessed 29 August 2006].

- Pagano, M. (1993) "Financial Markets and Growth: An Overview", *European Economic Review*, Vol. 37, Issue 2-3: pp. 613-622.
- Pallant, J. (2003) SPSS Survival Manual. 3rd ed., Philadelphia: Open University Press.
- Palmer, M. (1971) "The Application of Psychological Testing to Entrepreneurial Potential", *California Management Review*, Vol. 13, Issue 3: pp. 32-38.
- Palmerino, M. (1999) "Take a Quality Approach to Qualitative Research", *Marketing News*, Vol. 33: pp. 35-36.
- Pardy, R. (1992) "Institutional Reform in Emerging Securities Markets", *The World Bank Policy Research, Working Papers*, WPS 907.
- Parente, S.L. and Prescoott, E.C. (1993) "Changes in the Wealth of Nations", *Federal Reserve Bank of Minneapolis Quarterly Review*, Vol. 17: pp. 3-16.
- Patrick, H. (1966) "Financial Development and Economic Development in Underdeveloped Countries", *Economic Development and Culture Change*, Vol. 14, No. 2: pp. 174-189.
- Patrick, H.T. and Wai, U. (1973) "Stock and Bond Issues and Capital Markets in Less Developed Countries", *International Monetary Fund*, *Staff Papers*, World Bank, Washington, D.C.
- Paul, K. (2004) Libya Disarmament: A Model for US Policy, Arms Control Today [online] Available at: http://www.accessmylibrary.com/coms2/summary_0286-3818035 [15 August 2007].
- Pawelka, P. and Boeckh, A. (2004) Patrimonial Capitalism: Economic Reform and Economic Order in the Arab World [online] Available at: http://www.deposit.ddb.de/cgibin [Accessed 14 January 2007].
- Phillips, P.C. (1986) "Understanding Spurious Regressions in Econometrics", *Journal of Econometrics*, Vol. 33, Issue 3: pp. 311-340.
- Phillips, P.C. and Perron, P. (1988) "Testing for a Unit Root in Time Series Regression", *Biometrica*, Vol. 75, No. 2: pp. 335-346.
- Philippe, B. (2007) WTO, the European-Mediterranean Agreement and the Unified Arab Market: A European Perspective [online] Available at: http://www.worldbank.org> [Accessed 12 March 2007].
- Polatoglu, V. and Ekin, S. (2001) An Empirical Investigation of Turkish Consumers", Acceptance of Interment Banking Service," *Journal of Services Marketing*, Vol. 19, No. 4: pp. 156-165.
- Porter, M.E. and Chairman, D.Y. (2006) "National Economic Strategy: An Assessment of the Competitiveness of the Libyan Arab Jamahiriya", *The General Planning Council of Libya*, CERA: Monitor Group, UK.
- Poshakwale, S. (1997) *Time Series Properties of Emerging and Developed Stock Markets: An Analysis and Comparison*. PhD. Unpublished Thesis, UK: Manchester University.
- Pratten, C. (1993) The Stock Market. Department of Applied Economics, University of Cambridge.
- Punch, P.K. (2005) Introduction to Social Science: Quantitative and Qualitative Approaches. 2nd ed., London: Sage.

- Puzzanghera, J. and Reynolds, M. (2009) Obama vows Tougher Financial Regulations while Pressing Nations on Stimulus [online] Available at: http://www.latimes.com/news [Accessed 14 March 2009].
- Peavy, J.W. (1992) "Returns on Initial Public Offerings of Closed-end Funds", *Review of Financial Studies*, Vol. 3, Issue 4: pp. 695-708.
- Quah, D. (1993) "Empirical Cross-section Dynamics in Economic growth", *European Economic Review*, Vol. 37, Issue 2-3: pp. 426-434.
- Rajan, R.G. (1995) "Insiders and Outsider: The Choice between Informed and Arms Length Debt", *Journal of Finance*, Vol. 47, Issue 4: pp. 1367-1400.
- Rajan, R.G. and Zingales, L. (1996) "Financial Dependence and Growth", *NBER*, *Working Paper*, No. 5758.
- Rajan, R.G. and Zingales, L. (1998) "Financial Development and Growth", American Economic Review, Vol. 88, Issue 3: pp. 559-586.
- Rapach, D.E. (2002) "The Long-run Relationship between Inflation and Real Stock Prices", Journal of Macroeconomic, Vol. 24, Issue 3: pp. 331-351.
- Rapport, R.N. (1970) "Three Dilemmas in Action Research", *Human Relations*, Vol. 23, No. 4: pp. 499-513.
- Ratner, M. (1993) "A Co-integration Test of the Impact of Foreign Exchange Rates on U.S. Stock Market Prices", *Global Finance Journal*, Vol. 4, Issue 2: pp. 93-101.
- Rebelo, S. (1991) "Long-run Policy Analysis and Long-run Growth", *Journal of Political Economy*, Vol. 99, No. 3: pp. 500-521.
- Remenyi, D., Williams, B., Money, A. and Swartz, E. (1998) *Doing Research in Business and Management-An Introduction to Process and Method*. London: Publications Ltd SAGE.
- Rioja, F. and Valev, N. (2004) "Does One Size Fit All?: A Re-examination of the Finance and Growth Relationship", *Journal of Development Economics*, Vol. 74, Issue 2: pp. 429-447.
- Ritter, L.S., Sliber, W. and Udell, G.F. (1997) *Principles of Money, Banking, and Financial Management*. New York: R.R. Donnelley and Sons Company.
- Robinson, J. (1952) *The Generalisation of General Theory in the Rate of Interest and Other Essays*. London: Macmillan.
- Robson, C. (2002) *Real World Research: A Resource for Social Scientists and Practitioners*. 2nd ed., Oxford: Blackwell Publishers.
- Rodrik, D. (1996) "Understanding Economic Policy Reform", *Journal of Economic Literature*, Vol. 34, Issue 1: pp. 9-41.
- Roger, N. (1999) *Recent Trends in Private Participation in Infrastructure* [online] Available at: http://www.transport-americas.org/RecentTrendsPPI.pdf> [Accessed 27 March 2007].
- Roll, R. (1989) "Industrial Structure and the Comparative Behaviour of International Stock Market Indices", *Journal of Finance*, Vol. 47, No. 1: pp. 3-41.

- Romer, D. (1993) Credit Channel or Credit Action? An Interpretation of the Post-war Transmission Mechanism. In: Wyoming, J., ed. *Changing Capital Markets: Implication for Monetary policy*. Federal Reserve Bank of Kansas City.
- Romer, M. (1986) "Increasing Returns and Long Run Growth", *Journal of Political Economy*, Vol. 94, No. 5: pp. 1002-1037.
- Romer, M. (1987) Crazy Explanations for the Productivity Slowdown. In: Fischer, S. and Blancherd, O.J., eds. *NBER, Macroeconomics Annual*. Cambridge, MA: MIT Press: pp. 163-202.
- Romer, M. (1990) "Endogenous Technical Change", *Journal of Political Economy*, Vol. 98, No. 5: pp. 571-5102.
- Rosenberg, M. (2009) *Population Growth Rates* [online] Available at: http://geography.about.com/od/populationgeography/a/populationgrow.htm> [Accessed 15 August 2009].
- Rousseau, P.L. and Sylla, R. (2001) "Financial Systems, Economic Growth and Globalisation", *NBER*, *Working Paper*, No. W8323.
- Rousseau, P.L. and Wachhtel, P. (1998) "Financial Intermediation and Economic Performance: Historical Evidence from Five Industrial Countries", *Journal of Money, Credit and Banking*, Vol. 30, Issue 4: pp. 657-678.
- Rousseau, P.L. and Wachhtel, P. (2000) "Equity Markets and Growth: Cross Country Evidence on Timing and Outcomes", *Journal of Banking and Finance*, Vol. 24, Issue 12: pp. 613-622.
- Rust, L. (1993) "Observations: Parents and Children Shopping Together: A New Approach to Qualitative Analysis of Observational Data", *Journal of Advertising Research*, Vol. 33, Issue 4: pp. 65-70.
- Rowntree, D. (1991) Statistics Without Tears, a Primer for Non-mathematicians. London: Harmondsworth, Penguin.
- Sachs, J. (1994) "Shock Therapy in Poland: Perspective of Five Years," *The Tanner Lectures on Human Values*, University of Utah, April, Vol. 6-7: pp. 1-26.
- Sader, F. (1995) "Privatising Public Enterprises and Foreign Investment in Developing Countries, 1988-1993", Foreign Investment Advisory Service Occasional Paper, No. 5, World Bank, Washington, D.C.
- Sala-i-Martin, X. (1995) "The Classical Approach to Convergence Analysis", CEPR, Discussion Papers, No. 1254.
- Sala-i-Martin, X. (1996) "The Classical Approach to Convergence Analysis", *Economic Journal*, Vol. 106, Issue 437: pp. 1019-1036.
- Sala-i-Martin, X. (1997) "I Just Ran Two Million Regressions", American Economic Review, Vol. 87, No. 2: pp. 178-183.
- Saleh, M. (2001) Accounting Information Disclosure and Accountability: Cases from Libya. PhD. Unpublished Thesis, UK: Sheffield Hallam University.
- Salehi-Isfahani, D. (2002) Population, Human Capital, and Economic Growth in Iran. In: Sirageldin, I., ed. *Human Capital: Population Economics in the Middle East*. London: I.B. Tauris and Cairo: American University of Cairo Press.

- Salehi-Isfahani, D. (2005) "Human Resources in Iran: Potentials and Challenges", *Iranian Studies*, Vol. 38, No. 1: pp. 117-147.
- Sarantakos, S. (1998) Social Research. London: MacMillan Press Ltd.
- Sarma, A. and Gupta, M. (2002) "A Decade of Fiscal Reforms in India", *International Studies Program, Working Paper Series*, No. 0204.
- Saunders, M., Lewis, P. and Adrian, T. (2007) *Research Methods for Business Students*. 4th ed., London: Prentice Hall.
- Schipke, A. and Taylor, A.M. (1994) The Economic of Transformation: Theory and Practice in the New Market Economies. Springer-Verlag.
- Schmollinger, C. (2008) *Oil is Little Changed After Falling as Investors Sell Contracts* [online] Available at: http://www.bloomberg.com [Accessed 6 August 2008].
- Schumpeter, A. (1912) *The Theory of Economic Development*. Cambridge: AM, Harved University Press.
- Schwert, G.W. (1990) "Stock Returns and Real Activity: A Century of Evidence", *Journal of Finance*, Vol. 45, No. 4: pp. 1237-1257.
- Sekaran, U. (2003) *Research Methods for Business: A Skill-building Approach*. 4th ed., John New York: Wiley and Sons.
- Seppala, J.I. (2000) "The Term Structure of Real Interest Rates: Theory and Evidence from the U.K. Index-linked Bonds", *Economic Society*, World Congress 2000 Contributed Papers, No. 0245.
- Shah, A. (1984) "Crowding out, Capital Accumulation, the Stock Market and Money-financed Fiscal Policy", *Journal of Money, Credit and Banking*, Vol. 16, Part1: pp. 461-473.
- Sharma, D. (2002) "The Differential Effect of Environmental Dimensionality, Size and Structure on Budget System Characteristics in Hotels", *Management Accounting Research*, Vol. 13, No. 1: pp. 101-130.
- Shaw, E. (1973) Financial Deepening in Economic Development. New York: Oxford University Press.
- Sherman, P. (2009) *Obama's Goals at G-20: Concerted Stimulus, Regulatory Reform* [online] Available at: http://capitabeat.com/?p=2288 [Accessed 14 March 2009].
- Shiller, R.J. and Beltratti, A.E. (1992) "Stock Prices and Bond Yields: Can Their Comovements be Explained in Terms of Present Value Models", *Journal of Monetary Economics*, Vol. 30, Issue 1: pp. 25-46.
- Siegel, S.N. and Castellan, N.J. (1988) *Nonparametric Statistics for the Behavioural Science*. 2nd ed., New York: McGraw-Hill.
- Singh, A. (1992a) "The Stock Market and Economic Development: Should Developing Countries Encourage Stock Market?", *Working Paper*, No. 49, University of Cambridge.
- Singh, A. (1992b) "An Importation Approach to the Structure of the Financial System", *United Nation Conformance on Trade and Development, Discussion Paper*, No. 46.
- Singh, A. (1993) "The Stock Market and Economic Development: Should Developing Countries Encourage Stock Markets?", UNCTAD Review, pp. 1-28.

- Singh, A. (1996) "Pension Reform, the Stock Market, Capital Formation and Economic Growth: A Critical Commentary on the World Bank's Proposals", *Centre for Economic Policy Analysis, Working Paper*, No. 2, New York.
- Singh, A. (1997) "Financial Liberalisation, Stock Markets and Economic Development", *Economic Journal*, Vol. 107: pp. 771-782.
- Singh, A. (1999) "Should Africa Promote Stock Market Capitalism?", *Journal of Informational Development*, Vol. 11, Issue 3: pp. 343-365.
- Singh, A. and Hamid, J. (1992) "Corporate Financial Structure in Developing Countries", *IMF World Bank Series Papers*, No. 1.
- Sinha, D. and Macri, J. (1999) "Financial Development and Economic Growth: The Case of Eight Asian Countries", *Macquarie Economic Research*, Paper No.7, Macquarie University, Sydney.
- Singh, A. and Weisse, B. (1998) "Emerging Stock Markets, Portfolio Capital Flows and long-term Economic Growth: Micro and Macroeconomic Perspectives", *Journal World Development*, Vol. 26, Issue 4: pp. 607-622.
- Sirri, R. and Tufano, P. (1995) The Economic of Pooling. In: Dwight, B., ed. The *Global Financial System: A Functional Perspective*. Boston: Harvard Business School Press.
- Smith, A. (1776) An Inquiry into the Nature and Causes of the Wealth of Nations. London: Methuen and Co, Ltd.
- Smith, G. (2004) Random Walks in Middle Eastern Stock Market. London: University of London, UK.
- Solnik, B. (1973) "Note on the Validity of the Random Walk for European Stock Prices", *Journal of Finance*, Vol. 2, Issue 5: pp. 1151-1159.
- Solow, R. (1956) "A Contribution to the Theory of Economic Growth", *Quarterly Journal of Economics*, Vol. 70, No. 1: pp. 65-94.
- Solow, R. (2000) Growth Theory an Exposition. 2nd ed., New York Oxford: Oxford University press.
- SPSS Statistical Package for the Social Sciences (2003) CD-ROM. US: SPSS Inc.
- Srinivasan, T.N. (2003) "Stanford Center for International Development", Indian Economic Reforms: A Stocktaking, *Working Paper*, No. 190: pp. 1-36.
- Staehr, K. (2006) "Reforms and Economic Growth in Transition Economics: Complementary, Sequencing and Speed", *The European Journal of Comparative Economics*, Vol. 2, Issue 2: pp. 177-202.
- St John, R. (2006) *Historical Dictionary of Libya*. 4th ed., the University of America: Scarecrow Press, Inc.
- St John, R. (2008) North Africa, Politics, Region, and the Limits of Transformation. Edited by Zoubir, H.Y. and Fernández, H., Foreword, A by Quandt, W.B, Routledge, Taylor and Francis Group.
- Stapley, N.F. (1986) *The Stock Market: A Guide for the Private Investors*. London: Woodhead-Faulkner in Association with Laing and Cruickshank Investment Management Services.
- Starbuck, W. (2004) "Methodological Challenges Posed by Measures of Performance", *Journal of Management and Governance*, Vol. 8, Issue 4: pp. 337-343.

- Stiglitz, J. (1985) "Credit Markets and the Control of Capital", *Journal of Money, Credit and Banking*, Vol. 17, No. 2: pp. 133-152.
- Stiglitz, J. (1987) "The Cause and Consequences of the Dependence of Quality on Price", *Journal of Economic Literature*, Vol. 25, No. 1: pp. 1-48.
- Stiglitz, J. (1989) "Financial Markets and Development", *Oxford Review of Economic Policy*, Vol. 5, Issue 4: pp. 55-68.
- Stiglitz, J. and Ellerman, D. (2001) "Not Poles Apart: 'Whither Reform?' and 'Whence Reform'", *Journal of Economic Policy Reform*, Vol. 4, Issue 4: pp. 325-338.
- Stock, J.H. (2001) "Macro-econometrics", Journal of Econometrics, Vol. 100, No. 1: pp. 29-32.
- Stone, D. (1990) "The Emerging Markets and Strategic Asset Allocation", *Working Paper*, Frankel Russell Co. Tacoma, Washington, D.C.
- Strauss, A. and Corbin, J. (1998) *Basic of Qualitative Research: Techniques and Procedures for Developing Grounded Theory*. 2nd ed., Sage: Publications, Inc.
- Stulz, R. (1997) "International Portfolio Flows and Securities Markets", *Working paper*, No. 99-3, Ohio State University.
- Stulz, R. (2000) "Financial Structure, Corporate Finance and Economic Growth", *International Review* of Finance, Vol. 1, Issue 1: pp. 11-38.
- Sturzeneqqer, F. and Tommasi, M. (1998) The Political Economy of Reform. UK: MIT Press.
- Subrahamanyam, A. and Titman, S. (1999) "The Going-public Decision and the Developments of Financial Markets", *Journal of Finance*, Vol. 54, Issue 3: pp. 1045-1082.
- Sudman, S. and Bradburn, N.M. (1983) Asking Questions: A Practical Guide to Questionnaire Design. Oxford: San Francisco, Jossey-Bass.
- Summers, R. and Heston, A. (1991) "The Penn World Table (Mark 5): An Expanded Set of International Comparisons, 1950-1988", *Quarterly Journal of Economics*, Vol. 106, No. 2: pp. 327-368.
- Svensson, L.E.O. (1994a) "Estimating and Interpreting Foreword Interest Rates: Sweden 1992-1994", *Working Series Paper*, No. 579.
- Svensson, L.E.O. (1994b) "Monetary Policy with Flexible Exchange Rates and Forward Interest Rates as Indicators," *CEPR*, *Discussion Papers*, No. 941.
- Swan, T. (1956) "Economic Growth and Capital Accumulation", *Economic Record*, Vol. 32, No. 63: pp. 334-361.
- Taylor, P. (2008) "Bankruptcy Filing Reveals it to be the Biggest in US History", *The Daily Telegraph*, Tuesday, 16 September 2008, p. 7.
- Tellis, W. (1997) *Application of a Case Study Methodology. The Qualitative Report* [online] Available at: http://www.nova.edu/ssss/QR/QR3-2/tellis1.html [Accessed 6 August 2009].
- Thirlwall, A.P. (2006) *Growth and Development with Special Reference to Developing Economies*. New York: Eighth-ed, Palgrave Macmillan.

- Ticehurst, G.W. and Veal, A.J. (2000) Business Research Methods: A Managerial Approach. Pearson Education: NSW, Australia.
- Tim, K. (2004) Gaddafi Talks of Olive Branches and Business Deals but the Old Threat of Violence Still Remains. London: The Independent.
- Time, B. (2004) French Airliner Victims Families get US\$ 170m from Libya [online] Available at: http://english.people.com.cn/200401/09/eng20040109_136.shtml> [Accessed 15 Jun 2008].
- Tobin, J. (1969) "A General Equilibrium Approach to Monetary Theory", *Journal of Money, Credit* and Banking, Vol. 1, No. 1: pp. 15-29.
- Todaro, M. and Smith, S. (2006) *Economic Development*. 9th ed., London: Harlow, Addison Wesley.
- Tridico, P. (2006) "Institutional Change and Human Development in Transition Economies", *Working Paper*, No.59, Washington, D.C.
- Tsuru, K. (2000) "Finance and Growth: Some Theoretical Considerations and a Review of the Empirical Literature", *Working Paper*, No. 228, Organisation for Economic Co-operation and Development (OECD), Prise.
- Tulba, A. and Fhaima, G. (2004) "Assessing the Privatisation Policy in Libyan Economy: Motivations, and Positive and Negative Influences", *The Privatisation in Libyan Economy*, Conference Hold on 29-30 December, *Centre of Economic Sciences*, (in Arabic), Benghazi: Libya.
- UNDP United Nations Development Programme (2001) *Human Development Report 2001*. Oxford: Oxford University Press.
- Van Maanen, J. (1983) Qualitative Methodology. London: Beverly Hills, UK.
- Villagomez, F.A. (1997) "Private Saving, Interest Rates and Liquidity Constraints in LDCs: Recent Evidence", *Applied Economics*, Vol. 29, Issue 5: pp. 607-615.
- Vittas, D. (1992) "The Role of Non-bank Financial Intermediaries in Egypt and other MENA Countries", *World Bank Policy Research, Working Paper*, No. 1892.
- Wargler, J. (2000) "Financial Markets and Allocation of Capital", *Journal of Financial Economics*, Vol. 58, Issue 1-2: pp. 187-214.
- Wass, V.J. and Wells, P.E. (1994) *Principles and Practice in Business and Management Research*. Dartmouth: Brookfield, UK.
- Weil, P. (1993) "Precautionary Savings and the Permanent Income Hypothesis", *Review of Economic Studies*, Vol. 60, Issue 2: pp. 367-383.
- West, S., Finch, J. and Curran, P. (1995) Structural Equation Models with Nonnormal Variables: Problems and Remedies. In: Hoyle, R., Ed. Structural Equation Modelling: Concepts, Issue and Applications. Thousand Oaks, CH: Sage, pp. 56-75.
- Weston, F. and Brigham, E. (1979) Managerial Finance. London: Hold Rinehart and Winston.
- WIDER World Institute for Development Economics Research (1990) "Foreign Portfolio Investment in Emerging Equity Markets", *World Institute for Development Economics Research of the United Nations University*, Study Group Series (WIDER).
- Wilcox, J.W. (1992) "Global Investing in Emerging Markets", *Financial Analysts Journal*, January-February, pp. 15-9.

- Williamson, J. (1994) "The Political Economy of Policy Reform", Institute for International Economics, World Bank, Washington, D.C.
- Williamson, J. (1999) "Economic Reform: Contents Progress, Prospects", Institute for International Economics, World Bank, Washington, D.C, pp. 1-17.
- Williamson, J. (2004) "A Short History of the Washington Consensus", Paper Commissioned by Foundation CIDOB for A Conference, 24-25 September, from the Washington Consensus Towards A New Global Governance, Barcelona.
- Williamson, J. and Institute, P. (1990) What Washington Means by Policy Reform. Speeches, Testimony, Papers.
- Wood, A. (2008) Asian Shares Fall Sharply as Crisis Spreads [online] Available at: http://www.ft.com/cms/s/0/57b73> [Accessed 16 September 2008].
- World Bank (1989) *Financial Systems and Development*. World Development Report. New York: Oxford University Press.
- World Bank (1993) *The East Asian Miracle: Economic Growth and Public Policy*. Oxford University Press, New York.
- World Bank (1994) "Philippines Private Sector Assessment", World Bank, Vol. 1, Washington, D.C.
- World Bank (1995a) World Development Report 1995. New York: Oxford University Press.
- World Bank (1995b) *Bureaucrats in Business: The Economics and Politics of Government Ownership.* New York: Oxford University Press.
- Wright, J. (1969) Libya. London: Ernest Benn Limited.
- Wurgler, J. (2000) "Financial Markets and the Allocation of Capital", *Journal of Financial Economics*, Vol. 58, Issue 1-2: pp. 187-214.
- Yavas, V., Benkenstein, M. and Stuhldreier, U. (2004) "Relationships between Service Quality and Behavioural Outcomes: A Study of Private Bank Customers in Germany", *International Journal of Bank Marketing*, Vol. 22, No. 2: pp. 144-157.
- Yazid, A. (2001) *Perceptions and Practices of Financial Risk Management in Malaysia*. PhD. Unpublished Thesis, UK: Glasgow Caledonian University.
- Yin, R. (1994) Case Study Research: Design and Methods. California: Sage, Publications, Inc.
- Yousef, T.M. (1998) "Demography, Capital Dependency, and Growth in MENA", *ERF*, *Working Paper* No.9801.
- Yusuf, S. and Peters, R.K. (1984) "Savings Behaviour and its Implications for Domestic Resource Mobilisation: The Case of Republic of Korea", World Bank Staff, Working Paper, No. 628, World Bank, Washington, D.C.
- Zhao, M. (2006) "External Liberalisation and the Evolution of China's Exchange System: An Empirical Approach", *World Bank China Office Research, Working Paper*, No. 4.
- Zhu, A., Ash, M. and Pollin, R. (2004) "Stock Market Liquidity and Economic Growth: A Critical Appraisal of the Levine/Zervos Model", *International Review of Applied Economics*, Vol. 18: pp. 63-71.

APPENDICES

Appendix A: General Information about Libya

This appendix is designed to provide the reader with the background of the country in which this study is conducted. Different issues will be highlighted in the following sections, including figures, tables and the rationale behind this discussion: a brief review of the main changes in the political and economical systems since the independence of country.

A.1 SUBSEQUENT STEPS

Following the 1969 revolution the fundamental structure of the Libyan economy altered. Several steps were taken by the revolutionary government in order to reform the existing economic situation, where the nation was transformed from a constitutional monarchy into a revolutionary republic. In the first year the Libyan government nationalised foreign banks and their local branches. Looking back to the early period (1963-1969), the government, at that juncture, required a change of the banking system through "libyanisation" and nationalisation of foreign banking operations in Libya, transforming them into Libyan banking corporations in which Libyans contributed not less than 51 *per cent* of their capital and participated in the management and direction of their activities in a manner serving the national economic welfare. The results of such efforts are shown in the following banks and corporations (Masoud, 1998):

- 1. Sahara Bank: established on 11 July 1964, replaced the Cecilia Bank. The distribution of its capital was:
 - 51 per cent Libyan, 29 per cent Bank of America and 20 per cent Bank of Cecelia.
- 2. Commercial Bank: newly established on 16 June 1964, with the following rates:
- 51 per cent Libyan and 49 per cent Central Bank of England
- 3. African Banking Corporation: established on 11 September 1964, instead of the Algerian Tunisian Real Estate Bank, with the following rates:
 - 51 *per cent* Libyan and 49 *per cent* the Algerian Tunisian Bank.
- 4. North Africa Bank: established on 1 January 1966, as a substitute for the British Bank for the Middle East, where its capital was distributed as:
 - 52 per cent Libyan, 35 per cent the British Bank for the Middle East, 10 per cent Morgan Bank
- 5. Bank of the Arab Evolution: established on 1 August 1967, replacing the Bank of Egypt, with its ownership:
 - 51 *per cent* Libyan and 49 *per cent* Bank of Egypt.

Until the end of this period (1963-1969), the trading bank system operated with five libyanised banks and four foreign banks which refused to undergo "libyanisation", despite the separation of their capitals and needs from their main headquarters. These were the Bank of Roma, Barclays, Napoli and the Arabian Bank. The second specialisation bank was established, known as the Libyan Real-Estate Industrial Bank, as per the law issued on 7 September 1965 which allocated one half of its capital to industrial investment and the other half to real estate capital. March 1977 witnessed the declaration of the Libyan Jamahiriya. This is a state of masses involving the transformative experiment of both regulation and executive powers of their own political and economic affairs to the people. Additionally, the new regime succeeded in altering oil prices and production per-capita income. Table A.3.3, presents some significant indicators within the Libyan economy during the period 1970-2006. In the early 1970s per-capita income was 2500 US\$, increasing sharply to 12,091.2 US\$ to reach the highest per-capita income in North Africa; then it decreased to 8,327.3 US\$ towards the end of 2006 due to fluctuation of oil prices. Likewise, GDP growth rate was estimated at 5 billion US\$ in 1970, increasing to 36.8 billion US\$ in 1980 and then to approximately 49.7 billion US\$ by the end of 2006. The oil sector contributed more than 40 *per cent* of GDP. During mid-1980, however, the Libyan economy was severely affected by the slump in oil revenues due to low barrel prices, with a decline in revenue from 6.9 billion US\$ in 1980 to about 5.9 billion US\$ in 1988.

Since 1961 oil revenues have remained the key source of income, providing the Libyan economy with most of its hard currency. In 1992 the economy was affected by UN and US sanctions. Consequently, oil exports and other goods declined to a value of 6.8 billion US\$ in 2000 and jumped billions US\$ to 38.2 by the end of 2006 due to surging world oil supply demand. Libya is determined to reduce its dependence on the oil sector by promoting the private sector and supporting small and medium-sized enterprises to investing in other economic sectors: natural

gas; agricultural; industry; tourism; fisheries. Therefore, as mentioned in previous discussion, to understand Libya's economy it is necessary to comprehend its revenue sources and the structure of the economy. The Libyan economy has additionally become more successful compared with oil-producing nations (see Table A.3.13). For instance, Libya's oil price increased in the 70s from 2.23 US\$ per barrel in 1969 to 25.28 US\$ per barrel in 1979 (CBL, 1979). Then during 2003-2008, the figure rose from 17.2 US\$ in 1999 to 86.0 US\$ in 2008 (Table A.3.4). Due to favourable world conditions, oil revenue to Libya in the external current account surplus rose sharply from 15.4 *per cent* of GDP in 2003 to 25 *per cent* in 2004, then to about 52 *per cent* in 2006. Accordingly, Libya's gross international reserves increased to 18.9 billion US\$ in 2003 and 24.6 billion US\$ in 2004.

Libya, as a member of the Organisation of Petroleum Exporting Countries (OPEC), holds the largest established oil reserves among African nations, followed by Nigeria and Algeria, where Libya had total proven oil reserves of 41.5 billion barrels at the end of January 2007, up from 39.1 billion barrels in 2006. Some 80 *per cent* of Libyan proven oil reserves are located in the Sirte basin area, which is responsible for about 90 *per cent* of the country's oil output. This attracted worldwide attention to the post-reform programme of the Libyan economy, as it was believed that increased economic development bore a significant influence on the economic future of the MENA regions and the world economic situation. Figure A 1.1 displays the five leading African proven oil reserves during 2007.





Source: Libya Energy Data, Statistics and Analysis (2007).

According to the IMF report (2007), Libya still faces key challenges in the medium term due to its strong financial position and the favourable global (environmental) conditions, which provide an opportunity to take additional forward steps to address certain challenges: containing the increase in spending and enhancing its quality; developing the financial sector and infrastructure; supporting management of oil savings to defend intergenerational equity; improving the domestic business climate; increasing the productive base in order to reduce oil dependence. All these factors aid the government to enjoy a fast-growing GDP, including the labour force. The subsequent main subsections explain evolution of the Libyan situation and how the political landscape altered and the economy improved: Subsection A.1.1 discusses the political scene and determinants of economic reform; Subsection A.1.2 concerns growth and structure of the Libyan economy. These issues are now discussed.

A.1.1 Political Determinants of Economic Reform

Col. Al-Qaddafi has ruled Libya since 1969, when he and a group of officers overthrew King Idris. Borrowing from the Islamic world, his regime adopted Arab Socialist ideas, which refused to establish a "*third way*" after communism and capitalism⁶⁴. In the early 1970s, the Libyan economy was driven by the Third Universal Theory'

⁶⁴ Col. Al-Qaddafi based the Third Universal theory on nationalism and religion, two forces he described as the paramount drives moving history and humankind. Nationalism was considered to be the natural result of the world's *racial* and *cultural* diversity and, therefore, a significant and productive force. For instance, Arab nationalism was considered to have especially high and glorious roots in the ancient past. Due to the Arab nation being the product of an age-old civilisation based on the divine and universal message of Islam (John, 2006), Qaddafi argued that it had the right, as a duty, to be the bearer of the TUT to the world.

(TUT) which is based on the "*Green Book*"⁶⁵. The original framework of this theory was written by Col. Al-Qaddafi. It opted for a socialist state and a state-planned Libyan economy for "socioeconomic transformation". Investment was essentially state-driven and trade and price controls, along with subsidies, were widespread. Economic performance was strictly constrained by stifling government interference in the economy and an unfavourable business climate (US Commercial Service, 2006). In 1973 Qaddafi proclaimed the "cultural" or "popular revolution" and introduced the new idea of "people's power", which includes the formation of "people's committees" in schools, universities, hospitals, workplaces and administrative districts.

Theoretically, the GPCs meetings became a forum for Libyans for all social and economic transformations, which is the area of economy and planning consultant to the peak based on the popular annual process of GPCs (conference) with superficial similarities to the western-style Congress or Parliament. The group consists of representatives of the staff of the Basic People's Congresses (BPC), which became the key administrative units, the secretaries of unions, associations and professional syndicates, the secretariat of the GPC and the Governor of the Central Bank of Libya (CBL). During the conference, usually held in March, the popular committees choose their secretariats which, in turn, appoint the committee "*trustees*", equivalent to the ministers in a western parliamentary system. Only the General People are the delegates who discuss issues relating to national security and foreign policy, the national budget and oil policy. The General People's Supreme Committee and Sebha were declared by President Al-Qaddafi as "the Libyan", a new Arab expression meaning "State Masses". Literature on the politics of economic reform, however, has usually focused on overcoming obstacles to launching reform programmes. The principal assumption is that any government attempting stabilisation and market-oriented economic reforms, whether democratic or not, must cope with political instability inherent in the nature of these measures. Both stabilisation and structural reform impose costs that are immediate, certain and often concentrated on specific solutions.

Since the removal of UN sanctions in 1999, Libya has been gradually implementing measures to reform and adopt fresh steps to liberalise its still largely socialist economy. It ceased supporting terrorism, bridged its relations with European countries and ended animosity with the US by settling the Lockerbie issue and paying compensation. It reached a point when it admitted to owning weapons of mass destruction and agreed to destroy them, in addition to stating that Israel represented no danger to its national security. However, it is only since the lifting of UN sanctions and all US Libya-specific trade sanctions in September 2003 and 2004, respectively, that the pace of reform has accelerated somewhat, with the implementation of measures aimed at enhancing the role of the private sector in the economy (IMF, 2005). On 28 June 2004 the US renewed direct diplomatic relations with Libya. On 20 September of the same year President Bush terminated the state of emergency against Libya by executive order. This action eliminated most of the economic sanctions imposed on Libya and led to the release of Libyan assets frozen in the US.

A.1.2 Growth and Structure of the Libyan Economy

Following the revolution in September 1969, growth and structure of the Libyan economy began in many areas: social; political; legal and economic. In developing the structure and reframing of the Libyan economy, the economic and political policy on various environmental and social factors has moved forward (see Table A.3.5). Fulfilment of society's needs for commodities and services, on the one hand, and finally reducing dependence on oil income and utilisation of natural and human resources, on the other, have characterised the political policy and economic development in Libya during three periods.

⁶⁵ The Green Book is one written by the Libyan leader, Col. Al-Qaddafi, first published in 1975, expounding his views on democracy and his political philosophy via the theory of TUT, called "Freedom, Social and Unity" following the Egyptian model influenced by the revolutionary ideas of the Egyptian leader Jamal Abdul Nasser. He explained that in his theory all previous theories undertake the economic problem either from the angle of ownership of any of the elements of production, or form that of wages for production. Therefore, to resolve the problem of production field due to the fact that they are based on a wage system. This system deprives workers of any right to the products being produced, whether a society or a private establishment.

APPENDICES

The First Period 1955-1969

This period is recognised as financial, to a large extent, starting in the mid-1950s when the oil companies began to enter Libyan territories for oil extraction. Oil was first discovered in 1958 and by 1962 export commenced, consisting of about 30 *per cent* of the local product in total. "*It's the main resource for the currency*" (Alfaitori, 2003: 178). An economic expert, Farley (1971), stated that the Libyan economy, prior to the discovery of oil, was in a backward state because there were no indications of economic growth existing at that time. This sector is indirectly relevant to other economic sectors such as education, health, electricity, the construction industry and agriculture, etc. The Libyan economy, however, following the 1969 revolution, has developed within the last three decades to adopt a socialist philosophy, which encourages public ownership and variant national income due to the independence of oil wealth. This, in a way, has enabled the country to switch to productive assets and to replace the achieved revenues from new resources instead of the oil revenues, continuously and regularly (El-Qumati, 1993). Nonetheless, laws were issued from the trade economy organisation. Regulations controlling trade and economic policy can be summarised thus:

- Law supervising finance issued in 1955
- Export and import law no.59 in 1957
- Law to regulate export and import no. 59 in 1957
- Treasury minister's decision no. 2 in 1962 regarding customs tariffs
- Supervision of foreign expenditure
- Quantity limitations on imports (import permits).

The Second Period 1970-1998

The plan for the *economic development programme* for 1996-2000 was based on economic and social transformations in the production and service sectors, aimed at providing goods and services that fulfilled the needs of the local market and lowered importation levels. Investment expenditure on industrial, agricultural and staple goods sectors reached 11 million LD in three decades (1970-2002), with 36 million US\$ spent, using an exchange rate of 3.3. Furthermore, the level of investment rate was about 1,574 million LD (CBL, 2002). Following a strategic policy to encourage exports, the Libyan government established a board for developing exports and marketing them in foreign outlets (i.e. presenting attractive prices, powerful advertisements and, sometimes, selling on credit). However, Libyan exports were subject to three basic shocks during the 1970s and 80s; the impact of said shocks was generally positive on the Libyan economy. Firstly, oil prices had increased sharply in 1973/74 and 1979/80 (price per barrel increased four times) due to the Arab-Israeli and Iran-Iraq wars (see Table A.3.6). The maximum oil production capacity was 332 million barrels per day by 1970 and the barrel price increased from 22.3 to 26.2 US\$ just one year later. The price of oil increased four times during the 1970-1973 period, sustaining economic growth and prosperity (Libya's revolution in year 30).

Secondly, during the early 1980s and late 1990s the political and oil crises led to the nationalisation of Libyan oil companies (see Table A.3.7). Fisher (1990) argued that, during the 1980s, the Libyan economy was deeply affected by the low price of oil. This, in turn, negatively affected the government's general budget along with the international economic depression and, most importantly, the US embargo of Libya. As the leading importer of its oil, US economic sanctions affected heavily upon the Libyan economy (OPEC). This situation hugely affected economic transformation plans. In the 1970s and early 1980s, for example, the amount designated for the *fifth plan* (1981-1985) was to be 185 million LD or the equivalent of 55 billion US\$; but it had to wait until 1981 so that the conditions could be met. Subsequently, an increase in public expenditure weakened the *transformation plans* to include projects of construction an increase of between 2,345 million LD in 1980 and 15,007 million LD in 1999 was required. Therefore, the government had to plug the financial gap between what was planned for and the actual amounts available for these projects (Faiadh and Almejairy, 2003). Table A.3.7 summarises the percentages of Libyan economic growth between 1962-2000.

The Third Period 1999-2007

This period is recognised as experiencing higher oil export revenues. Libya underwent strong economic growth during 2003, with real GDP estimated to have increased by around 2.7 and 3.8 *per cent*, up from 0.2 to 1.5 *per cent* growth in 2002 (see Table A.3.8). For 2004, a real GDP growth of 2.3 and 2.6 *per cent* was fore cast, with

consumer price inflation of 1.9 and 3.5 *per cent*. One study indicated that the increase or decrease in oil prices of 10 *per cent* caused an increase or decrease in the GDP of 3 *per cent* (UNDR, 1999).

Although by the late 1990s and the beginning of this century the Libyan economy had benefited significantly from the *open door policy* in terms of rate of growth, it had, at the same time, become heavily dependent on the oil sector, a source of national income and foreign exchange. Oil exports constituted more than 96 *per cent* of total exports, which represented the leading source of foreign currency. In addition to the high rate of economic vulnerability, the ratio of imports to GDP accounted for between 25 and 30 *per cent* and the ratio of trade to GDP increased by more than 58 *per cent* on average during the latter part of the period and reached 69.1 *per cent* in 2005, when Libya imported more than 70 *per cent* of its goods from overseas, as reported in Table A.3.9.

Regarding the comparison in terms of the performance of Africa's Ten Largest Economies (ATLEs). The higher growth of the ATLEs was attributable to the strong performance of five out of six North African countries belonging to this group: Algeria; Morocco; Tunisia; Libya and Sudan, as Table A.3.10 displays. These countries registered growth rates of over 5 *per cent* in 2004. In the remaining North African state, Egypt, the economy grew at 4.1 *per cent*. In 2006 Egypt increased by 6.12 *per cent*, whereas Algeria, probably more disproportionately affected by negative domestic developments, grew by only 3.6 *per cent*. Angola is estimated to have had a high growth rate of 18.6 *per cent* in 2006 compared to 11.2 *per cent* in 2004, as a result of cessation of its civil war in 2002. As already indicated, the challenges mentioned are beginning to be addressed. For instance, the pace of economic and structural reform in Libya has improved since the lifting of the UN and US trade sanctions when other developing countries' rates grew worldwide with an average annual growth rate of real GDP around 0.3 *per cent* since 1999 (see Chapter Five, Figure 5.5).

A.2 LIBYAN STOCK MARKET

As early mentioned in Chapter Two, with the introduction of an "open door policy" during 1999, the Libyan economy considered how economic reform programmes may benefit businesses and how they may be managed. They reactivated the private sector ownership and aimed to attract foreign investment including mobile domestic investment and allocation of resources. Additionally, the Libyan economy sought guidance from emerging countries, those which have undergone the experiment of financial markets. These issues are discussed subsequently.

A.2.1 What the Libyan Economy Can Learn from Emerging Countries

The inherent relationship between stock market and economic development of a country is well understood by Libyan economists. Over the last decade or so, many Middle Eastern and other emerging economies and developed countries worldwide have established the stock markets. These have rendered it easy for such countries to increase their own private sector investment within their own economies and economic infrastructures. Along with this investment, a knock-on-effect has increased trade and, indeed, established a solid base for technological trade in such nations. In 1996, however, the number of stock markets throughout the developing world continued to rise to constitute almost 60 *per cent* of the entire world stock markets with a total capital of some 1.9 trillion US\$. It is suggested that, in those countries, liquidity has increased even faster, and turnover leapt from 26 to 85 *per cent* of emerging market capitalisation between 1985 and 1999 (Levine, 2003). Latin America witnessed the greatest decline in market capitalisation (down 36 *per cent* post-July 1997) before recovering by mid-1999. Asia, too, declined 9.2 *per cent* in June 1997 to peak by the end of 1999. However, the Middle Eastern stock markets have a different efficiency from equity stock markets in developed countries. This is because many are smaller, some are new and others have been successfully closed to foreign investors. The whole Middle Eastern market accounted for approximately 0.9 *per cent* of world stock market capitalisation and 8.0 *per cent* of emerging market capitalisation by the end of 2001 (Smith, 2007).

Gentzoglanis (2007) tested four Arab nations' stock market performance and economic growth. He found that, the share performance is very weak in Egypt's stock market, particularly after the 11 September (2001) attacks and the reduction in world economic growth. Foreign brokers' transactions fell to about 6 *per cent* in March 2002 from a total 25 *per cent*, whilst the Hermes share index was at its lowest level for 8 years. Morocco was another example of a poorly performing stock exchange, attributable to a number of factors: the poor results of listed companies; government failure to float some capital of privatised companies on the national stock exchange; lack of investor interest in emerging markets. Further, in January 2002, the launch of new indexes failed to increase

interest in stocks and the stock market failed to renew. The key share index (IGB) tumbled about 7.4 *per cent*, after a 15.3 *per cent* decline in 2000, with 13 *per cent* fewer transactions of 60 *per cent* less by 2000. Similarly, in Tunisia stock exchange activity was very weak and limited as the issues accounted for 90 *per cent* of the primary market. There was a lack of foreign investor interest in the financial market. Differently, Jordan's "Amman Financial Market" is one of the most open to foreign investors and the fastest growing stock market in the region with a market capitalisation close to 5 billion US\$, providing the ratio of a market capitalisation to GDP of about 77 *per cent*. The market capitalisation growth rate has risen by 158 *per cent* during the last five years. Jordan, however, introduced a modern Securities' Law to create a regulatory body by separating the regulatory function from the technical side of the stock market. This led to an increase in investors' confidence to invest. Also, the flow of information among the market institutions, investors and participants was made transparent and the creation of sophisticated and administrative functions within market institutions helped Jordan's financial stock market to a regional boost. Likewise, Jordan's financial market is well placed to be more competitive, as the government created four other bodies, i.e. the Jordan Stock Exchange (JSE), Jordan Stock Depository (JSD) and Jordan Securities Exchange industry to represent the private sector (JSEC) besides providing proper training in securities' dealing.

A.2.2 Characteristics of the Libyan Stock Market

The Libyan stock market consists of a broad range of participants which include companies, individual investors, institutional investors, financial instructors and dealers with assigned responsibilities. Trading in the market is based on a continuous trading system. Each company has an assigned board on the trading office and all trades have to be executed *vis a vis* a licensed broker in the office. By mid-2008 there were 15 brokerage firms licensed by the Libyan stock market to trade in the market. They can also buy and sell securities on behalf of their customers against commission and occasionally for their own account. While their legal unit and nature takes the form of a limited liability company the two may comprise a partnership functioning solely as a broker. Although investors in the Libyan stock market, particularly in the secondary market, are of assorted groups, the breakdown of various types of investors is not recognised. There is no available information on the percentage of individual investors as opposed to institutional investors; therefore, this type of information is vital. Clearly, the primary institutional investors in Libya consist of insurance companies, banking institutions, the Libyan Investment Corporation (government agency), saving funds, and different employee provident and investment funds. Whilst the first two institutions have been mainly net buyers in the market, they do not participate frequently on the selling side.

A.2.2.1 The Market Trading System

As mentioned in previous discussion (Chapter Five) the security traded on the Libyan stock market is divided into two groups: equities and debt instrument. The buying and selling of orders is handled by licensed brokers and implemented on a continuous action basis during trading hours. The passing of customer orders by telephone or verbally, therefore, is acceptable as long as such orders are subsequently confirmed either via order forms accordingly completed and signed by customers, or by a corresponding statement transmitted in written form. Trading for the officer occurs in units, each worth LD 50 of nominal value of the shares. Most shares have a nominal value of LD 1, with the exception of shares whose average ranged between 8 to 10 LD, such as the Wahda Bank, Tripoli Regional Bank, Cement Regional Company, Suq Al-Gumaa Regional Bank and United Company for Insurance with a nominal value of LD 10. Therefore, among brokers take place a spot basis at (T+1); the prices of shares are quoted in LD which increases with the greatest change of 5 per cent on the previous day's close compulsory on the daily movement of a share. Consequently, a ceiling is applied to prevent large price fluctuations to eliminate unnecessary speculation and to protect the interest of small investors. The exception to this rule occurs only when there is a stock dividend and a price adjustment is required or when a company is originally listed in which case its price is floated for 15 minutes or so and the base price is then set for that day. Evidently, the Libyan stock market provides an up-to-date record of all transactions and all traded shares are registered. The registration might occur on the day following the trading day, when the brokers record the transaction with the Libyan stock market. The latter is given two subsequent business days to update its records and to deliver all transactions (tighten) to the issuing companies who should resultingly update their records within three or so business days.
A.2.2.2 Available Market Information

As is well known, in any particular investment business area it is significant to bear all available information databases. The investor will have to consider factors such as company reports, statement sheets, government deregulation policy, economic growth and many other aspects besides. In Libya, however, the source of information available about investment and business in company shares can be divided into four groups: online databases available, company reports, stock market publications and brokers' research.

A.2.2.2.1 Online Databases Available

On 17 February 2008, the Libyan stock market started implementation of the new Electronic Trading System under the name "*Libyan Stock Market*" within a website access "*www.lsm.gov.ly*". This new system allowed market deals to be more easily processed and helped to provide information that is more accurate. The Libyan market relies on this electronic testing system which includes five brokerage firms via the request of five agents to record their statements and book the assets. In addition, they review the mechanisms of work of conservation and how to register requests for proposal for all brokerage firms and accounting market general index, which begin by 1000 points besides controlling commerce programmes. This is considered a qualitative leap for the market in order to provide more information transparency and safety for both investors and traders by entering all the orders into the computer system, matching supply and demand for securities electronically setting the network and applying prices. Since it started providing information, the market has experienced growth in a number of transactions. For instance, the number of listed companies has increased from 3 in 2006 to more than 7 in 2008. Furthermore, market capitalisation increased from 254,543,455 million LD in mid-2007 to approximately 1,021,115,386.730 million LD in mid-2008.

A.2.2.2.2 Company Reports

Although the most truthful and direct sources of information are created by company reports, all the listed companies are required by law to publish their annual reports during the first four months following the end of the financial year. These include a profit and loss account and a balance sheet. In the former, charges for depreciation interest on loans and overdrafts, investment income, charges in corporate tax, proposed and paid dividends, pension and compensation, auditors' remuneration and turnover are listed. One of the significant conditions of admission into the listed stock market is that the company should be prepared to provide shareholders with sufficient information for its assessment. To achieve this, companies are required to enter into a general agreement with the stock market for the provision of information availability. One of the provisions is to prepare a half-yearly report to be sent to shareholders. This should include statements of profit and loss, and comparative figures for the matching period should be provided. Supplementary information may also be contained in the Chairman's Report. From a browse *vis a vis* annual reports, as the key sections include notice of the annual general meeting, List of Directors, Chairman's Review, Secretary and Auditors, Director's Report, Report of the Auditors and the Accounts. The balance sheet includes terms such as loans, overdrafts and details of share capital, current and other assets including listed and unlisted investments, loans to directors, details of valuation of certain assets and any alteration in the company's assets.

A.2.2.2.3 Stock Market Publications

The Libyan stock market has been active in providing significant information about the listed companies.

1 Annual Report

The Libyan stock market provides a published annual report on the year's market activities. For instance, it notes the number of transactions made, the number of shares traded and their market value. New issues offered by companies, companies' authorised capital and other similar information are included. The report additionally includes all the licensed brokers, the market achievements and the chairman's view regarding future plans.

2 Monthly Statistical Bulletin

A monthly statistical bulletin is published by the national stock market, containing valuable data and the financial ratios of listed companies. The bulletin also includes sectoral data, cumulative market data and individual company data.

3 Companies' Guide

The Libyan stock market publishes the national shareholding companies' guide on an annual basis in order to provide interested parties with a reference that contains significant information needed by shareholders. The guide gives valuable data and the financial ratios of listed companies, it additionally features the number of shareholders, ownership ratios, the number of employees in each financial limit of the economy and their balance sheets and profit and loss statement for the past three to five years. Furthermore the results and regulations of the market, bank, company and insurance laws are published.

4 Daily Official List

The traded share prices are quoted on a daily and weekly basis via the local Arabic and English newspapers, reporting the total number of traded shares, their market value and the number of transactions made, the closing prices and the nominal value of quoted shares. Therefore, a short daily price quotation is published in the Libyan media (television and radio). The closing prices of Libyan shares listed and traded are quoted and transmitted by the network worldwide. Likewise, private ownership companies have begun participating in the disclosure process by facilitating access to supply more market information by computer network and telecommunication services.

A.2.2.2.4 Brokers' Research

The brokers in the Libyan stock market still undergo research on the listed companies for several reasons, largely, the small markets. However, some brokerage firms have started conducting activities in the market, which also maintain a database on prices and the company outcome. This information is usually supplied to key investors when required. There are also several international institutions that provide this kind of information about the market and companies in the stock market. For instance, IMF and World Bank databases cover the Libyan situation in their emerging market reports, in addition to MENA and AMF which publish databases covering Libya in their literature.

A.2.3 Agreement for Joint Co-operation

The agreement for joint co-operation partnerships is threefold: agreement with Arab Society of Certified Accountants; London Stock Exchange and University of Reading. These three issues are discussed next.

Agreement with Arab Society of Certified Accountants

Jordan signed an agreement in October 2007 for joint co-operation between the Libyan stock market and the Arab Society of Certified Accountants (ASCA). Current policy of the convention within the Libyan stock market is rehabilitation and development of employee skills in accounting and auditing and enrichment of their knowledge by examining and reviewing updated methods of international accounting. ASCA publicity for the Libyan stock market is available online (network) under the framework of scientific programmes agreed upon.

• Agreement with London Stock Exchange

On 15 October 2007 an agreement was signed in London for joint co-operation between the Libyan stock market and the London Stock Exchange in the area of training and skills development. The convention benefits from the lengthy LSE experience in this area which is used for the rehabilitation of Libyan stock market cadres to perform their duties to the fullest extent in addition to reviewing the rules and regulations in force in the Libyan stock market and developing the commensurate changes required by each stage. The convention additionally stipulates the need for participation of the Libyan side in each symposia and conference held under the auspices of the LSE.

APPENDICES

• Agreement with the University of Reading

The Libyan stock market signed a joint co-operation agreement on 17th October 2007 with the University of Reading in the area of securities' trading in securities. Under this agreement technical assistance and training will be provided in both Libya and Britain to market intermediaries and workers on the disclosure of financial data, arbitration and dispute resolution, governance and the preparation of legal rules and regulations to enhance the performance of censorship and protect users. In addition, it will provide consultancy for the development of the financial markets.

A.3 The Country's Economic Statistics

	Tabl	e A 3.1]	Populati	ion, Lal	oour Fo	rce and	l Fertili	ty				
		Populatio	on Growth	1	Gro	wth of La	abour		Fer	tility Rate	, Total	
Connectories	(Annual %)			Force (Annual %)				(Biı	ths per F	emale)		
Countries	1960	1997	1975	2005	1975	1995	1996	1000	2002	1970	1995	2000
	1988	2002	2005	2015	1984	2002	2005	1900	2002	1975	2002	2005
Algeria	2.9	1.6	2.4	1.5	3.2	3.4	15.3	6.0	2.8	7.4	3.8	2.5
Egypt	2.5	1.9	2.1	1.7	2.2	3.0	11.0	4.8	3.1	5.9	3.4	3.2
Iran	3.5	1.2	2.4	1.3	3.0	2.7	11.5	5.6	2.0	6.4	2.8	2.1
Iraq	3.5	2.1	NA	NA	2.8	3.0	1.1	6.3	4.1	NA	5.3	NA
Jordan	3.1	3.0	3.5	2.2	2.1	4.1	NA	7.2	3.5	7.8	4.9	3.5
Kuwait	7.2	3.4	3.3	2.2	7.5	5.7	1.1	4.8	2.5	6.9	2.9	2.3
Lebanon	1.5	1.4	1.3	1.0	-0.1	2.7	NA	3.3	2.2	4.8	2.7	2.3
Libya	4.2	2.0	2.9	1.9	4.3	2.1	NA	6.8	3.3	7.6	3.8	3.0
Morocco	2.6	1.7	1.9	1.2	3.2	2.5	11.0	4.8	2.8	6.9	3.1	2.5
Mauritania	2.4	1.6	2.7	2.4	NA	2.0	NA	6.5	3.2	6.6	5.5	4.8
Oman	3.6	2.6	3.4	2.0	5.7	2.3	NA	7.2	4.0	7.2	5.9	3.7
Saudi Arabia	4.3	2.7	3.9	2.1	5.1	2.9	5.2	7.2	5.3	7.3	5.8	3.8
Syria	3.4	2.5	3.1	2.2	3.1	4.1	11.7	6.7	3.4	7.5	4.0	3.5
Tunisia	2.2	1.2	1.9	1.0	3.7	2.5	14.2	4.0	2.1	6.2	2.6	2.0
Turkey	2.4	1.7	1.9	1.2	1.4	2.5	9.9	3.5	2.2	5.3	2.5	2.2
UAE	10.5	4.9	6.8	2.5	21.8	4.4	2.3	4.8	3.0	6.4	3.4	2.5
Yemen	2.4	2.9	3.6	2.9	1.7	3.0	11.5	7.0	6.0	8.7	7.6	6.0
MENA	2.7	1.9	2.6	1.9	2.8	2.9	NA	3.5	3.1	6.7	4.1	3.6
East Asia and Pacific	2.0	1.0	1.3	0.7	2.3	1.2	NA	3.6	2.1	5.0	1.8	1.9
South Asia	2.4	1.8	2.1	1.5	2.3	2.4	NA	2.7	3.2	5.5	3.3	3.2
Sub-Saharan Africa	2.8	2.4	2.8	2.3	2.6	2.6		5.2	5.1	6.8	5.5	5.5
Latin America and the Caribbean	2.5	1.5	1.8	1.2	3.2	2.2	NA	6.0	2.5	5.5	2.7	2.5
Developing countries	2.3	2.0	1.9	1.3	3.1	2.8	NA	1.9	2.3	5.4	3.0	2.9
World	1.8	1.1	1.6	1.1	2.7	2.1	NA	2.6	2.1	4.5	2.7	2.6

Notice a: NA: Not Applicable.

Source: Human Development Report (1990, 2000, 2006 and 2007/2008).

Table A 3.2 The Libyar	n Economy by	Sectors, 1958-1969
------------------------	--------------	--------------------

Economia Sactors	1958	3	1062 1060 8
Economic Sectors	Value (million LD)	Share (per cent)	1902-1909
GDP	52.2	100	22.6
Agriculture, Forestry and Fishing	13.6	26.1	4.1
Mining and Quarrying (Inc. oil)	3.6	6.9	46.7
Manufacturing	6	11.5	8.4
Construction	1.8	3.4	19.9
Transport and Communication	2.9	5.6	16.6
Wholesale and Retail Trade	7.3	14	12.7
Services Gathering	7.5	14.3	13.4
Banking, Insurance, Dwelling and Social Services	9.5	18.2	NA
Export of Goods and Service	NA ^b	NA	31.7
Import of Goods and Service	NA	NA	10.3
Private Consumption	NA	NA	6.7
Government Consumption	NA	NA	21.6

Notice a: Average annual growth rate.

Notice b: NA= Not Applicable.

Source: The Economic development of Libya (1960); CBL, statistically monthly bulletins various issues.

Economic Sectors	Units	Scale	1970	1980	1990	2000	2006
GDP	US\$	Billions	5.0	36.8	28.9	36.1	49.7
GDP per-capita	US\$	Units	2500	12,091.2	6,674.3	6,807.8	8,327.3
Population	Persons	Millions	2.0	3.040	4.334	5.306	5.970
Labour Force	Persons	Thousands	400	1000	1000	1200	2000
Oil Revenue	US\$	Billions	2.7	6.9	6.8	7.2	9.8
Imports of Goods and Services	US\$	Billions	0.8	3.7	3.9	4.7	12.9
Exports of Goods and Services	US\$	Billions	2.7	5.4	6.8	13.5	38.2
Current Account Balance	US\$	Billions	0.9	8.2	2.1	11.4	25.6
Trade Balance	LD	Millions	589.2	1616.7	2234.0	3310.1	28401.6

Table A 3.3 Libyan Economy Indicators, 1970-2006

Source: Economic Indicators of the Libyan Economy 1970-1982; IMF (2007) and CBL, Various Issues.

Table A	3.4 Libyan	Crude Oil	Production	and Prices	during	Reform P	rogramme
							0

Description	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008
Crude oil production (millions of barrels per day)	1.32	1.35	1.32	1.20	1.53	1.62	1.69	1.72	1.80	2.00
Libyan crude oil price (US\$/bbl)	17.2	27.0	23.3	24.4	28.2	36.9	51.9	62.5	69.3	86.0
Source: IMF (2005, 2007).										

Libya's Past	Libya's Emerging Future
Significant isolation from the outside world	Increasing integration with the outside world
Focus on equality of living conditions and social standards	Greater opportunities for individual achievement and involvement in the productive sector
Oil revenues as the main source of national prosperity	Oil revenues supplemented by wealth created in other parts of the economy
Government's central focus on the distribution of oil revenues to address social needs	Government increasingly working with the private sector to enable the creation of wealth in competitive markets

Table A 3.5 Libya Entering a New Phase in its Development

Source: Libyan data base (2006).

Table A 3.6 The Main Political Factors Affecting Oil Production, 1973-1998

Year	Past Political Factors	Results
1973	Arab-Israeli war	Increased oil prices
1979	Iranian crisis	Increased oil prices
1980s	• Different measures taken by industrialised countries to reduce their dependence on OPEC oil	• Reduced oil production (oil exports and oil prices)
	OPEC operation to control prices, oil production and oil quantities	Reduced oil production and oil exports
1982- 1988	War between Libya and Chad	Reduced oil production and oil exports
1986	US ban on importing oil from Libya	Reduced oil exports
1992- 1998	UN sanctions on Libyan economy	Reduced oil production and oil exports

Source: Developed for this research from Subsection A.2.2: Growth and Structure of the Libyan Economy.

Economic Sectors	1970-1982	1983-1990	1962-1990	1991-2000
GDP	13.2	-2.5	11.8	8.9
Agriculture, Forestry and Fishing	4.6	11.8	6.70	9.8
Mining and Quarrying (Inc. oil)	14.3	-7.8	17.8	22.8
Manufacturing	18.6	7.0	13.0	5.9
Construction	15.3	-1.8	12.0	5.4
Transport and Communication	18.1	-0.4	13.1	8.7
Wholesale and Retail Trade	17	-2.1	10.9	9.2
Services Gathering	14.0	2.7	11.1	10.2
Export of Goods and Service	12.4	-0.4	14.7	11.7
Imports of Goods and Service	18.4	-7.1	9.5	9.2
Private Consumption	14.1	-3.3	7.9	6.8
Government Consumption	16.8	-2.6	13.2	8.2

Table A 3.7 Average Annual Growth Rates of the Libyan Economy by Sectors, 1962-2000

Source: Calculated from CBL, statistically series and monthly bulletins, various issues.

Table A 3.8 The Main Political Factors Affecting Oil Production, 1999-2007

Year	Past Political Factors	Results
1999- 2003	The lifting of UN sanctions	Increased oil prices
2003-	• Iraq War and terrorist acts against oil infrastructure in Saudi Arabia.	
2007	• Increased demand for oil from countries such as China and the US and weakening of the US\$ against other main currencies.	
	• Speculation and lack of supplies, within prosperity speculation in energy markets.	Increased oil prices
	• Nuclear weapons' crisis between Iran (the 4 th largest world oil exporter) and Western countries. Also, the retreat in crude from Nigeria, (the 8 th largest source country) since February 2006 due to armed attacks on the oil industry, plus non-restoration of Iraq's previous production capacity.	

Source: Developed for this research from Subsection A.2.2: Growth and Structure of the Libyan Economy.

Table A 3.	9 The I	Rate of	Economic	Open E	Joor Policy	, 1999-2005
------------	---------	---------	----------	--------	--------------------	-------------

(Million LD)				•			
Economic Sector Years	1999	2000	2001	2002	2003	2004	2005
Exports and Re-exports	36,822	52,215	53,940	101,770	148,066	208,483	311,480
Imports	19,286	19,114	26,604	55,857	55,979	82,552	79,535
Total of Trade	56,108	71,329	80,544	157,627	204,045	291,035	391,015.0
Total of GDP	140,752	177,756	180,791	259,141	317,318	414,862	560,252
Rate of Opening Policy (%) ^a	39.9	40.1	44.6	60.8	64.3	70.2	69.8
Notice a: Researcher's calcu	lation						

Source: CBL (2006, 2007).

Eco. Sectors	GDP At C (bil	Current US\$ llions)	Populatior	n (millions)	GDP Rate (p	Growth per cent)	Country Total Afri	Weight in can GDP
Country	2004	2006	2004	2006	2004	2006	2004	2006
South Africa	216,772	255,272	46,461	47,482	4.8	5.0	8.4	6.5
Egypt	78,802	107,375	69,330	72,131	4.1	6.8	7.2	8.8
Algeria	85,027	113,888	32,364	33,494	5.2	3.6	9.1	4.7
Nigeria	71,533	116,488	142,655	149,877	6.0	6.5	10.5	8.4
Morocco	56,392	65,405	29,839	30,436	5.2	8.0	9.1	10.3
Tunisia	28,255	30,837	9,930	10,172	6.0	5.4	10.5	7.0
Libya	30,475	50,445	5,740	5,970	5.0	5.6	8.7	7.2
Sudan	21,685	36,218	34,474	37,442	5.1	11.8	8.9	15.2
Angola	19,800	45,167	14,973	15,864	11.2	18.6	19.6	24.0
Kenya	16,249	22,819	32,808	34,046	4.6	6.1	8.0	7.9
Total	625.0	843.9	418.6	436.9	57.2	77.4	100	100

Table A 3.10 Africa's Ten Largest Economies, 2004-2006

Source: IMF, WEO (2007).

			Form of
No	Company	Location	Ownership
			Transfer
1	The Libyan Iron and Steel Company	Misratah	Joint Stock Co.
2	The Socialist Ports	Misratah	Joint Stock Co.
3	Arab Cement	Al Markab	Joint Stock Co.
4	The National Company for Fodder and Mills	Tripoli	Joint Stock Co.
5	Libyan Airlines	Tripoli	Joint Stock Co.
6	Libyan Provisions	Tripoli	Joint Stock Co.
7	Handling, Ground Services and Maintenance Company in Airports	Tripoli	Joint Stock Co.
8	Public Tobacco	Tripoli	Joint Stock Co.
9	The Libyan Company for Poultry and Livestock	Tripoli	Joint Stock Co.
10	The National General Company for Maritime Transport	Tripoli	Joint Stock Co.
11	The Libyan Insurance Company	Al Zawiya	Joint Stock Co.
12	El Bariga for Oil Marketing	Al Zawiya	Joint Stock Co.
13	General Pipeline	Benghazi	Joint Stock Co.
14	Libyan Cement	Benghazi	Joint Stock Co.
15	General Wire	Benghazi	Joint Stock Co.
16	The National Company for Pharmaceuticals and Medical Supplies	Benghazi	Joint Stock Co.
17	Trucks and Buses	Tajoura	Joint Stock Co.
18	General Electronics	Tajoura	Joint Stock Co.
19	Libyan Tractors	Tajoura	Joint Stock Co.
~			

 Table A 3.11 Transferred Ownership Companies in the Libyan Stock Market

Source: GPC (2004).

(Million LD)

Year	Oil and N Act	atural Gas ivity	Total Non-oil Acti	Natural Gas vity	Trade	Current Account Balance Percent of	GDP
	Value	Per cent	Value	Per cent	Balance	GDP	
1966	529.2	62.8	313.9	37.2	211.7	4.9	843.1
1967	544.4	62.9	321.7	37.1	248.2	-5.8	866.1
1968	564.9	62.6	337.2	37.4	435.2	-12.3	902.1
1969	754.7	61.7	468.3	38.3	499.4	-23.3	1223.0
1970	812.6	63.1	475.7	36.9	589.2	-36.3	1288.3
1971	922.7	58.2	663.8	41.8	597.7	-19.1	1586.5
1972	920.6	52.5	832.4	47.5	946.4	-7.8	1753.0
1973	1131.0	51.9	1050.5	48.1	1132.5	-305.0	2182.5
1974	2385.3	62.9	1410.4	37.1	1215.8	-26.8	3795.7
1975	1961.1	53.4	1713.2	46.6	1550.8	-77.0	3674.3
1976	2750.0	57.7	2018.1	42.3	2676.5	-46.5	4768.1
1977	3275.9	58.4	2336.8	41.6	1761.2	-18.6	5612.7
1978	2808.7	51.1	2687.4	48.9	1067.1	-54.0	5496.1
1979	4545.3	59.8	3057.7	40.2	1472.3	-76.7	7603.0
1980	6525.7	61.8	4028.1	38.2	746.0	22.348	10553.8
1981	4403.3	50.0	4395.5	50.0	1925.1	-12.48	8798.8
1982	4235.8	47.4	4696.6	52.6	1784.5	-4.986	8932.4
1983	3823.6	44.9	4688.1	55.1	1831.8	-5.444	8511.7
1984	3209.8	41.1	4594.9	58.9	1458.7	-5.195	7852.1
1985	3500.4	44.6	4351.7	55.4	1357.8	6.86	6960.7
1986	2595.8	37.3	4364.9	62.7	393.2	-0.733	6011.6
1987	1875.4	31.2	4136.2	68.8	107.6	-4.955	6186.0
1988	1570.0	25.4	4616.0	74.6	192.7	-7.717	7191.0
1989	2055.5	28.6	5135.5	71.4	873.2	-4.086	8246.9
1990	3243.8	39.3	5003.1	60.7	1925.4	7.423	8757.3
1991	3104.3	35.7	5653.0	64.6	1585.5	1.35	9231.9
1992	2925.7	31.7	6306.2	68.3	1616.7	5.523	9137.7
1993	2460.1	26.9	6677.6	73.1	766.3	-2.893	9670.8
1994	2892.9	29.9	6777.9	70.1	1629.3	0.555	10672.3
1995	3380	31.7	7292.3	68.3	1493.6	16.587	12372.3
1996	3960.3	32.1	8367.0	67.9	1663.9	12.868	13800.5
1997	4505.8	32.6	9294.7	67.4	1317.0	11.959	12610.6
1998	2786.0	22.1	9824.6	77.9	170.3	7.842	14075.2
1999	3995.9	28.4	10079.3	71.6	1753.6	14.033	17620.2
2000	7761.9	42.1	10695.0	57.9	3310.1	31.536	18456.9
2001	7564.1	35.0	14054.5	65.0	2733.6	13.002	21618.7
2002	15209.8	50.1	15120.7	49.9	4591.3	3.331	30330.5
2003	21514.4	57.6	15846.3	42.4	9208.7	21.92	37360.7
2004	30848.4	64.1	17257.0	35.9	12593.1	24.316	48105.4
2005	46205.7	69.5	20245.0	30.5	23194.5	41.834	66450.7
2006	58358.1	72.3	22371.8	27.7	28401.6	51.582	80729.9
2007*	61834.2	69.3	27426.1	30.7	29402.2	42.545	89260.3

Notice a: * Preliminary data.

Source: IMF (2008); CBL, yearly statistical series, and monthly bulletin, various issue; General Planning Broad, Economics and Social Indicators (1962-2000).

Table A 3.13 Libya	a: Basic Econor	nic and Financ	ial Indicators	, 1997-2008
--------------------	-----------------	----------------	----------------	-------------

Indicators	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008
(Annual percentage change, u	nless otherwi	se specified	<i>d</i>)									
National Income and Prices			, ,									
Real GDP	5.2	-3.6	0.7	2.3	0.5	-0.2	5.9	5.0	6.3	5.2	6.8	8.8
No hydrocarbons	2.0	1.8	1.9	3.0	6.8	4.7	2.2	4.1	5.5	6.1	7.5	8.0
Hydrocarbons	4.9	3.8	5.1	4.9	7.8	9.8	17.7	7.4	8.3	2.8	4.8	10.9
Nominal GDP in billions of LD	13.1	14.5	14.7	17.7	18.1	24.6	30.8	39.8	54.5	65.2	73.7	95.7
Nominal GDP in billions of US\$	37.1	31.3	30.4	34.4	28.5	19.2	24.0	30.5	41.7	49.7	58.6	76.3
Per-capita GDP in billions of US\$	4.8	5.2	5.9	6.6	5.6	3.6	4.3	5.3	7.1	8.3	9.6	12.3
CPI inflation (average)	3.6	3.7	2.6	-2.9	-8.8	-9.8	-2.1	-2.2	2.0	3.4	7.0	8.0
CPI inflation (e-o-p)	2.6	-2.9	-8.8	2.8	2.9	-2.1	-1.3	-3.5	3.0	7.2	NA	NA
(In per cent of	of GDP)											
Central government finance												
Revenue	35.6	34.8	39.8	42.1	44.1	46.7	53.9	58.5	68.6	72.2	74.0	76.1
Of which: Hydrocarbon	23.7	20.0	17.4	27.2	29.0	35.6	47.1	50.6	63.7	66.8	66.7	69.7
Expenditure and net lending	36.0	38.9	34.3	32.6	44.4	42.9	39.1	41.2	38.6	33.1	39.3	42.4
Of which: Capital expenditure	7.6	6.2	8.3	10.3	10.5	13.7	9.2	17.7	19.8	17.9	22.6	25.8
Overall position (deficit -)	-0.4	-4.1	5.5	9.5	-0.3	3.9	14.8	17.4	30.0	39.1	34.7	33.7
Non-oil deficit	-23.0	-22.5	-11.9	-17.7	-29.3	-31.8	-32.3	-33.3	-33.7	-27.7	-32.0	-36.0
(Changes as a per cent of beggi	ng of the year	r money sto	ock)									
Money and credit												
Money and quasi-money	3.4	6.7	5.8	1.9	20.5	5.3	8.1	9.2	29.2	20.2	31.0	24.6
Net credit to the government	-5.4	1.9	-7.3	-29.7	-1.9	-10.7	-32.8	-104.2	-58.5	-110.2	-82.4	-99.2
Credit to the economy	4.6	3.9	5.2	4.9	5.9	6.2	6.6	-0.7	4.5	6.2	4.6	5.5
Of which: Credit to the private sector	0.5	0.9	1.1	0.8	1.2	0.5	-1.1	1.1	0.8	1.6	0.7	1.4
Deposit rate (1-year deposits, in <i>per cent</i>)	5.5	5.5	5.5	5.5	5.5	5.5	5.5	4.5	4.5	4.5	4.5	NA
(In billions of US\$ unless	otherwise in	dicated)										
Balance of payments								• • • •				<i>(</i>) <i>(</i>
Exports, f.o.b.	9.9	6.0	7.2	12.1	9.0	8.3	14.6	20.4	31.3	39.2	44.5	60.6
Of which: Hydrocarbons	9.1	5.6	6.7	11.6	8.5	8.1	14.2	19.5	30.4	38.2	43.4	59.3
Imports. F.o.b.	7.2	5.6	4.7	4.1	5.3	7.4	7.2	8.8	11.2	12.9	18.6	25.3
Current account balance	1.9	-0.4	1.6	7.0	2.4	-0.2	5.2	/.4	17.4	25.6	23.5	31.2
(As per cent of GDP)	5.1	-1.2	5.4	20.5	8.6	-1.2	21.5	24.3	41.8	51.6	40.2	40.9
Reserves		(7	(7	12.0	14.0	10.7	10.5	25.6	20.2	50.2	02.1	115.4
Gross official reserves	7.6	6./	6./	12.0	14.2	13.7	19.5	25.6	39.3	59.2	83.1	115.4
(In months of next year's imports of GNFS)	7.8	8.5	14.1	27.2	25.6	18.7	21.9	22.7	30.1	31.5	31.5	36.1
Exchange rate 1/		<u> </u>	0.5	0.5	0.6		1.00	1.20	1.01		1.04	
Official exchange rate (LD/US\$, period average)	0.4	0.4	0.5	0.5	0.6	1.3	1.28	1.30	1.31	1.31	1.26	NA
Official exchange rate (LD/US\$, end of period)	0.4	0.5	0.5	0.5	0.7	1.2	1.30	1.24	1.35	1.28	1.27	NA
Real effective exchange rate (<i>per cent</i> change)	2.2	3.3	1.9	1.9	1.6	NA	-13.91	-8.99	-1.75	-0.33	-0.98	NA
Crude oil production (millions of barrels per day)	1.20	1.25	1.32	1.35	1.32	1.20	1.53	1.62	1.69	1.72	1.80	2.00

Source: UN United Nations (2005), IMF (2007) Country Report No.07/149 and CBL (2008).

(Thousand	Barrels per Do	ay)					• ·			
Countries	1980	1985	1990	1995	2000	2001	2002	2003	2004	2005
Algeria	1,106.00	1,037.00	1,175.00	1,201.80	1,253.91	1,310.00	1,305.98	1,611.04	1,676.78	1,797.32
Egypt	895.00	887.00	873.00	920.00	768.33	719.66	715.42	712.67	673.38	658.20
Iran	1,662.00	2,250.00	3,088.00	3,643.22	3,724.30	3,723.30	3,444.70	3,742.80	4,001.43	4,138.58
Iraq	2,514.00	1,433.00	2,040.00	360.00	2,570.68	2,390.00	2,023.00	1,308.25	2,011.47	1,877.66
Jordan	0.0	0.0	0.33	0.03	0.04	0.04	0.02	0.02	0.02	0.02
Kuwait	1,656.00	1,023.00	1,175.00	2,057.41	2,078.50	1,997.50	1,894.16	2,136.00	2,375.70	2,529.18
Lebanon	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Libya	1,787.00	1,059.00	1,375.00	1,390.00	1,410.03	1,366.55	1,318.52	1,420.52	1,515.22	1,633.11
Morocco	1.00	1.00	0.31	0.30	0.30	0.40	0.26	0.20	0.20	0.50
Mauritania	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Oman	282.00	498.00	685.00	851.29	970.00	913.00	896.72	819.00	750.96	774.30
Qatar	472.00	301.00	406.00	442.00	737.20	714.15	679.11	715.00	782.54	835.00
Saudi Arabia	9,900.00	3,388.00	6,410.00	8,231.23	8,403.80	8,031.10	7,634.40	8,775.00	9,100.82	9,550.14
Sudan	0.0	0.0	0.0	0.0	186.00	209.13	238.77	269.74	342.51	350.00
Syrian	164.00	178.00	388.00	575.00	522.83	483.94	472.23	463.98	446.06	432.04
Tunisia	110.00	114.00	93.00	89.15	78.67	69.58	75.81	75.00	79.76	75.00
Turkey	46.00	41.00	73.00	67.47	52.68	48.00	46.67	45.17	42.42	43.67
UAE	1,709.00	1,193.00	2,117.00	2,233.00	2,367.83	2,205.00	2,082.00	2,348.14	2,478.26	2,535.42
Yemen	0.0	0.0	193.00	344.96	437.76	440.87	439.35	429.40	404.38	400.15
Middle East	18,442.00		18,979.02	21,823.01	20,936.38	19,602.19	20,775.27	22,388.72	22,388.72	23,107.58
Africa	6,125.00	5,371.00	6,432.31	6,954.20	7,527.15	7,525.85	7,520.1	8,156.53	8,864.01	9,645.15
Asia and Oceania	4,848.30	5,758.45	6,468.21	7,043.44	7,537.60	7,453.85	7,444.49	7,330.91	7,428.62	7,444.09
North America	11,968.00	13,187.00	11,461.33	10,982.48	10,810.61	10,957.63	11,093.36	11,357.31	11,200.28	10,880.82
World	59,557.56	53,966.19	60,491.68	62,384.88	68,494.63	68,101.03	67,167.93	69,448.44	72,512.48	73,806.86

Table A 3.14 MENA Crude Oil Production Including Libya, 1980-2005

Source: IEA (2005).

Countries	1985	1986	1987	1988	1989	1990	1991	1992	1993	1994	1995	1996	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006
Africa	2.6	2.4	1.8	4.0	3.0	2.3	0.3	1.2	0.4	2.5	2.9	5.7	3.2	2.8	2.6	3.1	4.2	3.6	4.6	5.5	5.2	5.7
Sub-Saharan Africa	1.7	2.5	2.3	4.5	2.9	2.1	-0.2	-1.8	0.9	2.2	3.6	5.3	3.8	1.9	2.6	3.4	4.2	3.6	4.2	5.6	5.5	5.8
Middle East	2.7	-1	2.9	-1.7	4.3	5.2	7.6	5.4	1.8	2.1	2.9	4.8	4.5	3.9	2.0	5.4	3.2	4.3	6.6	5.4	5.9	5.7
Central & Eastern Europe	2.4	3.8	3.6	1.7	1.3	-1.5	-7.7	-1.9	2.1	1.0	5.8	4.9	4.2	2.9	0.6	5	0.3	4.4	4.7	6.5	5.3	5.2
South and East Asia	7.3	6.4	7.6	9.1	5.9	5.3	6.2	9.1	9.4	9.8	9.1	8.4	6.7	4.3	6.3	7.0	6.1	7.0	8.4	8.8	8.6	8.2
Latin America	3.0	4.6	3.3	0.7	1.4	0.2	3.8	3.4	4.2	5.1	1.5	3.7	5.2	2.3	0.5	3.9	0.5	0.0	2.2	5.6	4.3	4.3

Table A 3.15 Real GDP Growth Rate Specific Countries in the World, 1985-2006

Source: IMF (2007) World Bank Outlook Database.

Tuble II cito initiation fuel opecific countries in the filotia, 1700 200	Table A 3.16	Inflation R	ate Specific	Countries in t	the World,	1985-2006
---	---------------------	-------------	--------------	-----------------------	------------	-----------

Countries	1985	1986	1987	1988	1989	1990	1991	1992	1993	1994	1995	1996	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006
Africa	14.2	17.5	17.2	17.7	20.4	15.4	36.8	43.9	36.7	49.8	33.8	28.4	13.7	9.3	11.9	13.6	12.7	9.9	10.8	8.1	8.5	9.1
Sub-Saharan Africa	15.6	19.3	20.8	21.5	24.4	17.6	42.9	52.1	44.5	61.1	38.6	33.9	16.6	10.9	15	17.4	15.7	12.3	13.4	9.7	10.6	10.7
Middle East	16.2	14.3	17.7	14.3	18.2	20.9	9.3	22.4	21.3	22	24.4	26.3	24.4	26.3	24.6	19.1	23.8	29.4	26.0	25.0	28.7	33.5
Central & Eastern Europe	16.7	16.2	20.3	36.3	78.8	119.5	89.9	86.1	75.3	59	40.6	36.6	51.7	32.7	23.0	22.8	19.4	14.7	9.2	6.1	4.8	4.1
South and East Asia	8.7	9.5	9.9	15.4	12.5	7.1	9.4	8.5	10.6	15.8	13.1	8.1	4.8	7.7	2.4	1.8	2.6	2.0	2.5	4.2	3.6	3.9
Latin America	149.4	90.2	131.2	250.5	444.6	567.6	151.3	164.8	215.8	221.4	36.4	19.8	11.9	9.0	8.2	7.6	6.1	8.9	10.5	6.5	6.3	5.8

Source: IMF (2007) World Bank Outlook Database.

100013317 Central Obvermient in <i>per cent</i> of OD1 Specific Countries in the world, $1205-2000$

Countries	1985	1986	1987	1988	1989	1990	1991	1992	1993	1994	1995	1996	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006
Africa	-0.6	-3.3	-1.4	-3	-2.2	-1.1	-1.5	-2	-2.1	-2.9	-3.7	-1.2	-1.4	-4.5	-3.5	1.6	0.1	-1.6	-0.4	0.1	1.9	2.6
Sub-Saharan Africa	-0.4	-2.7	-1.4	-3.2	-2.1	-1.4	-2.3	-2.5	-2.4	-2.7	-3.5	-1.8	-2.5	-5.4	-4.4	-0.2	-2.2	-3.5	-2.8	-2.1	-1.1	0.3
Middle East	-1.7	-5.4	-1.9	-2.6	-0.2	0.7	-15.9	-5.1	-4.8	-0.9	0.7	2.9	1.9	-5.0	2.3	11.1	6.2	4.6	4.6	8.1	12.4	19.1
Central & Eastern Europe	-1.3	-1.5	-0.7	0.3	0.8	-0.9	0.8	0.2	-2.8	1.2	-1.3	-3.2	-3.7	-3.1	-4.4	-5.3	-2.7	-3.5	-4.3	-5.7	-5.2	-5.5
South and East Asia	-2.9	-2.2	-1.0	-1.7	-2.0	-1.3	-1.0	-1.0	-2.3	-1.2	-2.3	-1.9	0.4	2.5	2.3	2.0	1.7	2.7	2.9	2.7	3.9	3.6
Latin America	-0.3	-2.3	-1.4	-1.1	-0.5	-0.1	-1.5	-2.7	-3.3	-3.2	-2.2	-2.2	-3.3	-4.5	-3.2	-2.4	-2.8	-0.9	0.4	0.9	1.2	0.8

Source: IMF (2007) World Bank Outlook Database.

Table A 3.18 Difference between Savings and Investment Specific Countries in the World, 1985-2006

85 19	986	1987	1988	1989	1990	1991	1992	1993	1994	1995	1996	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006
.5 22	2.4	20.1	20.6	20.6	19.8	20.6	20.4	19.5	20.2	20.0	19.1	19.1	20.9	20.7	19.3	20.0	19.0	20.6	21.6	21.1	22.0
.7 19	9.1	17.4	19.1	18.3	17.8	18.7	17.5	17.3	18.4	18.4	17.7	18.1	19.5	18.9	17.5	18.0	16.4	18.4	19.2	19.2	19.7
.0 21	1.2	20.5	21.0	22.1	23.6	26.1	27.6	24.4	21.8	22.4	23.2	22.7	24.3	21.9	20.1	21.6	22.4	22.9	22.8	22.0	21.9
.2 28	8.6	28.2	28.1	27.1	27.6	23.6	20.6	22.6	21.1	23.4	24.4	25.2	25.1	24.5	25.0	22.0	22.5	22.9	24.3	24.0	24.5
.2 28	8.5	27.8	29.5	30.3	30.3	30.7	31.2	34.9	33.9	35.0	33.8	32.4	29.0	28.4	28.2	28.9	29.7	31.9	33.4	34.3	35.6
.3 19	9.4	21.9	22.2	22.1	19.6	19.7	20.1	20.7	21.5	21.3	21.2	22.8	22.4	20.6	21.1	20.1	19.4	19.6	20.6	20.7	21.0
8	5 19 5 22 7 19 0 22 2 24 2 24 3 19	5 1986 5 22.4 7 19.1 0 21.2 2 28.6 2 28.5 3 19.4	5 1986 1987 5 22.4 20.1 7 19.1 17.4 0 21.2 20.5 2 28.6 28.2 2 28.5 27.8 3 19.4 21.9	5 1986 1987 1988 5 22.4 20.1 20.6 7 19.1 17.4 19.1 0 21.2 20.5 21.0 2 28.6 28.2 28.1 2 28.5 27.8 29.5 3 19.4 21.9 22.2	5 1986 1987 1988 1989 5 22.4 20.1 20.6 20.6 7 19.1 17.4 19.1 18.3 0 21.2 20.5 21.0 22.1 2 28.6 28.2 28.1 27.1 2 28.5 27.8 29.5 30.3 3 19.4 21.9 22.2 22.1	5 1986 1987 1988 1989 1990 5 22.4 20.1 20.6 20.6 19.8 7 19.1 17.4 19.1 18.3 17.8 0 21.2 20.5 21.0 22.1 23.6 2 28.6 28.2 28.1 27.1 27.6 2 28.5 27.8 29.5 30.3 30.3 3 19.4 21.9 22.2 22.1 19.6	5 1986 1987 1988 1989 1990 1991 5 22.4 20.1 20.6 20.6 19.8 20.6 7 19.1 17.4 19.1 18.3 17.8 18.7 0 21.2 20.5 21.0 22.1 23.6 26.1 2 28.6 28.2 28.1 27.1 27.6 23.6 2 28.5 27.8 29.5 30.3 30.3 30.7 3 19.4 21.9 22.2 22.1 19.6 19.7	5 1986 1987 1988 1989 1990 1991 1992 5 22.4 20.1 20.6 20.6 19.8 20.6 20.4 7 19.1 17.4 19.1 18.3 17.8 18.7 17.5 0 21.2 20.5 21.0 22.1 23.6 26.1 27.6 2 28.6 28.2 28.1 27.1 27.6 23.6 20.6 2 28.5 27.8 29.5 30.3 30.3 30.7 31.2 3 19.4 21.9 22.2 22.1 19.6 19.7 20.1	5 1986 1987 1988 1989 1990 1991 1992 1993 5 22.4 20.1 20.6 20.6 19.8 20.6 20.4 19.5 7 19.1 17.4 19.1 18.3 17.8 18.7 17.5 17.3 0 21.2 20.5 21.0 22.1 23.6 26.1 27.6 24.4 2 28.6 28.2 28.1 27.1 27.6 23.6 20.6 22.6 2 28.5 27.8 29.5 30.3 30.3 30.7 31.2 34.9 3 19.4 21.9 22.2 22.1 19.6 19.7 20.1 20.7	5 1986 1987 1988 1989 1990 1991 1992 1993 1994 5 22.4 20.1 20.6 20.6 19.8 20.6 20.4 19.5 20.2 7 19.1 17.4 19.1 18.3 17.8 18.7 17.5 17.3 18.4 0 21.2 20.5 21.0 22.1 23.6 26.1 27.6 24.4 21.8 2 28.6 28.2 28.1 27.1 27.6 23.6 20.6 22.6 21.1 2 28.5 27.8 29.5 30.3 30.3 30.7 31.2 34.9 33.9 3 19.4 21.9 22.2 22.1 19.6 19.7 20.1 20.7 21.5	5 1986 1987 1988 1989 1990 1991 1992 1993 1994 1995 5 22.4 20.1 20.6 20.6 19.8 20.6 20.4 19.5 20.2 20.0 7 19.1 17.4 19.1 18.3 17.8 18.7 17.5 17.3 18.4 18.4 0 21.2 20.5 21.0 22.1 23.6 26.1 27.6 24.4 21.8 22.4 2 28.6 28.2 28.1 27.1 27.6 23.6 20.6 22.6 21.1 23.4 2 28.5 27.8 29.5 30.3 30.3 30.7 31.2 34.9 33.9 35.0 3 19.4 21.9 22.2 22.1 19.6 19.7 20.1 20.7 21.5 21.3	5 1986 1987 1988 1989 1990 1991 1992 1993 1994 1995 1996 5 22.4 20.1 20.6 20.6 19.8 20.6 20.4 19.5 20.2 20.0 19.1 7 19.1 17.4 19.1 18.3 17.8 18.7 17.5 17.3 18.4 18.4 17.7 0 21.2 20.5 21.0 22.1 23.6 26.1 27.6 24.4 21.8 22.4 23.2 2 28.6 28.2 28.1 27.1 27.6 23.6 20.6 22.6 21.1 23.4 24.4 2 28.5 27.8 29.5 30.3 30.3 30.7 31.2 34.9 33.9 35.0 33.8 3 19.4 21.9 22.2 22.1 19.6 19.7 20.1 20.7 21.5 21.3 21.2	5 1986 1987 1988 1989 1990 1991 1992 1993 1994 1995 1996 1997 5 22.4 20.1 20.6 20.6 19.8 20.6 20.4 19.5 20.2 20.0 19.1 19.1 7 19.1 17.4 19.1 18.3 17.8 18.7 17.5 17.3 18.4 18.4 17.7 18.1 0 21.2 20.5 21.0 22.1 23.6 26.1 27.6 24.4 21.8 22.4 23.2 22.7 2 28.6 28.2 28.1 27.1 27.6 23.6 20.6 22.6 21.1 23.4 24.4 25.2 2 28.5 27.8 29.5 30.3 30.3 30.7 31.2 34.9 33.9 35.0 33.8 32.4 3 19.4 21.9 22.2 22.1 19.6 19.7 20.1 20.7 21.5 21.3 21.2 22.8	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	5 1986 1987 1988 1989 1990 1991 1992 1993 1994 1995 1996 1997 1998 1999 5 22.4 20.1 20.6 20.6 19.8 20.6 20.4 19.5 20.2 20.0 19.1 19.1 20.9 20.7 7 19.1 17.4 19.1 18.3 17.8 18.7 17.5 17.3 18.4 18.4 17.7 18.1 19.5 18.9 0 21.2 20.5 21.0 22.1 23.6 26.1 27.6 24.4 21.8 22.4 23.2 22.7 24.3 21.9 2 28.6 28.2 28.1 27.1 27.6 23.6 20.6 21.1 23.4 24.4 25.2 25.1 24.5 2 28.5 27.8 29.5 30.3 30.7 31.2 34.9 33.9 35.0 33.8 32.4 29.0 28.4 3 19.4 21.9 22.2 22.1 19.6 19.7 20.1 20	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	5 1986 1987 1988 1989 1990 1991 1992 1993 1994 1995 1996 1997 1998 1999 2000 2001 2002 5 22.4 20.1 20.6 20.6 19.8 20.6 20.4 19.5 20.2 20.0 19.1 19.1 20.9 20.7 19.3 20.0 19.0 7 19.1 17.4 19.1 18.3 17.8 18.7 17.5 17.3 18.4 18.4 17.7 18.1 19.5 18.9 17.5 18.0 16.4 0 21.2 20.5 21.0 22.1 23.6 26.6 22.6 21.1 23.4 23.2 22.7 24.3 21.9 20.1 21.6 22.4 2 28.6 28.2 28.1 27.1 27.6 23.6 20.6 22.6 21.1 23.4 24.4 25.2 25.1 24.5 25.0 22.0 22.5 2 28.5 27.8 29.5 30.3 30.3 30.7 31.2 <t< td=""><td>$\begin{array}{c ccccccccccccccccccccccccccccccccccc$</td><td>$\begin{array}{c ccccccccccccccccccccccccccccccccccc$</td><td>$\begin{array}{c ccccccccccccccccccccccccccccccccccc$</td></t<>	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$

Source: IMF (2007) World Bank Outlook Database.



Country	Market Name	Web Site	Established Year	Number of List Companies	Value Traded (Million US\$)	Shares Traded (Million)	Market Capitalisation Million US\$	AMF Index ^c
Algeria	Algeria Stock Exchange	www.cosob.org/	1996	2	0.207	0.035	97	67.2
Bahrain	Bahrain Stock Exchange	www.bahrainstock.com	1989	51	1,069	851	27,016	209.4
Egypt	Cairo & Alex. Stock Exchange	www.egyptse.com/	1888	435	64,772	15,752	138,828	353.1
Jordan	Amman Stock Exchange	www.ase.com.jo/	1976	245	17,424	4,479	41,233	522.6
Kuwait	Kuwait Stock Exchange	www.kuwaitse.com	1977	196	130,896	70,417	135,362	444.4
Lebanon	Beirut Stock Exchange	www.bse.com.lb/	1920	15	994	114.96	10,894	135.0
Libya ^a	Libyan Stock Market	www.lsm.gov.ly	2006	3	25,958	254	NA	NA
Morocco	Casablanca Stock Exchange	www.casablanca-bourse.com/	1929	73	22,009	262.05	75,495	571.9
Oman	Muscat Securities Exchange	www.msm-oman.com/	1988	125	5,211	2,742	23,086	177.9
Palestine	Palestine Stock Exchange	www.p-s-e.com	1995	35	817	301	2,474	189.5
Qatar	Doha Securities Exchange	www.dsm.com.qa/dsmsite/	1959	40	29,927	3,411	95,505	478.7
Saudi Arabia	Saudi Stock Market	www.tadawul.com	1934	116	682,287	57,886	518,984	555.6
Sudan	Khartoum Stock Exchange	www.dsm.com.qa/dsmsite/	1962	52	893	9,410	4,931	182.9
Tunisia	Tunis Stock Exchange	www.bvmt.com.tn/	1969	51	727	96,65	5,339	58.4
United	Abu Dhabi Securities Exchange	www.adsm.ae	2000	64	47,746	52,067	121,128	358.9
Emirates	Dubai Financial Market	www.dfm.ae	2000	55	103,297	105,257	138,179	592.4
		Totals		1550	1,134,028	577,019	1,338,550	328.7 °
Arabia Sudan Tunisia United Emirates	Khartoum Stock Exchange Tunis Stock Exchange Abu Dhabi Securities Exchange Dubai Financial Market	www.tadawdr.com www.dsm.com.qa/dsmsite/ www.bvmt.com.tn/ www.adsm.ae www.dfm.ae Totals	1954 1962 1969 2000 2000	52 51 64 55 1550	893 727 47,746 103,297 1,134,028	9,410 96,65 52,067 105,257 577,019	4,931 5,339 121,128 138,179 1,338,550	182.9 58.4 358.9 592.4 328.7 °

Table A 3.19 Summary Information and Indicators in Arab Stock Market at the End of period 2007

Notice a: Data available from Libyan a stock market in 30/09/2007.

Notice b: N/A Not Applicable.

Notice c: Composite Index.

Source: Arab Monetary Fund, Arab Market Data Base (2008).

APPENDICES



Appendix B: Research Methodology Diagram



- Notice a: Area 1 (SMPV) stock market performance variables including (SMPMZ) market size; (SMPML) market liquidity; (SMPMA) market activity; (SMPMC) market concentration . Area 2 (ERPV) economic reform programme variables including (ERPD) deregulation; (ERPC) corporatisation; (ERPV) privatisation; (ERPL) liberalisation. Area 3 (MERPV) macro-economic reform variables including (MERIR) interest rate; (MERER) exchange rate; (MERIFR) inflation rate; (MERBD) budget deficit; (MERPCI) per-capita income; (MERGDP) GDP growth rate. Area 4 (FMV) financial maturity variables including (FMBA) banking assets ratio; (FMDC) domestic credit ratio; (FMMC) market capitalisation ratio. Area 5 (GMV) growth model including (SM) stock market variables MC/GDP market capitalisation ratio; VT/GDP value trade ratio; TURNOVER turnover ratio; BS banking sector variables BAR banking asset ratio; DC/GDP domestic credit ratio; (MEP) macro-economic; PCYG real per-capita GDP growth rate; POP population growth; EDUC secondary school enrolment rate; INVES investment ratio; CI conditioning information variables ER economic freedom; PS political stability; OPEN export + imports/GDP; INF inflation rate.
- Notice b: Area 6 (Future Research) including (SMP) SML stock market liberalisation; SMVOL stock market volatility; VTR/VOL value traded ratio divided by volatility; TOR/VOL turnover ratio divided by volatility; (ERP) including ERPD deregulation; ERPC corporatisation; ERPV privatisation; (MERP) including CA capital accumulation; HC human capital; PS private savings' rates; PG productivity growth; FDI foreign direct investment; PR profitability ratio; TC technical changes; EE economic efficiency; (GM) other developed countries transferable model such as Australia; Finland; France; Germany; Greece; Japan; Singapore; Spain; UK and US.
- Notice c: H: testing research hypothesise.

Appendix C: Covering Letter from the Researcher Sent with the Self-administered Questionnaires



LIBYA'S ECONOMIC REFORM PROGRAMME AND THE CASE FOR A STOCK MARKET QUESTIONNAIRE SURVEY

Dear Participant

I am a PhD candidate currently studying in the Business School of the University of Huddersfield, United Kingdom, conducting research into "*Libya's Economic Reform Programme and the Case for a Stock Market*".

C Aim of Survey

The primary aim of this survey is to identify the most important factors that might affect the function of the Libyan stock market.

C Non-Response to Questions

If you feel unable to answer any questions, please leave them blank or insert comments such as 'N/A' or 'Don't know'. If you wish to make additional comments at any point please feel free to do so either by using spare specify on the questionnaire or by providing a covering note.

Outcome

All responses will be treated with the utmost confidence. The results of the survey will be used for research purposes only and no attempt will be made to identify an individual or organisation. I would very much appreciate your participation, since the success of this research depends upon your response.

C Instructions

Please attempt to answer every question; there are no right or wrong answers. I am seeking your judgement or opinion only. The survey consists of 10 pages and should take no longer than 30 minutes to complete. I very much hope you will be able to participate and feel sure that you will find the subject matter interesting. Thank you very much for your help in advance. Please return the completed questionnaire in the addressed envelope provided.

I look forward to receiving your reply.

Sincerely yours,

Najeb Masoud

Business School University of Huddersfield HD1 3DH Telephone: 07757261306 E-mail: <u>najeb2000@hotmail.com</u> Dr.John Coock

Direct of Studies Businesses school Huddersfield University HD1 3DH Tel: +44 (0) 1484 472086 E-mail: j.a.cook @hud.ac.uk





Appendix E: Covering Latter form Cultural Affairs the Libyan People's Bureau Sent with Self-administered Questionnaires in (Arabic)

The People's Bureau	إفالعلوقات الدراغة بين الشعوب وليدت بين الحكومات	(لمكتر - والشعبي
of The Great Socialist		للجراهيريني العربير تبير الليبر ومجتم
People's Libyan Arab Jamahiriya	I NOOF	الشعبة (الاير تر (للة (لغظم)
London		ريد ايد ريد من رويد ورد ان
Cultural Affairs		لالشيئور للتف افية
		التاريخ: الصيف /2007

الموضوع/ افــــادة

الى من يهمه الأمر

المر فقات:

بعد التحية،،،

يفيد الملحق الثقافى الليبي فى بريطانيا بأن الطالب/ نجيب محمد حمودة مسعود مبتعث من أمانه اللجنة الشعبية العامة للتعليم العالي، لتحضير درجة الدكتوراه من جامعة هدرزفيلد فى مجال "الاقتصاد والاسواق المالية". وهو الأن بصدد القيام برحلة علمية الى ليبيا لتوزيع استمارات الاستبانه الى المؤسسات المالية المصرفية والشركات وجمع المعلومات المطلوبة لموضوع بحثه. نأمل تقديم المساعدات الممكنة لتسهيل مهمت العلمية.

وقد صدرت هذه الافادة بناء على طلبة لتقديمها الى جهات الاختصاص.

والسلام عليكم ورحمة الله وبركاته،،،

الشئون الثقافية المكتوم الشعري - لنحن

مورة المي : - الملف الش<mark>حصي -</mark>

www.culturecorner.org.uk

61-62 Ennismore Gardens, London SW7 1NH Tel: 020 7581 4142 Fax: 020 7581 2393

Appendix F: Free Order Form



LIBYA'S ECONOMIC REFORM PROGRAMME AND THE CASE FOR A STOCK MARKET

FREE ORDER FORM QUESTIONNAIRE

As a participant of this questionnaire, you are entitled to receive a free copy of the research findings. These should be of value to finance, in particular to those involved the economic reform programmes and the stock market. Your cooperation and participation in this research will be of great benefit, as the findings will:

- Enable better understanding of the economic reform programme and stock market performance currently supporting the Libyan economy.
- Enable comparison of supporting economic reform programme and stock market performance with other financial markets in both Libya and other countries.
- Enable better understanding of economic reform programmes and stock market performance, so that their impact upon finance and the economy may be strengthened in the future.

If you are interested in obtaining a free copy of the result of this study, please complete your details here and attach this order form with the completed questionnaire.

Name:	(optional)
Title:	· · · · ·
E-mail:	
Address:	

Thank you for your time and co-operation in completing this questionnaire.

-						
	Universit HUDDERS	y of FIEL D				
	LIBYA'S ECONOMIC REFORM PROGRAMME A	ND THE CASE FOR A S	TOCK MARKET			
	اليبيا والحال بالنسبة لسوق الأوراق المالية	برنامج الإصلاح الافتصادي في				
	OUESTIONNAIDE	SUDVEV				
	QUESTIONNAIRE متبالة الدراسة					
	Section A: Demographic			قسم الأول : سكّاني]	
The p	urpose of this section is to collect general information about respondents and their financial institutions. Please tick ($$)) في الخانة المناسبة في إلاسئلة	ىاتهم المالية. <u>بر</u> جى وضع علامة (√)	لمعلومات العامة عن المجيبين ومؤسس	ٍ الحصول على ا	الغرض من هذا القسم هو
the ap	propriate box for each the following questions.					التالية
A.1	Your gender is:).«	س 1: ما جنسك
1 2	L Male L Female				اللى	ددر سر دما هر ع را <i>ف</i>
A.2	$\Box \text{ Under } 30 \Box \Box 30.40$			65-51 50-41	40-30	س <u>2</u> : ما بھو عمرت اقل من 30
	$\Box 41-50$ $\Box 51-65$			05 51 50 41	40 50	50 0 0
A.3	Your highest education level is:			لت علية	رى تعليميا تحصا	س 3: ما هو أعلى مست
	☐ High school ☐ College degree ☐ Bachelor's degree		البكالوريوس	دبلوم عالي درجة	عامة	شهادة الثانوية ا
	☐ Master's degree ☐ Doctorate Other, please specify		يرجى التحديد	درجة دكتوراه أخرى		درجة الماجستير
A.4	Your academic background:				الاكاديمي	س 4: ما هو تخصصك
	□ Accounting and finance □ Economics □ Business studies □	إدارة التسويق	الأعمال المصرفية	صاد الدراسات التجارية	ة اقت	المحا سبة والمالي
	□ Money and banking □ Management and marketing Other, please specify				دید	أخرى، يرجى الت
A.5	Your current position:	stitististi.	t i ti ersti ti .	1.811 . 11	وظيفي الحالي	س 5: ما هو مرکزك ال
	□ General manager □ Senior manager □ Administrative manager	مدير الفوى العاملة	مدير العلاقات العامة	ير المدير الاداري	משושב מב בנונ	المدير العام
A.6	What is your bank category:				<u>تتعامل معه</u>	س 6: ما هو البنك الذي
	\Box Central bank of Libya \Box Commercial bank $\Box\Box$ Foreign bank \Box		نك أجنبى	المصرف التجاري ب	کزی ت	مصرف ليبيا المرّ
	□ Investment bank □ Private bank Other, please specify		خرى، يرّجى التحد	المصرف الخاص أ.		بنك الاستثمار
A.7	How many years work experience do you have:		a constant a set	الخبرة في حياتك العملية	للي عدد سنوات	س 7: بِشْكَل عام كم إجد
	\Box Less than 3 years \Box \Box 3-6 years \Box 7 to 10 years \Box More than 10 years		الحتر من ()[سنوات	ت 7 إلى () [سنوات	3-6 سنوا اند بر ۱۱ مار ا	افل من 3 سنوات
А.ð	Number of employees:	3210 2106 im	2105 1341		طعين العاملين	س 8: کم يبنغ عدد المو اقل من 850
	□ Between 2106-3210 □ Over 3210	5210-2100 	2105-15410			ا <u>کثر</u> من 3210
A.9	Which of the following type of shareholders best represents your company's majority ownership:			يمكن أن تصنف	لقطاعات التالية	س 9: إلى أي نوع من ا
	Public company Private company Other, please specify		برجي التحديد	خاص أخرى، ب	قطاع	قطاع عام
A.10	Your company's total assets (Millions LD):		٤ - ١	مليون د.ل)	أصول الشركة (س 10: كم تقدر إجمالي
	$\Box \text{ Low (Less than 10)} \qquad \Box \text{ Medium (10-30)} \qquad \Box \text{ High (More or = 30)}$		کتر او = 30	متوسطة (10-30) كبيرة (<u> </u>	صغيرة (اقل من
A.11	Y OUR CURRENT MARKET REGION:		مارية	، محادة ماقارمية	لحالي الحاص بك اقارور	س 11: منطقة السوق ا
			ىوىيە	به محبیه و رسیمیه	<u>(</u>)	محيب

Appendix G: Economic Reform Programme and Stock Market Performance Questionnaires (Original in Design)

A.12	Your industry: Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and transport Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and transport Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and transport Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining Imanufacturing and transport Imanufacturing and mining Imanufacturing and mining Imanufacturing and mining I						ى أي نوع من القطاعات الصناعية يمكن أن تصنف مؤسستكم سناعة التحويلية والتعدين التجارة الطاقة، النفط والغاز راعة المالية وسائل الإعلام الخدمات تصالات والنقل والنباء أخرى، يرجيه التحديد	س 12: إلم الم الزر الإن
A.13	What are the fiscal revenues attributed to business activities?						ا هي الإيرادات الضريبية التي نسبت إلى الأنشطة التجارية؟ منذ ت	س 13: ما
	\Box Low \Box \Box Medium \Box High						حقصة ملوسطة عالية 	م
	Section B: Economic Reform Programme						القسم الثاني: برنامج الإصلاح الاقتصادي	
This s	ection investigates the structural change caused by environmental conditions in the economic reform program	nme	ات	معلوم	ل على	لحصوا	بحث في التغيير الهيكلي الناجم عن الظروف البئيه في متغيرات برنامج الإصلاح الاقتصادي. والهدف من ذلك هو ال	هذا القسم ي
variab	les. The aim is to obtain information about the Libyan economic reform programme in the stock market structure.				1 01	i di	ج الإصلاح الاقتصادي الليبي في تركيبه هيكله سوق اسهم الاور اق الماليه. المسلح الاقتصادي الليبي في تركيبه هيكله سوق اسهم الاور اق الماليه.	حول برنام
	Please tick the box that best reflects your answe	er wh	ere:		لزيمة	;در م	الرجاء وصع إساره على الرقم الا	
Code	i i = Strongly disagree د الوقق بسب	ree no	or disa	agree	حايد : انترمة	م ۱	ال الله عنه ع S = Strongly agree الله الله عنه ع S = Strongly agree الله الله الله الله الله الله الله ال	
Coue	Desegulation	500	bre		<u>سېچ</u> ،	,	التقرير والت	- .)
R 1	Economic reform is an important factor in the regulation of the stock market	1	2	3	4	5	التصريبات الإصلاح الاقتصادي بشكل عاملا معما في تنظره سمق الأمد إق الملاية	R 1
B.1 B.2	Economic reform is an important factor in the regulation of the stock market	1	2	3	4	5	ا پر مصادع او مصادی پیشن عاماد مهاما دی مصبح منوبی او و رای این بید. انتظاره سره قد الاُم داده الله الدة أمد حدد الاسانژه از	B.1 B.2
B.2 B.3	Regulation of the stock manufect is good for intestinent	1	2	3	4	5	يشجو استقرار الثير دوان الشركات الاستثمار في سمق الأسم المالية.	B.3
B.4	The laws of the stock market help to promote invest much stock market	1	2	3	4	5	ي بي مسرر (مسريات مسريات موريات ومسادر في مسري المنهم مسية). قد اندن مسدق الأسيم المالية قد تساعد في تعزيز الاستثمار في مسوق المال	B.4
B 5	Regulation is a important element to market and promote investment for the stock market	1	2	3	4	5	توريبي سري ترسم مدير = = = = ي تريبر تريسة را لاسعة المالية . القوانين معمة لتسويق وترويح الاستثمار في سوق الأسعة المالية .	B.5
B.6	Regulation to the stock market and promote investment of the stock market	1	2	3	4	5	اللوابين لهذ السوين والروين ، وتسدر في سرى ، يسم ، تساهد ألاده ات التشر بعدة المساهمة في تحسين سوق الأسهم المالية	B.6
B.7	The laws of the stock market play a crucial role in limiting the value of companies' shares	1	2	3	4	5	ق اندن سوق الأوراق المالية تلعب دورا جاسما في تحديد قيمة أسعو الشركات	B.7
B.8	Simplified procedures affect investment in the stock market	1	2	3	4	5	توري حروري جروري ير رو يَوَدُرُ الأحد إمات المُدسَطَة عليه الاستثمار في سوق الأوراق المالية	B.8
2.0	Cornoratisation	-	-		•	e		10.0
B.9	Corroratisation is an important factor in a market-based economy	1	2	3	4	5	التشار كيات تشكل عاملا معما في اقتصاد قائم على السوق	B.9
B.10	The corporation of companies plays a vital role in contributing to the stock market	1	2	3	4	5	التشار كيات جزء من الشركات والتي تلعب دورا جدوبا في المساهمة في سوق الأوراق المالية	B.10
B.11	Public sector assigns laws for endorsing the investment in the stock market	1	2	3	4	5	ر جي بربر القطاع العاد القوانين لتصديق على الاستثمار في سوق الأوراق المالية	B.11
B.12	The Librar government encourages corrorate companies to invest managers in the stock market	1	2	3	4	5	يستسف مسيح مدم علو المير كات للإستثمار في سوق الأسهم المالية الحكومة الليبية تشجع مدراء الشركات للإستثمار في سوق الأسهم المالية	B.12
	Managers of corporate companies play a crucial role in encuraging prospective customers to invest in the					-	مدراء الشركات بلعيون دورا حاسما في تشجيع الزيائن المتوقعين للإستثمار في اقتصاد قائم على السوق	
B.13	market-based economy	1	2	3	4	5		B.13
B.14	The following strategic targets are important for the future of the firm:		•				إنَّ الأهداف الإستر اتيجية التالية مهمة لمستقبل الشركة:	B.14
	B.14.1 Providing job opportunities	1	2	3	4	5	B.14.1 توفير فرص عمل	
	B.14.2 Providing domestic market goods and services	1	2	3	4	5	B.14.2 تزويد الاسوق المحلية بالسلع والخدمات	
	B.14.3 Increasing profitability	1	2	3	4	5	B.14.3 زيادة الربحيه	
	B.14.4 Repaying borrowing	1	2	3	4	5	B.14.4 سداد الافتراض	
	B.14.5 Expanding the firm	1	2	3	4	5	B.14.5 توسع الشركة	
	Privatisation						الخصخصة	
B.15	Private sector is important in strengthening the corporate market	1	2	3	4	5	القطاع الخاص مهم في تعزيز سوق الشركات	B.15
B.16	Privatisation is a new term in the stock market	1	2	3	4	5	الخصخصة هي اصطلاح جديد في سوق الأسهم المالية	B.16
B.17	Privatisation moves the Libyan economy towards a stock market	1	2	3	4	5	الخصخصة تساعد انتقال الاقتصاد الليبي نحو اقتصاد سوق الاسهم المالية	B.17
B.18	Privatisation transforms the economy from a socialist to a market economy	1	2	3	4	5	الخصخصة تحول الاقتصاد من اقتصاد اشتراكي إلى اقتصاد السوق	B.18
B.19	Privatisation of commercial banking sector	1	2	3	4	5	خصخصة القطاع المصرفي التجاري	B.19
B.20	Your company encourages you to borrow money for a business deal as long as it is profitable	1	2	3	4	5	شركتك تشجّعك على اقتراض الأموال لصفقة تجارية، طالما انه مربح	B.20
B.21	The Libyan government motivates the private sector to invest in the stock market	1	2	3	4	5	تحقر الحكومة الليبية القطاع الخاص للاستثمار في سوق الاسهم المالية	B.21
B.22	Privatisation is the real reason for contribution of productivity growth	1	2	3	4	5	الخصخصة هي السبب الحقيقي للمساهمة في نمو معدل الإنتاج	B.22
B.23	Technology tools such as internet telecommunications are crucial in the privatised system	1	2	3	4	5	ادوات تقنية التكنولوجية مثل الانترنت والاتصالات السلكية واللاسلكية تشكل عاملا حاسما في نظام الخصخصة	B.23
B.24	The private sector can co-operate with stock market authorities to increase competitive advantages	1	2	3	4	5	يمكن للقطاع الخاص ان يتعاون مع سلطات سوق الاور اق المالية لزيادة المزايا التنافسية	B.24
	Liberalisation	-	-	1			لتحرير)
B.25	Liberalisation of trade and investment regimes as well as deregulation and privatisation of government business activities generate opportunities for expansion of trade investment and technology flows	1	2	3	4	5	تحرير انظمة الاستثمار والتجارة بالإضافة إلى إلغاء قيود التشريعات التنظيمية وخصخصة الفعاليات التجارية الحكومية يولدان فرص الاستثمار والتوسّع التجاري وتدفق التقنية	B.25
B.26	Liberalisation restricts the movement of product and factor flows across borders involving goods, services,	1	2	3	4	5	يحدّد التحرير حركة تدفق المنتجات وعوامل الإنتاج عبر الحدود التي تنطوي على تدفقات السلع والخدمات	B.26

	investment, financial capital, technology and labour						والاستثمار ورؤوس الأموال والتكنولوجيا والعمل	
B.27	Liberalisation has led to a lower cost of transactions and global business improvements in transport,	1	2	3	4	5	التحرير يؤدي إلى انخفاض تكلفة الصفقات وأجراء تحسينات في المعاملات التجارية العالمية في مجالات النقل	B.27
	communication and information technology networks	-	-	ĩ	·	•	وشبكات تقنية الاتصالات وتكنولوجيا المعلومات	
B.28	Rapid liberalisation (dismantling and lowering of tariff/non-tariff barriers) and deregulation, both at national	1	2	3	4	5	التحرير السريع يؤدي إلى (تفكيك وتخفيض التعريفه أو الحواجز غير الجمركية)، ورفع القيود على التشريعات	B.28
D.20	and international levels, aids trade	-	-	č	•	e	التنظيمية على الصعيدين الوطني والدولي، والتجاري الت	D.2 0
B.29	Liberalisation may affect specific components of capital flows including debt (financial and non-financial	1	2	3	4	5	التحرير قد يؤتر على مكونات محددة من تدفقات رؤوس الأموال بما في ذلك الديون (المالية وغير المالية على	B.29
	borrowing, and lending), equity (portfolio) investment in stock markets), and foreign direct investment		<u> </u>	Ľ		-	الاقتراض والإقراض) والإنصاف (المحفظه) الاستثمار في اسواق الاسهم)، والاستثمار الاجنبي المباتس 	2.22
B.30	Other (Please specify)						اخرى، يرجى التحديد	B.30
D 44						-	ا ـــــــــــــــــــــــــــــــــــ	D. M
B.31	Which of the following factors affect your firm's decisions about issuing shares?	ļ	2	3	4	5	ِ لِلَى أَي دَرِجَهُ مَنْ الْعُوْامَلُ الْتَالَيَّهُ نَوْشَرُ عَلَى قُرْارَاتَكَ الْحَاسَمَةُ حَوْلَ إِصَدَارَ الأسْهُمَ؟ 	B.31
	Please tick the box that best reflects your answe	er whe	ere:		لايمة	جدره	الرجاع وضع إشارة على الرقم الا	
<u> </u>	1 = Strongly disagree لا اوقاق بسد 2 = Disagree لا اوقاق بسد 3 = iNeither age	ee noi	r disa	igree	محايد	2	او فق بسده S = Strongly agree او فق 4 = Agree	
Code	Statements	Scol	re	4	الشيج	-	البيانات	رت ۲۵۱۱
B.31.1	Product environment	1	2	3	4	5	حمايه بيبه المنتج	B.31.1
B.31.2	Whether recent profits have been sufficient to fund activities	1	2	3	4	5	الارباح الاخيرة كانت كافية لتمويل التشاطات	B.31.2
B.31.3	Shares are the cheapest source of funds	1	2	3	4	5	الاسهم ارخص مصدر الاموال	B.31.3
B.31.4	Shares are the least risky source of funds	1	2	3	4	5	الاسهم اقل مخاطرة لمصدر الاموال	B.31.4
B.31.5	Issuing shares gives investors a better impression of a firm's prospects than using debt	1	2	3	4	5	يعطي إصدار الأسهم المستثمرين انطباع راسخ عند احتمالات استخدام السحب	B.31.5
B.31.6	Fulfilling legal requirements regarding capital	1	2	3	4	5	الوفاء بالمتطلبات القانونية فيما يتعلق بالراسمال	B.31.6
B.31.7	Other (Please specify)						الخرى، يرجى التحديد	B.31.7
						سيهم:	نف البيانات التالية من 1 الـ 5، حيث 1 حدًا عن مقمة م5 مقمة حدًا، بخصوص في إز اتك الحاسمة حمل اصدار الأ	
	Please tick the box that best reflects your answ 1 = Strongly disagree لا أوفق بشد ع Strongly disagree الا أوفق بشد ع Strongly disagree	er whe	e re: r disa	Igree	۲ ملائمة حايد	سهم: أكثر م م	ف البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص فراراتك الحاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الأ أوفق بشدة 5 = Strongly agree أوفق ع	الرجاع للك
Code	Please tick the box that best reflects your answer 1 = Strongly disagree لا أو فق بشد 3 = Neither age Statements	er whe ee not	e re: r disa re	igree لغ	ر للائمة حايد النتيج	سیهم: اکثر م م	ف البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص فراراتك الحاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الأ أوفق بشدة 5 = Strongly agree أوفق 4 = Agree	، ربع عد ر.ت
Code B.32.1	Please tick the box that best reflects your answer 1 = Strongly disagree ۲ 2 = Disagree ۲ 3 = Neither age Statements To increase stock market efficiency and reduce wastage	er whe ree nor Scor 1	ere: r disa re 2	igree ž	بلائمة للائمة حايد النتيج 4	سبهم: أكثر م م	ف البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص فراراتك الحاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الا أوفق بشدة 5 = Strongly agree أوفق على المقالة البياتات لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد	،ربو عمر ر.ت B.32.1
Code B.32.1 B.32.2	Please tick the box that best reflects your answer 1 = Strongly disagree ٧ 2 = Disagree ٧ 3 = Neither agr Statements To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective	er whe ee nor Scor 1	ere: r disa re 2 2	igree 3 3 3	لائمة حايد النتيج 4 4	استهم؟ أكثر م 5 5	ف البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص فراراتك الحاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الا أوفق بشدة S = Strongly agree أوفق على الميتات البياتات لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد للحفاظ على زيادة حصتها في السوق لإرضاء الزبائن المحتملين	ريت B.32.1 B.32.2
Code B.32.1 B.32.2 B.32.3	Please tick the box that best reflects your answer 1 = Strongly disagree ٧ 3 = Neither age 3 = Neither age 3 = Neither age Statements To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector	er where nor Scor 1 1 1	ere: r disa re 2 2 2	igree 3 3 3 3	لائمة حايد النتيج 4 4	اسهم؟ اکثر م 5 5 5	ف البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص فراراتك الحاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الا البياتات الاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد للحفاظ على زيادة حصتها في السوق لإرضاء الزبائن المحتملين لتعزيز العلاقة بين القطاعين العام والخاص	ر.ت B.32.1 B.32.2 B.32.3
Code B.32.1 B.32.2 B.32.3 B.32.4	Please tick the box that best reflects your answer 1 = Strongly disagree ۲ 2 = Disagree ۲ 3 = Neither age Statements To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation	er whe ee not Scot 1 1 1 1	ere: r disa re 2 2 2 2 2	agree 3 3 3 3 3	لائمة حايد النتيج 4 4 4	سیهم: کثر م 5 5 5 5	ف البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص فراراتك الحاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الا البياتات لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد للحفاظ على زيادة حصتها في السوق لإرضاء الزبائن المحتملين لتعزيز العلاقة بين القطاعين العام والخاص لإضافة القوة إلى سمعة الشركة	ريت B.32.1 B.32.2 B.32.3 B.32.4
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5	Please tick the box that best reflects your answer 1 = Strongly disagree ۲ 2 = Disagree ۲ 3 = Neither age To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To improve stock market service quality	er whe ee nor Scor 1 1 1 1 1 1	ere: r disa re 2 2 2 2 2 2 2 2	agree 3 3 3 3 3 3	للائمة حايد النتيج 4 4 4 4 4	سهم: کثر م 5 5 5 5 5	ف البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص فراراتك الحاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الأ الوفق بشدة 5 = Strongly agree أوفق عملة علي الرجاء لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد للحفاظ على زيادة حصتها في السوق لإرضاء الزبائن المحتملين لتمزيز العلاقة بين القطاعين العام والخاص لإضافة القوة إلى سمعة الشركة لتحسين نوعية خدمة سوق الأسهم المالية	<u>تريت</u> B.32.1 B.32.2 B.32.3 B.32.4 B.32.5
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6	Please tick the box that best reflects your answer 1 = Strongly disagree ۲ 2 = Disagree ۲ 3 = Neither age To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To improve stock market service quality To encourage quality awareness within stock markets	er whe ee nor Scor 1 1 1 1 1 1 1 1	ere: r disa re 2 2 2 2 2 2 2 2 2 2 2	agree 3 3 3 3 3 3 3 3 3	للائمة حايد حايد 4 4 4 4 4 4 4	سهم: ککثر م 5 5 5 5 5 5	ف البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص فراراتك الحاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الأ الوفق بشدة 5 = Strongly agree أوفق عملة علي الرقم الأ لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد للحفاظ على زيادة حصتها في السوق لإرضاء الزبائن المحتملين لتمزيز العلاقة بين القطاعين العام والخاص لإضافة القوة إلى سمعة الشركة لتحسين نوعية خدمة سوق الأسهم المالية لتشجيع الوعي الممتاز ضمن أسواق الأسهم المالية	<u>تريت</u> B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7	Please tick the box that best reflects your answer 1 = Strongly disagree ٧ 3 = Neither age Statements To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To encourage quality awareness within stock markets Other (Please specify) Other (Please specify)	er whee ee nor 1 1 1 1 1 1 1 1	ere: r disa re 2 2 2 2 2 2 2 2 2 2 2 2 2	agree 3 3 3 3 3 3 3 3	للائمة حايد النتيج 4 4 4 4 4 4 4	سیهم: کثر م 5 5 5 5 5 5	ف البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص فراراتك الحاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الأ الوفق بشدة S = Strongly agree أوفق عملة على الرقم الأ لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد للحفاظ على زيادة حصتها فى السوق لإرضاء الزبائن المحتملين لتفزيز العلاقة بين القطاعين العام والخاص لإضافة القوة إلى سمعة الشركة لتتميين نوعية خدمة سوق الأسهم المالية أخرى، يرجى التحديد	تريت B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7	Please tick the box that best reflects your answer 1 = Strongly disagree ٧ 3 = Neither age Statements To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To encourage quality awareness within stock markets Other (Please specify)	er whee ee nor Scor 1 1 1 1 1 1 1 1	ere: r disa re 2 2 2 2 2 2 2 2 2 2 2	igree 3 3 3 3 3 3 3 3	لانمة حايد النتيج 4 4 4 4 4 4	سهم: کثر م 5 5 5 5 5 5 5	ف البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص فراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الأ الوفق بشدة S = Strongly agree أوفق علي المقا لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد للحفاظ على زيادة حصتها في السوق لإرضاء الزبائن المحتملين لتعزيز العلاقة بين القطاعين العام والخاص لإضافة القوة إلى سمعة الشركة لتصبين نوعية خدمة سوق الأسهم المالية أخرى، يرجى التحديد	<u>ترين</u> B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7	Please tick the box that best reflects your answer 1 = Strongly disagree لا لوفق بشد 2 = Disagree ع لا لوفق بشد 3 = Neither agr Statements To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To improve stock market service quality To encourage quality awareness within stock markets Other (Please specify) Section C: Macro-economic Reform	er whee ee nor 1 1 1 1 1 1 1 1 1	reisa reisa 2 2 2 2 2 2 2 2 2 2	agree 3 3 3 3 3 3 3 3 3 3 3 3 3	للائمة حايد 4 4 4 4 4 4 4	سهم: کثر م 5 5 5 5 5 5	عب البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص فراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الأ الوفق بشدة S = Strongly agree أوفق عامة عليه الميانة لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد الخطاط على زيادة حصتها في السوق لإرضاء الزبائن المحتملين التعزيز العلاقة القوة إلى سمعة الشركة لتصبين نوعية خدمة سوق الأسهم المالية أخرى، يرجى التحديد القسم الثالث: الإصلاح الاقتصاد الكلي	<u> </u>
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 The pur	Please tick the box that best reflects your answer 1 = Strongly disagree ٧ 3 = Neither age Statements To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To encourage quality awareness within stock markets Other (Please specify) Section C: Macro-economic Reform pose of this section is to test for structural change in environmental conditions in macro-economic reform	r whe ee nor Scor 1 1 1 1 1 1 1 1 1 1	ere: r disa re 2 2 2 2 2 2 2 2 2 2 2 2 2 7 2	agree 3 3 3 3 3 3 4 6 6 6 6 6 6 7 7 7 7 7 7 7 7 7 7 7 7 7	للائمة حايد 4 4 4 4 4 4 4 4 4 4 8	مسهم: اکثر م 5 5 5 5 5 5 5 5 5 5	عب البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص فراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الأ الوفق بشدة S = Strongly agree أوفق علي المقالا لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد لتعزيز العلاقة بين القطاعين العام والخاص لإضافة القوة إلى سمعة الشركة لتشجيع الوعي الممتاز ضمن أسواق الأسهم المالية أخرى، يرجى التحديد القسم الثالث: الإصلاح الإقتصاد الكلي r opinion about the macro-economic reform activities in financial market performan	<u> </u>
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 The pur	Please tick the box that best reflects your answer 1 = Strongly disagree ٧ 3 = Neither age Statements To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To encourage quality awareness within stock markets Other (Please specify) Section C: Macro-economic Reform pose of this section is to test for structural change in environmental conditions in macro-economic reform Utilidation in the section is to test for structural change in environmental conditions in macro-economic reform	r whe ee nor <u>Scor</u> 1 1 1 1 1 1 1 1 1	re: re: 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	agree 3 3 3 3 3 3 3 3 4 5 6 6 1 1 1 1 1 1 1 1 1 1 1 1 1	للائمة حايد 4 4 4 4 4 4 4 4 2	مسهم: (کفتر م 5 5 5 5 5 5 5 5 5 5 5 5 9 9 9 9 9 9 9	عب البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمه و5 مهمه جدا، بخصوص فراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الأ الوفق بشدة S = Strongly agree أوفق عليان المقائد لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد لتعزيز العلاقة بين القطاعين العام والخاص لإضافة القوة إلى سمعة الشركة لتشجيع الوعي الممتاز ضمن أسواق الأسهم المالية أخرى، يرجى التحديد القسم الثالث: الإصلاح الاقتصاد الكلي منا القسم هو اختبار العوامل التي تؤثر في التغيير الهيكلي في الظروف ألبنبه على أداء متغيرات الإصلاح الاقتصاد الاقتصاد الكلي ن هذا القسم هو اختبار العوامل التي تؤثر في التغيير الهيكلي في الظروف ألبنبه على أداء متغيرات الإصلاح الاقتصاد الكلي	ربت B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 nce. الغرض مز
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 The pur	Please tick the box that best reflects your answer 1 = Strongly disagree ٧ 3 = Neither age Statements To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To encourage quality awareness within stock markets Other (Please specify) Section C: Macro-economic Reform pose of this section is to test for structural change in environmental conditions in macro-economic reform 0 things be again of the box that best reflects your answer	r whe ee noi 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	ere: r disa re 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	ي ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت	للانمة حايد التتيج 4 4 4 4 4 4 4 4 4 4 9 1 2 الر	مسهم: (کثر م 5 5 5 5 5 5 5 5 5 5 5 5 5	عب البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص قراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الأ الوفق بشدة S = Strongly agree أوفق علي المقات الستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد الحفاظ على زيادة حصتها في السوق لإرضاء الزبائن المحتملين التعزيز العلاقة بين القطاعين العام والخاص لإضافة القوة إلى سمعة الشركة التصرين نوعية خدمة سوق الأسهم المالية التشجيع الوعي الممتاز ضمن أسواق الأسهم المالية أخرى، يرجى التحديد القسم الثالث: الاصلاح الاقتصاد الكلي ropinion about the macro-economic reform activities in financial market performat ن هذا القسم هو اختبار العوامل التي تؤثر في التغيير الهيكلي في الظروف ألبنبه على أداء متغير ات الاصلاح الاقتصاد الرجاء وضع بأشارة على الرقم الأ	ريت B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 nce.
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 The pur	Please tick the box that best reflects your answer 1 = Strongly disagree ٧ 3 = Neither age Statements To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To encourage quality awareness within stock markets Other (Please specify) Section C: Macro-economic Reform pose of this section is to test for structural change in environmental conditions in macro-economic reform. 0 things be like the box that best reflects your answer 1 = Strongly disagree ٧ legio the box that best reflects your answer	r whe ee nor Scor 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	re: 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	igree 3 3 3 3 3 3 4 ease Eleise classe	للائمة حايد للتتيج 4 4 4 4 4 4 4 4 4 4 4 4 4 4 2 . الر مات	مسهم: أكثر م م 5 5 5 5 5 5 5 5 5 5 5 2 4 2 شر م	عب البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص قراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الا السيالية الا لاستكمال كفاءة سوق الأوراق المالية و تقليل الفاقد للحفاظ على زيادة حصتها في السوق لإرضاء الزبائن المحتملين لتعزيز العلاقة بين القطاعين العام والخاص لإضافة القوة إلى سمعة الشركة تتشبيع الوعي الممتاز ضمن أسواق الأسهم المالية أخرى، يرجى التحديد القسم الثالث: الاصلاح الاقتصاد الكلي م المحتم حصق إشارة على الرقم الأ ن هذا القسم هو اختبار العوامل التي تؤثر في التغبير اليويكلي في الظروف ألبنبه على أداء متغير ات الاصلاح الاقتصاد ن هذا القسم هو اختبار العوامل التي تؤثر في التغبير اليويكلي في الظروف ألبنبه على أداء متغير ات الاصلاح الاقتصاد الرجاء وضع إشارة على الرقم الأ وفق بشدة Strongly agree عالم عارجة على الرقم الأ	ريت B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 nce.
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 The pur	Please tick the box that best reflects your answer 1 = Strongly disagree ٧ 3 = Neither age Statements To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To encourage quality awareness within stock markets Other (Please specify) Section C: Macro-economic Reform pose of this section is to test for structural change in environmental conditions in macro-economic reform. 0 timeduc 1 = Strongly disagree ٧ legio unic. 1 = Strongly disagree ٧ legio unic. 3 = Neithe Statements	er whe ee nor <u>Scor</u> 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	re: 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	gree	للائمة حايد 4 4 4 4 4 4 4 4 4 4 4 4 5 1 1 1 1 1 1 1	السميم أكثر م 5 5 5 5 5 5 5 5 5 5 5 5 2 7 2 2 2 2 2 2	عب البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص قراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الا لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد للحفاظ على زيادة حصتها في السوق لإرضاء الزبائن المحتملين لإضافة القوة إلى سمعة الشركة لإضافة القوة إلى سمعة الشركة التصبين نوعية خدمة سوق الأسهم المالية التشجيع الوعي الممتاز ضمن أسواق الأسهم المالية أخرى، يرجى التحديد القسم الثالث: الإصلاح الاقتصاد الكلي من من القسم هو اختبار العوامل التي تؤنثر في التغيير اليويكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد ن هذا القسم هو اختبار العوامل التي تؤنثر في التغيير اليويكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد الرجاء وضع إشارة على الرقم الأن الوقق بشدة S = Strongly agree على أداء متغيرات الاصلاح الاقتصاد البياتات	ر بت B.32.1 B.32.2 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 The pur	Please tick the box that best reflects your answer 1 = Strongly disagree ٢ ٤	r whe ee nor <u>Scor</u> 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	ere: r disa re 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	gree	حايد حايد 4 4 4 4 4 4 4 4 4 4 4 5 9 9 9 9 9 9 9 1 1 1 1 1 1 1 1 1 1 1 1	العشر م اكثر م م 5 5 5 5 5 5 5 5 5 5 5 5 2 2 2 2 4 2 4 2	عب البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص قراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الا الاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد التعزيز العلاقة بين القطاعين العام والخاص لإضافة القوة إلى سمعة الشركة التصريز نوع خدمة سوق الأسهم المالية التشجيع الوعي الممتاز ضمن أسواق الأسهم المالية أخرى، يرجى التحديد القسم الثالث: الاصلاح الاقتصاد الكلي رهذا القسم هو اختبار العوامل التي تؤثر في التغريلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد الكلي ن هذا القسم هو اختبار العوامل التي تؤثر في التغيير اليويكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد الرجاء وضع إشارة على الرقم الأ المعم القلدة القوة الي العوامل التي تؤثر في التغيير اليويكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد المعم المواق بشدة على المقار على المقارة على الرواق الأسهم المالية الرجاء وضع إشارة على الرقم الأ المعار الفائدة الفقة النبية على أداء متغيرات الاصلاح الاقتصاد الكلي الرجاء وضع إشارة على الرقم الأ	ريت B.32.1 B.32.2 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 The pur Code C.1	Please tick the box that best reflects your answer 1 = Strongly disagree ٧ 3 = Neither age Statements 3 3 To increase stock market efficiency and reduce wastage 3 3 To maintain increased market share to satisfy customer perspective 3 3 To strengthen the relationship between public and private sector 5 5 To add strength to the company's reputation 7 5 5 To encourage quality awareness within stock markets 0 5 5 Other (Please specify) 5 5 5 5 Section C: Macro-economic Reform 5 5 5 5 pose of this section is to test for structural change in environmental conditions in macro-economic reform 5 5 5 Utidatic loud: 1 2 2 1 3 3 1 1 Statements The policy of interest rate	r whe ee nor Scor 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	re: r disa re 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	gree 3 3 3 3 3 3 3 3 3 3 3 3 4 4 5 6 6 6 7 7 7 7 7 7 7 7 7 7 7 7 7	للانمة حايد 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	مسهم: محتر م ح ح ع محتر م محتر م ح اید الکل الکل محتر م محتر م	عب البيانات التالية من 1 إلى 5، حيث 1 جدا عير مهمة و5 مهمة جدا، بخصوص قراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الا لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد للحفاظ على زيادة حصتها في السوق لإرضاء الزبائن المحتملين لإضافة القوة إلى سمعة الشركة لاضافة القوة إلى سمعة الشركة التصبين نو عية خدمة سوق الأسهم المالية التشريع الو عي الممتاز ضمن أسواق الأسهم المالية التشريع الو عي الممتاز ضمن أسواق الأسهم المالية التشريع الو عي الممتاز ضمن أسواق الأسهم المالية القسم الثالث: الاصلاح الاقتصاد الكلي د هذا القسم هو اختبار العوامل التي تؤثر في التغيير اليوكلي في الظروف ألبنبه على أداء متغير ات الاصلاح الاقتصاد ن هذا القسم هو اختبار العوامل التي تؤثر في التغيير اليوكلي في الظروف ألبنبه على أداء متغير ات الاصلاح الاقتصاد ن هذا القسم هو اختبار العوامل التي تؤثر في التغيير اليوكلي في الظروف ألبنبه على أداء متغير ات الاصلاح الاقتصاد الموق بشارة على الرقم الأ م الرجاء وضع إشارة على الرقم الأ م الرجاء وضع المؤلز على الرقم الأ سياسة سعر الفائدة	ر.ت B.32.1 B.32.2 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 The pur Code C.1 C.2	Please tick the box that best reflects your answer 1 = Strongly disagree ۲ 2 = Disagree 9 3 = Neither age To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective	r whe ee nor Scor 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	re: r disa re 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	gree 3 3 3 3 3 3 3 3 3 3 4 4 5 6 6 7 7 7 7 7 7 7 7 7 7 7 7 7	للائمة التتيج - حايد - 4 - 4 - 4 - 4 - 4 - 4 - 4 - 4 - 4 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1	مسهم: أكثر م 5 5 5 5 5 5 5 5 5 5 5 2 2 2 2 2 2 2 3 2 5 5 5 5	عب البيانات التالية من 1 إلى 5، حيث 1 جذا عير مهمة و5 مهمة جذا، بخصوص قراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الا الوفق بشدة S = Strongly agree اليباتات السنكمال كفاءة سوق الأوراق المالية وتقليل الفاقد للحفاظ على زيادة حصتها في السوق لإرضاء الزبائن المحتملين لإضافة القوة إلى سمعة الشركة لإضافة القوة إلى سمعة الشركة التصبين نو عية خدمة سوق الأسهم المالية التشجيع الو عي الممتاز ضمن أسواق الأسهم المالية التشجيع الو عي الممتاز ضمن أسواق الأسهم المالية التشجيع الو عي الممتاز ضمن أسواق الأسهم المالية القسم الثالث: الاصلاح الاقتصاد الكلي م الرجاء وضع إشارة على الرقم الا ن هذا القسم هو اختبار العوامل التي تؤثر في التغيير الهيكلي في الظروف ألبنبه على أداء متغير ات الاصلاح الاقتصاد القسم الثالث: الاصلاح الاقتصاد الكلي م الرجاء وضع إشارة على الرقم الأ الميان معاد الله علي المقادة في المقادة المطلوب سيزداد و من ثم ستنخفض أسعار الأسه عند زيادة سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستنخفض أسعار الأسه	تر.ت B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 nce. الغرض من C.1 C.2
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 The pur Code C.1 C.2 C.3	Please tick the box that best reflects your answer 1 = Strongly disagree ٧ 3 = Neither agr To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To encourage quality awareness within stock markets Other (Please specify) Please tick the box that best reflects your answer Section C: Macro-economic Reform Please tick the box that best reflects your answer 0 ther (Please specify) I = Strongly disagree I = Strongly disagree 1 = Strongly disagree V legg under the box that best reflects your answer 1 = Strongly disagree V legg under the box that best reflects your answer 1 = Strongly disagree V legg under the box that best reflects your answer 1 = Strongly disagree V legg under the box that best reflects your answer 1 = Strongly disagree V legg under the section is to test for structural change in environmental conditions in macro-economic reform 0 the four the box that best reflects your answer I = Strongly disagree I = Statements 1 = Strongly disagree V legg under the box that best reflects your answer I = Box under thegg under 1 = Str	r whe ee nor Scor 1 1 1 1 1 1 1 1 1 1 1 1 1 r whe scor Scor 1 1 1 1	re: r disa re 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	ي ي ي ي ي ي ي ي ي ي ي ي ي ي	للائمة بالتتيج بالتتيج بالتيج بالريمة بالتيج بالتيج بالتيج بالتيج	مسهم: کفتر م 5 5 5 5 5 5 5 5 5 5 5 5 5	عب البيانات التالية من 1 إلى 5، حيث 1 جذا عير مهمة و5 مهمة جذا، بخصوص قراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الا لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد للحفاظ على زيادة حصتها في السوق لإرضاء الزبائن المحتملين لإضافة القوة إلى سمعة الشركة لاضفة القوة إلى سمعة الشركة التصبين نوعية خدمة سوق الأسهم المالية التشجيع الو عي الممتاز ضمن أسواق الأسهم المالية م الرجاء وضع إشارة على الرقم الأن اليجاء وضع إشارة على الرقم الأن البياتات الما الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه يزئر سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه يزئر سعر الفائدة على سوق الأسهم المالية	تر.ت B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 Ince. الغرض مز C.1 C.2 C.3
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 The pur Code C.1 C.2 C.3 C.4	Please tick the box that best reflects your answer 1 = Strongly disagree ٧ أوفق بشد 3 = Neither age To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To increase specify) To encourage quality awareness within stock markets Other (Please specify) Section C: Macro-economic Reform section C: Macro-economic Reform Please tick the box that best reflects your answer 1 = Strongly disagree ٧ أوفق بشد 2 = Disagree 1 = Strongly disagree ٧ أوفق بشد 3 = Neither 1 = Strongly disagree ٧ أوفق بشد 3 = Neither 1 = Strongly disagree ٧ أوفق بشد 3 = Neither 1 = Strongly disagree ٧ أوفق بشد 3 = Neither 1 = Strongly disagree ٧ أوفق بشد 3 = Neither 1 = Strongly disagree ٧ أوفق بشد 3 = Neither The policy of interest rate Y أوفق بشد 3 = Neither The policy of interest rate When interest rates increase, the required rates of return will increase, then stock prices will decrease Interest rates affect the stock market The flow of money depends upon the interest rates either to the stock market or the banking sector <td>r whore score 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1</td> <td>re: r disa re 2 2 2 2 2 2 2 2 2 2 2 2 2</td> <td>ي ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت</td> <td>للائمة حايد 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4</td> <td>مسهم: ککثر م 5 5 5 5 5 5 5 5 5 5 5 5 5</td> <td>عب البيانات التالية من 1 إلى 5، حيث 1 جذا عير مهمة و5 مهمة جذا، بخصوص قراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الا لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد للحفاظ على زيادة حصتها في السوق لإرضاء الزبائن المحتملين لتحزيز العلاقة بين القطاعين العام والخاص لإضافة القوة إلى سمعة الشركة التصبين نوعية خدمة سوق الأسهم المالية التشجيع الوعي الممتاز ضمن أسواق الأسهم المالية التشجيع الوعي الممتاز ضمن أسواق الأسهم المالية التشجيع الوعي الممتاز ضمن أسواق الأسهم المالية م الرجاء وضع إشارة على الزعم الا ن هذا القسم هو اختبار العوامل التي تؤثر في التغيير الهيكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد ن هذا القسم هو اختبار العوامل التي تؤثر في التغيير الهيكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد م الرجاء وضع إشارة على الرقم الأم م الرجاء وضع إشارة على الرقم الأم الا م الرجاء وضع إشارة على الرقم الأم الم سياسة سعر الفائدة م عنر زيادة سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه يزئر سعر الفائدة على سوق الأسهم المالية يزئر سعر الفائدة على سوق الأسهم المالية البياع المصر في يؤف على أسعار الفائدة أما على سوق الأسهم المالية أو القطاع المصر في يزئر سعر الفائدة من معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه يزئر سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه يزئر سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه يزئر سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه يزئر سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستنخفض أسعار الأسه يزئر سعر الفائدة فان معدل العائدة أما على سوق الأسهم المالية أو القطاع المصر في</td> <td>تر.ت B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 Ince. الغرض مز C.1 C.2 C.3 C.4</td>	r whore score 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	re: r disa re 2 2 2 2 2 2 2 2 2 2 2 2 2	ي ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت	للائمة حايد 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	مسهم: ککثر م 5 5 5 5 5 5 5 5 5 5 5 5 5	عب البيانات التالية من 1 إلى 5، حيث 1 جذا عير مهمة و5 مهمة جذا، بخصوص قراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الا لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد للحفاظ على زيادة حصتها في السوق لإرضاء الزبائن المحتملين لتحزيز العلاقة بين القطاعين العام والخاص لإضافة القوة إلى سمعة الشركة التصبين نوعية خدمة سوق الأسهم المالية التشجيع الوعي الممتاز ضمن أسواق الأسهم المالية التشجيع الوعي الممتاز ضمن أسواق الأسهم المالية التشجيع الوعي الممتاز ضمن أسواق الأسهم المالية م الرجاء وضع إشارة على الزعم الا ن هذا القسم هو اختبار العوامل التي تؤثر في التغيير الهيكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد ن هذا القسم هو اختبار العوامل التي تؤثر في التغيير الهيكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد م الرجاء وضع إشارة على الرقم الأم م الرجاء وضع إشارة على الرقم الأم الا م الرجاء وضع إشارة على الرقم الأم الم سياسة سعر الفائدة م عنر زيادة سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه يزئر سعر الفائدة على سوق الأسهم المالية يزئر سعر الفائدة على سوق الأسهم المالية البياع المصر في يؤف على أسعار الفائدة أما على سوق الأسهم المالية أو القطاع المصر في يزئر سعر الفائدة من معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه يزئر سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه يزئر سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه يزئر سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه يزئر سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستنخفض أسعار الأسه يزئر سعر الفائدة فان معدل العائدة أما على سوق الأسهم المالية أو القطاع المصر في	تر.ت B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 Ince. الغرض مز C.1 C.2 C.3 C.4
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 The pur Code C.1 C.2 C.3 C.4 C.5	Please tick the box that best reflects your answer 1 = Strongly disagree Y 3 = Neither age Statements To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To increase stock market service quality To encourage quality awareness within stock markets Other (Please specify) Section C: Macro-economic Reform Please tick the box that best reflects your answer 0 this section is to test for structural change in environmental conditions in macro-economic reform Output: Please tick the box that best reflects your answer 1 = Strongly disagree Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y <td colspan="2</td> <td>r whet ee nor 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1</td> <td>re: r disa re 2 2 2 2 2 2 2 2 2 2 2 2 2</td> <td>ي جاء ارا طisagget عجاء ارا عرباء ارا عرباع ارا مرام ار عرباع ارا مرام ار عرباع ارا مرام ار مرام ار عرباع ار عرباع ار عرباع ار عرباع ار عرباع الم الم الم الم الم الم الم الم الم الم</td> <td>للائمة حايد حايد 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4</td> <td>مسهم: محتر م ح ح you کوتر م کوتر م ح اید الکل ح ی 5 5 5 5 5 5 5 5 5 5 5 5 5</td> <td>للبيانات التالية من 1 إلى 5، حيث 1 جذا عير مهمة و5 مهمة جذا، بخصوص قراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الا السنكمال كفاءة سوق الأوراق المالية وتقليل الفاقد لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد لإضافة القوة إلى سمعة الشركة لإضافة القوة إلى سمعة الشركة التحبين نوعية خدمة سوق الأسهم المالية التسبيع الوعي الممتاز ضمن أسواق الأسهم المالية القسم الثالث: الإصلاح الاقتصاد الكلي فذا القسم هو اختبار العوامل التي تؤثر في التغيير الييكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد ن منا القسم الثالث: الإصلاح الاقتصاد الكلي فذا القسم هو اختبار العوامل التي تؤثر في التغيير الييكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد ن منا القسم هو اختبار العوامل التي تؤثر في التغيير الييكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد الكلي م الرجاء وضع إشارة على الرقم الا القسم الثالث: الاصلاح الاقتصاد الكلي م الرجاء وضع إشارة على الرقم الا القسم الثالث الم الي تؤثر في التغيير الييكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد م الرجاء وضع إشارة على الرقم الا م الرجاء وضع إشارة على الرقم الا م الرجاء وضع إشارة على الرقم الا م الرجاء وضع إشارة على المقارة الاقصل م الرجاء وضع إشارة على الرقم الا م الرجاء وضع إشارة على الرقم الا م الرجاء وضع إشارة على الرقم الا م الرجاء وضع إشارة على المقارة المقالية المقارة على الرقم الا م الرجاء وضع إشارة على الرقم الا م الرجاء وضع إشارة على المقادة المقالو من ثم ستتخفض أسعار الأسه م البية سعر الفائدة ال معل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه ت مزيراد معر الفائدة على سوق الأسهم المالية م ترير سعر الفائدة على سوق الأسهم المالية أو القطاع المصر في م ترير سعر الفائدة على أسار الفائدة أما على سوق الأسهم المالية أو القطاع المصر في م ترير سعر الفائدة الفاسم الفائدة أما على سوق الأسهم المالية أو القطاع المصر في المندر ال</td> <td>تربت بر بت B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 C.1 C.1 C.2 C.3 C.4 C.5</td>	r whet ee nor 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	re: r disa re 2 2 2 2 2 2 2 2 2 2 2 2 2	ي جاء ارا طisagget عجاء ارا عرباء ارا عرباع ارا مرام ار عرباع ارا مرام ار عرباع ارا مرام ار مرام ار عرباع ار عرباع ار عرباع ار عرباع ار عرباع الم	للائمة حايد حايد 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	مسهم: محتر م ح ح you کوتر م کوتر م ح اید الکل ح ی 5 5 5 5 5 5 5 5 5 5 5 5 5	للبيانات التالية من 1 إلى 5، حيث 1 جذا عير مهمة و5 مهمة جذا، بخصوص قراراتك الخاسمة حول إصدار الا الرجاء وضع إشارة على الرقم الا السنكمال كفاءة سوق الأوراق المالية وتقليل الفاقد لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد لإضافة القوة إلى سمعة الشركة لإضافة القوة إلى سمعة الشركة التحبين نوعية خدمة سوق الأسهم المالية التسبيع الوعي الممتاز ضمن أسواق الأسهم المالية القسم الثالث: الإصلاح الاقتصاد الكلي فذا القسم هو اختبار العوامل التي تؤثر في التغيير الييكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد ن منا القسم الثالث: الإصلاح الاقتصاد الكلي فذا القسم هو اختبار العوامل التي تؤثر في التغيير الييكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد ن منا القسم هو اختبار العوامل التي تؤثر في التغيير الييكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد الكلي م الرجاء وضع إشارة على الرقم الا القسم الثالث: الاصلاح الاقتصاد الكلي م الرجاء وضع إشارة على الرقم الا القسم الثالث الم الي تؤثر في التغيير الييكلي في الظروف ألبنبه على أداء متغيرات الاصلاح الاقتصاد م الرجاء وضع إشارة على الرقم الا م الرجاء وضع إشارة على الرقم الا م الرجاء وضع إشارة على الرقم الا م الرجاء وضع إشارة على المقارة الاقصل م الرجاء وضع إشارة على الرقم الا م الرجاء وضع إشارة على الرقم الا م الرجاء وضع إشارة على الرقم الا م الرجاء وضع إشارة على المقارة المقالية المقارة على الرقم الا م الرجاء وضع إشارة على الرقم الا م الرجاء وضع إشارة على المقادة المقالو من ثم ستتخفض أسعار الأسه م البية سعر الفائدة ال معل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه ت مزيراد معر الفائدة على سوق الأسهم المالية م ترير سعر الفائدة على سوق الأسهم المالية أو القطاع المصر في م ترير سعر الفائدة على أسار الفائدة أما على سوق الأسهم المالية أو القطاع المصر في م ترير سعر الفائدة الفاسم الفائدة أما على سوق الأسهم المالية أو القطاع المصر في المندر ال	تربت بر بت B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 C.1 C.1 C.2 C.3 C.4 C.5
Code B.32.1 B.32.2 B.32.3 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 The purp Code C.1 C.2 C.3 C.4 C.5	Please tick the box that best reflects your answer 1 = Strongly disagree ٧ 3 = Neither age Statements To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To increase quality awareness within stock markets Other (Please specify) Section C: Macro-economic Reform Please tick the box that best reflects your answer Other (Please specify) Section C: Macro-economic Reform pose of this section is to test for structural change in environmental conditions in macro-economic reform Other (Please specify) Section C: Macro-economic Reform Dose of this section is to test for structural change in environmental conditions in macro-economic reform Output: 1 = Strongly disagree ٧ 3 = Neithe Disagree 9 3 = Neithe Statements Interest Rates The policy of interest rate When interest rates increase, the required r	r where e norms	re: r disaa re 2 2 2 2 2 2 2 2 2 2 2 2 2	ي ي ي ي ي ي ي ي ي ي ي ي ي ي	للائمة للائمة للتنيج للائمة للائمة rree للائمة للائمة للائمة للائمة للائمة	مسهم: المتر م المتر م المت المتر م المتر م المت م المت م المتر م المت المتر م المتر م المتر م المتر م المتر م المتر م المتر م المتر م المت م المتر م المتر م المتر م المتر م المت م المتر م المت م المت م المت م المتر م المتم م المتر م المتر م المتم م الم المتم م المتم م الم المم م الم الم المم م المم م المم المم مم المم مم المم مم المم مم المم مم المم مم الم	عب البيانات التالية من 1 إلى 5، حيث 1 جذا عير مهمة و5 مهمة جذا، بخصوص قراراتك الخاسمة حول إصادار الا الرجاء وضع إشارة على الرقم الا لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد لإضافة القوة إلى سمعة الشركة لاضافة القوة إلى سمعة الشركة لتحبين نوعية خدمة سوق الأسهم المالية القسم الثالث: الإصلاح الاقتصاد الكلي فرى، يرجى التحديد القسم الثالث: الإصلاح الاقتصاد الكلي فرى يرجى التحديد القسم الثالث: الإصلاح الاقتصاد الكلي في الظروف ألبنبه على أداء متغير ات الإصلاح الاقتصاد القسم الثالث المعاد القسم هو اختبار العوامل التي تؤثر في التغيير الييكلي في الظروف ألبنبه على أداء متغير ات الإصلاح الاقتصاد القسم الثالث المعاد القصاد الكلي في القلوة على الموالي المعاد الكلي القسم الثالث المعاد القصاد الكلي عدر إيادة سعر الفائدة من معدل العائد المعاد الموالي البيانات سعار الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستنخفض أسعار الأسه سعار الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستنخفض أسعار الأسه يزئر سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستنخفض أسعار الأسه تحرير سعر الفائدة من معدل العائدة أما على سوق الأسهم المالية أو القطاع المصر في تحرير سعر الفائدة المعاد الفائدة أما على سوق الأسهم المالية أو القطاع المصر في الندة تحرير سعر الفائدة المعاد الفائدة أما على سوق الأسهم المالية أو القطاع المصر في إلى الماد الفائدة أما على سوق الأسهم المالية أو القطاع المصر في المور في الائمي الفائدة أما على سوق الأسهم المالية أو القطاع المصر في الموائدة أما على سوق الأسهم المالية أو القطاع المصر في أو الفائدة أما على سوق الأسهم المالية أو القطاع المصر في أولي الفائدة أما على سوق الأسهم المالية أو القطاع المصر في المور أول المؤائدة أما على سوق الأسهم المالية أو القطاع المصر في أوليا الموائد المولي المور اليؤلي المور اليؤلي المور اليونا المائد المولي الموال المائد المور المونا الموال الفائد أمالي أول المور الموال الموال يؤف على أساد	تريت بريت B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 B.32.6 B.32.7 C.1 C.1 C.2 C.3 C.4 C.5
Code B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 The purp Code C.1 C.2 C.3 C.4 C.5	Please tick the box that best reflects your answer 1 = Strongly disagre ۲ أوفق بند Statements To increase stock market efficiency and reduce wastage To maintain increased market share to satisfy customer perspective To strengthen the relationship between public and private sector To add strength to the company's reputation To increase specify) To encourage quality awareness within stock markets Other (Please specify) Section C: Macro-economic Reform section C: Macro-economic Reform J you all a strengt disagree Y 0 ther (Please specify) J Section C: Macro-economic Reform J you all a strengt disagree Y 0 ther (Please tick the box that best reflects your answer 1 = Strongly disagree Y 0 Y 3 = Neither Statements Y 1 = Strongly disagree Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y Y	r when or score in a	re: 2 2 2 2 2 2 2 2 2 2 2 2 2	ي ا ا ا ا ا ا ا ا ا ا ا ا ا ا ا ا ا ا ا	للائمة حايد التنيج 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	استهم: اعتر م 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	ع البيانات التالية من 1 إلى 5، حيث 1 جذا عير مهمة و5 مهمة جذا، بخصوص قراراتك الخاسمة حول إصادار الا الرجاء وضع إشارة على الرقم الا لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد لاستكمال كفاءة سوق الأوراق المالية وتقليل الفاقد لتعزيز العلاقة بين القطاعين العام والخاص لإضافة القوة إلى سمعة الشركة لتحسين نوعية خدمة سوق الأسهم المالية التسبيع الوعي الممتاز ضمن أسواق الأسهم المالية أخرى، يرجى التحديد القسم الثالث: الإصلاح الاقتصاد الكلي منابع علي أذاء منغيرات الاصلاح الاقتصاد الكلي فران يرجى التحديد القسم الثالث: الإصلاح الاقتصاد الكلي القسم الثالث: الإصلاح الاقتصاد الكلي القسم الثالث: الإصلاح الاقتصاد الكلي عدر القسم الثالث: الإصلاح الاقتصاد الكلي القسم الثالث: الإصلاح الاقتصاد الكلي عدر القسم هو اختبار العوامل التي توثر في التغيير الهيكلي في الظروف ألبنيه على أذاء منغيرات الاصلاح الاقتصاد القسم الثالث: الإصلاح الاقتصاد الكلي القسم الثالث: الإصلاح الاقتصاد الكلي عدر القسم هو اختبار العوامل التي توثر في التغيير الهيكلي في الظروف ألبنيه على أذاء منغيرات الاصلاح الاقتصاد القسم الثالث: الإصلاح الاقتصاد الكلي عدر القسم مو اختبار العوامل التي توثر في التغيير الهيكلي في الظروف ألبنيه على أذاء منغيرات الاصلاح الاقتصاد العمال القلدة على معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه عدر زيادة سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه يوثر سعر الفائدة فان معدل العائد المطلوب سيزداد و من ثم ستتخفض أسعار الأسه تحرير سعر الفائدة ما معدار العائدة أما على سوق الأسهم المالية أو القطاع المصر في تحرير سعر الفائدة	تريت بريت B.32.1 B.32.2 B.32.3 B.32.4 B.32.5 B.32.6 B.32.7 C.1 C.1 C.2 C.3 C.4 C.5

	Exchange Rates						سعار الصرف	l.
C.6	There are no distinct monetary policies in relation to the exchange rate policy	1	2	3	4	5	لا توجد سياسات نقدية واضحة بالنسبة لسياسة أسعار الصرف	C.6
C.7	The movement in exchange rate may alter foreign investors' demand for domestic stock prices compared to other countries	1	2	3	4	5	الحركة في سعر الصرف قد يغيّر مطلب المستثمرين الأجانب لأسعار الأسهم المحلية بالمقارنة مع بلدان أخرى	C.7
C.8	It is important to reduce fluctuations between domestic and foreign currency	1	2	3	4	5	من المهم تخفيض التقابات بين العملة المحلية والأجنبية	C.8
C.9	The exchange rate may influence investment in the local currency market compared with other markets	1	2	3	4	5	سعر الصرف قد يغيّر الاستثمار في سوق العملات المحليّ بالمقارنة مع الأسواق الأخرى	C.9
C.10	Exchange rate fluctuations have influence upon the stock prices	1	2	3	4	5	تقأبات سعر الصرف لها التأثيرات على أسعار الأسهم	C.10
C.11	The exchange rate plays a vital role in investing in the stock market	1	2	3	4	5	يلعب سعر الصرف دورا حيويا للاستثمار في سوق الأسهم المالية	C.11
	Inflation Rates						لتضخم	.1
C.12	When the rate of inflation rises, stock prices fall due to the effect of inflation on the historic cost method of depreciation and taxation of nominal capital gains	1	2	3	4	5	عندما يرتفع معدل التضخم فان أسعار الأسهم تتخفض بسبب تأثير طريقة الكلفة التاريخية للتلف والنظام الضريبي من مكاسب رأس المال الاسمي	C.12
C.13	Any increase in the inflation rate reduces the expected real returns to investors from holding stocks	1	2	3	4	5	أي زيادة في معدل التضخم تحوّل العائدات الحقيقية المتوقعة إلى المستثمرين من حصّة الأسهم	C.13
C.14	The inflation rate impacts strongly on social life	1	2	3	4	5	معدل التصخم يؤثر بقوة على الحياة الاجتماعية	C.14
C.15	Inflation increases the price of goods and services	1	2	3	4	5	التضخم يزيد من أسعار السلع والخدمات	C.15
C.16	Inflation decreases currency value	1	2	3	4	5	التضخم يؤدى إلى انخفاض قيمة العملة	C.16
C.17	Inflation causes unemployment and reduction of job opportunities	1	2	3	4	5	يسبّب التضّخم في زيادة العمالة والحد من فرص العمل	C.17
	Budget Deficit						العجز في الموازنة العامة	
C.18	Budget deficit is considered to be the excess of government expenditure over time	1	2	3	4	5	العجز في الميز انية يعبر عن الزيادة في الإنفاق الحكومي على مر الزمن	C.18
C.19	Reducing the budget deficit overcomes unemployment problems	1	2	3	4	5	خفض العجز في الميز انية يؤدي إلى التغلب على مشاكل البطالة	C.19
C.20	Budget deficit prohibits the increase of public resources	1	2	3	4	5	يمنع العجز في الميز انية زيادة الموارد العامة	C.20
C.21	Budget deficit can lead to increased inflation and interest rates	1	2	3	4	5	عجز الموازنة العامة يمكن أن يؤدي إلى زيادة معدلات التضخم وأسعار الفائدة	C.21
C.22	A low value in budget deficit reduces government spending	1	2	3	4	5	انخفاض قيمة العجز في الموازنة العامة يخفض الإنفاق الحكومي	C.22
C.23	The exploitation of private sector tools contributes to reduction of the budget deficit	1	2	3	4	5	استغلال أدوات القطاع الخاص والتي تساهم في تخفيض العجز في الميزانية	C.23
	Per -capita Income						نصيب الفرد من الدخل	
C.24	Per-capita income can be reflected in development of the Libyan economy	1	2	3	4	5	نصيب الفرد من الدخل يمكن أن ينعكس على تطوير الاقتصاد الليبي	C.24
C.25	Per-capita income is one of the driving forces towards a stock market	1	2	3	4	5	نصيب الفرد من الدخل يمثل إحدى القوى الدافعة نحو سوق الأسهم المالية	C.25
C.26	The human development index is strongly affected by per capita income	1	2	3	4	5	مؤشر التنمية البشرية يؤثر بقوة على معدل دخل الفرد	C.26
C.27	The per-capita income of individuals depends upon population growth	1	2	3	4	5	متوسط الدخل الفردي للافراد يعتمد على النمو السكاني	C.27
C.28	Fiscal discipline determines the per-capita income that satisfies GDP growth rate	1	2	3	4	5	الانضباط المالي يحدد نصيب الفرد من الدخل التي تستوفي من معدل نمو الناتج المحلي الإجمالي	C.28
	Real GDP Growth Rate						الناتج المحلي الإجمالي	
C.29	Higher GDP growth leads to increase in companies' cash flows	1	2	3	4	5	ارتفاع نمو الناتج المحلي الإجمالي يؤدي إلى زيادة التدفقات النقدية في الشركات	C.29
C.30	Real GDP growth rate is necessary to increase output	1	2	3	4	5	معدل نمو الناتج المحلي الإجمالي الحقيقي أمر ضروري لزيادة الناتج	C.30
C.31	When increasing the per-capita of GDP, the Libyan economy must be competitive in the stock market	1	2	3	4	5	عند زيادة نصيب الفر د من الناتج المحلي الإجمالي فان الاقتصاد الليبي يجب أن يكون نتافسي في سوق الأسهم المالية	C.31
C.32	There is a strong growth rate in export and import trade	1	2	3	4	5	هناك أسباب قوية لزيادة معدل النمو في تجارة الصادر ات والوار دات	C.32
C.33	Oil prices impact on GDP of the stock market	1	2	3	4	5	أسعار النفط تؤثر على الناتج المحلي الإجمالي في سوق الأسهم المالية	C.33
C.34	New improved and varied products, goods and services which increased in the economic growth rate	1	2	3	4	5	التحسنات الجديدة والمتنوعة للمنتجات والسلع والخدمات تؤدي إلى ارتفاع في معدل نمو الاقتصاد	C.34
C.35	Growth in GDP reduces unemployment	1	2	3	4	5	النمو في الناتج المحلي الإجمالي يخفّض البطالة	C.35
C.36	Investment return growth is high in their economy	1	2	3	4	5	عائد الاستثمار هو ارتفاع معدل النمو في الاقتصاد	C.36
C.37	Other (Please specify)						أخرى، يرجى التحديد	C.37
C.38	To what extent do you think each of the following statements increase/ decrease the contribution of f economic growth rate?	actor	rs to				في رأيك، إلى إي العوامل الثالية تساهم بالزيادة أو النقصان في معدل النمو الاقتصادي؟	C.38
	Please tick the box that best reflects your answer 1 = Strongly disagree لا لوفق بشد 2 = Disagree لا لوفق بشد	e r wh r agre	ere:	disa	لائمة Igree	ڈکثر ما حاید	الرجاء وضع اشارة على الرقم الا أوفق بشدة 5 = Strongly agree أوفق مشدة 4 = Agree	
Code	Statements	Sco	ore		لنتيجة	ti	البيانات	ر ت
C.38.1	Growth of output	1	2	3	4	5	نمو الإنتاج	C.38.1
C.38.2	Contribution of human capital	1	2	3	4	5	مساهمة رأس المال البشري	C.38.2
C.38.3	Contribution of capital	1	2	3	4	5	مساهمة ر أس المال	C.38.3

C.38.4	Contribution of labour	1	2	3	4	5	مساهمة العمالة	C.38.4
C.38.5	Total factor productivity growth (TFP)	1	2	3	4	5	نمو الانتاجية الكلية لعوامل الانتاج	C.38.5
C.38.6	Other (Please specify)		-	0	•	v	ار جو بيد مي التحديد أخرى، بر جي التحديد	C.38.6
0.0010								0.0010
G 20	When undertaking any form of business activity, how concerned are you about each of the financial ris	sks rei	fer to	fina	ncial	l event	ts as following, which are capable of causing harm to firms?	G 20
C.39	والتي تسبب في اعقة أداء الشركات؟	ِ التالي،	لنحو	لية علم	ث المال	ن الأحداث	عند القيام بأي شكل من أشكال النشاط الاقتصادي، كيف يمكنك معينة كل من المخاطر المالية التي تشير إلى	C.39
	Please tick the box that best reflects your answer	wher	e:		ثمة	لأكثر ملا	الرجاء وضع إشارة على الرقم ال	
1	= Un concerned غير مهتما 2 = Slightly concerned غير مهتما 3 = Moderately concerne	مهتم d				4 = Ve	ery concerned مهتم على الإطلاق 5 = Extremely concerned مهتم جدا	
Code	Statements	Sco	re	٦	النتيج		البيانات	رت
C.39.1	Interest rates	1	2	3	4	5	أسعار الفائدة	C.39.1
C.39.2	Foreign rates	1	2	3	4	5	المعدالات الأجنبية	C.39.2
C.39.3	Commodity prices	1	2	3	4	5	أسعار السلع	C.39.3
C.39.4	Indirect instruments of monetary policy	1	2	3	4	5	أدوات السباسة النقدية الغير مباشرة	C.39.4
C 39 5	Other (Please specify)						أخرى، يرجى التحديد	C.39.5
0.57.5								
	Please rank the following statements there are several problematic issues/ obstacles factors which	might	t influ	ience	e doi	ing bu	siness with Libya as a country/ economy and to rank from 1 to 5, where 1	
C.40	represents no obstacles and 5 major obstacles. Please answer these questions by selecting the option that	at you	ı thin	k bes	st ref	lects y	your opinion.	C.40
	يق الترتيب من 1 إلى 5، حيث 1 لا عقبات و5 عقبات الرئيسية.	، عن طر	الاستله	ن هذه	ابه عز	ضل الاج	هناك عدة مشاكل أو عقبات والتي قد تؤثر في أبرام الصفقات التجارية مع الاقتصاد الليبي. الرجاء اختار افض	
	Please tick the box that best reflects your answer	• wher	re:		لائمه	الاكثر ملا	الرجاء وضع إشارة على الرقم ا	_
	الأتوجد عقبة المسيطة 2 = Minor obstacle لأتوجد عقبة 3 = Moderate	e obsta	ıcle	معتدلة	عقبة		عقبة كبيرة 5 = Major obstacle عقبة	
Code	Statements	Sco	re	à	النتيج		البياثات	رت
C.40.1	Policy instability	1	2	3	4	5	عدم استقرار السياسة	C.40.1
C.40.2	Inefficient government bureaucracy	1	2	3	4	5	عدم كفاءة الحكومة البيروقراطية	C.40.2
C.40.3	Restrictive labour regulations	1	2	3	4	5	التقيديه في التشريعات التنظيمية للعمالة	C.40.3
C.40.4	Inadequate supply of infrastructure	1	2	3	4	5	عدم كفاية الإمدادات من البنية التحتية	C.40.4
C.40.5	Inadequately educated workforce	1	2	3	4	5	نقص في القوى العاملة المتعلمه	C.40.5
C.40.6	Corruption	1	2	3	4	5	الفساد (الإداري أو الاقتصادي أو المالي)	C.40.6
C.40.7	Foreign currency regulations	1	2	3	4	5	أنظمة العملة الأجنبية	C.40.7
C.40.8	Poor work ethic of national labour force	1	2	3	4	5	ضعف أخلاقيات العمل في قوة العمل الوطنية	C.40.8
C.40.9	Government instability/coups	1	2	3	4	5	عدم الاستقرار الحكومي/الانقلابات	C.40.9
C.40.10	Tax regulation	1	2	3	4	5	التشريعات الضريبية	C.40.10
C.40.11	Crime and theft	1	2	3	4	5	الجريمة والسرقة	C.40.11
C.40.12	Tax rates	1	2	3	4	5	المعدلات الضريبية	C.40.12
C.40.13	Access to financing	1	2	3	4	5	الحصول على التمويل	C.40.13
C.40.14	Voice and Accountability	1	2	3	4	5	المشاركة والمساءلة	C.40.14
C.40.15	Other (Please specify)						أخرى، يرجى التحديد	C.40.15
	Section D: Financial Maturity						القسم الرابع: النضج المالي	
This section	on aims to examine the change in financial maturity factors which link stock markets, banks and corpo	rate fi	nanc	ing d	ecisi	ons wi	ith an understanding of the relationship between financial patterns and econom	nic growth
rate.								
	علاقة بين الأنماط المالية ومعدل النمو الاقتصادي.	، بفهم ال	قرارات	موّل ال	التي ت	لشركات	سم لدراسة التغيير في النضج المالي بناء على العوامل التي تربط بين أسواق الأوراق المالية والمصارف وال	يهدف هذا الق
	Please tick the box that best reflects your answe	er whe	ere:		ملائمة	م الأكثر ه	الرجاء وضع إشارة على الرقم	
	1 = Strongly disagree لا أوفق بشد 2 = Disagree 3 = Neithe	r agree	e nor	disag	ree	محايد	أوفق بشدة 5 = Strongly agree أوفق 4 = Agree	
Code	Statements	Sco	re	ä	النتيج		البيانات	رت
B	anking Assets Ratio of GDP						سبة الأصول المصرفية من الناتج المحلي الإجمالي	<u>نا</u>
D.1	Deposit money bank assets/total banking sector assets	1	2	3	4	5	داع أصول أموال المصارف إلى مجموع أصول القطاع المصرفي	J D.1
D.2	Banking assets of GDP can improve growth rate in the stock market	1	2	3	4	5	لأصول المصرفية للناتج المحلي الإجمالي يمكن أن تحسّن معدل النمو في سوق الأسهم المالية	// D.2
D.3	Financial intermediaries get larger measured by total assets relative to GDP	1	2	3	4	5	بقاس الوسطاء الماليون بكبر مجموع الأصول إلى الناتج المحلي الإجمالي	D.3
D.4	The low ratio of broad money to GDP indicates low monetary depth of the economy	1	2	3	4	5	خفاض نسبة النقود بمعناها الواسع إلى الناتج المحلي الإجمالي تشير إلى انخفاض النقد في عمق الاقتصاد.	il D.4

	-								the state that and the state of the the		
~ -	Domestic	c Credit Ratio of GDP							المحلي من النائج المحلي الإجمالي	سبه لانتمان	
D.5	There is I	limited access to loans and credit facilities	1	2	3	4	5		4 فرص الحصول على الفروض والتسهيلات الانتمانية منابع بشريبية	هناك محدودي	D.5
D.6	Have you	a experienced any of the following problems?	1	2	3	4	5		مع المساكل التالية؟	هل لك تجرّبة	D.6
	D.6.1	Loan application rejected	1	2	3	4	5		رفض طلب الفرض	D.6.1	
	D.6.2	Charges too high	1	2	3	4	5		رسوم عالية جدا	D.6.2	
	D.6.3	Relationship difficulties	1	2	3	4	5		صعوبة في العلاقة	D.6.3	
	D.6.4	Bank errors	1	2	3	4	5		اخطاء البنك	D.6.4	
	D.6.5	Red tape (Bureaucracy)	1	2	3	4	5		الروتين (البيروقراطية)	D.6.5	
	D.6.6	Interest rates too high	1	2	3	4	5		سعر فائدة عالي جدا	D.6.6	
D.7	Domestic	c credit of GDP is one of the measurements of financial maturity of the stock market systems	s 1	2	3	4	5	موق الأسهم المالي	ي من الناتج المحلي الإجمالي أحد مقابيس النضج المالي لأنظمة «	الائتمان المحا	D.7
D.8	Domestic	c credit of GDP ability is considered to represent financial maturity in the stock market	1	2	3	4	5	ق الأسهم	ي من الناتج المحلي الإجمالي يعبر عن قدرة النضبج المالي في سو	الائتمان المحا	D.8
	Stock Ma	arket Capitalisation Ratio of GDP							لسوقية الي الناتج المحلي الإجمال	نسبة القيمة ا	
DA	Stock ma	arket capitalisation of GDP represents one aspect of financial maturity in the market-	-based	2	2	4	5	سج المالي في اقتصاد قائم على	موق الأسهم المالية من الناتج المحلي الإجمالي واحد من سمة النض	تمثل رسملة ا	DO
D.9	economy		1	2	3	4	5			السوق	D.9
D.10	Stock ma	rket capitalisation of GDP increases investment in the stock market	1	2	3	4	5	لأسهم المالية	بوق الأسهم المالية من الناتج المحلى الإجمالي استثمار في سوق ا	تزيد رسملة س	D.10
D.11	Stock ma	rkets become larger, as measured by market capitalisation relative to GDP	1	2	3	4	5	المحلى الإجمالي	, تصبح أكبر ، إذا ما قيست مَّن قبل نَّسبة رسمَّلة السوق إلى الناتج ا	أسواق الأسهم	D.11
D.12	Independ	lence of the central bank	1	2	3	4	5		رف المركزي	استقلال المص	D.12
D.13	Other (Pl	ease specify)							التحديد	أخرى، يرجى	D.13
	,										
D.14	What so	urce(s) of finance does your firm use?							وال الذي تستخدمها شركتكم؟	ما مصدر الاه	D.14
		Please tick the box that best reflects your	answer whe	re:		زئمة	الأكثر ملا	الرجاء وضع إشارة على الرقم ا			
								ada ana ang kana ang Ang kana ang			
		2= Used with no success	2 11 1 '	а	1	. 1		c			
	1= Not u	تم استخدامه بدون نجاح sed لم يستخدم	3 = Used Wi	th a n	10dei	rate d	legree o	of success 4	دامه بنجاح کبیر (Used with a great deal of success دامه بنجاح کبیر	تم استخ	
				متوسط	بنجاح	حدامه ب	ىم است				
Code		Statements	Sco	re	- 4	النتيجه	1		البيانات		رت
D.14.1	Bank loa	ns	1	2	3	4	5		مصرفية	القروض ال	D.14.1
D.14.2	Retrained	1 earning	1	2	3	4	5		للربح	اعادة تشغيا	D.14.2
D.14.3	Trade cre	edit (supplies)	1	2	3	4	5		جاري (العرض)	الائتمان الت	D.14.3
D.14.4	Governm	ent subsidies	1	2	3	4	5		حكومية	الإعانات ال	D.14.4
D.14.5	Foreign s	sources	1	2	3	4	5		أجنبية	المصادر ا	D.14.5
D.14.6	Other (Pl	ease specify)							مي التحديد	أخرى، ير.	D.14.6
	How ofte	en during a year does your firm follow its target stock market when financing its invest	tment?					اتها؟	ل السنة شركتك تهدف لمتابعة سوق الأسهم عند تمويل استثمار	کم مرة خلا	
D.15											D.15
		Please tick all those that apply:						ينطبق:	يرجى وضع علامة أمام كل ما ي		
Code		Statements							البيانات		رت
D.15.1		Never							أبد		D.15.1
D.15.2		Rarely (once per year)							نادرا (مرة في السنة)		D.15.2
D.15.3		Often (twice per year)							في أغلب الأحيان (مرتين في السنة)		D.15.3
D.15.4		Very often (approximately every three months)							في أغلب الأحيان (كل ثلاثة أشهر)		D.15.4
D.15.5											D 15 5
		Systematically (approximately every month)							بانتظام (کل شهر تقریبا)		D.13.3
D.15.6		Systematically (approximately every month) During the analysis of each investment project							بانتظام (کل شهر تقریبا) أثناء تحلیل کلّ مشر و ع استثمار ی		D.15.6
D.15.6 D.15.7	Other (Pl	Systematically (approximately every month) During the analysis of each investment project ease specify)							بانتظام (كل شهر تقريبا) أثناء تحليل كلّ مشروع استثماري م. التحديد	أخرى، بر.	D.15.6 D.15.7
D.15.6 D.15.7	Other (Pl	Systematically (approximately every month) During the analysis of each investment project ease specify)							بانتظام (كل شهر تقريبا) أثناء تحليل كلّ مشروع استثماري بي التحديد	أخرى، ير.	D.15.6 D.15.7
D.15.6 D.15.7	Other (Pl	Systematically (approximately every month) During the analysis of each investment project ease specify) Please tick the box that best reflects your	answer whe	re:		تمة	الأكثر ملا	الرجاء وضع اشارة على الرقم ا	بانتظام (كل شهر تقريبا) أثناء تحليل كلّ مشروع استثماري بي التحديد	أخرى، ير.	D.15.6 D.15.7
D.15.6 D.15.7	Other (Pl = Very uni	Systematically (approximately every month) During the analysis of each investment project ease specify) Please tick the box that best reflects your mportant غير مهمة 2= Unimportant عبر مهمة 3=	answer whe	re: آکد	غير مت	نمة خ	الأكثر ملا	الرجاء وضع إشارة على الرقم ا معة 1 Important	بانتظام (كل شهر تقريبا) أثناء تحليل كلّ مشروع استثماري بي التحديد مة حدا 5= Very important	أخرى، ير.	D.15.6 D.15.7
D.15.6 D.15.7	Other (Pl = Very uni	Systematically (approximately every month) During the analysis of each investment project ease specify) Please tick the box that best reflects your mportant غير مهمة 2= Unimportant عبر مهمة	answer whe Uncertain Bank-Based	r e: آکد	غير مت	نمة Mark	الأكثر ملا et-Bas	الرجاء وضع إشارة على الرقم ا يمة Important = sed I	بانتظام (كل شهر تقريبا) اثناء تحليل كلّ مشروع استثماري بى التحديد مة جدا 5= Very important	أخرى، ير. مه	D.15.6 D.15.7
D.15.6 D.15.7	Other (Pl = Very uni	Systematically (approximately every month) During the analysis of each investment project ease specify) Please tick the box that best reflects your mportant عبر مهمة 2= Unimportant عبر مهمة 3= Statements	answer whe Uncertain Bank-Based اقاعدة- المصر في	re: اکد	غير مت	انمة Mark بوقية	الأكثر ملا ket-Bas أناحدة- الس	الرجاء وضع إشارة على الرقم ا يمة Important = sed	بانتظام (كل شهر تقريبا) اثناء تحليل كلّ مشروع استثماري بى التحديد مة جدا 5= Very important البياتات	أخرى، ير. مه	D.13.5 D.15.6 D.15.7
D.15.6 D.15.7	Other (Pl = Very uni	Systematically (approximately every month) During the analysis of each investment project ease specify) Please tick the box that best reflects your mportant حدا غير مهمة 2= Unimportant Statements Important Important Statements Important <td colspan="</td> <td>answer whe Uncertain Bank-Based افاعدة- المصرفي 2 3 4</td> <td>re: یکر ا</td> <td>غير مت ا</td> <td>ينمة Mark سوقية 2</td> <td>الأكثر ملا ket-Bas قاعدة- الس 3 2</td> <td>الرجاء وضع إشارة على الرقم ا يمة Important = sed الق 4 5 </td> <td>بانتظام (كل شهر تقريبا) أثناء تحليل كلّ مشروع استثماري بى التحديد مة جدا 5= Very important البياتات</td> <td>أخرى، ير. مه</td> <td><u>D.15.5</u> D.15.7</td>	answer whe Uncertain Bank-Based افاعدة- المصرفي 2 3 4	re: یکر ا	غير مت ا	ينمة Mark سوقية 2	الأكثر ملا ket-Bas قاعدة- الس 3 2	الرجاء وضع إشارة على الرقم ا يمة Important = sed الق 4 5	بانتظام (كل شهر تقريبا) أثناء تحليل كلّ مشروع استثماري بى التحديد مة جدا 5= Very important البياتات	أخرى، ير. مه	<u>D.15.5</u> D.15.7
D.15.6 D.15.7 1 Code D.16.1 D.16.2	Other (Pl = Very uni Mobilisii Identifyii	Systematically (approximately every month) During the analysis of each investment project ease specify) Please tick the box that best reflects your mportant عبر مهمه 2= Unimportant عبر مهمه 3= Statements ag saving ng good investment 1	answer whe Uncertain Bank-Based لقاعدة- المصرفي 2 3 4 2 3 4	re: اکد 5 5	غير مت ا 1	نمة Mark سوقية 2	الأكثر ملا ket-Bas تاعدة- الس 3 4 3 4	الرجاء وضع إشارة على الرقم ا همة Important = sed الق 4 5 4 5	بانتظام (كل شهر تقريبا) أثناء تحليل كلّ مشروع استثماري بي التحديد مه جدا 5= Very important البياتات خار	اخری، یر. مه تعیئة الاد تحدید جی	<u>D.15.5</u> D.15.6 D.15.7 <u></u> D.16.1 D.16.2

D.16.3	Allocating capital	1	2	3	4	5	1	2	3	4	D.16 تخصيص الرأسمال	6.3
D.16.4	Providing risk management tools	1	2	3	4	5	1	2	3	4	D.16 توفير أدوات لإدارة المخاطر D.16	6.4
D.16.5	Exerting sound corporate control	1	2	3	4	5	1	2	3	4	D.16 ممارسة سليمة لمراقبة الشركات	6.5
D.16.6	Justifying problems associated with excessively powerful banks	1	2	3	4	5	1	2	3	4	D.16 تبرير المشاكل المفرطة والمرتبطة بالسلطة المصرفية D.16	6.6
D.16.7	Increasing transparency and accountability	1	2	3	4	5	1	2	3	4	D.16 زيادة الشفافية والمساءلة	6.7
D.16.8	Structure activity	1	2	3	4	5	1	2	3	4	D.16 هيكل النشاط	6.8
D.16.9	Structure size	1	2	3	4	5	1	2	3	4	D.16 هيكل الحجم D.16	6.9
D.16.10	Structure efficiency	1	2	3	4	5	1	2	3	4	D.16 هيكل الكفاءة	6.10
D.16.11	Structure regulatory	1	2	3	4	5	1	2	3	4	D.16 الهيكل التنظيمي	6.11
D.16.12	Finance activity	1	2	3	4	5	1	2	3	4	D.16 تمويل النشاط	6.12
D.16.13	Finance size	1	2	3	4	5	1	2	3	4	 D.16 حجم التمويل	6.13
D.16.14	Finance efficiency	1	2	3	4	5	1	2	3	4	D.16 الكفاءة المالية	6.14
D.16.15	Other (Please specify)		_	-	-	-	_	_	-		D.16 أخرى، برجي التحديد	6.15
		1	2	3	4	5	1	2	3	4	5	
	Section E: Stock Market Performanc										ألقسم الخامس : أداء سوق الأوراق المالية	
This sect	on aims to examine the change in stock market performance variables after the introduction o	f the	econ	omic	refo	rm					a maker be to be a first set of a start set of the start	
programm	ie.										هذا الفسم إلى ذر أسه تأثير متغيرات أذاء سوق الأسهم على برتامج الإصلاح الأقتصادي.	يهدف ،
1.0	Please tick the box that best reflect	s you	r ans	swer	whe	re:		لائمة	کثر ما	قم الأذ	الرجاء وضع إشارة على الرة	
	لا أوفق 2 = Disagree لا أوفق بشد 1 = Strongly disagree	3 =	= Nei	ther a	Igree	nor	disag	gree	حايد		أوفق بشدة 5 = Strongly agree أوفق 4 = Agree	
Code	Statements				Sco	ore		لنتيجة	ti	Τ	الييانات	رت
	Market Size										حجم السوق	-
E.1	Market capitalisation is considered to be a measure of market size in the stock market			[1	2	3	4	5	T	ا سُوْق كَفَايَة رأس المال يعتبر بمثابة أداة من حجم السوق في سوق الأو راق المالية	E.1
E.2	Market capitalisation is a percentage of GDP in the stock market				1	2	3	4	5		ا سوق كفاية رأس المال كنسبه مئويه من الناتج المحلي الإجمالي في سوق الأسهم المالية	E.2
E.3	The number of listed companies is an important factor for determining market size				1	2	3	4	5		جدد الشركات المدرجة تمثل عامل هام لتحديد حجم السوق	E.3
E.4	The volume of shares listed enhances the stock market size				1	2	3	4	5		ججم الأسهم المدرجة في السوق تعزز من حجم السوق في سوق الأور اق المالية	E.4
E.5	The number of financial intermediaries increases the market size in the stock market				1	2	3	4	5		حد مؤسسات الوساطة المالية بزيد من حجم السوق في سوق الأور اق المالية. FF	E.5
2.10	Market Liquidity				-	-	e				يد و و و پر پرون وي	
	The total value-traded in market capitalisation contributes to viability and accessibility of the sto	ck ma	rket						1	T	الحمالي القيمة المتداولة في رسملة البيوق تساهم في ضمان الاستمر اربة و إمكانية الوصول الى سوق الأسهم	
E.6		•11 1110			1	2	3	4	5		المالية المالية	E.6
E.7	The ratio of total value-traded to GDP impacts on the stock market				1	2	3	4	5		E النسبة من إجمالي القيمة المتداولة إلى الناتج المحلي الإجمالي لها تأثير ات على سوق الاسهم المالية	E .7
E.8	The ratio of total value-traded shares helps in increasing volume of shares listed of market li stock market	quidit	y in	the	1	2	3	4	5	ق	النسبة من إجمالي قيمة الاسهم المتداولة يساعد في زيادة حجم الاسهم المدرجة في السوق ذات السيوله في سوق E الاسهم المالية	E .8
E.9	Turnover ratios are indicators of the market's ability to trade significant positions				1	2	3	4	5		ا نسب الدور أن تعد مؤشر مهم للحكم على قدرة السوق للمتجارة	E.9
	Market Activity						-				نشاط السوقي	
E.10	The value of trade is part of market activity				1	2	3	4	5		ا قيمة التجارة هي جزء من نشاط السوق	E.10
E.11	Volume of trade plays an important role in the stock market				1	2	3	4	5		بلعب حجم التجارة دورا مهما في سوق الأسهم المالية	E.11
E.12	A number of transitions are considered to be market activities that affect the stock market				1	2	3	4	5		عدد من التحو لات تعبر عن نشاطات السوق التي تؤثر على سوق الأسهم المالية	E.12
E.13	The increased number of traded companies increases market activities in the stock market				1	2	3	4	5		از دياد عدد الشركات المتداولة في السوق تزيد من نشاطات السوق في سوق الأسهم المال	E.13
E.14	The increased value of new issues, including capital, generates market activities in the stock mar	ket			1	2	3	4	5		القيمة المتزايدة من الاصدر ات الجديدة، بما في ذلك رأس المال تولد نشاطات للسوق في سوق الأسهم المالية –	E.14
E.15	The value of new issues acts as a percentage of GDP in the stock market				1	2	3	4	5		ـــــــــــــــــــــــــــــــــــــ	E.15
	Market Concentration				-	_	-	-	-	-	ـــــــــــــــــــــــــــــــــــــ	
	The percentage of the largest company's share in market capitalisation increases market capital	lisatio	on sh	are						(j	النسبة المئوية من أسهم الشركات الكبري لها حصبة في رسملة السوق والتي تزيد من حصبة اسهم رسملة السوق	
E.16	in the stock market				1	2	3	4	5		E في سوق الأسهم المالية	E.16
E.17	The percentage of largest companies affects the value traded in the stock market				1	2	3	4	5		E النسبة المئويه في الشركات الكبري تؤثر على القيمة المتداولة في سوق الاوراق المالية	±.17
E.18	Other (Please specify)										E أخرى، يرجى التحديد	£.18
h												

E.19	For the following statements, please indicate to what degree they would encourage the successful estable	lishn	nent				بع علي نجاح إنشاء سوق الأوراق المالية.	ن خلال البيانات التالية، يرجى الإشارة إلى أي درجة أنها تشم	E.19 م
	of a stock market in the Libyan economy.				<i>.</i>	N			
	العلمة المعالي معالي معالي معالي معالي	agre	ere:	r dis	sagre	ر مرد ايد e	الرجاع وصلع إلمارة على الرد أوفق 4 = Agree	اوفق بشدة 5 = Strongly agree	
Code	Statements	Sc	core		يجة	111		البيانات	رت
E.19.1	Lack of quality companies	1	2		3 4	5		ضعف جودة الشركات	E.19.1
E.19.2	Lack of collateral (security)	1	2	3	3 4	5		الافتقار إلى ضمانات (أمنية	E.19.2
E.19.3	Lack of good trading record	1	2	3	3 4	5		الافتقار إلى السجل التجاري الجيد	E.19.3
E.19.4	Provide a new technology as the market needs	1	2	3	3 4	. 5		توفر تقنية حديثة حسب حاجة السوق	E.19.4
E.19.5	Poor relationships with banks	1	2	3	3 4	5		ضعف العلاقات مع البنوك	E.19.5
E.19.6	Geographical location	1	2	~	3 4	5		الموقع الجغر افي	E.19.6
E.19.7	Deterioration in the state of the economy	1	2	~	3 4	5		تدهور حالة الاقتصاد	E.19.7
E.19.8	The suppliers of finance are in small and/or underdeveloped sector	1	2	~	3 4	5	نطاع	حصة الموردين الماليين قد تكون صغيرة أو غير متطورة ال	E.19.8
E.19.9	Proximity to international markets	1	2	3	3 4	5		القرب من الأسواق الدولية	E.19.9
E.19.10	Inability to convince lenders of profitability of the investments	1	2	3	3 4	5		عدم القدرة على إقناع المقرضين من ربحية الاستثمار ات	E.19.10
E.19.11	Other (Please specify)							أخرى، يرجى التحديد	E.19.11
	In the process of gathering information about the corporation in which the stock market operates, d	iffer	ent s	our	ces o	of in	ation are available for companie	s. From the following information sources,	
E.20	could you rate the importance of each source to the stock market effects?								E.20
					f			e studies to the this to set the this	
	ن اهمية كل من المصادر ادناء لعملية نقيم السوق المالي؟ منابع	ید مدی	اء نحد	لرج	، ادناه ا ۳	ومات	⁴ إلى عدة مصادر للمعلومات من ضمن مصادر	عند الفيام بعملية نفيم عمليات السوق المالي عادة نلجا الشرك	
1=	Please tick the box that best reflects your answer Very unimportant غير مهمة 2= Unimportant عبر مهمة 3= Uncertain	whe متأكد	re: غير ہ		٩	ملائم	الرجاء وصع إشاره على الرقم مهمة 4= Important	مهمة جدا 5= Very important	
Code	Statements	Sc	core		يجة	التت		البيانات	رت
E.20.1	Companies' personnel at headquarters	1	2	3	3 4	5		موظفي الشركة في الإدارة العامة	E.20.1
E.20.2	Companies' personnel abroad	1	2	3	3 4	. 5		موظفي الشركة في الفروع	E.20.2
E.20.3	Banks operating locally and abroad	1	2	3	3 4	1 5		البنوك العاملة محليا وخارجيا	E.20.3
E.20.4	Other companies operating locally and abroad	1	2	~,	3 4	5		الشركات العاملة محليا وخارجيا	E.20.4
E.20.5	Media (e.g. television, radio, newspapers)	1	2	~	3 4	5		وسائل الأعلام (مثل التلفزيون والإذاعة والصحف	E.20.5
E.20.6	Trade associations	1	2	~	3 4	5		الاتحادات التجارية	E.20.6
E.20.7	International organisations (e.g. International Monetary Fund)	1	2	3	3 4	5		منظمات دولية (مثل صندوق النقد الدولي)	E.20.7
E.20.8	Business magazines	1	2	3	3 4	5		المجلات التجارية	E.20.8
E.20.9	Academics	1	2	3	3 4	5		الأكاديميون	E.20.9
E.20.10	Libyan embassies abroad	1	2	3	3 4	5		السفار ات الليبية في الخار ج	E.20.10
E.20.11	Governmental domestic agencies (e.g. Libyan Investment Board)	1	2	3	3 4	5		الوكالات الحكومية المحلية (مثل مجلس الاستثمار الليبي	E.20.11
E.20.12	Other (Please specify)							أخرى، يرجى التحديد	E.20.12
E.21	In your opinion, which of the following factor/s may negatively affect the stock market?						وَثِر سلبا على سوق الأسهم؟	في رأيك، إلى أي درجة أنت مهتم بالأحداث أدناه والتي قد ت	E.21
	Please tick the box that best reflects your answer	whe	re:		ā	ملائم	الرجاء وضع إشارة على الرقم	· · · · · · · · · · · · · · · · · · ·	
	ا عبر مهتم العالي = Not concerned عبر مهتم العالي = Not concerned عبر مهتما = Not concerned عبر مهتما = 1 = Not concerned	ىھتم	<u>م</u>		4	$= V\epsilon$	ncerned مهتم جدا 5 = Ex	tremely concerned مهتم على الإطلاق	
Code	Statements	Sc	core		يجة	<u>111</u>		البيانات	رت
E.21.1	Political events (e.g. revolutions, political instability, riots, corruption, expropriation, and terrorism).	1	2	3	3 4	1 5	ال الشغب، الفساد، نزع الملكية والإر هاب).	أحداث سياسية (مثل الثورات، عدم الاستقرار السياسي، أعم	E.21.1
E.21.2	Financial events (e.g. foreign exchange rates, interest rates or commodity prices currency inconvertibility,	1	_				الفائدة أو أسعار السلع، العقوبات الاقتصادية،	الأحداث المالية (مثل سعر صرف العملات الأجنبية، أسعار	E.21.2
	economic sanctions, restrictions on ownership, and taxation).	1	2	1	, 4	1	-	فرض قيود على الملكية والضر ائب).	
E.21.3	Cultural events (e.g. language, education, religion).	1	2	3	3 4	5		الأحداث الثقافية (مثل اللغة، التعليم والدين).	E.21.3
E.21.4	Other (Please specify)							أخرى، يرجى التحديد	E.21.4

In the process of gathering information about the stock market in which corporation floated. When اثناء عملية جمع المعلومات حول الشركة التي تتبع سوق الأسهم المالية إلى أي درجة أنت قلق حول كل من اللازمات E.22 In the process of gathering information about the stock market in which corporation floated. When In the process of gathering information about the stock market in which corporation floated. When In the process of gathering information about the stock market in which corporation floated. When In the process of gathering information about the stock market in which corporation floated. When In the process of gathering information of business activity, how concerned are you about each of the following financial crises in terms of their potential adverse impact upon your business? E.2									
	Please tick the box that best reflects your answer	wher	·e:		لائمة	کثر م	الرجاء وضع إشارة على الرقم الأ	•	
1	= Not concerned مهتم قلیلا 2 = Slightly concerned غیر مهتما 3 = Moderately concerned	مهتم			4 = V	/ery	مهتم على الإطلاق Extremely concerned مهتم جدا concerned		
E.22.1	Political instability						عدم الاستقرار السياسي	E.22.1	
E.22.2	Economic sanctions	1	2	3	4	5	العقوبات الاقتصادية	E.22.2	
E.22.3	Currency inconvertibility	1	2	3	4	5	عدم قابلية تحويل العملة المحلية إلى عملة صعبة	E.22.3	
E.22.4	Import and/or export restrictions	1	2	3	4	5	القيود على الاستيراد والتصدير	E.22.4	
E.22.5	Ownership and/or personnel restrictions	1	2	3	4	5	القيود على الملكية أو تعين الموظفين	E.22.5	
E.22.6	Investment climate	1	2	3	4	5	مناخ الاستثمار	E.22.6	
E.22.7	Terrorism	1	2	3	4	5	الإرهاب	E.22.7	
E.22.8	Demonstrations, riots and insurrection	1	2	3	4	5	تظاهرات وأعمال الشغب والتمرد	E.22.8	
E.22.9	Corruption	1	2	3	4	5	الفساد	E.22.9	
E.22.10	Wars	1	2	3	4	5	الحروب	E.22.10	
E.22.11	Taxation restrictions	1	2	3	4	5	القيود الضريبية	E.22.11	
E.22.12	Other (Please specify)						أخرى، يرجى التحديد	E.22.12	
E.23	In the following table there are several problematic factors which might influence the success of stock performance. Please answer these questions by selecting the option that you think best reflects your op	k mar inion.	ket	نله	ذه الأسنا	عن هذ	في الجدول النّالي هناك عدّة عوامل قد تعيق أو تؤثر على نجاح أداء سوق الأوراق المالية. الرجاء الأجابه : عن طريق اختيار أفضل الإجابة والتي تعتقد إنها تعكس رأيك.	E.23	
	Please tick the box that best reflects your answer	wher	·e:		لائمة	کثر م	الرجاء وضع إشارة على الرقم الأ		
1 =	الا توجد عقبة بسيطة عقبة بسيطة 2 = Minor obstacle لا توجد عقبة 3 = Moderate obst	acle	حتدلة	عقبة ه			عقبة كبيرة 5 = Major obstacle عقبة		
Code	Statements	Sc	ore	Ä	النتيج		البياتات	رت	
E.23.1	There are no clear monetary policies in relation to the exchange rate	1	2	3	4	5	لا توجد سياسات نقدية واضحة بالنسبة لسعر الصرف	E.23.1	
E.23.2	There are high rates of taxes and tariffs	1	2	3	4	5	هناك ارتفاع معدل الضرائب والتعريفات الجمركية	E.23.2	
E.23.3	Disorganisation and corruption exists in financial institutions	1	2	3	4	5	عدم التنظيم وظهور الفساد في المؤسسات المالية	E.23.3	
E.23.4	There is excessive regulation and bureaucracy	1	2	3	4	5	هناك الإفراط في التشريعات والبيروقر اطية	E.23.4	
E.23.5	There is a lack of data and information	1	2	3	4	5	نقص في البيانات والمعلومات	E.23.5	
E.23.6	There is a lack of discussion on stock market performance	1	2	3	4	5	نقص في إبداء الراي والحوار حول أداء سوق الأوراق المالية	E.23.6	
E.23.7	There are weaknesses in skills within the workforce	1	2	3	4	5	هناك نقاط ضعف في المهار ات داخل القوى العاملة	E.23.7	
E.23.8	There are political/business risks	1	2	3	4	5	هناك مخاطر سياسية وتجارية	E.23.8	
E.23.9	Oil prices shocks	1	2	3	4	5	المضاربة في أسعار النفط	E.23.9	
E.23.10	Other (Please specify)						أخرى، يرجى التحديد	E.23.10	
E.24	Which of the following do you consider to be the best practise model experiments for the Libyan stock	mark	et?	لى	لهبق عا	ي يند	من خلال البيانات التالية هناك، العديد من التجارب النموذجية لعدة دول، في رأيك ما هو النموذج الأفضل الذ حالة سوق الأوراق المالية الليبي؟	E.24	
	Please tick the box that best reflects your answe	r whe	ere:	2	ملائمة	لأكثر	الرجاء وضع إشارة على الرقم ا		
	1 = Strongly disagree لا أوفق بشد 2 = Disagree 3 = Neither	agree	e nor	disag	ree .	محايد	أوفق بشدة 5 = Strongly agree أوفق 4 = A		
Code	Statements	Sc	ore	ة	النتيجا		البيانات	رت	
E.24.1	Middle Eastern and North African Countries	1	2	3	4	5	بلدان الشرق الأوسط وشمال أفريقيا	E.24.1	
E.24.2	Far Eastern Countries	1	2	3	4	5	بلد الصين	E.24.2	
E.24.3	Indian Subcontinent Countries	1	2	3	4	5	بلد الهند	E.24.3	
E.24.4	Central and Eastern European Countries	1	2	3	4	5	بلدان أوروبا الوسطي والشرقية	E.24.4	
E.24.5	Latin American Countries	1	2	3	4	5	بلدان أمريكا اللآنينية	E.24.5	
E.24.6	Other (Please specify)						أخرى، يرجى التحديد	E.24.6	

APPENDICES

E.25	To whom are the results of the stock market assessment reported? (Please state the title only)		E.25
		إلى من ترسل تقييم نتائج سوق الأوراق المالية؟ (يرجى ذكر عنوان فقط)	
Code	Statements	البياتات	رت
E.25.A			E.25.A
E.25.B			E.25.B
E.25.C			E.25.C
E.25.D			E.25.D
E.25.E			E.25.E
E.25.F			E.25.F
E.25.G			E.25.G

Please write any comments which you feel would be helpful to this research.	الرجاء تدوين أية ملاحظات ترى إنها مفيدة لهذة الدراسة.
If you are interested in obtaining a free copy of the result of this study, please complete your details here an	d attach this order form with the completed questionnaire.
صة بك أدناه مرفقة مع استمارة الاستبيان.	إذا كنت ترغب في الحصول على نسخة مجانية من نتيجة هذه الدراسة، يرجى استكمال التفاصيل الخا
Name:	الاسم.
E-mail:	البريد الكتروني:
Address:	العنوان:

,	Thank you for your time and co-operation in completing this questionnaire
	شكرا جزيلا على وقتك والتعاون معنا في إنجاز هذا الاستبيان

Appendix H: Covering Letter from the University of Huddersfield with Regard to Conducting the Semi-structured Interviews



20/09/2007

Dear Participant

Thank you very much for your participation in the previous survey designed as a methods of collecting data. Initial analysis of the survey has revealed interesting findings. As one of the successful players in the economic reform and the financial market, these findings now need to be discussed with you. I should therefore be most grateful if you would agree to be interviewed within the next two weeks. The location, date and time are to be arranged at your convenience.

The general theme of the interview will ceater upon two main issues.

- The economic reform programme, which should represent on of the ssuccessful factors of Libyan stock market performance in the future.
- The practices of an economic reform programme upon the Libyan economy and financial institutions.

We guarantee that all information provided and data collected will be regarded as strictly confidential and will be used solely for academic research in which no individual economic or financial system will be identified. If you have any queries about this interview, please contact researcher or Dr. Jhon Coock the on the numbers or e-mails provided below.

Your participation is very much appreciated.

Sincerely yours,

Najeb Masoud

Business School University of Huddersfield HD1 3DH Telephone: 07757261306 E-mail: <u>najeb2000@hotmail.com</u> Dr. John Coock

Direct of Studies Businesses school Huddersfield University HD1 3DH Tel: +44 (0) 1484 472086 E-mail: j.a.cook @hud.ac.uk

Appendix I: Interview Questions



Appendix J: Respondent Interviews

Approach methods of qualitative data collection in the form of *in-depth interviews* were used in this research. 14 respondents were selected as the target for *interviews* from those not targeted in the questionnaire. This approach is based on:

Respondent A: General Manager of Central Bank. Despite the respondent being "very" enthusiastic about the banking system in terms of economic reform programme, there is no formal responsibility assigned for an individual (s) to conduct the banking reform process, nor is any effort made by any individual in the banking system to do so. This bank, therefore, was classified as one of the most important among financial institutions.

Respondent B: General Manager of the Stock Market. This individual is "extremely" concerned about economic reform programmes in terms of financial institutions. He is formally responsible for conducting the programme of economic reform. The macro-economic reform and financial maturity are utilised as a valuable source of information. Even though a questionnaire suggests the use of qualitative techniques in an economic reform programme, the interview does not.

Respondent C: Executive Manager/ Investment Department of Business. The business has never assigned formal responsibility for an individual (s) to conduct the economic reform programme even though its executive manager is "very" concerned about the financial market. The main sources of information for this manager are personnel at headquarters, colleagues overseas and other institutions. This manager is satisfied with his approach to their economic reform and business environment.

Respondent D: Senior General Manager Marketing. This respondent is formally responsible for conducting an economic reform programme. The manager is "very" concerned about economic reform in terms of any unfavourable consequences upon the financial market. The programme is, therefore, conducted occasionally and is motivated by proposing investment and/or re-investment in a certain country. This manager is "very" satisfied with his approach to the process of stock market.

Respondent E: General Manager of Banking Sector. Despite being "very" concerned about the economic reform programme and financial maturity, formal responsibility for an individual (s) to conduct the economic reform programme has never been assigned. In conducting the programme, the manager relies on various sources of information, including other banks and financial institutions thought to be knowledgeable in specific areas. This manager, however, is satisfied with his approach to an economic reform programme.

Respondent F: Marketing Manager/Corporate Marketing Manager of Finance Marketing Strategy and Plans. The manager is "very" concerned about developing existing business strategy and agreeing the budget and the creation of new job opportunities. He relies on his support for business development activities including events' organisation, collateral development, and website and database management. Finally, he is "very" satisfied with his experience of operating within a professional service environment.

Respondent G: Manager of Public Sector. The latter is italicise concerned about the private sector but "very" concerned about the public sector. In conducting the process, the manager can realise diverse sources of

APPENDICES

information including public sector competition. This manager, though, is satisfied with his approach to professional service experience in terms of an economic reform programme.

Respondent H: General Manager of Private Sector. The general manager is "moderately" concerned about ownership in terms of any unfavourable consequences concerning companies. There is no formal responsibility for an individual (s) to perform the ownership process but the related activities are conducted by the general management of companies. The manager relies on banks and other companies, usually similar in nature, to obtain required data. He is dissatisfied with current arrangements regarding ownership of companies.

Respondent I: Manufacturing Improvement Manager. The manufacturing manager is "very" concerned about economic reform programme relating any unfavourable consequences upon manufacturing efficiencies. He is, though, "very" satisfied with his approach and in delivering successful investment projects.

Respondent J: Financial Manager of Planning. The manager is "very" concerned about the economic reform programme, direct, organisation control and implementing the financial policies and systems of establishment. He obtains information and prepares various financial reports for senior management. He is satisfied with his approach to the companies via the private sector and government.

Respondent K: Accounting Disclosure Manager. Despite the manager being "very" concerned about economic reform, formal responsibility for an individual(s) to conduct the economic reform programme has never been assigned. In conducting the programme, the manager relies on different sources of information, including formal accounting authorities such as company regulations, securities, commission and stock market disclosure requirements. Nonetheless, this manager is satisfied with his approach to an economic reform programme.

Respondent L: Managers of Financial Regulations of Financial Institutions. The financial manager of regulation is "very" concerned about the economic reform programme in terms of the financial market to certain requirements by a government or non-governmental organisation. This manager is formally responsible for supervision, restriction and guidelines. Even though, a questionnaire suggests the use of qualitative and quantitative techniques in the Libyan financial arena and economy.

Respondent M: Director of the Economic Research Centre. The respondent is "very" enthusiastic about the multi-disciplinary research centre that brings together undergraduates, graduates, post-doctoral fellows and academics from a wide range of academic disciplines to research and study the Libyan economy related issues and topics. The manager provides a wider societal role and serves as a key element in the Libyan economic reform programme.

Respondent N: Director of the Tax Department, Finance Ministry. The Taxes manager is "extremely" concerned to provide the most important revenue source for the government about the economic reform programme in terms of financial institutions. This manager is formally responsible for conducting the source of fiscal revenue; tax is a key economic player of macro-economic reform and fiscal policy requirements and greatly affects the Libyan economy and social development.

7

Appendix K: Chapter Seven: Presentation of the Findings

K.1 Analysis of the Results Related to Participants and Organisations Characteristics

K.1.1 Sample Description Characteristics

Table K 1.1 Distribution of Responses by Demographics of Gender and Age (PERSONAL) Sample Groups ^a

Statements	Gro CI	up 1 LM	Group 2 CYM		Group 3 PVS		Group 4 PCS		Group 5 FLI	
Sample Size	4	.9	43		41		38		32	
Gender										
Male (Per cent)	71.4		67.4		75.6		57.9		62.5	
Female (Per cent)	28.6		32.6		24.4		42.1		37.5	
Age	Freq.	Per.	Freq.	Per.	Freq.	Per.	Freq.	Per.	Freq.	Per.
Under 30	3	6.1	5	11.6	3	7.3	6	15.8	3	9.4
30-40	15	30.6	16	37.2	14	34.1	13	34.2	12	37.5
41-50	19	38.8	17	39.5	16	39.0	14	36.8	13	40.6
51-65	12	24.5	5	11.6	8	19.5	5	13.2	4	12.5
Total	49	100.0	43	100.0	41	100.0	38	100.0	32	100.0

Notice a: Number of total valid responses =203.

Source: Analysis of questionnaire survey data.

Fahle K 1 2 Dist	tribution of Resnans	es by Education	Levels (FDUCAT)	(ON) of Particinants ^a
	in induction of Acopolis	cs by Education	LUCIS (LDUCAT)	(Or t) of I al therpathes

Qualifications	Group 1 CLM		Group 2 CYM		Group 3 PVS		Group 4 PCS		Group 5 FLI	
	Freq.	Per.								
High School	N/A	N/A	N/A	N/A	N/A	N/A	1	2.6	N/A	N/A
College Degree	6	12.2	9	20.9	12	29.3	8	21.1	9	28.1
Bachelor's Degree	26	53.1	18	41.9	21	51.2	24	63.2	16	50.0
Master's Degree	8	16.3	13	30.2	6	14.6	4	10.5	5	15.6
Doctorate	9	18.4	3	7.0	2	4.9	1	2.6	2	6.3
Total	49	100.0	43	100.0	41	100.0	38	100.0	32	100.0

Notice a: Number of total valid responses =203.

Source: Analysis of questionnaire survey data.

Table IX 1.5 Distribution of Responses by Academic Dackground (SODJECT) of Tarticipant	Table	K 1	.3	Distribution	of	Responses by	Academic	Background	(SUBJECT)) of Participants
--	-------	-----	----	--------------	----	--------------	----------	------------	-----------	-------------------

Academic Background	Group 1 CLM		Group 2 CYM		Group 3 PVS		Group 4 PCS		Group 5 FLI	
	Freq.	Per.								
Accounting and Finance	19	38.8	12	27.9	10	24.4	8	21.1	11	34.4
Economics	9	18.4	10	23.3	8	19.5	6	15.8	9	28.1
Business Studies	4	8.2	4	9.3	5	12.2	3	7.9	4	12.5
Money and Banking	6	12.2	9	20.9	5	12.2	7	18.4	5	15.6
Management and Marketing	11	22.4	8	18.6	13	31.7	14	36.8	3	9.4
Total	49	100.0	43	100.0	41	100.0	38	100.0	32	100.0

Notice a: Number of total valid responses =203.

Source: Analysis of questionnaire survey data.

Table K 1.4 Distribution of Responses by Current Position (POSITION) of Participants ^a

Current Position	Group 1 CLM		Group 2 CYM		Group 3 PVS		Group 4 PCS		Group 5 FLI	
	Freq.	Per.								
General Manager	2	4.1	6	14.0	7	17.1	5	13.2	7	21.9
Senior Manager	6	12.2	7	16.3	5	12.2	8	21.1	7	21.9
Administrative Manager	19	38.8	13	30.2	12	29.3	10	26.3	9	28.1
Employment Manager	17	34.7	10	23.3	11	26.8	6	15.8	4	12.5
Public Relationship Manager	5	10.2	7	16.3	6	14.6	9	23.7	5	15.6
Total	49	100.0	43	100.0	41	100.0	38	100.0	32	100.0

Notice a: Number of total valid responses =203.

Source: Analysis of questionnaire survey data.



Figure K 1.1 Distribution of Responses by Type of Bank Category (LOCATION)^a

Notice a: Number of total valid responses =203.

Source: Analysis of questionnaire survey data.

Table K 1.5 Distribution of Responses by Work Experience (EXPERIENCE) of Participants ^a

	Group 1		Gro	Group 2		Group 3		Group 4		up 5
Academic Background	CLM		СҮМ		PVS		PCS		FLI	
	Freq.	Per.	Freq.	Per.	Freq.	Per.	Freq.	Per.	Freq.	Per.
Less than 3 years	2	4.1	3	7.0	6	14.6	5	13.2	4	12.5
3-6 years	8	16.3	7	16.3	8	19.5	7	18.4	9	28.1
7 to 10 years	21	42.9	13	30.2	14	34.1	11	28.9	8	25.0
More than 10 years	18	36.7	20	46.5	13	31.7	15	39.5	11	34.4
Total	49	100.0	43	100.0	41	100.0	38	100.0	32	100.0
Chi-Square	19.	000	15.	.326	4.3	366	6.2	211	3.2	250
Asymp. Sig.	0.0	000	0.0	002	0.2	225	0.1	102	0.3	355

Notice a: Number of total valid responses =203.

Source: Analysis of questionnaire survey data.





Notice a: Number of total valid responses =203.

Source: Analysis of questionnaire survey data.





Notice a: Number of total valid responses =203.

Source: Developed from questionnaire survey data analysis from Subsection 6.3.1.8: Ownership.



Figure K 1.4 Distribution of Responses Companies by Total Assets (SIZE) in LD Million ^a

Notice a: Number of total valid responses =203.

Source: Developed from questionnaire survey data analysis from Subsection 6.3.1.9: Size.

Figure K 1.5 Distribution of Responses Companies by Number of Operating Markets (MARKET) a



Notice a: Number of total valid responses =203.

Source: Developed from questionnaire survey data analysis from Subsection 6.3.1.10: Current Market Region.





Notice a: Number of total valid responses =203.

Source: Developed from questionnaire survey data analysis from Subsection 6.3.1.11: Industry.





Notice a: Number of total valid responses =203.

Source: Developed from questionnaire survey data analysis from Subsection 6.3.1.12: Revenue.

K.2 Correlations of Coefficient among Characteristics

This part provides a contribution by describing the correlation analysis using *SPSS (2003)* packages used to measure the basic statistical relationship between characteristics (i.e. personnel, education, subject, position and location) that are common to both sample and population. Other variables of characteristics correlated of the size of financial institutions. They are classified into three categories representing small, medium and large-sized. Large-sized companies possess more assets, larger numbers of employees, industry and ownership companies, which operate in more current markets than do medium or small-sized entities. However, the higher the number of operating markets, the higher the number of industries.

The output of Spearman's Rho test⁶⁶, as depicted in Table K 2.1, is to test whether there is a significant correlation between the five characteristic variable ratings. Findings illustrate that there is no linear relationship between correlation coefficient among the variables of characteristics. Spearman's Rho test detected insignificant correlation between personal characteristics of age of employee and other characteristics which have facilities: SUBJECT (Spearman's Rho = 0.062, P > 0.05, 2-tailed) and LOCATION (Spearman's Rho = 0.026, P > 0.05, 2-tailed). These results indicate that, as one variable (personal) increases, the other variables increase too. The output of Spearman's Rho test further reveals that there is no linear relationship significant correlation between education and current POSITION of employee (Spearman's Rho = 0.076, P > 0.05, 2-tailed) and bank category LOCATION (Spearman's Rho = 0.096, P > 0.05, 2-tailed). The first correlation means that, the higher the level of education, the higher the location of bank category.

	Spearman's Rho (2-tailed) (Valid N=203)												
Characteristics	PERSONAL	EDUCATION	SUBJECT	POSITION	LOCATION								
Personal	-	$r^{a} = -0.051$ $P^{b} = 0.471$	r = 0.062 p = 0.383	r = -0.061 p = 0.378	r = 0.026 p = 0.709								
Education		-	r = -0.035 p = 0.623	r = 0.076 p = 0.279	r = 0.052 p = 0.465								
Subject			-	r = -0.086 p = 0.225	r = -0.052 p = 0.462								
Position				-	r = -0.029 p = 0.681								
Location					-								
Notice a: S	pearman's Rho correla	tion coefficient tes	t(r).										
			1										

Notice b: Probability (p) of correlation coefficient value is Sig. (2-tailed) relationship between two variables.

Source: Analysis of questionnaire survey data.

Further analysis of response characteristics were negative linear relationship between personal variable and the qualification of EDUCATION degree (Spearman's Rho = -0.025, p > 0.05, 2-tailed) and POSITION (Spearman's Rho = -0.061, p > 0.05, 2-tailed). This result indicates that, as personal ratings increase, the level of education ratings decrease. Therefore, as each personal perception of age of employee varies, these ratings may not represent reliable indicators of the actual performance of qualification of education level degree and current position of job responsibility. Nevertheless, there were significant negative correlations between level of education and SUBJECT (Spearman's Rho = -0.035, p > 0.05, 2-tailed) academic background SUBJECT with ratings of POSITION (Spearman's Rho = -0.086, p > 0.05, 2-tailed) and subject and LOCATION (Spearman's Rho = -0.063, p > 0.05, 2-tailed). The output of Spearman's Rho test, as depicted in Table K 2.2, reveals that there is a significant positive correlation coefficient between size of total assets in million LD and the number of company employees EMPLOY (Spearman's Rho = 0.188*, p < 0.01, 2-tailed). This result of correlation suggests that the greater the company's total assets, the higher the company's number of employees, and vice versa. Similar significant positive correlations are found between the variable of size and the variables of EXPRIENCE, market and OWNERSHIP. The output of Spearman's Rho test further reveals that there is insignificant

⁶⁶ The requirements and interpretation of Spearman's Rho are discussed in Chapter Six: Subsection 6.10.1.2.1.2.
correlation between size and income of REVENUE (Spearman's Rho = -0.173^* , p < 0.05, 2-tailed) and the number of INDUSTRY of which companies have facilities (Spearman's Rho = -0.046, p > 0.05, 2-tailed).

Characteristics	Spearman's Rho (2-tailed) (N=203)							
Characteristics	SIZE	EMPLOYEES	EXPERIENCE	REVENUE	MARKET	INDUSTRY	OWNERSHIP	
Size ^a	-	$r = 0.188^{**}$ p = 0.007	r = 0.025 p = 0.722	r = -0.173* p = 0.013	r = 0.074 p = 0.294	r = -0.046 p = 0.516	r = 0.056 p = 0.428	
Employees		-	r = -0.012 p = 0.868	r = -0.002 p = 0.976	r = -0.069 p = 0.329	r = 0.002 p = 0.974	r = 0.080 p = 0.258	
Experience			-	r = 0.082 p = 0.246	r = 0.050 p = 0.481	r = 0.077 p = 0.274	r = -0.018 p = 0.801	
Revenue				-	r = -0.052 p = 0.464	r = -0.010 p = 0.884	r = -0.059 p = 0.402	
Market					-	r = 0.096 p = 0.175	r = -0.082 p = 0.244	
Industry						-	r = -0.117 p = 0.096	
Ownership							-	
M.C.			1 I D 111					

Notice a: Size measured by total assets in LD million.

Notice b: * Correlation is significant at the 0.05 level (2-tailed).** Correlation is significant at the 0.01 level (2-tailed).

Source: Analysis of questionnaire survey data.

The second correlation means that, the higher the number of employees in companies, the higher the number of industry and ownership which have no linear relationship correlation between a company's number of employees and INDUSTRY (Spearman's Rho = 0.002, p > 0.05, 2-tailed) and OWNERSHIP companies (Spearman's Rho = 0.080, p > 0.05, 2-tailed), while there is negative relationship between the number of employees and EXPERIENCE (Spearman's Rho = -0.012, p > 0.05, 2-tailed). Income of REVENUE (Spearman's Rho = -0.002, p > 0.05, 2-tailed) and the operative current MARKET (Spearman's Rho = -0.069, p > 0.05, 2-tailed) as correlation suggests the higher the company's number of employees, the less experience, lower rate of revenue. The third correlation means there is no linear relationship between experience and REVENUE (Spearman's Rho = 0.082, p > 0.05, 2-tailed), market (Spearman's Rho = 0.050, p > 0.05, 2-tailed) and number of INDUSTRY (Spearman's Rho = 0.077, p > 0.05, 2-tailed). There was no significant correlation between the amount of experience and number of OWNERSHIP companies (Spearman's Rho = -0.018, p > 0.05, 2-tailed). The fourth correlation suggests there is insignificant negative correlation between a company's revenue and the degree of MARKET (Spearman's Rho = -0.052, p > 0.05, 2-tailed), INDUSTRY (Spearman's Rho = -0.010, p > 0.05, 2tailed) and OWNERSHIP companies (Spearman's Rho = -0.059, p > 0.05, 2-tailed). The fifth correlation means that the higher the number of operating markets, the higher the number of industries of which there is no linear correlation between current market and the number of INDUSTRY companies (Spearman's Rho = 0.096, p > 0.0960.05, 2-tailed). There was insignificant correlation between market and number of OWNERSHIP companies (Spearman's Rho = -0.082, p > 0.05, 2-tailed). The final correlation means that there is negative correlation between the number of industry and the OWNERSHIP companies (Spearman's Rho = -0.117, p > 0.05, 2-tailed).

Table K 2.3	Company	Ownership a	nd Other (Company-sp	ecific Cha	racteristics

Characteristics	Ownersl	hip Categor	ies (Median),	(Valid N=2	03) Com	panies
Characteristics	Public $(N = 95)$		Private ($N = 108$)			
SIZE	29.0)4		32.06	5	
EMPLOYEES	110)4		746		
EXPERIENCE	7			9		
REVENUE (per cent)	14.6		21.0			
MARKET	9		11			
	Cross-tabl	ulation (Ind	ustry/Ownersh	nip)		
INDUCTOV	Number of Companies					
INDUSTRI	Frequency	Per cent	Frequency	Per cent	Total	Per cent
Manufacturing	39	45.3	47	54.7	86	100
Non-manufacturing	56	47.9	61	52.1	117	100

Source: Analysis of questionnaire survey data.

For further analysis of the characteristics of the responding companies, in order to gain a rich description of capital market practices, responses were analysed based on sector (SIZE, EMPLOYEES, EXPERIENCE, REVENUE, MARKET), OWNERSHIP (public and private companies) and INDUSTRY (manufacturing and non-manufacturing). In doing so, the *median*, as displayed in Table K 2.3, is an analysis of private companies owning more total assets (*median* 32.06 Million LD), have more years in experience (*median* 9), greater revenue from total productivity (*median* 21.0) and operate in more markets (*median* 11) than is the case for public companies. These companies, however, have more employees (*median* 1104) than their private contemporaries (*median* 746). There are a few differences between private and public companies in terms of employment, goals and receipt of government subsidies. Sun et al. (2002) argued that public companies differ in terms of choice of social and political goals, while private companies are more concerned about the ability to perform in terms of employment of staff than public companies. Studies by Dewenter and Malalatesta (2001) and Sun et al. (2002), among others, conducted empirical support for the proposal that public ownership is less efficient than private.

Furthermore, development of the public manufacturing sector in Libya featured rapid decisions incorporating most industrial and trade activities with the reverse decision to privatise. This decision started simply, featuring Act no.9 of 1992 in relation to engagement in economic activities, leading to the more comprehensive GPC resolution (no.30) of 2003 issuing regulations to organise ownership of companies and other public economic units. Table K 2.3 illustrates cross-tabulating company's industry categories with ownership and that there are forty-seven (54.7 *per cent*) private manufacturing companies versus thirty-nine (45.3 *per cent*) public, while the vast majority of responding private companies sixty-one (52.1 *per cent*) are non-manufacturing, versus fifty-six (47.9 *per cent*) public companies of disparate sizes.

K.3 Practices of the Libyan Economic Characteristics

K.3.1 Other Recommended Economic Reforms

K.3.1.1 Financing Policy

Questionnaire Findings

Respondents to the questionnaires were presented with six questions-derived potential reasons likely to cause Libyan companies to issue shares. The respondents were asked to rate degrees of concern relating to these reasons on a 5-point rating scale. Based on the means of concern indicated towards share issuing policies, presented in Table K 3.1, the highest ranked statement was, 'Fulfilling legal requirements regarding capital is ranked as the most important factor' (mean 4.01). Analysis reflects that the second most important factor was 'Issuing shares gives investors a better impression of a firm's prospects than using debt' (mean 3.96). These results were crosschecked with interview results which illustrate that there are more supply-side effects influencing financing policy than demand-side effects in the Libyan business environment as respondents place a high value on the factor relating to inability to secure funds via other sources of finance.

Code Rank		Statements		Valid N=203	
coue	Runn			Means	SD
B31.1	5	Product environment	67.5	3.7438	1.22005
B31.2	3	Whether recent profits have been sufficient to fund activities	75.9	3.9163	1.12922
B31.3	6	Shares are the cheapest source of funds	64.0	3.6601	1.24985
B31.4	4	Shares are the least risky source of funds	73.4	3.8719	1.15328
B31.5	2	Issuing shares gives investors a better impression of a firm's prospects than using debt	78.3	3.9606	1.13819
B31.6	1	Fulfilling legal requirements regarding capital	79.3	4.0148	1.06927
Notico	at T	he percentage of respondents who seered 4 "agree" and 5 "strongly agree	»?"		

Table K 3.1 Descriptive Statistics for Financing Policy for Issuing Shares

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree". Source: Analysis of questionnaire survey data.

Interview Findings

Respondents from the companies involved were further asked to indicate techniques relating to the financing policy in which their companies operate. As interviewees were not required to tick or rate a pre-formulated written set of answers, the relevant measure was the number of times any given technique was mentioned. The interview findings, as depicted in Table K 3.2, demonstrate that shares are the right of participating in a company which is the instrument which proves that right negotiable according to the rules of commercial law, and a

contributor right to participate in public associations, the right to vote in the election and the right of priority in subscription when the capital increases, as it affords the right to receive a portion of profits when a company goes into liquidation.

Table K 3.2 Respondents' Interviews to Financing Policy				
	Policy of Financa	Valid N=11 intervie	ws ^a	
	Toncy of Finance	Number of Mentions	Per cent	
Any financ	cing policy	11	100.0	
Any non-fi	inancing policy	3	27.3	
Notice a:	Details totalled more than	N = 11 and 100 per cent due	to duplicate resp	

Notice a: Details totalled more than N = 11 and 100 *per cent* due to duplicate responses. Source: Analysis of the data obtained from semi-structured interviews.

Interviewees were further required to explain reasons behind relying on particular techniques. The common justification for relying on financing policy for interviewees was the factors influencing the rise or decline in stock index and, thus, the specific information required (Respondent F). Other significant policies are: the performance of the company and management efficiency (Respondent F); profitability of the company (Respondent H); financial position (Respondent J); the company's growth prospects and the performance of the company to which it belongs in a certain period of time where there is a sector of significant growth or contraction (Respondent I). The country's economy: the country's economic situations always affect share prices causing them to rise when the economy is recovering and to fall when the economy is in recession (Respondent B); affect share prices positively or negatively (Respondent E).

K.3.1.2 Firms' Decisions

Questionnaire Findings

Respondents to the questionnaires were presented with six questions-derived potential about firms' decisions concerning issuing shares. The respondents were asked to rate degrees of concern relating to these reasons on a 5-point rating scale. Based on the means and *per cent* of concern indicated towards firms' decisions, presented in Table K 3.3, the highest ranked statement was, 'To maintain increased market share to satisfy the customer perspective' (mean 3.99, *per cent* 79.3). Analysis reflects that the second most important factor was 'To strengthen the relationship between the public and private sector' (mean 3.91, *per cent* 75.3). These results were crosschecked with interview results which illustrate that the most active factors in these decisions are to increase stock market efficiency and reduce wastage. This finding is consistent with the conclusion of Harrison and Paton (2004: 220) who noted that evidence of stock market efficiency is helpful in deciding whether transition economies in Europe are able to satisfy the key criterion of having a functioning market economy as stipulated in the Copenhagen Criteria for European Union membership. Lim et al. (2007) argued that an efficient stock market encourages efficiency in the allocation of resources and would better equip an economy to cope with the competitive pressure within the Union.

Codo Ponk		Stataments		Valid N=203	
Coue	Kalik	Statements		Means	SD
B32.1	4	To increase stock market efficiency and reduce wastage	69.9	3.8128	1.16654
B32.2	1	To maintain increased market share to satisfy customer perspective	79.3	3.9852	1.14095
B32.3	2	To strengthen the relationship between public and private sector	75.3	3.9064	1.19243
B32.4	6	To add strength to the company reputation	65.5	3.6995	1.17032
B32.5	3	To improve stock market service quality	72.9	3.8571	1.17072
B32.6	5	To encourage quality awareness within stock markets	69.5	3.8128	1.17078

 Table K 3.3 Descriptive Statistics for Most Important Factors for Firms' Decisions

Notice a: The percentage of respondents who scored 4 "important" and 5 "very important". Source: Analysis of the data obtained from semi-structured interviews.

Interview Findings

Respondents in the firms interviewed were additionally asked from where they obtained information about decisions which their firms made. Since interviewees were not required to rate or tick a pre-formulated written set of answers, the relevant measure was the number of times any given source was mentioned. Reduction in the nominal value of shares raised will, without doubt, broaden the base of shareholders in the company that could lead to a degree of balance via the distribution of shares in the hands of a large base of contributors (Respondent

F). At the level of primary market, "market version" will, without doubt, reduce the nominal value of shares to one LD, would greatly assist in facilitating the initial proposal and would collect as much as possible from the investments of small investors who would benefit greatly from reducing the nominal value of shares (Respondent C). On the level of market circulation, it is supposed to be a decision by the financial market, based on the new corporate system in their company (Respondent D). "I believe the secondary market will greatly benefit from retail, as it will make it accessible to small investors who were reluctant to engage in some of the shares due to the high values based on previous nominal value" (Respondent B). Interviewees were additionally required to explain their concerns pertaining to decision-making within firms. Retail shares will lead to the entry of new investors to the market assisting reducing the margin of fluctuation that could lead the entire market to a new phase of balanced investment, decrease speculation over shares and reduce the proportion of speculative margin in market. This, in turn, may bring additional benefit to the market through traffic balanced low-risk investors (Respondents C and F).

K.3.1.3 Histogram of Economic Reform Programme Independent Variables

Figure K 3.1, labelled *Histogram*⁶⁷, displays the *frequency* distribution of the normal distribution curve. *SPSS* draws a curve on the histogram to show the shape of this distribution. Residuals occurring at 4.107 are beyond the right tail of the curve; furthermore, many of the residuals are fairly close. Some of the columns of residuals are over the curve and others are below it.



Figure K 3.1 The Histogram of Normal Distribution of ERP Independent Variables

Source: Developed from questionnaire survey data analysis.

K.3.2 Macro-economic Reform Condition

K.3.2.1 Contribution of Output Factors and TFP Growth to Economic Growth

Questionnaire Findings

Respondents to the questionnaires were presented with five questions-derived potential contribution of factors output and TFP growth to economic growth. The respondents were asked to rate degrees of concern relating to these contributions on a 5-point rating scale. Based on the means of concern indicated towards share issuing policies, presented in Table K 3.4, the highest ranked statement was, 'Growth of output' (mean 3.90). Analysis reflects that the second most important factor was 'Contribution of capital' (mean 3.85). The statement, 'Contribution of human capital' (Mean 3.70) was ranked lowest and therefore these laws were considered the least important source of TFP growth of those listed. According to North (1991), stock market creation can increase the rate of economic growth by reducing the costs of exchanging ownership rights in firms. Another study by Bencivenga and Smith (1991) classified new stock markets as increasing economic growth by lowering the holdings of liquid assets and increasing the growth rate of physical capital. These results were checked with the interview results which demonstrate that the most active factors in these decisions are to improve growth of output and increase capital of accumulation.

⁶⁷ The rationale and assumptions of using a *Histogram* were discussed in Chapter Six: Subsection 6.10.1.2.2.

Cada	Donk	Statements	07. a	Valid N=203	
Coue	Nalik	Statements	70	Means	SD
C.38.1	1	Growth of output	77.4	3.9015	1.00992
C.38.2	3	Contribution of labour	70.5	3.7340	1.16389
C.38.3	2	Contribution of capital	76.3	3.8522	1.17643
C.38.4	5	Contribution of human capital	67.9	3.6995	1.24811
C.38.5	3	Total Factor Productivity growth (TFP)	70.5	3.7340	1.16389
NT. (1	T1				

Table K 3.4 Descriptive Statistics for Contribution of Output and TFP in Economic Growth

Notice a: The percentage of respondents who scored 4 "increase" and 5 "strongly increase". Source: Analysis of the data obtained from semi-structured interviews.

Interview Findings

Interviewees were requested to indicate techniques they relied on most in the contribution of output factors and TFP growth involved in their economic growth. As they were not required to tick or rate a pre-formulated written set of answers, the relevant measure was the number of times any given technique was mentioned. The economy is composed of input resources and capabilities and capacities of physical, human capital and labour available to the community, and termed "factors of production", and activities and events conducted by various institutions, units and establishments in the community and to crystallise these activities in three major economic areas: "investment, savings and consumption". Accordingly, a product of economic processes is output and goods and services crystallised (Respondent M).

K.3.2.2 Macro-economic Reform Risks

Questionnaire Findings

Respondents to the questionnaire survey a list of four questions-derived macro-economic risks. They were requested, with reference to the macro-economic reform, to rate degrees of concern on a 5-point rating scale. Based on the ranking and *per cent* of the macro-economic risks score, as presented in Table K 3.5, the macro-economic risks issues that concerned respondents the most were firstly that 'Foreign rates' (*per cent* 80.3). The next highest rank concerned 'Indirect instruments of monetary policy' (*per cent* 77.4). The third highest ranking is that 'Commodity prices' (*per cent* 64.1). Consistent with the previous study, Palmer (1971: 32) defines risk-taking as "the willingness to commit to a course of action which may result in rewards or penalties associated with success or failure". Subsequently, Weston and Brigham (1979) argue that a firm's capital structure represents the financial risk that firm could face. In other words, as stated by Barton and Gordon (1987), the amount of funds that could be borrowed by the companies depends, to some extent, on the amount of risk these companies can bear, and, therefore, the top management's risk taking propensity will affect the firm's capital structure.

	Table K 5.5 Descriptive Statistics for Macro-economic Kisks							
Cada	ada Dank Statamanta		07 a	Valid N=203				
Code	Kalik	Statements	-70 -	Means	SD			
C.39.1	4	Interest rates	62.6	3.6601	1.19725			
C.39.2	1	Foreign rates	80.3	3.9951	1.11470			
C.39.3	3	Commodity prices	64.1	3.5764	1.28504			
C.39.4	2	Indirect instruments of monetary policy	77.4	3.9458	1.28504			

Table K 3.5 Descriptive Statistics for Macro-economic Risks

Notice a: The percentage of respondents who scored 4 "very concerned" and 5 "extremely concerned". Source: Analysis of the data obtained from semi-structured interviews.

Interview Findings

The general managers of CBL were asked about the macro-economic reform risks which most concerned them. Since interviewees were not required to rank or order or rate risks, the relevant measure was simply the number of times any given risk was mentioned. Interest rates have a significant impact and substantial pricing of securities, the amendment to attract investment portfolios and savings or expulsion and to determine the level of the stock market, to attract capital investment, to facilitate bank credit profitability of the banking sector and growth rates in the national economy.

K.3.2.3 The Most Problematic Issues/Obstacles for Conducting Business with Libya as a Country/Economy

Questionnaire Findings

Questionnaire respondents were presented with a list of fourteen questions-derived obstacles to conducting business with Libya as a country/economy. They were requested to rate degrees of concern with the obstacles on a 5-point rating scale. Based on the means and *per cent*, as presented in Table K 3.6, the areas seen as presenting the greatest obstacles were 'Inadequate supply of infrastructure' (mean 4.21, 87.1 *per cent*); 'Access to financing' (mean 4.18, 86.2 *per cent*); 'Voice and accountability' (mean 4.14, 84.2 *per cent*); 'Inefficient government bureaucracy' (mean 4.06, 82.3 *per cent*) and 'Corruption' (mean 3.89, 76.4 *per cent*). The statements, 'Crime and theft' (mean 1.10, 5.9 *per cent*) and 'Government instability/coups' (mean 2.12, 6.9 *per cent*) were ranked lowest and, therefore, these issues were considered by respondents least likely to obstruct business with Libya. Tulba and Fhaima (2004) argue that the Libyan economy achieved high growth rates during the 1970s (the first and second development plans), suffered negative growth rates in the early 1980s and achieved slightly positive rates between 1985 and 2002. However, it is argued that most of the objectives of the development plans have not been achieved due to the shortage of skilled and semi-skilled manpower in the Libyan public sector (Saleh, 2001) and the domination of the state over economic activities that led to the misuse of economic resources, lower productivity levels, higher production costs, lower quality, weak control in the public sector and lower return on capital (Alqadhafi, 2002).

Cada	odo Statamenta		Valid	N=203
Code	Statements	70	Means	SD
C.40.1	Policy instability	71.9	3.8227	1.18909
C.40.2	Inefficient government bureaucracy	82.3	4.0640	1.07207
C.40.3	Restrictive labour regulations	53.7	3.4138	1.28826
C.40.4	Inadequate supply of infrastructure	87.1	4.2069	0.94740
C.40.5	Inadequately educated workforce	53.7	3.4138	1.28826
C.40.6	Corruption	76.4	3.8818	1.01757
C.40.7	Foreign currency regulations	62.1	3.5714	1.26982
C.40.8	Poor work ethic of national labour force	37.5	3.1330	1.33759
C.40.9	Government instability/coups	6.9	2.1182	0.96770
C.40.10	Tax regulation	9.3	2.2217	1.05545

Table K 3.6 Descriptive Statistics for most Problematic Issues/Obstacles for Conducting I	Business v	with
Libya as a Country/Economy		

Notice a: The percentage of respondents who scored 4 "obstacle" and 5 "major obstacle". Source: Analysis of the data obtained from semi-structured interviews.

5.9

32.0

86.2

84.2

1.9951

2.7586

4.1773

4.1429

0.96214

1.35919

0.94820

0.93039

Interview Findings

C.40.11

C.40.12

C.40.13

C.40.14

Crime and theft

Access to financing

Voice and accountability

Tax rates

Managers in the companies interviewed were asked about key problematic issues involved in business with the Libyan economy which most concerned them. Since interviewees were not required to rank the obstacles in order, the relevant measure was simply the number of times any given obstacle was mentioned. Inadequate supply of infrastructure, as reported in Table K 3.7, headed the list, followed by a lack of data and information required by investors (reported by seven respondents) and lack of coordination between administrative authorities and corruption (reported by six interviewees).

Table K 3.7 Respondents' Views on most Problematic Issues/Obstacles for Conducting Business with Libya as a Country/Economy

Obstaalos	Valid N=12 interviews ^a		
Obstacles	Number of Mentions	Per cent	
Inadequate supply of infrastructure	8	66.7	
Lack of data and information required by investors	7	58.3	
Lack of coordination between the administrative authorities	6	50.0	
Corruption	6	50.0	
Policy instability	4	33.3	
Access to financing	3	25.0	

Notice a: Details totalled more than N = 12 and 100 *per cent* due to duplicate responses. Source: Analysis of the data obtained from semi-structured interviews.

K.3.2.4 Histogram of Macro-economic Reform Independent Variables

Figure K 3.2, labelled *Histogram*, illustrates frequency distribution of the normal distribution curve of macroeconomic reform independent variables. SPSS draws a curve on the histogram to illustrate the shape of this distribution where residuals were occurring at (3.7927), beyond the right tail of the curve and many of the residuals are quite closely placed. Furthermore, some of the columns of residuals are above the curve and others below it.





Developed from questionnaire survey data analysis. Source:

K.3.3 **Financial Maturity Practice**

K.3.3.1 The Choice of Sources of Funds

Questionnaire Findings

Respondents to the questionnaires were given a list of five questions-derived financial sources of funds. The respondents were asked to rate degrees of concern relating to these financial sources on a 5-point rating scale. The findings, presented in Table K 3.8, show respondents preferred to use in their companies 'Bank loans' (38.9 per cent), 'Retrained earning' (41.9 per cent), 'Trade credit (supplies)' (35.5 per cent), 'Government subsidies' (27.1 per cent) and 'Foreign sources' (19.7 per cent). Meanwhile, the most successful factor was rated as 'Retrained earning' (32.5 per cent), followed by 'Bank loans' (30.0 per cent). Michaelas (1998) argues that companies will be more dependent on bank credit as they develop. This may imply that Libyan companies prefer to be financed in order of inside funds, outside trade credit from suppliers, outside debt and outside equity is more favoured by non-manufacturing companies. 'Trade credit (supplies)' and 'Foreign sources' were ranked lowest and therefore these sources were considered the least preferred choice of funds of those listed.

	Table K 3.8 Descriptive	e Statistics	for the (Choice of S	ources o	of Funds		
Cada	Statements		Use		Success ^a			
Code	Statements	Ν	%	Valid N	Ν	%	Valid N	
D.14.1	Bank loans	79	38.9	203	61	30.0	203	
D.14.2	Retrained earning	85	41.9	203	66	32.5	203	
D.14.3	Trade credit (supplies)	72	35.5	203	52	25.6	203	
D.14.4	Government subsidies	55	27.1	203	55	27.1	203	
D.14.5	Foreign sources	40	19.7	203	39	19.2	203	

Notice a: Percentage of firms that score 3 for "used with a moderate degree of success" and 4 for "used with a great deal of success"; details totalled more than 100 per cent due to duplicate responses.

Source: Developed from questionnaire survey data analysis.

Interview Findings

Interviewees were asked how the results of choice of funds sources were used in the decision-making process. Since interviewees were not given a pre-formulated written set of answers, the relevant measure was the number of times any given use was mentioned. The source of finance was used in: "bank loans" (Respondent G); "retrained earning" of strategic planning (Respondent F); establishing "foreign sources" (Respondents C and E). Respondents (J) and (N), on the other hand, were unaware of the finance source utilised in their investment.

K.3.3.2 The Frequency of Firms' Investment

Questionnaire Findings

Respondents to the questionnaires were presented with six questions-derived frequency of investment by firms and requested to rank them in relation to the stock market performance. Based on the *per cent* of frequency of firms' investment, as presented in Table K 3.9, the frequency of firms' investment that concerns respondents the most were that 'During the analysis of each investment project' (21.2 *per cent*). Analysis reflects that the highest ranked concern was 'Often (twice per year)' (20.2 *per cent*). These results were crosschecked with interview results revealing that most firms carefully select their fund sources when financing investment projects. According to Myers (2001), investment decisions can affected by the use of long-term debt when shareholders might perceive that the gains from new investment will be used to pay off existing debtholders. This situation might lead shareholders to encourage managers to pass up profitable projects. He added that short-term debt could be used to mitigate the conflict between shareholders and debtholders.

Code	Donk	Statements		Valid N=203
Code	Kalik	Statements	Statements Per cent C	Cumulative Per cent
D.15.1	6	Never	11.8	11.8
D.15.2	4	Rarely (once per year)	15.3	27.1
D.15.3	2	Often (twice per year)	20.2	47.3
D.15.4	3	Very often (approximately every three months)	17.7	65.0
D.15.5	5	Systematically (approximately every month)	13.8	78.8
D.15.6	1	During the analysis of each investment project	21.2	100.0
		Total	100.0	

Table K 3.9 Descriptive Statistics for Frequency of Firms' Investment

Source: Analysis of the data obtained from semi-structured interviews.

Interview Findings

Respondents in firms interviewed were requested to indicate the techniques they relied on most in a firm's investment pertinent to in their financial maturity. Since interviewees were not given a pre-formulated written set of answers, the relevant measure was the number of times any use was mentioned. The most frequent reason revealed behind the occasional approach to a firm's investment was attributed during analysis of each investment project (Respondents C and F). As such, the improving financial performance has become one of the most a significant objectives, if not the most, as described by the Respondent (J), "now we are required to improve the financial performance and generate profits to make the company attractive to investors, unlike the past when the profitability was not a priority".

K.3.2.3 Comparison between Bank-based and Market-based Financial Systems for Promoting Longterm Economic Growth

Questionnaire Findings

Questionnaire respondents were presented with fourteen questions-derived comparisons between bank-based and market-based systems. They were requested, with reference to the which is better for promoting economic growth in the Libyan economy, to rate degrees of concern on a 5-point rating scale. Based on the means and percentage of bank-based ranging from 81.3 *per cent* to 29.6 *per cent* and market-based ranging from 80.8 *per cent* to 27.1 *per cent* score, as presented in Table K 3.10, the bank-based systems that concern respondents the most were that 'Mobilising saving' (mean 4.0739, 81.3 *per cent*), 'Regulatory structure' (mean 4.02, 79.3 *per cent*), 'Justifying problems associated with excessively powerful banks' (mean 3.10, 78.4 *per cent*) and 'Structure activity' (mean 3.92, 74.9 *per cent*). This was followed by market-based system issues that concerned respondents the most as firstly 'Identifying good investment' (mean 4.07, 80.8 *per cent*), then 'Allocating capital' (mean 3.99, 78.8 *per cent*), next 'Increasing transparency and accountability' (mean 3.95, 75.4 *per cent*) and finally 'Finance activity' (mean 3.84, 72.4 *per cent*).

In addition, these results provide some explanation as found in Demirgüe-Kent and Levine (2004: 1) who added: "Financial systems tend to be more market-based in higher income countries, where stock markets also become more active and efficient than banks. Financial systems also tend to be more market-based, even after controlling

for income, in countries with a common law tradition, strong protection of shareholder rights, good accounting standards, low levels of corruption, and no explicit deposit insurance". The findings are in line, too, with the view of Levine and Zerovs (1998) who explained that a stock market with greater liquidity implies faster economic growth no matter what the level of banking system development. Similar findings in the literature were found by Rajan and Zingles (1998) who empirically established that bank-based systems are better at promoting growth in countries with weak shareholder and poor legal systems and property rights. They further stated that market-based legal systems improve.

Table K 3.10 Descriptive Statistics for Comparison between Bank-based and Market-based

Cada	Statementa		Bank-	based			Marke	t-based	
Code	Statements	Rank	Means	SD	% ^a	Rank	Means	SD	% ^a
D.16.1	Mobilising savings	1	4.0739	1.03859	81.3	5	3.7143	1.10227	70.0
D.16.2	Identifying good investment	7	3.5517	1.25904	60.6	1	4.0690	1.03655	80.8
D.16.3	Allocating capital	5	3.7734	1.08449	70.9	2	3.9901	1.11244	78.8
D.16.4	Providing risk management tools	9	3.4187	1.38487	51.7	6	3.7291	1.20662	67.4
D.16.5	Exerting sound corporate control	8	3.4975	1.29880	58.1	10	3.2660	1.39596	49.7
D.16.6	Justifying problems associated with excessively powerful banks	3	3.9951	1.04122	78.4	11	3.2562	1.30627	46.2
D.16.7	Increasing transparency and accountability	6	3.6946	1.23694	65.5	3	3.9507	1.09794	75.4
D.16.8	Structure activity	4	3.9212	1.19558	74.9	14	2.8867	1.21132	27.1
D.16.9	Structure size	10	3.2512	1.41454	50.7	13	2.8325	1.33925	29.5
D.16.10	Structure efficiency	11	3.3005	1.35462	46.3	12	3.0985	1.34243	37.9
D.16.11	Structure regulatory	2	4.0246	1.07831	79.3	8	3.5222	1.26775	57.6
D.16.12	Finance activity	13	2.8621	1.32407	31.1	4	3.8374	1.21793	72.4
D.16.13	Finance size	14	2.8571	1.26032	29.6	9	3.4581	1.39020	54.2
D.16.14	Finance efficiency	12	2.9754	1.36954	38.4	7	3.6108	1.25125	63.1

Notice a: Percentage of comparison that score 1 "very unimportant" and 5 "very important". Source: Analysis of the data obtained from semi-structured interviews.

K.3.2.4 Histogram of Financial Maturity Variables

Figure K 3.3, labelled *Histogram*, illustrates the *frequency* distribution of the normal distribution curve of financial maturity independent variables. *SPSS* draws a curve on the histogram to demonstrate the shape of this distribution. Residuals were occurring at (3.7188), beyond the right tail of the curve. Additionally, many of the residuals are fairly close. Finally, some of the columns of residuals are above the curve and others are below it.





Source: Developed from questionnaire survey data analysis.

K.3.4 Practices of Libyan Stock Market Performance

The aim of this section is to present and analyse data related to the research's objective. Data was obtained from both self-administered questionnaire surveys and semi-structured interviews regarding the process of the practices of the Libyan stock market performance. The section is divided into seven main subsections. Subsection K 3.4.1 describes the ranking of most and least important for the successful establishment of a stock market in the Libyan economy. Subsection K 3.4.2 presents the importance of institutional source which may affect the stock market regarding the international business environment. Subsection K 3.4.3 presents the stock market effect. Sub-section K 3.4.4 explains the financial crisis. Subsection K 3.4.5 reveals obstacles and constraints of stock market performance. Subsection K 3.4.6 considers the best practise model experiments for the Libyan stock

market, while results of stock market assessment are reported in subsection K 3.4.7. Finally, Subsection K 3.4.8 presents the histogram of stock market performance variables.

K.3.4.1 Establishment a Stock Market in the Libyan Economy

Questionnaire Findings

Questionnaire respondents were presented with a list of ten questions-derived successful establishment of a stock market in the Libyan economy and requested to rate degrees of concern on a 5-point rating scale. Based on the rank and means, the output of tests, as presented in Table K 3.11, indicated that only four sources had a mean of more than 3.00. These sources are 'Geographical location' (mean 3.89); 'Proximity to international markets' (mean 3.80); 'Lack of collateral (security)' (mean 3.35); 'Provide new technology as the market demands' (mean 3.36); 'Lack of quality company' (mean 3.09). This finding is consistent with the conclusion of World Bank (1994) that the central geographical location of Libya between the developed economies in the West and Growing economies of North Africa has reduced transport costs and increased the significance of the Libyan oil market, which rise more foreign currency to the country. The establishment of stock markets in developing countries and the opening of them to foreign security house as well as to foreign portfolio investors can be viewed as a part of this global liberalisation project (Singh, 1992). The other six sources, however, had means of less than 3.00.

Code	Code Statements -		Valid N=20	13
Coue			Means ^a	SD
E.19.1	Lack of quality companies	4	3.0887	1.40086
E.19.2	Lack of collateral (security)	6	2.7291	1.29374
E.19.3	Lack of good trading record	8	2.5862	1.19247
E.19.4	Provide new technology as the market needs	3	3.3547	1.36517
E.19.5	Poor relationships with banks	10	2.4975	1.13615
E.19.6	Geographical location	1	3.8867	1.13101
E.19.7	Deterioration in the state of the economy	9	2.5419	1.16550
E.19.8	The suppliers of finance are in small and/or underdeveloped sector	5	2.8966	1.31033
E.19.9	Proximity to international markets	2	3.7980	1.13609
E.19.10	Inability to convince lenders of the profitability of the investments	7	2.6453	1.27517

Table K 3.11 Descriptive Statistics for Ranking of Successful Adoption of a Stock Market

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree". Source: Analysis of the data obtained from semi-structured interviews.

Interview Findings

Respondents interviewed in the stock market were requested to indicate techniques they relied on most in the establishment of a stock market. They were not required to tick or rate a pre-formulated written set of answers, the relevant measure was the number of times any given technique was mentioned. The stock market is highly critical of Libyan economic development due to the absence of stock market sources which may have negative effects (Respondent B). This may additionally encourage many significant reform programmes such as deregulation, privatisation and liberalisation, attracting foreign capital investment and establishing free zones (Respondent E). This highlights the significance of the role that could be played by the existence of an effective market for securities in Libya in the process of economic and social development especially given the natural and human resources enjoyed by Libya's excellent geographical location, near the coasts of the southern shores of Europe and, therefore, in the proximity of African and European markets (Respondents M and F). This, in addition to the availability of natural resources such as oil and gas, and chemical products, iron and steel products, provides opportunities for investment capital in the most significant areas: manufacturing, construction, fishing and desert tourism, communications and aviation industries based on oil (Respondent D).

K.3.4.2 International Business Environment

Questionnaire Findings

Questionnaire respondents were presented with a list of eleven questions-derived sources of information regarding the international business environment within the Libyan economy and requested to rate degrees of concern on a 5-point rating scale. Based on the rank and means, the output of tests, as presented in Table K 3.12, indicated that only five sources had a mean of more than 3.00: 'Companies' personnel abroad' (mean 4.05); 'Companies' personnel at headquarters' (mean 3.10); 'Banks operating locally and abroad' (mean 3.73); 'International organisations' (mean 3.44); 'Other companies operating locally and abroad' (mean 3.09). According to law number 5 in 1997, any investment project has to contribute to the economy by: (1) increasing the export of goods; (2) providing work opportunities for Libyans, enabling them to gain technical experience and to use modern technology; (3) providing services to the national economy; (4) strengthening the link between the existing economic activities; (5) making use or help in making use of local raw materials; (6) contributing to the growth and development of the remote or underdeveloped areas. In addition, the investment project may not be nationalised, seized, frozen or subjected to actions of the same impact except through the passing of a law or by a court decision. The last six sources, therefore, had a mean of less than 3.00.

Code	Code Statements		Valid N=20)3
Coue	Statements	Rank	Means ^a	SD
E.20.1	Companies' personnel at headquarters	2	3.9901	1.13447
E.20.2	Companies' personnel abroad	1	4.0542	1.09545
E.20.3	Banks operating locally and abroad	3	3.7340	1.27356
E.20.4	Other companies operating locally and abroad	5	3.1527	1.34287
E.20.5	Media (e.g. television, radio, newspapers)	6	2.7980	1.19139
E.20.6	Trade associations	9	2.1232	0.97473
E.20.7	International organisations (e.g. International Monetary Fund)	4	3.4433	1.33145
E.20.8	Business magazines	11	1.8670	0.89942
E.20.9	Academics	7	2.5665	1.16859
E.20.10	Libyan embassies abroad	10	1.9951	0.96214
E.20.11	Governmental domestic agencies (e.g. Libyan Investment Board)	8	2.2315	1.06268

 Table K 3.12 Descriptive Statistics for International Business Environment

Notice a: Percentage of comparison that score 1 "very unimportant" and 5 "very important". Source: Analysis of the data obtained from semi-structured interviews.

Interview Findings

Respondents interviewed in the business arena were additionally asked from where they obtained information about the international business environment. Since interviewees were not required to rate or tick a preformulated written set of answers, the relevant measure was the number of times any given source was mentioned. Interview findings, as displayed in Table K 3.13, suggested a large degree of dependence on human resources (i.e. a company's personnel at headquarters and abroad, managers of banking and companies locally and overseas, academics). A number mentioned non-human sources (i.e. media, magazines, trade and international organisations) as available sources of business information.

Table K 3.13 Respondents' Interviews for International Business Environment

Source of huginess information	Valid N=11 interviews ^a			
Source of business information	Number of Mentions	Per cent		
Any human source	11	100.0		
Any non-human source	3	27.3		
Notice of Details totalled more then	N = 11 and 100 ner cent due	to duplicate race		

Notice a: Details totalled more than N = 11 and 100 *per cent* due to duplicate responses. Source: Analysis of the data obtained from semi-structured interviews.

Interviewees were further required to explain why they used a particular source within the international business environment. The most significant external bank systems were reported to be "more sensitive" to political policy change than other concerns (Respondent E) and businesses within similar spheres were reported as available sources since they have "similar interests" (Respondent D). The other available source was insignificant to international business for the following reasons: media was used merely as an "indicator" but not as a major source (Respondent G); Libyan embassies were provided as "politically oriented" (Respondent L); business magazines were considered to introduce "general" information (Respondent J). Furthermore, the other sources

were not perceived as significant to respondents due to the view that academics may lack "specific knowledge" and that it was difficult to "judge" the "quality" of such individuals (Respondent M). Similarly, the government's "capability" and external consultants were also considered as hard to judge. Such agencies and consultants were thought not to take business-specific characteristics into account when introducing investment (Respondents C and K).

K.3.4.3 Stock Market Effect

Questionnaire Findings

Respondents to the questionnaires were presented with a list of three questions-derived by political, financial, cultural events and requested, in relation to the stock market effectives, to rate degrees of concern on a 5-point rating scale. Based on the ranking and percentage of effective scores, as shown in Table K 3.14, the political events that concerns respondents the most were 'Revolutions, political instability, riots, corruption, expropriation and terrorism' (75.4 *per cent*). The second highest ranked concern was 'Financial events' (71.5 *per cent*). The statement, 'Cultural events (e.g. language, education, religion)' (62.6 *per cent*) was ranked lowest. The remaining respondents represent other events such as 'Landslides; hurricane or flooding; earthquakes; volcanic activity' (10.8 *per cent*). Questionnaire respondents were additionally required to rate their agreement that a political impact stock market would pose a threat to their economy. It is further evident that cultural factors were found to have a less significant impact than political and finance events. This suggests that people take them into account largely for their economic conditions. Engrman and Sokoloff (1997; 2002), Acemoglu et al. (2001) and Easterly and Levine (2003) show that political stability, legal system, cultural and geographical factors influence the financial system and are required to understand more about the role of financial factors in the process of economic growth rate.

Table K 3.14 Descriptive Statistics for Stock Market Effect

Cada	Code Statements		Valid N=203		
Code			Rank	Means	SD
E.21.1	Political events (e.g. revolutions, political instability, riots, corruption, expropriation, and terrorism)	75.4	1	3.9310	1.08325
E.21.2	Financial events (e.g. foreign exchange rates, interest rates or commodity prices currency inconvertibility, economic sanctions, restrictions on ownership, and taxation)	71.5	2	3.8818	1.15862
E.21.3	Cultural events (e.g. language, education, religion)	62.6	3	3.6798	1.13055
E.21.4	Other events ^b (e.g. natural events such as: landslides; hurricane or flooding; earthquakes; volcanic activity)	10.8	4	3.7742	1.11683

Notice a:The percentage of respondents who scored 4 "very concerned" and 5 "extremely concerned".Notice b:Added from responses, with 32 responses from a total of 203 questionnaires distributed.

Source: Analysis of the data obtained from semi-structured interviews.

Interview Findings

Interviewees were further required to explain their concerns about stock market effectives. Nearly all those questioned had difficulty in responding. Accordingly, their concerns were "general" and could not be expressed in terms of specific effects that might cause constraints upon their operations.

K.3.4.4 Financial Crisis

Questionnaire Findings

Respondents to the questionnaires were presented with eleven questions-derived sources of financial crisis and were requested to rate their degrees of concern on a five-point rating scale, where 1 stood for "not concerned" and 5 stood for "extremely concerned". Based on the mean and *per cent* of financial crisis scores, as presented in Table K 3.15, the financial crises that concern respondents, in descending order were: 'Economic sanctions' (mean 4.21, 90.2 *per cent*); 'Political instability' (mean 4.14, 87.7 *per cent*); 'Wars' (mean 4.08, 84.7 *per cent*); 'Demonstrations, riots and insurrection' (mean 4.01, 80.8 *per cent*); 'Terrorism' (mean 3.71, 62.5 *per cent*); 'Corruption' (mean 3.17, 42.9 *per cent*); 'Investment climate' (mean 2.77, 27.6 *per cent*); 'Currency inconvertibility' (mean 2.57, 20.7 *per cent*); 'Taxation restrictions' (mean 2.51, 17.7 *per cent*); 'Ownership

and/or personnel restrictions' (mean 2.47, 15.8 per cent); 'Import and/or export restrictions' (mean 2.44, 14.3 per cent). On the other hand, Fayad (2000) states that the Libyan economy was affected by the world recession in the 1980s and, therefore, the Libyan government was unable to fund its development plans and, as a consequence, many projects were cancelled.

Cada	Statements	01	Valid N=203			
Code	Code Statements %		Rank	Means ^a	SD	
E.22.1	Political instability	87.7	2	4.1429	0.98719	
E.22.2	Economic sanctions	90.2	1	4.2118	0.94369	
E.22.3	Currency inconvertibility	20.7	8	2.5714	1.21402	
E.22.4	Import and/or export restrictions	14.3	11	2.4433	1.11270	
E.22.5	Ownership and/or personnel restrictions	15.8	10	2.4680	1.14438	
E.22.6	Investment climate	27.6	7	2.7734	1.30037	
E.22.7	Terrorism	62.5	5	3.7094	1.20200	
E.22.8	Demonstrations, riots and insurrection	80.8	4	4.0099	1.08996	
E.22.9	Corruption	42.9	6	3.1576	1.35879	
E.22.10	Wars	84.7	3	4.0788	1.01166	
E.22.11	Taxation restrictions	17.7	9	2.5123	1.16619	
Notice a:	Respondents ranked the relative im	portance of	f eleven s	sources rar	nging from	
	important" to "least important" the	is a higher	<i>mean</i> ind	icates orea	ater imports	

Table K 3.15 Descrip	otive Statistics for F	Relative Importance of	of Financial Crisis
----------------------	-------------------------------	------------------------	---------------------

he "most to least important", thus a higher *mean* indicates greater importance. Source: Analysis of the data obtained from semi-structured interviews.

Interview Findings

Interviewees were asked how the results of financial crisis affected their decision-making process. As they were not given a pre-formulated written set of answers, the relevant measure was the number of times any given aspect was mentioned. Key reasons behind financial crisis were mentioned as: "political instability" (Respondent F); "economic instability" (Respondent M). The IMF pointed out in 1998 that lack of economic stability is one of the most significant factors behind financial crisis; "investment climate" of strategic planning (Respondent D); "corruption" is economic reform programme effective (Respondent K).

K.3.4.5 Obstacles and Constraints of Stock Market Performance

Questionnaire Findings

Respondents to the questionnaires were presented with nine questions-derived potential obstacles and constraints of stock market performance to rate degrees of concern on a 5-point rating scale. Based on the mean and per cent of obstacles of the stock market score, as presented in Table K 3.16, stock market obstacles that concern respondents, in descending order, were: 'Excessive regulations and bureaucracy' (mean 4.15, 84.2 per cent); 'Lack of distinct monetary policies in relation to exchange rate' (mean 3.94, 75.4 per cent); 'Lack of data and information' (mean 3.77, 68.5 per cent); 'Disorganisation and corruption within financial institutions' (mean 3.55, 61.1 per cent); 'Political/business risks' (mean 3.40, 55.2 per cent); 'Oil price shocks' (mean 3.30, 51.7 per cent); 'Paucity of discussion stock market performance' (mean 3.23, 48.7 per cent); 'Weaknesses in skills within the workforce' (mean 3.11, 44.3 per cent). It was reported in Chapter Two that the oil discovery had brought to light a number of problems. The small population, which had been an advantage when the economy largely depended on foreign aid, became a handicap when the demand for non-agricultural labour suddenly increased (Wright, 1969). Lack of experienced management, the shortage and cost of skilled labour, competition from foreign goods and the need to import many raw materials, meant that the country's industry was in a desperate need of state support and protection for healthy survival (Wright, 1969).

Code	Statements		Valid N=203	
Coue			Means	SD
E.23.1	There are no clear monetary policies in relation to the exchange rate	75.4	3.9409	1.10644
E.23.2	There are high rates of taxes and tariffs	43.8	3.1330	1.35962
E.23.3	Disorganisation and corruption exists in financial institutions	61.1	3.5468	1.33184
E.23.4	There is excessive regulation and bureaucracy	84.2	4.1478	0.93228
E.23.5	There is a lack of data and information	68.5	3.7685	1.20666
E.23.6	There is a lack of discussion on stock market performance	48.7	3.2266	1.40645
E.23.7	There are weaknesses in skills within the workforce	44.3	3.1133	1.37947
E.23.8	There are political/business risks	55.2	3.3793	1.39268
E.23.9	Oil price shocks	51.7	3.2956	1.38640

Table K 3.16 Descriptive Statistics for Obstacles and Constraints of Stock Market Performance

Notice a: The percentage of respondents who scored 4 "obstacle" and 5 "major obstacle". Source: Analysis of the data obtained from semi-structured interviews.

Interview Findings

Interviewees were required to explain their concerns about negative effects of the obstacles of a stock market. Since interviewees were not required to rank in order or rate problems, the relevant measure was simply the number of times any given problem was mentioned. Primary reasons behind the absence of a stock market which causes many negative effects were mentioned as: unclear political and economic environment (Respondent M); delayed deregulation and privatisation programme (Respondents H and L); increase in the number of bankrupt companies in the long-term (Respondent J); led to increase in the risk of financial investment (Respondent C); scarcity of data and information (Respondent E); lack of accounting disclosure and transparency (Respondent K); oil price shocks (Respondent D).

K.3.4.6 Lessons which Could be Learned from the Experience of Other Countries

Questionnaire Findings

Questionnaire respondents were presented with a list of five questions-derived mutually exclusive choices of best model experiments for Libyan stock market performance. Based on rank the rating used was the mean and *per cent*. Findings are presented in Table K 3.17. Respondents were most concerned with 'Middle Eastern and North African' countries first (mean 4.10, 85.7 *per cent*), 'Central and Eastern European' countries second (mean 3.92, 74.9 *per cent*), 'Far Eastern' third (mean 3.76, 68.0 *per cent*), Latin American fourth (mean 3.56, 56.1 *per cent*). The least other ranked stock markets practice experience, however, was the 'Indian subcontinent' (mean 3.18, 42.9 *per cent*).

Code	Statements		Valid N=203					
Code	Statements	Rank	% ^a	Means	SD			
E.24.1	Middle Eastern and North African	1	85.7	4.0985	1.01968			
E.24.2	Far Eastern	3	68.0	3.7635	1.22404			
E.24.3	Indian Subcontinent	5	42.9	3.1823	1.30546			
E.24.4	Central and Eastern European	2	74.9	3.9212	1.13613			
E.24.5	Latin American	4	56.1	3.5567	1.19839			
NT /*	FD1 C 1 1	1 4 46 22	1 7 44	1	**			

 Table K 3.17 The Best Practise Model Experiments for the Libyan Stock Market

Notice a: The percentage of respondents who scored 4 "agree" and 5 "strongly agree". Source: Analysis of the data obtained from semi-structured interviews.

Interview Findings

Respondents interviewed within the Libyan stock market were asked to indicate the models they relied on most in learning from the experience of other countries within the domestic stock market area. Since interviewees were not required to tick or rate a pre-formulated written set of answers, the relevant measure was the number of times any given model was mentioned. Table K 3.18 demonstrates that interview findings confirm findings from the questionnaires; most interviewees were likely to rely on one or more developing country's model.

14	<i>JIC IX 5.10 IXCS</i>	politicities interviews to Auopted	u WIOUCIS
	Madala	Valid N=12 intervie	ws ^a
Widels		Number of Mentions	Per cent
Developing	country	10	83.3
Developed c	ountry	7	58.3
Notice a:	Details totalle	ed more than $N = 12$ and $100 per c$	cent due to duplicat

Table K 3.18 Respondents'	Interviews to Ado	pted Models
---------------------------	-------------------	-------------

sponses. Source: Analysis of the data obtained from semi-structured interviews.

Interviewees were additionally required to mention reasons for relying on developing country models more often than developed country models: Libya began its stock market later than most emerging countries (i.e. Egypt, Saudi Arabia, Jordan, Tunisia, UAE, etc) (Respondent M). The first practical action of the state towards a stock market started in 2006 following the new economic reform programme of 1999 vis a vis private sector enterprises in 2000 by establishment of the General Board of Ownership (GBO)68 (Respondents B and H); assessing the strengths and weaknesses of stock markets in MENA countries (Respondent C). Furthermore, to complete the picture, there is a model, modernisation, and this model is close to Libya, which depends on stabilisation of the political vision with the events of economic reforms. This model was applied in China (Respondent M), so there is the circulation of policy power (Respondent B), the task of development in the issue of economic reform, if we report the recent economic development Libya evolved in 2004 to 2006 significantly (Respondent H). What happens then in Libya needs a certain degree of attention, the model of modernisation opportunity for technical institutions within Libyan society, besides agreement on a political vision, and identifying whether we require dialogue to reform or modernise, and finally the vision of political action. Perhaps the issue is not economic but primarily we need to modernise the political wheel (Respondents A and M).

K.3.4.7 Report of the Stock Market Assessment

Respondents to the questionnaires were required to state the title(s) to which the outcomes were reported. Findings within the organisation that assign formal responsibility to personnel in order to conduct the stock market performance are presented in Table K 3.19. 78.6 per cent of respondents reported to 'General managers', 64.3 per cent to a 'Chairperson or chief executive' (50.0 per cent), (42.9 per cent) 'Investment manager', (35.7 per cent) 'Planning manager', 'Financial management' 28.6 per cent, 'Risk management' 14.3 per cent, and 'Legal management' 7.1 per cent.

Cada	<u>Statement</u>	Valid	<i>l N=14</i> ^a	_
Code	Statements	Ν	Per cent	_
E.25.A	General manager	11	78.6	
E.25.B	Chairperson	9	64.3	
E.25.C	Chief executive	7	50.0	
E.25.D	Investment manager	6	42.9	
E.25.E	Planning manager	5	35.7	
E.25.F	Financial management	4	28.6	
E.25.G	Risk management	2	14.3	
E.25.H	Legal management	1	7.1	
Notice a:	Details added from compa	nies to more th	an $N = 14$ or	100 per cent due to
	reproduction responses.			
Source:	Analysis of the data obtain	ed from semi-s	tructured inte	erviews.

K.3.4.8 Histogram of Stock Market Performance Variables

The Histogram in Figure K 3.4 displays the frequency distribution of the normal a bell-shaped distribution curve of stock market performance dependent variables. SPSS draws a curve on the histogram to demonstrate the shape of this distribution the residual were occurring at (3.7104), beyond the right tail of the curve. Also, many of the residuals are fairly close. Furthermore, some of the columns of residuals are above the curve and others are under it.

⁶⁸ GBO was established by the Libyan government, which is directly affiliated to the GPC to implement and control privatisation in accordance with GPC decree no.198 of 2000.



Source: Developed from questionnaire survey data analysis.

K.4 The Data Collected by Interviews

This part encompasses the analysis and interpretation of data gathered from this study, as outlined in the methodology and presentation chapters⁶⁹. Specifically, the appendix contains the following points:

- An analysis of the interviews conducted with Libyan experts to investigate the impact of an emerging stock market in the Libyan economy
- Survey statements investigated by managers of financial and non-financial institutions.

As mentioned in Chapter Six, a series of semi-structured interviews were used to gather data concerning the research questions related to this part of the study, from each participant within the research study. Interviews were conducted between September 2007 and the end of January 2008. Each interview included 8 questions and took, on average 30 to 60 minutes. Three significant factors helped the researcher to conduct interviews successfully. Firstly, the interviews were conducted in Arabic i.e. the researcher's mother tongue, which enabled the researcher to understand each statement taken during interviews. Secondly, the researcher believed much of what Easterby-Smith et al. (2002) emphasised regarding the importance of arranging interviews with enough time between them to allow the researcher sufficient opportunity to write or type a transcript or set of notes and to consider the data, and to analyse and explore some of the related issues. Finally, the researcher had predetermined the questions to be asked in all the interviews, since their nature was semi-structured, which facilitated the analysis process to some extent. All interview participants provided valuable and interesting insights for the research project. Interview questions were divided into three sections. Section K 4.1 represented the importance of establishing a stock market in Libya. Section K 4.3 illustrated the expected role of policies of the Libya stock market.

K.4.1 The Importance of Establishing a Stock Market in Libya

According to the first question related to establishing a stock market in Libya, Table K 4.1 shows that most interviewees agreed that the stock market is very important to the Libyan economy and is needed for increasing the growth rate because the absence of a stock market causes many negative effects. Some argued that it caused an increase in the number of bankrupt companies over a period and encouraged the capital sums to invest abroad.

⁶⁹ The reasons favouring the use of personal interview method: the nature of the research, which may sometimes require accurate and detailed information, lack of information, the desire to acquire new information that has not been published and the desire to verify information related to the study.

Code Statements		Valid N=14 interviews ^a				
coue	Suunnas	Number of montions	Por cont			
Q 1 Do you think a stock market is needed in Libya? And why?		Number of mentions Fer ce				
Very Import	ant	12	86			
Not Importa	nt	2	14			
	Total	14	100			
Notice a:	Notice a: Details totalled more than $N = 14$ and 100 <i>per cent</i> due to duplicate responses.					

Table K 4.1 Respondents' Responses to Question 1

Source: Analysis of the data obtained from semi-structured interviews.

In response to question two (Table K 4.2), most participants believe that the stock market should be independent and not controlled by the Libyan Central Bank, but that it might be inspected by other institutions, such as the Economic Fund and Social Development or the Ministry of Economy and Trade to protect the investors.

Table K 4.2 Respondents' Responses to Question 2

Code	Statements	Valid N=14 interviews ^a	
Q 2	The majority of Libyans state "The Libyan Central Bank control the Libyan stock market" Do you agree with that? Why? If you disagree, do you think the stock market should be independent or controlled by others (the Economic Fund and Social Development or Ministry of Financial Control?	Number of mentions	Per cent
Controlled by Libyan Central Bank		2	14.3
 Independent 	nt without any control	4	28.6
 Independent and inspected by the Economic Fund and Social Development 		1	07.1
 Independent and inspected by the Economy and Trade ministry 		7	50.0
	Total	14	100.0
Mating	Details totalled means then N 14 and 100 means and does to dealisate as		

Notice a: Details totalled more than N = 14 and 100 per cent due to duplicate responses. Source: Analysis of the data obtained from semi-structured interviews.

As mentioned early in Chapter Two, the Libyan General Peoples' Committee issued Resolution 49 in 2002 concerning the executive regulations of law 21 in 2001. Articles 48-50 of this resolution allow the establishment of a stock market. Based on this resolution and Resolution 242 in 2003, the committee awarded the Libyan Central Bank the right to establish a stock market. However, Resolution 105 in 2005 cancelled that decision and gave it to the Libyan General Peoples' Committee for Economy and Trade. Consequently, the Libyan stock market did not start working process until the end of 2006. Respondent interviewees were asked in question three about the main prerequisites for a successful stock market in Libya. Interview findings mentioned that the Libyan stock market is expected to encourage investors to invest in securities (Respondent A), regulate the isse, and dealing of securities, allow easy liquidation of securities (Respondent D) and improve the level of financial disclosure and competition between Libyan economic sectors (Respondent K). Interviewees were further required to explain their concerns regarding the key prerequisites. Libya, like many other developing countries, has made remarkable progress in privatising its enterprises by transferring ownership from the state to private citizens (Respondent H). This has created many new shareholders (Respondent J). Now Libya needs to develop the stock market and related institutions to handle the large volume of share trading that is likely to occur after further privatisation programmes (Respondent L).

K.4.2 The Relationship between the Libyan Economic Reform Programme and the Stock Market

Interviewees were required to explain why there are differences among countries with regard to the information needed from the market-based system (Respondent F) and any international comparison of stock markets should take into account socio-economic effects, especially the change from a centrally-planned to a market-based system (Respondent F). These differences are due to factors such as national culture, corporate culture, legislation and national institutions. The change in the market-based system in the developing economy of Libya, has experienced a huge reform from a centrally-controlled to a market-based system and has the distinctive characteristic of being heavily reliant on the oil industry as the main resource for the economy (Respondent M). Table K 4.3 shows that, when an investment bank brings a new company to flotation, its shares are issued on the primary market. If this company subsequently decides to raise further finance by issuing bonds, these are also sold on the primary market. Similarly, if a company decides to expand using either equity finance or bond finance, the additional shares or bonds are traded on the primary market. The secondary market is that in which existing securities are subsequently traded. As mentioned in Chapters Two and Five, the Libyan stock market will

play an important role in Libyan economic growth rate by speeding up economic reform programmes, increasing the GDP, helping in restructuring the Libyan economy and other reform programmes to proceed faster.

Code	St	Valid N=14 interviews ^a					
Q 4	What is the relationship between the Libya performance and could you please indicate th impact of the stock market in relation to the fo	Number of mentions	Per cent				
		High	Low	Do not	No		
		impact	impact	know	impact		
Financi	al turnover of the organisation	4	3	2	1	10	71.4
Reputat	tion of the organisation	6	2	4	0	12	85.7
Quality	of goods and products	6	1	3	1	11	78.6
Profital	bility	7	3	2	2	14	100
Product	tivity	8	1	3	0	12	85.7
Market	share	3	3	5	2	13	92.9
Employ	vee satisfaction	4	2	1	1	8	57.1
Cost re	duction	7	1	3	2	13	92.9

Table K 4.3 Respondents' Responses to Question 4

Notice a: Details totalled more than N = 14 and 100 *per cent* due to duplicate response. Source: Analysis of the data obtained from semi-structured interviews.

Respondents interviewed were asked in question five to indicate the effectiveness of the post-sanction economic reforms in transforming the Libyan economy from bank-based to one that is market-based. Respondent D explained that different sanctions were imposed on Libya by the US and the UN. The US banned imports of Libyan crude oil in 1981 and extended it later to include direct trade, commercial contracts and travel activities. The UN imposed an embargo in 1992 after the accusation of two Libyan citizens of involvement in the crash of an American aeroplane in Scotland in 1988. The UN embargo was eased in 1999 and completed in 2003 after the country accepted responsibility for the crash, while the US embargo was ended in 2004 (Respondent A).

Interviewees were additionally required to explain their concerns, especially in the way the former socialist and transition countries have demonstrated the relationship between economic reforms and changes in the marketeconomy system. China, for example, which is considered to be one of the fastest-growing economies in the world (Respondent M), provides an example of the impact of business environment change on economic reform programme. Respondent D argued that the phenomenon of economic reform from a centrally-planned to a capitalist system had caught the attention of researchers, especially in the field of micro and macro-economics, since the start of economic reforms in 1979. Respondent M highlighted the main features of the Chinese accounting system, including the role of marketing policy, the regulatory environment, ownership methods, and recent improvement in the system. Respondents M and H argued that the continuing change in the Chinese economic system since liberalisation has influenced economic reform practices. Interviewees were also required to explain concerns relating to the Libyan economy framework. The process of transforming the Libyan economy from a centrally-planned to a market-based system, which was launched in the late 1980s and involved fundamental change in the regulatory context, has terminated domination of the state-owned sector over the Libyan market, which proved to be responsible for many deficiencies obstructing the growth of the economy (Respondent H and M). Hence, a new business environment has been created, within which Libyan companies, especially those that are state-owned, encounter increased competition causing deterioration in their financial performances (Respondent F).

Interviewees were additionally required in question six to explain why financial maturity has altered. Respondent M emphasised that Libyan financial maturity has changed as a result of economic policies. After the revolution the government required all banks operating in Libya to be under state control (Respondent E). Respondents A and E argued that by 1970 all banks became state-owned enterprises, including foreign banks whose names were then changed. For instance, Barclays became Al-Jamahiriya and Banco Di Roma became Al-Ummah Bank. By the process of merging, the number of commercial banks was reduced to five: three of these (National Trade Bank, Jamahiriya Bank and Ummah Bank) are completely state-owned and at least 70 *per cent* of the shares in the other two (Sahara Bank and Wahda Bank) are state-owned. Moreover, the Bank of Commerce and Development was established in 1996 and currently comprises five branches and four agencies in different regions in Libya. This bank is considered to be the fastest growing bank in Libya, with advanced technology and services crowned by the issuing of Visa Cards for the first time in the country in 2005. Interviewees were further required to explain why, despite government efforts to build a strong financial sector that could support private

investment initiatives (Respondent F), the Libyan banking sector plays no significant role in improving the economy and is still in its very early stages compared to those of developed and developing countries (Respondent E). However, this liquid industry is very promising for foreign investors as it generates 60 to 70 billion US\$ in revenue each year (Respondent C).

K.4.3 The Expected Role of Policies of the Stock Market in Libya

In question seven respondents were asked about the type of techniques they use in their company's activities in the financial market and whether the decision-making process of the financial market copes with the firms in the market. As pointed out by Respondents D and F, there is no evidence of the use of so-called advanced management decision techniques, such as activity-based costing and the balanced scorecard, the financial perspective from the view of the shareholders, the customer perspective, the strategy for creating value in the customer's view, the internal business perspective in transitional and developing countries. To obtain factual evidence, respondents were asked whether they use or have considered using these techniques. Respondent L argued that the changes introduced in the state regulations have transferred these firms from socio-oriented to profit-oriented entities. Government support has been eliminated and greater autonomy has been given to the management of these firms to control their operational activities, including the product-pricing process, deregulation, privatisation of the economy and market competition. The other organisation-specific factors to consider are those such as ownership, industry type, size and several inter-organisation factors, such as dissatisfaction with existing management policy, change in organisational structure, availability of competent firms, encouraging top management and alterations in production methods (Respondents G and H). Regarding the final question related to strengths and weakness in the structure of the Libyan economy, the answer is primarily based on the results of interview analysis as reported in Table K 4.4.

Table K 4.4 Respondents' Responses to Question 8

Code Statements	Valid N=14 interviews ^a
Q 8 In your view, what are the strengths and weaknesses in the structure of the Libyan economy, and wh needed in your organisation to improve financial market activities?	hat efforts are Strengths Weaknesses
• Offering many alternative solutions to the problems that face the Libyan economy such as (the weakness of the creation of new jobs and a reduction in implement rate and transferable ownership enterprises.	eLD, ✓
• Encouraging increased production and improvements in the level of competition within the Libyan economy.	\checkmark
• Reduce the risk of financial investment and provide improved production for investors.	\checkmark
• Raising individual income levels and improvement in the level of financial disclosure in the Libyan economy.	\checkmark
• Contribute to a reduction in the financing costs of ownership and debt.	\checkmark
• Motivation of companies' management to improve its efficiency and cause it completion between companies.	\checkmark
Absence of regulation that supports the operation of stock markets	\checkmark
An unclear political and economic environment policy in Libya.	\checkmark
• Libyan banking system is not sufficiently well developed.	\checkmark
Low levels of financial disclosure and transparency	\checkmark
• Corruption is another critical problem and may occur risk to foreign investment and business environment	\checkmark
• Weakness of public awareness of the role of investment and the role of the stock market in the economy.	\checkmark

Notice a: Details totalled more than N = 14 and 100 per cent due to duplicate response.

Source: Analysis of the data obtained from semi-structured interviews.

K.5 Template Analysis

This part encompasses the template analysis technique of data gathered from this study, as outlined in the qualitative analysis were mentioned in chapter methodology and presented in presentation chapters.

Objective One: To assess the success of the economic reform programme in Libya, specifically the deregulation, corporatisation, privatisation and liberalisation that has led to the inception of the Libyan stock exchange.

Questionnaire	Research Issues	Semi-structured Interview	Combined Analysis of Questionnaire and Interview results		
(Question No)	Issue Explored	Objective No	General Conclusions		
B.1			Removal of controls imposed by government and policies that		
B.2			cause negative rates of real interest in order to encourage the		
B.3			efficient operation of market.		
B.4	Deregulation	1	To understand the basis for competition and rationale for		
B.5	Deregulation	1	regulating institutions government bodies.		
B.6			To deregulate private sector, financial system, information		
B.7			quality, accountability, availability and transparency in order to		
B.8			reduce corruption.		
B.9					
B.10			To establish a structural reform process which provides incentive and efficiency in order to improve economic and		
B.11	Corporatisation	1	public accountability performance.		
B.12	Corporatisation	1	To distance the entermine form direct encourses and all in		
B.13			order to improve competitive neutrality.		
B.14					
B.15			To correct the failure of markets and restructure industries to		
B.16			efficiently use all resources.		
B.17			To impact policy package in order to establish a large number		
B.18			of new banks, institutions, corporations and companies.		
B.19			To develop profit maximisation initiative in order to make a		
B.20	Privatisation	1	privatised industry more competitive and efficient.		
B.21			The sine estimate sector increases are seen in the		
B.22			development process.		
B.23					
B.24			To reduce price costs, employment, demand for factors and increase in productivity.		
B.25			Reform of government includes: fiscal and tax reform, pension,		
B.26			health and welfare system in order to improve social security		
B.27	T 11 11 71	1	system and reacted p social infrastructure.		
B.28	Liberalisation	1	To liberalise trade finance, prices, insurance, foreign direct investment, economic fiscal and monetary policy.		
B.29			To reduce transaction costs.		

Source: Developed for this research Chapter Seven: Subsection 7.4.1

Objective Two: To identify the benefit of a stock market to the Libyan economy. This will involve examining the relationship between macro-economic reform variables and stock market performance variables.

Image: Interest Rate Objective No General Conclusions C.1	Questionnaire	Research Issues	Semi-structured Interview	Combined Analysis of Questionnaire and Interview results		
C.1 C.2 C.3 Interest Rate 2 To reform existing framework C.4 Interest Rate 2 To allow lower interest rates and renewed macro policy flexibility C.5 C.5 More power for central banks To fully liberalise interest rates C.6 Real exchange rate arbs influenced by nominal devaluations and restrictive demand policies To reduce exchange rate instability C.7 To reduce transaction costs To reduce transaction costs To reduce public expenditure and government balance C.10 Inflation Rate 2 Is percentage increase in the price of goods and services in an economy usually measured by the consumer price index and the product price index C.17 Inflation Rate 2 To reduce tub general price level C.18 C.19 To reduce tub general price level To reduce public debt C.21 Per-capita Income 2 To reduce public debt To educe public debt C.23 Per-capita Income 2 To advess current income regime To develop real individual income C.24 Per-capita Income 2 To advess current income regime To educe tub advess regending C.25 C.26 Per-capita Income	(Question No)	Issue Explored	Objective No	General Conclusions		
$ \begin{array}{ c c c c } \hline C.2 \\ \hline C.3 \\ \hline C.4 \\ \hline C.5 \\ \hline \hline C.5 \\ \hline \hline C.5 \\ \hline \hline C.6 \\ \hline C.6 \\ \hline C.7 \\ \hline C.7 \\ \hline C.7 \\ \hline C.7 \\ \hline C.8 \\ \hline C.7 \\ \hline C.8 \\ \hline C.7 \\ \hline C.8 \\ \hline C.7 \\ \hline C.9 \\ \hline C.10 \\ \hline \hline C.10 \\ \hline \hline \hline C.10 \\ \hline \hline C.11 \\ \hline \hline C.11 \\ \hline C.12 \\ \hline C.12 \\ \hline C.13 \\ \hline C.13 \\ \hline C.14 \\ \hline Inflation Rate \\ \hline C.15 \\ \hline C.16 \\ \hline \hline C.17 \\ \hline C.18 \\ \hline C.17 \\ \hline C.18 \\ \hline C.19 \\ \hline C.12 \\ \hline C.19 \\ \hline C.19 \\ \hline C.19 \\ \hline C.19 \\ \hline C.20 \\ \hline C.10 \\ \hline C.11 \\ \hline D.15 \\ \hline C.18 \\ \hline C.19 \\ \hline C.21 \\ \hline C.19 \\ \hline C.22 \\ \hline C.23 \\ \hline C.23 \\ \hline C.23 \\ \hline C.24 \\ \hline C.25 \\ \hline C.25 \\ \hline Rate A Budget Deficit \\ \hline C.2 \\ \hline C.22 \\ \hline C.23 \\ \hline C.23 \\ \hline C.23 \\ \hline C.24 \\ \hline C.25 \\ \hline C.25 \\ \hline C.25 \\ \hline C.26 \\ \hline C.27 \\ \hline C.27 \\ \hline C.27 \\ \hline C.23 \\ \hline C.23 \\ \hline C.24 \\ \hline C.25 \\ \hline C.25 \\ \hline C.25 \\ \hline C.26 \\ \hline C.27 \\ \hline C.28 \\ \hline C.27 \\ \hline C.28 \\ \hline C.27 \\ \hline C.27 \\ \hline C.27 \\ \hline C.28 \\ \hline C.27 \\ \hline C.27 \\ \hline C.28 \\ \hline C.27 \\ \hline C.27 \\ \hline C.28 \\ \hline C.28 \\ \hline C.29 \\ \hline C.29 \\ \hline D.10 $	C.1			To reform existing framework		
C.3 C.4Interest Rate2To allow lower interest rates and renewed macro policy flexibilityC.5To allow lower interest rates and renewed macro policy flexibilityC.5More power for central banks 	C.2			The key area of monetary policy		
C.4 Instantify C.5 More power for central banks C.6 Fully liberalise interest rates C.6 Real exchange rate can be influenced by nominal devaluations and restrictive demand policies C.7 C.9 C.10 To reduce exchange rate instability C.9 To reduce exchange rate instability C.10 To reduce public expenditure and government balance S.11 Inflation Rate 2 C.12 Inflation Rate 2 C.14 Inflation Rate 2 C.15 Inflation Rate 2 C.16 To reduce public expenditure and government balance S. percentage increase in the price of goods and services in an economy usually measured by the consumer price index and the product price index C.16 To reduce the general price level C.17 To reduce badget C.20 Per-capita Income 2 C.21 Per capita Income 2 C.23 Per capita income, additional economic benefit from extra income C.24 Per-capita Income 2 C.25 Per-capita Income 2 C.26 Per-capita In	C.3	Interest Rate	2	To allow lower interest rates and renewed macro policy		
C.5 More power for central banks C.6 To fully liberalise interest rates C.7 Real exchange rate can be influenced by nominal devaluations and restrictive demand policies C.8 To reduce exchange rate can be influenced by nominal devaluations and restrictive demand policies C.9 To reduce exchange rate can be influenced by nominal devaluations and restrictive demand policies C.10 To reduce exchange rate instability C.11 To reduce transaction costs C.12 Inflation Rate 2 C.14 Inflation Rate 2 C.15 Inflation Rate 2 C.16 To reduce the general price loyed Is percentage increase in the price of goods and services in an economy usually measured by the consumer price index and the product price index C.16 To reduce the general price level To reduce public debt C.20 Budget Deficit 2 To reduce public debt C.23 To shift emphasis to priority areas To address current income regime C.24 Per capita income To address current income regime C.26 Per-capita income 2 To address current income regime C.27 To address current income regime 10 addre	C.4			flexibility		
C.6To fully liberalise interest ratesC.6Real exchange rateReal exchange rate can be influenced by nominal devaluations and restrictive demand policiesC.7C.8To reduce exchange rate instabilityC.9To reduce exchange rate instabilityTo reduce public expenditure and government balanceC.10Inflation Rate2Is percentage change in the overall level of prices usually per annumC.14Inflation Rate2Is percentage increase in the price of goods and services in an economy usually measured by the consumer price index and the product price indexC.16To reduce public debtTo reduce public debtC.19C.20To reduce public debtC.21Budget Deficit2C.23Per-capita Income2C.24Per-capita Income2C.25Real GDP Growth Rate2C.31Real GDP Growth Rate2C.33Real GDP Growth Rate2C.34C.34Conomic services and increase competitionC.35C.34Conomic Activation of the payment systemC.35C.34Conomic Activation of the payment system of the rest rate of the rate of t	C.5			More power for central banks		
C.6 Real exchange rate can be initianced by nominal devaluations and restrictive demand policies C.7 and restrictive demand policies C.9 To reduce transaction costs C.10 To reduce transaction costs C.11 To reduce public expenditure and government balance S.11 Inflation Rate 2 C.13 Is percentage increase in the price of goods and services in an economy usually measured by the consumer price index and the product price index C.16 To reduce public expenditure and government balance C.14 Is percentage increase in the price of goods and services in an economy usually measured by the consumer price index and the product price index C.16 To reduce public works' spending C.17 To reduce public works' spending C.20 To reduce public works' spending C.21 To shift emphasis to priority areas C.22 Per capita income 2 C.23 To address current income regime To complete modernisation of the payment system C.24 Per capita income To increase or decrease economic output C.23 Is one of the ways of measuring the size of economy and market' business, scale economic and increase competition C.23 Real GDP Growth Rate				To fully liberalise interest rates		
C.3 C.3 C.9 Exchange rate 2 To reduce exchange rate instability C.10 To reduce transaction costs To reduce public expenditure and government balance C.11 Inflation Rate 2 Is percentage change in the overall level of prices usually per annum C.14 Inflation Rate 2 Is percentage increase in the price of goods and services in an economy usually measured by the consumer price index and the product price index C.16 C.17 To reduce the general price level C.18 To reduce public debt To reduce public debt C.20 Budget Deficit 2 To reduce public debt C.21 C.23 To shift emphasis to priority areas C.24 C.25 Per-capita Income 2 To address current income regime To develop real individual income C.23 C.24 Per capita income, additional economic benefit from extra income To complete modernisation of the payment system C.23 Es and GDP Growth Rate 2 So ne of the ways of measuring the size of economy and market value of all goods and services produce by labour and property located in a country at specific year's prices in period of time C.33 C.34 C.35 To link markets' business, scale eco	C.6			Real exchange rate can be influenced by nominal devaluations and restrictive demand policies		
C.3 C.9Exchange rate2To reduce exchange rate instabilityC.9C.10To reduce transaction costsC.11To reduce public expenditure and government balanceC.12Is percentage change in the overall level of prices usually per annumC.13Inflation Rate2C.16Is percentage increase in the price of goods and services in an economy usually measured by the consumer price index and the product price indexC.16To reduce the general price levelC.17To reduce the general price levelC.18To reduce public debtC.19To reduce public debtC.20Budget Deficit2C.21Deficit2C.23Per-capita Income2C.24Per capita income, additional economic benefit from extra incomeC.26Per-capita Income2C.27To complete modernisation of the payment systemC.23Real GDP Growth Rate2C.31C.33Is one of the ways of measuring the size of economy and property located in a country at specific year's prices in period of timeC.33C.34To link market 's business, scale economics and increase competitionC.34C.35To link market's business, scale economics and increase competition	C.7					
C.9 To reduce transaction costs C.10 To reduce public expenditure and government balance C.12 Inflation Rate 2 C.13 Is percentage change in the overall level of prices usually per annum C.14 Inflation Rate 2 C.16 Is percentage increase in the price of goods and services in an economy usually measured by the consumer price index and the product price index C.16 To reduce the general price level C.17 To reduce ubget C.19 To reduce public works' spending C.20 To reduce public works' spending C.21 To shift emphasis to priority areas C.22 To address current income regime C.23 To address current income regime C.26 Per-capita income To complete modernisation of the payment system C.26 Per-capita income Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of time C.23 Is one of the ways of measuring the size of economies and increase competition C.34 To ink markets' business, scale economies and increase competition C.35 To inerease or decrease anonic output </td <td>C.0</td> <td>Exchange rate</td> <td>2</td> <td>To reduce exchange rate instability</td>	C.0	Exchange rate	2	To reduce exchange rate instability		
C.10To reduce public expenditure and government balanceC.11Inflation Rate2Is percentage change in the overall level of prices usually per annumC.13Inflation Rate2Is percentage change in the overall level of prices usually per annumC.14Inflation Rate2Is percentage increase in the price of goods and services in an economy usually measured by the consumer price index and the product price indexC.16C.17To reduce the general price levelC.18To reduce the general price levelC.19Perc20To reduce public debtC.20To reduce public debtC.21To shift emphasis to priority areasC.22Per-capita Income2C.23Per-capita Income2C.26Per-capita Income2C.27C.28To complete modernisation of the payment systemC.29Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.31C.34To link markets' business, scale economies and increase competitionC.34To link markets' business, scale economies and increase competition	C 10			To reduce transaction costs		
C.12 Inflation Rate 2 Is percentage change in the overall level of prices usually per annum C.13 Inflation Rate 2 Is percentage increase in the price of goods and services in an economy usually measured by the consumer price index and the product price index C.16 Inflation Rate 2 Is percentage increase in the price of goods and services in an economy usually measured by the consumer price index and the product price index C.16 To reduce the general price level To reduce budget C.19 To reduce public debt To reduce public debt C.20 Budget Deficit 2 To reduce public works' spending C.21 C.23 To shift emphasis to priority areas C.24 Per-capita income 2 C.25 Per-capita Income 2 C.26 Per-capita Income 2 C.27 To develop real individual income To complete modernisation of the payment system C.29 Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of time C.31 C.34 2 Is one of the ways of measuring the size of economis and increase competition C.33 C.34 2<	C 11			To reduce public expenditure and government balance		
C.13Inflation Rate2annumC.14Inflation Rate2Is percentage increase in the price of goods and services in an economy usually measured by the consumer price index and the product price indexC.16C.17To reduce the general price levelC.18To reduce the general price levelC.19C.20C.20Budget Deficit2C.21Budget Deficit2C.22To reduce public debtC.23To shift emphasis to priority areasC.24Per capita incomeC.25Per-capita Income2C.26Per-capita Income2C.27To develop real individual incomeC.28To complete modernisation of the payment systemC.30So no of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.31C.342C.35To increase or decrease economic outputTo ink market's business, scale economics and increase competitionTo inprove government services and improve infrastructure rules	C.12			Is percentage change in the overall level of prices usually per		
C.14 C.15 C.16Inflation Rate2Is percentage increase in the price of goods and services in an economy usually measured by the consumer price index and the product price indexC.16To reduce the general price levelTo reduce the general price levelC.17To reduce the general price levelTo reduce budgetC.19To reduce public debtTo reduce public debtC.20To reduce public works' spendingTo shift emphasis to priority areasC.21Per-capita Income2Per capita income, additional economic benefit from extra incomeC.25Per-capita Income2To address current income regimeC.26Per-capita Income2To complete modernisation of the payment systemC.29Real GDP Growth Rate2Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.33C.34To increase or decrease economic output To link markets' business, scale economics and increase competitionC.35To improve government services and improve infrastructure rules	C.13			annum		
C.15Inflation Rate2Inflation Rate2C.16C.17C.17C.18To reduce the general price levelC.19To reduce budgetC.20To reduce public debtC.21To reduce public debtC.22To shift emphasis to priority areasC.23Per capita income, additional economic benefit from extraC.25To address current income regimeC.26Per-capita Income2To complete modernisation of the payment systemC.29Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.31C.33C.34C.35C.34C.35C.34C.35	C.14			Is percentage increase in the price of goods and services in an		
C.16product price indexC.17To reduce the general price levelC.18To reduce the general price levelC.19To reduce budgetC.20To reduce public debtC.21To reduce public works' spendingC.22To shift emphasis to priority areasC.23Per-capita IncomeC.24Per-capita IncomeC.25Per-capita IncomeC.26Per-capita IncomeC.27To address current income regimeC.28To complete modernisation of the payment systemC.30Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.33Real GDP Growth Rate2C.34To increase or decrease economic outputTo ink markets' business, scale economies and increase competitionTo improve government services and improve infrastructure rules	C.15	Inflation Rate	2	economy usually measured by the consumer price index and the		
C.17To reduce the general price levelC.18	C.16			product price index		
C.18To reduce budgetC.20Budget Deficit2To reduce public debtC.21To reduce public works' spendingTo shift emphasis to priority areasC.23C.24Per capita incomePer capita income, additional economic benefit from extra incomeC.24Per-capita Income2To address current income regimeC.26Per-capita Income2To develop real individual incomeC.27To complete modernisation of the payment systemTo complete modernisation of the payment systemC.29Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.31C.32Real GDP Growth Rate2C.34To increase or decrease economic output To link market's business, scale economies and increase competition To improve government services and improve infrastructure miles	C.17			To reduce the general price level		
C.19To reduce budgetC.20Budget Deficit2To reduce public debtC.21To reduce public works' spendingTo shift emphasis to priority areasC.23Per capita incomePer capita income, additional economic benefit from extra incomeC.24Per capita incomePer capita income regimeC.26Per-capita Income2To address current income regimeC.27To complete modernisation of the payment systemC.29Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.31Real GDP Growth Rate2C.35Per capita income terease economic output To link markets' business, scale economies and increase competitionC.35To improve government services and improve infrastructure miles	C.18			To reduce budget		
C.20 C.21Budget Deficit2To reduce public debtC.21C.22To reduce public works' spending To shift emphasis to priority areasC.23Per capita incomePer capita income, additional economic benefit from extra incomeC.26Per-capita Income2Per capita income regime To develop real individual income To complete modernisation of the payment systemC.29Real GDP Growth Rate2Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.35C.342To link markets' business, scale economics and increase competitionC.35C.34To link markets' business, scale economics and increase competition	C.19					
C.21 C.22Budget Deficit2To reduce public works' spending To shift emphasis to priority areasC.23C.24 C.25Per capita incomePer capita income, additional economic benefit from extra incomeC.26 C.27Per-capita Income2To address current income regime To develop real individual income To complete modernisation of the payment systemC.29 C.29Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.31 C.32 C.342Is one of the markets' business, scale economics and increase competition To link markets' business, scale economies and increase competition To improve government services and improve infrastructure rules	C.20	Pudgat Dafiait	2	To reduce public debt		
C.22 C.23To shift emphasis to priority areasC.24 C.25Per-capita Income2C.26 C.27Per-capita Income2C.27To address current income regime To develop real individual incomeC.29 C.30Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.31 C.32 C.33Real GDP Growth Rate2C.34 C.35Real GDP Growth Rate2C.35 C.34Real GDP Growth Rate2	C.21	Budget Denett	2	To reduce public works' spending		
C.23To simil emphasis to priority areasC.24Per capita incomeC.25Per-capita IncomeC.26Per-capita IncomeC.27To address current income regimeC.28To develop real individual incomeC.29To complete modernisation of the payment systemC.30Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.312C.32Real GDP Growth RateC.342C.35To link markets' business, scale economics and increase competitionTo improve government services and improve infrastructure miles	C.22			To shift amphasis to priority areas		
C.24Per capita incomePer capita income, additional economic benefit from extra incomeC.26Per-capita Income2To address current income regime To develop real individual incomeC.27To develop real individual incomeTo complete modernisation of the payment systemC.28Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.312To increase or decrease economic output To link markets' business, scale economies and increase competitionC.34C.35To improve government services and improve infrastructure miles	C.23			To shift emphasis to priority areas		
C.25Per-capita Income2To address current income regimeC.27To address current income regimeTo develop real individual incomeC.28To complete modernisation of the payment systemC.29Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.31Real GDP Growth Rate2C.33Per-capita Income2C.34To increase or decrease economic output To link markets' business, scale economies and increase competitionC.35To improve government services and improve infrastructure miles	C.24			Per capita income, additional economic benefit from extra		
C.26Per-capita Income2To address current income regimeC.27To develop real individual incomeC.28To complete modernisation of the payment systemC.29Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.31C.32Real GDP Growth Rate2C.33C.342C.35To link markets' business, scale economics and increase competitionC.36To improve government services and improve infrastructure miles	C.25			The later of the l		
C.27 To develop real individual income C.28 To complete modernisation of the payment system C.29 Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of time C.30 Real GDP Growth Rate 2 C.34 To link markets' business, scale economics and increase competition C.35 To improve government services and improve infrastructure miles	C.26	Per-capita Income	2	To address current income regime		
C.28 To complete modernisation of the payment system C.29 Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of time C.31 Real GDP Growth Rate 2 C.33 To increase or decrease economic output To link markets' business, scale economies and increase competition To improve government services and improve infrastructure miles	C.27			To develop real individual income		
C.29Is one of the ways of measuring the size of economy and market value of all goods and services produced by labour and property located in a country at specific year's prices in period of timeC.31C.32Real GDP Growth Rate2To increase or decrease economic output To link markets' business, scale economies and increase competitionC.34C.35To improve government services and improve infrastructure miles	C.28			To complete modernisation of the payment system		
C.30 Indice value of all goods and services produced by labour and property located in a country at specific year's prices in period of time C.32 Real GDP Growth Rate 2 C.33 C.34 C.35 C.36	C.29			Is one of the ways of measuring the size of economy and		
C.31 of time C.32 Real GDP Growth Rate C.33 C.34 C.35 C.36	C.30			property located in a country at specific year's prices in period		
C.32 Real GDP Growth Rate 2 To increase or decrease economic output C.33 C.34 To link markets' business, scale economies and increase competition C.35 To improve government services and improve infrastructure rules	C.31			of time		
C.35 To link markets' business, scale economies and increase competition C.35 To improve government services and improve infrastructure miles	C.32	Real GDP Growth	2	To increase or decrease economic output		
C.34 competition C.35 To improve government services and improve infrastructure mles	C.33	Katt		To link markets' business, scale economies and increase		
C.35 To improve government services and improve infrastructure rules	C.34			competition		
1 AD	C.33			To improve government services and improve infrastructure rules		

Source: Developed for this research Chapter Seven: Subsection 7.4.2

Questionnaire	Research Issues	Semi-structured Interview	Combined Analysis of Questionnaire and Interview results	
(Question No)	Issue Explored	Objective No	General Conclusions	
D.1			To indicate a depth of bank intermediation and size of bank sector	
D.2	Deulius conte Detia	3	To more a deput of bank intermediation and size of bank sector	
D.3	Banking assets Ratio		To provide levels of financial intermediation in country	
D.4			To measure maturity intermediation or asset transformation	
D.5			To indicate high level of banking intermediation by using high share of domestic	
D.6		3	credit in GDP or high loans to GDP	
D.7	Domestic Credit Ratio		To accept deposits to the private sector divided by GDP using the ratio of loans to GDP	
D.8			To terminate directed credit allocations	
			To improve the environment for banking activity	
D.9			To understand the main features of national competition policy in Libya	
			To measure stock market activity	
D.10	Stock Market		To measure stock market size	
D.11	Capitalisation Ratio	3	To measure the value of listed domestic shares on domestic exchange divided by	
D 12			GDP	
D.12			To increase liquidity	

Objective Three: To assess the performance of market-based economies, with particular reference to the emerging economy of Libya, and to evaluate current and best practice in financial deregulation.

Source: Developed for this research Chapter Seven: Subsection 7.4.3

Dependent variables: As mentioned in Chapter Six: Figure 6.3, each economic reform programme, macroeconomic reform and financial maturity variables will be tested as an independent variables with stock market performance variables as dependent variables, where significant relationships between the explanatory variables and the stock market performance variables exist.

Questionnaire	Research Issues	Semi-structured Interview	Combined Analysis of Questionnaire and Interview results		
(Question No)	Issue Explored	Dependent Verbalise	General Conclusions		
E.1			To provide a more comprehensive picture of stock market		
E.2			Use the total value-traded as a percentage of GDP		
E.3	Market Size	1	To increase the market conitalisation ratio		
E.4					
E.5			Is not a very good predictor of economic performance		
E.6			To provide a high turnover ratio indicator in order to reduce		
E.7			transaction costs		
E.8	Market Liquidity	1	To reduce price-risk exposure		
EO			To increase turnover ratio		
E.9			Total value traded ratio		
B.10					
B.11			To increase value-traded ratio which reduce higher transaction costs		
B.12	Maulaat Aatiaitaa	1	To increase value-traded ratio which reduce higher transaction costs		
B.13	Market Activity	1	As measured by total value, traded ratio is a good predictor of		
B.14			economic growth rate		
E.15					
			To reflect the degree of competition in the market		
E.16	Market Concentration	1	Is a criterion that can be used to rank or order various distributions of firms' shares of the total production such as: alternatively; total capacity or total reserves in stock market		
E.17	Concentration		A market concentration measure should be decreasing or at least non- increasing with the degree of symmetry between the firms' share. Also, should be decreasing the number of firms in the stock market		

Source: Developed for this research Chapter Seven: Subsection 7.4.4

It is important to mention that, the researcher tended not to use computer-assisted qualitative data analysis software (CAQDAS)⁷⁰ to analyse the qualitative data. That was due to the following reasons: firstly, the number of interviews was 14, so it was easy to work on them manually; secondly, the researcher needed to be part of analysis process, involved in it, observe data, think of the data, read it again and again. Finally, there are a number of (CAQDAS) software which may vary in relation to the type of facilities that they offer and potentially in their usefulness for different analytic situation, however, only one piece of software was readily accessible or available for the researcher. This type of software requires experience, ability and time to experiment with different software, requiring experiences with a range of software so the most appropriate choice can be made. Seidle (1991) argues, in Saunders et al. (2007), that in a case where CAQDAS software is used by inexperienced researcher, this will move the attention into analytical practices and conceptual problems rather than, into depth analysis, this is referred to the fact that analyst become more concerned with analysis based on quantification than exploration of meaning. Nigel King (personal communication by email. 29 July 2009) suggested to me that⁷¹:

"No you don't need any software. I find a useful method if to write the potential themes on post-it notes and you can then move them around until you have sorted put the final structure of your template. Cheers Nigel"

As mentioned earlier, in Chapter Six, Subsection 6.10.2, the analysis of interviews' qualitative data was organised. The first step includes preparing a clear text transcription from the replies of the interviewee. The second phase is the analysis of the text transcriptions by which the key words, phrases and paragraphs relevant to the enquiry are marked and highlighted in different colours. Moreover, paragraphs to be cited from the transcription are underlined and marked in a specific colour. Finally, relevant paragraphs are extracted and organised in tables under different headings. The main objective of this phase is to summarise the transcriptions, to identify the relevant themes, to translate and classify interviewee opinions, views and perceptions into subjects. The third phase consists of creating coding categories in to which subjects are grouped. The central objective of this phase is to group subjects under a particular title or a heading. The final step of the process is to display and present the findings of the analysis using many techniques, such as quotation, citation, diagrams and tables.

⁷⁰ See more details on analysing qualitative data software (CAQDAS) polished article by Tanuka Bhowmick (2006) Buildingan Exploratory Visual Analysis Tool for Qualitative Researchers [online] available at:<http://www.geovista.psu.edu/publications/2006/Bhowmick_AutoCarto_QualRes_06.pdf> [Accessed 3 September 2009].

⁷¹ Nigel King, Professor in Applied Psychology; University of Huddersfield. Nigel has developed a website devoted to the template analysis approach available at: http://www.hud.ac.uk/hhs/research/template/index.htm.

Appendix L: Chapter Eight: Statistics Tests; Appropriate Use

L.1 RESULTS FROM MODELLING ECONOMIC REFORM PROGRAMME

L.1.1 The Assumption of Collinearity Diagnostics

Table L 1.1 produce *SPSS* output, of labelled *Collinearity Diagnostics*, which included eigenvalues of the scaled, cross-products matrix, the condition index and the variance proportions for each predictor. The *eigenvalues* values provided some idea as to how accurate this regression model is. If the eigenvalues are fairly similar then the measured model is likely to be unchanged by small changes in the other measured variables (Field, 2005). The *condition index* is another measure of explaining these eigenvalues and represents the square root of the ratio of largest eigenvalue, so the condition index will always be 1 to the eigenvalue of interest.

	Tuble Entri Commentity Diagnostics							
Madal Dimension		Figonyoluo	Condition	Variance Portions				
Widdei	Widder Dimension	Eigenvalue	Index	(Constant)	ERPD	ERPC	ERPP	ERPL
1	1	4.941	1.000	0.00	0.00	0.00	0.00	0.00
	2	0.029	13.164	0.00	0.01	0.01	0.14	0.89
	3	0.022	15.015	0.02	0.03	0.02	0.80	0.05
	4	0.006	28.562	0.01	0.69	0.39	0.00	0.01
	5	0.003	43.996	0.97	0.27	0.58	0.05	0.05

Table L.1.1 Collinearity Diagnostics ^(a)

Notice b:a Dependent Variable: SMP.

Source: Analysis of SPSS output from questionnaire survey data.

For these data, the final dimension has a condition index of 43.996, which is large compared to the other dimensions. The variance proportions considered between 0 and 1 and for each predictor should be distributed across different dimensions (or eigenvalues). For this model, each predictor has most of its variance loading onto a separate dimension: ERPL had 89 *per cent* of variance on dimension 2, ERPP has 80 *per cent* of variance on dimension 3 and ERPD has 69 *per cent* of variance on dimension 4. The result of this data analysis of collinearity diagnostics found no multicollinearity between predictor variables in this regression analysis. This result is also considered by Spearman's Rho correlation (see Chapter Eight, Table 8.2).

L.1.2 Checking Assumptions

As a final stage in the data analysis, the two assumptions of *homoscedasticity* and *linearity* are checked⁷². Figure L 1.1 displays a plot of the regression standardised residual against the regression standardised predicted values. The graph should represent a random array of dots evenly dispersed around zero. If this graph funnels out, then it is likely that there is homoscedasticity in the data analysis, but if there is any sort of curve in this graph, then it is probable data have broken the assumption of linearity.

⁷² Homscedasticity means that variance of errors is the same across all levels of the independent variables X. When the variance of errors differs at separate values of the independent variable X, heteroscedasticity is indicated. When heteroscedasticity is marked, it can lead to serious distortion of findings and seriously weaken the analysis, thus increasing possibility of a Type I error. Heteroscedasticity can take many forms, such as a bow-tie or fan shape. Linearity, as presented by standard multiple regression, can only accurately estimate relationship between dependent and independent variables if the relationships are linear in nature. Due to there being many instances in social sciences in which nonlinear relationships occur, it is essential to examine analyses of such relationship for nonlinearity. If the actual relationship between independent variables X and the dependent variable Y is not linear, as being assumed, results of regression analysis will underestimate the true relationship. This underestimation carries two risks: increased chance of a Type II error for that independent variable X and, in the case of multiple regression, an increased risk of Type I errors (overestimation) for other independent variables X that share variance with that independent variable X (Hair et al., 2005).

Figure L 1.1 Scatterplot of Standardised Residuals Against Standardised Predicted Values/Multiple Regression



Source: Analysis of SPSS output from questionnaire survey data.

Figure L 1.1 illustrates the scatter test of how points are randomly and evenly dispersed throughout the plot. This pattern is indicative of a situation in which assumptions of linearity and homoscedasticity have been met and considered. Based on *normality of* test *residuals*, the histogram and normal probability plot of the study's data outlined in Figure L 1.2 and Figure L 1.3, findings were that the histogram should appear like a normal distribution (a bell-shaped curve)⁷³. The output of *SPSS* draws a curve on the histogram to display the shape of this distribution. The normal distribution of histogram in Figure L 1.2 presents distribution as almost normal. This is in contrast to the situation in which a histogram exhibits a non-normal distribution that is extremely skewed (unsymmetrical).





Source: Analysis of SPSS output from questionnaire survey data.

The normal probability plot further explains deviations from normality distribution. The straight line in this plot represents a normal distribution and the points represent the observation residuals. Therefore, in a perfectly distributed data set, most points will lie off the line. For the normal probability plot demonstrated in Figure L 1.3, it can be observed that, in general, observed residuals follow the line. Whilst there is some deviation away from that line, these are not far from it and so the overall trend for these residuals is that they follow the line representing a normal distribution.





Source: Analysis of SPSS output from questionnaire survey data.

⁷³ Normal distribution: it is assumed that residuals in the model are random, normally, distributed variables with a mean of 0. Therefore, differences between the model and observed data are most frequently zero or very close to zero. Differences much greater than zero happen only infrequently (Field, 2005).

L.2 RESULTS FROM MODELLING MACRO-ECONOMIC REFORM

L.2.1 The Assumption of Collinearity Diagnostics

SPSS output produced Table L 2.1, labelled *Collinearity Diagnostics*, which included eigenvalues of the scaled, cross-products matrix, the condition index and the variance proportions for each predictor.

	Dimension	Eigenvalue	Condition	Variance Portions						
Model			Index	(Constant)	MERITR	MERER	MERIFR	MERBD	MERPCI	MERGDP
1	1	6.897	1.000	0.00	0.00	0.00	0.00	0.00	0.00	0.00
	2	0.029	15.475	0.00	0.93	0.01	0.01	0.01	0.04	0.04
	3	0.022	17.759	0.00	0.01	0.13	0.01	0.12	0.04	0.73
	4	0.020	18.685	0.00	0.01	0.03	0.02	0.05	0.80	0.00
	5	0.017	20.118	0.00	0.00	0.07	0.31	0.52	0.00	0.03
	6	0.012	23.850	0.00	0.00	0.57	0.54	0.08	0.05	0.00
	7	0.003	45.807	1.00	0.05	0.19	0.12	0.23	0.07	0.19

 Table L 2.1 Collinearity Diagnostics ^(a)

Notice c: a Dependent Variable: SMP

Source: Analysis of SPSS output from questionnaire survey data.

For these data, the final dimension has a condition index of 45.807, which is large compared to the other dimensions. Variance proportions are considered between 0 and 1, and each predictor should be distributed across a range of dimensions or eigenvalues. For this model, each predictor has most of its variance loading onto a different dimension: MERITR has 93 *per cent* of variance on dimension 2, MERPCI has 80 *per cent* of variance on dimension 4 and MERGDP has 73 *per cent* of variance on dimension 3. The result of this data analysis of collinearity diagnostics found no multicollinearity between predictor variables in this regression analysis. This result is also considered by Spearman's Rho correlation (see Chapter Eight, Table 8.11).

L.2.2 Checking Assumptions

Figure L 2.1 displays a plot of the regression standardised residual against regression standardised predicted values. The graph should resemble a random array of dots evenly dispersed around zero. If this graph funnels out, then the likelihood is there is homoscedasticity in data analysis. However, if there is any sort of curve in this graph then the likelihood is that data have broken the assumption of linearity.

Figure L 2.1 Scatterplot of Standardised Residuals against Standardised Predicted Values/Multiple Regression



Based on the normality test of residuals, the histogram and normal probability plot of the study's data outlined in Figure L 2.2 and Figure L 2.3, findings were that the histogram should appear as a normal distribution (a bell-shaped curve). Output of *SPSS* draws a curve on the histogram to display this distribution's shape. The normal distribution of histogram in Figure L 2.2 presents the distribution as being almost normal. This is in contrast to the situation in which a histogram exhibits a non-normal distribution that is extremely skewed (unsymmetrical).



Source: Analysis of SPSS output from questionnaire survey data.

The normal probability plot additionally explains deviations from normal distribution. The straight line in this plot represents a normal distribution and the points represent observation residuals. Therefore, in a perfectly distributed data set, all points will lie off the line. For the normal probability plot, demonstrated in Figure L 2.3, it can be noted that, in general, the observed residuals follow the line. While there is some deviation away from that line, these are not distant from it, and so the overall trend for these residuals is that they follow the line representing a normal distribution pattern.





Source: Analysis of SPSS output from questionnaire survey data.

L.3 RESULTS FROM MODELLING FINANCIAL MATURITY

L.3.1 The Assumption of Collinearity Diagnostics

SPSS output produced Table L 3.1, of labelled *Collinearity Diagnostics*, which included eigenvalues of the scaled, cross-products matrix, the condition index and variance proportions for each predictor.

Madal	Dimension	Eigenvalue	Condition Index	Variance Portions				
Model				(Constant)	FMBA	FMDC	FMSMC	
1	1	3.903	1.000	0.00	0.00	0.00	0.00	
	2	0.051	8.733	0.00	0.07	0.78	0.15	
	3	0.035	10.623	0.02	0.32	0.01	0.76	
	4	0.012	18.254	0.98	0.60	0.20	0.09	

 Table L 3.1 Collinearity Diagnostics ^(a)

Notice d: a Dependent Variable: SMP

Source: Analysis of SPSS output from questionnaire survey data.

For these data the variance proportions are considered between 0 and 1, and each predictor should be distributed across different dimensions or eigenvalues. For this model, each predictor has most of its variance loading onto a separate dimension: FMDC has 78 *per cent* of variance on dimension 2, FMSMC has 76 *per cent* of variance on dimension 3 and FMBA has 60 *per cent* of variance on dimension 4. The result of this data analysis of collinearity diagnostics found no multicollinearity between predictor variables in this regression analysis. This result was also considered in line with Spearman's Rho correlation (see Chapter Eight, Table 8.22).

L.3.2 Checking Assumptions

Figure L 3.1 displays a plot of the regression standardised residual against regression standardised predicted values. The graph should resemble a random array of dots evenly dispersed around zero. If this graph funnels out, then it is highly likely there is homoscedasticity in the data analysis. If there is any sort of curve in the graph, then it is probable data have broken the assumption of linearity.

Figure L 3.1 Scatterplot of Standardised Residuals against Standardised Predicted Values/Multiple Regression



Source: Analysis of *SPSS* output from questionnaire survey data.

Based on the normality test of residuals, the histogram and normal probability plot of the study's data, as outlined in Figure L 3.2 and Figure L 3.3, findings were that the histogram should resemble a normal distribution (a bellshaped curve). The output of *SPSS* draws a curve on the histogram to display this distribution's shape. The normal distribution of the histogram in Figure L 3.2 presents the distribution as almost normal. This is in contrast with the situation in which a histogram exhibits a non-normal distribution that is extremely skewed (unsymmetrical).



Source: Analysis of SPSS output from questionnaire survey data.

The normal probability plot additionally explains deviations from normality distribution. The straight line in this plot represents a normal distribution and the points represent observation residuals. Therefore, in a perfectly distributed data set, all points will lie off the line. For the normal probability plot demonstrated in Figure L 3.3, it can be observed that, in general, the observed residuals follow the line. Whilst there is some deviation away from that line, these are not far from it and so the overall trend is that they follow the line representing a normal distribution.







L.4 Regression model Background

Linear regression analysis originated from statistics and has been widely used in econometrics statistics testing.

L.4.1 Simple Linear Regression Model (SLRM)

Simple regression seeks to produce an outcome variable from a single predictor variable. The following general equation the idea that can predict any data using:

Setting up the Model

$$outcome_{i} = (mod \ el_{i}) + error_{i}$$
(L.1)

The equation being predicted for a particular person is the dependent variable or the regression which will denote it by Y. The predictor variable by whatever model fits the linear model or model based on a straight line denoted by X, is also called the independent variable or the explanatory variable plus some kind of error.

The probabilistic regression model or the equation for a strength line is given by the formula as:

$$Y = \alpha + \beta X$$

Y is endogenous, X is exogenous, etc. This implies X determines Y exactly, a deterministic relationship. However if you make Y stochastic then the simple linear model is:

$$Y = \alpha + \beta X + \varepsilon \tag{L.2}$$

where \mathcal{E} is an rv with some distribution as displayed in Figure L 4.1.

Figure L 4.1 Potential \mathcal{E} , termed the error



Source: Developed for this research from research model.

Actually, \mathcal{E} is a *sequence* of r.v. which is denoted by the *error* term, one value of X. When X is fixed in the sample or non-stochastic then all the stochastic part of Y is determined by this error process, \mathcal{E} . In the previous discussion, it's possible to test how the linear relationship between two variables could be measured using simple linear correlation coefficient. Regression analysis, on the other hand, attempts to determine the functional relationship between two or more variables. However, the most basic regression model is the simpler regression model which is a bivariate linear regression. In this case one variable is being predicted by another. Thus, the relationship using correlation analysis could be tested. The probabilistic regression model can be expressed as:

$$Y_i = \alpha + \beta X_i + \varepsilon_i \qquad i = 1, ..., n \tag{L.3}$$

for cross-section data, or

$$Y_t = \alpha + \beta X_t + \mathcal{E}_t \qquad i = 1, ..., t$$
 (L.4)

for time series data

The general linear model (GLM) used in this research which is defined⁷⁴:

Y-Intercept Slopes or Coefficients Random Error

$$Y_{i} = \beta_{0} + \beta_{1}X_{i1} + \beta_{2}X_{i2} + \dots + \beta_{k} + X_{ik} + \varepsilon_{i}$$
Dependent Variable Independent (Explanatory) Variables
(L.5)

Where Y_i is the outcome variable being predicted is the dependent variable or the *regressand*. The predictor variable, denoted by X_i , is *i* th participants' score on the predictor variable and called either the independent variable or the explanatory variable, β_1 is the coefficient of the first predictor X_1 , β_2 is the coefficient of the second predictor X_2 , and β_0 is the intercept of that line. β_k is the coefficient of the *n* th predictor X_k . The portion $\alpha + \beta X_i$ represents the deterministic type of the probabilistic model (see Figure L 4.2).

Figure L 4.2 Simple Linear Regression Model



Source: Introduction to the linear Regression Model (2008).

There is a residual term \mathcal{E}_i which represents the model's random component and the observed value of Y for the *i* th participant, which plays a significant role in regression analysis. The Ordinary Least Squares (OLS) estimate of $\beta_i = \overline{Y}_i$, so the residual $\mathcal{E}_i, ..., \mathcal{E}_n$ are defined by:

$$\Rightarrow \mathcal{E}_i = Y_i - Y_i \quad \text{or}$$
$$\Rightarrow \mathcal{E}_i = Y_i - \bar{Y_i}$$

Thus, α and β are unobtainable and should be estimated by using the sample statistics a and b. Thus, in equation L.3 these could be re-expressed as:

$$\boldsymbol{Y}_{i} = \boldsymbol{a} + \boldsymbol{b}\boldsymbol{X}_{i} + \boldsymbol{\mathcal{E}}_{i} \tag{L.6}$$

$$Y_{i} = b_{0} + b_{1}X_{i1} + b_{2}X_{i2} + \dots + b_{k} + K_{ik} + \mathcal{E}_{i}$$

⁷⁴ Estimate Regression Equation can be written as:

There is no disparity between this equation and L.5. They merely use different letters to represent the coefficients. The sole difference is that this equation b_{e} instead of β_{e} and both versions are the same.

Appendix M: Chapter Nine: Finance and Economic Growth

In this appendix we present the empirical research that discusses a link between the financial system and stock market development, and economic growth. Table M.1 presents a summary of the main previous empirical evidence on finance and economic growth.

Author(s) (Year)	Sample size and period	Independent Variables	Dependent Variables	Findings
King and Levine (1993b)	80 countries 1960-1989	Bank development = total liquid liabilities/GDP or M3/GDP, ratio of claims on non-financial private sector to domestic credit, log of enrolment, government. Expenditures to GDP, inflation rate, (exports + imports)/GDP	GYP = real per capita GDP growth, GK = growth rate of per capita stock, INV=ratio of average annual investment to GDP, bank credit/ (deposit money + bank credit)	Positive relationship between growth and banking; FIs enhance efficiency of allocation, therefore affecting LR economic growth.
Levine (1998)	47 countries 1976-1993	Legal origin enforcement output (RGDP/capita), creditor rights, bank development	Bank development = (domestic credit/GDP); RGDP/capita growth	Legal systems influence banking development; legal system (as a component of the banking system) is positively associated with economic growth, capital accumulation, and productivity growth.
La Porta et al. (1997)	49 countries 1976-1993	GDP growth, log of GDP, rule of law, French origin, Scandinavian origin, German origin, anti-director rights, one share	External capital (market capitalisation/GNP), domestic firms/ population; IPOs/population, Debt/GDP	Civil law countries (French particularly) have the least developed capital markets.
Levine and Zervos (1998)	47 countries 1976-1993	Turnover = value of the trades of domestic shares as a share of market capitalisation, value traded = value of the trades of domestic shares as a share of GDP, capitalisation = value of domestic shares as a share of GDP, and volatility = measure of stock return volatility	Output growth = real per-capita GDP growth, capital stock growth = real per-capita capital stock growth, productivity growth = output growth-capital stock growth, average savings = savings as a <i>per cent</i> of GDP	Initial levels of turnover and bank credit are positive and significant in explaining subsequent growth (all measures).
Garcia and Liu (1999)	15 countries 1980-1995	Domestic credit/GDP, M3/GDP, stock market liquidity = total value traded/GDP, turnover = total value traded/market capitalisation.	Market capitalisation	Financial intermediary development and stock market development are complements.
Rousseau and Wachtel (2000)	47 countries 1980-1995	Per-capita market capitalisation, per-capita value traded, market capitalisation/GDP, value traded/GDP, real per-capita M3	Real per-capita GDP growth	Causality runs from stock market indicators to economic activity.
Levine et al. (2000)	71 countries 1960-1995	Legal origin: English, French, German, liquid liabilities, commercial central bank, private credit	Real per-capita GDP growth, liquid liabilities, commercial central bank, private credit	Financial intermediary development exerts a positive, significant impact on economic (GDP) growth.
Levine (2001)	48 countries 1980-1995	Regulatory dummy, legal system, market capitalisation/GDP, total value traded ratio (dollar trading value/GDP), overhead costs, activity: (total value traded intermediary private credits)/GDP, size: (market capitalisation intermediary private credits)/GDP, efficiency: (total value traded/bank overhead cost ratio)	Real per-capita GDP growth	Financial services view is significant as opposed to models for only banking or only market based views. Activity, efficiency, and aggregate principal components have a positive and significant influence on real per-capita GDP growth.
Beck and Levine (2003)	40 countries 1976-1998	Turnover ratio, bank credit = claims on sector/GDP, schooling, govt. consumption, trade openness, inflation black market premium	Real per-capita GDP growth	Use system panel estimator (with 5-year averages and cross-country OLS) and GMM estimators: find turnover and bank credit are positive and significant. GMM difference estimator: neither bank credit nor turnover are significant. GMM level estimator: only turnover is positive and significant. The use of annual data only finds that ONLY turnover is significant and positive; bank credit is insignificant.
Zhu et al. (2004)	47 countries 1976-1993	Stock market liquidity (turnover), banking sector	Annual growth GDP/capita	Stock market liquidity is not statistically significant in explaining GDP growth in the presence of (Asian market) outliers.

Table M.1 Summary of Empirical Findings on Finance and Economic Growth ^a

Notice a: Further information relating to these empirical findings can be found under the author's name in the research reference.